

ETNAY



9610070159

Office of Hydropower Licensing

September 1996

FERC/FEIS-0075

FINAL ENVIRONMENTAL IMPACT STATEMENT

Upper Penobscot River Basin Maine Volume 1

Ripogenus Hydroelectric Project (FERC No. 2572) -005 Penobscot Mills Hydroelectric Projecter-DOCKETED (FERC No. 2458) - 619 OCT - 1 1996

FEDERAL ENERGY REGULATORY COMMISSION OFFICE OF HYDROPOWER LICENSING

FINAL ENVIRONMENTAL IMPACT STATEMENT

LICENSING TWO EXISTING HYDROELECTRIC PROJECTS IN THE UPPER PENOBSCOT RIVER BASIN

FERC Project Nos.

2572 Ripogenus 2458 Penobscot Mills

1

1

0-44

Applicant:

Great Northern Paper, Inc.

Additional copies of the FEIS are available from:

Federal Energy Regulatory Commission Public Reference and Files Maintenance Branch 888 First Street, N.E., Room 2A Washington, DC 20426

September 1996

FEDERAL ENERGY REGULATORY COMMISSION Washington, DC 20428

TO THE AGENCY OR INDIVIDUAL ADDRESSED

Attached is the Final Environmental Impact Statement (FEIS) for relicensing the Ripogenus (FERC No. 2572) and Penobscot Mills (FERC No. 2458) hydroelectric projects, Maine. The FEIS is contained in two volumes. Volume 1 includes the text of the FEIS through Appendix D. Volume 2 is Appendix E, comments on the DEIS and Commission staff's responses to those comment. This FEIS was prepared pursuant to requirements of the National Environmental Policy Act (NEPA) and the Commission's regulations implementing NEPA (18 CFR Part 380).

The FEIS documents the views of government agencies, nongovernmental organizations, affected Indian tribes, the public, the license applicant, and the Commission's staff. It contains staff's recommendations about licensing the Ripogenus and Penobscot Mills projects in the upper Penobscot River Basin.

Any Commission order issued pursuant to this document will be subject to the Commission's rehearing process under 18 CFR Section 185.713. Requests for rehearing must be filed within 30 days of the date of issuance of the subject order.

Before the Commission makes a decision on relicensing these projects, it will take into account all concerns relevant to the public interest. This FEIS will be part of the record from which the Commission will make its decision.

Attachment

COVER SHEET

- a. Title: New Licenses for Two Existing Projects in the Upper Penobscot River Basin, FERC Project Nos. 2572 and 2458
- b. Final Environmental Impact Statement
- c. Lead Agency: Federal Energy Regulatory Commission

d. Abstract: Great Northern Paper, Inc. (GNP) filed applications for new licenses (relicense) for two existing projects on the West Branch of the Penobscot River in Maine. GNP proposes changes in the operation and some minor construction to provide environmental and recreational enhancements requested by resource management agencies and other nongovernmental groups.

Primary resource issues are potential effects on (1) water quality and quantity, (2) fisheries resources, (3) shoreline preservation, and (4) socioeconomics. The Commission's staff also evaluated two alternatives to GNP's proposal.

The staff's preliminary recommendation is to relicense the projects as proposed with the additional enhancements prescribed in Alternative 2.

- e. Contact: Edward R. Meyer Federal Energy Regulatory Commission Office of Hydropower Licensing 888 First Street, N.E. Washington, DC 20426 (202) 208-7998
- f. This FEIS, prepared by the Commission's staff in connection with applications filed by GNP for FERC Projects Nos. 2572 and 2458 is being made available to the public on or about September 1996, as required by NEPA and the Commission's Regulations Implementing NEPA (18 CFR Part 380).

i

FOREWORD

The Federal Energy Regulatory Commission (FERC or Commission), pursuant to the Federal Power Act (FPA)¹ and the U.S. Department of Energy Organization Act² is authorized to issue licenses for terms up to 50 years for the construction and operation of nonfederal hydroelectric developments subject to its jurisdiction, on the necessary conditions:

(T)hat the project adopted...shall be such as in the judgement of the Commission will be best adapted to a comprehensive plan for improving or developing a waterway or waterways for the use or benefit of interstate or foreign commerce, for the improvement and utilization of water power development, for the adequate protection, mitigation, and enhancement of fish and wildlife (including related spawning grounds and habitat), and for other beneficial public uses, including irrigation, flood control, water supply, and recreational and other purposes referred to in section $4(e)...^3$

The Commission may require such other conditions not inconsistent with the provisions of the FPA as may be found necessary to provide for the various public interests to be served by the project.⁴ Compliance with such conditions during the license period is required. Section 385.206 (1987) of the Commission's Rules of Practice and Procedure allows any person objecting to a licensee's compliance or noncompliance with such conditions to file a complaint noting the basis for such objection for the Commission's consideration.⁵

Section 401(a)(1) of the Clean Water Act (CWA) requires an applicant for a federal license or permit for any activity that may result in a discharge into navigable waters of the United States to provide to the licensing or permitting agency a certification from the state in which the discharge originates that such discharge will comply with certain sections of the CWA. A state Water Quality Certificate (WQC), therefore, is a prerequisite for obtaining a Commission license. The Commission's past practice has been to include all state water quality conditions in any order issuing a project license; however, as stated in *Tunbridge Mill Corporation* (68 FERC ¶ 61,078, 1994), under Section 401(d), states may lawfully impose only conditions related to water quality. In examining the conditions proposed in the Penobscot Mills and Ripogenus WQCs in section 5 of this FEIS, we follow the principles discussed in *Tunbridge Mill*.

Commission staff is aware of <u>PUD No. 1 of Jefferson County v. Washington Dept.</u> of Ecology. (U.S. Sup. Ct. No. 92-1911, May 31, 1994). As appropriate, the license order in this proceeding will address the relevance of the issues discussed in <u>Jefferson County</u>.

² Public Law 95-91, 91 Stat. 556 (1977).

¹ 16 U.S.C. Sec. 791(a)-825(r), as amended by Electric Consumers Protection Act of 1986, Public Law 99-495 (1986).

³ 16 U.S.C. Sec. 803(a).

^{4 16} U.S.C. Sec. 803(g).

⁵ 18 CFR Sec. 385.206 (1987).

TABLE OF CONTENTS

.

.....

a constantino a constantino a constantino a constantino a

Ρ	8	a	e
•	-	22	-

VOLUME 1	
FOREWOR	D
LIST OF FI	GURES
LIST OF TA	ABLES
ACRONYM	S AND ABBREVIATIONS xvii
EXECUTIV	E SUMMARY
1.0 PUF 1.1 1.2 1.3	POSE AND NEED FOR ACTION 1-1 PURPOSE OF ACTION 1-1 NEED FOR POWER 1-1 SCOPE OF THE EIS 1-4 1.3.1 Introduction 1-4 1.3.2 Scoping 1-4 1.3.3 Cumulative impacts 1-4
2.0 PRC 2.1	POSED ACTION AND ALTERNATIVES 2-1 APPLICANT'S PROPOSAL 2-1 2.1.1 Description of Projects 2-1 2.1.2 Operation of Projects 2-7 2.1.3 Proposed Environmental Measures 2-12
2.2	STATUTORY REQUIREMENTS2-172.2.1Water Quality Certification Conditions2-172.2.2Section 18 Fishway Prescriptions2-21
2.3	PROJECT ALTERNATIVES 2-21 2.3.1 Alternative 1 2-22 2.3.2 Alternative 2 2-23 2.3.3 No-Action Alternative 2-23 2.3.4 Alternative Considered but Eliminated from Detailed Study 2-23
2.4	ECONOMIC COMPARISON OF ALTERNATIVES 2-24 2.4.1 Annual Power Value 2-25 2.4.2 Annual Project Cost 2-30 2.4.3 Annual Net Benefits 2-31 2.4.4 Cost of Power 2-33
3.0 AFFI 3.1 3.2 3.3	ECTED ENVIRONMENT 3-1 GENERAL SETTING 3-1 GEOLOGY AND SOILS 3-1 STREAMFLOW 3-4

	3.3.1	Regional	. 3-4
	3.3.2	Site-specific	. 3-5
3.4	WATER	QUALITY	. 3-5
	3.4.1	Regional	. 3-5
	3.4.2	Site-specific	. 3-8
3.5	FISHER	IES RESOURCES	3-12
	3.5.1	Regional	3-12
	3.5.2	Site-specific	3-17
3.6	WETLA	NDS	3-21
	3.6.1	Basinwide	3-21
	3.6.2	Ripogenus Project	3-21
	3.6.3	Penobscot Mills Project	3-23
3.7	TERRES	TRIAL RESOURCES	3-24
	3.7.1	Basinwide	3-24
	3.7.2	Ripogenus Project	3-25
	3.7.3	Penobscot Mills Project	3-27
3.8	THREA	TENED AND ENDANGERED SPECIES	3-29
	3.8.1	Basinwide	3-29
	3.8.2	Ripogenus Project	3-29
	3.8.3	Penobscot Mills Project	3-30
3.9	RIVER A	AND LAND MANAGEMENT PLANS	3-30
	3.9.1	Maine Rivers Study	3-30
	3.9.2	State of Maine Comprehensive Rivers Management Plan	3-30
	3.9.3	Statewide Comprehensive Outdoor Recreation Plan	3-31
	3.9.4	Maine Strategic Plans, Vol. II, Inland Fisheries	3-31
• • •	3.9.5	Maine Statewide River Fisheries Management Plan	3-31
3.10	RECREA	TION RESOURCES	3-31
	3.10.1	Regional	3-31
	3.10.2	National Designations	3-33
	3.10.3	Existing Recreation Facilities/Opportunities in the Project	0.00
	2 1 0 4		3-33
	3.10.4		3-34
	3.10.5	Access for Boople with Dischilities	3-40
2 1 1			3-41
3.11	2 1 1 1	De 103065	3-41
	2112		3-41
	3.11.2	Proposed Land Liege	3-48
3 1 2			3-40
5.12	2 1 2 1		2 40
	2 1 2 2		3-43
3 1 3			3-50
0.13	3 1 2 1	Regional	3-31
	3 13 2	Site-snecific	3-01
3,14	SOCIOE	CONOMIC RESOURCES	3-53
	3.14.1	Demographic Conditions	3-54
	0 1 4 0		0-54
	3.14.2	WUDICIDAL Services	

4.0	ENVI	RONME	NTAL IMPACTS	1
	4.1	GEOLO	OGY AND SOILS	
		4.1.1	Applicant's Proposal 4-1	I
		4.1.2	Alternatives 1 and 2 4-1	I
		4.1.3	No-action Alternative 4-1	l
	4.2	STREA	MFLOW	l
		4.2.1	Applicant's Proposal 4-2	2
		4.2.2	Alternative 1 4-4	ł
		4.2.3	Alternative 2 4-6	5
		4.2.4	No-action Alternative 4-7	1
		4.2.5	Summary 4-7	/
	4.3	WATEF	R QUALITY	\$
		4.3.1	Applicant's Proposal 4-8	\$
		4.3.2	Alternative 1 4-13	\$
		4.3.3	Alternative 2 4-14	ŀ
		4.3.4	No-action Alternative 4-15	,
	4.4	FISHER	IES RESOURCES 4-15)
		4.4.1	Applicant's Proposal 4-15	j
		4.4.2	Alternative 1	1
		4.4.3	Alternative 2 4-30)
		4.4.4	No-action Alternative 4-32	
		4.4.5	Summary 4-32	
	4.5	WETLA	NDS	1
		4.5.1	Applicant's Proposal 4-33	;
		4.5.2	Alternative 1 4-37	1
		4.5.3	Alternative 2 4-39	ł
		4.5.4	No-action Alternative 4-40)
	4.6	TERRES	STRIAL RESOURCES 4-41	
		4.6.1	Applicant's Proposal 4-41	
		4.6.2	Alternative 1 4-42	
		4.6.3	Alternative 2 4-42	
		4.6.4	No-action Alternative 4-42	
	4.7	THREA	TENED AND ENDANGERED SPECIES	
		4.7.1	Applicant's Proposal 4-43	
		4.7.2	Alternative 1 4-45	
		4.7.3	Alternative 2 4-45	
		4.7.4	No-action Alternative 4-45	
	4.8	RECREA	ATION RESOURCES 4-45	
		4.8.1	Applicant's Proposal 4-46	
		4.8.2	Alternative 1 4-52	
		4.8.3	Alternative 2 4-54	
		4.8.4	No-action Alternative 4-56	
	4.9 L	AND US	iΕ	
		4.9.1	Applicant's Proposal 4-57	
		4.9.2	Alternative 1 4-60	
		4.9.3	Alternative 2 4-61	
		4.9.4	No Action 4-64	

.

	4.10	AESTHETIC RESOURCES	4-64
		4.10.1 Applicant's Proposal	4-64
		4.10.2 Alternative 1	4-66
		4.10.3 Alternative 2	4-67
		4.10.4 No-action Alternative	4-67
	4.11	CULTURAL RESOURCES	4-68
		4.11.1 Applicant's Proposal	4-68
		4.11.2 Alternative 1	4-70
		4.11.3 Alternative 2	4-70
		4.11.4 No-action Alternative	4-70
	4.12	SOCIOECONOMIC RESOURCES	4-70
		4.12.1 Applicant's Proposal	4-71
		4.12.2 Alternative 1	4-72
		4.12.3 Alternative 2	4-74
		4.12.4 No-action Alternative	4-75
	4.13	UNAVOIDABLE ADVERSE IMPACTS	4-75
	4.14	IRREVERSIBLE AND IRRETRIEVABLE COMMITMENT OF RESOURCES .	4-75
	4.15	RELATIONSHIP BETWEEN SHORT-TERM USES AND LONG-TERM	
		PRODUCTIVITY	4-75
5.0	STAF	F'S CONCLUSIONS	. 5-1
0.0	5.1		5-1
	5.2	PROJECT ALTERNATIVES	5-2
	5.3	ENVIRONMENTAL COMPARISON OF ALTERNATIVES	. 5-2
		5.3.1 Streamflow	. 5-7
		5.3.2 Fisheries Enhancement	. 5-7
		5.3.3 Wetlands	. 5-9
		5.3.4 Land Use	5-10
		5.3.5 Economic and Socioeconomic Effects	5-12
	5.4	SUMMARY OF FINDINGS AND RECOMMENDATIONS	5-15
	5.5	STATUTORY REQUIREMENTS	5-16
		5.5.1 Water Quality Certificate Conditions	5-16
		5.5.2 Section 18 Prescriptions	5-19
	5.6	SECTION 10(J) RECOMMENDATIONS	5-19
		5.6.1 Fish and Wildlife Agency Recommendations	5-19
		5.6.2 Preliminary Determination of Inconsistency	5-21
		5.6.3 10(j) Meeting	5-22
	5.7	CONSISTENCY WITH COMPREHENSIVE PLANS AND OTHER	
		RESOURCE PLANS	5-24
6.0	LITER	ATURE CITED	. 6-1
70	1107 /		
7.0	131 (UF FREFAREND	. /-1
8.0	LIST	OF RECIPIENTS	. 8-1

APPENDICES

ļ

Α	DAMS IN THE PENOBSCOT RIVER BASIN	A-1
В	FISHERIES RESOURCES	B-1
С	WETLANDS AND TERRESTRIAL RESOURCES	C-1
D	GNP'S WATER-USE MODEL	D-1
E	RESPONSES TO COMMENTS ON THE DEIS (VOLUME 2)	E-1

.

LIST OF FIGURES

	raye
1-1	West Branch of the Penobscot River Basin showing the Ripogenus and Penobscot Mills projects 1-2
2-1	Ripogenus Project map
2-2	Penobscot Mills Project map 2-4
2-3	Ripogenus Project and Penobscot Mills Project normal pool elevations, storage capacities, maximum generation, flows, and proposed minimum flows
3-1	Existing dams within the Penobscot River Basin and project area
3-2	Existing dams on the West Branch of the Penobscot River
3-3	Annual flow duration curve at McKay station for the period 1970-1985 3-7
3-4	Annual flow duration curve at Dolby station for the period 1972-1985 3-7
3-5	Travel regions in Maine showing location of Penobscot Mills Project area
3-6	Trends in recreational use of GNP's West Branch District
3-7	Summary of summer activity within GNP's West Branch District
4-1	Total weighted usable area for salmon in Upper Gorge at various flows 4-17
4-2	Total weighted useable area for salmon and smallmouth bass in Millinocket Stream at various flows 4-20
4-3	Total weighted useable area for salmon in Back Channel at various flows 4-22
4-4	Wetlands enhancement sites evaluated for feasibility by GNP at the Ripogenus Project, 4-6 August 1992 4-36
4-5	Wetlands enhancement sites evaluated for feasibility by GNP at the Penobscot Mills Project, 4-6 August 1992 4-38
4-6	Whitewater boating experience versus flow for the West Branch 4-48

LIST OF TABLES

	Pa	ge
1-1	West Branch Penobscot River Basin hydroelectric projects evaluated in this FEIS	-1
2-1	Summary of facilities and operations 2	-8
2-2	West Branch Penobscot River Basin hydroelectric project water quality certification	17
2-3	Economic analysis parameters 2-:	26
2-4	Incremental average annual power production losses	27
2-5	Incremental annual power values by alternative	29
2-6	Capital cost by alternative	30
2-7	Incremental annual project costs 2-3	31
2-8	Annual net benefits by alternative 2-3	32
2-9	Cost of power by alternative 2-3	33
3-1	West Branch impoundments upstream of the Ripogenus and Penobscot Mills projects	-4
3-2	Impoundment characteristics of developments in the Ripogenus and Penobscot Mills projects	-6
3-3	Peak discharge of West Branch impoundments	-8
3-4	Major point sources on the Penobscot River	-9
3-5	Water quality classification of the Penobscot River Basin	0
3-6	State of Maine river and stream water quality standards for dissolved oxygen	1
3-7	Range and mean of mercury concentrations in sediments and fish tissues from project waters and selected Maine lakes	3
3-8	Recreation facilities summary 3-3	35

3-9	Summary of significance classifications of lakes in the Ripogenus and Penobscot Mills project areas	3-42
3-10	Summary of zoning classifications	3-44
3-11	Population and rate of growth for Penobscot and Piscataquis counties and Millinocket/E. Millinocket	3-55
3-12	Housing units in Penobscot and Piscataquis counties and Millinocket/ E. Millinocket, 1980-1990	3-55
3-13	Employment by industrial sector for Maine and Piscataquis and Penobscot counties, 1990	3-56
3-14	Labor force participation in Maine and Penobscot and Piscataquis counties, 1990-92	3-57
3-15	Per capita and household incomes for Maine, Penobscot and Piscataquis counties and Millinocket/E. Millinocket, 1980-1990	3-58
4-1	Water use under GNP's proposed water-use plan, Alternative 1, and Alterna- tive 2, for the Ripogenus and Penobscot Mills projects	4-3
4-2	Mean concentration of mercury in sediments from selected lakes	4-9
4-3	Average mercury concentrations found in fish and mussels from reservoirs and reference lakes during the summer of 1992	4-10
4-4	Spawning habitat for key impoundment species	4-23
4-5	Changes in elevation, surface area, and dates of minimum and maximum elevation in Ripogenus and North Twin impoundments for the project alternatives	1-24
4-6	Minimum flow schedule	1-46
5-1	Summary of enhancement measures in the Applicant's Proposal and alternative for the Ripogenus and Penobscot Mills projects	s 5-3
5-2	Summary comparison of environmental and economic consequences of the Applicant's Proposal and alternatives for the Ripogenus and Penobscot Mills projects	5-4
5-3	Incremental annual net benefits by alternative	5-13
5-4	Levelized net annual costs of the staff's final recommendation	5-17

5-5	Summary of the staff's conclusions about Maine 401 Water Quality Certificate conditions for the Ripogenus Project	5-18
5-6	Summary of the staff's conclusions about Maine 401 Water Quality Certificate conditions for the Millinocket Lake Storage dam of the Penobscot Mills Project	5-18
5-7	Summary of the staff's conclusions about Maine 401 Water Quality Certificate conditions for the North Twin, Millinocket, Dolby, and East Millinocket developments of the Penobscot Mills Project	5-19
5-8	Ripogenus Project: Summary of Interior's 10(j) recommendations and their associated costs	5-20
5-9	Penobscot Mills Project: Summary of Interior's 10(j) recommendations and their associated costs	5-21
5-10	Relative cost comparison for flows to the Back Channel	5-23

ACRONYMS AND ABBREVIATIONS

.

401 WQC	Section 401 Water Quality Certification
AC	acres
ACHP	Advisory Council for Historic Preservation
ADA	Americans with Disabilities Act of 1990
AF	acre teet
AMC	Appalachian Mountain Club
ASRSC	Atlantic Sea-Run Salmon Commission
AWA	American Whitewater Affiliation
bcf	billion cubic feet
BHE	Bangor Hydro-Electric Company
BOD5	five-day biological oxygen demand
BPR	Marine Bureau of Parks and Recreation
Btu/kWh	British thermal units per kilowatt hour
°C	Celsius degree
cfs	cubic feet per second
CI	Conservation Intervenors (American Rivers, American
	Whitewater Affiliation, Appalachian Mountain Club,
	Conservation Law Foundation, Maine Audubon Society)
COE	Department of the Army, Corps of Engineers
Commission or FERC	Federal Energy Regulatory Commission
CRMP	Cultural Resource Management Plan
D	Development District(s)
D-CL	Commercial Development District
D-GN	General Development District
D-RS	Residential Development District
DEIS	draft environmental impact statement
DEP	Maine Department of Environmental Protection
DIFW	Maine Department of Inland Fisheries and Wildlife
DMR	Maine Department of Marine Resources
DO	dissolved oxygen
DOC	Maine Department of Conservation
EIS	environmental impact statement
EMAP	Environmental Monitoring and Assessment Program
FOS	Energy Optimization System
FPA	Environmental Protection Agency
°F	Fahrenheit degree
FFIS	final environmental impact statement
FEMA	Federal Emergency Management Agency
FPA	Federal Power Act
FS	Department of Agriculture Forest Service
FWS	Department of the Interior Fish and Wildlife Service
GPA	Great Ponde water quality class A
GNP	Great Northern Paper Inc
bo	boreanower
uh in the second s	

Hz	hertz
IFIM	Instream Flow Incremental Methodology
ITM	International Toxics Monitoring program
ITS	Maine Interconnecting Trail System
kV	kilovolt
kW	kilowatt
lb	pound(s)
LURC	Maine Land Use Regulation Commission
Μ	General Management District(s)
MEPRO	Maine Professional River Outfitters
MFPA	Maine Forest Practices Act
mgd	million(s) of gallons per day
МНРС	Maine Historic Preservation Commission
MM Btu	million British thermal units
MSPO	Maine State Planning Office
MUN-IND	industrial zone subject to local zoning regulations
MUN-SL	shoreline zone subject to local zoning regulations
MW	megawatts
MWh	megawatt hour(s)
NERC	North American Electric Reliability Council
NPCC	Northeast Power Coordinating Council
NPS	National Park Service
NRPA	Natural Resources Protection Act
Р	Protection District(s)
P-AL	Accessible Lakes Protection District
P-GP	Great Pond Protection District
P-RP	Resource Plan Protection District
P-SL	Shoreline Plan Protection District
PA	Programmatic Agreement
PIN	Penobscot Indian Nation
PLD	Penobscot Log Driving Company
PMF	probable maximum flood
ppm	parts per million
SCORP	Maine Statewide Comprehensive Outdoor Recreation Plan
SD	scoping document
SDF	spillway design flood
SHPO	State Historic Preservation Officer
SMP	Shoreline Management Plan
SOSLA	Save Our Scenic Lakes Association
SPF	standard project flood
TSS	total suspended solids
TU	Trout Unlimited
µg/g	micrograms per gram
USGS	United States Geological Survey
WUP	water use plan

EXECUTIVE SUMMARY

This Final Environmental Impact Statement (FEIS) evaluates the potential sitespecific and cumulative environmental consequences, economic costs, and related benefits associated with the proposed changes in operation and minor construction at the Ripogenus and Penobscot Mills projects on the West Branch of the Penobscot River in Piscataquis and Penobscot counties, Maine. The Federal Energy Regulatory Commission (FERC or Commission) is considering the applications for relicensing submitted by Great Northern Paper, Inc. (GNP), in December 1991 for the Ripogenus (FERC No. 2572) and Penobscot Mills (FERC No. 2458) projects. GNP proposes actions to provide environmental and recreational enhancements requested by resource management agencies and other interested, nongovernmental parties during initial consultations for relicensing.

The developments evaluated in this relicensing process have existed for many years, in at least one case more than 100. GNP proposes no major construction or project modifications but seeks to operate the projects nearly as it has over the past 50 years. Several intervenors seek enhancements beyond those proposed by GNP to improve existing fisheries, recreational opportunities, and other environmental conditions. This document evaluates all environmental and economic issues associated with licensing the projects but focuses on two primary issues: whether GNP should be required to release flows substantially higher than it proposes into Upper Gorge, Millinocket Stream, and the Back Channel to enhance fisheries resources and recreational opportunities, and what measures should be taken to preserve the natural character of shorelines of project impoundments.

We evaluated three alternatives using the No-action Alternative as a baseline for comparison. The Applicant's Proposal includes several categories of enhancements such as minimum flows, recreational facilities, further studies, and strict schedules for impoundment draw-downs. Alternative 1 includes enhancements of fisheries and other measures similar to those requested by conservation intervenors and some agencies. Alternative 2 includes enhancement measures intermediate between those proposed by GNP and those in Alternative 1.

APPLICANT'S PROPOSAL

GNP proposes no new developments and would continue to use the projects to provide power for its paper mills in Millinocket and East Millinocket. GNP would continue to operate the Ripogenus, Millinocket Lake, and North Twin developments for water storage and the Millinocket, Dolby, and East Millinocket developments as run-of-river.

Minimum flows in the West Branch below McKay station would be increased to enhance fisheries and whitewater recreation. Minimum flow in Upper Gorge would be 100 cfs between July 1 and September 30, and leakage (about 12 cfs) during the rest of the year. Flows in the Back Channel would remain at leakage, except for spillage events similar to those that have occurred historically. Impoundment fluctuations on North Twin would be scheduled to minimize impacts on lake trout spawning, which would improve the likelihood of achieving DIFW's goal to establish a self-sustaining lake trout population in that impoundment. Millinocket Lake would have a draw-down limit of 10 feet (minimum elevation of 470) and minimum flow in Millinocket Stream would be 60 cfs from between May 1 and October 15. GNP proposes no changes in lake-level management on Ripogenus impoundment.

GNP proposes to stock 500 brook trout per year in Millinocket Stream to enhance a spring recreational fishery. The Holbrook Stream nursery area is proposed to increase the amount of nursery habitat along the West Branch to increase the self-sustaining stock of landlocked salmon. A wildlife management plan would be implemented in the Back Channel area. GNP also proposes several enhancements, including improving boat ramps, and constructing changing facilities for whitewater boaters and rafters and additional parking facilities throughout the project area.

In addition, GNP would participate in monitoring to investigate mercury concentrations, DO levels, lake trout reproduction, and recreational use.

During the DEIS comment period, GNP proposed to adopt a Memorandum of Understanding (MOU) with the state of Maine to contribute a 250-foot conservation easement along approximately 73 shoreline miles within the Ripogenus Project area, and a 500-foot conservation easement for approximately 5 shoreline miles outside the Ripogenus Project area. The proposed easements would be donated to the state of Maine for the term of the license and would not be incorporated into the project boundaries. GNP proposes no conservation easements for the Penobscot Mills Project area.

ALTERNATIVE 1

Minimum flows in Upper Gorge would be 100 cfs between July 1 and September 30, and 50 cfs during the remainder of the year. Millinocket Stream would receive yearround minimum flows of 60 cfs. Minimum flows in the West Branch would be nearly the same as in the Applicant's Proposal. The most significant difference between this alternative and the Applicant's Proposal is that year-round flows of 350 cfs would be released to the Back Channel.

This alternative also includes a 500-foot boundary expansion (i.e., 500-foot building setback with 250-foot vegetative buffer) around the impoundments at both projects and wetlands enhancements at four locations within the area of the two projects. Other operational and enhancement measures would be the same as those proposed by GNP.

ALTERNATIVE 2

Flows in Upper Gorge and the West Branch would be nearly the same as those under the Applicant's Proposal. Millinocket Stream would receive a minimum flow of 60 cfs year round. We examined the potential value and economic feasibility of 30 cfs in the Upper Gorge and several levels of flow in the Back Channel ranging up to a year-round flow of 165 cfs to define this alternative. Comparing potential fisheries benefits with costs indicated that flow levels beyond leakage are not justified given the minimal benefit to key fish species and the adverse economic effect on GNP; therefore, Alternative 2 includes only leakage and spillage flows in the Back Channel and only leakage from fall through spring in the Upper Gorge. Alternative 2 includes two options: (1) a 250-foot conservation easement outside the Ripogenus Project boundary for approximately 73 shoreline miles of GNP-owned property as defined under the terms of GNP's proposed MOU; or (2) a 200-foot expansion of the project boundary on GNP-owned lands for the Ripogenus Project area. For Penobscot Mills, the project boundary would be expanded 200 feet on GNP-owned land along the impoundments under both options. Within the proposed boundary expansion areas, existing structures would be grandfathered. This alternative also proposes wetlands enhancement at two sites in the Penobscot Mills Project area and one in the Ripogenus Project area. All other operational and enhancement measures are the same as those in the Applicant's Proposal.

ANALYSIS OF ENVIRONMENTAL CONSEQUENCES

We evaluated the environmental consequences of the Applicant's Proposal and three alternatives. During the scoping process and in our environmental analysis, we identified five resource areas that encompass the major issues in this licensing process: streamflow, fisheries, wetlands, land use and socioeconomics.

GNP developed a water-use model for the West Branch river basin and, based on model output, concluded that it could not provide flow releases of the magnitude sought for fisheries enhancement by conservation intervenors and others (as represented in Alternative 1), given other water management constraints (e.g., impoundment draw-down limits, legally required minimum flow at Millinocket). Conservation intervenors contend that GNP's model is not valid and that model results are unreliable. We evaluated the model and determined that it is appropriate for analyzing this issue. GNP provided extensive model output that supported the staff's conclusion that nearly all flow enhancements could be implemented in wet and average years and are also feasible under dry and worst-case conditions with some effects on other water use constraints. We concluded, therefore, that water availability is not a limiting factor in adopting any of the alternatives.

We concluded that enhancement measures proposed by GNP would substantially enhance the landlocked salmon stock in the West Branch below McKay station and contribute to establishing a lake trout stock in North Twin impoundment. Alternative 1 would provide significant enhancement of aquatic habitats in the Back Channel, Millinocket Stream, and Upper Gorge, but this enhancement would not substantially increase regional abundance of key fish species, in particular landlocked salmon, beyond the enhancement provided under the Applicant's Proposal. Under Alternative 1, the 350-cfs flow in Back Channel could produce a standing stock of several hundred legal-sized landlocked salmon; 60-cfs flows in Millinocket Stream would yield fewer salmon than in the Back Channel, and year-round Upper Gorge flows would yield virtually none. In Alternative 2, winter flows in Millinocket Stream beyond those proposed by GNP would substantially enhance salmon habitat in that stream, but the increase in regional availability of salmon would be small.

GNP proposes no enhancement of wetlands. Alternative 1 includes four wetland projects that would enhance about twice the acreage of wetlands believed to be negatively affected by current project operations. Alternative 2 includes three wetlands enhancement

projects with an acreage about equal to that affected by project operations for substantially less cost than the enhancements included in Alternative 1.

Land use issues relate to potential future uses of forested lands surrounding the projects. GNP proposes no change of present project boundaries, which are generally at the high water marks of project impoundments. Alternative 1 includes a 500-foot building setback with a 250-foot vegetative buffer around impoundments at both projects. We concluded that this expansion of project boundaries would preserve regional aesthetics and protect water quality and riparian and terrestrial habitats, but at a very significant cost (approximately \$24.6 million). Alternative 2 includes a 200-foot boundary expansion for GNP-owned lands for the Penobscot Mills Project area and 250-foot easements on GNP-owned lands for approximately 73 shoreline miles of the Ripogenus Project area. We concluded that the reduced expansion would be adequate to meet aesthetic, water quality, and wildlife protection objectives at no direct cost to GNP.

We evaluated GNP's claims that losing generation by providing high flows, such as those included in Alternative 1, would affect its business operations severely and could lead to substantial layoffs and adverse effects on the local and regional economy. We conclude that environmental enhancement alternatives that specify more than leakage flows in the Back Channel would adversely affect the economics of coated paper production at the Millinocket and East Millinocket mills and would result in relatively meager environmental benefits in the project region. Although requiring wetlands enhancements and building setbacks also would add significantly to GNP's relicensing costs, the levelized annual costs for these measures in Alternative 2 are substantially less than those associated with the Back Channel flows included in Alternative 1.

ECONOMIC AND FINANCIAL EVALUATION

Staff compared the costs of construction, lost generation, and cost of replacement power associated with the Applicant's Proposal and Alternatives 1 and 2. The pulp and paper products industry is highly competitive and energy intensive. GNP's competitive position depends on the availability of a reliable source of inexpensive electric power. Imposing additional production costs for replacement power would adversely affect the economic viability of the Millinocket and East Millinocket paper mills. GNP indicated that any changes of project operations that would add incremental costs beyond those associated with its proposed water use plan would threaten the long-term viability of the mills, thereby threatening the economic base of the towns.

GNP can produce hydroelectric power at no significant incremental cost to the company. In comparison, the pre-tax cost of obtaining an equivalent amount of replacement power from another power producer would be \$73.92 per MWh. Any reduction of power production for environmental enhancement, therefore, would result in additional costs for GNP. Except for the No-Action Alternative, all alternatives would reduce generation from the Ripogenus and Penobscot Mills projects by diverting flows to improve aquatic habitats and enhance wetland areas, recreation, and lake aesthetics. Alternatives that would release more than leakage flows through the Back Channel would have the greatest effect on power generation, reducing power production by 5.0 to 7.0

percent of anticipated 1994 power production from the two projects and by 5.4 to 8.2 percent of Penobscot Mills power generation. Alternatives that would release only leakage through the Back Channel would reduce power production by 3.1 percent from the two projects.

The annual cost of the alternatives ranges from \$0.877 million to \$4.49 million (1996 dollars). Annual costs of GNP's proposal (\$0.877 million) would be less than costs associated with all other alternatives. GNP's cost for replacement power under Alternative 1, which specifies a minimum flow of 350 cfs in the Back Channel, would be approximately \$3.21 million per year. The total annual cost of this alternative would be \$4.49 million. Alternative 2, which specifies only leakage flows in the Back Channel would cost GNP \$0.959 million annually, which is slightly higher than the cost associated with GNP's proposal. Staff would not expect the incremental cost increase associated with Alternative 2 to alter GNP's competitive position and thus have any significant socioeconomic consequences.

RECOMMENDED ACTION

Based our independent analysis pursuant to Sections 4(e), 10(a)(1), and 10(a)(2) of the FPA, we conclude that issuing licenses for the Ripogenus and Penobscot Mills projects, with the enhancement measures defined in Alternative 2 and other special license conditions, would permit the best comprehensive development of the West Branch of the Penobscot River for the following reasons:

(1) Alternative 2, which includes enhancement of flows in Millinocket Stream beyond those proposed by GNP but otherwise specifies flow releases and impoundmentlevel restrictions nearly identical to those included in the Applicant's Proposal, would substantially enhance existing fish stocks (particularly landlocked salmon) in the project region. Flows beyond those included in Alternative 2 in Upper Gorge, Millinocket Stream, and the Back Channel probably would not further enhance regional fish stocks.

(2) The minimal enhancement of regional fish stocks associated with higher flow releases specified in Alternative 1 and considered as options for defining Alternative 2 would not cause significant increases in fisheries resources or recreational angling in the region and, thus, would contribute little to the regional economy.

(3) High-quality recreational fishing waters are abundant in the project region; our estimates indicate that creating a limited amount of additional, year-round, riverine habitat in Upper Gorge and the Back Channel (Alternative 1), is not required to meet increasing recreational fishing demand, even if the additional habitat enhanced regional fish stocks.

(4) Flows for recreational boating negotiated between GNP and some whitewater boating groups and included in Alternative 2 offer some enhancement of existing recreational opportunities and would not adversely affect existing landlocked salmon populations in the West Branch.

(5) The wetlands enhancements (affecting about 280 acres) included in Alternative 2 are sufficient to enhance the estimated acreage of wetlands that are adversely affected

by impoundment draw-downs and would eliminate a long-standing adverse effect of the projects on the regional ecosystem.

(6) The proposed 250-foot conservation easements or the 200-foot boundary expansion on GNP-owned lands for the Ripogenus Project area, and the proposed 200-foot boundary expansion for GNP-owned land within the Penobscot Mills Project area (Alternative 2) would control potential future development and ensure protection of aesthetic and recreational resources within the project area. The proposed easements and boundary expansions would also protect against habitat modifications that could adversely affect terrestrial wildlife and vegetation in valuable riparian habitat around the impoundments.

(7) Continued limitation of nongeneration flow releases, particularly in the Back Channel, would ensure GNP's access to relatively inexpensive hydroelectric power necessary to maintain its paper production capability cost-effectively, thereby not placing greater economic stress on GNP and providing some protection for local municipalities and citizens against significant adverse effects on regional employment and socioeconomics.

(8) The fisheries and recreational enhancements that would result from the higher flows associated with Alternative 1 would be limited in magnitude, are not needed to meet recreational demand, and would have little effect on the local economy. In contrast, the wetlands enhancements, conservation easements, and project boundary expansion included in Alternative 2 would improve and protect habitat elements critical to the aesthetics and ecology of the project areas for the term of the project license. These habitats would otherwise be vulnerable to continued or future degradation. The enhancements included in Alternative 2 also are much less costly than boundary expansions and flow releases specified in Alternative 1.

(9) The 500-foot expansion of the project boundary around project impoundments under Alternative 1 would provide resource benefits only marginally greater than the proposed 250-foot conservation easements or 200-foot boundary expansion on GNPowned lands for the Ripogenus Project area and 200-foot project boundary expansion for the Penobscot Mills Project area under Alternative 2, at a much higher cost (estimated at \$24.6 million).

1.0 PURPOSE AND NEED FOR ACTION

1.1 PURPOSE OF ACTION

The Federal Energy Regulatory Commission (Commission or FERC), under authority of the Federal Power Act (FPA), may issue licenses for up to 50 years for construction, operation, and maintenance of nonfederal hydroelectric developments. The proposed action by the Commission is to determine if licenses should be renewed for two hydroelectric projects on the West Branch of the Penobscot River, in Piscataquis and Penobscot counties, Maine (table 1-1; figure 1-1). This final environmental impact statement (FEIS) was prepared as required by the National Environmental Policy Act (NEPA)¹ and FERC regulations to provide the Commission with descriptions and evaluations of the potentially significant environmental and developmental effects of the projects.

Table 1-1. West Bran FEIS (Sou	ble 1-1. West Branch Penobscot River Basin hydroelectric projects evaluated in this FEIS (Source: GNP, Staff)						
Project	Project Number	Installed Capacity (MW)					
Ripogenus	2572	37.5					
Penobscot Mills Millinocket Lake Sto Development North Twin Develop Millinocket Develop Dolby Development East Millinocket Dev	2458 prage pment ment t velopment	70.6 (total) 0.0* 7.0 35.8 20.9 6.9					
Total		108.1					
* Storage facility							

1.2 NEED FOR POWER

Great Northern Paper, Inc. (GNP) of Millinocket, Maine, applied for new licenses for two projects on the West Branch of the Penobscot River: (1) Ripogenus (FERC No. 2572), and (2) Penobscot Mills (FERC No. 2458). FERC issued the original licenses for the Ripogenus and Penobscot Mills projects on December 20, 1968, and April 1, 1962, respectively. GNP, a wholly owned subsidiary of Bowater Corporation, owns both projects. GNP's hydroelectric generating resources supply electric capacity and energy to

¹ P.L. 91-90, 42 U.S.C. 4341 (January 1, 1970), as amended by P.L. 94-52 (July 3, 1975) and P.L. 94-83 (August 9, 1975)



Figure 1-1. West Branch of the Penobscot River Basin showing the Ripogenus and Penobscot Mills projects (modified after Figure C-1, GNP 1991a; many lakes and tributaries have been omitted for clarity)

meet a portion of the company's requirements at its paper-making operations in the towns of Millinocket and East Millinocket, Maine.

The pulp and paper industry is highly competitive and energy intensive. Recent modernization of paper machines at East Millinocket indicates that the industry's efforts to improve paper quality and productivity have increased usage of electric energy in pulp and paper production. In addition, paper-making capacity exceeds the demand for product in many of GNP's markets. GNP's presence in certain markets and its competitive position in general, therefore, depend upon the availability of a reliable source of inexpensive electric power.

GNP has been involved in paper-making operations at Millinocket since 1900 and at East Millinocket since 1907 and has modernized the hydroelectric developments to meet the increased energy requirements of its pulp and paper operations. Because of the degree of integration between the Penobscot River developments and GNP's pulp- and paper-making operations, the Millinocket and East Millinocket mills (see figure 1-1) depend heavily upon hydroelectric energy produced by the Penobscot Mills and Ripogenus projects.

GNP supplies all of its electric power through a combination of hydroelectric/ hydromechanical, cogeneration, and condensing turbine capacity. In addition, the company has a tie-line with Bangor Hydro-Electric Company (BHE) to deliver up to 15 megawatts (MW) of interruptible, purchased power. The Ripogenus and Penobscot Mills projects were expected to supply approximately 37 percent of the company's sustainable generation capacity and nearly 50 percent of its energy needs for 1994. Cogeneration was expected to supply 27 percent of GNP's sustainable capacity and 36 percent of the energy in 1994, and condensing turbines were expected to supply peaking capacity and about 6 percent of energy. In addition, the Penobscot Mills and Ripogenus hydroelectric projects were expected to supply 77 percent of the 40-hertz (Hz) energy that would be consumed by the Millinocket and East Millinocket mills in 1994. Forty-hertz energy is a critical input to the pulp-making operations at these mills.

GNP has limited options for replacing lost hydroelectric capacity. Cogeneration currently satisfies a significant portion (36 percent) of GNP's energy demand. Cogeneration, however, depends upon the amount of steam generated for use in the manufacturing process, and the existing cogeneration system takes full advantage of available steam. Furthermore, recent modernization of the mill to reduce energy usage by conserving steam has reduced the amount of steam available, resulting in less cogeneration capacity. Cogeneration, therefore, has little potential for offsetting any reduction in GNP's other power sources. Section 2.4.2 discusses the feasibility of using alternative power sources, primarily from a cost perspective.

The lengthy period of GNP's electric-supply service for the Millinocket and East Millinocket mills has established an adequate foundation for both the short-term and long-term needs for electric power equivalent to the combined outputs of the projects.

1.3 SCOPE OF THE EIS

1.3.1 Introduction

On April 20, 1993, FERC issued notice that it would prepare an environmental impact statement (EIS) for the Ripogenus and Penobscot Mills projects. Scoping Document 1 (SD1) was issued in August 1993 to provide information about the project to the public and resource agencies and to solicit comments on the scope of the EIS.

1.3.2 Scoping

We conducted two scoping meetings on August 25, 1993. All interested individuals, organizations, and agencies were invited to attend both meetings to help identify environmental issues that should be analyzed in the EIS. Daytime and evening meetings were held at Stearns High School, Millinocket, Maine. The staff compiled notes on the daytime meeting, and a court reporter recorded the evening meeting. The Commission placed all statements (oral and written) in the public record for the project.

Following the scoping meetings and comment period, FERC reviewed and revised SD1; Scoping Document 2 (SD2) was distributed to all interested parties in November 1993. The content of the DEIS was based on comments obtained through the scoping process . Principal modifications of the scope include adding a resource-oriented alternative that includes larger flow enhancements than the applicant proposed, and including socioeconomics as a major issue. Issues addressed in this FEIS include (1) geology and soils, (2) streamflow, (3) water quality, (4) fishery resources, (5) wetlands, (6) terrestrial resources, (7) threatened and endangered species, (8) recreation resources, (9) land uses, (10) aesthetic resources, (11) cultural resources, and (12) socioeconomic resources. No new issues were identified by respondents to the DEIS.

1.3.3 Cumulative Impacts

To make sound licensing decisions the Commission analyzes both site-specific and cumulative impacts of hydropower development. Cumulative impacts are the additive or interactive impacts on resources caused by multiple developments within a river basin. Individually small or other seemingly minor impacts of past, present, and reasonably foreseeable future actions taken by several agencies or persons, when added together in space and time, may result in combined or cumulative impacts that have serious environmental consequences.

This FEIS addresses the potential for interaction between the effects of the projects and those of other developments in the river basin for appropriate resources. The FEIS identifies and describes the status of resources within the geographic area selected for the analysis and assesses the potential for the projects to contribute to cumulative impacts. The effects of past and present activities are reflected in the existing environment and provide a context for determining the potential for cumulative impacts.

During scoping, agencies and individuals questioned whether or not limiting the scope of the cumulative environmental analysis of the Penobscot Mills and Ripogenus projects to the project boundaries would allow adequate consideration of potential regional cumulative impacts, including the entire West Branch of the Penobscot River Basin. Participants in the scoping process also suggested that the cumulative assessment should encompass several West Branch developments upstream of the Ripogenus Project.

The staff offered the following clarification and revisions during final scoping and we have revised them slightly in response to comments on the DEIS.

- (1) Each resource evaluated in the Upper Penobscot River EIS will be evaluated cumulatively over the appropriate geographic range for that resource. For example, the Commission will assess the impacts on landlocked Atlantic salmon over the potential range of what is considered to be the West Branch stock of that species.
- (2) This FEIS does not encompass other projects within the Penobscot River Basin that are presently in the relicensing process, specifically the lower Penobscot River projects (Basin Mills, FERC Project No. 10981; Milford, FERC Project No. 2534; and Stillwater, FERC Project No. 2712). A separate multiple-project FEIS will be prepared for those projects, which are approximately 70 river miles downstream of Penobscot Mills. The Commission did not combine these projects into a single, basinwide EIS for these reasons:
 - The resources of concern differ substantially. The major environmental concern at the Basin Mills/lower Penobscot projects is restoring anadromous fish, particularly Atlantic salmon. At the upper Penobscot River projects (Ripogenus/Penobscot Mills), the major concerns are potential economic impacts on the licensee and local municipalities, water use and allocation, mercury contamination, enhancement of landlocked Atlantic salmon populations, recreation access, and whitewater boating.
 - Key resources in the upper and lower Penobscot are generally exclusive of each other. There are no anadromous Atlantic salmon in the vicinity of Ripogenus and Penobscot Mills, and Penobscot River restoration plans do not include any restoration activities in the West Branch. Most wetlands within the basin are confined to the vicinity of Ripogenus and Penobscot Mills and are not contiguous with the isolated wetland areas in the vicinity of the lower Penobscot projects. Commercial and individual whitewater boating occurs almost exclusively in the upper portions of the river basin.
- (3) Section 4 of the FEIS considers the need to expand the project boundaries or provide conservation easements to account for watershed activities that may influence project operations and environmental effects associated with those operations.

2.0 PROPOSED ACTION AND ALTERNATIVES

2.1 APPLICANT'S PROPOSAL

GNP proposes changes in operation and some minor construction at the Ripogenus and Penobscot Mills projects. These changes would provide environmental and recreational enhancements requested by resource management agencies and other interested nongovernmental groups during initial consultation for relicensing. This section describes the projects and GNP's proposed enhancements.

2.1.1 Description of Projects

2.1.1.1 Ripogenus Project

The principal features of the Ripogenus Project (figure 2-1) are an impoundment (Ripogenus Lake), a dam, a powerhouse, a bypass reach of about 3,900 feet in the Upper Gorge area, and appurtenant facilities. In its existing condition, the project has a total nameplate generator capacity of 37.5 MW at 40 Hz and an average annual generation of about 234,000 megawatt-hours (MWh). The dependable capacity is 22.4 MW based on a project flow of 1,800 cubic feet per second (cfs), with an 85 percent exceedance value and a net head of 173 feet.

The project includes:

- a 795-foot-long concrete gravity dam, consisting of a 480-foot-long ogee spillway section with a crest elevation of 929.6 feet National Geodetic Vertical Datum (NGVD) at a maximum height of 83 feet; topped with 22 stop-log gates, each about 17 feet wide by 11 feet high; and 2 crest gates, each about 17 feet wide by 11 feet high;
- a 37-foot-wide tunnel intake section with a 16-foot-diameter, concrete-lined tunnel about 3,850 feet long; a 44-foot-diameter by 104-foot-high surge tank that rises about 54 feet above grade; and three concrete-lined, steel penstocks, each 10 feet in diameter and ranging about 100 to 136 feet in length, all protected by trashracks of 3 by %-inch steel bars with 2%-inch openings;
- a 179-foot-long gate section with 4 deep waste gates, each about 9 feet high by 8 feet wide; and 2 timber gates, each about 14 feet high by 6 feet wide;
- a 100-foot-long earth embankment with a crest elevation of 942.6 feet;
- a concrete, steel, and brick powerhouse that is about 76 feet high by 45 feet wide by 130 feet long, is equipped with three vertical-shaft generating units with a total rated capacity of 37,530 kilowatts (kW), and has a hydraulic capacity of 3,500 cfs and a designed head ranging from 165 to 175 feet;



Figure 2-1. Ripogenus Project map (modified after figure G-3; GNP, 1991a)

- a 20.8-mile-long impoundment (consisting of Ripogenus Lake, Caribou Lake, Chesuncook Lake, Moose Pond, Brandy Pond, and Black Pond) with a surface area of about 29,270 acres (AC), a gross storage capacity of 710,000 acre-feet (AF), a usable storage capacity of 688,705 AF, a normal pool headwater elevation of 941.6 feet, and tailwater elevation of 758.5 feet;
- a 30.2-mile-long, 115-kilovolt (kV) transmission line; and
- appurtenant facilities.

2.1.1.2 Penobscot Mills Project

The Penobscot Mills Project (figure 2-2) consists of four discrete generating facilities and one storage/pump station development. Beginning with the most upstream, the developments are the Millinocket Lake Storage Development (which is located northeast of the North Twin Development and contains a pumping station), North Twin, Millinocket, Dolby, and East Millinocket.

Penobscot Mills has a total installed capacity of 70.6 MW and an average annual generation of about 386,400 MWh. The dependable capacity is 36.4 MW based on a project flow of 2,800 cfs with an 85 percent exceedance value. In 1994, GNP shut down the pulpwood grinding lines (letter from B. Stetson, GNP, September 2, 1994). GNP stated that the five hydromechanical units at the Millinocket development, which are equipped with synchronous motors, will continue to generate electric power to support increased production at the East Millinocket mill.

The principal features of the existing Penobscot Mills Project are five dams, five impoundments, four powerhouses, and appurtenant facilities.

Millinocket Lake Storage Development. This development, which is separate and distinct from the Millinocket Development, is used strictly for storage. Water is either released through the dam and down the Millinocket Stream or pumped through a pumping station into the North Twin impoundment to increase generation at the North Twin and Millinocket developments. This development has no hydroelectric-generating facilities. It consists of:

- a 635-foot-long concrete and earth-filled dam, with a 462-foot-long earthen embankment that has a crest elevation ranging from 485.6 feet to 487.0 feet; two spillway sections, totaling about 115 feet with a crest elevation of 480.0 feet; separated by a 58-foot-long intake section with four lift gates that are 8 feet wide by 9 feet high and a log sluice gate that is 8 feet wide by 10 feet high;
- a concrete, steel, and brick pumping station that is about 25 feet wide by 53 feet long, equipped with 2, vertical, wet-pit pumps, each with a capacity of 122 cfs protected by trashracks of 3/16-inch steel bars with 1-inch openings; driven by 2 induction motors, each with a capacity of 250 hp; discharging into 2 underground 4.5-foot-diameter pipes; about 544 feet long; that lead to the outlet structure at North Twin impoundment, which has 2 steel gates about 6 feet high by 6 feet wide;



Figure 2-2. Penobscot Mills Project map (modified after figure G-5; GNP, 1991b)

- an impoundment known as Millinocket Lake with a surface area of about 8,640 AC, a usable storage capacity of 45,370 AF, and a normal maximum pool headwater elevation of 480.0 feet; and
- an appurtenant facility.

North Twin Development. North Twin consists of:

- a 1,051-foot-long concrete earth-filled gravity dam, with a maximum height of 35 feet consisting of two earth wings with concrete core walls totaling about 500 feet long, of which 309 feet is topped with a paved roadway, and 100 feet is topped with a parapet wall with crest elevations that vary from 498.60 feet to 494.62 feet;
- a 34-foot-long concrete weir fishway with two deep-gated log sluice sections;
- a 114-foot-long by 37-foot-wide intake section with trashracks of %-inch steel bars with 2%-inch openings;
- a 117-foot-long concrete spillway with two Taintor gates, each 27 feet high by 50 feet wide, with an invert elevation of 464.62 feet; and 6 auxiliary earth dikes totaling about 2,530 feet long;
- a concrete, steel, and brick powerhouse that is integral to the dam and about 50 feet wide by 114 feet long, equipped with 2 vertical Francis turbine/generator units and 1 vertical Kaplan turbine/generator unit with a total rated capacity of 6,972 kW; a total hydraulic capacity of 4,500 cfs, a net head of 28 feet, and an average annual generation of 47,300 MWh; and a tailrace of six bays, each measuring 14 feet wide, and bordered by a 28-foot-long concrete retaining wall;
- an impoundment (consisting of Elbow Lake, North Twin Lake, South Twin Lake, Pemadumcook Lake, and Ambajejus Lake) that is about 11.8 miles long with a surface area of about 17,790 AC, gross storage capacity of 346,000 AF, a usable storage capacity of 344,355 AF, a normal pool headwater elevation of 491.92 feet, and tailwater elevation of 460.7 feet;
- a 4.2-mile-long, 34.5 kV transmission line; and
- appurtenant facilities.

Millinocket Development. Millinocket, which is separate and distinct from Millinocket Lake Storage Development, consists of:

 a 1,262-foot-long concrete gravity and stone dam, at the outlet of Quakish Lake, with a maximum height of 27 feet, which consists of a concrete gravity overflow section about 300 feet long with a crest elevation of 458.95 feet; 2 concrete gravity sections totaling about 786 feet long with a crest elevation of 456.20 feet, topped with 30-inch-high flashboards, and separated by a 52-foot-long wastegate structure with 4 steel gates; 8 auxiliary earth dikes totaling about 5,769 feet long; and a 124foot-long headgate section, with 10 steel gates, each about 8 feet high by 11 feet wide; and a sluiceway about 10 feet high by 12 feet wide;

- an intake section extending from the headgates located at the outlet of Quakish Lake through Ferguson Pond to the intake structure at Ferguson Pond outlet, consisting of a canal section that is about 150 feet wide by 1,400 feet long, separated from the Back Channel by a concrete gravity section with a crest elevation of 458.2 feet, topped with 6-inch flashboards; a concrete and wood frame intake structure with 6 gates that each measure 12.5 feet wide by 12.5 feet high, which control the flow into 6, 10-foot-diameter penstocks that are 1,007 to 1,024 feet long, lead to six units in the Grinder Room, and are protected by trashracks of %-inch steel bars with 2%-inch openings; and one gate, measuring 13.5 feet wide by 13.5 feet high that controls the flow into an 11-foot-diameter, 1,024-foot-long penstock that leads to two units in the generator room and is protected by a trashrack of %-inch steel bars with 2½-inch openings;
- a concrete, steel, and brick powerhouse, about 52 feet wide by 112 feet long, equipped with eight horizontal Francis turbine/generator units with a total installed capacity of 35,782 kW, a hydraulic capacity of 5,000 cfs, a net head of 108 feet, and an average annual generation of 203,300 MWh;
- a tailrace of 7 bays, each measuring 14 feet wide;
- an impoundment (consisting of Quakish Lake and Ferguson Pond) with a surface area of about 1,344 AC, a gross storage capacity of 8,100 AF, negligible usable storage capacity, a normal pool headwater elevation of 458.7 feet, and tailwater elevation of 347.4 feet;
- a 300-foot-long, 34.5 kV, transmission line; and
- appurtenant facilities.

Dolby Development. Dolby consists of:

- a 1,395-foot-long concrete gravity and earth-filled dam with a maximum height of 66 feet that consists of a 521-foot-long concrete gravity spillway section with a crest elevation of 332.2 feet, topped with 4-foot flashboards, separated by a 76-foot waste gate structure with 6 gates, each about 6 feet by 9 feet, and by a 34-foot log sluice section; an earthen dike with core walls about 550 feet long topped with a 12-foot-wide travel path; and a 209-foot-long headgate section, with 9 gates, protected by 3 sets of trashracks of %-inch steel bars with 1%-inch openings and 4 sets of trashracks of %-inch steel bars with 2%-inch openings;
- a concrete, steel, and brick powerhouse, about 115 feet wide by 167 feet long, and an addition that is 82 feet wide by 36 feet long, equipped with 3 horizontal Francis turbine/generator units, 3 inclined turbine/generator units, and 1 vertical Kaplan turbine/generator unit, with a total rated capacity of 20,886 kW, a hydraulic

capacity of 6,000 cfs, a net head of 49 feet, and an average annual generation of 98,100 MWh;

- a tailrace with 8 discharge bays;
- A 2.3-mile impoundment (known as Dolby Pond) with a surface area of about 2,048 AC, a gross storage capacity of 41,956 AF, negligible usable storage capacity, and a normal pool headwater elevation of 338.2 feet and tailwater elevation of 287.2 feet;
- a 2-mile-long, 34.5-kV, 60-Hz transmission line and a 6.8-mile-long, 33.0- to 34.5kV, 40-Hz transmission line; and
- appurtenant facilities.

East Millinocket Development. East Millinocket consists of:

- a 571-foot-long concrete and earth-filled gravity dam, with a maximum height of 28 feet that consists of a 116-foot-long earth embankment with a tip elevation of 295.2 feet; a 300-foot-long concrete gravity spillway section with a crest elevation of 283.2 feet that is topped with 4-foot flashboards separated by a 59-foot-long waste gate structure with 2 gates, each about 23 feet wide; a 7-foot-long timber crib section; and a 146-foot-long intake section with 12 gates, about 9 feet high by 11 feet wide, protected by trashracks of %-inch steel bars with 1¼-inch openings;
- a concrete, steel, and brick powerhouse, about 56 feet wide by 147 feet long, equipped with 6 horizontal Francis turbine/generator units with a total rated capacity of 6,936 kW at 60 Hz, a hydraulic capacity of 4,200 cfs, a net head of 24 feet, and an average annual generation of 37,700 MWh;
- a tailrace that is about 1,050 feet long by 110 feet wide, with 6 discharge bays;
- an impoundment (consisting of a 1.9-mile stretch of the West Branch of the Penobscot River) with a surface area of about 128 AC, a gross storage capacity of 1,950 AF, negligible usable storage capacity, normal pool headwater elevation of 287.2 feet, and tailwater elevation of 261.5 feet; and
- appurtenant facilities.

Table 2-1 summarizes project facilities for the developments of the Ripogenus and Penobscot Mills projects.

2.1.2 Operation of Projects

The Ripogenus Project operates as a storage development on an annual basis, whereas the Penobscot Mills developments operate in different modes. Millinocket Lake operates as a storage development where water is pumped into the North Twin impoundment. GNP operates North Twin as a storage development on an annual basis. Millinocket, Dolby,

Table 2-1. Summary of facilities and operations (Source: GNP, Staff)							
	Ripogenus	Millinocket Lake	North Twin	Millinocket	Dolby	East Millinocket	
Impoundment Surface Area (acres)	29,270	8,640	17,790	1,344	2,048	128	
Normal Headwater Elevation (ft)	941.6	480.0	491.9	458.7	336.2	287.2	
Normal Tailwater Elevation (ft)	758.5		460.7	347.4	287.2	261.5	
Draw-down Capability (ft)	44.0	6.2	22.0	••			
Gross Storage (acre-ft)	710,000	45,370	346,000	8,100	41,956	1,950	
Usable Storage (acre-ft)	688,705	45,370	344,355				
Hydraulic Capacity (cfs)	3,500		4,500	5,000	6,000	4,200	
Installed Capacity (KW)	37,530	••	6,972	35,782	20,886	6,936	
Annual Energy (MWh)	234,000		47,300	203,300	98,100	37,700	

and East Millinocket operate as run-of-river (outflow equals inflow) developments, with a small flow-reregulating capability.

The ability to manage flow from the storage impoundments at the Ripogenus and Penobscot Mills projects allows GNP to maximize the combined annual energy generation from these projects by generally keeping the flow of water within each development's hydraulic capacity. Figure 2-3 shows the relationships between the projects and their developments. Storage is operated on an annual cycle to dampen the extremes of seasonal variation in runoff by storing the majority of spring runoff to provide sustained flows during the remainder of the year, when precipitation is significantly less. Water is released at a controlled rate during the entire year according to a system-rule curve (based on providing an average daily flow of 2,400 cfs at Millinocket with a 1 percent probability of running out of storage) that produces a relatively even flow in the West Branch throughout the year. This minimizes spillage at the dams, which can occur when flows exceed a facility's hydraulic capacity. Flow is reserved for later use to provide maximum sustained baseload energy generation and protect populated areas downstream from flooding during the high spring runoff. The total volume of usable storage in the West Branch system is 1,347,600 AF.



- PROTOCOLOGICA - PROTOCOLOGICA

encontraction (Contraction (Con

Figure 2-3. Ripogenus Project and Penobscot Mills Project normal pool elevations (in feet NGVD), storage capacities (in billion cubic feet or bcf), maximum generation, flows and capacities, and proposed minimum flows (in cfs; Source: GNP, 1991a, 1991b)
The Ripogenus and Penobscot Mills projects are part of an integrated system that GNP developed to provide baseload electrical and mechanical power for the Millinocket and East Millinocket mills. GNP monitors the energy demand of the manufacturing facilities through a computerized dispatch system known as the Energy Optimization System (EOS) and operates the storage and generation facilities of the system to meet that demand efficiently. The total generating capacity of the Penobscot Mills/Ripogenus projects is 108,106 kW; 37,836 kW of this power is generated at 60 Hz, and 70,270 kW is generated at 40 Hz. The 60-Hz component of GNP's load is supplied from hydroelectric generation, cogeneration, condensing, and purchased power. North Twin, East Millinocket, and portions of Millinocket and Dolby generate at 60 Hz. The 40-Hz network supplies power to the large wood grinder motors at East Millinocket mill and is supplied by the Ripogenus Project and portions of the Millinocket and Dolby facilities. Forty-hertz power cannot be purchased from, or sold to, any other public utilities because they all use a 60-Hz network. Overall, GNP supplies about 90 percent of the energy needed for its mills and purchases the remainder from BHE. Occasionally, GNP sells a small amount of 60-Hz power to BHE.

The EOS dispatch center in Millinocket maintains remote monitoring and control of each development. This real-time, computerized monitoring and control system allows the dispatcher to receive information on impoundment levels, discharge, and unit settings for all the projects and power demands at the manufacturing facilities; select the best units and settings for maximum system efficiency; and adjust unit flows and settings accordingly. The dispatcher can meet instantaneous demands, optimize performance, and maximize annual energy production of the power system.

2.1.2.1 Ripogenus Project

The Ripogenus impoundment provides 51 percent of the total volume of usable storage in the West Branch system. During high spring flows, GNP manages outflow and limits spillage by exploiting the usable storage capacity of the impoundment. When the impoundment is at its normal maximum operating level, GNP opens the deep waste gates and crest gates. With all gates open, the dam can pass the probable maximum flood (PMF) of 65,800 cfs.

By storing spring flows, GNP can release a more consistent flow to the West Branch throughout the remainder of the year. Although the Ripogenus impoundment has a potential draw-down of 44 feet, draw-down has not exceeded 30.6 feet since 1970 and has averaged 16.5 feet. Over the same period, flow at McKay station ranged from 200 cfs to 15,185 cfs, averaging 2,699 cfs.

The project's bypass reach, the 4,730-foot-long Upper Gorge between the Ripogenus dam and McKay station, currently receives only leakage flows estimated at 12 cfs and spillage, which occurs approximately 11 percent of the time.

2.1.2.2 Penobscot Mills Project

The Penobscot Mills Project consists of five developments. The North Twin impoundment is at the upstream end of the project and receives flow from the Ripogenus Project. Releases from North Twin pass through the four generation facilities of the nobscot Mills Project: North Twin, Millinocket, Dolby, and East Millinocket. Water from nocket Lake is either pumped to North Twin or discharged through the dam to Millinocket am, which enters the West Branch immediately downstream of the Millinocket elopment. The total volume of usable storage in the Penobscot Mills Project is 389,725 which is 29 percent of the West Branch total.

North Twin Development. The North Twin Development operates in a storage mode annual basis. During high flows, GNP manages outflow and limits spillage by exploiting ailable usable storage capacity of the impoundment. When the impoundment reaches mal maximum operating level, GNP opens the Taintor gates. The development has a num discharge capacity of 72,000 cfs and can pass the PMF of 64,300 cfs.

The North Twin impoundment has a potential draw-down of 22 feet; however, annual down has ranged only from 6.1 feet to 17.6 feet since 1970. GNP releases water from um storage to maximize generation at North Twin by maintaining a higher headpond on.

FERC staff estimates that the minimum river flow at North Twin is 1,980 cfs (during mill wn); the mean is 3,543 cfs, and the maximum is 32,008 cfs. Spillage occurs nately 20 percent of the time.

Millinocket Lake Storage Development. This development is used principally to plement flow for power generation through the North Twin and Millinocket developments. this mode of operation, water is pumped from Millinocket Lake into Ambajejus Lake (part of the North Twin impoundment) at a rate of up to 244 cfs. Alternatively, water can be spilled over the dam at Millinocket Lake into Millinocket Stream, which enters the West Branch downstream of the Millinocket Development. Water that is spilled from the lake, therefore, is available for generation only at the Dolby and East Millinocket developments, whereas water pumped into Ambajejus Lake is available at all four generation facilities. Likewise, a minimum flow of 20 cfs and any other spillage released at the dam to Millinocket Stream is unavailable for power generation at the North Twin and Millinocket developments. Millinocket Lake has a potential draw-down of 6.2 feet, and the dam at the lake can withstand the PMF of 4,400 cfs.

Millinocket Development. The Millinocket impoundment has negligible storage capacity; therefore, the development is operated in run-of-river mode, with minor fluctuations in water level based on inflow from the North Twin Development. These fluctuations are typically caused by varying demand at the manufacturing facilities for either 40-Hz or 60-Hz electrical power. The average daily variation between North Twin and Millinocket outflows is 130 cfs.

FERC staff estimates river flows at Millinocket for the period of record are a minimum of 1,980 cfs (during a mill shut-down), a mean of 3,568 cfs, and a maximum of 32,230 cfs. Spillage occurs approximately 12 percent of the time when flows exceed the facility's hydraulic capacity.

The flashboards on the spillway of Stone dam will fail when overtopped by 1.5 feet (at 460.2 feet elevation), thereby spilling flows to Back Channel. The dam can pass 109,000 cfs

with the water level at the deck of the gatehouse (at 464.2 feet elevation), and the dam and dikes can withstand the PMF of 64,800 cfs.

Dolby Development. The Dolby impoundment has negligible storage capacity; therefore, the Dolby Development is operated in run-of-river mode, with minor fluctuations in water level based on inflow from the Millinocket Development. As described above for Millinocket, these fluctuations are typically the result of load variations at the manufacturing facilities. Operation is also a function of flow from the Millinocket Lake Storage Development by way of Millinocket Stream.

During the period of record of GNP's flow analysis, the mean river flow at the Dolby Development was 3,979 cfs; the minimum river flow was 1,837 cfs (during a mill shut- dow and the maximum river flow was 35,947 cfs. Spillage occurs approximately 5 percent of the time when river flow exceeds the facility's hydraulic capacity.

GNP opens the waste gates, which have a combined flow capacity of 3,600 cfs, du high flows to maintain the flashboards in place. The flashboards on the spillway will fail w overtopped by 1.5 feet. The development structures can withstand the PMF of 64,000 cfs the waste gates open and the flashboards removed.

East Millinocket Development. The East Millinocket impoundment has negligib' storage capacity; therefore, the development is operated in run-of-river mode, with minor fluctuations in water level based on inflow from the Dolby Development. These fluctuations, are typically caused by load variations at the manufacturing facilities.

FERC staff estimates that the minimum river flow at East Millinocket for the period of record was 1,840 cfs (during a mill shut-down); the mean was 3,985 cfs, and the maximum was 35,998 cfs. Spillage occurs approximately 28 percent of the time when river flow exceeds the facility's hydraulic capacity.

GNP opens the waste gates, which have a combined flow capacity of 2,000 cfs, to maintain the flashboards in place. The flashboards on the spillway will fail when overtopped by 1.5 feet (at an elevation of 288.7 feet). The development structures can withstand the PMF of 64,000 cfs (at an elevation of 293.1 feet).

2.1.3 Proposed Environmental Measures

Although FERC licenses the two projects evaluated in this FEIS separately, many of the resources within the two project areas are interdependent, particularly resources associated with streamflow and water quality; therefore, associating GNP's proposed enhancements specifically with one project or the other is not always accurate or appropriate. We describe GNP's proposed enhancements according to the integrated nature of the resources they would affect, indicating the specific project area to be enhanced where appropriate.

2.1.3.1 Geology and Soils

GNP intends to develop an erosion and sedimentation control plan that will specify control measures and procedures to limit erosion and sedimentation during construction of enhancements of recreation access in the Ripogenus Project area. GNP proposes no construction and, therefore, no mitigation for geology and soils/groundwater in the Penobscot Mills Project area.

2.1.3.2 Streamflow

Figure 2-3 illustrates project storage capacities, maximum generation plans, and GNP's proposed minimum flows. GNP proposes to supply flows of 100 cfs in Upper Gorge between July 1 and September 30 and leakage flows of approximately 12 cfs during the rest of the year. Supplemental summer "attraction flows" of 100 cfs would be provided to attract adult salmon into Upper Gorge in an effort to increase angling opportunities. Timing these releases would be coordinated with state and federal fisheries agencies. Below McKay station, GNP proposes the following flows (as modified to reflect the agreement between GNP and Maine Professional River Outfitters on recreational boating flows):

- outage flows of 400 cfs;
- rafting flows of 1,800 to 2,300 cfs between May 1 and September 15, depending on day of week and water availability (see figure 2-3 for details);
- rafting flows of 2,300 cfs (normal years) or 2,200 cfs (wet or dry years) on Saturdays and Sundays between September 16 and October 1;
- spawning flows of at least 1,300 cfs between October 15 and November 15;
- incubation flows equal to or greater than spawning flows from November 16 through June 7; and
- rafting or incubation flows, whichever is greater, continuously from May 1 through June 7.

Flows released to Millinocket Stream would be increased to 60 cfs between May 1 and October 15 but would remain at 20 cfs during the remainder of the year. Back Channel flows would remain at leakage of 2 to 5 cfs, except during periods of high inflow, when water may be spilled at Stone dam. Flows below Millinocket, Dolby, and East Millinocket developments would continue to be at least 2,000 cfs.

2.1.3.3 Water Quality

GNP contends that project operations do not affect mercury levels within project impoundments and that mercury levels are statistically no different than levels in other impoundments and natural lakes within the area. GNP, therefore, does not propose to conduct further studies other than those with which it is participating voluntarily already: the Environmental Protection Agency's (EPA) Environmental Monitoring and Assessment Program (EMAP) and the International Toxics Monitoring (ITM) program begun in 1992 by several New England states and Canadian provinces.

GNP contends that Back Channel does not have to meet water quality standards because it is not classified by the state. GNP also maintains that studying low dissolved oxygen (DO) in Dolby Pond is unnecessary and that Dolby Pond meets state water quality standards under current project operation.

2.1.3.4 Fisheries

GNP does not propose to provide anadromous fish passage because the state has no plans to restore anadromous Atlantic salmon to the West Branch. GNP agreed to modify and maintain the North Twin fishway. GNP proposes to provide water to a channel that parallels the West Branch near Holbrook Stream to enhance the salmon nursery habitat in the West Branch. Additionally, GNP proposes to regulate seasonal draw-downs of the North Twin impoundment so that it reaches its lowest level before lake trout spawning begins (October 15). GNP also proposes to implement a Millinocket Stream Fisheries Management Plan in which GNP would stock more and larger brook trout to Millinocket Stream. GNP also proposes to increase flow in this reach to 60 cfs between May 1 and October 15.

2.1.3.5 Wetlands

GNP proposes to enhance wetlands at the North Twin impoundment at the Deep Cove East, Deep Cove West, and Wadleigh Brook sites. The proposed measures would enhance existing vegetation at these sites by increasing water retention time during draw-downs.

2.1.3.6 Terrestrial

GNP (1992c) proposes to implement a Back Channel Wildlife Habitat Management Plan to enhance the value of this area for wildlife. The plan would include snag (standing dead trees) management and even-aged and uneven-aged management of the adjacent forest. The goals of this management plan would be to increase vertical stratification and maintain riparian vegetation, while increasing and improving habitat within the area by selective timber harvesting.

2.1.3.7 Threatened and Endangered Species

GNP contends that present project operations have no impacts on threatened and endangered species (GNP, 1993c) and that the documented bioaccumulation of mercury in bald eagles is unrelated to project operation. GNP (1993c) believes that no mitigation measures are necessary because the proposed operational changes and construction would not adversely affect bald eagles.

2.1.3.8 Recreation

GNP proposes to maintain existing recreation facilities within the project areas and to enhance recreation opportunities by modifying existing reservoir operation and providing additional recreation facilities. For the Penobscot Mills Project, GNP proposes to:

- provide relatively stable water levels in the North Twin impoundment from May 1 through mid-August, unless GNP is unable to maintain the required 2,000 cfs minimum flow below Millinocket. This action would address concerns raised by the Maine Department of Conservation and the Save Our Scenic Lakes Association regarding the potential effects of fluctuating lake elevations on recreational activities;
- provide space for three vehicles and five trailered vehicles at the Dead Man's Curve boat access site and for four vehicles at the Route 157 boat access along Dolby Pond within 1 year following issuance of the new license. This action would respond to a recommendation by the Maine Department of Conservation;
- improve the Green Bridge boat access site above Quakish Lake by removing boulders and other obstructions at the shoreline and placing additional gravel in the parking area to accommodate approximately six additional vehicles. These improvements would be completed within 1 year following the issuance of a new license. This action would respond to recommendations by the Maine Department of Conservation; and
- assess the adequacy of the existing recreation facilities within the project area once every 10 years for the term of the new license in consultation with the Maine Bureau of Parks and Recreation.

For the Ripogenus impoundment, GNP proposes to:

- eliminate access fees for noncommercial, day use within the company's gate system for Maine residents, as a result of an agreement with the Fin and Feather Club (see license order);
- improve the informal boat launch on Caribou Lake by removing several trees, placing gravel to provide a pull-ahead area, and making minor improvements to expand the existing parking area;
- assess the adequacy of existing recreation facilities within the project area once every 10 years for the term of the new license in consultation with the Maine Bureau of Parks and Recreation; and
- attempt to schedule routine dam maintenance on week days between Labor Day and Memorial Day to avoid affecting downstream rafting.

For Upper Gorge, GNP proposes to:

 provide a minimum flow of 100 cfs in Upper Gorge from July 1 to September 30 to increase fishing opportunity (leakage would be provided during the remainder of the year); and • maintain the abandoned dirt road that extends from Ripogenus dam along the top of Upper Gorge for pedestrian use (a gate has been installed at its entrance to restrict vehicular access).

For McKay station, GNP proposes to:

- provide the recreational flows outlined in section 2.1.3.2;
- modify the existing vehicle gate to allow unrestricted pedestrian access to the river 24 hours a day;
- install two concrete vault privies at the parking lot adjacent to the McKay station access road;
- establish a telephone message system that would be updated twice daily to provide information about flow conditions along the West Branch downstream of McKay station and any scheduled releases from the dam; and
- review the advisability of whitewater races below McKay station once the organizer of the event notifies GNP in writing of the event and provides documentation of appropriate state approvals.

For Never's Corner, GNP proposes to construct a set of changing rooms at the takeout for whitewater boaters.

2.1.3.9 Land Use

GNP proposes to donate a 250-foot conservation easement to the state of Maine for approximately 73 shoreline miles within the Ripogenus Project area. GNP proposes no easements for the Penobscot Mills Project area. The proposed easements would not be included within the project boundary.

2.1.3.10 Aesthetics

GNP does not propose to reduce impoundment draw-downs to improve aesthetics. GNP contends that reducing draw-downs would be incompatible with its Water Use Plan (WUP) and the flow required for different locations throughout the project areas.

2.1.3.11 Cultural Resources

In consultation with the State Historic Preservation Officer (SHPO), GNP developed two, site-specific Cultural Resource Management Plans (CRMPs). These plans describe data recovery and preservation activities for all identified sites.

2.2 STATUTORY REQUIREMENTS

2.2.1 Water Quality Certification Conditions

The Maine Department of Environmental Protection (DEP) and the Maine Land Use Regulation Commission (LURC) issued Section 401 Water Quality Certification (401 WQC) for the Ripogenus and Penobscot Mills projects (table 2-2).

Section 401(a)(1) of the Clean Water Act (CWA) requires an applicant for a federal license or permit for any activity that may result in a discharge into navigable waters of the United States to provide to the licensing or permitting agency a certification from the state in which the discharge originates that such discharge will comply with certain sections of the CWA. A state Water Quality Certificate (WQC), therefore, is a prerequisite for obtaining a Commission license. The Commission's past practice has been to include all state water quality conditions in any order issuing a project license; however, as stated in *Tunbridge Mill Corporation* (68 FERC ¶ 61,078, 1994), under Section 401(d), states may lawfully impose only conditions related to water quality. In examining the conditions proposed in the Penobscot Mills and Ripogenus WQCs in section 5 of this FEIS, we follow the principles discussed in *Tunbridge*.

Table 2-2. West Branch Penobscot River Basin hydroelectric project water quality certification (Source: GNP, Staff)							
Project Project Number Date Granted							
Ripogenus	2572	4/15/93					
Millinocket Lake Storage	2458	4/22/93					
Penobscot Mills	2458	4/22/93					

2.2.1.1 Ripogenus Project

LURC granted certification that there is a reasonable assurance that the continued operation of the Ripogenus Project, as proposed by the applicant and modified herein, would not violate applicable water quality standards, subject to eight conditions (Boulter, 1993).

- 1. Minimum Flows
 - A. Except as temporarily modified by operating emergencies beyond the applicant's control, as defined below, the Ripogenus Project must be operated such that minimum flows are maintained as proposed by the applicant (including 100 cfs seasonal flow in Upper Gorge), and described in this decision, with the following exceptions:
 - (1) a minimum flow of at least 12 cfs must be maintained in Upper Gorge at all times; and

- (2) during the period from May 1 to September 15, a daytime recreational flow of at least 2,000 cfs must be provided during normal months (as defined in the WUP and described above), and of at least 1800 cfs during wet or dry months. These flows must be achieved as soon as possible following any outage.
- B. Operating emergencies beyond the applicant's control include, but may not be limited to, equipment failure or other temporary abnormal operating condition, generating unit operation or interruption under power supply emergencies, and orders from local, state, or federal law enforcement or public safety authorities.
- C. Flows required to promote salmon spawning and incubation, as described in the application, shall be established each year in consultation with the Maine Department of Inland Fisheries and Wildlife.
- 2. The applicant shall continue to keep accurate flow records and monitor recreational use and shall review the effectiveness of its WUP with the Commission, other interested resource agencies, and affected private groups every 10 years over the term of the new project license.
- 3. The applicant shall conduct a study similar in scope to that conducted by the applicant for Millinocket Lake in 1992 to assess mercury concentrations in aquatic life in the Ripogenus impoundment and the West Branch. Such study shall include sampling of game fish in the project area. A work plan and proposed schedule for the study shall be submitted to the Commission for its review and approval within 90 days of the effective date of this certification. Such study, including all data collected, shall be submitted to the Commission upon completion. In addition, the applicant shall cooperate in a study to be conducted by the DEP and the EPA to determine the interrelationship and impacts of atmospheric deposition and water level fluctuations on concentrations of mercury and other heavy metals on aquatic life in the project waters.
- 4. The applicant shall, within six months of FERC relicensing or upon such a schedule as may be established by FERC, submit a plan for implementing all proposed fisheries, fishing, recreational, habitat, and navigational enhancements specified herein and those specified in its FERC application. This plan shall be reviewed by and must receive approval of the LURC.
- 5. This approval is limited to and includes the proposals and plans contained in the application and supporting documents submitted by the applicant except as herein specified. All variances from the plans and proposals contained in said documents are subject to the review and approval of the Commission prior to implementation.
- 6. The applicant shall secure and appropriately comply with all applicable federal, state, and local licenses, permits, authorizations, conditions, agreements, and orders required for the operation of the project.

- 7. This water quality certification shall be effective on the date of issuance of a new hydropower project license by the FERC and shall expire with the expiration of the FERC license.
- 8. This water quality certification does not constitute an approval of any aspects of the applicant's WUP or any other elements of its proposal that deal with upstream impoundments, as these are to be subject to future licensing by the FERC and water quality certification by the State.

2.2.1.2 Penobscot Mills Project

The Maine DEP certified that the continued operation of the Penobscot Mills Project would not violate applicable water quality standards, subject to 17 special conditions (Marriott, 1993). Six of these conditions pertain specifically to Millinocket Lake Storage dam, which has a separate WQC. (Maine DEP expressly waived its authority to certify that continued operation of the Penobscot Mills Project will meet applicable water quality standards in the section of the West Branch known as the Back Channel.)

Millinocket Lake Storage Dam.

- 1. Except as temporarily modified by operating emergencies beyond the applicant's control, a minimum flow of 60 cfs shall be maintained from the Millinocket Lake Storage dam to Millinocket Stream from May 1 to October 15 annually, and a minimum flow of 20 cfs shall be maintained during the remainder of the year.
- 2. Except as temporarily modified by approved maintenance activities, inflows to the project area, or operating emergencies beyond the applicant's control, the water level in Millinocket Lake shall be maintained between elevations 470.0 feet and 480.0 feet mean sea level (MSL) while providing water to maintain North Twin impoundment levels.
- 3. The applicant shall implement and monitor the results of an upgraded spring and fall brook trout stocking program as specified in Millinocket Stream. The Department reserves the right to approve a revised stocking program if deemed necessary to meet the goal of providing a seasonal fishery in Millinocket Stream.
- 4. The applicant shall provide enhancement of existing wetlands in the Penobscot Mills Project area, and shall submit plans for evaluating, implementing, and monitoring these enhancements.
- 5. The applicant shall cooperate in a study to be conducted by the DEP and the EPA to determine the interrelationship and impacts of atmospheric deposition and water level fluctuations on concentrations of mercury and other toxic metals on aquatic life in the project waters.
- 6. The applicant shall consult with the Department of Conservation regarding the need for a study to mark and remove submerged hazards to navigation in

Millinocket Lake, and shall, if requested by Conservation, submit plans for investigating and for marking and/or removing hazards in the lake.

North Twin, Millinocket, Dolby, and East Millinocket Hydroelectric Developments.

- 1. Except as temporarily modified by operating emergencies beyond the applicant's control, the Millinocket, Dolby, and East Millinocket developments shall be operated as run-of-river facilities while providing an instantaneous minimum flow of 2,000 cfs to the West Branch at Millinocket.
- 2. Except as temporarily modified by approved maintenance activities, inflows to the project area, or by operating emergencies beyond the applicant's control, the water level in the North Twin impoundment shall be maintained at or above the lake trout spawning/incubation level for the period on or about October 15 through May 1 annually, and shall be maintained at a relatively stable level from May 1 through mid-August annually, unless the minimum flow of 2,000 cfs cannot be maintained at Millinocket.
- 3. The applicant shall investigate the extent to which DO deficits in the Dolby impoundment are due to discharges from the Millinocket Mill, and shall submit the results of the investigation, along with a discussion of possible corrective actions, to the Department in conjunction with the next renewal of the Waste Discharge License for the Millinocket Mill.
- 4. The applicant shall cooperate in a study to be conducted by the DEP and the EPA to determine the interrelationship and impacts of atmospheric deposition and water level fluctuations on concentrations of mercury and other toxic metals on aquatic life in the project waters.
- 5. The applicant shall undertake appropriate repairs and/or modifications to the existing North Twin fishway.
- 6. The applicant shall conduct a study as specified to monitor togue (lake trout) reproductive success in the North Twin impoundment following licensing.
- 7. The applicant shall provide enhancement of existing wetlands in the Penobscot Mills Project area, and shall submit plans for evaluating, implementing, and monitoring these enhancements.
- 8. The applicant shall consult with the Department of Conservation regarding the need for a study to mark and remove submerged hazards to navigation in the North Twin impoundment, and shall, if requested by Conservation, submit plans for investigation and for marking and/or removing hazards in the impoundment.
- 9. The applicant shall improve existing recreational access facilities in the project area by: providing parking areas for four vehicles at the Route 157 Causeway site and for three vehicles and five trailered vehicles at the Dead Man's Curve site; and removing boulders at the boat launch and adding gravel to expand the

size of the parking area at the boat put-in site located upstream of Quakish Lake at the Green Bridge.

- 10. This approval is limited to and includes the proposals and plans contained in the application and supporting documents submitted and affirmed to by the applicant. All variances from the plans and proposals contained in said documents are subject to the review and approval of the Board or Department prior to implementation.
- 11. The applicant shall secure and appropriately comply with all applicable federal, state and local licenses, permits, authorizations, conditions, agreements and orders required for the operation of the project.

2.2.2 Section 18 Fishway Prescriptions

Section 18 of the FPA provides the Secretary of the U.S. Department of Interior (Interior) the authority to prescribe fishways² at Commission-licensed projects. Interior filed the following measures for license implementation pursuant to Section 18 for the Ripogenus and Penobscot Mills projects (letters from J. Deason, Interior, May 24, 1993).

Any license issued by the Commission for the Ripogenus Project or the Penobscot Mills Project should include the following special article:

The Secretary of the Interior's authority to prescribe the construction, operation, and maintenance of fishways is reserved under Section 18 of the Federal Power Act.

2.3 PROJECT ALTERNATIVES

GNP proposes no modifications of existing impoundments and structures at the Ripogenus and Penobscot Mills projects but would provide several measures intended to enhance resources in the project area. Conservation intervenors (CI³) and EPA, Interior, PIN, TU and others requested that new licenses for the projects include enhancement measures (e.g., minimum flow releases) substantially greater than those proposed by GNP.

We evaluated the projects as proposed by GNP (Applicant's Proposal) and defined two alternatives to provide a sound basis for identifying the appropriate balance between developmental and nondevelopmental uses of the waterway. The Applicant's Proposal includes all enhancement measures proposed by GNP and some recreation enhancements that GNP agreed to through settlement agreements with several intervenors during scoping.

² Section 18 of the FPA states: "The Commission shall require the construction, maintenance, and operation by a licensee at its own expense of. . . such fishways as may be prescribed by the Secretary of Commerce or the Secretary of Interior, as appropriate."

³ The Conservation Intervenors consist of the following individual groups: American Rivers, American Whitewater Affiliation, Appalachian Mountain Club, Conservation Law Foundation, and Maine Audubon Society.

Alternative 1 includes measures proposed by the CI and other parties to maximize or substantially increase benefits for various resources, particularly fisheries, without regard to economic effect for GNP.

Alternative 2 includes enhancement measures intermediate between those sought by the CI and those proposed by GNP to enable us to evaluate a complete range of alternatives. To establish the specific measures appropriate for this alternative, we considered the benefits and effects of several different minimum flow levels for the Back Channel and Millinocket Stream before selecting the specific flows to include in Alternative 2 (see section 4).

Under the No-action Alternative, the projects would continue to operate as they do now.

2.3.1 Alternative 1

This alternative reflects the requests and interests of the CI, Interior, EPA, and others.

Under this alternative, the proposed project would be modified to reduce adverse effects or enhance environmental values beyond the level proposed by the applicant. The following measures would be implemented:

- maintain flows in Upper Gorge of 100 cfs year from July 1 to September 30 and 50 cfs during the remainder of the year;
- provide a minimum flow of at least 1,422 cfs in the West Branch below McKay station between October 15 and June 7
- provide greater stability of project impoundment levels;
- enhance wetlands both onsite and offsite for both projects;
- create building setbacks and vegetative buffers within the watershed to prevent development and potential subsequent degradation of water quality;
- maintain year-round flows of 60 to 80 cfs in Millinocket Stream;
- provide flows of 350 to 500 cfs in Back Channel;
- operate Millinocket, Dolby, and East Millinocket in instantaneous run-of-river mode;
- provide artificial nesting platforms for common loon and other aquatic birds; and
- assess the adequacy of existing recreational facilities within the project areas once every 6 years for the term of the new licenses in consultation with Fish and Wildlife Service (FWS), National Park Service (NPS), Maine Department of Inland Fisheries and Wildlife (DIFW), Maine Department of Conservation (DOC), and the Penobscot Indian Nation (PIN).

2.3.2 Alternative 2

FERC staff developed this alternative to present a level of enhancement that is intermediate between Alternative 1 and the No-action Alternative. Under this alternative, we considered adopting the following actions in addition to those proposed by GNP:

- instituting flows in Upper Gorge of 100 cfs from July 1 to September 30 and between leakage (12 cfs) and 30 cfs during the remainder of the year;
- provide a minimum flow of at least 1,422 cfs in the West Branch below McKay station between October 15 and June 7
- instituting flows in Back Channel between leakage and 165 cfs;
- instituting flows in Millinocket Stream of 60 cfs from May 1 to October 15 and 60 cfs or inflow during the remainder of the year;
- delaying and modifying fall draw-down at storage impoundments; and
- establishing shoreline protection measures within the project areas to protect existing aesthetic and recreational resources;
- enhancing wetlands on North Twin and Ripogenus impoundments; and
- providing artificial nesting platforms for common loon and other aquatic birds; and
- assess the adequacy of existing recreational facilities within the project areas once every 6 years for the term of the new licenses in consultation with FWS, NPS, DIFW, DOC, and the Maine Bureau of Parks and Recreation (BPR).

2.3.3 No-Action Alternative

Under the No-action Alternative, the projects would continue to operate under the terms and conditions of the existing licenses, and no new environmental protection or enhancement measures would be implemented. We use this alternative to establish baseline environmental conditions to compare with other alternatives.

2.3.4 Alternatives Considered but Eliminated from Detailed Study

We considered several other alternatives to the applicant's relicensing proposal but eliminated them from detailed study because they are not reasonable under the circumstances of this case:

- federal takeover and operation of the projects;
- issuing a nonpower license; and
- decommissioning the projects.

We do not consider federal takeover to be a reasonable alternative. Federal takeover and operation of the projects would require Congressional approval. Although that alone would not preclude further consideration of this alternative, no evidence indicates that federal takeover should be recommended to Congress. No party has suggested that federal takeover would be appropriate, and no federal agency has expressed an interest in operating the projects.

Issuing a nonpower license would not provide long-term resolution of the issues presented. A nonpower license is a temporary license that the Commission terminates whenever it determines that another governmental agency will assume regulatory authority for and supervision over the lands and facilities covered by the nonpower license. In this case, no agency has suggested its willingness or ability to do so. No party has sought a nonpower license, and we have no basis for concluding that the projects should no longer be used to produce power; therefore, a nonpower license is not a realistic alternative for relicensing.

The projects could be decommissioned with or without dam removal. Either alternative would involve denying the application and terminating the existing license with appropriate conditions. No participant has suggested that dam removal would be appropriate in this case, and we have no basis for recommending it. The existing developments provide extensive recreation opportunities for local residents and visitors to the region and also supply water for local municipalities. Dam removal, therefore, is not a reasonable alternative to relicensing the projects with appropriate mitigation and enhancement measures.

The second decommissioning alternative would involve retaining the dam and disabling or removing equipment used to generate power. Project works would remain in place and could be used for historic or other purposes. This would require us to identify another government agency that is willing and able to assume regulatory control and supervision of the remaining facilities. No agency has stepped forward, and no participant has advocated this alternative, nor have we any basis for recommending it. Because the power supplied by the projects is needed, we would also have to identify a source of replacement power. In these circumstances, we do not consider removal of electric generating equipment to be a reasonable alternative.

2.4 ECONOMIC COMPARISON OF ALTERNATIVES

In this section the staff analyzes the economic benefits of the proposed alternatives and determines the economic effects on the project of the various measures intended to provide environmental and recreational benefits. The staff performed an incremental analysis to estimate the annual net benefit (ANB) of each alternative. This analysis is incremental in that it considers only the power benefits (or costs) and the project costs produced by each alternative in comparison with the No-action Alternative. An alternative's ANB has two components:

• The annual power value, which consists of the value at current market prices of the change in annual power generation. This can be positive or negative depending on whether annual power generation increases or decreases.

• The annual project cost, which consists of its annualized capital cost and its annual operating and maintenance (O&M) cost

The incremental ANB for an alternative is obtained by subtracting its annual project cost from its annual power value.

We estimated the incremental ANBs using current (1996) costs and market values. We did not consider future inflation or escalation of market prices or construction costs. Our analysis assumed beginning operations immediately, amortizing project capital costs over the period of analysis, and holding power values constant at current levels during that time. The alternatives consisted of various combinations of operational modifications (all of which would reduce power generation) and physical improvements designed to provide environmental benefits (e.g., conservation easements and enhancements of wetlands and fisheries). The alternatives under consideration were

- the Applicant's Proposal;
- Alternative 1, based on the recommendations of the CI and some agencies;
- Alternative 2 (Final Recommendation), developed by the Commission's staff;
- Alternative 2 (Intermediate Flow); and
- the No-action Alternative.

Table 2-3 summarizes the key parameters used in the economic analysis.

Section 2.4.1 presents the calculation of the annual power value for each alternative. Section 2.4.2 presents the derivation of the annual project cost. Section 2.4.3 presents the incremental ANBs of each alternative and an estimate of the cost of power.

2.4.1 Annual Power Value

The five operational aspects of the alternatives under consideration that would affect annual power production are:

- flows from the Ripogenus impoundment to the West Branch of the Penobscot River (Ripogenus Project),
- flows in the Upper Gorge (Ripogenus Project),
- elevations in the North Twin impoundment (Penobscot Mills Project),
- flows in Millinocket Stream (Penobscot Mills Project), and
- flows in the Back Channel (Penobscot Mills Project).

Table 2-3. Economic analysis parameters (Source: Staff, GNP)						
Parameter Value						
Benefit/Cost Base Year	1996					
Terms of Analysis & Financing	30 years					
Depreciation Period	20 years					
Interest & Discount Rates	10 percent					
Annual Construction Cost Escalation Rate 0 percent						
Annual Power Value Escalation Rate	0 percent					
O&M Expense Rate	\$7.17/MWh					
Annual O&M Expense Escalation Rate	0 percent					
Federal Tax Rate	34 percent					
State and Local Tax Rate	3 percent					
Current Value of Alternative Power* \$73.92/MWh						
 Equivalent to the cost of obtaining replacement power in the form of firm purchases from Bangor Hydro-Electric. 						

Table 2-4 summarizes the incremental losses in annual power generation associated with operational modifications for the alternatives compared with the No-action Alternative. Based on the project's current annual power output of 620,400 MWh (including hydroelectrical and hydromechanical units), the four alternatives would produce small losses in annual power output. These declines would range from 3 percent under the Applicant's Proposal to 7 percent under Alternative 1. The release of minimum flows to the Back Channel at the Penobscot Mill's Project would cause the most significant reduction in annual generation.

2.4.1.1 Applicant's Proposal

The operational changes under the Applicant's Proposal would entail

- providing a flow between 2,000 cfs and 2,300 cfs (depending on day of week) in normal years and between 1,800 cfs and 2,200 cfs (depending on day of week) in wet or dry years in the West Branch below McKay station during the recreation season;
- providing a minimum flow of 100 cfs in the Upper Gorge from July 1 to September 30 (with leakage from October 1 to June 30);
- maintaining relatively stable water elevations in the North Twin impoundment from May 1 to August 15;

 providing a minimum flow of 60 cfs in Millinocket Stream from May 1 to October 15 and 20 cfs from October 16 to April 30; and

ล

• providing leakage in the Back Channel.

Table 2-4. Incremental* average annual power production losses (MWh) (Source: Staff, GNP)							
	Applicant's Proposal	Alternative 1	Alternative 2: Final Recommendation	Alternative 2: Intermediate Flow			
Ripogenus							
West Branch Recreation Flow	3,300	3,300	3,300	3,300			
Upper Gorge Minimum Flow	2,200	5,355	2,200	3,787			
Ripogenus Total	5,500	8,655	5,500	7,087			
Penobscot Mills							
North Twin Elevation	5,100	5,100	5,100	5,100			
Millinocket Stream Minimum Flow	350	771	771	771			
Back Channel Minimum Flow	0	20,800	0	9,900			
Penobscot Mills Total	5,450	26,671 5,871		15,771			
Efficiency Reductions							
Hydroelectric Efficiency	4,100	4,100	4,100	4,100			
Steam Efficiency	4,000	4,000	4,000	4,000			
Efficiency Reductions Total	8,100	8,100	8,100	8,100			
Total	19,050	43,426	19,471	30,958			
* Relative to the No-action Alternative (current facilities and operations).							

Table 2-4 shows that the total annual decline in power production would be 19,050 MWh, which is the lowest decline of the four alternatives.

2.4.1.2 Alternative 1

The Cl's recommendations for West Branch flows and North Twin elevations were the same as under the Applicant's Proposal, but regarding the other operational aspects, this alternative would entail

• providing a minimum flow of 100 cfs in the Upper Gorge from July 1 to September 30 and 50 cfs from October 1 to June 30;

- providing a minimum flow between 60 cfs and 80 cfs in Millinocket Stream yearround; and
- providing a minimum flow between 350 cfs and 500 cfs in the Back Channel.

The annual losses in power production caused by these recommendations are shown in Table 2-4. This alternative would cause the largest annual decline in power generation, a loss of 43,426 MWh.

2.4.1.3 Alternative 2 (Final Recommendation)

The provisions for West Branch flows, Upper Gorge flows, North Twin elevations, and Back Channel flows were the same under this alternative as under the Applicant's Proposal, but this alternative would entail providing a minimum flow of 60 cfs in Millinocket Stream from May 1 to October 15 and 60 cfs or inflow from October 16 to April 30. The annual decline in power output under this alternative would be 19,471 MWh.

2.4.1.4 Alternative 2 (Intermediate Flow)

This alternative is the same as Alternative 2 (Final Recommendation) except for the following two differences:

- Flow in the Upper Gorge flow between October 1 and June 30 would be 30 cfs instead of 12 cfs; and
- the minimum flow in Back Channel would be 165 cfs year-round rather than leakage flow

Because of these two higher flows, this alternative would produce a decline in annual power generation of 30,958 MWh as shown in Table 2-5.

2.4.1.5 No-action Alternative

The No-action Alternative would involve no operational modifications. The current annual power output of 620,400 MWh from the two dams would be maintained.

2.4.1.6 Summary

The applicant states that the minimum flow requirements and draw-down limits in its proposal would impose operational constraints that would reduce the generating efficiency of its system under certain flow conditions. The applicant estimated the resulting average annual power losses at 4,100 MWh caused by reductions in hydroelectric operating efficiency and 4,000 MWh caused by reductions in steam generating efficiency. We allocated these losses on a system-wide basis, rather than to either project individually, and applied them to all the alternatives other than the No-action Alternative.

Table 2-5 presents the annual power value for each alternative. The annual power values were based on a current power cost of \$73.92/MWh. This represents the marginal

cost that would be incurred by the applicant in making purchases for replacement firm power from Bangor Hydro-Electric (BHE). Our cost estimate is calculated from BHE's sales of electricity to industrial customers that were reported in the 1994 FERC Form 1. A representative of BHE said (April, 1996) that their rates had not changed in several years so this cost is still applicable (personal communication, Jeff Wood, BHE, April, 1996).

Table 2-5. Incremental* annual power values in dollars by alternative (Source: Staff, GNP) (1996\$)							
	Applicant's Proposal	Alternative 1	Alternative 2: Final Recommendation	Alternative 2: Intermediate Flow			
Ripogenus			·····	<u></u>			
West Branch Recreation Flow	- \$243,900	- \$243,900	- \$243,900	- \$243,900			
Upper Gorge Minimum Flow	- \$162,600	- \$395,800	- \$162,600	-\$279,900			
Ripogenus Total	- \$406,500	- \$639,700	- \$406,500	-\$523,800			
Penobscot Mills							
North Twin Elevation	- \$377,000	- \$377,000	-\$377,000	- \$377,000			
Millinocket Stream Minimum Flow	- \$25,900	-\$57,000	- \$57,000	-\$57,000			
Back Channel Minimum Flow	\$0	-\$1,537,500	\$ 0	-\$731,800			
Penobscot Mills Total	-\$402,900	-\$1,971 ,500	-\$405,700	-\$1,137,500			
Efficiency Reductions							
Hydroelectric Efficiency	- \$303,100	- \$303,100	-\$303,100	-\$303,100			
Steam Efficiency	- \$295,700	- \$295,700	-\$295,700	-\$295,700			
Efficiency Reductions Total	- \$598,800	- \$598,800	-\$598,800	-\$598,800			
Total	-\$1,408,200	-\$3,210,000	-\$1,439,300	-\$2,288,400			
* Relative to the No-action Alternative (current facilities and operations).							

The applicant indicated that annual O&M expenses are directly related to the amount of power generated, and would decline for each alternative where annual power production is reduced. The resulting reduction, or savings in annual O&M costs, would partially offset the cost of obtaining replacement power. The figure provided by the applicant indicates a reduction in annual O&M costs of \$7.17/MWh. We determined the reduction in current year O&M costs that would occur under each alternative and included it in our analysis. We discuss these reductions in expenses here because they are related to power generation, but they are not included in Table 2-5 because O&M expenses represent a portion of project costs, distinct from replacement (alternative) energy costs and, therefore, were included in the calculation of annual project costs.

2.4.2 Annual Project Cost

The annual project cost for an alternative consists of its annualized capital cost and its annual O&M costs. The capital costs for each of the alternatives are presented in Table 2-6. Approximately 94 percent of the capital cost of Alternative 1 is for the conservation easements; this is not a depreciable cost, and the staff accounted for this fact in its analysis. The No-action Alternative has no physical improvements and no capital cost.

Table 2-6. Capital cost by alternative (Source: Staff, GNP) (1996\$)							
	Applicant's Proposal	Alternative 1	Alternative 2: Final Recommendation	Alternative 2: Intermediate Flow			
Ripogenus							
Wetlands Enhancements	\$0	\$1,285.600	\$702,700	\$702,700			
Conservation Easements	\$0	\$8,316,500	\$0	\$0			
Fisheries Enhancements	\$0	\$20,800	\$20,800	\$20,800			
Ripogenus Total	<u>\$0</u>	\$9.622.900	\$723.500	\$723.500			
Penobscot Mills							
Wetlands Enhancements	\$0	\$47,900	\$47,900	\$47, 9 00			
Conservation Easements	\$0	\$16,353,300	\$0	\$0			
Penobscot Mills Total	<u>\$0</u>	\$16.401.200	\$47.900	\$47,900			
Total	\$0	\$26,024,100	\$771,400	\$771,400			
* Relative to the No-action Alternative (current facilities and operations).							

The staff annualized the capital cost of each alternative based on the financing assumptions listed in Table 2-3 and added the change in annual O&M costs to estimate the incremental annual project cost. These costs are presented in Table 2-7; they are incremental in that they present the change in annual project cost that would occur under each alternative compared with the No-action Alternative.

The staff's analysis indicates that the annual project cost would be negative for all alternatives but Alternative 1 as shown in Table 2-7. A negative annual project cost (i.e., a cost savings) would occur for the following reasons:

- Annual O&M costs would decline because of the reduction in annual power generation as discussed in section 2.4.1.6.
- Minimal or no capital cost would produce minimal or no annualized capital costs (i.e., debt service payments). The exception is Alternative 1 that would have significant annual debt service payments.

• Tax benefits are produced when the net income for a component is negative (i.e., annual before-tax expenses exceed revenues).

Table 2-7. Incremental* annual project costs (Source: Staff, GNP)							
Applicant's Alternative 1 Alternative 2: Alternative 2: Proposal Final Intermediate Flow Recommendation							
Ripogenus	-\$164,300	\$666,300	-\$104,950	-\$152,300			
Penobscot Mills	-\$162,800	\$814,800	-\$171,400	-\$467,100			
Efficiency Reductions Total	-\$203,500	-\$203,500	-\$203,500	-\$203,500			
Total	-\$530,600	\$1,277,600	-\$479,800	-\$822,900			
* Relative to the No-action Alternative (current facilities and operations).							

The staff's analysis showed that for three of the four alternatives the combination of a reduction in annual O&M costs and tax benefits would exceed other annual costs such as debt service and fixed O&M costs (e.g., insurance and local taxes), producing negative annual project costs. This result occurs because O&M expenses and tax benefits are related to power production, which declines under all the alternatives; the other annual costs are related to capital costs, which are minimal or zero for three of the four alternatives. A negative annual project cost indicates that O&M expenses related to power production and tax expenses decreased more than the increase in annualized capital costs and other O&M costs.

2.4.3 Annual Net Benefits

All the alternatives would result in negative annual net benefits, as shown in Table 2-8. This would be due to the combination of the loss in annual power generation that would occur under each alternative, and its associated annual project costs. Even where the annual project costs would be negative (i.e., a cost reduction), the corresponding loss in revenues would be larger, still resulting in a negative ANB. The combined effects of the loss in efficiency and increased minimum flows would produce a loss in annual power generation. The net result would be an increase in the cost of generating power at the two projects. The Applicant's Proposal would reduce current-year net benefits by \$877,600 (i.e., would cost the applicant \$877,600 in the current year). Alternative 1 and Alternative 2 (Final Recommendation) would cost the applicant \$4,487,600 and \$959,500, respectively, in the current year. Alternative 2 (Intermediate Flow) has higher costs than Alternative 2 (Final Recommendation) because of the higher power losses that would result from the 165 cfs minimum flow in the back channel of the Penobscot Mills project. Alternative 1 would have the highest costs (i.e., it has the largest negative ANB) because of the high cost of acquiring the conservation easements.

The incremental analysis presents the economic costs to the applicant's shareholders of adding important environmental and recreational enhancements at the Penobscot and

Table 2-8. Annual net benefits by alternative ^a (Source: Staff)												
	Api	plicant's Proposa	1		Alternative 1		Alternative 2: Final Recommendation			Alternative 2: Intermediate Flow		
Project & Component	Power Value	Annual Project Cost ^a	ANB	Power Value	Annual Project Cost ^e	ANB	Power Value	Annual Project Cost ^a	ANB	Power Value	Annual Project Cost ^a	ANB
Ripogenus Project												
Maintain stable flows in west br. of Penobscot River below McKay Station	-\$243,900	-\$98.600	-\$145,300	-\$243,900	-\$98,600	-\$145.300	-\$243,900	-\$98,600	- \$ 145,300	-\$243,900	-\$98.600	\$145,300
Min. flow in upper gorge	-\$162,600	-\$65,700	\$96,900	-\$395,800	-\$159,900	-\$235,900	-\$162,600	-\$65,700	-\$96,900	-\$279,900	-\$113,100	-\$166,800
Wetlands enhancements	\$0	\$ 0	\$0	\$0	\$105,600	\$105,600	\$ 0	\$57,700	-\$57,700	\$0	\$55,700	\$57,700
Holbrook stream fisheries enhancement	\$ 0	\$ 0	\$ 0	s o	\$1,700	\$1,700	\$ 0	\$1,700	-\$1,700	\$ 0	\$1.700	-\$1,700
Conservation easement	\$ 0	\$0	\$0	\$0	\$817,500	-\$817,500	\$ 0	S O	\$0	\$ 0	\$ 0	\$0
Sublotal Ripogenus	-\$406,500	-\$164,300	\$242,200	-\$639,700	\$666,300	-\$1,306,000	-\$406,500	-\$104,900	-\$301,600	-\$523,800	\$152,300	\$371,500
Penobacot Mills Project												
Main. rel. stable water levels in the North Twin impound, during summer season	-\$377.000	-\$152.300	-\$224,700	-\$377,000	-\$152,300	-\$224,700	-\$377,000	-\$152,300	-\$224,700	-\$377,000	-\$152,300	·\$224,700
Wetlands enhancements	\$0	\$ 0	\$0	\$0	\$3,900	-\$3,900	\$0	\$3,900	-\$3.900	\$ 0	\$3,900	-\$3,900
Conservation easement	\$0	\$ 0	\$0	so s o	\$1,607.400	-\$1,607,400	\$0	\$0	\$0	\$ 0	\$ 0	\$0
Min. flow in Millinocket Stream	-\$25.900	-\$10,500	-\$15,400	-\$57,000	-\$23,000	-\$34,000	-\$57,000	-\$23,000	-\$34,000	- \$ 57,000	-\$23,000	-\$34,000
Min. flow in back channel	\$0	\$ 0	\$0	-\$1,537,500	-\$621.200	-\$916,300	\$0	S 0	\$ 0	-\$731,800	- \$29 5,700	-\$436,100
Subtotal Penobscot	-\$402,900	-\$162,800	-\$240,100	-\$1,971,500	\$814.800	-\$2,786,300	-\$434,000	-\$171,400	\$262.600	-\$1,165,800	-\$467,100	\$698,700
Operating Impacts			\$0									
Hydro operating efficiency losses	-\$303,100	-\$103,000	-\$200,100	-\$303,100	-\$103,000	-\$200,100	-\$303,100	-\$103.000	-\$200,100	-\$303,100	-\$103,000	-\$200,100
Steam Generation Losses	-\$295,700	-\$100,500	-\$195,200	-\$295,700	-\$100,500	\$195,200	-\$295,700	-\$100.500	-\$195,200	-\$295,700	-\$100,500	\$195,200
Subtotal Operating	\$598,800	-\$203,500	-\$395,300	-\$598,800	\$203,500	-\$395,300	-\$598,800	-\$203,500	-\$395,300	-\$598,800	-\$203,500	\$395,300
Total (Both Projects)	-\$1,408.200	\$530,600	- \$ 877,600	\$3,210,000	\$1,277,600	-\$4,487,600	-\$1,439,300	-\$479,800	-\$959.500	\$2,288,400	-\$822,900	\$1,465.500

* Relative to the No-action Alternative.

* A minus sign (-) under the Annual Project Cost Column indicates a reduction in costs, or a savings.

2-32

Ripogenus projects. Because there would be no power enhancements, including them in the existing projects would reduce the projects' current total ANBs.

2.4.4 Cost of Power

Another way of applying the staff's economic analysis is to compare the cost of power under each alternative to that of the least-cost power alternative. The least-cost power alternative is purchasing power from BHE, or a current cost of 73.92 mills/kWh. The staff estimated the increase in the cost of generating power at the Penobscot and Ripogenus projects that would occur under the four alternatives. This analysis considers both the total annual power output that would occur if an alternative was implemented, and the annual cost of producing the total annual power output. These results are presented in Table 2-9.

Table 2-9. Cost of power by alternative (Source: Staff)						
Alternative	Cost of Power (mills/kWh)	ANB Over the Least-cost Alternative (mills/kWh)	% Change in Cost of Power from No-action Alternative			
Least-cost alternative: purchase from BHE	73.92	NA				
No-action	34.10	39.82				
Applicant's proposal	34.30	39.62	0.6%			
Alternative 1	38.67	35.25	13.4%			
Alternative 2 (Final Recommendation)	34.41	39.51	0.9%			
Alternative 2 (Intermediate Flow)	34.50	39.42	1.2%			

The staff estimated the power cost of the No-action alternative based on a current book value for the two dams of \$31,949,500 in 1995 dollars and on the O&M cost of \$7.17/MWh. As shown in Table 2-9, we estimate the current power cost of the existing projects to be 34.1 mills/kWh, indicating that the two projects are currently capable of producing power at a cost well below that of the least-cost power alternative.

Table 2-9 shows that implementing any of the alternatives would result in only a small absolute increase and a small percent increase in the cost of producing power at the two projects. The maximum cost increase would be 4.57 mills/kWh under Alternative 1. The maximum increase in the cost of power in percentage terms would be 13.4 percent under Alternative 1. The Applicant's Proposal would have a minimal adverse effect on the cost of power, increasing it by 0.2 mills/kWh, or about 0.6 percent. The largest increase in the cost of power would be caused by the requirement to purchase conservation easements as shown by

highest power costs for Alternative 1 in Table 2-9. Table 2-9 further confirms that the proposed recreation, conservation, and environmental enhancements for both versions of Alternative 2 can be made at both projects with only a small increase in the cost of power. Even with these improvements the cost of power would be about half that of the least-cost alternative.

3.0 AFFECTED ENVIRONMENT

3.1 GENERAL SETTING

The Penobscot River Basin, which covers 8,750 square miles, is the largest river basin in Maine and the second largest in the Northeast, after the Connecticut River Basin (figure 3-1). Headwaters arise at elevations of between 800 and 1,200 feet (Baum, 1983). The basin, which is 95 percent forested, is located in central Maine and empties into Penobscot Bay about 20 miles south of Bangor. Average annual precipitation in the region is approximately 42 inches as rainfall. This includes snowfall, which averages 95 inches (Cutting, 1959).

A principal physiographic feature of the northern part of the basin is 5,267-foot- high Mt. Katahdin, the state's highest peak, located in the 200,000-acre Baxter State Park. Several large impoundments and a variety of headwaters in the basin feed the Penobscot River (figure 3-2).

Throughout the 19th and 20th centuries, the West Branch, which drains about 2,100 square miles, has been used for lumber and paper-making. These industries used water resources to transport materials and products and in industrial processes. The growth of the towns of Millinocket and East Millinocket paralleled the growth of the mills, which GNP has operated since they were built. The West Branch economy continues to be extremely dependent on the lumber and paper industries. The region is also home to the Penobscot Indian Nation (PIN), much of whose cultural heritage is closely associated with the river and the resources it provides.

Construction of dams, which began on the estuarine tributaries during the late 1700's, has strongly influenced the environmental resources and socioeconomics within the basin. The first mainstem dam was built in the mid-1820's in the Old Town-Milford area. Other early mainstem dams include the Great Works dam, constructed just after 1830, and a dam at McMahon's Falls in Veazie constructed in 1833. Figure 3-1 depicts dams in the basin, and they are listed in Appendix A. This FEIS covers projects on the West Branch, which joins the main stem approximately 3 miles downstream of the East Millinocket Development (figure 3-1). One-hundred thirty-seven dams have been constructed on the West Branch and its tributaries since logging began in 1828 to facilitate the transportation of timber downstream to more populated areas. Construction of dams within the Ripogenus and Penobscot Mills project areas began in 1846; however, Stone dam, the oldest existing dam, was completed in 1899 (figure 3-2). Table 3-1 lists impoundments located upstream of the Ripogenus and Penobscot Mills projects, and figure 3-2 shows their locations.

3.2 GEOLOGY AND SOILS

The Penobscot River Basin is located in the New England upland physiographic region. The area has gentle slopes interspersed with occasional mountains or monadnocks of resistant rocks composed of metamorphic bedrock (shale, slate, and schist) with igneous intrusions. The main stem of the Penobscot River is a lower area; hills generally rise 300 to



Figure 3-1. Existing dams within the Penobscot River Basin and the project area; the key for identification of numbered dams is presented in Appendix A (Source: Staff)





400 feet. The terrain of the West Branch region is characterized by rolling hills and mountains ranging from 2,000 feet to more than 5,000 feet (Mt. Katahdin).

Table 3-1.West Branch impoundments upstream of the Ripogenus and PenobscotMills projects (Source: GNP, Staff)						
Impoundment	Map No.	Usable Storage (acre-feet)	Drainage Area (square miles)			
Rainbow	102	5.900	10.7			
Nesowadnehunk	103	9.734	14.3			
Harrington	107	7.691	39.6			
Umbazooksus	109	9.458	25.7			
Loon	111	10.537	59.9			
Dole Pond	113	3.857	42.8			
Long Pond	114	3.007	14.4			
Penobscot	116		15.8			
Total Unlicensed		58,265 ^(a)	223.1			
FERC No. 2634		·				
Ragged Lake	108	30.500	34			
Caucoongomic	110	42,500	165			
Seboomook	112	106,700	469			
Canada Falls	115	23.400	164			
Total Project 2634		203.100	832			
Grand Total		261.365 ^(b)	1.055.1			
 (a) 2.5 billion cubic feet (bcf (b) 11.4 bcf)					

3.3 STREAMFLOW

-

3.3.1 Regional

In 1841, construction of Lock dam (No. 105 on figure 3-1), Telos dam (No. 104 on figure 3-1), and Telos Cut altered the natural hydrology of the Penobscot River Basin. These facilities diverted flow from the Saint John River Basin to the Penobscot River Basin for log driving. Water is still diverted from Telos, Chamberlain, Round Pond, and Allagash lakes, although log driving no longer takes place. An agreement between the Maine Department of Inland Fisheries and Wildlife (DIFW) and the East Branch Improvement Company mandates a minimum flow of 150 cfs in the East Branch of the Penobscot River from Grand Lake

Matagamon (figure 3-1). GNP is required to maintain a minimum flow of 2,000 cfs from the West Branch of the Penobscot River. A 1911 court decree specifies flow proportions within the Stillwater River (a portion of the Penobscot River that branches off from the main stem in Milford and rejoins it in Orono) and the Penobscot River. These flow proportions are 30 percent and 70 percent, respectively, for average flow conditions; as flows decrease below average, the proportion of flow to the Stillwater decreases, reaching 216 cfs (9 percent) at a Penobscot River flow of 2,400 cfs.

Regulation of storage and flow in upper branches of the basin moderates discharge throughout the lower basin. In general, upstream storage in the west branch increases flows to the downstream areas during low-flow periods and reduces flows during higher spring flows. Flooding is a relatively minor problem in the Penobscot; damages occur infrequently in larger towns such as Bangor, Orono, and Old Town. At present, no flood-damage reduction projects are planned in the basin, although most communities participate in the Federal Emergency Management Agency (FEMA) National Flood Insurance Program.

The average annual runoff throughout the basin is about 1.7 cfs per square mile of watershed; this is equivalent to about 22 inches per year, or 55 percent of the mean annual precipitation. More than 40 percent of the runoff occurs during March, April, and May; the remainder is distributed uniformly throughout the rest of the year.

3.3.2 Site-specific

Figure 3-2 shows dams and impoundments located in the West Branch of the Penobscot River. Table 3-1 presents usable storage capacities and drainage areas for impoundments upstream of the Ripogenus and Penobscot Mills projects. Total usable storage for these upstream impoundments is approximately 11.4 billion cubic feet (bcf), or 19 percent of the total impounded storage in the West Branch. Eight of these projects are unlicensed, and four of them make up the Great Northern Storage Project (FERC Project No. 2634); none have power generating facilities. Table 3-2 lists characteristics of the impoundments associated with the Ripogenus and Penobscot Mills projects.

The average flows in the West Branch at McKay station and Dolby station are 2,699 cfs and 3,979 cfs, respectively. Figures 3-3 and 3-4 show annual flow-duration curves for the West Branch downstream of McKay station and Dolby station. Table 3-3 lists the peak discharges for the PMF, spillway design flood, and standard project flood flows. The highest flow recorded at Ripogenus dam during the period of study (1970 to 1985) was 15,185 cfs; the lowest flow was 200 cfs. At Dolby dam, the highest flow recorded was 35,947 cfs; the lowest was 2,000 cfs. GNP calculated the unregulated 7Q10 flows (the lowest flow that would be expected to occur over a 10-year period for a duration of 7 days) as 81 cfs at Ripogenus dam and 126 cfs at Dolby dam (in the absence of any flow regulation).

3.4 WATER QUALITY

3.4.1 Regional

Water quality in the Penobscot River, including the West Branch, has varied widely since the West Branch was settled. Pollution sources during this period included saw mills,

Table 3-2. Impoundment characteristics of developments in the Ripogenus and Penobscot Mills projects (Source: GNP, Staff)								
Dev elop ment	Impoundment, Lake Names	Map No.	Gross Storage (acre-feet)	Usable Storage (acre-feet)	Usable Storage (billion cubic feet)	Normal Maximum Water Surface Area (acres)	Normal Maximum Water Surface Elevation (feet USGS)	
Ripogenus	Ripogenus Lake, Chesuncook Lake, Caribou Lake, Moose Pond, Black Pond, Brandy Pond	106	710,000	688,705	30.0	29,270	941.6	
East Millinocket	East Millinocket	96	1,950	0	0	128	287.2	
Dolby	Dolby Pond	97	41,956	0	0	2,048	336.2	
Millinocket	Ferguson Pond	99		0	0	262	457.4	
Millinocket	Quakish Lake	99	8,100	0	0	1,344	458.7	
North Twin	North Twin Lake, South Twin Lake, Elbow Lake, Pemadumcook Lake, Ambajejus Lake	100	346,000	344,000	15.0	17,790	491.9	
Millinocket Lake Storage	Millinocket Lake	98	45,370	45,370	2.0	8,640	480.0	

7

F



Figure 3-3. Annual flow duration curve at McKay station for the period 1970-1985



Figure 3-4. Annual flow duration curve at Dolby station for the period 1972-1985

shoe factories, leather tanning operations, pulp and paper manufacturing plants, and untreated municipal sewage (Baum, 1983). Minimum flows required from the Penobscot Mills project has lessened the impact of some of these pollution sources by providing dilution flows during low flow periods. Since the early 1800's, when demand on land and water resources of the basin increased sharply, periodic efforts have been made to improve water quality. Such efforts have often coincided with recognition that Atlantic salmon stocks were being adversely affected. The most recent effort followed passage of the Federal Water Pollution Control Act of 1972, which required using improved technologies to treat all waste sources by 1977. Additionally, this act set the goal of achieving B-2 water quality classification (swimmable-fishable) by 1983. Today, water quality in the Penobscot is much improved, particularly in the West Branch, because of these regulations, changes in industrial processes, and mill closings. Table 3-4 lists major sources of municipal and industrial discharges, and table 3-5 lists water quality classification for various reaches of the Penobscot River. Table 3-6 defines the water use and standards for each classification.

Table 3-3. Peak discharge of West Branch impoundments (Source: GNP, Staff)						
Dam	License No.	Peak Discharge PMF/SDF (cfs)	Peak Discharge SPF (cfs)			
Canada Fails	2634	16,700	-			
Seboomook	2634	42,500	-			
Caucomgomoc	2634	10,400	-			
Ragged Lake (SDF)	2634	600	-			
Ripogenus	2572	65,800	51,300			
North Twin	2458 64,300		49,700			
Stone	2458	64,800	50,700			
Millinocket Lake	2458	4,400	2,100			
Dolby	2458	64,000	53,700			
East Millinocket	2458	64,000	53,700			
Weldon	2520	95,700	87,100			
PMF = probable maximum flood SDF = spillway design flood SPF = standard project flood						

3.4.2 Site-specific

3.4.2.1 Conventional Water Quality

Ripogenus. GNP collected water quality data in 1986, 1987, and 1988 to characterize existing DO and temperature conditions in the impoundment, Upper Gorge, and the West Branch. DO within the impoundment ranged from 3.3 parts per million (ppm) to 10.0 ppm during the summer, and saturation values ranged between 31 and 104 percent. Temperature ranged from 6°C to 25°C during the same period. In Upper Gorge, DO ranged from 7.4 ppm to 12.3 ppm between May and November; saturation ranged from 75 to 100 percent.

Table 3-4. Major point sources on the Penobscot River (Source: Mitnick, 1991)						
	ndustrial Pulp an	License	• Limits	Ţ <u></u>		
Licensee	Location	Receiving Water	Flow (mgd)	Mo. Ave. BOD5 (Ib/day)	Daily Max. BOD5 (Ib/day)	Treatment
Great Northern Paper	Millinocket	West Branch	43	13,700	20,500	Secondary
Great Northern Paper	E. Millinocket	West Branch	30	10,574	18,980	Secondary
Lincoln Paper	Lincoln	Main stem	13.5	4,772	9,176	Secondary
James River	Old Town	Main stem	24.4	7,500	18,000	Secondary
Champion Inter.	Bucksport	Estuary	16	3,100	10,000	Secondary
	Municipal			License	e Limits	T
Licensee	Receiving Water	Flow (mgd)	Mo. Ave. BOD5 (Ib/day)	Weekiy Ave. BOD5 (ib/day)	Daily Max. BOD5 (Ib/day)	Treatment
Millinocket	West Branch	2.33	375	563	625	Secondary
Lincoln	Main stem	1.07	268	402	446	Secondary
Howland	Main stem	0.15				Untreated
Old Town	Main stem	1.7	425	638	709	Secondary
Orono*	Main stem	1.84	384	614	690	Secondary
Veazie	Main stem	0.19	48	71	80	Secondary
Bangor	Main stem	9				Primary
Brewer*	Main stem	3.03	758	1,137	1,264	Secondary
Winterport	Estuary	0.11				Primary
Bucksport	Estuary	0.323				Primary
* Industrial input						

Temperature ranged from 7°C to 18°C during the same period. DO values for the West Branch at McKay station were between 7.3 ppm and 11.6 ppm. Saturation values ranged between 82.6 and 101 percent. Temperatures ranged from 5°C to 22°C.

Table 3-5.Water quality classification of the Penobscot River Basin (Source: Mitnick, 1991; Boulter, 1993; and Marriott, 1993)			
Class	Description		
GPA	Ripogenus impoundment		
В	West Branch from Ripogenus dam to outlet of Ferguson Pond, except North Twin impoundment		
GPA	North Twin impoundment		
GPA	Millinocket Lake		
Α	Millinocket Stream from Millinocket Lake dam to railroad bridge		
В	Millinccket Stream from railroad bridge to West Branch Canal		
С	Millinocket Stream from West Branch Canal to the confluence with West Branch		
С	From outlet of Ferguson Pond to confluence with East Branch, including all impoundments		
С	Main stem, from the confluence with the East Branch to confluence with Mattawamkeag, including all impoundments		
В	Main stem, from the confluence with the Mattawamkeag River to the confluence with Cambolasse Stream		
С	Main stem, from the confluence with Cambolasse Stream to the confluence with Piscataquis River, including all impoundments		
В	Main stem, from the confluence with Piscataquis River to the Maine Central Railroad bridge in Bangor-Brewer		
С	Main stem, from the Maine Central Railroad bridge in Bangor-Brewer to the confluence with Reeds Brook		
SC	Penobscot estuary, entire length		

Penobscot Mills. Water quality conditions in the Penobscot Mills Project area, which vary widely, reflect the variety of habitat types and trophic conditions there. A water quality sampling program conducted by GNP between 1986 and 1988 showed that DO values and saturation values were usually, but not always, above the standards for the respective classification of each riverine segment or impoundment. DEP (Marriott, 1993) classified Ferguson Pond, Quakish Lake, Dolby Pond, and East Millinocket impoundment as riverine rather than GPA (table 3-5), although they exhibit some lacustrine characteristics. Accordingly, they must meet numerical standards, unlike North Twin and Millinocket Lake, which are classified as GPA. Quakish Lake, Ferguson Pond, and Dolby Pond exhibit weak thermal stratification. The combination of substantial amounts of organic matter and thermal stratification causes frequent DO deficits in the deeper strata of these impoundments (GNP, 1991b). East Millinocket, which is a true riverine impoundment, does not experience such

deficits. DEP (Marriott, 1993) concluded that, although DO levels in Dolby Pond are occasionally substandard, these events do not seem to be caused by project operations. DEP, however, did require GNP to conduct further studies of the extent and cause of the DO deficits in Dolby Pond as a condition of the WQC. DEP did not require such studies in Quakish Lake or Ferguson Pond. Riverine segments of the Penobscot Mills Project all met or exceeded the respective water quality standards. Water temperatures in project impoundments and riverine segments rarely exceed 25°C.

Table 3-6.State of Maine river and stream water quality standards for dissolved oxygen (Source: Mitnick, 1991; Boulter, 1993; Marriott, 1993)			
Classification	Water Uses	Standard	
A	Drinking water supply after disinfection Fishing Recreation in and on the water Industrial process and cooling water supply Hydroelectric power generation Navigation Habitat for fish and other aquatic life	≥7 ppm or ≥75 percent saturation ^(a)	
GPA	Drinking water supply after disinfection Fishing Recreation in and on the water Industrial process and cooling water supply Hydroelectric power generation Navigation Habitat for fish and other aquatic life	no numerical standards	
В	Water supply (after treatment) Fishing Recreation in and on the water Industrial process and cooling water supply Hydroelectric power generation Navigation Habitat for fish and other aquatic life	≥7 ppm or ≥75 percent saturation ^(b)	
С	Water supply (after treatment) Fishing Recreation in and on the water Industrial process and cooling water supply Hydroelectric power generation Navigation Habitat for fish and other aquatic life	≥5 ppm or ≥60 percent saturation ^(a)	
 ^(a) Except that in identified salmonid spawning areas where water quality is sufficient to ensure spawning, egg incubation, and survival of early life stages, water quality sufficient for these purposes is to be maintained. ^(b) Except that for the period from October 1st to May 14th, the 7-day mean DO concentration is not to be less than 9.5 ppm and the 1-day minimum DO concentration is not be less than 8.0 ppm in identified fish spawning areas. 			
3.4.2.2 Toxics

Mercury is toxic above certain concentrations. Methylated mercury (the bioavailable form) in the water column or sediments of a water body can accumulate in organisms throughout the food chain, especially at the higher trophic levels represented by predatory fish, bald eagles, and other consumers of fish. EPA's fish consumption guideline is 0.6 micrograms per gram (μ g/g) of total mercury in edible fish tissue (EPA, 1993). Some samples collected from project waters have exceeded this guideline. Independent studies by FWS and DIFW indicate that bald eagles nesting near the Dolby impoundment contain concentrations of mercury almost 25 times higher than concentrations in eagles nesting in less industrialized parts of Maine.

The Conservation Intervenors (CI), Trout Unlimited (TU; American Rivers et al., 1992), EPA (1992), Interior (FWS, 1992), and PIN (1992) all have expressed concern about mercury concentrations within the project areas.

In 1988, GNP sampled sediment from project waters to investigate possible relationships between project operations and mercury concentrations. Sample sites were located upstream and downstream of mill discharges. Concentrations in sediments below discharges ranged from 0.06 to 1.60 μ g/g, whereas concentrations in sediments above discharges ranged from less than 0.03 to 0.3 μ g/g. For comparison, mercury concentrations in sediments from Schoodic Lake and Sabathday Lake were 0.55 μ g/g and 0.59 μ g/g, respectively.

GNP funded studies of mercury concentration in the tissues of fish from the project area during 1992. Mean concentrations of mercury were 0.87 μ g/g in Dolby Pond chain pickerel, 1.09 μ g/g in North Twin lake trout, and 1.20 μ g/g in Millinocket Lake trout. Table 3-7 summarizes data from both of these studies and other studies in selected Maine lakes. Various theories have been proposed to explain the origin of mercury within project and nonproject waters. Mercury can originate both from natural sources, such as weathering of bedrock or flooding and subsequent leaching from topsoil, and from various industrial processes that release mercury-contaminated effluent (LURC, 1993). Atmospheric deposition also has been suggested as a primary source of mercury loadings.

3.5 FISHERIES RESOURCES

3.5.1 Regional

The Penobscot drainage once supported large spawning runs of Atlantic salmon, alewife, and American shad in addition to populations of resident species such as brook trout, lake trout, and blueback trout, which is an uncommon and unique salmonid found in several lakes in the basin. Baum (1983) estimated that, before 1800, the Penobscot salmon run ranged between 40,000 and 75,000 adult fish. No estimates of the size of the historic salmon run in the West Branch are available. Anadromous clupeids are reported to have traveled as much as 200 miles upstream from the mouth of the river (Atkins, 1887), including into the West Branch. These spawning runs decreased substantially as the basin was developed for timber-related industry. Both pollution and dam building contributed to the decline.

Table 3-7. Range and mean of mercury concentrations (μg/g) in sediments and fish tissues from project waters and selected Maine lakes. Data from GNP (1991b)						
	Sediments	Chain Pickerel ^(b)	Lake Trout ^(b)	Smallmouth Bass ^(b)		
East Millinocket impoundment ^(a,c)	<u>0.06 - 0.40</u> 0.29	<u>0.58 - 1.23</u> 0.87				
Dolby Pond ^(a,c)	<u>0.28 - 1.60</u> 0.86	<u>0.58 - 1.23</u> 0.87		<u>0.50 - 1.24</u> 0.91		
Mattamiscontis Lake		<u>0.34 - 0.93</u> 0.58				
Molunkus Lake		<u>0.35 - 1.32</u> 0.68		<u>0.34 - 1.28</u> 0.63		
North Twin impoundment ^(a)	<u>0.1 - 0.3</u> 0.22		<u>0.76 - 1.50</u> 1.09			
Millinocket Lake ^(a)	<u>0.1 - 0.2</u> 0.13		<u>0.69 - 3.20</u> 1.20			
Carr Pond			<u>0.43 - 0.78</u> 0.66			
Debsconeag Lake	<u>0.03 - 0.2</u> 0.14		<u>0.20 - 0.31</u> 0.27			
Schoodic Lake	<u>NA - NA</u> 0.55		<u>0.15 - 0.50</u> 0.28			
Sabathday Lake	<u>NA - NA</u> 0.59					
 ^(a) Project water ^(b) Fillets ^(c) Downstream of mill 						

Today, although the lower Penobscot is the site of an active restoration program for anadromous Atlantic salmon, the West Branch supports a high-quality fishery for landlocked Atlantic salmon (Baum, 1983). In addition, the West Branch contains a mixture of warmwater and coldwater species, depending on the local habitat conditions. Generally, warmwater species are found in the shallower impoundments, and coldwater species inhabit deeper impoundments and riverine reaches (Appendix B). Smallmouth bass and pickerel, both introduced to the drainage in the 1800's, are important fisheries, especially in shallower impoundments. Other species of recreational importance are lake trout, brook trout, burbot, white perch, various sunfish species, and smelt. The West Branch also supports a typical Maine assemblage of nongame species including creek chub, fallfish, longnose dace, blacknose dace, white sucker, and slimy sculpin. There are no anadromous fish in the West Branch because there are no fish passage facilities, and DIFW's management goals do not include restoring anadromous fish in this portion of the river basin. State and federal fisheries agencies have no current plans to restore anadromous fish to the West Branch but Interior has reserved its authority to prescribe fishways under Section 18 of the FPA in the future.

The following sections present life history information for key species.

3.5.1.1 Landlocked Atlantic Salmon

The life histories of landlocked salmon and anadromous salmon are similar except that landlocked subadults reside in freshwater impoundments rather than in the ocean while growing to sexual maturity. Spawning, nursery, resting, and holding habitat requirements are the same (FWS, 1989). This section describes the physical habitat requirements of Atlantic salmon as excerpted from the Final EIS for Atlantic Salmon Restoration in New England (FWS, 1989).

Spawning Habitat. Good spawning habitat contains sufficient gravel areas with substrate material of 0.5 to 4 inches in diameter (Peterson, 1978; Warner, 1963) to permit movement of well-oxygenated water through redds (nests). Free movement of water through the substrate is critical because salmon eggs may be deposited as deep as 12 inches (Warner, 1963). Water temperatures during the spawning period normally range from 45°F to 50°F (Jordan, 1981). Water temperatures lower than 50°F are desirable for normal egg development; 43°F is optimum (Peterson et al., 1977).

Salmon within the project areas originate from both hatchery stockings and natural reproduction. The Ripogenus Project area supports a completely self-sustaining population of landlocked salmon. Spawning occurs in seven of the tributaries to the impoundment, including the West Branch. Spawning occurs in the main stem and tributaries between McKay station and North Twin impoundment. This reach is unusual because the fish spend their entire life cycle in the riverine environment rather than moving seasonally from lake to river, as do most other known populations.

Most of the fish in the impoundments and riverine reaches within the Penobscot Mills Project area are of hatchery origin. Salmon populations in the North Twin impoundment and Millinocket Lake comprise 82 percent and 91 percent hatchery fish, respectively, based on data from creel censuses and routine field sampling. Wild fish in North Twin and Millinocket Lake are most likely from the West Branch or the tributaries of the impoundments, several of which contain documented spawning and nursery habitat. Downstream of North Twin, salmon occur in smaller numbers and are generally of hatchery origin. The West Branch between North Twin and Quakish Lake has a hatchery contribution of approximately 80 percent. There are no tributaries in this reach, although spawning is known to occur in the main stem. Dolby Pond, Millinocket Stream, Quakish Lake/Ferguson Pond, East Millinocket impoundment, and the remaining riverine reaches of the Penobscot Mills Project area support small populations of salmon, mostly of hatchery origin. Spawning has been documented in Millinocket Stream.

Nursery Habitat. Salmon nursery habitat is typically composed of shallow riffle areas interspersed with deeper riffles and pools. Substrate material ranging from 0.5 inches to more than 9 inches in diameter affords adequate cover for the juvenile salmon (Knight, 1981). Juvenile salmon will grow at water temperatures below 45°F (Symons, 1979), and growth is optimal in streams with daily peaks of 72°F to 77°F (Elson, 1975). Water temperatures higher than 83°F can be harmful to young salmon (Fry, 1947).

Resting and Holding Areas. Adult salmon prefer to rest in pools, which provide temporary refuge from swift currents. These pools usually lack cover and can be warmer than stream portions used as holding areas.

Holding areas, which are normally close to the spawning grounds, consist of pools with the cover, depth, temperature, and water velocities preferred by adult salmon. Pools with gravel substrate and water velocities of less than 1.6 feet per second are preferred (Frenette et al., 1972). Optimum water temperatures in adult holding areas are 50°F to 54°F, but temperatures of 60°F and daily fluctuations to 77°F are tolerated if the water cools to 68°F or less at night (Elson, 1975).

3.5.1.2 American Eel

The Penobscot drainage supports a commercial fishery for American eel, a catadromous fish. Adult eels from the Penobscot River, like all other eels produced along the east coast of the United States, migrate to the Sargasso Sea, where they spawn and die (Bertin 1956). Larvae produced in the Sargasso migrate to coastal regions, where they undergo a two-stage transformation into "glass eels" and then into elvers (immature eels). Elvers migrate into freshwater rivers and lakes. Upon reaching adulthood (8 to 13 years), female eels migrate great distances upstream. Males, however, usually remain in the coastal region for 5 to 20 years before returning to the Sargasso Sea.

Upstream passage for eels can be provided by modifications which are fairly simple, compared to upstream passage needs of salmon. The basic requirements are a pipe or flume of some type, with approximately 2 inches of water flowing through it. Often, an existing ice or trash sluiceway can be used. Synthetic bristles are placed within the fishway which makes it easier for the juvenile eels to ascend the fishway. Clay (1995) describes upstream eel fishways using this basic design concept which have been used successfully at dams as high as 68 meters.

Downstream passage for eels is usually achieved using fishways designed for other species such as salmon. However, any screens or bars which are used in front of the intakes, they need to be designed to exclude fish of the same size as an adult eel. The

passage facilities also need to be operating during the period when adult eels are migrating to spawn. Site-specific studies may be required to determine when this occurs although generally it is thought to occur in most rivers during late summer or fall (Flagg, undated).

Specific design parameters for eel fishways are not as critical as those for many other species because of the eel's natural ability to negotiate instream blockages. Eels are known, for example, to get past dams on their upstream migration by moving over land adjacent to rivers. This usually occurs at night while it's raining. The moisture from the wet grass, leaves, etc. allows the eels to survive out of water long enough to reach the forebay. Eels are also known to climb almost vertical faces of structures such as spillways, as long as the face is wet (Flagg, undated). These abilities help explain why eel populations persist in even heavily impounded rivers despite the lack of fish passage facilities. Often eels are the only migratory fish species which remains after other species such as shad, salmon, or striped bass have disappeared. Eels appear in the Ripogenus impoundment, more than 100 river miles upstream of Bangor and upstream of 11 operating hydroelectric projects. Large numbers of elvers also appear annually in the fishway at the Great Northern Paper Company's Mattaceunk Project (BHE, 1993c).

3.5.1.3 Other Species

The brook trout fishery in the West Branch declined in importance during the 1950's as the popularity of the landlocked salmon fishery increased (GNP, 1991a). Brook trout remain common, although not as abundant as salmon. GNP conducted electrofishing surveys in 1984 on six West Branch tributaries to estimate brook trout densities: Holbrook Stream, Carry Pond Brook, Harrington Hill Brook, Trout Brook, Rocky Pond Brook, and Fowler Brook. Density estimates averaged 21.4 trout (both young-of-year and adult) per 100 square yards of habitat (range: 0.9 to 41.0). GNP's annual salmon redd surveys provide additional information about brook trout distribution and reproductive status. GNP personnel have observed brook trout redds in the Horserace, upper Abol Deadwater, Trout Brook, Katahdin Stream, and Abol Stream.

Smelt that are landlocked in the West Branch spawn in tributaries during early spring. They migrate into streams in the evening and return to the lake the next morning for several consecutive days. Each day, the proportion of females in the run increases. Eggs hatch in approximately 7 to 10 days, depending on water temperature. Within the project area, smelt are found in all impoundments and provide important forage for landlocked salmon. GNP studied "smelt drift," or the movement of smelt from one project water to another via spillage or turbine entrainment, from 1984 to 1989. Smelt drift in the Ripogenus Project area averaged 28,085 pounds per year (GNP, 1991a). Smelt drift has been documented at all of the Penobscot Mills developments in tailrace drift-net sampling conducted by GNP from 1986 to 1989. Although highly variable, total smelt drift at Millinocket, Dolby, and East Millinocket is substantial during certain years. In 1986, for example, the total smelt drift at Dolby between March and December was estimated at 20,370 pounds.

Lake trout spawn during fall at depths between 6 inches and 15 feet. Eggs are deposited between rocks and are not covered with gravel, as are the eggs of most other salmonids. Impoundment draw-down during spawning periods can cause dewatering of eggs and poor reproductive success and is, therefore, of concern in project impoundments.

Smallmouth bass spawn during spring in the littoral zone of lakes and ponds and in low-velocity areas of rivers. Eggs are deposited in depressions, and the nest is guarded until the fry emerge and seek shelter in aquatic vegetation.

3.5.2 Site-specific

Fisheries resources in the project areas include those in impoundments and their tributaries, the West Branch of the Penobscot River and its tributaries, and bypass reaches. The following sections characterize fisheries resources in Ripogenus impoundment, Millinocket Lake, North Twin impoundment, Quakish Lake, Ferguson Pond, Dolby Pond, Upper Gorge, Millinocket Stream, Back Channel, and riverine sections of the West Branch of the Penobscot River, including habitat types and fish community composition (section 3.10 presents recreational fisheries data). Table 3-2 presents impoundment characteristics.

3.5.2.1 Ripogenus Impoundment

Habitat. Ripogenus impoundment is 20.8 miles long and includes Ripogenus Lake, Caribou Lake, Chesuncook Lake, and Black and Brandy ponds. The former three lakes are typically oligotrophic, with maximum depths of 150 feet, 84 feet, and 115 feet, respectively. In contrast, Black and Brandy ponds are shallower and mesotrophic; however, no bathymetric data for the ponds are available. The total area of the impoundment is approximately 29,270 acres. Assuming full-pool conditions and a lake volume of 710,000 acre-feet (GNP, 1991a), average depth is approximately 24 feet. Water level fluctuations, which depend on the project's mode of operation, affect both quality and quantity of littoral zone habitat.

Fish Community. The Ripogenus impoundment supports warmwater and coldwater species. From 1986 to 1989 Charles Ritzi Associates, consultants to GNP, studied the fish community in the project area, including Ripogenus impoundment. Ritzi collected 24 species of fish representing 14 families, including salmon, brook trout, lake whitefish, burbot, smelt, white perch, yellow perch, and several nongame species (GNP, 1991a; table 3-6). No threatened or endangered species were collected or observed during sampling.

3.5.2.2 West Branch

A STREET GARAGE

Habitat. The West Branch flows for 20.8 miles between McKay station and Ambajejus Lake (North Twin impoundment). This segment of the river and its tributaries provides habitat necessary for spawning, nursery, and adult life for a variety of species. The habitat is most suitable for salmonids, including landlocked Atlantic salmon and brook trout. According to DIFW and GNP, the limiting factor determining the abundance of salmon in the West Branch is young-of-year nursery habitat.

Fish Community. The West Branch fish community is nearly the same as that in Ripogenus impoundment, with the addition of six species that prefer riverine habitat, such as round whitefish and redbreast sunfish (Appendix B, table 2). Landlocked salmon is the dominant gamefish in this reach.

3.5.2.3 Upper Gorge

Habitat. Upper Gorge is a historical river channel that is bypassed and now receives minimal flows, except for spillage. The approximately 3,900-foot channel between Ripogenus dam and McKay station has an average gradient of 2.5 percent, with substrate of ledge, boulder, and cobble. Upper Gorge is bordered by steep, nearly sheer, rock walls throughout its length. Under existing flow management regimes (resulting in either very low or very high flows), the habitat is marginal; many of the pools provide suitable habitat for salmon or trout only at certain flows. Habitat surveys indicate that Upper Gorge has virtually no spawning habitat for salmon or trout, based on the presence of redds. Young-of-year trout are presumed to originate from tributaries (GNP, 1991a).

Fish Community. Upper Gorge supports approximately half of the species found in the West Branch (Appendix B, table 2) and contains no species unique to the project area. Brook trout and landlocked salmon inhabit Upper Gorge. The salmon probably enter Upper Gorge during periods of spillage.

3.5.2.4 North Twin Impoundment

Habitat. North Twin impoundment, which is also known as the Pemadumcook Chain of Lakes, is an oligotrophic, 17,790-acre lake comprising five separate lake basins (Ambajejus, Pemadumcook, North Twin, South Twin, and Elbow lakes). Except for Elbow Lake, which is relatively shallow, maximum depths of the basins range from 55 to 101 feet. Mean depth of the entire impoundment is 27.7 feet. As in other impoundments in the project area, water management causes seasonal draw-downs of up to 17 feet. Including the West Branch, the impoundment has 10 tributaries, which provide some spawning habitat for salmon, trout, and smelt. The shoreline of the impoundment is mostly large rocks and boulders and lacks large areas of sand or gravel substrate. Numerous islands and boulder reefs add to the quantity of fish habitat.

Fish Community. North Twin has a mixed coldwater and warmwater community including several game species and the typical Maine assemblage of minnows and suckers. Appendix B, table 1, lists all species found in the impoundment. Smelt provide the forage that supports salmon and lake trout. Salmon in North Twin are mostly stocked, whereas Ripogenus supports a self-sustaining population. Brook trout are found in the tributaries, but they are not abundant.

3.5.2.5 Millinocket Lake

Habitat. Millinocket Lake is similar to North Twin impoundment physically and biologically. Surface area is 8,640 acres at normal pool elevation, and the mean and maximum depths are 23.6 and 86 feet, respectively. Three streams enter Millinocket Lake, and the main outlet is at the dam, where water is released into Millinocket Stream. Additional outflow occurs at the pumping station into North Twin impoundment. Sandy Stream provides spawning habitat for both salmon and brook trout, but Mud Brook contains only brook trout spawning habitat.

Fish Community. Millinocket Lake supports a mixed warmwater/coldwater fish community that, except for salmon and lake trout, is self-sustaining. Appendix B, table 1, lists species in the impoundment. The fish community is similar to that in the North Twin and Ripogenus impoundments.

3.5.2.6 Quakish Lake/Ferguson Pond

Habitat. Quakish Lake and Ferguson Pond were formed in 1899 by the construction of Stone dam and a system of dikes. They have surface areas of 1,344 and 262 acres, respectively. Both are relatively shallow and productive compared to other lakes in the project areas. Quakish Lake's mean and maximum depths are 10.5 and 37 feet, whereas those of Ferguson Pond are 8.5 and 19 feet. Both lakes contain numerous stumps and submerged accumulations of pulpwood. Substrates are generally fine, with large areas of soft, organically rich bottom. The 1,600-foot section below the Stone dam gatehouse is a steep-banked riverine area approximately 5 feet deep with good riparian vegetation and adult salmon habitat.

Fish Community. The fish communities in Quakish Lake and Ferguson Pond are typical of Maine mesotrophic lakes and include warmwater and coldwater species. Salmon and lake trout are believed to originate from upstream stockings. See Appendix B, table 1, for a complete list of species. Pickerel and white perch are the dominant gamefish.

3.5.2.7 Dolby Pond

Habitat. Dolby Pond is a mesotrophic 2,048-acre impoundment with maximum depth of 46 feet and mean depth of 20 feet. It is separated from Upper Dolby Pond, which is relatively shallow, by State Route 157. The pond has three tributaries, including the West Branch. Stumps and large accumulations of pulpwood are common, and there are a variety of substrate types. Aquatic vegetation is abundant.

Fish Community. In addition to the typical New England assemblage of minnows and suckers, Dolby Pond supports a diverse community of warmwater and coldwater species. Appendix B, table 1, lists species found in the impoundment. Smallmouth bass, pickerel, white perch, salmon, and several species of sunfish, smelt, and yellow perch are found in Dolby Pond.

3.5.2.8 East Millinocket Impoundment

Habitat. East Millinocket impoundment is a mesotrophic, riverine impoundment with a surface area of 128 acres, mean depth of 11 feet, and maximum depth of 24 feet. The banks are steep and forested. The upstream portion of the impoundment is the Dolby tailrace, which has a relatively constant flow because Dolby dam is operated as run-of-river. Besides the West Branch, the impoundment has one other small tributary.

Fish Community. Because angler use is low, not much is known about the fish community in the impoundment except in the Dolby tailrace, where nearly all sampling has occurred. The fish fauna probably is similar to that in other riverine segments of the project area. Appendix B, table 1, lists the known species.

3.5.2.9 West Branch - North Twin Dam to Quakish Lake

Habitat. This section of the West Branch contains a variety of kinds of riverine habitat with a range of substrates, velocities, depths, and instream cover. There are no tributaries; however, there is some habitat for young-of-year, parr, and spawning in the main stem. Overall, habitat is best suited for adult salmon.

Fish Community. Appendix B, table 2, shows the fish species collected from this section of river. Generally, species richness is much less here than in upstream sections of the West Branch; however, salmon, brook trout, and several nongame species are present.

3.5.2.10 West Branch - Millinocket Development Tailrace to Dolby Pond

Habitat. This section is approximately 3 miles long and includes Shad Pond, a shallow, lacustrine area near the junction of Back Channel and the West Branch. No spawning has been documented, although some habitat is suitable for spawning. Lack of cover makes the reach poor habitat for young-of-year and parr. There are no tributaries.

Fish Community. Species composition is similar to that in the West Branch from North Twin to Quakish Lake (Appendix B, table 2).

3.5.2.11 Millinocket Stream

Habitat. This 7.9-mile stream section connects the Millinocket Lake dam with the Millinocket tailrace. Habitat is suitable for young-of-year and parr salmon and brook trout. Adult salmon habitat is limited by lack of deep pools. Little Smith Brook is the only tributary. Brook trout is the dominant gamefish in Millinocket Stream, and the population is supplemented with annual stockings.

Fish Community. This reach contains a mixture of warmwater and coldwater species, including smallmouth bass and brook trout (Appendix B, table 2).

3.5.2.12 Back Channel

Habitat. The Back Channel was the original river channel before it was diverted for log driving and hydroelectric generation. Currently, it receives leakage flows of 2 to 5 cfs, except during spillage, when it receives flows in excess of 29,000 cfs. Because of this flow regime, the Back Channel contains very little fish habitat, although potential habitat is extensive. GNP conducted an Instream Flow Incremental Methodology Study (IFIM) for the 4.5-mile-long the Back Channel in 1988 to estimate the flows necessary to provide habitat for various life stages of landlocked salmon. GNP considered flows ranging from leakage (2 to 5 cfs) to 2,000 cfs. The results indicated that total weighted usable area (WUA; a measure of habitat quantity) was maximized for all life stages except adult salmon at flows between 170 and 500 cfs. Adult salmon habitat increased incrementally up to 2,000 cfs to 2,000 cfs. Electrofishing surveys have documented the presence of small populations of brook trout where tributaries enter the Back Channel; there are three tributaries along this reach. Grand Falls, where an approximately 15 foot drop in stream bottom occurs, is located approximately

2.7 miles below Stone dam. This falls is a natural barrier to upstream migration of warmwater species, such as smallmouth bass and pickerel. It is also likely to be an upstream migration barrier to young life stages of salmonids, including landlocked Atlantic salmon, although it is probably passable by adult salmon at higher river flows. However, the actual flow threshold at which Grand Falls becomes negotiable to salmon is undocumented, and no run of landlocked salmon presently exists in the Back Channel.

Fish Community. Because of the limited amount of habitat under current flow conditions, the fish community in the Back Channel is limited (Appendix B, table 2).

3.5.2.13 East Millinocket Tailrace

Habitat. This 300-yard-long section is contiguous with the Medway impoundment (FERC Project No. 2666). The 3- to 5-foot depths, heavy substrate, and moderate-to-high velocity make it most suitable for adult salmon, although there are small amounts of habitat for young-of-year and parr.

Fish Community. Few surveys have been conducted within this reach because angler use is extremely low. According to GNP personnel, only one angler was observed between 1986 and 1989. Appendix B, table 2, lists species documented in this section of the river.

3.6 WETLANDS

3.6.1 Basinwide

About 30 percent of the total land area of Maine is estimated to have been wetlands before the 19th century. Current figures show that about 24.5 percent of the total area of the state is wetlands (Dahl, 1990). Since the beginning of the 19th century, wetlands throughout the Penobscot River Basin have been altered or inundated to some degree by large-scale projects such as dams for the logging industry, water storage, and hydroelectric generation. Predominant wetlands types within the Penobscot River Basin are palustrine emergent marsh (primarily bogs), palustrine scrub/shrub, and wetland floodplain forests (Tiner, 1984).

3.6.2 Ripogenus Project

More than 1,251 acres of wetlands border the Ripogenus impoundment (Appendix C, figures 1 through 3). Alder thickets constitute 118 acres, wet meadow/emergent marsh 282 acres, wet meadow/open water 66 acres, and open bog/wet meadow 785 acres. The areas of conifer swamp/wet forest and deep marsh/aquatic bed have not been measured or estimated. Wetlands surveys conducted by GNP indicate that these are generally discrete plant communities that are not entirely hydrologically dependent on the Ripogenus impoundment (GNP, 1991a). The following are brief descriptions of the principal wetlands types:

 Alder thickets, composed primarily of dense stands of speckled alder, occur as a discontinuous band along the shorelines of the Ripogenus impoundment and the deadwaters⁴ of the West Branch. (Appendix C, table 3, lists the plant species found in alder thickets in the vicinity of the Ripogenus Project.)

- Wet meadow/emergent marsh and wet meadow/open water occur along the shorelines of Ripogenus impoundment and West Branch deadwater, often in association with beaver dams. The most common species include grasses, sedges, and rushes. (Appendix C, table 5, lists the plant species found in the wet meadow/emergent marsh and wet meadow/open water habitat in the Ripogenus Project area.)
- Open bog/wet meadow occurs on the periphery of the impoundments and in association with streams and beaver ponds. The most abundant plants include the shrubs leatherleaf, Labrador tea, bog rosemary, and bog laurel. (Appendix C, table 6, lists plant species found in open bog/wet meadow habitat in the Ripogenus Project area.)
- Conifer swamp/wet forest is found primarily in the deadwater areas along the shorelines of the West Branch. Principal tree species include northern white cedar, black spruce, green ash, yellow birch, and red maple. The predominant shrubs are the overstory species interspersed with speckled alder, striped maple, and skunk currant. (Appendix C, table 4, lists the plant species found in the conifer swamp/wet forest in the Ripogenus Project area.)
- Deep marsh/aquatic bed is found in still waters of ponds and lake embayments with water depths of 2 to 10 feet, when sufficient fine substrates are present. Predominant plants include manna grass, pond lily, pondweed, burreed, and smartweed. (Appendix C, table 7, lists plant species found in deep marsh/aquatic bed habitat in the Ripogenus Project area.)

The following major systems represent wetlands around the Ripogenus impoundment:

- The 437-acre Brandy Pond wetlands system is located at the confluence of the Upper West Branch, Caucomgomoc Stream, and Umbazooksus Stream. It is a large open bog/wet meadow, grading outward (toward the Ripogenus impoundment) to wet meadow/emergent marsh.
- The Quaker Brook system is approximately 183 acres and is located north of Caribou Lake on the west side of Chesuncook Lake. The lower half of the system is wet meadow/emergent marsh along both banks of Quaker Brook, and it is inundated at normal high water level for part of the year. The upper half of the system is open bog/wet meadow, where surface water is present most of the year.
- The shoreline zone of the Ripogenus impoundment includes the area directly above the water line that is exposed during annual draw-downs of the impoundment. The substrate ranges from ledge and large boulders to smaller

Deadwaters are backwaters, coves, or other sheltered environments where wetlands are often present.

rocks and cobbles, with a few areas of coarse granular sand. Predominant plants within the shoreline zone are typical of those adapted to fluctuating water regime and disturbance. Many areas are dominated by "weedy" plant species. There are some scattered areas of submerged aquatic vegetation and emergent plants in the lower end of the dewatered zone. These areas are believed to be most affected by draw-downs.

 The Upper Gorge area between Ripogenus dam and McKay station contains predominantly very rocky substrate with high, steep-sided rock walls, which has restricted wetlands development. Low (leakage) flow from Ripogenus dam also has restricted wetlands development. Vegetation is very sparse in general in Upper Gorge.

Of the 528 acres of riparian wetlands bordering West Branch in the vicinity of the Ripogenus Project, most are associated with deadwaters. These deadwaters consist primarily of alder thickets, conifer swamp/wet forest, wet meadow/emergent marsh, wet meadow/open water, and open bog/wet meadow. These wetlands closely resemble those in unregulated lakes in remote parts of Maine (GNP, 1991a). The riparian zone wetlands between McKay station and Ambajejus Lake are characterized by alder thickets and open patches of grasses and sedges. Drier riparian sites at slightly higher elevations are dominated by hardwood and softwood mixed forests.

3.6.3 Penobscot Mills Project

Primary wetlands types found throughout the Penobscot Mills Project area include alder thickets, conifer swamp/wet forest, wet meadow/emergent marsh and wet meadow/open water, open bog/wet meadow, and deep marsh/aquatic bed. A total of 296 acres of wetlands border the North Twin impoundment (Appendix C, figure 4). Principal kinds of wetlands at North Twin are wet meadow/emergent marsh (180 acres), open bog/wet meadow (114 acres), and wet meadow/open water (2 acres). The three principal wetlands systems in the vicinity of North Twin are:

- White Horse Island wetland complex, which is wet meadow/emergent marsh, with open water east and south of the island.
- Stephenson's Landing wetlands, which consists of two distinct types, wet meadow/emergent marsh and open bog/wet meadow. The wetlands are flooded during high water.
- Impoundment shoreline wetlands, which are characterized by plants tolerant of fluctuating water levels during draw-downs and human disturbance. The substrate of the dewatered shoreline zone ranges from ledge and large boulders to smaller rocks and cobbles and coarse, granular sand.

Millinocket Lake encompasses about 709 acres of wetlands (Appendix C, figure 5). Wetlands types include wet meadow/emergent marsh (383 acres), open bog/wet meadow

(316 acres), alder thickets (8 acres), and wet meadow/open water (2 acres). There are three major wetland system's at Millinocket Lake:

- Pickerel Cove wetlands consists of two wet meadow/emergent marshes.
- The Grant Brook wetlands complex consists of open bog/wet meadow.
- Impoundment shoreline wetlands are characterized by plants found in constantly wet soils.

GNP identified 159 acres of wetlands, predominantly wet meadow/emergent marsh, at Quakish Lake (Appendix C, figure 6). These wetlands are primarily herbaceous with occasional scattered shrubs. Shoreline zones are absent around Quakish, Ferguson, Dolby, and East Millinocket impoundments because water levels are stable (GNP, 1991b).

GNP identified 219 acres of wetlands at Dolby Pond; most of these are wet meadow/emergent marsh (Appendix C, figure 7). These wetlands are predominantly herbaceous.

The East Millinocket impoundment area encompasses very few wetlands because of past development of the Millinocket Mill complex and the associated mill yards on its eastern shore (GNP, 1991b).

Riverine wetlands border Millinocket Stream and sections of the West Branch in the vicinity of the Penobscot Mills Project:

- Millinocket Stream has small areas of emergent marsh with scattered shrubs along the shoreline. Areas of emergent and scrub/shrub wetlands occur in several small deadwaters that have developed along the stream.
- Sections of the West Branch from North Twin impoundment to Quakish Lake and from the Millinocket tailrace to Dolby Pond contain the same wetlands types as Millinocket Stream. Shad Pond, which could be considered a portion of the West Branch, supports abundant emergent and deep marsh wetlands.
- The Back Channel consists of a main channel and an overflow channel. Five tributary streams flow into the main channel. The substrate of the main channel is predominantly large cobbles and boulders, but silt and sand have accumulated in a few sections, allowing development of small areas of wetlands. Narrow areas of riparian scrub/shrub wetlands occur in narrow margins along both the main channel and the overflow channel.

3.7 TERRESTRIAL RESOURCES

3.7.1 Basinwide

Approximately 95 percent of the Penobscot River Basin is forested (Pierce et al., 1993). Historically, the predominant vegetation type throughout northern Maine, including the

Penobscot River Basin, was spruce-fir forest. Currently, predominant vegetation type is spruce-fir forest, grading to pine-hardwood forests to the south and east (Pierce et al., 1993). Red spruce and balsam fir are the predominant trees in the spruce-fir forest. Principal trees in the pine-hardwood forest are oaks and white pine. In addition, a substantial portion of the basin is palustrine-forested, scrub/shrub, and emergent wetlands. Agricultural land also is found in parts of the northeastern section of the river basin (Cutting, 1959). The basin is relatively flat; average slope is less than 5 percent (Pierce et al., 1993).

There was little commercial use of northern Maine forests until the early-to-mid 1800's, when the first dams were built, and the late 1800's and early 1900's, when commercial logging began. Historically, forest-product companies have owned and used large tracts of forest to guarantee steady supplies of wood for their lumber mills (Harper et al., 1992). Large tracts of forest were cut heavily to supply the lumber mills; other tracts were lost through inundation by the dams. Another historic factor that reduced regional forests was the series of severe budworm epidemics between 1913 and 1919 (Pierce et al., 1993). In addition, much of the forested land in northern Maine (especially parcels on or adjacent to water) is currently under increasing commercial development pressure, especially from the vacation home industry (J. Sewall Co., 1993).

Wildlife species associated with the Penobscot River Basin include animals that live in the flood plain (e.g., moose, white-tailed deer, beaver, red-winged blackbird), feed in the water (e.g., waterfowl, turtles, frogs, otter, mink), feed along the shoreline (e.g., raccoon, fox, bear, shorebirds), live on or in the banks (e.g., otter, mink, muskrat), or feed in flight (e.g., swallow, swift, bats).

3.7.2 Ripogenus Project

The following sections describe terrestrial resources at the project boundaries⁵ and in the immediately surrounding area. Lands bordering the project are predominantly forested; hardwoods and softwoods occur in varying proportions. Softwoods are generally predominant; however, hardwoods predominate in some forests. GNP has not quantified these forest resources. Wetlands also occur within the project boundaries and the immediately surrounding area (see section 3.6). The project area encompasses one Maine State Registered Area Critical (RCA), Gero Island (see section 3.7.2.3). Wildlife of the Ripogenus Project area is typical of that found throughout northern Maine.

3.7.2.1 Softwood-dominated Mixed Forest

Softwood-dominated mixed forest is the most abundant vegetation type throughout the Ripogenus Project area. It occurs within scattered, large sections of the project area and the immediately surrounding area and is typically interspersed with somewhat smaller stands of hardwood-dominated forest. Predominant overstory trees in the softwood-dominated mixed forest are red spruce and balsam fir. White pine also is abundant, but does not occur as

⁵ In this resource section, the Ripogenus Project boundaries are the Ripogenus impoundment to the high water mark and immediately adjacent lands.

frequently. Subdominant trees in the softwood-dominated mixed forest include red maple, vellow birch, paper birch, and balsam poplar.

The shrub layer in the softwood-dominated mixed forest is composed predominantly of sapling trees of the overstory species. Other principal species within this layer include striped maple, sugar maple, shadbush, and skunk currant. The most predominant plants within the herbaceous layer in the softwood-dominated mixed forest are wood sorrel, bunchberry, and star flower. Other commonly observed herbaceous plants include goldthread, enchanter's nightshade, bristly wintergreen, and twinflower. (Appendix C, table 1, lists the softwood-dominated mixed forest project area.)

3.7.2.2 Hardwood-dominated Mixed Forest

Hardwood-dominated mixed forest is the subdominant vegetation type within the Ripogenus Project area and the immediately surrounding area. Hardwood-dominated forest generally occurs in more scattered discontinuous parcels and somewhat smaller stands than softwood-dominated forests. Within the hardwood-dominated parcels the predominant species vary according to amount of shading, density of cover, position on slopes, and proximity to impoundment shorelines.

In shaded areas around the impoundment shorelines at the Ripogenus Project, predominant trees are big toothed aspen, trembling aspen, paper birch, and yellow birch. Principal shrub species along the shores are speckled alder and willows. Hardwoods are much more predominant in the mixed stands on flat areas and on lower slopes away from the shoreline (particularly southern slopes). Principal species are beech, yellow birch, and paper birch. Other less prominent trees in flat areas and on lower slopes away from the shoreline include sugar maple, white ash, and red maple. Northern red oak is a subdominant species at scattered locations along the riparian zone of the West Branch.

The shrub layer consists of overstory regeneration; American beech and sugar maple are the most abundant species. Other predominant shrubs and trees within the shrub layer include striped maple, mountain maple, shadbush, beaked hazel, and witch hobble. Locations with this shrub layer mix indicate a typical regional northern hardwood climax stage (GNP, 1991a). Primary plants in the herbaceous layer of the hardwood-dominated mixed forest are oak fern, Solomon's seal, star flower, shinleaf, spinulose wood fern, and sarsaparilla. (Appendix C, table 2, lists the plant species of the hardwood-dominated mixed forest in the Ripogenus Project area.)

3.7.2.3 State-listed Critical Areas

Gero Island, which is located near the northern boundary of the Ripogenus Project at the northern end of Chesuncook Lake, is a Maine RCA (RCA 534). The island carries a stand of old growth white pine, which is considered rare in Maine and of scenic value. Maine's RCAs are considered to be of statewide significance.

Other areas of special interest that meet technical criteria for RCAs but that are not registered because of other considerations are called Qualified, but Unregistered Areas. The Ripogenus Project area encompasses four such areas: Ripogenus Gorge (PCA 39Q), the

Cribworks (PCA Q40), Big Ambejackmockamus Falls (PCA Q41), and Nesowadnehunk Falls (PCA 1Q). Ripogenus Gorge, located directly east of the Ripogenus dam, qualifies because of its geologic features, the presence of a rare plant, its scenic qualities, and its relatively undisturbed state. The Cribworks, Big Ambejackmockamus Falls, and Nesowadnehunk Falls are downstream of McKay station and qualify because of their scenic features (GNP, 1991a).

3.7.2.4 Wildlife

The Ripogenus Project is within the range⁶ of 48 mammal, 158 bird, and 24 reptile and amphibian species (Appendix C, tables 8 through 10). GNP's wildlife surveys documented 11 reptile and amphibian, 86 bird, and 19 mammal species in the Ripogenus Project area (GNP, 1991a).

Reptile and amphibian species diversity is relatively low in the project area (and in the region in general) due to the harsh, long winters typical of northern latitudes (45° 40' north). (Appendix C, table 10, lists species observed in the Ripogenus Project area during the GNP reptile and amphibian survey.) About 35 of the 86 bird species observed in the vicinity nest there; another 31 species probably nest nearby and forage in the project area, and about 20 species probably migrate through the project area (GNP, 1991a; Appendix C, table 9).

As indicated by GNP's mammal surveys, the highest density of small mammals (such as deer mice) occurs in the hardwood-dominated forest. Other common mammals are coyote, bobcat, black bear, white-tailed deer, and moose. In the softwood-dominated mixed forest and conifer swamp/wet forest, red squirrel, fisher, deer mouse, southern red-backed vole, northern flying squirrel, and snowshoe hare are the predominant mammals. Marten, coyote, bobcat, black bear, white-tailed deer, and moose are also found. The mammals observed most frequently in the riparian and impoundment shoreline habitats are moose and muskrat. In the wet meadow/emergent marsh, meadow jumping mice, muskrat, beaver, river otter, meadow vole, and red fox are typical species. In alder thickets, raccoons, meadow jumping mice, ermine, snowshoe hare, and white-tailed deer are typical. In open bog/wet meadow habitat, browsing and grazing animals such as moose, white-tailed deer, meadow vole, and southern bog lemming are typical. Deep marsh/aquatic bed habitat is used almost exclusively by beaver and muskrat and occasionally by moose (GNP, 1991a). (Appendix C, table 8, lists mammals observed in the Ripogenus Project area.)

3.7.3 Penobscot Mills Project

The following sections describe terrestrial resources within the project boundaries and the immediately surrounding area. Lands bordering the project are predominantly forested; hardwood-dominated and softwood-dominated forests occur in approximately equal proportions. GNP has not quantified these forest resources. Wetlands also occur within the project boundaries and the immediately surrounding area (see section 3.6). Wildlife of the Penobscot Mills Project is typical of that found throughout northern Maine.

⁶ A species' range is the geographic boundaries within which conditions are appropriate for the species.

3.7.3.1 Softwood-dominated Mixed Forest

In contrast to the Ripogenus Project, softwood-dominated and hardwood-dominated forests occur in approximately equal proportions within the Penobscot Mills Project area (GNP, 1991b). In general, hemlock and white pine are the predominant trees on flat, low-lying areas behind the impoundments; red pine and northern white cedar occur less frequently. As slopes and elevation increase, red spruce and balsam fir increase in abundance. Hardwood trees, including beech, paper birch, yellow birch, and sugar maple, occur in limited numbers in these areas. These hardwoods occur in much lower proportions than the softwoods in the softwood-dominated mixed forest (GNP, 1991b).

The shrub and herbaceous layer of the softwood-dominated mixed forest is similar to that found at the Ripogenus Project. (Appendix C, table 1, lists the softwood-dominated mixed forest species in the Penobscot Mills Project area.)

3.7.3.2 Hardwood-dominated Mixed Forest

In low, flat areas, shade-intolerant hardwood species such as trembling aspen, big toothed aspen, red maple, and paper birch predominate. Principal understory plants are lowbush blueberry, huckleberry, wintergreen, and bracken. Plant species composition in some areas of the project may be a function of the recent high incidence of forest fires in the project area (GNP, 1991b). All of these shade-intolerant species are early colonizers of burned areas (Barbour, 1980).

As slopes and elevation increase, beech, sugar maple, red maple, white ash, and northern red oak predominate. Less predominant trees on the slopes are paper birch and yellow birch, and softwood species including hemlock, balsam fir, and red spruce. The shrub and herbaceous layers of the hardwood-dominated mixed forest are similar to those at the Ripogenus Project. (Appendix C, table 2, lists the hardwood-dominated mixed forest species found in the Penobscot Mills Project area.)

3.7.3.3 Wildlife

The Penobscot Mills Project is within the range of 48 mammal, 158 bird, and 23 reptile and amphibian species (Appendix C, tables 8 through 10). GNP observed 9 reptile and amphibian, 80 bird, and 21 mammal species in the Penobscot Mills Project area during habitat-based surveys. GNP did not conduct population-based surveys.

Because of the harsh, long winters, reptile and amphibian species diversity is relatively low in the project area (and in the region in general). (Species observed in the vicinity of the Ripogenus Project during the GNP reptile and amphibian survey are shown in Appendix C, table 10.) About 32 of the 86 bird species observed in the area nest there; another 28 species probably nest nearby and forage in the project area, and about 20 species probably migrate through the area (GNP, 1991b; Appendix C, table 9).

Mammals found in the Penobscot Mills Project area are similar to those at the Ripogenus Project. (Appendix C, table 8, lists mammals observed in the Penobscot Mills Project area.)

3.8 THREATENED AND ENDANGERED SPECIES

3.8.1 Basinwide

The bald eagle (*Haliaeetus leucocephalus*), a federally and state-listed endangered species, breeds in many locations throughout the West Branch Basin. As of 1991, there were 123 known nesting pairs of bald eagles in Maine (Welch, 1991). The West Branch Basin is one of the most important nesting and wintering areas for bald eagles in Maine (MSPO, 1987).

3.8.2 Ripogenus Project

In a letter dated February 12, 1990, FWS stated that the bald eagle is the only federally listed species known to occur in the Ripogenus Project area, but that the ranges of the long-tailed shrew (*Sorex dispar*), a C3 candidate for federal listing,⁷ and North American lynx (*Felis lynx canadensis*), a C2 candidate for federal listing,⁸ overlap the project area. GNP indicated that neither species was observed during surveys.

As of 1992, there were four known active bald eagle nests in the Ripogenus Project area; in 1993 there were three active nests.⁹ The nests are in the northeastern, southwestern, and southern-most portions (along the West Branch) of the project. Other eagle pairs occupy territories in the area but are not nesting.

Common loon (*Gavia immer*), a bird species of concern in Maine, breeds primarily in the Chesuncook Lake portion of the Ripogenus impoundment. Other adult loons with chicks were observed in Brandy Pond during bird surveys conducted on June 15, 1987, (GNP, 1991a). Ring-billed gull (*Larus delawarensis*), considered an uncommon breeding species for the project area, and common tern (*Sterna hirundo*), cited in Maine's List of Threatened and Endangered Animals as a species of special concern, both breed on Gull Island and Tern Ledge, located near the northern end of Chesuncook Lake (GNP, 1991a).

In a letter dated February 21, 1995, FWS indicated that several additional species of concern (that were not considered in the DEIS) may occur in the vicinity of the projects and that they should be considered in the FEIS. These species, all federal C2 candidates, include yellow lampmussel (*Lampsilis cariosa*), brook floater (*Alasmidonta varicosa*), extra striped snaketail dragonfly (*Ophiogomphus anomalus*), and midget snaketail dragonfly (*Ophiogomphus howeii*). The extra striped snaketail dragonfly and the midget snaketail dragonfly have also been proposed as state listed threatened and endangered species, respectively. The FWS indicated that exuvia (exoskeletons shed during molting) of midget

⁷ Animals listed under the C3 category have proven to be more abundant or widespread than previously believed, or are not subject to any identifiable threat. In this context, the Commission did not consider this species important in its review of the project.

⁸ C2 indicates that federal listing as threatened or endangered is possibly appropriate, but conclusive data on biological vulnerability and threat are not currently available to support such listing.

⁹ Data for two of the 1992 eagle nest sites were not available for 1993.

snaketail dragonfly were observed in the West Branch in 1994. FWS stated that information pertaining to the distribution of all these species is severely limited.

No federally listed threatened or endangered plant species are known to occur within the Ripogenus Project area; however, two plants on Maine's official list of rare plants, purple clematis (*Clematis occidentalis*) and northern woodsia (*Woodsia alpina*), occupy the banks of the West Branch between McKay station and Ambajejus Lake (GNP, 1991a). A stand of old growth pine, which is considered rare in Maine, is located on Gero Island in Chesuncook Lake.

3.8.3 Penobscot Mills Project

As of 1992, there were two known active bald eagle nests in the Penobscot Mills Project area. During 1993 there also were two active nests; however, only one of the two nests was the same during both years. The nests are located between the North Twin Development and Stone dam and to the west of the Dolby Development. Other eagle pairs occupy territories in the area but are not nesting.

Common loon breeds in the North Twin impoundment, Millinocket Lake, and Quakish Lake. It also could breed in Dolby Pond and in other areas of the Penobscot Mills Project (GNP, 1991b).

Orono sedge (*Carex oronensis*), a C2 species, was observed a considerable distance from the North Twin impoundment shoreline during a botanical survey conducted by GNP, but it was never found within the limits of the Penobscot Mills Project (GNP, 1991b).

3.9 RIVER AND LAND MANAGEMENT PLANS

The state has developed several river and land management plans for the lower Penobscot River Basin. Some plans are considered "comprehensive plans" under Section 10 (a)(2) of the FPA.

3.9.1 Maine Rivers Study

The Maine Rivers Study (DOC, 1982) is an inventory of natural, economic, and recreational resources of the state rivers of Maine. This study identified the river stretches in Maine that stand out as remarkable from statewide and regional perspectives. The significant resource values of the West Branch are geologic, ecologic, inland fishery, scenic, boating, historic, and canoe touring. The study notes that the West Branch is eligible for inclusion in the National System for Wild and Scenic Rivers and that the West Branch is one of the state's highest quality fishery resources.

3.9.2 State of Maine Comprehensive Rivers Management Plan

The three-volume Maine Comprehensive Rivers Management Plan (MSPO, 1987) is a compilation of plans, parts of plans, state laws, executive orders, and maps produced by a variety of state agencies and elected officials. Volume 1 is an executive order mandating designation of river sections that merit special protection, a report on the projected

contribution of hydroelectric generation to meeting Maine's electricity needs in 1990 and 2000, and a statewide fisheries plan that specifies river-by-river fishery considerations and fish passage needs at existing and proposed hydroelectric generating facilities. Volume 2 contains the 1982 Maine Rivers Study. Volume 3 provides a discussion of the laws, implementing orders and plans, and river-specific plans necessary to implement a statewide Maine Rivers Policy.

3.9.3 Statewide Comprehensive Outdoor Recreation Plan

The Maine Statewide Comprehensive Outdoor Recreation Planning Program (SCORP; MDOC, 1988, 1993) is the basis for a 5-year plan for managing Maine recreation resources. The SCORP contains an inventory of current recreation resources, a demand forecast, and discussion of policies. The SCORP divides the state into geographic regions to identify regional differences in use based on the origin of users. The West Branch is in the Katahdin/Moosehead travel region of Maine, where the most popular recreational activities are fishing, camping, and hunting. The SCORP recognizes the need in the region for boat and fishing access, canoe access, family camping, and primitive camping.

3.9.4 Maine Strategic Plans, Vol. II, Inland Fisheries

This plan established goals to (1) maintain optimum population levels of freshwater fishes and associated aquatic species; (2) maintain optimum quality, quantity, and diversity of habitat; and (3) provide for optimum and diverse uses of freshwater fishes for sportfishing, aesthetic, economic, ecologic, scientific, and educational purposes. The plan further states the specific objectives of the Maine DIFW: (1) to provide for a projected demand of approximately 436,700 licensed and unlicensed anglers and 2.6 million angler-days, (2) to provide for a combined harvest of approximately 3.1 million fish of all species from lakes and 674,000 fish of coldwater species from streams, and (3) to maintain fishing quality for the major gamefish species at approximately current levels.

3.9.5 Maine Statewide River Fisheries Management Plan

The DIFW prepared this plan for The Governor's Cabinet Committee on Hydropower Policy. Goals and objectives are general: (1) maintain optimum population levels of freshwater fishes and associated aquatic species; (2) maintain optimum quality, quantity, and diversity of habitat; and (3) provide for optimum and diverse use of freshwater fishes for sportfishing, aesthetic, economic, scientific, and educational purposes. The plan also states that the Maine DIFW intends to review and act on proposed dam projects on a case-by-case basis.

3.10 RECREATION RESOURCES

3.10.1 Regional

The project areas are located in the Katahdin/Moosehead Travel Region, the largest of the eight travel regions in Maine (figure 3-5). This area is rich in natural resources and offers abundant outdoor recreation opportunities. A disproportionately large percentage of total state park land (79 percent), including Baxter State Park (202,064 acres), is within this region.



Figure 3-5. Travel regions in Maine showing location of Penobscot Mills Project area (Source: SCORP, 1988)

Mount Katahdin and other peaks within Baxter State Park are visible from much of the project area, which is only a few miles from the park boundary. The West Branch of the Penobscot River and other recreation facilities within the project areas provide additional recreation opportunities for Baxter State Park visitors. Much of the land there is owned by GNP, which allows public access for recreation.

3.10.2 National Designations

The West Branch of the Penobscot River within the project boundaries has not been designated as a National Wild and Scenic River, although it is eligible and included on the National Rivers Inventory maintained by Interior. The state of Maine has not requested federal designation of the river as a wild, scenic, or recreational river.

A short section of the Appalachian National Scenic Trail, a part of the National Trails System, crosses the project area on the western edge of Pemadumcook Lake. In 1990, GNP granted a conservation easement to the state of Maine to protect the entire Appalachian Trail within GNP's land holdings. A short segment of the Maine Interconnecting Trail System (ITS) is located along the northern shore of Pemadumcook Lake, a popular snowmobile route.

No National Wilderness Areas or National Natural Landmarks are within the project areas or would be affected by the proposed projects.

3.10.3 Existing Recreation Facilities/Opportunities in the Project Region

The project areas offer a wide array of private and commercial recreation opportunities. The kinds and extent of development along the shores and on the islands of the project impoundments are varied. Residential and recreation development is a function of the accessibility of the impoundment, the natural limitations on construction because of wetlands and soil types, and the zoning districts established by LURC.

In the past, GNP awarded leases to current and retired employees for use as private camps. GNP terminated this practice in the early 1970's and has not issued new leases since that time (GNP, 1991a, 1991b). The approximately 900 existing leases are transferable and renewable annually. Nearly 80 percent of these are concentrated along the shoreline of the North Twin impoundment (Ambajejus, South Twin, and North Twin lakes). Most leased sites are developed with summer cottages, and many have floating or fixed docks.

Eighteen commercial facilities exist throughout the project area, and most are located on land leased from GNP (1991a, 1991b). Most of these are camps offering swimming, boat rental, guide services, and convenience goods.

Public and private recreational facilities include 6 public boat launches, 4 private boat launches, 4 public boat put-ins, 2 informal boat put-ins, 8 permitted campsites (which require a fire permit from the Maine Bureau of Forestry), 15 authorized campsites (which have been approved by the landowner, the Maine Bureau of Parks and Recreation, and the Bureau of Forestry and do not require a fire permit), and several formal and informal picnic areas.

The six project impoundments total nearly 60,000 acres of water surface. The 29,270acre impoundment created by Ripogenus dam is the second largest lake in the state. All of the impoundments except East Millinocket (128 acres) are available for recreation.

The West Branch between Ripogenus dam and Ambajejus Lake is heavily used for whitewater boating and fishing. In cooperation with commercial rafting outfitters, GNP provided put-in facilities at McKay station (including a gravel parking area for outfitter buses, a gravel raft-staging area, a generator/compressor to inflate the rafts, and two portable toilets), five launch sites along the river, and take-out facilities at Never's Corner. Other informal river access points for canoes, kayaks, and fishing exist throughout the area. GNP recently modified the security fence at McKay station to allow 24-hour pedestrian access.

GNP owns hundreds of miles of current and former logging roads surrounding the project areas, which are open to the public and are used by hikers, hunters, cross-country skiers, bird watchers, and others. A popular trail follows the eastern rim of Upper Gorge from Ripogenus dam to below McKay station.

Table 3-8 summarizes the recreation facilities and opportunities available at each project.

Several public and private roads provide access to the project area, including State Route 157, which links East Millinocket and Millinocket and provides access to Dolby Pond; State Route 11, which runs southwest from Millinocket and serves the Twin, Elbow, and Quakish lakes; State Road, which provides access to Millinocket and Ambajejus lakes and Baxter State Park; and Golden Road, GNP's privately owned haul road that is used primarily for transporting cut timber to the mills but also provides access to the Ripogenus Development and the Upper West Branch area.

3.10.4 Current Recreational Use

Current recreational uses of the project areas include fishing, swimming, boating, hunting, ice fishing, camping, hiking, snowmobiling, canoeing, cross-country skiing, waterskiing, sailing, and sightseeing. Most land-based recreation occurs on nonproject lands owned by GNP.

3.10.4.1 Ripogenus Project

The Ripogenus Project area is part of GNP's West Branch District, a 1.2-million-acre commercial forest. Access to the district is controlled through three checkpoints: Debsconeag, Sias Hill, and Twenty-Mile. Approximately 55 to 60 percent of visitors enter the district through the Debsconeag checkpoint, the closest access point to Interstate 95 and Millinocket (GNP, 1991a). Sias Hill accounts for 25 to 30 percent of visitors, and Twenty-Mile accounts for approximately 15 percent. Figure 3-6 summarizes recreational- use trends for the GNP's West Branch District.

Since 1985, use of the West Branch District has remained relatively constant at 150,000 visitors per year, but the number of visitor-days dropped in 1987 after GNP instituted access fees. According to GNP's gate data, most visitors come in July and August, or during

Table 3-8. Recreation facilities summary (Source: Staff)							
Lakes	Surface Area	Recreationa I Use	Activities	Facilities	# Private Leases	# Commer- cial Leases	Access Points
RIPOGENUS PROJECT - Ripogenus Lake - Caribou Lake - Chesuncook Lake	29,270 acres	Light	Swimming Fishing Ice Fishing Snowmobiling X-C Skiing Boating Canoeing Camping Sightseeing	- 18 campsites	60	3	2 Public Boat Launches • Chesuncook Dam Point • Umbazooksus Stream 1 Private Boat Launch • Allagash Gateway Campsite 1 Informal Boat Launch • Western Shore of Caribou Lake
PENOBSCOT MILLS PROJECT NORTH TWIN • N. Twin Lake • S. Twin Lake • Pemadumcook Lake • Ambajejus Lake • Elbow Lake	17,790 acres	Heavy	Fishing Boating Waterskiing Swimming Snowmobiling Ice Fishing Canoeing X-C Skiing	S. Twin Lake Picnic Area · Ambajejus Lake Beach · 3 campsites	703	4	 2 Public Boat Launches Ambajejus Lake Partridge Cove (S. Twin) 2 Private Boat Launches Bartons Marina North Woods Trading Post 1 Public Boat Put-in Norcross (Elbow Lake)
MILLINOCKET LAKE	8,640 acres	Heavy	Fishing Boating Waterskiing Swimming Snowrnobiling Ice Fishing Canoeing X-C Skiing	2 campsites	120	7	2 Public Boat Launches Millinocket Lake Millinocket Dam 1 Private Boat Launch Robinson Twin Pines Camp 1 Informal Boat Put-in Near Robinson Twin Pines Camp
MILLINOCKET · Quakish Lake · Ferguson Pond	1,606 acres	Light	Fishing Canoeing Ice Fishing		0	0	1 Public Boat Put-in • Green Bridge
DOLBY POND	2,048 acres	Light	Fishing Canoeing Ice Fishing	Dolby Picnic Area	0	0	2 Public Boat Put-ins · Dead Man's Curve (Jerry Brook) · Rt. 157 (Dolby Flowage)
E. MILLINOCKET	128 acres	Restricted	None		0	0	None





3-36

the prime fishing and hunting months of June and November (GNP, 1991a). Figure 3-7 summarizes summer activity for 1986, the last year that GNP asked visitors to specify the purpose of their visits when registering at the checkpoint. According to 1987 gate records, 65 percent of visitors to the West Branch District were from Maine.

GNP charges fees for accessing Ripogenus Project lands and other nonproject areas between May 1 and November 30. These fees defray GNP's cost to maintain use of its land for public recreational use (GNP, 1991a). An \$8-per-vehicle or \$48-per-season access fee for nonresidents and a \$4-per-vehicle or \$24-per-season fee for Maine residents allows nearly unlimited access to all of GNP's West Branch District. GNP also charges commercial rafting outfitters a license-use fee of \$4 per rafting customer and a \$3 access fee per customer.

GNP estimates approximately 4,000 to 5,000 angler-days of fishing on the Ripogenus impoundment during the April 1 to September 30 open-water fishing season, 2500 to 4000 angler-days during the winter ice fishing season (January 1 to March 31), and 9900 camping days (GNP, 1991a). In addition to the salmon fishery, the impoundment also supports important fisheries for lake trout, burbot, and lake whitefish.

Current operations of the Ripogenus impoundment result in annual water level fluctuations of as much as 44 feet. Since 1972, the maximum draw-down during the summer recreation period (between Memorial Day and Labor Day) has been 15.25 feet; the average has been 3.5 feet.

In 1987, GNP surveyed residents of Millinocket, East Millinocket, and Medway regarding their recreational use of the project area (GNP, 1991a). Nearly 30 percent of the respondents reported that the Ripogenus Project area is the most important location for their outdoor recreation; only 5 percent felt that the area is overused.

Camping, fishing, whitewater boating, and sightseeing are popular along the West Branch. The area experiences approximately 47,000 camping days each year. The West Branch also is recognized as one of the world's premier landlocked salmon fisheries, and GNP estimates that approximately 5,900 to 8,000 angler-days occur along the river between April 1 and September 30 (GNP, 1991a). Fishing quality is high at 0.41 fish per angler-day (state goal is 0.2). The average size of creeled salmon is 18.5 inches and 2.25 pounds (state goals are 17 inches and 1.75 pounds). According to 1986 GNP gate data, fishing accounted for the highest rate of visitor days (27 percent) for the West Branch District. The West Branch and its tributaries support a brook trout fishery with average annual catches of 584 fish.

The West Branch is one of the most challenging whitewater boating rivers in the eastern United States. It contains 1 of only 2 significant Class V rapids in New England and 5 of only 40 rapids identified by the Maine Critical Area Program as having statewide significance. This river is suitable for expert level kayaking, rafting, and guided canoe touring (with portages). In Maine, the commercial boating value of the West Branch is second only to the Kennebec River (DOC, 1982).

The popularity of commercial whitewater rafting has increased enormously since it began in the project area in 1977; however, after peaking in 1985 at nearly 19,000, the number of customers has averaged approximately 18,600 and declined to 17,962 in 1989 and



Figure 3-7. Summary of summer activity within GNP's West Branch District

3-38

16,441 in 1990 (GNP, 1991a). In 1985, the Maine legislature established a maximum of 560 commercial passengers per day on the West Branch.

In 1990, GNP surveyed commercial whitewater rafting customers and private whitewater boaters who used the West Branch below Ripogenus dam regarding the quality of recreation experience and spending patterns (GNP, 1991a). Nearly all of the whitewater rafting customers (97 percent) were very or extremely satisfied with their experience. Eightyfive percent stated they would definitely return, and 100 percent stated that they would recommend rafting on the Penobscot River to others. Quality of whitewater and safety were the most important factors affecting a rafter's choice of boating locations.

More than 90 percent of whitewater boaters were very or extremely satisfied with their experience; 93 percent stated they would definitely return, and 98 percent stated that they would recommend the West Branch to other paddlers. Paddlers preferred flows above 3,000 cfs and below 2,000 cfs. Respondents paddling at intermediate flows (2,000 to 2,600 cfs) preferred higher, more predictable flows. Many paddlers who responded as somewhat, not very, or not at all satisfied with the flow levels had frequent boating experience on the West Branch, averaging 43 trips.

The upper Penobscot River also provides one of the best opportunities for multiday wilderness cance trips in the eastern United States. The 83-mile route from Seboornook Lake to Millinocket is a popular trip. Some suggest that Upper Gorge may offer whitewater boating opportunities. Upper Gorge currently receives only leakage (12 cfs) and occasional spillage flows.

The Maine Department of Conservation (DOC) states that the limited development in the project area; the extensive, high-quality natural resources; the existing recreational facilities; and the amount of public use of the river make the Ripogenus Project area one of the most important recreational areas in Maine (letter from C.W. Ten Broeck, DOC, November 5, 1990).

3.10.4.2 Penobscot Mills Project

In 1987, GNP surveyed year-round residents, leaseholders, and visitors to the Penobscot Mills Project area to assess recreational use patterns (GNP, 1991b). The survey showed that more than 80 percent of the recreational visitors are Maine residents, and nearly 60 percent of the total are local residents. More than 50 percent of the local residents identify the Penobscot Mills Project area as their principal outdoor recreation area and believe the impoundments receive "about the right amount of use."

Current operations of the Penobscot Mills Development result in annual water level fluctuations of as much as 22 feet at the North Twin impoundment (North Twin, South Twin, Pemadumcook, and Elbow lakes) and 6.2 feet at Millinocket Lake. Since 1972, the North Twin impoundment has sustained a maximum draw-down of 9.7 feet during the summer recreation period; average draw-down was 1.5 feet. During the same time, Millinocket Lake experienced a maximum draw-down of 3.4 feet and an average draw-down of 0.6 feet. Drawdowns expose boaters and waterskiers to navigation hazards, increase the distance to the water for swimmers, and make use of private docks and public boat launches (i.e., Barton's) more difficult.

The Millinocket, Dolby, and East Millinocket developments operate in run-of-river mode; maintenance and flashboard failure cause minimal water level fluctuations.

Several entities suggest that Millinocket Stream and the Back Channel may offer whitewater boating opportunities (letter from D. Sosland, CI, February 29, 1992). The "AMC River Guide" describes the 7.8-mile section of Millinocket Stream from Millinocket dam to the bridge at Millinocket as containing Class II rapids and quickwater. Currently, the stream receives a minimum flow of 20 cfs and is navigable only during spillage periods. The "AMC River Guide" describes the 4.5-mile the Back Channel as offering Class III and Class IV rapids mixed with flatwater. The Back Channel is navigable during spillage periods. Recreational fisheries in the Penobscot Mills Project area include both coldwater and warmwater species. The landlocked salmon fisheries of North Twin impoundment and Millinocket Lake receive approximately 5,000 and 2,400 angler-days of use, respectively. Anglers also target lake trout, burbot, and lake whitefish. Estimated use of Dolby Pond ranges between 1,200 and 1,800 angler-days annually, directed primarily at smallmouth bass, chain pickerel, and white perch. Millinocket Stream is stocked annually with brook trout and contains smallmouth bass and some salmon that drop down from Millinocket Lake. Quakish Lake, Ferguson Pond, and the riverine sections within the project area also receive some fishing pressure.

3.10.5 Future Demand for Recreation

The 1988 SCORP evaluated the adequacy of 16 kinds of outdoor recreation activities by travel region and projected deficiencies in the Katahdin/Moosehead Travel Region for boating/fish access, canoe access, and family and primitive camping. These needs are regional, not necessarily project-specific. DOC stated that existing recreational facilities within the project area, including water access, are adequate to meet current demand.

The 1993 SCORP also projected statewide trends over the next 5 to 10 years for various outdoor recreational activities including:

- Moderate-growth activities (0.9 to 3.0 percent increase in annual user days)
 - canceing and kayaking
 - cross country skiing
 - lake and pond fishing
 - hunting
- Small-to no-growth activities (0.9 percent increase to 0.9 decrease in total annual user days)
 - primitive camping
 - river and stream fishing
 - hiking
 - ice fishing
 - snowmobiling

The 1993 SCORP recognizes that recreational use in the northern Maine Woods is increasing. Visitor days have increased a total of 26.5 percent since 1980, although annual growth rates have slowed to 1.2 percent since 1987. Hunting, camping, hiking, and rafting visits are the major contributors to this growth.

3.10.6 Access for People with Disabilities

No specially designed access to recreational facilities is available for people with disabilities.

3.11 LAND USE ISSUES

3.11.1 Regional

The Ripogenus Project and Penobscot Mills Project areas lie within a remote, forested region of northern Maine. Commercial forestry operations are the predominant land use, and the small towns in the region developed around past and present forestry operations.

GNP owns roughly 2.1 million acres of woodland in northern Maine, much of which lies within the watershed of the West Branch of the Penobscot River. GNP owns in fee or holds flowage rights to all the project lands and most land surrounding the project area. Development within the region, primarily by GNP, includes forestry-related access roads, paper mills, and hydropower facilities. Recreation development consists of private and commercial recreational facilities, predominantly lakefront cabins and seasonal camps leased by GNP (1991a, 1991b).

In 1986, LURC evaluated more than 1,500 lakes larger than 10 acres as part of the Maine Wildland Lake Assessment. The assessment rated the value of seven resources: fish and wildlife, scenic quality, shoreline character, botanic features, cultural and historic resources, and physical features (hydrology and geology). LURC combined these ratings to obtain a cumulative resource value and established three resource classes: lakes of statewide significance, lakes of regional significance, and lakes of local or unknown significance. Although the overall project area includes only 22 percent (337 lakes) of the rated lakes, LURC classified seven lakes within the Penobscot Mills Project area and six lakes within the Ripogenus Project area as lakes of statewide significance (table 3-9). The remaining lake in the Penobscot Mills Project area and two lakes within the Ripogenus Project area are rated as having regional significance. LURC did not classify any of the lakes within the project region as having local or unknown significance (Land & Water Assoc., 1993).

3.11.1.1 Maine Land Use Regulation Commission

LURC, which was created in 1969, is responsible for planning, zoning, and land-use regulation for Maine's unorganized townships. Most of the Ripogenus Project and Penobscot Mills Project areas are within unorganized townships.

Lake/Pond Maine Wildlande	Lake Momt	Development		Protection			Municipal		Land Management			
	Assessment Significance	nt Class e	D-GN	D-CI	D-RS	P-GP	P-AL	P-RP	P-SL	M-SL	M-IND	Agency
Ripogenus											,	
Caribou Lake	Statewide	3			35%	65%						ML
Chesuncook Lake	Statewide	2				j –	83%		17%			ML, MB
Debsconeag D.	Statewide	1						100%				
Ripogenus Lake	Regional	2	20%				80%					ML
TOTAL RIP	OGENUS		3%		5%	10%	60%	14%	8%			
Penobscot Mills					·							
Ambajejus L.	Statewide	3,5	7%		53%	40%				Ĩ		ML
Dolby Pond	Statewide	7		7%		65%				28%		ML. EM, MI
East Millinocket	Not Rated	NR							50%		50%	ML, EM
Elbow Lake	Statewide	3				100%						ML
Ferguson P.	Regional	7							ŀ	100%		MI
Millinocket L.	Statewide	7	1%		12%	87%						ML
N. Twin L.	Statewide	3			33%	67%						
Pemadumcook L.	Statewide	3			10%	90%						ML
Quakish	Statewide	7				100%						ML
S. Twin L.	Statewide	3,5			15%	85%	Į			700		
Shad P.	Not Rated					30%	<u> </u>			70%		
TOTAL PE	NOBSCOT		1%	1%	13%	72%			2%	10%	2%	<u></u>
TOTAL COMBINED		2%	1%	11%	52%	20%	3%	5%	4%	4%		

LURC established three resource-based zoning districts to ensure compatibility of future development with existing land use and natural resources. The zoning districts include Protection, Development, and General Management Districts:

- <u>Protection Districts</u> (P) are areas in which development would jeopardize unusual or fragile natural resources.
- <u>Development Districts</u> (D) are designated areas of existing residential, commercial, industrial, or recreational development where future compatible development is encouraged.
- <u>General Management Districts</u> (M) are existing and recommended areas of commercial forest-product or agricultural use (LURC, 1991).

Each of the three broad zoning districts is divided into subdistricts with specific landuse standards. Two protection subdistricts, Great Pond (P-GP) and Accessible Lakes (P-AL), and one management subdistrict, General Management (M-GN), encompass most of the land area within and adjacent to the Ripogenus Project and the Penobscot Mills Project areas. Other subdistricts within the project areas include development: General Development (D-GN), Residential Development (D-RS), and Commercial Development (D-CI); and protection: Resource Plan (P-RP) and Shoreland Plan (P-SL). Portions of the project areas are within the towns of Millinocket and East Millinocket and subject to local zoning regulations (MUN-SL and MUN-IND). Table 3-10 summarizes the zoning regulations for the Ripogenus Project and Penobscot Mills Project areas.

General land use standards apply to all subdistricts, unless otherwise specified within a particular subdistrict, and include shoreline frontage specifications, minimal setbacks, maximum building height, and vegetative buffers. A minimum of 200 feet of shoreline frontage per dwelling unit is required for residential development along lakes and ponds; 300 feet of shoreline frontage is required for commercial, industrial, or other land uses. A minimum of 150 feet of shoreline frontage per residential dwelling unit is required along streams; 200 feet for commercial, industrial, or other land uses.

LURC requires a minimum setback of 100 feet from the shoreline of lakes and ponds and 75 feet from stream or river shorelines. Structures built within 500 feet of the normal high water mark can be no higher than existing screening vegetation or 25 feet, whichever is greater. Vegetative buffers must be maintained within 75 feet of the high water mark of streams, and 100 feet of the high water mark of lakes and ponds (LURC, 1991).

Protection districts encompass nearly 75 percent of the total lake shoreline in the region (table 3-9). P-GP, which extends 250 feet from the normal high water mark of the designated lakes, encompasses most of the land within both project areas. This subdistrict is used to regulate development and land use to protect the recreation potential, fishery habitat, and scenic character of the designated area.

P-AL is found only in the Ripogenus Project area, along most of the shoreline of Chesuncook Lake (about 80 percent) and Ripogenus Lake (about 80 percent). The P-AL

Table 3-10. Summary of zoning classifications (Source: LURC, 1991)								
Zone	Linear Feet Shoreline Per Structure	Set ba ck	Vegetativ e Buffer	Timber Harvest	Residential Dwellings	Subdiv./ Multi. Unit	Camp- grounds	Perm & Non- Perm Docks/ Boat Ramps
M-GN D-CI D-GN D-RS MUN-SL P-RP P-AL P-GP ^(a) P-SL1 ^(a) &SL2	N/A 300' 200'/300' ^(b) 200' 100 1 mile 200'/300' 150'/200'	N/A 150' 100'/150' 100' 75' 500' 100' 100' 100'/150' 75'/100'	N/A 100' 100' 75' 50' 100' 100' 75'	Yes P P Std/P P Std Std	PN PP PP P	Х Р Р Р Р Р Р Р Р Р Р Р Р Р Р Р Р Р Р Р	P P No SP SP SP SP	N/A P/Yes SP/Yes SP/Yes SP/Yes SP/Yes SP/Yes
 ^(a) Timber harvesting regulations prohibit clear cutting within 50 feet of the high water mark. ^(b) Includes both residential development/commercial and other development. MUN-SL - extends 75' of highwater mark P - permit P-AL - extends 500' of highwater mark (Class 2 lakes) P-GP - extends 250' of highwater mark P-RP - extends 500' of highwater mark P-SL1 - extends 250' of highwater mark P-SL2 - extends 75' of highwater mark 								

.

subdistrict extends 500 feet from the normal high water mark. In this zone, single-family detached dwellings are allowed by permit at the rate of one dwelling unit per shoreline mile.

P-RP provides the means for more efficient and effective management of single or multiple protection subdistricts. The P-RP subdistrict permits landowners to develop a resource management plan for a land area. The resource management plan must follow standards developed by LURC and be submitted for review and approval for designation. Upon approval by LURC, land use activities are allowed in accordance with the plan. The P-RP subdistrict applies to areas along the West Branch of the Penobscot River, from Ripogenus dam to Ambajejus Lake within the Ripogenus Project area. The conservation easement there is a required component of the resource management plan. P-SL applies to Chesuncook Lake (about 17 percent) and East Millinocket (about 50 percent).

The M-GN subdistrict encompasses a few small areas within the project region, and most of the land adjacent to it. The M-GN subdistrict areas permit forestry and agricultural management activities. LURC (1991) decided that land areas within this district do not require the special protection provided by the protection subdistricts.

Development districts constitute approximately 15 percent of the total area of lake shoreline in the combined project region. D-GN, which recognizes existing development and encourages compatible development within and adjacent to these areas, is found along previously developed portions of Ripogenus and Ambajejus lakes. D-RS, which encourages concentration of residential development within and adjacent to existing residential areas, is found along Caribou Lake in the Ripogenus Project area and Ambajejus, Millinocket, and Twin lakes in the Penobscot Mills Project area. D-RS requires clustering residential development to protect shorelines along lakes classified as Class 3 (Ambajejus and South Twin). D-CI is found only along a small portion of Dolby Pond.

LURC timber harvesting regulations for P-SL1 and P-GP protection subdistricts include:

- no clearcutting within 50 feet of the normal high water mark, and harvesting activities within this area must maintain a well-distributed stand of trees to protect the aesthetic and recreational value and water quality of the area;
- at distances between 50 feet and 250 feet, harvesting activities may not create single openings greater than 14,000 square feet, and single canopy openings of over 10,000 square feet may not be closer than 100 feet apart; and
- harvesting within 250 feet of the normal high water mark may not remove more than 40 percent of the volume on each acre in a 10-year period.

For P-AL districts the harvesting provisions for 50 to 250 feet described above would apply to the full 500-foot protection zone.

LURC (1991) and DOC (1992) require maintaining unscarified filter strips between the exposed mineral soil and the normal high water mark of surface water areas as indicated below:

Average Slope of Land Between Exposed Mineral Soil and Normal High Water Mark (percent)	Width of Strip Between Exposed Mineral Soil and Normal High Water Mark (feet along surface of the ground)
0	25
10	45
20	65
30	85
40	105
50	125
60	145
70	165

In 1990 LURC established seven management classes based partially on the ratings developed in the Wildlands Lakes Assessment to provide protection plans and development guidelines for designated lakes (LURC, 1990a, 1990b). LURC incorporated some of these guidelines into the land use districts and standards zoning regulations.

Four of the seven management classes apply to the lakes and ponds within the project region. Debsonceag Deadwaters is rated Class 1, and the guidelines include prohibiting development within one-quarter mile of shorelines and restricting permanent vehicular access. Chesuncook and Ripogenus lakes are rated Class 2, and guidelines include restricting density to one dwelling unit per mile of shoreline for areas within 500 feet of the shoreline. LURC incorporated these guidelines into the P-AL subdistrict.

Class 3 designation applies to most lakes and ponds within the Penobscot Mills Project area, including North Twin Lake, Elbow Lake, Pemadumcook Lake, portions of Ambajejus Lake and South Twin Lake, and to Caribou Lake in the Ripogenus Project area. Guidelines for this classification include conserving natural resources while supporting responsible development.

Developed portions of Ambajejus and South Twin lakes are rated Class 5, and guidelines include measures to maintain natural qualities and enhance scenic values. Cluster development is required, except where inappropriate due to site characteristics. Finally, guidelines for Class 7 lakes (Dolby Pond, Ferguson Pond, Millinocket Lake, and Quakish Lake) involve managing lakes for multiple uses, including resource conservation, recreation, and timber production.

3.11.1.2 Shoreland Zoning Act

The Shoreland Zoning Act of 1971 requires municipalities to establish land-use controls for designated shoreland areas. This act protects water quality, aquatic and wildlife habitats, and historical and cultural resources; conserves natural resources; preserves open space; and anticipates and responds to development impacts within shoreland areas.

The towns of Millinocket and East Millinocket encompass approximately 10 percent of the lake shoreline in the project region. These towns adopted zoning regulations that conform

to requirements stated in the Shoreland Zoning Act, including a shoreline zone (MUN-SL) and an industrial zone (MUN-IND). Designated shoreland areas within the project region include sections of Millinocket Lake, Shad Pond, Dolby Pond, and the Back Channel. The areas extend 250 feet from the high water mark of designated lakes and ponds, and 75 feet from the high water mark of designated streams.

Land use controls provided in the Shoreland Zoning Act include a minimum lot area and minimum 100-foot shoreline frontage; structure setbacks of 75 feet; clearing limitations, including a minimum of 75 feet of vegetative buffer along the shoreline; timber harvesting limitations; erosion and sedimentation control; sewage disposal; and provisions for nonconforming uses. The primary land use controls applied within the project region include vegetative buffers and use of selective-cut timbering methods along the stream channels (DEP, 1992).

3.11.1.3 Natural Resources Protection Act (NRPA)

The NRPA establishes a regulatory permitting process to prevent degradation and encourage protection and enhancement of natural resources. Activities that require a permit include removal and displacement of soil, sand, vegetation and other materials; draining; filling; and construction, repair, or alteration of any permanent structure. Permits are processed and regulated by the Maine DEP.

The proposed activity must meet environmental standards set by this legislation. These standards maintain that the proposed activities must not interfere with scenic, aesthetic, recreation, or navigational uses; cause unreasonable soil erosion or sedimentation; unreasonably harm vegetation, aquatic species, wildlife, or habitat; interfere with natural water flow; lower water quality; cause an increase in flooding; or cross river segments identified as outstanding (unless with specified provisions; DEP, 1993a, 1993b).

3.11.1.4 Maine Forest Practices Act (MFPA)

The MFPA provides a consistent and comprehensive perspective on the role of Maine's vast forest resources. It is a catalyst to encourage and promote sustained-yield management and use of forests and related resources. Many acres of forest in Maine are privately owned, and the act provides a means to regulate timber harvesting.

Timber harvest regulations include standards for clear cutting (areas over 50 acres require a forest management plan); regeneration standards (within 5 years of completing timber harvest); and notification before harvest. Forest management and harvest plans must be updated every 10 years and prepared by a licensed professional forester. These plans must outline activities to regenerate, improve, and harvest standing timber crops. Forest management and harvest plans also must include locations of water bodies and wildlife habitats identified by the DIFW.

Failure to comply with regulations leads to fines enforced by state, county, or municipal law enforcement officers. Municipalities considering adopting new ordinances to regulate timber harvest must consult with the Bureau of Forestry (State of Maine, 1989).
3.11.2 Site-specific

3.11.2.1 Ripogenus Project

The primary land uses within the Ripogenus Project area are forestry, water storage, and hydropower facilities. A few small recreational developments occur along the impoundment shorelines. GNP granted approximately 60 leases for private camps and commercial campsites on Ripogenus, Chesuncook, and Caribou lakes.

In 1981, GNP donated a conservation easement defined in the Resource Protection Plan for the Penobscot Waterway (LURC, 1981) starting 400 feet below Ripogenus dam and continuing down to the inlet at Ambajejus Lake. The conservation easement incorporates the lands wholly owned by GNP within 500 feet of the normal high water mark of each side of the Penobscot River measured as a horizontal distance landward of such high water mark, including islands lying within the Penobscot River (LURC, 1981).

The conservation easement established under the LURC P-RP zone prohibits residential and commercial development within 500 feet of the shoreline. Timber harvesting practices within the easement must follow LURC standard regulations requiring written notification of LURC before any harvesting. The Maine Bureau of Parks and Recreation regulates the resource management and recreational use of lands within the easement and on Gero Island in Chesuncook Lake (GNP, 1993b).

LURC regulates most land within both project areas, except for land within Millinocket and East Millinocket. Both towns are developing and updating comprehensive plans.

3.11.2.2 Penobscot Mills Project

Primary land uses within the Penobscot Mills Project area include hydroelectric facilities, impoundments, forestry, pulp and paper industry, and recreation. Most impoundment shoreline consists of undeveloped woodlands and several developed areas. Both private and commercial development are concentrated along Ambajejus, South Twin, North Twin, and Millinocket lakes.

GNP awarded leases to current and retired employees but has issued no new leases since the early 1970's. All existing leases are transferable and renewable annually. In the Penobscot Project area, GNP has granted approximately 823 leases for private camps, commercial camping, guiding services, and boat launches on Ambajejus, South Twin, North Twin, Pemadumcook, and Millinocket lakes.

Land uses in Millinocket and East Millinocket include residential, commercial, industrial, institutional, and rural/undeveloped land. GNP owns most of the undeveloped land within the town boundaries; primary uses are open space and a landfill (East Millinocket, 1993).

3.11.3 Proposed Land Uses

Future land use within the project region is expected to remain similar to current use. The three land regulatory agencies within the project region (LURC, Millinocket, and East Millinocket) establish goals to protect natural resources and accommodate reasonable growth and development. Future land use issues (pertinent to the project region) anticipated by LURC continue current issues related to river protection, lake protection, forestry regulations, and development.

In its draft comprehensive plan, East Millinocket proposes to implement more structured land use regulations. The proposed districts would include more specific shoreline zoning related to resource protection and land use categories. The town of Millinocket stated land use policies in its draft comprehensive plan: "to regulate, through land use zoning, site development review, permitting, and quality tax records, all future growth and development of the town, recognizing all locational limitations, the local character, and economic importance of the land area" (Millinocket, 1992).

Projected land use development was determined in the Northern Forest Lands Study (USDA Forest Service and Governor's Task Force on Northern Lands, 1990). Large tracts of private forest may remain, but they probably will be concentrated away from accessible lakes and public roads. Forest with the highest value for recreation development is the most vulnerable to changes in land use, including lake shore, river frontage, scenic vistas, and access to alpine ski areas.

The study also indicated that changes in the land base will further encourage forest industry to pursue land development projects. Lakeshores in all but the most inaccessible regions of the Northern Forest probably will be developed, with cottages and year-round homes dotting the shorelines (FS, 1990). Land with lake frontage is in the highest demand. In a 3-year period, 50 percent of all applications for permits to subdivide land involved lakeshore property (FS, 1990).

3.12 AESTHETIC RESOURCES

3.12.1 Regional

The aesthetic resources of the region include diverse water, land-form, and vegetative patterns. The area provides numerous opportunities for viewing wildlife in vast forested lands of diverse vegetation. Distant views include Mount Katahdin and other peaks within Baxter State Park. Visual resources related to water include the flatwater expanses of many lakes and the West Branch of the Penobscot River.

As part of the Maine Wildland Lakes Assessment, LURC (1990b) conducted the Scenic Lakes Evaluation in Maine's unorganized towns using a multistep procedure to identify lakes with the greatest scenic value. The rating criteria included elevation changes in areas surrounding the lakes, shoreline configuration, foreground and background relief, vegetation diversity along the shoreline, presence of inharmonious development, and presence of special features (such as extreme water clarity or wildlife viewing). Four lakes within the project areas were rated as having "outstanding scenic significance," and two were rated as being "significant."

Structures within both project areas are primarily limited to seasonal camp buildings and project structures, including dams, powerhouses, pumping stations, and paper mills. Several national historic structures, such as the Ambajejus Boom House and Chesuncook Village, are noted historic and visual landmarks of the region.

Shoreline development within the region consists primarily of seasonal private and commercial camps. The type and extent of development vary considerably (Dewan & Assoc., 1990a, 1990b). The shorelines of the most heavily developed area are irregular with many inlets and peninsulas.

Forestry clear cutting practices often have visual impacts within the region. LURC and the MFPA regulate forestry practices within the Ripogenus Project and Penobscot Mills Project areas. Timber harvesting regulations define the size and location of allowable clear cutting and provide measures for limiting the visual impacts related to forestry practices.

3.12.2 Site-specific

3.12.2.1 Ripogenus Project

Significant visual elements within the Ripogenus Project area include the Ripogenus impoundment, Ripogenus dam, McKay station, and the transmission line from McKay station to Millinocket. Ripogenus dam creates a 29,270-acre impoundment consisting of three lakes and three ponds. The shorelines of these lakes and ponds are largely undeveloped, and the primary uses of the surrounding land are forestry (timber production and transportation) and recreation.

The transmission line between McKay station and Millinocket is supported primarily by wooden structures that blend into the surrounding forest. The Ripogenus dam itself is visually significant within the region. Sightseers are drawn to the dam to view its massive size and engineering accomplishments. The impoundment created by the Ripogenus dam provides many views of lakes and shorelines. None of the lakes within the Ripogenus Project area have scenic ratings; however, LURC did identify Ripogenus Lake and Chesuncook Lake as warranting further evaluation (Dewan & Assoc., 1990b).

Annual draw-down levels for the Ripogenus impoundment averaged 16.5 feet and ranged up to 30.6 feet during the period of record for GNP's flow analysis (1970 to 1985). Draw-downs expose the shoreline.

Development along the impoundment occurs in three areas: Chesuncook dam, Chesuncook Village, and along the western shoreline of Caribou Lake. Development consists of private and commercial camps, most of which are seasonal camps on sites leased from GNP. Structures are diverse and located primarily along the shorelines.

Upper Gorge is also an important scenic destination. Stretches of Upper Gorge contain sheer vertical walls that rise 75 feet above the riverbed. Motorists can view the West Branch from points along Golden Road from Abol Ridge to McKay station (Dewan & Assoc., 1990b).

3.12.2.2 Penobscot Mills Project

Significant visual resources of the Penobscot Mills Project include the five impoundments and associated structures. Most of the project structures are positioned in inconspicuous locations and are screened from major public highways. The principal access to the project area, State Route 11, offers periodic views of the lakes, ponds, and surrounding distant hills (Dewan & Assoc., 1990a).

All lakes in the North Twin impoundment (except Ambajejus Lake) are designated as "outstanding scenic resources" because of numerous islands, shorelines and beaches, diverse vegetation, and views of Mount Katahdin. Ambajejus Lake and Millinocket Lake are rated as "significant scenic resources." No other lakes within the Penobscot Mills Project area are rated for scenic resources value (Dewan & Assoc., 1990a).

The two most developed areas within the project area are the south shore of South Twin Lake and the eastern end of North Twin Lake. The typical development is seasonal camps on land leased from GNP. The typical camp is a one-story structure of modest design and often includes outside storage areas for recreational equipment, household items, or firewood (Dewan & Assoc., 1990a).

GNP controls flow in the West Branch. The primary visual influence of flow management relates to the visual character of several impoundment shorelines and bypass reaches. Project operations cause water levels in Millinocket Lake and the North Twin impoundment to fluctuate. Draw-downs result in seasonal fluctuations in shoreline elevations of up to 6.2 feet (up to 3.4 feet in summer) for Millinocket Lake and up to 22 feet (up to 9.7 feet in summer) for North Twin impoundment. Draw-down of impoundment levels increases shoreline exposure, typically of gravel and boulders.

Diversion of water flow from the bypass reaches exposes rocky streambeds. Current flows in the Back Channel, the bypass reach from Stone dam to Shad Pond, average 2 to 5 cfs, and flows in Millinocket Stream from Millinocket Lake dam down to the town of Millinocket average 20 cfs.

3.13 CULTURAL RESOURCES

3.13.1 Regional

3.13.1.1 Prehistoric

A few site discoveries indicate that Paleo-Indians camped and hunted large game within the region, specifically within the Ripogenus Project and Penobscot Mills Project areas, around 9000 to 7000 B.C. Evidence of the archaic period occupations, typically hunter-gatherer groups, is more common than evidence of the Paleo-Indian in the Penobscot drainage. These populations range from the early archaic (ca. 7000 to 5500 B.C.), to middle archaic (ca. 5500 to 4000 B.C.), and late archaic (4000 to 1000 B.C.). Late archaic period sites, the most common sites identified in the project region, are found along the lower Penobscot River (GNP, 1991a, 1991b). Woodland era populations (ca. 1000 B.C. to 1550 A.D.) occupied the entire Penobscot River drainage basin and nearby waterways. The woodland culture of this era continued earlier hunter-gatherer patterns of subsistence. The Penobscot Indians, a riverine tribe, resided along the streams and lakes within the project area. The Penobscot Indians, whose population numbered as many as 3,300 in the early 1600's, were reduced to 200 people by the late 1700's. Various aboriginal groups remained widespread in the region until the late 18th century, when they gathered along the main stretch of the lower Penobscot River (GNP, 1991a, 1991b).

3.13.1.2 Historic

Euro-American occupation began with settlement of the Penobscot River valley after the resolution of the French and Indian War in the 1760's. Before 1828, activities within the region focused on exploration and trapping. Demand for timber led to a number of exploratory surveys during the late 18th and early 19th centuries. Joseph Treat surveyed the Penobscot Mills area for land and timber resources in 1820 under an agreement with the state of Maine. In 1829, Thomas Fowler, Sr. established the first permanent Euro-American settlement in the local area (GNP, 1991a, 1991b).

3.13.1.3 Pulp and Paper Industry

The first large-scale lumbering operation within the region began in 1828 along the West Branch near present-day East Millinocket. Primary settlements along the West Branch consisted of lumber camps and small outposts. Timber companies constructed log-driving dams at narrow or steep falls to improve transport of harvested logs to mills downstream. The Chesuncook dam, built in 1840 at the outlet of Chesuncook Lake, was one of the earliest dams in the region (GNP, 1991a, 1991b).

In 1846, the Maine legislature chartered the Penobscot Log Driving Company (PLD) to lessen the cost and difficulty of small drives within the region. PLD contracted with numerous companies to construct and improve dams, canals, sluices, and booms. Construction included approximately 137 dams on the West Branch and surrounding tributaries.

GNP, founded in 1898, quickly became the largest lumbering company in the region. Originally, the company owned 338,000 acres, predominantly within the region of the West Branch of the Penobscot River. GNP expanded its land holding to include more than 2.1 million acres; contracted with hundreds of loggers; and quickly became a great, regional economic influence (Rolde, 1990).

Millinocket mill, constructed in 1899, was the first mill in the region and the largest newsprint mill in the United States. By 1900, the population of Millinocket rose to 2000 people, and the town incorporated on March 16, 1901. GNP built a second mill, the East Millinocket mill, in 1906, and by 1907 the town of East Millinocket incorporated. Hardy S. Fergusen, recognized as one of the world's foremost pulp and paper mill engineers, designed Millinocket mill, East Millinocket mill, and Ripogenus dam (GNP, 1991a, 1991b).

3.13.2 Site-specific

3.13.2.1 Ripogenus Project

GNP built several dams within the Ripogenus Project area to meet increasing needs for water storage and power for the mills. The Chesuncook dam, built in the 1840's, was the earliest. Subsequent dams built to increase water storage capacity included a small dam (1865) downstream from Chesuncook dam, and a timber crib dam (1887) upstream of the Ripogenus dryway dam.

Ripogenus dam, built between 1915 and 1916, significantly increased the water storage capacity and size of the impoundment, submerging the former Chesuncook dam. Ripogenus dam has remained virtually unchanged since that time. In 1950, GNP built McKay station at the lower end of Ripogenus Gorge (GNP, 1991a).

National Register and Significant Sites. Phase I and Phase II archeological surveys identified 130 aboriginal sites; 73 were attributed to known periods of prehistory and history, and at least 36 sites were considered potentially eligible for the National Register (GNP, 1991a). Subsequent research and consultation with the SHPO led to a final recommendation that 15 sites be considered for Phase III mitigation (GNP, 1992b).

Chesuncook Village is listed in the National Register of Historic Places, recognizing its importance as a frontier logging settlement. The village incorporates the original log shanty constructed in 1849, the Chesuncook House built in 1863, and a boom house constructed at the head of Chesuncook Lake in 1911. GNP moved the boom house and converted it to a church in 1923.

Ripogenus dam, a significant achievement in controlling water resources in Maine, is eligible for the National Register of Historic Places. A draft National Register nomination completed in 1992 specifies the contributions of Ripogenus dam to the broad patterns of history, significant engineering, and industrial development in the Penobscot River Valley (GNP, 1991a).

Overall Significance. The significance of the sites identified within the project region relates to both regional and local aboriginal history and prehistory. Past studies in Maine focused upon coastal locations, limiting the archeological surveys of inland riverine areas and populations. Archeological research for the Ripogenus and Penobscot Mills and projects provides valuable information for formulating models of interior aboriginal adaptations over time. This research is significant beyond local applications and contributes to regional studies throughout Maine (GNP, 1991a, 1991b).

3.13.2.2 Penobscot Mills Project

The Penobscot Mills Project consists of four developments along the West Branch and one located at the outlet of Millinocket Lake. North Twin dam and Millinocket Lake dam are on sites of previous dams. Dams built during mill construction, Stone dam (1899), North Twin dam (1903), and East Millinocket dam (1906), remain primarily unchanged. The Dolby Development was constructed in 1906 because the East Millinocket impoundment, which was originally intended to be larger, was inadequate. Dolby station originally served as a combination pulp mill and generating station and was converted from a hydromechanical to hydroelectric station in 1925. The dam created Dolby Pond and has remained unchanged since construction (GNP, 1991b). The Millinocket Development includes Millinocket Lake dam, built in 1910, and a pumping station and pumphouse built in 1950.

National Register and Significant Sites. Preliminary archeological surveys conducted during 1981 and 1985 indicated that the West Branch area below McKay station had been an occasional prehistoric route; there was no evidence of permanent or long-term settlement. Aboriginal populations occupied the Penobscot River drainage during the late Paleo-Indian period and throughout later periods. Artifacts discovered include lithic, ceramic, and organic remains (GNP, 1991b).

The University of Maine at Farmington Research Center conducted further archeological surveys and testing during 1987 and 1988 to identify sites within the project region that may be eligible for listing on the National Register of Historic Places. A total of 124 aboriginal sites were identified and verified; 78 sites were attributed to known periods of prehistory and history, and at least 25 sites were considered potentially eligible for the National Register. Subsequent research and consultation with the Maine SHPO led to a final recommendation that seven sites be considered for mitigation (GNP, 1992a).

The Ambajejus Boom House, built in 1907 at the head of Ambajejus Lake, is one of the most significant remaining buildings of the lumbering culture of the West Branch. The Ambajejus Boom House is listed in the National Register, recognizing the importance of logging operations on the West Branch (GNP, 1991b).

3.14 SOCIOECONOMIC RESOURCES

The Ripogenus and Penobscot Mills projects affect the socioeconomics of the north central Maine counties of Penobscot and Piscataquis and the incorporated towns of Millinocket and East Millinocket. In comments received during scoping, intervenors, residents, and businesses in this region all indicated that continued operation and economic vitality of the GNP paper mills is essential to provide employment and municipal tax revenues that support community services.

3.14.1 Demographic Conditions

3.14.1.1 Population

Although population in the region grew modestly (approximately 7 percent) between 1980 and 1990 (table 3-11), the population of the Millinocket urban area (including Millinocket, East Millinocket, and Medway) decreased by 6.5 percent. The decrease is largely the result of worsening economic conditions and declining GNP employment levels (Millinocket, 1992). This represents the second largest loss in population in Maine during the 1980's.

Total Population	1980	1990	Rate of Change (%) 1980-1990
Penobscot County	137.015	146,601 ^(a)	7.0
Piscataguis County	17,634	18,653 ^(b)	5.8
Millinocket	7,567	6,959 ^(a)	-8.1
E. Millinocket	2,372	2,166	-8.7
Medway	1,871	1,922	2.7

^(b) Bureau of the Census, 1980, 1990a.

The number of housing units grew moderately in Penobscot County and Millinocket during the 1980's, whereas Piscataquis County experienced tremendous growth in new housing because of the recreation homes established primarily by Maine residents in the scenic lakes region of upper Piscataquis County (table 3-12).

Table 3-12. Hous	 Housing units in Penobscot and Piscataquis counties and Millinocket/E.								
Millin	Millinocket, 1980-1990 (Source: Bureau of the Census, 1980, 1990a)								
	Housing Units	Housing Units	Rate of Change						
	(1980)	(1990)	1980-1990						
Penobscot County	49,416	61,359	24.1						
Piscataquis County	7,109	13,194	85.0						
Millinocket	2,715	2,867	5.5						
E. Millinocket	NR	846	NA						

3.14.1.2 Employment

A CASE OF A CASE

Manufacturing accounts for about 16 percent of the employment in Penobscot and Piscataquis Counties (table 3-13). Major employers in the project region are (Millinocket, 1992):

- Bowater/GNP (1,037 employees);
- Millinocket Regional Hospital (205 employees);
- Millinocket Machine and Foundry Company (20 employees); and
- Bangor and Aroostook Railroad (75 employees).

GNP employs approximately 2,000 people in the Millinocket area, which represents about one-third of all jobs in the area. Using the Maine Department of Labor employment multiplier of 1.61, GNP supports more than 3,200 jobs in the area. GNP pays more than \$121 million in wages/salaries/benefits within Maine, much of which is concentrated in the Millinocket area. Since 1990, the labor force has increased modestly across the state

	Maine Employment by Sector	Percent by Sector	Penobscot County Employment by Sector	Percent by Sector	Piscataquis County Employment by Sector	Percent by Sector
Farm	12,065	1.8%	910	1.1%	236	2.9%
Ag. Serv., Forestry & Fish.	11,184	1.6%	801	1.0%	130	1.6%
Mining	305	0.0%	30	0.0%	D	
Construction	38,923	5.7%	3,768	4.6%	286	3.5%
Manufacturing	98,8 93	14.6%	11,909	14.6%	2,449	30.4%
Transportation and public utilities	27,023	4.0%	4,548	5.6%	419	5.2%
Wholesale trade	26,466	3.9%	3,533	4.3%	D	
Retail trade	130,400	19.2%	16,120	19.8%	1,417	17.6%
Finance, insurance, and real estate	39,292	5.8%	3,247	4.0%	274	3.4%
Services	186,507	27.5%	21,682	26.6%	1,592	19.7%
Government						
Federal, civilian	16,159	2.4%	1,178	1.4%	62	0.8%
Military	14,049	2.1%	847	1.0%	103	1.3%
State and local	77,627	11.4%	12,800	15.7%	1,093	13.6%
Total	678,893	100.0%	81,373	100.0%	8,061	100.0%

~

and in the project region. During that period, however, unemployment in Millinocket remained consistently above the state average, often by several percentage points (table 3-14). GNP announced plans to eliminate 200 additional positions during 1994.

3.14.1.3 Income

Traditionally, Millinocket has had one of the highest income levels in the state, in large part because of the high wages paid by GNP (table 3-15). The 1990 labor contract for the mills established only minor increases in wages for the next 10 years. Declining employment opportunities and incomes may erode the town's ability to provide community services (Millinocket, 1992).

3.14.2 Municipal Services and Government Revenue

Millinocket and East Millinocket provide most public services in the project region, including schools, police, fire and emergency services, and recreation. School officials expect a slight decline in enrollment through 1997; consequently, space in the schools is sufficient.

East Millinocket's total revenue for 1991 was \$5.7 million; Millinocket's total revenue was \$13.16 million. Property taxes were 65 percent and 63 percent of the total revenue of the towns, respectively.

Table 3-14.	Labor force participation in Maine and Penobscot and Piscataquis counties, 1990-92 (Source: Maine Department of Labor Statistical Handbook, 1990, 1991, 1992)								
		Labor Force	Number Employed	Number Unemployed	Unemployment Rate				
1992 - Maine		662,000	615,000	47,000	7.1				
Penobscot Co) <i>.</i>	71,820	66,520	5,300	7.4				
Piscataquis C	0.	8,860	8,110	760	8.5				
Millinocket		4,390	3,990	400	9.1				
1991 - Maine		647,000	598,000	49,000	7.5				
Penobscot Co).	70,980	65,320	5,660	8.0				
Piscataquis C	0.	8,730	7,910	830	9.5				
Millinocket		4,240	3,810	430	10.2				
1990 - Maine		635,000	603,000	33,000	5.1				
Penobscot Co).	69,540	65,810	3,740	5.4				
Piscataquis C	0.	8,590	8,030	570	6,6				
Millinocket		4,260	3,980	280	6.5				

Table 3-15.	able 3-15. Per capita and household incomes for Maine, Penobscot and Piscat counties and Millinocket/E. Millinocket, 1980-1990 (Source: Bureau Census)						
		1980 Household Income (per capita)	1990 Median Household Income (per capita)				
Maine		\$13,816	\$27,854				
		(\$5,768)	(\$12,957)				
Penobscot Co	ounty	\$14,181	\$26,631				
	-	(\$5,593)	(\$12,231)				
Piscataquis C	ounty	\$12,260	\$22,132				
	-	(\$4,990)	(\$9,919)				
Millinocket/		\$19,840	\$35,610				
E. Millinocket		(\$6,829)*	(\$13,216)				

Millinocket and East Millinocket rely on industry, especially GNP, to pay a substantial proportion of local taxes. Bowater/GNP is the primary taxpayer for both Millinocket and East Millinocket, representing 72 percent and 88 percent of the towns' total property tax revenues, respectively.

4.0 ENVIRONMENTAL IMPACTS

Although FERC licenses the Ripogenus and Penobscot projects separately, many of the effects of operating the two projects are linked to such an extent that they cannot be considered independently (e.g., effects on streamflow); therefore, our environmental analyses are described according to the integrated or project-specific effects of the Applicant's Proposal and alternatives, as appropriate.

4.1 GEOLOGY AND SOILS

4.1.1 Applicant's Proposal

Construction activities associated with GNP's proposed enhancements at the Ripogenus Project would cause the only effects on the geology and soils in the project areas. Improvement of boat ramps, construction of changing facilities and additional parking facilities, and development of the Holbrook Stream nursery area could cause minor, short-term impacts. Local disturbances of the vegetation and soil could cause sediment to run off into adjacent waters if not properly contained. GNP has not yet stated how it would minimize these potential impacts. We conclude that a sediment control plan prepared and submitted in accordance with local, state, and federal regulations would ensure that adequate precautions are taken during construction of the applicant's proposed enhancements. GNP proposes no construction at the Penobscot Mills Project; therefore, there would be no construction-related effects or other effects on geology and soils in that area.

4.1.2 Alternatives 1 and 2

These alternatives do not include additional construction activities beyond those proposed by GNP that would affect geology and soils; therefore, these alternatives would have the same effects as the Applicant's Proposal.

4.1.3 No-action Alternative

Under the No-action Alternative, the project would not be modified, and geology and soils would not be affected.

4.2 STREAMFLOW

This section presents our analysis of streamflow at the Ripogenus and Penobscot Mills projects for all alternatives and the effects of alternative flow management on water uses in the West Branch of the Penobscot River. Because flow management at the Ripogenus Project affects the feasibility of managing flows downstream at the developments of the Penobscot Mills Project, we analyzed the effects of the alternatives on streamflow for both projects together.

The feasibility of increasing minimum flow releases from several developments and meeting numerous, apparently conflicting, water management objectives has been a major

point of contention between GNP and several intervenors. GNP developed a water use model to evaluate the consequences of various flow releases at all developments. The Cl questioned the validity of the model structure and the accuracy of the output. We evaluated GNP's water use model and concluded that it was appropriate for investigating streamflow issues at the Ripogenus and Penobscot Mills projects (see Appendix D for our evaluation of the model).

GNP conducted numerous model runs and provided extensive results representing various combinations of minimum flows and impoundment draw-down constraints specified by the staff (under the No-action Alternative, existing flows would continue, and no feasibility analysis was necessary). As noted in section 2.3, to define Alternative 2 we considered a range of flows intermediate between those proposed by GNP and those sought by the CI and other parties for greater resource enhancement. In this section, we describe only one set of minimum flows for all developments under the Applicant's Proposal, Alternative 1, and the No-action Alternative, but a range of minimum flows in the Back Channel under Alternative 2 to establish the feasibility of those flows. Table 4-1 summarizes the details of these combinations of flows; sections 4.2.1 through 4.2.3 contain our discussion.

4.2.1 Applicant's Proposal

As described in section 3.3, the West Branch of the Penobscot River includes 20 dams and impoundments, 6 of which are included in the proposed relicensing evaluated in this FEIS. Nearly all the flows within the West Branch are controlled by projects operated by GNP, and GNP manages its system to maximize sustained power generation for its mills in Millinocket and East Millinocket (GNP, 1991a, 1991b). Lake levels were never controlled for other purposes, although the dams and impoundments provide coincidental flood control benefits for the entire river basin. The only minimum flows provided were 200 cfs below McKay station, 20 cfs below Millinocket Lake Storage dam, and 2,000 cfs at Millinocket.

During consultation, various agencies and interest groups requested several changes in project operations for environmental enhancements, including flows and lake level management for recreation, wildlife, and fisheries. To determine the feasibility of meeting various, and sometimes conflicting, flow and lake level management requests, GNP developed a water-use model for the major elements of its hydro system. Appendix D provides details of the model, including which elements of the West Branch it can simulate, and the comments of agencies and interest groups on the model itself. GNP used the model to determine which of the various requested flows could be accommodated and based its determination of the cost of replacement power on the model. Based on these results, GNP developed a Water Use Plan (WUP) to meet as many of the flow requests as are hydrologically and economically feasible. Table 4-1 lists the flows and lake levels proposed as GNP's WUP for the Ripogenus and Penobscot Mills projects.

The water-use model can be used to compare Ripogenus and North Twin impoundment levels and outflows under the proposed WUP with historical operations (same as the No-action Alternative). GNP simulated all the years from 1976 to 1990 and selected a wet year, an average year, and a dry year, defined according to total inflow during the simulation period (see Appendix D, figures D-6 through D-8). GNP constructed the worst-case year (Appendix D, figure D-9) by combining data for the least amount of total available water Table 4-1. Water use under GNP's proposed water-use plan (Applicant's Proposal), Alternative 1, and Alternative 2, for the Ripogenus and Penobscot Mills projects (Source: GNP, 1991a, 1991b, 1993a, 1993b, 1994; staff)

Water Body	Minimum Flow or Lake Level	Purpose	Time Period
Ripogenus Project			
Upper Gorge All Alternatives Applicant's Proposal Alternative 1 Alternative 2	100 cfs leakage (12 cfs) 50 cfs 30 cfs	salmon fishery - salmon fishery salmon fishery	Jul 1 - Sep 30 Oct 1 - Jun 30 Oct 1 - Jun 30 Oct 1 - Jun 30
West Branch below McKay station (All alternatives)	1,800 - 2,300 ^(s)	rafting	May 1 - Oct 15 (0830-1700)
	1,000 cfs	salmon habitat	Jun 8 - Sep 15 (1700-0830)
	400 ^(b) 1,000 cfs 1,300 cfs (A.P.) 1,422 cfs (Alt. 1 & 2) ≥ 1,300 cfs (A.P.) ≥ 1,422 cfs (Alt. 1 & 2)	outage North Twin draw-down salmon spawning salmon incubation	year-round Sep 16 - Oct 14 Oct 15 - Nov 15 Oct 15 - Nov 15 Nov 16 - Jun 7 Nov 16 - Jun 7
Penobscot Mills Project			
North Twin and Millinocket Lake	"relatively stable" water levels	wildlife, wetlands, recreation, aesthetics	May 1 - Aug 15
alternatives)	minimum lake level	lake trout spawning (North Twin)	Oct 15 - Nov 5
	> = minimum lake level	lake trout incubation (North Twin)	Nov 6 - May 1
Back Channel Applicant's Proposal Alternative 1 Alternative 2	leakage (2-5 cfs) 350 - 500 cfs leakage - 165 cfs	- fishery habitat fishery habitat	year-round year-round year-round
Millinocket Stream Applicant's Proposal Alternative 1 Alternative 2	60 cfs leakage (20 cfs) 60 cfs 60 cfs 60 cfs or inflow	fishery habitat fishery habitat fishery habitat	May 1 - Oct 15 Oct 16 - Apr 30 year-round May 1 - Oct 15 Oct 16 - Apr 30
West Branch downstream of Millinocket (All alternatives)	2,000 cfs instantaneously (required by state law and 401 WQC)	water quality	year-round

^(a) Proposed whitewater recreation flows depend on the day of the week and whether the year is wet or dry; see Section 4.8 for a complete description of these flows.

(b) Other scheduled flows would be resumed as quickly and practically as possible using releases from Ripogenus dam, if necessary, but in no event would the outage last more than 3 days.

for each week from the 15-year period of record available; it is not data from a single actual year but represents the least water available for the period of record to meet various flow requirements and enhancements.

Model results show that the enhancements for this alternative can be accommodated in all years (Appendix D, figures D-6 through D-9; table D-3). In comparison with existing conditions (the No-action Alternative), Ripogenus impoundment elevations may be reduced slightly to provide additional downstream flows. Flows from Ripogenus are more regulated to meet summer whitewater recreation needs and salmon spawning and incubation flows in the fall and winter. Outflows may be reduced during early fall to facilitate North Twin draw-down for lake trout spawning. North Twin impoundment elevations are regulated for several purposes, including to stabilize levels during the summer recreation period. The regulated period is followed by rapid draw-down from mid-August through September to reach a minimum level for lake trout spawning in late October. Impoundment levels are maintained at or above this minimum fall elevation to optimize lake trout incubation. North Twin outflows are at least 2000 cfs to maintain the required minimum flow at Millinocket; flows also are adjusted with the inflows from Ripogenus, Millinocket Lake, and other upstream sources to maintain the desired impoundment elevation. Section 4.4 contains specific details about the effects of GNP's flow proposals on fisheries; section 4.8 discusses effects of the proposed flows on recreation.

4.2.2 Alternative 1

Under this alternative, GNP would provide flows in addition to those in the Applicant's Proposal (table 4-1) in Upper Gorge (50 cfs instead of leakage during the non-summer period), Millinocket Stream (60 cfs year-round instead of seasonally), and the Back Channel (350 to 500 cfs year round instead of leakage). To evaluate flow feasibility with the water-use model, we considered only 350 cfs to the Back Channel. Additional flows to Upper Gorge and Millinocket Stream are too small to be evaluated with the water-use model, and we assumed them to be feasible within the context of overall water use. Although we evaluated the feasibility of only 350 cfs in the Back Channel, the merits of flows other than 350 cfs are discussed in the appropriate resource sections (particularly fisheries and socioeconomics).

Under this alternative, GNP would provide a flow of 50 cfs (38 cfs more than GNP's proposed leakage flow) in Upper Gorge from October 1 through June 30. This amount of flow would not affect water use in the basin, if flow from McKay station was reduced by this amount. If this flow were taken out of Ripogenus storage rather than being obtained by reducing discharge from McKay station, the 20,655 acre-feet of water required to produce a 50-cfs flow would reduce the impoundment elevation by approximately 1 foot, assuming no additional inflow. This additional flow could be provided without affecting water use within the basin. The primary consequence of increased flow would be lost generation and cost to GNP (see section 5.3).

This alternative also would provide a flow of 60 cfs to Millinocket Stream year-round, which is 196 more days of increased flow than proposed by GNP. This flow is equivalent to about 23,000 acre-feet and would reduce the water surface elevation in Millinocket Storage Lake by a maximum of 2.2 feet per year, assuming no additional storage from reduced pumping to North Twin. This reduction of elevation would adversely affect fisheries and

recreation, unless withdrawals to North Twin were reduced. Additional flow to Millinocket Stream could be provided within the context of overall water use within the West Branch, although effects on resources within the lake could be significant.

Model results comparing GNP's proposed WUP (only leakage to the Back Channel) with this alternative (350 cfs to the Back Channel) show that impoundment levels during the wet year and the average year with 350 cfs in the Back Channel would not be significantly different than levels under the Applicant's Proposal (Appendix D, figures D-10 and D-11). All other enhancements could be achieved as proposed. During a dry or worst-case year, however, Ripogenus impoundment levels would be reduced by several feet, and outflows from Ripogenus would have to be reduced below desired levels for 4 to 5 weeks to avoid using storage from the following year (Appendix D, figures D-12 and D-13). These lower flows would occur during the salmon incubation period and could adversely affect survival during the winter. Under this alternative, North Twin impoundment levels would be reduced during a dry year and a worst-case year (Appendix D, figures D-12 and D-13). Lake trout spawning and incubation levels could be maintained during the dry year, but during the worst-case year, the incubation level would drop a foot or more below the spawning level for several weeks during late winter. We estimate that insufficient water would be available annually to meet all flow objectives in 13 percent of years, based on the annual water availability from 1976 through 1990.

The model also shows that summer recreation levels for both the dry and the worstcase years would be lower and less stable than levels under the WUP. Outflows from North Twin during the dry and worst-case years could be maintained above the levels required to meet the minimum flow of 2,000 cfs at Millinocket and 350 cfs in the Back Channel, except for 4 weeks during March. During both years, flows from North Twin would have to be reduced, and the Back Channel flows would have to be curtailed to avoid using storage from the following year to provide the required flow. (The water use model as used in this simulation assumes that flow in the Back Channel and the minimum flow at Millinocket would not be reduced until storage was depleted, even if other resource objectives were not being met.) During the worst-case year, flows from North Twin during this 4-week period would be reduced well below the 2,000 cfs minimum flow required at Millinocket. This flow could be provided from storage for the following year because the North Twin impoundment elevation is higher at the end of the year than at the beginning. The modeling results may overestimate the outflow required from North Twin because GNP assumed that 2,610 cfs would be needed to provide 2,000 cfs at Millinocket, 350 cfs to the Back Channel, and an additional buffer of 260 cfs to account for flow regulation, gate setting, and control equipment sensitivities (letter from J. Carson, GNP, April 5, 1994). The amount of this additional buffer has never been quantified in detail. When spread out over the year, the additional flow for the buffer would be within the amount of flow reduction that occurred during the last 4 weeks of the simulation. During a dry year, the average flow available over the year is within 20 cfs of the amount GNP contends is required to meet flow objectives below North Twin, and for the worst-case year, the average flow is within 80 cfs of this value. Only slight reductions in minimum flow from North Twin, therefore, would be required to meet all flow and impoundment level enhancement objectives.

We note, however, that GNP will be required to establish water flows and impoundment levels in advance to meet resource objectives, such as spawning flows in West

19 JA DUS ALGO T, AN ALSO ANALYSIS.

Branch and impoundment recreation levels and lake trout spawning levels at North Twin. Requiring GNP to provide flows to the Back Channel in addition to the 2,000 cfs minimum flow through the mill at Millinocket will make it more difficult to meet the other objectives in years when a significant drop in water availability occurs after one or more operational modes have been established. These types of constraints cannot be simulated by the water use model or by other models such as HEC-5, which are based on simulations of historical water availability.

We conclude that the flow-related enhancements proposed by GNP and 350 cfs in the Back Channel are feasible for maintaining water availability, except during a very dry year, when some enhancements would be reduced. The cost to GNP and the reduction in generation, however, would be significant (see sections 2.4 and 5.3). In these dry years, the Back Channel and minimum flows through the mill at Millinocket would have to be reduced, over-year storage would have to be used, or other upstream resource enhancements would not be met.

4.2.3 Alternative 2

Under this alternative, GNP would provide flows of 30 cfs in Upper Gorge and Millinocket Stream instead of leakage during the non-summer period, and flows of up to 165 cfs in the Back Channel year-round. To evaluate flow feasibility with the water-use model, we evaluated only 165 cfs to the Back Channel (Appendix D, figures D-14 through D-17; table D-5); however, habitat improvements and costs associated with flows ranging from 50 to 165 cfs are discussed in the appropriate resource sections (particularly fisheries and socioeconomics).

We evaluated a flow of 30 cfs in Upper Gorge from October 1 through June 30. This flow would not affect water use in the basin, if flow from McKay station was reduced by this amount. If this flow was obtained from Ripogenus storage rather than by reducing discharge from McKay station, the 9,780 acre-feet of water required to produce a 30 cfs flow would reduce the impoundment elevation by approximately 0.5 feet, assuming no additional inflow. This additional flow could be provided without affecting water use within the basin. The primary consequence of increased flow would be lost generation and cost to GNP (see section 5.3).

We evaluated a flow of 30 cfs to Millinocket Stream from October 1 through April 30, which is 10 cfs more than proposed by GNP for 196 days. This flow is equivalent to about 3,900 acre-feet and would reduce the water surface elevation in Millinocket Storage Lake by a maximum of 0.8 feet per year, assuming no additional storage from reduced pumping to North Twin. Fisheries and recreation would be slightly affected by this reduction in elevation, unless withdrawals to North Twin were reduced. This additional flow to Millinocket Stream could be provided within the context of overall water use within the West Branch, and impacts on other resources would be minimal.

Results for simulations with 165 cfs in the Back Channel show little or no difference from the GNP WUP during the wet and average years (Appendix D, figures D-14 and D-15), and all other enhancements could be achieved as proposed. During a dry year or worst case year (Appendix D, figures D-16 and D-17), however, Ripogenus impoundment levels would be reduced by a few feet and outflows from Ripogenus would have to be reduced below desired levels for 4 to 5 weeks to avoid using storage from the following year. These lower flows would occur during the salmon incubation period and could adversely affect survival during the winter. During dry and worst-case years, North Twin impoundment levels would be lower under this alternative than under GNP's proposed WUP. Lake trout spawning and incubation levels could be maintained during a dry year, but during the worst-case year, the incubation level would drop a foot or more below the spawning level for several weeks during late winter. We estimate that sufficient water would be available to meet all flow objectives, based on the annual water availability from 1976 to 1990.

The model also shows that summer recreation levels for both the dry and the worstcase years would be lower and less stable than levels under the WUP, although not as low as under Alternative 1. Outflows from North Twin during the dry and worst-case years could be maintained above the levels required to meet the 2,000 cfs minimum flow at Millinocket and 165 cfs in the Back Channel at all times. These modeling results probably exaggerate the inability to meet all flow-related enhancements because GNP assumed that 2,420 cfs would be needed to provide 2,000 cfs at Millinocket, 165 cfs to the Back Channel, and an additional buffer of 255 cfs to account for flow regulation, gate setting, and control equipment sensitivities (letter from J. Carson, GNP, April 5, 1994). The amount of this additional buffer has never been quantified in detail.

We conclude that the flow-related enhancements proposed by GNP and flows as great as 165 cfs in the Back Channel are feasible for maintaining water availability, except during a very dry year, when some enhancements should be reduced. The cost to GNP and the reduction in generation, however, would be significant for substantial the Back Channel flow releases (see sections 2.4 and 5.3).

4.2.4 No-action Alternative

Under the No-action Alternative, existing project operations would not change, and existing streamflows or impoundment levels would not be affected. The existing minimum flows would continue at 200 cfs below McKay station, 20 cfs below Millinocket Lake dam, and 2,000 cfs at Millinocket Mills, and no additional flow or lake level enhancements would be provided. The flows and elevations would continue as illustrated in Appendix D, figures D-6 through D-9.

4.2.5 Summary

Because streamflow issues and relevant model output are complex, we provide this summary of our findings.

- Under the Applicant's Proposal, sufficient water would be available to provide all proposed flows and lake level enhancements, even under a worst-case scenario with minimum available water.
- Under Alternative 1, available water would be sufficient to provide most flow enhancements considered, including flows up to 350 cfs in the Back Channel, except during very dry years. During dry years, not all enhancements could be satisfied simultaneously. Some enhancements (e.g., salmon incubation flows in

the West Branch below McKay station, North Twin summer recreation impoundment levels) could be adversely affected, or storage from the following year could be depleted to maintain the Back Channel flow and the required minimum flow at Millinocket. Under a worst-case scenario, it may be necessary to reduce the 350-cfs flow to the Back Channel, the minimum flow through the mill at Millinocket, or the over-year storage from Ripogenus and North Twin to achieve other enhancements.

- Under Alternative 2, available water would be sufficient to provide all proposed flow enhancements, including flows up to 165 cfs in the Back Channel, except during very dry years. During dry years, not all enhancements could be satisfied simultaneously. Some enhancements (e.g., salmon incubation flows in the West Branch below McKay station, North Twin summer recreation impoundment levels) could be adversely affected for a short time, or storage from the following year could be depleted unless the Back Channel flow or the required minimum flow through the mill at Millinocket were curtailed.
- A draw-down limit on Ripogenus would preclude attainment of downstream uses, such as salmon incubation flows in the West Branch of the Penobscot River below McKay station and lake trout incubation levels in North Twin, particularly during dry years. In addition, flood control benefits would be significantly reduced even with a modest draw-down limit on Ripogenus, as is discussed further in Appendix D.

4.3 WATER QUALITY

Water quality throughout all Ripogenus and Penobscot Mills project waters is generally very good; nearly all waters meet state water quality standards for coldwater fisheries (see section 3.4). Only three, site-specific water quality issues were identified during the scoping process: mercury concentrations in the impoundments, DO concentration in Dolby Pond, and water quality in the Back Channel.

4.3.1 Applicant's Proposal

4.3.1.1 Impoundment Mercury Concentrations

Methyl mercury can accumulate in living tissue and cause toxic effects on the nervous systems of people and wildlife that consume contaminated fish. Several intervenors expressed concern about mercury contamination within the projects' reservoirs, the effects of project operations on the cycling of methyl mercury through the food chain, and eventual bioaccumulation of mercury in bald eagles nesting along the banks of the impoundments (FWS, 1992; EPA, 1992; American Rivers et al., 1992).

EPA expressed concern about existing high concentrations of mercury in the sediments of impoundments downstream from the town of Millinocket and the effects of such contamination on aquatic organisms, bald eagles, other wildlife, and humans. American Rivers and its affiliates also suggested that because of the affinity of methyl mercury for clay particles, resuspension of clay particles during reservoir draw-down (through wind and wave

action on exposed sediments during low water levels) could increase the rate of transmission of mercury through the food chain.

FERC requested additional information to determine if fluctuations in impoundment water levels are linked to mercury concentrations in fish and invertebrates. After consulting with FWS, DEP, PIN, and DIFW, GNP developed a study plan that included sampling fish and invertebrates indigenous to the basin for mercury contamination, sampling in reference lakes and impoundments, comparing tissue concentrations with EPA criteria for edible fish tissue, and comparing mercury concentrations in tissues of organisms from other Maine watersheds.

Based on the results of that study, GNP concluded that concentrations of mercury within the study lakes were within the range of concentrations typically found in other Maine watersheds. Mercury concentrations in sediments from Millinocket Lake (0.24 mg/kg) and Dolby Pond (0.69 mg/kg) were slightly higher than concentrations in the other lakes and impoundments (all of which averaged less than 0.19 mg/kg; table 4-2), but these differences were not statistically significant. Mercury concentrations in freshwater mussels were generally below 0.12 mg/kg (except Debsconeage Lake), and concentrations in mussels from the drawdown reservoirs were among the lowest observed in this study (table 4-3). Tissue concentrations in draw-down and reference impoundments were all below EPA criteria.

Table 4-2. Mean concentration of mercury in sediments from selected lakes (adapted from Environmental Science and Engineering, 1992)								
Water BodyNumber of SamplesMean Concentration of Mercury (mg/kg)								
Millinocket Lake (draw-down)	4	0.24						
North Twin Impoundment (draw-down)	8	0.13						
Debsconeage Lake (control)	3	0.19						
Carr Pond (control)	3	0.15						
Schoodic Lake (control)	4	0.09						
Dolby Pond	7	0.69						
Molunkus Lake (control)	4	0.10						
Mattamiscontis Lake (control)	3	0.15						

GNP also concluded that mercury concentrations were generally lower in forage and bottom feeding fish than in predatory fish (table 4-3). GNP found no statistically significant differences in the tissue concentrations of mercury in forage and bottom-feeding fish among the sampled lakes; however, lake trout from the North Twin impoundment and Millinocket Lake, both of which have fluctuating water levels, contained high concentrations of mercury.

GNP argues that the higher concentrations of mercury in lake trout do not indicate that periodic draw-down in the reservoirs promotes accumulation. GNP maintains that because lake trout are long lived and on the top of the food chain, they accumulate mercury naturally.

Table 4-3. Average mercury concentrations (mg/kg) found in fish and mussels from reservoirs and reference lakes during the summer of 1992 (adapted from Environmental Science and Engineering, 1992)										
Species	Kind of Sample	Trophic Status	Dolby Pond*	Mattam/scontis Lake	Molunkus Lake	North Twin Impoundment*	Millinocket Lake*	Carr Pond	Debsconeage Lake	Schoodic Lake
Lake Trout	Fillet	Predator	-	•		1.09	1.20	0.66	0.27	0.28
Smallmouth Bass	Fillet	Predator	0.91	-	0.63	-	-	-	-	•
Chain Pickerel	Fillet	Predator	0.87	0.58	0.68	-	-	-	-	-
Burbot	Whole Fish	Predator	-	-	•	0.59	0.23		-	-
Round Whitefish	Whole Fish	Bottom Feeder	-	-		*	0.35	-	-	0.18
Lake Whitefish	Whole Fish	Bottom Feeder	-	-	-	0.19		0.37	0.40	-
White Sucker	Whole Fish	Bottom Feeder		0.22	0.16	0.21	0.18	0.20	0.19	0.16
Rainbow Smelt	Whole Fish	Forage	0.29	·	•	0.21	0.14	0.21	-	-
Brown Bullhead	Whole Fish	Bottom Feeder	0.28	0.08	-	-	-	-	-	•
Freshwater Mussels	Composite	·	0.03	0.08	0.11	0.07	0.03	0.05	0.13	0.09
* Project waters										

GNP also states that because tissue concentrations were similar in forage and bottom-feeding fish from draw-down and control reservoirs, the data do not support the conclusion that existing project operations enhance movement of mercury through the food chain.

The most likely sources of mercury in the two impoundments in which water levels fluctuate periodically (North Twin impoundment and Millinocket Lake) are weathering of rock in the watershed and atmospheric inputs. GNP operates two paper mills (which could be point sources of mercury) near the project area, but the effluents of both mills are discharged downstream of the North Twin impoundment and Millinocket Lake. No data are available to document the mercury input to project waters from weathering, but there is no basis for concluding that such inputs to project waters would be different than inputs to control impoundments.

No evidence suggests that long-established impoundments (75 years in the present case) contribute to increased mercury concentrations or increased production of methyl mercury over background concentrations. Research has shown, however, that high concentrations of mercury are commonly found in fish from recently flooded freshwater reservoirs and acidified lakes (Gilmour and Henry, 1991). Stokes and Wren (1987) reviewed literature on mercury levels in new reservoirs in Canada and the United States and found that, 4 years after impoundment, concentrations of monomethyl mercury in fish were two to five times greater than preimpoundment concentrations. Mercury methylation rates may be enhanced during flooding by large additions of nutrients that stimulate the activity of soil bacteria (WHO, 1990; Stokes and Wren, 1987).

GNP's water quality data for project impoundments indicate that most of the water bodies within project boundaries are either neutral or very slightly acidic (GNP, 1991a, 1991b). One of the lowest pH levels (6.6) and the highest water color recorded (which may be indicative of high dissolved organic carbon attributable to humic acid) were found in the deep portion of upper Dolby Pond, the impoundment at which mercury concentrations were high in bald eagles. Although these data suggest that pH and dissolved organic carbon might be contributing to elevated concentrations of mercury in organisms in Dolby Pond (a phenomenon reported in the literature; Wiener et al., 1990), in general, the water quality data do not suggest that acidification or high dissolved organic carbon are likely to be major contributing factors in other project impoundments. The literature, furthermore, does not suggest a mechanism by which project operations could be responsible for low pH and high color that might contribute to mobilizing mercury.

The high mercury levels observed in lake trout are probably a result of the species' top position in the food chain and its longevity, both of which would increase the natural bioaccumulation of mercury. Mercury concentrations were lower in lake trout from control lakes; however, consumption of large amounts of rainbow smelt (the preferred prey for lake trout) in North Twin impoundment and Millinocket Lake may explain the difference. Smelt populations are more abundant in project waters than in other waters in Maine, and smelt is the major prey of lake trout in project waters (Environmental Science and Engineering, 1992). Although mercury concentrations in rainbow smelt were similar in samples collected from Carr Pond and the draw-down impoundments (table 4-3), the higher rate of smelt consumption by lake trout in project impoundments could cause faster and greater accumulation of mercury. Although no definitive explanation for higher mercury levels in some fish species in project

Шİ

waters is available at this time, all available information leads us to conclude that project operations probably are not the cause of elevated mercury concentrations.

American Rivers and its affiliates suggest that increased resuspension of clay particles complexed with methylated mercury during draw-down could mobilize mercury into the food chain. GNP's measurements of total suspended solids (TSS) revealed that TSS was not higher in draw-down reservoirs than in control reservoirs and lakes (Environmental Science and Engineering, 1992). Existing data, therefore, do not support the hypothesis that resuspension of clay is mobilizing methylated mercury in the draw-down reservoirs.

Based on our review and data presented by GNP, we conclude that current or proposed operation of the projects, including periodic draw-downs of the reservoirs, would not increase mercury levels in the impoundment or the production of methyl mercury.

4.3.1.2 Dolby Pond Dissolved Oxygen Levels

EPA and CI contend that GNP has not adequately investigated the cause(s) of low DO concentrations in bottom waters of Dolby Pond and the possible link to mill discharges (letters from R. Manfredonia EPA, May 21, 1993; D. Sosland, CI, September 3, 1993).

Low DO concentrations during warm months (as low as 0.1 mg/l) have been recorded but not explained (see section 3.4.2.1). Given the abundant accumulation of organic matter within the impoundment, DO deficits probably are natural phenomena caused by biological decomposition of organic matter (possibly aggravated by bark and debris accumulated from historical log drives) in the hypolimnion of the weakly stratified impoundment during warm periods of the year. Mill discharges, however, may enhance biological and chemical oxygen demand and contribute to low DO levels.

EPA renewed GNP's National Pollutant Discharge Elimination System permit in 1992, and DEP issued a WQC for this permit, which stated that "...the discharge described will not lower the quality of the receiving waters below the minimum requirement of their classification."

Although the record contains no evidence to suggest that the mill discharges cause the DO deficits, GNP proposes (in accordance with a DEP WQC condition) to study the relationship between mill discharges and DO concentrations in Dolby Pond and to evaluate possible means of resolving any problems identified. Although EPA suggests delaying licensing pending resolution of this issue, we conclude that a licensing delay is not warranted because the proposed project would continue to operate as it does under existing conditions, and no evidence indicates any current biological effect of what may be natural low DO events. The permitted discharge volume is 55.8 cfs, less than 3 percent of the minimum continuous flow from the Millinocket Development (2,000 cfs), and a much smaller percentage during normal and high flows. The magnitude and frequency of low DO events in Dolby Pond would not be altered under the Applicant's Proposal because GNP proposes no changes in project operation.

4.3.1.3 Back Channel Water Quality

CI, TU, and EPA initially identified their concerns about water quality in the Back Channel in their motions to intervene, dated August 1992. In more recent comments dated from May to September 1993, all parties state that the current flow regime is not adequate to meet the state water quality standards, particularly the designated use standards of navigation and aquatic life criteria (letters from M. Huntington, CI, April 30, 1992; C. Gauvin, TU, May 24, 1993; R. Manfredonia, EPA, May 21, 1993).

Under both existing and proposed operating conditions, the Back Channel would receive flows varying from leakage (2 to 5 cfs) to 29,000 cfs (periodic high-volume spillage). In its WQC for the Penobscot Mills Project, the DEP waived its authority to certify that the Back Channel would meet Class C water quality standards. No data in the record indicate that numerical standards for DO are not being met; however, we agree with various intervenors that water quality in the Back Channel is unlikely to meet the Class C standards for designated uses, which include navigation. We conclude that the proposed flow regime (leakage and spillage) would allow navigation through this 4.5-mile section of river only during spill conditions; however, there is no apparent demand for recreational boating in this river reach at present. (See section 4.4 for discussion of fish habitat as a designated use.)

Using existing conditions as the baseline, we conclude that GNP's proposed projects would not affect water quality in the Back Channel because GNP proposes no changes in the existing flow regime. We also conclude that there is no compelling basis for challenging the state's decision to waive certification for the Back Channel.

4.3.1.4 Cumulative Impacts

Initial construction of all the project developments converted free-flowing river segments into impoundments, which generally have very low flushing rates. Although this original habitat modification changed water quality (e.g., greater diurnal temperature variation and higher oxygen levels in rivers; stratification with lower DO in deep layers in impoundments), the Applicant's Proposal would not alter existing water quality regimes in project waters. We conclude, therefore, that the Applicant's Proposal would result in no new or additional cumulative impacts on or enhancements of water quality.

4.3.2 Alternative 1

4.3.2.1 Impoundment Mercury Concentrations

Under this alternative, flows would be increased to enhance fisheries throughout the project area, which would cause more rapid impoundment draw-down and expose larger areas of impoundment substrate. Our evaluation of the mercury issue indicates that project operations, including impoundment draw-down, probably do not increase mercury concentrations in project waters. This alternative, therefore, would create no potential for further elevating existing mercury concentrations, nor would it reduce existing mercury concentrations.

4.3.2.2 Dolby Pond Dissolved Oxygen

Increased flows in the Back Channel under this alternative (discussed in section 4.4) might enhance dissolved oxygen concentrations in water entering Shad Pond and then passing into Dolby Pond. Given the flow amounts and the large volumes of the impoundments, however, the possible slight increase in DO would not alter existing oxygen conditions in Dolby Pond. Oxygen conditions in Dolby Pond under this alternative, therefore, would be the same as under the Applicant's Proposal.

4.3.2.3 Back Channel Water Quality

This alternative includes proposed flows of 350 cfs in the Back Channel to maximize habitat for fry and juvenile salmon, as discussed in section 4.4. Although no low DO values or elevated temperatures have been documented in the Back Channel under existing conditions, the 350 cfs flow would reduce the potential for degraded water quality, enhancing aeration and flushing rates. Under this alternative, therefore, water quality probably would be enhanced and protected against potential degradation under worst-case summer conditions. Flows as high as 350 cfs still would not provide sufficient water for navigation of the Back Channel (see section 4.8); consequently, this flow regime probably would not be sufficient to meet all Class C water quality criteria, as requested by some intervenors.

4.3.2.4 Cumulative Impacts

Increased flow in the Back Channel is the only measure associated with this alternative that could alter existing cumulative impacts on water quality. We conclude that the potential slight improvement in DO concentrations would not affect oxygen conditions in downstream waters. There are, therefore, no cumulative water quality enhancements or impacts associated with this alternative.

4.3.3 Alternative 2

4.3.3.1 Impoundment Mercury Concentrations

Under this alternative, impoundment draw-downs would be somewhat greater than under the Applicant's Proposal, but less than under Alternative 1. Project operations, including impoundment draw-down, under the Applicant's Proposal probably would not increase mercury levels in project waters. This alternative would neither increase nor reduce existing mercury concentrations.

4.3.3.2 Dolby Pond Oxygen Levels

Under Alternative 2, less flow would be released into the Back Channel than under Alternative 1; therefore, the potential for oxygen enhancement would be less than under Alternative 1. The proposed flows in the Back Channel, therefore, would not affect low DO in Dolby Pond.

4.3.3.3 Back Channel Water Quality

Under Alternative 2, we considered a minimum flow release into the Back Channel that would be higher than leakage but less than the 350 cfs flow proposed in Alternative 1. As discussed under the Applicant's Proposal, any flow higher than leakage would increase flushing rate and aeration in the Back Channel and, thus, protect against water quality degradation during worst-case summer and early fall conditions, when water temperatures are at their seasonal peak. This alternative, therefore, provides greater protection for water quality than the Applicant's Proposal, but less than Alternative 1.

4.3.3.4 Cumulative Impacts

Because flow modifications proposed under this alternative are less than those proposed under Alternative 1, this alternative would result in no cumulative enhancement of or impact on existing water quality conditions in the project area.

4.3.4 No-action Alternative

Under the No-action Alternative, existing project operations would not change, and water quality conditions would not change.

4.4 FISHERIES RESOURCES

Increased flows to benefit fisheries at several locations in the project area is one of the key issues in this licensing process. During scoping, we identified five site-specific fisheries enhancements to be evaluated in this FEIS: minimum continuous flows in the Upper Gorge below Ripogenus dam (Ripogenus Project), minimum continuous flows in Millinocket Stream (Penobscot Mills Project), minimum continuous flows in the Back Channel (Penobscot Mills Project), restrictions on impoundment draw-down to protect impoundment fish stocks (both projects), and provisions for fish passage (both projects).

4.4.1 Applicant's Proposal

4.4.1.1 Upper Gorge Flows

In motions to intervene, PIN, FWS, CI, TU, and EPA stated that flows of 50 to 100 cfs should be provided year-round to enhance habitat for fish and other aquatic life. Since then, FWS commented that the Applicant's Proposal is adequate, given the state's goals of increasing fishing opportunity in this reach (letter from J. Deason, Interior, May 24, 1993). EPA requested provision of flows to meet criteria for aquatic life (letter from R. Manfredonia, EPA, May 21, 1993). PIN deferred to FWS and now accepts the Applicant's Proposal (letter from P. Bisulca, PIN, May 21, 1993). CI, by not changing its request, presumably continues to seek year-round flow enhancements. Other intervenors, however, still recommend higher year-round flows.

The present flow regime in Upper Gorge is leakage (approximately 12 cfs) in addition to spillage, which occurred during 12 of 20 years between 1970 and 1989. Spillage was as high as 11,600 cfs during some events (see section 4.2). Occasional scouring flows of this

magnitude generally occur in unregulated rivers and prevent siltation, which can degrade salmonid habitat by covering the substrate on which macroinvertebrates live and feed and on which salmonids spawn. Such high-volume flows, however, also may prevent early life stages, juveniles, and even adult fish from remaining in the high-gradient, bedrock pool habitat in Upper Gorge.

GNP proposes to increase flows to 100 cfs from July 1 to September 30, with leakage during the remainder of the year to attract adult salmon during the summer fishing season and increase fishing opportunity within this reach (see section 4.8 for further discussion of fishing opportunity). Based on GNP's 1989 habitat-based flow study, this flow enhancement would increase the amount of adult salmon habitat in the reach from 41,869 to 71,707 square feet (71 percent; figure 4-1). GNP did not conduct habitat analyses for any species other than salmon because these waters are managed only for this species.

DIFW, which consulted with GNP during the habitat study, accepts the Applicant's Proposal (LURC, 1993). DIFW has identified no overwintering or spawning habitat in Upper Gorge and, thus, no benefit for salmon from winter and spring flows. The scouring spillage flows during the spring also are likely to displace any fish that may occupy the area during the summer. The IFIM results for juveniles and fry (figure 4-1) show that fry habitat declines to nothing at 100 cfs, and that juvenile habitat declines precipitously at flows above 100 cfs, which supports DIFW's position. LURC adopted the Applicant's Proposal as a condition of its WQC.

The proposed seasonal flow enhancement would improve habitat for adult salmon and may benefit other aquatic organisms. Year-round flows of any magnitude also might enhance habitat for other fish species and other aquatic life seasonally, but any improvements probably would be eliminated by the periodic high-volume spillage. Year-round flows, therefore, would not provide significant long-term benefit to fisheries resources, but GNP's proposed seasonal flow increase probably would draw adult salmon into Upper Gorge and, thus, support additional recreational fisheries.

4.4.1.2 West Branch Flows

CI, TU, FWS, PIN, and EPA all requested an IFIM study of this reach to establish minimum flow requirements. The Commission initially requested such a study but later withdrew this request, stating that the study would provide no new information but would cause GNP considerable expense. The Commission decided against requiring an IFIM study in the West Branch because

- the existing salmon fishery meets or exceeds all of the state's management goals for catch per angler-day and average size per fish caught;
- GNP proposes to create additional nursery habitat along the West Branch (see section 2.1.3.4); and
- GNP's flow proposals were developed from abundant data about the West Branch fisheries resources.



Figure 4-1. Total weighted useable area (WUA) for salmon in Upper Gorge at various flows (Source: GNP, 1991a)

All intervenors continue to request an IFIM study (letters from C. Gauvin, TU, September 10, 1992; P. Bisulca, PIN, May 21, 1993; J. Deason, Interior, May 24, 1993; R. Manfredonia, EPA, May 21, 1993).

The existing minimum flow requirement below McKay station is 200 cfs. GNP proposes an elaborate schedule of flow enhancements related to salmon spawning and incubation and recreational boating. In addition GNP proposes a minimum, short-term (maximum 3 days), emergency outage flow of 400 cfs; flows would be restored to at least 711 cfs (summer aquatic base flow) via spillage from Ripogenus as soon as possible after any emergency shut-down.

GNP delivered its flow enhancement proposals, generally between 1,800 and 2,300 cfs, during consultation with resource agencies and representatives of recreational boating interests (see section 2.1.3.8). Impacts of the proposed flows on recreational resources, including fishing opportunity, are discussed in section 4.8. FWS maintains that fish habitat cannot be assessed accurately until a quantitative, habitat-based flow study (such as IFIM) is conducted (letter from G. Beckett, FWS, September 2, 1993). FWS also contends that the flows in the West Branch below McKay station are artificially high and recommends flows approximately half of those currently being released (letter from G. Beckett, FWS, September 2, 1993). FWS believes that the high flows may be reducing the suitability of salmon nursery habitat.

GNP does not regulate flows during the spawning or incubation periods; however, GNP now proposes to stabilize flows throughout these periods (October 15 to June 7). These flow regulations would increase spawning in the West Branch and survival in the redds, thereby increasing total reproductive success.

According to DIFW, flows of 1,000 cfs maintain a full channel, meaning that the entire channel is wet. GNP estimates that typical flows during the spawning season would range between 1,300 and 2,000 cfs. Based on this information, the proposed spawning and incubation flows would prevent exposure and desiccation of redds. The proposed flows, therefore, would improve reproductive success of salmon in the West Branch.

GNP proposes minimum outage flow enhancements of 400 cfs (automatic, short-term) and 711 cfs (as soon as possible). These flows would be implemented if an emergency required shutting down the turbines. If the outage lasts longer than 3 days, GNP proposes to provide flows of 1,000 cfs. Under historic operating conditions there was a minimum lag time of 20 to 30 minutes between system shut-down at McKay station and restoration of flow in the river while GNP personnel identified the problem at Millinocket and instructed the operator at Ripogenus dam to open the deep gate to restore flows of 200 cfs. The proposed automated system is already in place and enables flows of 400 cfs at McKay station within 10 to 15 minutes of shut-down. Thus, GNP's proposal provides for more immediate, higher flows during plant shut-down.

Impacts associated with outage flows are loss of habitat and degradation of remaining habitat. When flows cease, the only water in the channel is in deep pools. Depending on weather conditions, water temperature can increase, and DO can decrease. These extreme summer conditions can kill fish. Aquatic invertebrates and other aquatic organisms less

mobile than fish also can die from desiccation. Any operational changes that reduce the frequency or duration of outage-related conditions would improve the habitat for aquatic life, including fish, in the West Branch. Because GNP's proposals increase the magnitude of the outage flows and decrease the time necessary to restore flows, we conclude that they would provide a substantial improvement over historic conditions.

Through an agreement with parties interested in recreational boating GNP has generally guaranteed specified seasonal flows during daylight hours. These flows are discussed in sections 4.2 and 4.8. Because the proposed flows are similar to historical flows, they probably would not adversely affect the existing salmon stock.

GNP proposes to develop a salmon nursery area near Holbrook Pool on the West Branch between McKay station and North Twin impoundment. Although plans are not final, resource agencies support the proposal. According to GNP, this enhancement would create approximately 7,000 square yards (70, 100-square yard units) of spawning and nursery habitat (LURC, 1993). Using a production estimate of 5 parr per unit and a survival estimate of 70 percent, this amount of habitat could add 245, 3-year-old fish to the population per year. Although this enhancement is not included as a condition of the WQC, it should be adopted because of the potential benefits of additional nursery and spawning area in the West Branch.

4.4.1.3 Millinocket Stream Flows

CI, EPA, PIN, TU, and FWS all recommend higher year-round minimum flows in this stream despite the Applicant's Proposal to provide flow enhancements during the recreational fishing season (letters from P. Bisulca, PIN, May 21, 1993; D. Sosland, CI, September 3, 1993; R. Manfredonia, EPA, May 21, 1993; C. Gauvin, TU, May 24, 1993; J. Deason, Interior, May 24, 1993).

The existing year-round minimum flow requirement in Millinocket Stream is 20 cfs; however, flows are usually higher because of additional spillage from the dam. The aquatic base flow for this regulated stream is 60 cfs. GNP proposes a flow of 60 cfs from May 1 to October 15, and 20 cfs during the remainder of the year. Several intervenors request a year-round flow of 60 to 80 cfs based on the results of GNP's IFIM study, which was conducted in consultation with DIFW and FWS.

DEP adopted the Applicant's Proposal in the WQC for the project. IFIM studies of flows up to 80 cfs show that habitat for adult smallmouth bass is highest at flows of 30 to 50 cfs; spawning habitat is highest at 50 cfs. As figure 4-2 shows, however, the magnitude of change in habitat quantity is very small over the entire range of flows studied. Spawning smallmouth bass, the only life stage that would benefit from the proposed flow enhancement, would gain 16 percent more habitat. GNP's proposed flows would only slightly enhance smallmouth bass habitat in Millinocket Stream but would have no adverse impact.

WUA for early and late salmon fry declines slightly as flows increase, whereas WUA for juvenile, adult, and spawning salmon increases steadily up to 80 cfs (figure 4-2). Juvenile habitat increases about 15 percent between 20 and 30 cfs, with a lesser rate of increase at additional 10-cfs flow increments. Spawning habitat shows the next greatest increase with increased flow. The amount of spawning and adult habitat that could be created in Millinocket



Figure 4-2. Total weighted useable area (WUA) for salmon (panel A) and smallmouth bass (panel B) in Millinocket Stream at various flows (Source: GNP, 1991b)

4-20

Stream, however, is limited; IFIM results show that only 200 units of such habitat are present at flows of 80 cfs.

Although these IFIM data suggest that salmon may benefit from flows greater than those proposed by GNP, we conclude that the potential for population growth from increased flows would be small because of the small size of the current salmon population. Redd surveys conducted by GNP identified 21 salmon redds in 1986, 10 in 1987, and 9 in 1988. All redds were found in the upper 2 miles of the 7.9-mile affected stream section that connects Millinocket Lake dam and the Millinocket tailrace.

The presence of small numbers of wild juvenile salmon indicates some degree of successful reproduction; however, adult salmon habitat is not abundant because Millinocket stream has no deep pools (GNP, 1991b). All adult salmon reported in creel surveys of this reach were of hatchery origin and probably entered Millinocket Stream via spillage from Millinocket Lake. There is no indication that existing spawning and nursery habitat contribute to sustaining a wild salmon stock in Millinocket Stream.

GNP's proposed 60-cfs flow from May 1 to October 15 would significantly enhance juvenile and adult salmon habitat (35 percent and 48 percent, respectively). Habitat for early and late fry would be decreased by 15 percent and 3 percent, respectively. GNP's proposed 20-cfs flow during the remainder of the year would maintain existing conditions; however, redds created as a result of the 60-cfs flow provided during the fall spawning period might be desiccated or frozen during the winter because of exposure or decreased water depth when the flow regime returns to 20 cfs after October 15. Winter flows greater than 20 cfs would provide a greater degree of protection and enhancement for salmon redds; however, the salmon stock in Millinocket Stream is small, and a significant increase in the regional abundance of salmon is unlikely, even with enhancements.

GNP proposes to stock brook trout in Millinocket Stream annually. Although such stocking would probably enhance recreational fisheries (see section 4.8), it would not enhance resident fisheries resources. DIFW does not manage Millinocket Stream as brook trout habitat.

4.4.1.4 Back Channel Flows

GNP proposes no minimum flow for the Back Channel. IFIM studies show that the Back Channel encompasses between 200,000 and 400,000 square feet of habitat for juvenile and fry life stages of landlocked salmon under leakage flows, and no adult or spawning habitat (figure 4-3). Although juvenile habitat increases rapidly with increasing flow, only a very limited amount of adult and spawning habitat (about 100,000 square feet) is created at flows as high as 700 cfs. We conclude that GNP's proposed project would not significantly alter existing conditions, nor would it enhance existing fisheries habitat in the Back Channel.



Figure 4-3. Total weighted useable area (WUA) for salmon in Back Channel at various flows (Source: GNP, 1991b)

4.4.1.5 Impoundment Draw-downs

CI, FWS, TU, and PIN request reducing or eliminating draw-downs of project impoundments to protect littoral zone habitat for fish. TU and CI claim that impoundment draw-downs affect the quality and abundance of fish habitat (letters from M. Huntington, CI, April 30, 1992; C. Gauvin, TU, May 24, 1993).

Species that inhabit the littoral zone of the impoundment are displaced when their preferred habitat is dewatered. Other species that normally occupy deep water may spawn in shallow water, and dewatering during draw-down could affect their reproductive success, depending upon the magnitude and timing of the draw-downs. Impoundment draw-downs also can make tributaries inaccessible to species that migrate to habitat to spawn. Table 4-4 summarizes the spawning habitat requirements for various species and the seasons of spawning and incubation during which impoundment levels are critical.

Table 4-4. Spawning habitat for key impoundment species (Source: Smith, 1985)					
Species	Spawning Habitat and Season				
Lake trout	Lake shoreline at depths of 1 to 15 feet during October and November				
Lake whitefish	Lake shoreline at depths of approximately 25 feet and in tributaries during October/November				
Burbot	Lake shoreline at depths of 1 to 4 feet and in tributaries during February				
Smelt	Lake shoreline at depths of 1 to 5 feet and in tributaries during March				

North Twin, Millinocket Lake, and Ripogenus impoundments all experience substantial draw-downs because GNP operates them in annual storage mode. Project impoundments associated with run-of-river developments such as Dolby, Millinocket, and East Millinocket, however, do not experience significant draw-downs.

GNP's proposed project operations would have a minor effect on the timing and magnitude of draw-downs in Millinocket Lake or the Ripogenus impoundment. These impoundments potentially experience draw-downs of up to 6 and 44 feet, respectively. The WQC for the Millinocket Lake Storage dam contains a condition requiring GNP to maintain impoundment levels between 470 and 480 feet elevation; however, fluctuations within this range virtually eliminate any chance of natural reproduction for lake trout and lake whitefish in the impoundment itself by exposing the redds during incubation. Likewise, draw-downs eliminate natural reproduction of these two species within the Ripogenus impoundment (table 4-4). Existing populations are maintained by stocking (lake trout) or tributary spawning (lake whitefish). Reproductive success of burbot and smelt in Ripogenus is not jeopardized because, in most years, water levels are at their minimum during March and rise from April to

June (table 4-5) following spawning, which is generally between February and April for these species.

. . .

lable 4-5. Changes in elevation, surface area, and dates of minimum and maximum elevation in Ripogenus and North Twin impoundments for the project alterna- tives (Source: Staff)							
Ripogenus	No-Action	Applicant's Proposal	Alternative 1	Alternative 2			
Annual maximum change in elevation (feet)	26.3	26.4	27.0	26.4			
Change in surface area, 6/1-12/31 (acres)	13,000	12,500	13,700	13,700			
Approximate date of minimum elevation (average year)	March 21	March 21	March 21	March 21			
Approximate date of maximum elevation (average year)	May 15	May 15	May 15	May 15			
North Twin				·····			
Annual maximum change in elevation (feet)	13.3	13.1	13.2	13.6			
Change in surface area, 6/1-12/31 (acres)	2,400	2,200	2,300	2,300			
Approximate date of minimum elevation (average year)	April 21	October 15	October 31	October 31			
Approximate date of maximum elevation (average year)	May 15	May 15	May 15	May 15			

Because this FEIS assesses impacts using the existing conditions as a baseline, we conclude that continued operation of the projects under the Applicant's Proposal would not alter fish habitat in Millinocket Lake or Ripogenus impoundment. DIFW manages this portion of the West Branch for landlocked salmon, and it wishes to prevent the establishment of a self-sustaining population of lake trout in Ripogenus impoundment because such a population could adversely affect the smelt population. (The smelt population is, to a great extent, the basis for the high population of salmon below McKay station.) The Applicant's Proposal is consistent with DIFW management objectives for the project area because impoundment draw-downs would continue to preclude the reproductive success of lake trout in both Millinocket and Ripogenus impoundments.

Annual draw-downs at North Twin impoundment can be as high as 17 feet below normal pool level. Such draw-downs reduce the surface area of the impoundment by 3,200 acres, or 18 percent. Because DIFW is attempting to establish a self-sustaining lake trout population in North Twin, it requests that GNP monitor its draw-down schedule to minimize adverse effects on lake trout reproduction. Lake trout spawn between early October and mid-November. Although GNP does not propose draw-down limits, it would complete fall drawdown by October 15, then maintain the water level at or above this level until May 1, by which time the lake trout eggs will have hatched. DIFW accepted this proposal, and DEP included this enhancement as a condition of the WQC. The proposed schedule also would improve spawning habitat for burbot and lake whitefish. GNP also proposes a study to evaluate the reproductive success of lake trout and its correlation with water level management. We agree that implementing the proposed schedule of impoundment draw-downs would minimize any adverse effects on the lake trout population of North Twin. Table 4-5 illustrates that the magnitude of draw-downs under the Applicant's Proposal would decrease slightly (approximately 0.2 feet); this decrease is probably biologically insignificant. We agree that the proposed study would provide valuable information for managing the fishery to achieve DIFW's goal of a self-sustaining lake trout population. Table 4-5 shows that there is little difference in impoundment elevation and surface area fluctuation among the alternatives.

4.4.1.6 Fish Passage

GNP proposes to repair or modify the North Twin fish passage facility in consultation with DIFW and FWS. TU requests installation of both upstream and downstream fish passage facilities at Millinocket Lake Storage dam and downstream passage at Dolby, Millinocket, and East Millinocket dams (letter from C. Gauvin, TU, May 24, 1993). FWS reserves its authority to prescribe fish passage facilities (letter from J. Deason, Interior, May 24, 1993).

DIFW and GNP contend that construction of passage facilities is unnecessary because no anadromous species inhabit the project areas, and because passage would threaten the state's fisheries management objectives for the project waters. Smallmouth bass and pickerel, both introduced species, provide valuable sportfishing opportunities within the lower Penobscot Mills Project area. Adding passage facilities would increase their distribution, which would almost certainly result in direct competition with native species such as salmon, lake trout, and a variety of other game and nongame species. Although fish passage at dams is generally viewed as beneficial to fisheries resources, we agree with the state that, in this case, creating passage would facilitate the further, undesirable spread of introduced species throughout the West Branch. Limiting fish passage to North Twin would preserve a coldwater fish community that most closely resembles the native assemblage of species.

The North Twin fish passage facility was built in 1934 and repaired in 1984. In monitoring studies conducted during 1979 and 1986, 67 and 38 salmon were trapped after passing up the fishway. Only 1 of 38 salmon captured and tagged below the North Twin dam during 1986 and 1987 was later recaptured above the North Twin dam (GNP, 1991b). These data suggest that some salmon move through the existing passage facility, but that the number is limited.

The primary purpose for providing fish passage at North Twin is to allow salmon that passed over the North Twin dam to move back upstream to spawning areas in the West
Branch below McKay station. Few wild landlocked salmon are found downstream of North Twin dam, and most salmon fisheries below that point are supported almost totally by stocking. The North Twin fishway does not significantly contribute to sustaining existing fish stocks; however, because the facility already exists and is used by some salmon, there is no reason not to continue maintaining and monitoring this facility.

In considering the need for fish passage at Millinocket Lake, Dolby, Millinocket, and East Millinocket Mills dams, we note the absence of anadromous species and the lack of significant production habitat for landlocked salmon in portions of the project area where these dams are located. The project as proposed would not significantly affect passage or require installing passage facilities. Smelt drift, which appears to be the major fish entrainment phenomenon in the West Branch, has not adversely affected the smelt population and is the primary basis for maintenance of the existing salmon stocks throughout the system. We agree with DIFW that expanding the range of smallmouth bass and pickerel throughout the West Branch probably would adversely affect the excellent landlocked salmon fisheries there. Fish passage facilities would not benefit fisheries at these developments, might adversely affect landlocked salmon stocks, and, therefore, should not be constructed. PIN, with their comments on the DEIS, expressed concern regarding the status of the American eel oopulation in the Penobscot River basin and suggested that measures to enhance eel migration be incorporated into licensing alternatives for these projects. As discussed in section 3, there has been an apparent decline in the Penobscot River eel stock in recent decades. However, there have been no alterations of the Ripogenus and Penobscot Mills projects during this time, and thus there is no basis for concluding that the existing operation of these projects has had any impact on eels. Both juvenile and adult eel are robust migrants. capable of passing through very limited amounts of water and even over land during wet conditions (Flagg, undated). This is evident from the fact that eels occur throughout project waters under existing conditions. The existing leakage in the Back Channel is sufficient to ensure eel migration through that portion of the river, and, with no other modifications in project developments, eels will continue to move among project waters as they have during the past license term for the project. We conclude that no specific measures need to be taken to enhance eel passage at these projects.

4.4.1.7 Cumulative Impacts

No anadromous fish stocks occupy the West Branch, and project waters are inhabited by a mix of warmwater and coldwater species, depending on local habitat conditions (see section 3.5). Important populations of smallmouth bass and pickerel inhabit lower project impoundments, but impoundment and river reach populations do not interact (i.e., they do not migrate between waters); therefore, there would be no cumulative impacts on individual stocks of these species.

Landlocked Atlantic salmon is the only species that could experience cumulative impacts because its life history frequently involves residence and growth in impoundments and migration into tributaries or rivers to spawn. This kind of life history behavior, however, is not observed in the Ripogenus Project and Penobscot Mills Project areas. The primary wild and self-sustaining landlocked salmon stock in the project areas is located in the West Branch, downstream of the Ripogenus dam and McKay station. This salmon stock appears to spend its entire life in the riverine environment rather than rearing and maturing in North Twin impoundment and migrating into the West Branch to spawn (see section 3.5.1.1). Very limited salmon spawning occurs downstream of North Twin dam, and most fish present there are of hatchery origin.

To the extent that a small portion of the West Branch stock may move downstream past the North Twin dam, the fish passage facilities at that dam provide for migration back upstream to principal spawning grounds in the West Branch. We believe, however, that these fish represent an insignificant portion of the total salmon stock. Because no single landlocked salmon stock occupies all project waters, and the West Branch stock appears to be permanently resident in the riverine West Branch, we conclude that the projects as proposed would have no significant cumulative impacts on landlocked salmon stocks.

4.4.2 Alternative 1

4.4.2.1 Upper Gorge Flows

Under Alternative 1, minimum flows in Upper Gorge would be 100 cfs from July 1 to September 30 and 50 cfs during the remainder of the year. As in the Applicant's Proposal, this flow enhancement would increase the adult salmon habitat in Upper Gorge from 41,869 to 71,707 square feet during the summer. With flows of 50 cfs instead of leakage during the winter and spring, however, adult habitat would be 51 percent greater than under the Applicant's Proposal, increasing from 41,869 to 63,298 square feet. Juvenile habitat would increase by about 33 percent, but habitat for fry would decline by about 70 percent (figure 4-3).

Although the IFIM study provides a basis for concluding that physical habitat would increase significantly under this flow regime, Upper Gorge is subject to high-volume, high-velocity spills that negate the benefits of a continuous minimum flow by eliminating suitable habitat. These spillage flows are most likely to occur during winter and spring, and the effect of high-volume spillage would be most harmful for spawning and early life stages of salmon. The flow regime proposed for Alternative 1 probably would not enhance the production of salmon in Upper Gorge, but would simply provide additional overwintering habitat in an area where such habitat is not limiting population size.

The winter/spring flow enhancement in Upper Gorge under this alternative would not significantly enhance landlocked salmon stocks in this segment of the West Branch and would provide no lasting benefits to fisheries resources.

4.4.2.2 West Branch Flows

We identified no flow measures beyond those proposed by GNP for the West Branch that would provide further fisheries enhancement; therefore, for West Branch flows, this alternative would have the same impacts as the Applicant's Proposal.

4.4.2.3 Millinocket Stream Flows

Under Alternative 1, minimum flow in Millinocket Stream would be 60 cfs year- round. IFIM study findings (see figure 4-2) show that flows of this magnitude would double salmon spawning habitat compared with conditions under the existing 20-cfs flow, increase adult habitat by about 30 percent, cause small declines in fry habitat, and increase juvenile habitat by about 25 percent.

This flow would increase spawning habitat for smallmouth bass by about 10 percent; habitat for all other life stages would decrease (figure 4-2). Increased winter flows under this alternative would protect all species and life stages of fish from ice formation better than the current winter flow of 20 cfs does.

Only juvenile and adult life stages of salmon would be present in Millinocket Stream during the fall and winter. IFIM study results suggest that habitat for those life stages increases by 25 percent to 30 percent between 20 cfs and 60 cfs, suggesting that the 60 cfs flow regime might enhance overwintering populations of those life stages. Also, this winter flow would protect redds from ice formation and, thus, improve the reproductive success of salmon in Millinocket Stream. The very small number of spawning salmon under the existing 20 cfs flow (a maximum of 20 redds counted in the 3 years surveyed), however, leads us to conclude that such modest increases in seasonal habitat would produce only a small, if any, numerical increase in the salmon stock size compared with the GNP flow proposal. Downstream project waters have well-established warmwater fish communities that would compete with the salmon; therefore, we find that salmon production in Millinocket Stream would not increase significantly. We also conclude that this alternative would have no significant benefit for smallmouth bass.

We conclude that this alternative would enhance the existing, small salmon stock in Millinocket Stream but that this enhancement would not substantially increase the number of salmon available to the fishery in the area.

4.4.2.4 Back Channel Flows

Alternative 1 includes a year-round minimum flow of 350 cfs in the Back Channel. This flow was selected based on results of the IFIM study (figure 4-3), which show that this flow would increase the amount of habitat for all life stages of salmon and that the benefits for juveniles would be maximized (approximately 460 percent habitat increase compared with existing conditions with leakage flows).

Although not addressed in the IFIM study, species other than salmon, such as brook trout, eel and a variety of minnow and sucker species, could benefit from the flow increase, as would the macroinvertebrate community. Eel passage may be improved slightly, but we have no data by which to establish the extent of such improvement. Habitat for adult and spawning stages of landlocked salmon, the key gamefish of concern, would be very limited (i.e., about 80,000 square feet of spawning and adult habitat would be present at 350 cfs).

All life stages would have to be supported to establish a permanent self-sustaining salmon stock. Based on the quantity of WUA documented in the IFIM study (figure 4-3), the greatest enhancement possible in the Back Channel is creation of a nursery area for fry and juvenile salmon at 350 cfs. The very small amount of adult habitat available (less than 100,000 square feet) under this enhanced flow regime and the very limited amount of spawning habitat present suggests that reproduction adequate to saturate the increased fry

and nursery habitat would be unlikely. Grand Falls poses a barrier to movement of most fish from the migration of Shad Pond into a major stretch of the Back Channel under low flow conditions, and could hinder adults that matured downstream as they attempted to migrate upstream to the limited spawning habitat that would exist under this flow regime.

Young salmon reared in the Back Channel under the enhanced fisheries flows might contribute to adult populations in Shad Pond, Dolby Pond, or other waters downstream of the Millinocket Development, if they dispersed to those waters and survived. Spawning habitat is limited, however. Full production from the nursery habitat created would probably require stocking hatchery fry, a management action not substantially different than the existing DIFW program of stocking waters downstream of the Back Channel. Young salmon moving into Shad and Dolby ponds from the Back Channel nursery habitat would be exposed to predation by smallmouth bass and pickerel. Any salmon that survived through the juvenile stage would compete with these species for available forage, most likely smelt drift through Millinocket Development. In addition, the shallow, warm waters of the downstream impoundments are not good salmon habitat, as evidenced by the limited salmon fisheries supported by DIFW stocking efforts. Only three salmon were found in GNP's creel surveys conducted in Dolby Pond between 1986 and 1989 (GNP, 1991b). No salmon were found in Shad Pond surveys.

We also conclude that young life stages of a salmon established in the Back Channel in response to enhanced minimum flows probably would be displaced by periodic high-volume spillage (as great as 29,000 cfs). Recolonization by upstream movement of displaced juvenile fish would be constrained by Grand Falls, reinforcing our view that stocking would be required to make full use of the nursery area created by higher minimum flows.

Lack of water for enhancing flow in the Back Channel also could reduce the likelihood of establishing a sustainable salmon population in this reach. In dry years, water may not be available to provide all of the enhancements discussed under Alternative 1 (see section 4.2 and Appendix D for a thorough discussion of the water use model). In this case, enhancements based on WQC conditions, which are mandatory, would receive the highest priority. Other enhancements such as the Back Channel flows, would be provided only if there is enough water remaining after the mandatory conditions are met; therefore, fisheries resources in the Back Channel may be exposed to dewatering in dry years.

Our evaluation of the 350 cfs flow regime suggests that, although habitat availability for some salmon life stages would be significantly increased, this enhancement would not substantially increase the availability of adult landlocked salmon to area fisheries. The aquatic ecosystem, however, generally would be enhanced (e.g., increased benthic invertebrate abundance, increased populations of forage species such as minnows).

4.4.2.5 Impoundment Draw-downs

Under this alternative, flow enhancements in Upper Gorge, Millinocket Stream, and the Back Channel would cause minor changes in the magnitude of draw-downs on North Twin and Ripogenus impoundments (Appendix D, table D-4). Based on hydrological modeling conducted by GNP and reviewed by the staff, the maximum range of the impoundment levels over 12 months would be no more than 1.0 foot greater than historical conditions and no more than 0.5 foot greater than the Applicant's Proposal (Ripogenus), even if flows of 350 cfs were

provided in the Back Channel. In impoundments as large as North Twin and Ripogenus, these changes would be biologically insignificant; therefore, we conclude that this alternative would have no draw-down-related impacts on fisheries resources beyond those that occur under existing conditions or the Applicant's Proposal.

4.4.2.6 Fish Passage

We identified no fish passage measures beyond those proposed by GNP that would provide further fisheries enhancement; therefore, for fish passage, this alternative would have the same impacts as the Applicant's Proposal. To the extent that higher flows in the Back Channel and elsewhere might facilitate eel migration, this alternative may be more beneficial to eels than GNP's proposal.

4.4.2.7 Cumulative Impacts

Enhanced flows in Upper Gorge, Millinocket Stream, and the Back Channel may enhance site-specific fish stocks to some degree and benefit the aquatic ecosystem at those sites. Because single, projectwide stocks of fish species do not exist except for American eel, however, we conclude that these site-specific enhancements would have no cumulative impact on fisheries resources. These enhancements might have a slight beneficial effect on eels.

4.4.3 Alternative 2

4.4.3.1 Upper Gorge Flows

We identified no flow measures beyond those proposed by GNP for Upper Gorge that would provide further fisheries enhancement; therefore, for Upper Gorge flows, this alternative would have the same impacts as the Applicant's Proposal.

4.4.3.2 West Branch Flows

We identified no flow measures beyond those proposed by GNP for the West Branch that would provide further fisheries enhancement; therefore, for West Branch flows, this alternative would have the same impacts as the Applicant's Proposal.

4.4.3.3 Millinocket Stream Flows

We attempted to identify a flow regime for Millinocket Stream that would provide greater fishery benefits than the Applicant's Proposal but would have less economic impact on GNP than the flows proposed under Alternative 1. We noted previously that only juveniles, adults, and spawning redds would be present during winter. The IFIM study findings (figure 4-2) show that adult salmon habitat increases slightly between 20 cfs (GNP's proposed flow) and 30 cfs; juvenile salmon habitat increases by about 15 percent between these two flows, and there is a slight increase in smallmouth bass spawning habitat. This was the basis for our incorporating a 60 cfs/30 cfs flow regime in our DEIS analysis.

During 10(j) discussions with Interior, we modified our recommended flow regime to 60 cfs year round. A spring/summer flow of this magnitude results in a doubling of juvenile habitat that could enhance a resident salmon stock, and the 60-cfs flow during the winter incubation period could increase production by protecting redds and juveniles against the effects of ice formation. Adults would not benefit as much as juveniles would; however, adults probably have a greater range of movement than juveniles and, thus, could overwinter in downstream waters, including impoundments. As discussed for Alternative 1, however, Millinocket Stream has little value as a salmon production area for other project waters or for Millinocket Stream. We conclude that our proposed flow regime would provide some enhancement for salmon beyond that provided by GNP's proposal but would not contribute substantially to regional salmon stocks.

4.4.3.4 Back Channel Flows

Under Alternative 2 we considered releasing a continuous minimum flow in the Back Channel that is significantly greater than the existing leakage/spillage flows. We evaluated if the increased flow might permanently enhance the aquatic environment and the fish community and if such an environment would significantly enhance populations of important gamefish species, particularly landlocked salmon. The enhanced the Back Channel environment might increase salmon stocks in two ways: by creating a self-sustaining resident stock in the Back Channel that would support a fishery there, and by serving as production waters for salmon that would disperse downstream. Although we conclude that the additional flow would definitely increase total available habitat for many elements of the aquatic ecosystem, the landlocked salmon stock would not be substantially increased.

Figure 4-3 presents the changes in habitat quantity for various life stages of salmon at three flows (50 cfs, 100 cfs, and 165 cfs) that are intermediate between leakage and the 350 cfs proposed under Alternative 1. As described earlier, habitat for salmon fry and juveniles increases most (more than 100 percent and 300 percent, respectively) when flow increases from leakage to 50 cfs. Additional increases in flow continue to produce gains in habitat for all life stages, but at a reduced rate.

The value of the Back Channel for fisheries management is limited by lack of adequate spawning and adult habitat, probable high predation from warmwater spills in downstream waters, and spillage flows that would probably displace young life stages fish populations. Flow greater than the leakage/spillage flow regime included in the Applicant's Proposal, therefore, is not warranted because of the minimal benefits for regional fisheries. We did not include any additional flow release in Alternative 2; therefore, the impacts of this alternative would be the same as those of the Applicant's Proposal.

4.4.3.5 Impoundment Draw-downs

Under this alternative, flow enhancements in Upper Gorge, Millinocket Stream, and the Back Channel would cause minor changes in the magnitude of draw-downs in North Twin and Ripogenus impoundments (Appendix D, table D-5). The maximum range of the impoundment levels over a 12-month period would increase by no more than 0.5 feet (North Twin), even if flows of 165 cfs are provided in the Back Channel. These changes would be biologically

insignificant; therefore, this alternative would have no draw-down-related impacts on fisheries resources beyond those under existing conditions or the Applicant's Proposal.

4.4.3.6 Fish Passage

We identified no fish passage measures beyond those proposed by GNP that would provide further fisheries enhancement; therefore, for fish passage, this alternative would have the same impacts as the Applicant's Proposal.

4.4.3.7 Cumulative Impacts

The only different flow measure included in Alternative 2 is enhanced flow in Millinocket Stream. This flow may enhance site-specific fish stocks to some degree and benefit the aquatic ecosystem at the site. Because no single, projectwide stocks of fish species exist except for American eel, however, we conclude that this site-specific enhancement would have no cumulative impact on fisheries resources.

4.4.4 No-action Alternative

Under existing conditions, no change would be expected in the high-quality landlocked salmon fishery in the West Branch, and no fisheries enhancements would be provided elsewhere in the project area.

4.4.5 Summary

Because of the complexity of the proposed changes and other alternatives with regard to fisheries resources, a summary of our findings is provided. Effects of each alternative are contrasted against baseline conditions that would persist under the No-action Alternative.

- Under the Applicant's Proposal, adult salmon habitat in Upper Gorge would increase in summer and fall, but the additional habitat would produce no increase in salmon stock; the landlocked salmon stock in the West Branch below McKay would be enhanced because of additional protection from outage flows and additional spawning and nursery habitat (Holbrook Pool); certain life stages of salmon and smallmouth bass would benefit <u>seasonally</u> from increased flows in Millinocket Stream, but stock size would not change; and lake trout spawning success would probably increase in North Twin because of regulated draw-downs.
- Under Alternative 1, adult salmon habitat in Upper Gorge would increase <u>vear-round</u> but would produce no lasting enhancement of salmon stocks; enhancements of West Branch salmon stocks would be the same as under the Applicant's Proposal; certain life stages of salmon and smallmouth bass would benefit <u>vear-round</u> from increased minimum flows in Millinocket Stream, resulting in a small increase in regional salmon abundance; habitat for certain life stages of salmon would increase in response to increased minimum flows in the Back Channel, but a substantial self-sustaining resident landlocked salmon stock probably would not be established; lake trout populations in North Twin impoundment would benefit from

protection of redds because of regulated draw-downs; and impoundment fish stocks would not be affected by draw-downs.

• Under Alternative 2, effects on fisheries in Upper Gorge, West Branch, the impoundments and Millinocket Stream would be the same as for Alternative 1.

4.5 WETLANDS

4.5.1 Applicant's Proposal

4.5.1.1 Draw-down and Flow-related Effects

Ripogenus Project. Approximately 80 percent (1,000 acres) of the 1,251 acres of wetlands in the vicinity of the Ripogenus impoundment are at least partially hydrologically independent of the impoundment. GNP's proposed operation of the Ripogenus Project would continue to adversely affect the estimated 20 percent of wetlands that are hydrologically dependent on impoundment levels. Under GNP's proposed operation, these areas would continue to be exposed occasionally to freezing/thawing conditions during winter draw-downs. Summer drawdowns would continue to expose portions of the affected wetlands to occasional desiccation and scouring. Wetlands that are hydrologically associated with the impoundment contain plant species that are somewhat tolerant of fluctuating water regimes and are often not of high value to wildlife. This narrow band of affected wetlands occurs around the periphery of the impoundment and does not possess deep, organic substrates or sources of inflow independent of the impoundment (such as the 437-acre Brandy Pond wetlands).

As part of an agreement with Maine Professional River Outfitters (PRO), GNP agreed to provide seasonal daytime minimum flows of 2,400 cfs below McKay station from May 1 to September 15. Additionally, GNP agreed to provide a minimum flow of 1,800 cfs during wet or dry months, and a temporary flow of 400 cfs during power outages. The 528 acres of wetlands below McKay station on the West Branch are associated primarily with six deadwater areas (see section 3.7.3). GNP studied these deadwater wetlands under both the proposed, managed flow conditions (2,000 to 2,400 cfs) and unmanaged flow (1,000 cfs; GNP, 1992c). The results of the study indicate that flows of 2,000 to 2,400 cfs between Ripogenus dam and the North Twin impoundment in the West Branch are beneficial to these high-quality wetlands, whereas flows that simulate run-of-river conditions (1,000 cfs average) could result in an eventual net loss of about 90 acres of these wetlands.

Our review of GNP's managed/unmanaged study results shows that the proposed flows below McKay station would not adversely affect the six primary areas of deadwater wetlands on the West Branch. Because the proposed flows below McKay station would not be substantially different than existing flows, the proposed flows would not affect the current hydrological regime of deadwater wetlands. Indeed, such flows are likely to help to preserve their current values.

The applicant's proposed summer flows (July 1 to September 30) of 100 cfs and the 12 cfs minimum flow for the remainder of the year could enhance the sparsely vegetated (unquantified) wetlands in the river channel. Upper Gorge is extremely rocky and steep-sided, and these conditions have precluded substantial wetlands development. Under existing

conditions, Upper Gorge receives only leakage flows of about 12 cfs throughout most of the year and brief spillage flows up to 11,600 cfs during spring runoff. Continuous flows of 100 cfs during part of the growing season could enable any existing wetlands vegetation to expand into broader areas within Upper Gorge; however, wetlands vegetation in the vicinity would be further benefitted with a greater extended hydroperiod (see section 4.5.2).

We conclude that although most wetlands in the Ripogenus Project area would not be adversely affected by GNP's proposed operation, the negative effects of draw-downs on the limited amount of affected shoreline wetlands would decrease value to wildlife (see section 4.6.1.2). We estimate that the approximately 250 acres of shoreline wetlands around the periphery of the impoundment that are entirely dependent on impoundment levels would continue to be most affected by the project draw-downs.

Penobscot Mills Project. GNP's proposed operation of the Penobscot Mills Project would not have any additional adverse effects on wetlands resources beyond the effects of existing operations. Current operation of the Penobscot Mills Project negatively affects wetlands areas that are hydrologically dependent on the impoundments (i.e., no independent inflow from tributaries or moisture-retaining substrate), primarily narrow margins along the North Twin impoundment shoreline. These wetlands are affected during impoundment drawdowns, especially when the wetlands vegetation may be exposed to winter freezing/thawing conditions. Summer drawdowns would also continue to expose the affected wetlands to occasional desiccation and scouring. Wetlands around the peripheries of the Quakish, Ferguson, Dolby, and East Millinocket impoundments are not currently affected by project operations because water levels at these impoundments do not fluctuate significantly during run-of-river operation.

We estimate that approximately 75 percent (222 acres) of the 296 acres of the wetlands in the vicinity of the North Twin impoundment are at least partially hydrologically independent of the impoundment. Wetlands hydrologically associated with the North Twin impoundment are generally small and composed of plant species that are somewhat tolerant of fluctuating water regimes. These areas around the impoundment shorelines are characterized by substrates ranging from large boulders to small rocks and coarse sand; they lack the water storage capacity of deep organic substrates and independent sources of inflow. GNP's proposed operation of the Penobscot Mills Project would continue to adversely affect the estimated 25 percent of wetlands that are hydrologically dependent on North Twin impoundment levels.

Around Millinocket Lake, wetlands total 709 acres (see section 3.6). GNP suggests that most of these wetlands are hydrologically independent of the existing impoundment levels (e.g., Grant Brook wetlands). We concur, based partly on our field observations of the deep organic substrates and the independent sources of inflow in these wetlands during a dry period in August 1993.

Continued run-of-river operation at the Quakish Lake, Ferguson Pond, Dolby Pond, East Millinocket, and Millinocket Lake developments, and in the sections of the West Branch from the North Twin impoundment to Quakish Lake and from the Millinocket tailrace to Dolby Pond, would maintain stable, *status quo* wetlands conditions in these areas. Extensive wetlands are not present around Millinocket Stream because of the flow regulations on Millinocket Stream (currently, leakage flow from Millinocket impoundment, 20 cfs) and its surrounding topography. Several small deadwaters within Millinocket Stream support limited areas of emergent and scrub/shrub wetlands. Wetlands resources would benefit from the proposed 60-cfs flow from May 1 to October 15 (with a minimum flow of 20 cfs the remainder of the year). Extending the hydroperiod of these wetlands during the growing season would enhance the quantity and the quality of the existing wetlands, making these areas more valuable to wildlife; however, such wetlands enhancement could be greater with the permanent extension of the hydroperiod (see section 4.5.2).

Several small wetlands occur along the Back Channel, particularly in areas protected by beaver dams and large boulders. Currently, the main portion of the Back Channel receives only leakage flows from Stone dam (2 to 5 cfs) and inflows from several small tributaries. These flows would be maintained under the Applicant's Proposal. Existing wetlands along the Back Channel probably would remain stable under the proposed leakage flows.

GNP's proposed leakage flows would adequately maintain the existing small quantity of wetlands in the Back Channel. We estimate that wetlands resources probably would not be greatly expanded or enhanced, even with much higher continuous flows (such as proposed by FWS and CI and discussed in section 4.6.2), because of the very rocky substrate over virtually all of the Back Channel. Such increased flows could disrupt the existing beaver population. Losses of the beavers and their dams in certain areas of the Back Channel could adversely affect wetlands that are directly associated with the beaver dams. We conclude, therefore, that existing leakage flows are appropriate for the protection of existing wetlands resources along the Back Channel.

Most of the project wetlands would not be adversely affected by GNP's proposed operation, but the negative impacts on the small amount of affected shoreline wetlands caused by draw-downs should be compensated for (see section 4.5.1.2). Wetlands most affected by the project draw-downs are those areas around the periphery of the impoundment in which hydrology is entirely dependent on impoundment levels.

4.5.1.2 Wetlands Enhancements

Ripogenus Project. GNP offers no enhancements that would benefit wetlands at the Ripogenus impoundment. FWS and CI request that GNP provide both onsite and offsite wetlands mitigation (letters from D. Sosland, CI, September 3, 1993; G. Beckett, FWS, September 2, 1993). GNP identified Quaker Brook and Black Pond on the Ripogenus impoundment as potential candidates for wetlands enhancement but proposes no wetlands enhancement at either site (figure 4-4). GNP identified the two sites as having a total of about 350 acres of low-quality shoreline wetlands; both sites are directly adjacent to wetlands with high existing wildlife values.

GNP indicates that constructing dumped-earth structures to increase water retention during draw-downs is feasible at the Quaker Brook and Black Pond sites. GNP estimates that about 250 acres of wetlands would be enhanced at the Black Pond site, and 100 acres at the Quaker Brook site.



Figure 4-4. Wetlands enhancement sites evaluated for feasibility by GNP at the Ripogenus Project, 4-6 August 1992 (modified after figure 1, GNP, 1992e)

We believe that wetlands enhancements within the project area are appropriate to improve overall habitat conditions, particularly in currently marginal wetlands areas that are adjacent to ecologically valuable areas and presently affected by project operations. As we discuss below, we believe that the on-site candidate locations for wetlands enhancement (both at Ripogenus and Penobscot Mills projects) are of adequate size to compensate for the total wetlands acreage affected by project operations and, thus, we did not investigate potential offsite wetlands enhancement sites.

Penobscot Mills Project. GNP proposes wetland enhancements at the Penobscot Mills Project; the WQC also requires wetlands enhancements. FWS and CI request that GNP provide both onsite and offsite wetlands mitigation (letters from G. Beckett, FWS, September 2, 1993; D. Sosland, CI, September 3, 1993). GNP identified three sites on the North Twin impoundment, Deep Cove East, Deep Cove West, and Wadleigh Brook, as potential candidates for wetlands enhancement (figure 4-5). The three sites are all good candidates for wetlands enhancement (because they are currently low-quality wetlands that are adversely affected by impoundment draw-downs and are directly adjacent to higher quality wetlands.

Requiring wetlands enhancements within the project boundaries is appropriate. The Deep Cove East and Deep Cove West sites are good candidates for enhancement because they could increase the total acreage of project wetlands that possess high functional and wildlife values.

GNP proposes to provide increased water retention during project draw-downs by modifying an existing causeway at Deep Cove East and Deep Cove West and by constructing a dumped-earth structure at Wadleigh Brook. GNP estimates that about 30 acres of wetlands at the Deep Cove sites and about 15 acres at the Wadleigh Brook site would be enhanced. None of GNP's proposed enhancements would require complicated water control structures or other devices that would be difficult to operate.

GNP states that the proposed dumped-earth enhancement structure at Wadleigh Brook may not be acceptable to the wetlands regulatory agencies, and that the site also would involve potential access constraints because of existing wetlands; therefore, we conclude that the site is not appropriate for wetlands enhancement (see discussion of Alternative 2, section 4.6.3).

4.5.2 Alternative 1

Ripogenus Project. This alternative would have one potential minor impact on wetlands resources in addition to the impacts of GNP's proposal. Greater than normal drawdowns of the Ripogenus impoundment during dry years, primarily during winter months (to allow for increased flows below McKay station), could increase adverse effects on impoundment wetlands. We conclude, however, that these negative effects would be inconsequential because they would be so infrequent and of such short duration.

Under this alternative, both Quaker Brook and Black Pond would be used for wetlands enhancement. The combined sites would enhance about 350 acres (Quaker Brook, 100 acres; Black Pond, 250 acres) of existing, lesser-quality, shoreline wetlands. Enhancing both



Figure 4-5. Wetlands enhancement sites evaluated for feasibility by GNP at the Penobscot Mills Project, 4-6 August 1992 (modified after figure 1, GNP, 1992e)

Quaker Brook and Black Pond, however, would provide greater mitigation than is reasonable to compensate for the roughly 250 acres of presently affected wetlands (see section 4.6.3).

Upper Gorge flows of 100 cfs from July 1 to September 30 and 50 cfs during the remainder of the year would help to increase wetlands functions and values and the wildlife values of wetlands resources. Wetlands resources are generally lacking throughout Upper Gorge because of its rocky substrate, steep topography, and existing flows (see section 3.6). Additional flow enhancement would benefit Upper Gorge wetlands resources; however, the scope of potential new wetlands created or benefits to existing wetlands would be minimal in the context of wetlands resources throughout the project area.

Penobscot Mills Project. Implementing Alternative 1 would have two beneficial impacts, one potential minor adverse impact, and one adverse impact on wetlands resources in the Penobscot Mills Project area. Year-round flows of 60 cfs would benefit the existing areas of emergent and scrub/shrub wetlands in the several small deadwaters in Millinocket Stream. Permanent extension of the hydroperiod in these wetlands would better protect them from fluctuating (low) water levels, and additional small areas of wetlands could be created where sufficient organic substrate exists. Also, with the enhanced hydrology, plant communities that are more valuable to wildlife could colonize the existing wetlands. This additional flow enhancement would benefit Millinocket Stream wetlands resources.

Under this alternative, wetlands enhancements at the Deep Cove East and Deep Cove West sites would be implemented as proposed by GNP. The Wadleigh Brook site would be rejected because of existing wetlands access constraints. Crossing the existing wetlands with the equipment necessary to enhance the Wadleigh Brook site could damage them.

This alternative also would provide year-round minimum flows of 350 cfs in the Back Channel. Such flows probably would enhance some of the topographically higher existing wetlands along peripheral areas of the Back Channel but would flood others that are directly associated with beaver dams in the channel. These increased continuous flows could disrupt the existing beaver population. Loss of the beavers and their dams would result in at least some short-term loss of wetlands because of inundation along the Back Channel. Conversely, increased flows could create small areas of additional wetlands in peripheral areas and enhanceaquatic vegetation within the channel. These benefits would be relatively small due to the rocky substrate throughout the Back Channel. Additionally, any long-term gains in total wetlands area from such increased year-round flows are uncertain. Year-round minimum flows of 350 cfs in the Back Channel probably would provide only marginal net wetlands benefits.

4.5.3 Alternative 2

Ripogenus Project. Under Alternative 2, there would be one potential minor impact on wetlands resources in addition to those associated with GNP's proposal. Greater than normal Ripogenus impoundment draw-downs, primarily during winter months (to maintain flows below McKay station), could have minor, short-term adverse effects on impoundment wetlands. These negative effects would be so infrequent and of such short duration, however, that they would be inconsequential. Wetlands enhancements are appropriate at the Ripogenus Project and should be required. The Black Pond site is a good candidate for increasing the total acreage of project wetlands with high functional and wildlife values. The site supports about 250 acres of low-quality, shoreline wetlands that are directly adjacent to wetlands with high wildlife values. The scope of potential enhancement at this site is similar to the scope of existing degraded wetlands within the project area. GNP indicates that constructing a dumped-earth structure at the site could retain more water during project draw-downs. The structure, as described by GNP, would not require complicated water control valves or other devices that would be difficult to operate.

We conclude that GNP's proposed water retaining structures would be appropriate to enhance these wetlands; however, the dumped-earth structures may not be acceptable to the wetlands regulatory agencies.¹⁰ GNP should investigate additional types of enhancement structures for the two potential sites. We recommend that GNP file a detailed wetlands enhancement and monitoring plan that fully describes proposed activities at Black Pond with the Commission for approval. We do not recommend additional enhancements at Quaker Brook because enhancements at Black Pond (250 acres) would be adequate, and the cost of enhancement at Quaker Brook (only 100 acres enhanced for the same order of cost) would be disproportionately high.

Penobscot Mills Project. We believe that the two GNP proposed wetlands enhancements at the Penobscot Mills Project (Deep Cove East and West, totalling 30 acres) are appropriate, as we discussed under Applicant's Proposal. The Wadleigh Brook site is not acceptable for wetlands enhancement because of potential access problems, the potential for rejection of the proposed structure by the wetlands regulatory agencies, and the small potential enhancement, as we discussed above.

Alternative 2 would provide flows of 60 cfs from May 1 to October 15 and 30 cfs in the Millinocket Stream during the remainder of the year. Increased flows would benefit wetlands resources. Although this alternative would provide slightly more stable hydrology in Millinocket Stream wetlands than the GNP proposal would, we estimate that it would not provide the continuous, year-round hydrology of Alternative 1 (section 4.6.2) and, therefore, would not be as beneficial to wetlands.

Although wetlands are not particularly abundant in the Back Channel, they apparently provide relatively high wildlife values at year-round leakage flows. Our analysis indicates that neither the total area nor the quality of the wetlands for wildlife would be greatly increased. Wetlands resources, therefore, would be best served under the current leakage flows, in conjunction with GNP's the Back Channel Wildlife Habitat Management Plan.

4.5.4 No-action Alternative

The No-action Alternative would have no adverse effects on wetlands, but existing wetlands at the Ripogenus and the Penobscot Mills projects would not be enhanced.

¹⁰ GNP indicates that Maine DEP could require detailed analysis of alternatives and that "the acceptability [to Maine DEP] of utilizing a dumped-earth structure is presently unknown."

4.6 TERRESTRIAL RESOURCES

4.6.1 Applicant's Proposal

4.6.1.1 Site-specific Impacts

Ripogenus Project. Both CI and FWS are concerned about the use of herbicides along the existing 30.2-mile transmission line corridor between McKay station and Millinocket (letters from D. Sosland, CI, September 3, 1993; G. Beckett, FWS, September 2, 1993). CI maintains that the herbicide spray program degrades water quality and wildlife habitat within the corridor. The transmission corridor (which is all upland and is not influenced by the impoundment) is maintained by herbicides and selective cutting (80 percent of the right-of-way is maintained by herbicides, 20 percent by cutting; GNP, 1993c).

GNP specified areas of the corridor that have been and will be hand-cut and those that have been and will be sprayed between 1991 and 1998. GNP also provided information about environmental, timing, and safety requirements of its contractual agreement for aerial application of herbicides and data about the kinds of herbicides used. The agreement specifies that herbicides will continue to be applied annually by helicopter. GNP uses two herbicides: Banvel[®] 720 and Roundup[®]. The agreement also clearly states that all contractors must meet all federal and state application and disposal requirements, and that all herbicides used shall be approved for use by EPA and the Maine Department of Agriculture. GNP's contractual agreement also stipulates that herbicide contractors must maintain unsprayed buffers of 100 feet along all streams.

As described by GNP, the use of herbicides to control vegetation along the corridor would not adversely affect water quality or wildlife habitat. Alternatives to herbicides, such as increasing the proportion of cutting, could increase disturbance of wildlife by increasing the frequency of noise and direct human presence. Continued operation of the Ripogenus Project as proposed by GNP would not adversely affect or result in the loss of additional terrestrial resources. We do not object to GNP's use of herbicides in the corridor, provided that GNP remains in compliance with all applicable federal and state rules and regulations.

Penobscot Mills Project. The Back Channel Wildlife Habitat Management Plan, as presented by GNP, provides for the placement of duck boxes (exact number not specified) in appropriate locations in the overflow channel of the Back Channel. GNP indicates that the duck boxes would be cleaned and maintained annually in March, just before the nesting season. GNP's plan, however, does not name the party that would provide such annual maintenance or replace the duck boxes as needed. GNP also proposes to manage snags in the overflow channel of the Back Channel. Such management would maintain an optimal number of snags for use as perches by waterfowl and other avifauna.

GNP would manage approximately 2,300 acres of forest directly adjacent to the overflow channel of the Back Channel to enhance wildlife habitat under its plan. This would be accomplished by uneven-aged management of most of the forest to increase vertical stratification. Trees that produce hard mast (e.g., oaks, hickories) and trees that produce soft mast (e.g., maples, conifers) would be allowed to predominate in separate areas to maximize benefits to wildlife. Even-aged management would be used in areas along the tributaries that

flow into the main channel and in the southern part of the main channel, where beaver and grouse habitat could be created. The large white pines near the southern end also would be retained as perches and nesting trees for bald eagles.

Continued operation of the Penobscot Mills Project as proposed by GNP would not adversely affect or result in loss of additional terrestrial resources. We conclude that the Back Channel Wildlife Habitat Management Plan would be beneficial to terrestrial resources in the project area.

4.6.1.2 Cumulative Impacts

Operation of the Ripogenus and Penobscot Mills projects as proposed by GNP would have no adverse cumulative impacts on regional terrestrial resources because GNP proposes no significant alterations of terrestrial habitats.

4.6.2 Alternative 1

Operating the Ripogenus Project under Alternative 1 would cause no adverse effects on terrestrial resources. As indicated in section 2.3 and discussed in more detail in section 4.9, Alternative 1 includes establishing a 500-foot building setback with a 250-foot vegetative buffer around all project shorelines. Such measures would protect against future degradation of existing riparian habitat and habitats adjacent to the impoundment.

Operating the Penobscot Mills Project under Alternative 1 could have minor adverse effects on the existing beaver population in the Back Channel. The higher flows proposed under this alternative could force the beavers to move off the Back Channel to some of the smaller tributaries that flow into it. This could, in turn, adversely affect wetlands resources in the Back Channel (see section 4.5.2). Operating the Penobscot Mills Project under Alternative 1 would have no other adverse effects on terrestrial resources.

4.6.3 Alternative 2

Under Alternative 2, no adverse terrestrial impacts other than those due to the Applicant's Proposal would occur at the Ripogenus and Penobscot Mills projects. Under this alternative, however, we would require GNP to provide at least annual maintenance of the duck boxes specified in the Back Channel Wildlife Habitat Management Plan and to replace them as needed. Reports (to DIFW) of such maintenance and replacement would be required annually. As explained further in section 4.9, Alternative 2 includes establishing a 250-foot easement for 73 shoreline miles or a 200-foot boundary expansion on GNP-owned lands at the Ripogenus Project and a 200-foot boundary expansion on GNP-owned land around the impoundments at the Penobscot Mills Project. These increases would protect existing riparian habitat but less terrestrial habitat in the immediate vicinity of the impoundment than the larger zone proposed in Alternative 1.

4.6.4 No-action Alternative

The No-action Alternative would have no additional adverse effects on terrestrial resources.

4.7 THREATENED AND ENDANGERED SPECIES

4.7.1 Applicant's Proposal

Ripogenus Project. The bald eagle (*Haliaeetus leucocephalus*), a federally and statelisted endangered bird, breeds in several areas of the Ripogenus Project (see section 3.8.2). EPA, FWS, and other intervenors assert that impoundment level fluctuations are directly linked to elevated mercury levels in impoundment fish, and that this could adversely affect eagles.

GNP conducted a mercury study, which showed that project operations are not related to elevated mercury levels in fish within the project impoundment (see section 4.3). GNP believes that the bioaccumulation of mercury in the fish on which bald eagles feed is unrelated to project operation. The study showed that similar levels of mercury are found in fish from both unmanaged lakes (i.e., lakes without dams) and managed lakes throughout the region.

GNP asserts that the additional bioaccumulation studies proposed by several agencies and other intervenors are unjustified and that no further measures are necessary because no additional changes in operation are proposed. During the LURC hearings for the Ripogenus Project, GNP's wildlife experts and toxicologists testified that elevated mercury levels in the project impoundments are not related to impoundment fluctuations and that, although elevated levels of mercury in fish have been documented, no current evidence indicates that bald eagles are being adversely affected by exposure to mercury (LURC, 1993).

Other data show that although heavy metals such as mercury may be correlated with lower reproduction rates in raptors, these lower rates are also strongly linked to the simultaneous presence of other pollutants, such as DDE (metabolite of DDT), which has been found in most egg samples (Wiemeyer et al., 1984). Another study indicates that bald eagle nests along the West Branch of the Penobscot River and its associated lakes have an average production rate of 0.94 young per occupied nest, compared with the production level of 1.00 young per occupied nest regularly attained by other major breeding populations of eagles (Welch, 1991). Such differences in fecundity of hypothetically healthy populations and the existing West Branch population are probably statistically insignificant.

The newest research by Welch (1994) provides no new evidence that would indicate a positive correlation between mercury contamination and eagle productivity. The same is true for PCBs, but a positive correlation was found for DDE. Neither of these contaminants are associated with project operations. Further, this research provides no evidence that elevated levels of mercury in Maine's eagles are linked to impoundment fluctuations. The data show that mercury levels in the blood and feathers of eaglets is significantly higher in lacustrine situations than in marine, estuarine, or riverine situations in Maine; however, data from only a few fluctuating impoundments among a variety of lakes were apparently considered in the Welch (1994) data set. The research also points out that mercury concentrations recorded in the blood of Maine eaglets throughout the state are similar to levels observed throughout Florida, where very few hydroelectric impoundments exist, and that eaglets sampled in Oregon had blood mercury levels approximately five times higher than concentrations measured eaglets in Maine.

We conclude that nothing in the current literature directly or indirectly links impoundment draw-downs with elevated levels of mercury in bald eagles. Further, no evidence suggests that the fecundity of the eagle population at the Ripogenus Project is affected by current draw-down operations. We conclude that operation of the Ripogenus Project as proposed by GNP would have no demonstrable adverse effects on the bald eagle population in the project area.

The common loon (*Gavia immer*), a bird species of concern in Maine (described in section 3.8.2), breeds within the Chesuncook Lake portion of the Ripogenus impoundment. GNP indicated that operation of the impoundment may affect nesting success slightly; however, GNP stressed that nest site selection takes place early (late May to late June). In mid-May, water levels are typically within 1 foot of their highest annual level at the impoundment; therefore, most of the loon nesting cycle occurs after the danger of flooding has passed (GNP, 1991a). Also, according to GNP, receding water levels of more than 2 feet typically do not occur until the latter part of the nesting cycle, which minimizes some effects on the broods (because of timing) but increases other effects, such as predation. Impoundment fluctuations similarly affect common tern (*Sterna hirundo*) and ring-billed gull (*Larus delawarensis*), both uncommon breeding birds in Maine.

We conclude that detrimental effects on the common loon, common tern, and ringbilled gull may be occurring at the Ripogenus Project, and that enhancement measures may be appropriate (see Alternative 2, section 4.7.3).

FWS indicated that the long-tailed shrew (*Sorex dispar*), listed as a C3 species (more abundant than previously believed), and the North American lynx (*Felis lynx*), a C2 species for future federal listing, have ranges that overlap the project area (letter from G. Beckett, FWS, to J. Carson, GNP, February 12, 1990; see section 3.8.2). GNP did not identify either of these species during wildlife surveys; however, even if either or both of these species exists at the project, it is unlikely that proposed operations would have any adverse impacts on them because of their relative mobility. Because of its current listing status, we did not consider the long-tailed shrew an important species in terms of our review.

FWS also indicated in its letter of February 21, 1995, that several species whose ranges potentially overlap the project boundaries should be added to the list of potential species of concern for the projects. These species, all federal C2 candidates, include yellow lampmussel (*Lampsilis cariosa*), brook floater (*Alasmidonta varicosa*), extra striped snaketail dragonfly (*Ophiogomphus anomalus*), and midget snaketail dragonfly (*Ophiogomphus howeii*). The extra striped snaketail dragonfly and the midget snaketail dragonfly have also been proposed as state listed threatened and endangered species, respectively. FWS stated that information pertaining to the distribution of all these species is severely limited. These species have never been identified within the project boundaries, and it is not likely that proposed operations would affect them adversely.

Penobscot Mills Project. Bald eagles breed in several areas of the Penobscot Mills Project (see Ripogenus Project discussion).

Common loon, a bird species of concern in Maine (described in section 3.8.3), breeds on the North Twin impoundment, Millinocket Lake, and Quakish Lake (see discussion of impoundment level effects on common loon breeding for the Ripogenus Project). GNP's proposal to stabilize the North Twin impoundment levels from May 1 to September 15 should help to ameliorate possible impacts (i.e., loss of eggs and chicks) of flooding of nest sites during the early portion of the nesting cycle (mid-May to late June). Additional wildlife enhancements are necessary at the Penobscot Mills Project to compensate for these detrimental effects on the common loon (see Alternative 2, section 4.7.3).

The sedge *Carex oronensis*, a Category 2 species, was observed a considerable distance from the North Twin impoundment shoreline during a botanical survey conducted by GNP but was never found within the limits of the Penobscot Mills Project (see section 3.8.3). GNP's proposed operation of the project probably would not affect this plant adversely.

4.7.2 Alternative 1

This alternative, which reflects the concerns and requests of the CI, would have no adverse effects on threatened and endangered species in either project area, other than those that would occur under GNP's proposal. Several groups, including CI, however, suggest that artificial nesting platforms would help to alleviate some of the effects of impoundment fluctuations on nesting common loons and other aquatic birds (see discussion in section 4.7.3).

4.7.3 Alternative 2

Alternative 2 would produce no additional adverse impacts on threatened and endangered species. We recommend, however, that GNP confer with DIFW and FWS to preserve existing eagle perching areas and to investigate the appropriateness of creating new perching areas around the periphery of the impoundments.

Several groups, including CI, suggest that artificial nesting platforms would help to alleviate some of the impacts on nesting common loon due to impoundment fluctuation. Common tern and ring-billed gull, both uncommon breeding birds in Maine, also experience similar nesting impacts due to impoundment fluctuations. Providing artificial nesting platforms would be an effective and relatively inexpensive means to improve the nesting success of common loon and other aquatic birds.

We agree that GNP should use artificial, floating nest structures to help increase survival of nesting loons and other aquatic birds at impoundments at both the Ripogenus and the Penobscot Mills projects. GNP should develop plans for design and implementation of these structures in conjunction with DIFW.

4.7.4 No-action Alternative

The No-action Alternative would not affect threatened and endangered species.

4.8 RECREATION RESOURCES

Potential recreation impacts associated with the proposed projects relate to flows, water level fluctuations, fisheries, access, facilities, and cumulative impacts. In the following section, we discuss each of these impact areas under each alternative.

4.8.1 Applicant's Proposal

4.8.1.1 Recreation Flows

Proper river flows are critical to support recreational boating in both the impoundments and the river system affected by project operations.

Ripogenus Project. GNP voluntarily provides flows of at least 2,000 cfs during normal months and at least 1,800 cfs during wet or dry months throughout the primary recreation season (May 1 to September 15). As part of a settlement agreement with MEPRO, GNP would increase minimum daytime flows according to the schedule presented in table 4-6 and would establish a telephone message system to provide information about flow conditions along the West Branch and scheduled releases from Ripogenus Dam, notify a designated representative of MEPRO of any unscheduled releases from Ripogenus dam, and provide readily visible staff gauges calibrated to river flow levels at McKay station, Abol Bridge, and Telos Bridge (GNP, 1993b). Following an emergency outage, GNP would restore these flows as quickly as possible, but in no more than 3 days.

Table 4-6. Minimum flow schedule (cfs; Source: MEPRO, 1993)				
	May 1 to September 15			September 16 to October 1
	Sat & Sun	Mon & Fri	Tues to Thurs	Sat & Sun
Normal Year	2,300	2,200	2,000	2,300
Wet/Dry Year	2,200	2,000	1,800	2,200

Several agencies and interest groups recommend alternative minimum flows for the West Branch of the Penobscot River below McKay station, which is the principal whitewater boating segment in the project area. The American Whitewater Affiliation (AWA) and the National Park Service (NPS) both request a 2,400 cfs minimum flow for whitewater boating (letters from AWA, September 27, 1989; D. Haas, NPS, March 2, 1992), which was MEPRO's original recommendation (letter from MEPRO, October 18, 1989). FWS recommends reducing the flow below 2000 cfs to allow additional water storage to meet downstream fish and wildlife needs (letter from G. Beckett, FWS, to J. Carson, GNP, December 16, 1991).

Flow is one of the most important variables affecting the whitewater boating experience. Research indicates an inverted-U relation between recreation guality and flow (Brown et al., 1991); flows below a certain level are unusable. Above that minimum, recreation quality rises with flow, plateaus at an intermediate range, and then drops as flows reach unsafe levels. Figure 4-6 estimates this relationship for the West Branch based on the staff's evaluation of a video showing whitewater rafting at flows ranging from 1,400 cfs to 3,600 cfs (GNP, 1992c) and the results of the 1990 whitewater boating survey (GNP, 1991b). Although flow ranging from 1,700 to 3,000 cfs did not statistically affect the satisfaction of rafters, the survey did not address first-time rafters who may have had no basis for comparing alternative flow levels. Private boaters responding to the survey preferred higher (above 3,000 cfs) and lower flows (below 2,000 cfs); they found intermediate flows (2,000 to 2,600 cfs) less satisfying. Many of these paddlers had extensive experience on the West Branch, averaging 43 trips, and were more familiar with the river at various flows. MEPRO indicates that 1,800 cfs is the minimum flow necessary for a reasonable rafting experience, and approximately 2,400 cfs is necessary for a quality rafting experience that would encourage repeat business (letter from J. Connelly, MEPRO, to J. Carson, GNP, April 13, 1991).

Increasing the minimum flow at McKay station from 2,000 cfs to 2,300 cfs on weekends during years with normal rainfall would significantly improve the quality of whitewater rafting while at least maintaining the quality of paddling (figure 4-6). The West Branch already is recognized as one of the premier whitewater rafting rivers in New England, especially during the summer, when regulated flows enable sustained high-quality rafting. Given this existing reputation, increased minimum flows for recreation are not likely to attract significant numbers of new customers. Improving the recreational experience, however, probably would increase the number of repeat rafting customers, which both MEPRO and DOC contend is critical for the continued success of the commercial boating outfitters.

Although the proposed flows would be less than the 2,400 cfs recommended by AWA and NPS, these flows would be minimums, not averages. GNP reports that during 1992, when it voluntarily implemented major components of the WUP, daily flows during the boating season exceeded 2,400 cfs during daylight hours 64 percent of the time (GNP, 1993b). According to the WUP, GNP also would be able to maintain weekly average flows at McKay station above 2,400 cfs from May 1 to September 12 during an average rainfall year.

Upper Gorge. DOC recommended studying the potential for whitewater boating in Upper Gorge below Ripogenus dam (letter from DOC to GNP, April 28, 1987). Upper Gorge has been rafted successfully at flows as low as 1,100 cfs (GNP, 1991b). Currently, no commercial rafting occurs through Upper Gorge, and whitewater boating is generally limited to private boaters during occasional spillage events.

GNP does not propose flows for recreational boating in Upper Gorge. Under the Applicant's Proposal, whitewater boating in Upper Gorge would occur during occasional spillage events, as it does now.

Millinocket Stream and Back Channel. CI and NPS requested studies of the potential for whitewater boating on the Back Channel and Millinocket Stream (letters from D. Sosland, CI, February 29, 1992; D. Haas, NPS, March 2, 1992). FERC decided not to evaluate the potential for whitewater boating in these two streams in this FEIS because the



Figure 4-6. Whitewater boating experience versus flow for the West Branch (Source: Staff)

4-48

significant flows that would be required to provide recreational boating opportunities would jeopardize other environmental enhancements (FERC, 1993c).

Multiday Canoe Trips. The West Branch of the Penobscot River offers one of only a few remaining opportunities in the eastern United States for multiday canoe trips in a wilderness setting. GNP does not propose any construction activities that would interfere with such trips or would require additional portage. Loss of wilderness character because of shoreline development, forestry practices, and water level fluctuations is the major potential adverse effect on multiday canoe trips in this area (see sections 4.9 and 4.10).

4.8.1.2 Water Level Fluctuations

Water levels in Ripogenus, North Twin, and Millinocket Lake impoundments fluctuate seasonally. GNP proposes to maintain relatively stable water levels in North Twin impoundment from May 1 through mid-August, unless it is unable to maintain the required 2000 cfs minimum flow below Millinocket. GNP proposes to pump water from Millinocket Lake to maintain stable water levels in North Twin; this, combined with increased releases to Millinocket Stream, would increase water level fluctuations in Millinocket Lake during the summer recreation season.

Although several agencies and organizations recommend limiting draw-down of the Ripogenus, North Twin, and Millinocket Lake impoundments for various environmental reasons, only Save Our Scenic Lakes Association (SOSLA) specifically cites recreation. SOSLA contends that rapid draw-downs during the recreation season (May 1 to September 1) make boating, swimming, and general recreation hazardous, stressful, and expensive. SOSLA's concern focuses on the North Twin impoundment and Millinocket Lake, and it requests that the start of draw-down on North Twin be delayed until at least August 22 (SOSLA, 1993). Other owners of waterfront property on North Twin testified during scoping meetings that fluctuating water levels are not a problem (FERC, 1993b).

Ripogenus Project. The proposed WUP would increase water level fluctuations in the Ripogenus impoundment during dry years to satisfy other downstream demands for water. During a dry year, draw-downs allowed under the plan would not exceed the historical maximum draw-down (approximately 27 feet since 1976), which typically occurred during late winter. The annual average lake elevation, however, would be approximately 1 foot lower, and the average elevation during the summer recreation period would be approximately 0.5 feet lower. During average rainfall years, neither annual nor summer draw-down would increase. In general, lake elevations are highest in early summer because the impoundment stores high spring flows.

GNP identified 43 seasonal camps along the Ripogenus shoreline, only 7 of which have boat docks (GNP, 1992d). All seven boat docks are either on floats or fastened to wheels so that they can be adjusted to changing water levels. The two commercial operations on the impoundment both have floating docks.

The staff concludes that the proposed average reduction of 1 foot in water levels would not significantly affect boat access or recreational use at the Ripogenus impoundment. North Twin. Although the WUP does not establish firm limits on draw-down during the summer, it would dampen water level fluctuations. From 1976 to 1990, draw-down between May 15 and August 15 averaged 2.2 feet. Under the WUP, draw-down during this same period would have averaged 1.3 feet. During a dry year, the North Twin impoundment elevation historically averaged approximately 483 feet. Under the WUP, GNP would maintain water levels at an elevation of approximately 488 feet.

GNP proposes to maintain a relatively stable water level only until August 15 and then to initiate a relatively rapid draw-down of up to 2.0 feet per week to reach a minimum pool elevation before lake trout spawning season begins (October 15). This rapid draw-down would begin during the height of the summer recreation season, including the Labor Day holiday weekend.

GNP found that 50 percent of seasonal camps along the North Twin impoundment do not have boat docks, 40 percent have floating docks, and 10 percent have fixed docks. Water depths at the fixed docks ranged from 4.4 to 7.0 feet at normal pool elevation. All three commercial operations on North Twin have floating docks. Maximum summer draw-down at North Twin since 1972 is 9.7 feet, which renders all the fixed docks unusable. The average summer draw-down since 1972, however, is only 1.5 feet, leaving fixed docks largely unaffected. Under the proposed WUP, maximum draw-down would be approximately 4.5 feet during a dry year and 1.5 feet during a year with normal rainfall. The staff concludes that the proposed WUP would improve recreational use at the North Twin impoundment by reducing water level fluctuations and maintaining higher average pool elevations during the summer. This would be particularly beneficial for the North Twin impoundment because it is the most heavily used recreation area in the project area. Rapid draw-down after August 15, however, would conflict with recreational use and could create some safety concerns. A study of navigation hazards would be appropriate to ensure recreational boating safety.

Millinocket Lake. GNP proposes to pump water from Millinocket Lake to maintain relatively stable water levels in the North Twin impoundment. Combined with increased minimum flow releases to Millinocket Stream, this would result in increased draw-down of Millinocket Lake during the summer. GNP did not model water surface elevations on Millinocket Lake; therefore, although the exact extent of potential draw-down is unknown, it would be greater than the historic average.

GNP sampled seasonal camps on Millinocket Lake and found that 33 percent do not have docks, 19 percent have floating or movable docks, and 48 percent have fixed docks. Water depths at the fixed docks ranged from 1.8 to 7.6 feet at normal pool elevation. The two commercial operations on Millinocket Lake have a total of eight floating docks and one fixed dock with a water depth of 9 feet at normal pool elevation. Maximum summer draw-down at Millinocket Lake since 1972 is 3.4 feet, which would render some of the fixed docks unusable. The average summer draw-down since 1972, however, has been only 0.6 feet, which would not affect fixed docks. Under the proposed WUP, maximum draw-down would increase. The staff concludes that the WUP would adversely affect recreation on Millinocket Lake, but that a study of potential navigation hazards may be appropriate.

4.8.1.3 Recreational Fisheries

Upper Gorge. GNP proposes to increase flows to 100 cfs from July 1 to September 30; flows would remain at leakage (12 cfs) during the rest of the year. This flow enhancement is designed to attract adult salmon into Upper Gorge during the summer fishing season to increase fishing opportunity within this reach. This flow enhancement would increase the amount of adult salmon habitat in the reach by 71 percent. DIFW and FWS accept this proposal as adequate (letter from J. Deason, Interior, May 24, 1993). The proposed seasonal flow enhancement would improve habitat for salmon in Upper Gorge, especially habitat for the adult salmon desired by anglers. Although this enhancement would increase the number of river miles of landlocked salmon habitat in the West Branch watershed by less than 1 percent, the new habitat would be located in an area of high use and would offer additional habitat for a world-renowned salmon fishery.

West Branch. GNP proposes to enhance flows for salmon spawning and incubation in the West Branch below McKay station and to develop a salmon nursery area near Holbrook Pool. The staff estimates the salmon nursery could add 245, 3-year-old fish per year to the population (see section 4.4). The staff concludes that GNP's proposal would improve the reproductive success of salmon and enhance the recreational fishery in this area.

Millinocket Stream. GNP proposes to increase minimum flows to 60 cfs from May 1 to October 15 and to maintain 20 cfs during the remainder of the year. This flow enhancement would increase smallmouth bass spawning habitat by 16 percent but would not significantly affect salmon stocks (see section 4.4). The staff concludes that GNP's proposal would improve the bass fisheries in Millinocket Stream.

GNP also proposes to establish a put-and-take trout stocking program with 500 trout per year at 3 trout per pound. Most out-of-town anglers are attracted to the natural, highquality salmon and trout fishing opportunities in the area. DIFW estimates that there are more than 3,000 miles of native brook trout streams just in the Ripogenus/Moosehead Lake area (Region E); therefore, we conclude that the stocking program would benefit primarily local residents and would attract few out-of-town anglers.

Back Channel. GNP proposes to provide only leakage (2 to 5 cfs) in the Back Channel. This flow would not support any sport fishery in this reach. The staff concludes that existing conditions in the Back Channel would not change.

4.8.1.4 Recreation Access and Facilities

GNP maintains that existing recreation access to the impoundments is adequate and does not propose any additional access points. GNP contends that the number of water access points, campsites, and other recreation facilities adequately meets current demand at the Ripogenus Project (letter from C.W. Ten Broeck, DOC, to J. Carson, GNP, November 5, 1990).

The North Twin impoundment and Millinocket Lake, which are the impoundments used most heavily for recreation, have five and four access points for boats, respectively. These access points are located conveniently along the major roads in the area. There is no evidence of overuse or capacity limitation, except for parking at several water access points. GNP's proposed recreational enhancements would address parking concerns and provide amenities for whitewater boaters, which would improve the overall recreational experience within the project area. We conclude, therefore, that the numbers and locations of existing recreation facilities at both Ripogenus and Penobscot Mills are adequate. GNP agreed to assess the adequacy of existing recreation facilities within the project area every 10 years in consultation with the Maine Bureau of Parks and Recreation.

GNP did not receive comments from the public or from agencies about access for people with disabilities at the Ripogenus or Penobscot Mills projects. GNP must, however, comply with the Americans with Disabilities Act of 1990 (ADA). GNP intends to meet applicable accessibility standards at its recreation sites to the maximum extent possible. GNP proposes to provide marked handicapped-accessible parking spaces with proper surface compaction for wheelchair access at the Caribou Lake, McKay station, Green Bridge, Dead Man's Curve, and Route 157 boat launch facilities. The proposed privies at McKay station and the changing facilities at Never's Corner would meet ADA guidelines. This access would provide recreational boating and fishing opportunities for wheelchair users.

4.8.1.5 Cumulative Impacts

This alternative would enhance whitewater boating, fishing, and other water-based recreational activities along project impoundments and the West Branch.

4.8.2 Alternative 1

4.8.2.1 Recreation Flows

Regarding the proposed flow regime at McKay station, this alternative and its associated impacts would be the same as the Applicant's Proposal.

4.8.2.2 Water Level Fluctuations

Under this alternative, draw-downs of Ripogenus impoundment and Millinocket Lake would increase slightly because water stored in these impoundments would help to provide the 350 cfs flow to the Back Channel and to maintain relatively stable water levels in the North Twin impoundment.

At Ripogenus, this alternative would increase average draw-down by about 2 feet during the summer of a dry year as compared with historical conditions; effects would be negligible during a year with normal rainfall. The staff concludes that this increase in drawdown would not significantly affect boat access or recreational use at the impoundment because nearly all docks can adjust to water level fluctuations, and most boat launches are usable even during periods of low water.

At the North Twin impoundment, this alternative would reduce average draw-down during dry and normal years by approximately 3 feet and 1 foot, respectively, thereby improving recreational access and use.

GNP would use water stored in Millinocket Lake to provide increased minimum flows in Millinocket Stream and help maintain stable water levels in the North Twin impoundment. This alternative, therefore, would result in increased draw-down in Millinocket Lake relative to historic conditions. The staff predicts that draw-down at Millinocket Lake under this alternative would be greater than under the Applicant's Proposal because the Back Channel flows would be increased, resulting in greater recreational effects.

4.8.2.3 Recreational Fisheries

Alternative 1 proposes additional fishery enhancements in Upper Gorge, Millinocket Stream, and the Back Channel.

We conclude that increasing minimum flows from 12 cfs to 50 cfs from October 1 to June 30 would not provide any lasting benefit to fishery resources in Upper Gorge and would not improve recreational fishing (see section 4.4). Providing minimum flows of 60 cfs in Millinocket Stream also would result in a small, if any, numerical increase in salmon stocks and would not provide any significant benefit to smallmouth bass. The proposed flow regime, therefore, would not significantly improve recreational fishing in Millinocket Stream.

This alternative would provide a minimum flow of 350 cfs in the Back Channel, producing approximately 245, 3-year-old fish per year over the 4.5-mile reach. The estimated 245 adult salmon in the Back Channel represent only 61 fish per mile per year, well below the state standard of 83 fish per mile per year required for moderate quality fisheries (DIFW, 1991). The staff predicts that providing a 350-cfs minimum flow would improve fishing in the Back Channel, but not enough to attract a significant number of anglers because better, more accessible fishing opportunities are available along the West Branch. As discussed in section 4.4, other factors limit the value of a fishery in the Back Channel.

4.8.2.4 Recreation Access and Facilities

Interior recommends that the applicant conduct recreation monitoring studies to assess the adequacy of existing recreational facilities within the project areas once every six years for the term of the new licenses in consultation with FWS, NPS, DIFW, DOC, and PIN. The monitoring studies would include at a minimum the collection of annual recreation use data and would begin within 6 years of receiving a license. GNP would submit to the Commission a report to include (1) annual recreation use figures; (2) a discussion of the adequacy of its recreation facilities at the project site to meet recreation demand; (3) a description of the methodology used to collect all study data; (4) if there is a need for additional facilities, a recreation plan to accommodate recreation needs in the project area; (5) documentation of agency/tribal consultation and agency /tribal comments on the report after it has been prepared and provided to the agencies and PIN; and (6) specific descriptions of how the agency/PIN comments are accommodated by the report. GNP would allow a minimum of 30 days for the agencies and the PIN to comment and to make recommendations before filing the report with the Commission.

The recreational facilities within the project area are varied and managed by numerous agencies. Assessing recreational use and demand and coordinating the overall management of recreational resources is crucial to ensure adequate public recreational resources over the

term of the license. The staff finds the specified details for recreational monitoring to be appropriate to ensure adequate monitoring of recreational use and resources within the project area.

The effects of this alternative would be similar to those of the Applicant's Proposal (see section 4.8.1.4). Recreational access would be more difficult during dry years because of jower lake water levels.

4.8.2.5 Cumulative Impacts

The effects of this alternative would be the same as those of the Applicant's Proposal (see section 4.8.1.6), with the additional benefits of the proposed recreation monitoring every 6 years to ensure facilities are meeting recreational demand.

4.8.3 Alternative 2

4.8.3.1 Recreation Flows

Regarding the proposed flow regime at McKay station, this alternative and its associated impacts would be the same as the Applicant's Proposal. This alternative also would provide scheduled whitewater flow releases in Upper Gorge for expert daylight kayaking on two weekends during May. GNP would notify AWA and local whitewater boating interests at least 7 days before scheduled releases.

No estimates of the number of paddlers in the region who are sufficiently skilled to paddle this reach are readily available, but the carrying capacity of Upper Gorge would probably determine the number of users. The staff estimates that the per-day whitewater boating capacity of Upper Gorge is approximately 54 paddlers. This assumes 3 boaters per group (per AWA safety code), 9 hours available for access each day, and 30 minutes between group departures to preserve a wilderness experience. Flow releases would be limited to May to avoid conflicting with fishery goals. Releases during June or October could attract salmon into Upper Gorge; however, these fish would be stranded when flows are reduced on weekdays. From July 1 to September 30, GNP proposes flows of 100 cfs for fishery enhancement. Higher flows for whitewater boating during this period could flush salmon out of the reach. In addition, GNP often spills water during May. Under this alternative, GNP would simply notify whitewater interests of planned spillage. The cost to GNP in terms of lost power would be negligible because these flows usually would be spillage.

The staff concludes that two weekend releases during May would not conflict with fishery goals; would have negligible cost to GNP; and would provide whitewater boating opportunities for more than 200 paddlers, which should be adequate to satisfy the demand of expert boaters to paddle this reach. The staff recommends that GNP consult with whitewater interest groups regarding appropriate flows for the weekend releases during May.

4.8.3.2 Water Level Fluctuations

Under this alternative, draw-downs at Ripogenus impoundment and Millinocket Lake would increase slightly because water stored in these impoundments would help maintain relatively stable water levels in the North Twin impoundment.

At Ripogenus, this alternative would increase average draw-down by 1.5 feet during the summer of a dry year as compared with historical conditions; effects would be negligible during a year with normal rainfall. The staff concludes that an increase in draw-down of 1.5 feet during a dry year would not significantly affect boat access or recreational use at the impoundment because nearly all docks can adjust to water level fluctuations, and most boat launches are usable even during periods of low water.

At the North Twin impoundment, this alternative would reduce average draw-down by approximately 4 feet during dry years and 1 foot during normal years. This alternative also would extend the period during which GNP maintains relatively stable water levels in the North Twin impoundment from August 15 to August 22. This additional week would enhance recreation and provide more predictable water levels for boaters during the height of the summer recreation period at the most actively used impoundment in the project area.

Extending the period of stable water levels would reduce the time available to draw down the reservoir to its lowest elevation (approximately elevation 479 feet) by October 15 (for lake trout spawning) from 8.5 to 7.5 weeks. Reducing this draw-down period would not jeopardize GNP's draw-down requirements or exceed GNP's maximum draw-down rate of 2.0 feet per week. Maximum draw-down currently occurs when the impoundment is at maximum pool elevation (491.9), and water levels would need to be drawn down 12.9 feet over 8.5 weeks, or 1.5 feet per week. Under the worst-case scenario, Alternative 1 would decrease the impoundment level by 12.9 feet over 7.5 weeks, or 1.7 feet per week; this rate is still below GNP's recommended maximum. A 1-week extension would increase outflow by less than 5 percent (from 3,900 cfs to 4,100 cfs) during a year with normal rainfall. As long as inflow to North Twin does not exceed approximately 2,800 cfs, GNP would not have to spill water to achieve this draw-down. These proposed actions would improve recreational access and use. Extending the period of stable water levels by 2 weeks until August 29 would result in an average draw-down of 2.0 feet per week and probably would exceed GNP's maximum draw-down ratio.

GNP would use water stored in Millinocket Lake to provide increased minimum flows in Millinocket Stream and help maintain stable water levels in North Twin; consequently, this alternative would result in increased draw-down relative to historic conditions. The staff estimates that draw-down under this alternative would be slightly greater than under the Applicant's Proposal because of increased the Back Channel flows and extended stable water levels in North Twin; however, draw-down under this alternative would be less than under Alternative 1 because of reduced the Back Channel flows.

4.8.3.3 Recreational Fisherles

Proposed flows and effects on recreational opportunities in the Upper Gorge and West Branch would be the same as for the Applicant's proposal (see section 4.8.1.3). This alternative proposes a 60-cfs minimum flow year-round in Millinocket Stream. The proposed flow regime would provide only a small, if any, numerical increase in adult salmon (see section 4.4). This alternative, therefore, would provide similar effects on recreational fishing opportunities as described in section 4.8.1.3.

4.8.3.4 Recreation Access and Facilities

Under this alternative, GNP would conduct recreation monitoring studies every 6 years for the term of the licenses in consultation with various resource agencies. The monitoring studies would include at a minimum the collection of annual recreation use data and would begin within 6 years of license. GNP would submit to the Commission a report to include (1) annual recreation use figures; (2) a discussion of the adequacy of its recreation facilities at the project site to meet recreation demand; (3) a description of the methodology used to collect all study data; (4) if there is a need for additional facilities, a recreation plan proposed to accommodate recreation needs in the project area; (5) if there is need for additional facilities, design of recreational facilities would conform to the national standards established by the Architectural and Transportation Barriers Compliance Board pursuant to the Americans with Disabilities Act of 1990; (6) documentation of agency consultation and agency comments on the report after it has been prepared and provided to the agencies; and (7) specific descriptions of how the agency comments are accommodated by the report. GNP would allow a minimum of 30 days for the agencies to comment and to make recommendations before filing the report with the Commission.

Staff believes that assessing recreational use and demand and coordinating the overall management of recreational resources are crucial to ensure adequate public recreational resources over the term of the license. The proposed monitoring of recreational use and resources within the project area would provide incremental assessment and allow for potential improvements of recreational resources as recreational demand warrants.

The effects of this alternative would be similar to those of the Applicant's Proposal (see section 4.8.1.4). Recreational access would be more difficult during dry years because of lower lake water levels.

4.8.3.5 Cumulative Impacts

The effects of this alternative would be the same as those of the Applicant's Proposal (see section 4.8.1.6), with the additional benefits of the proposed recreation monitoring every 6 years to ensure that facilities are meeting recreational demand.

4.8.4 No-action Alternative

Under the No-action Alternative, the projects would continue to operate under the terms and conditions of the existing licenses, with no change in existing environmental conditions.

4.9 LAND USE

During scoping, resource agencies and intervenors noted the need for a comprehensive watershed management plan. GNP proposes no such plan. A watershed

protection plan is beyond the scope of FERC's EIS process because changing land use patterns throughout the watershed outside project boundaries is not a project activity.

4.9.1 Applicant's Proposal

Land use issues include shoreline development, timber harvesting, and expansion of project boundaries. Primary concerns are the potential negative impacts of future development and the need to establish more protective land use regulations.

During the DEIS comment period, GNP proposed additional measures for managing the shoreline in the Ripogenus Project. GNP proposes to adopt a Memorandum of Understanding (MOU) with the state of Maine to convey a renewable conservation easement within the Ripogenus Project area including approximately 73 miles of shorefront land within 250 feet of the normal high water mark in certain areas of Chesuncook and Ripogenus lakes and Brandy and Black ponds for the term of any FERC license. GNP also proposes a perpetual conservation easement outside the project area on approximately 5 miles of shorefront land within 500 feet of the normal high water mark of portions of Lobster Lake and the West Branch of the Penobscot River (letter from Angus King, Jr., Governor of Maine, and Donald McNeil, GNP, February 16, 1995). GNP proposes no conservation easements for the Penobscot Mills Project area.

The proposed conservation easements would be consistent with and become an addendum to the 1981 Resource Protection Plan and Recreation Management Plan for the Penobscot Waterway (see section 3.11.2.1). Under the proposed MOU, the state would manage recreational activity within the easements, and no additional commercial or residential development would be allowed. GNP and the state reserve the right to withdraw from the obligations put forth in the MOU if FERC imposes conditions or requirements for shoreline protection within the Ripogenus or Penobscot Mills Project areas which differ materially from the GNP-proposed easements (letter from Angus King, Jr., Governor of Maine, and Donald McNeil, GNP, February 16, 1995).

4.9.1.1 Shoreline Development

Under the Applicant's Proposal, the proposed 250-foot easements would guide shoreline development in a significant portion of the GNP-owned property within the Ripogenus Project area. GNP proposes no easements for the Penobscot Mills Project area, where LURC's existing regulations would guide shoreline development. GNP contends that LURC's current land use regulations adequately control the kind and amount of development within the project boundaries and area (GNP, 1993b).

The CI are concerned about the amount of waterfront development allowed under LURC's regulations and note that LURC's standards are not necessarily permanent. They contend that LURC is an independent state commission that can alter its regulations, approve rezonings, or even be abolished on its own, or pursuant to an act of the legislature (letter from Daniel Sosland, CLF et al., April 30, 1992).

LURC's regulations establish frontage restrictions along project shorelines. In the Penobscot Mills Project area, roughly 72 percent of the impoundment shorelines are zoned as P-GP, which restricts residential development to one dwelling unit per 200 linear feet of shoreline. Within the Ripogenus Project area, roughly 60 percent of the impoundment shoreline is zoned as P-AL, which limits development to one dwelling unit per linear mile of shoreline. The conservation easement along the West Branch of the Penobscot River established in 1981 encompasses roughly 14 percent of the Ripogenus Project and restricts development within 500 feet of the river.

The proposed conservation easements along 73 shoreline miles around Chesuncook and Ripogenus lakes and Brandy and Black ponds would limit development of commercial and residential structures within 250 feet of the shoreline. Both existing and proposed easements would allow structures related to generation of hydroelectric power, timber harvesting, mineral extraction, and the development, in accordance with LURC requirements, of camps and campgrounds on existing lease lots within the easement areas (LURC, 1981).

GNP owns approximately 50 percent of the Ripogenus Project shoreline and 70 percent of the Penobscot Mills Project shoreline. GNP awarded approximately 900 leases to current and retired employees for use as private camps. GNP terminated leasing during the early 1970's; however, GNP has not guaranteed that the leasing moratorium would remain in effect for the new license period. GNP could resume granting leases, and related development could occur within the project area. Presently, there are approximately 430 dwelling units along the Penobscot Mills Project shoreline and 70 units along the Ripogenus Project shoreline. Over the term of the license, the staff estimates residential development could increase by approximately 40 percent in the Penobscot Mills Project area, and by approximately 70 percent in the Ripogenus Project area. These estimates are based on LURC's current regulations, including subdivision regulations, and do not account for development limitations such as steep slopes, poor soils, wetlands, or access.

According to the Forest Service, forested lands with high recreation value, particularly on lakeshore and riverfront lands, are most vulnerable to changes in land use and potential development (FS, 1992). A LURC study of building permits granted between 1971 and 1991 showed that 53 percent of all residential development occurred along shorelines of lakes with high scenic value (lakes of statewide significance; see section 3.11.1). Approximately 70 percent of the lakes within the project areas are classified as having statewide significance and are particularly subject to the effects of future development.

For the Ripogenus Project, the proposed easements would not be incorporated into the project boundaries but would be donated to the state of Maine and maintained for the duration of the license period as put forth in the terms of the MOU. The staff concludes that the proposed easements for the Ripogenus Project area and the shoreline protection afforded by these easements would be adequate for the term of the license.

For the Penobscot Mills project, GNP proposes no easements and LURC's current regulations would apply. LURC's regulations are subject to change independently of project operations and license conditions. For example, between 1985 and 1992, LURC granted 17 of 23 petitions to rezone lands within P-GP districts (letter from Daniel Sosland, CI, September 3, 1993). The staff finds that although LURC's current regulations adequately manage shoreline development, these regulations are not directly tied to the license and long term

protection of the aesthetic and natural recreational experience would not be guaranteed for the term of the license.

4.9.1.2 Timber Harvesting Practices

The CI contend that GNP's forestry practices affect water quality, water quantity, recreation, fisheries, and wildlife habitat in the watershed (letter from Daniel Sosland, CI, September 3, 1993). GNP contends that current regulations provide adequate measures to ensure proper management of forest resources (GNP, 1993c). Timber harvesting practices within the project region are regulated by the MFPA and LURC regulations. The MFPA regulations include performance standards for clear-cutting, regeneration standards, notification before harvest, annual reports, and regularly updated forest management and harvest plans prepared by a licensed forester (State of Maine, 1989). Within the Ripogenus and Penobscot Mills Project areas, LURC timber harvesting regulations control the extent of vegetative clearing and provide vegetative filter strips related to the slope of the land (see section 3.11.1.1).

The most commonly recommended minimum buffer to protect water quality is 50 feet (Nieswand et al., 1990). Several other factors, most notably slope, also affect the necessary width of buffer strips (Nieswand et al., 1990; Budd et al., 1987; Joubert, 1985). The FS recommends a 75-foot buffer (15-foot undisturbed forest and 60-foot managed forest) for protection of water quality (FS, 1992). LURC's and MFPA's buffer requirements are generally consistent with recommended guidelines for protecting water resources. The staff concludes, therefore, that the existing timber harvesting practices and regulations adequately protect water quality.

4.9.1.3 Expansion of Project Boundaries

The CI and Interior recommend that GNP establish protection zones and shoreline conservation easements for all lands surrounding the project impoundments to provide long-term protection and enhancement of water quality, aesthetics, riparian wildlife habitat, and backcountry recreation opportunities (letter from Daniel Sosland, CLF et al., November 2, 1993; and letter from J. Deason, Interior, May 24, 1993).

During the DEIS comment period, GNP proposed conservation easements along approximately 73 shoreline miles for the Ripogenus Project area (see section 4.9.1). These easements would not be included within the project boundary but would be conveyed to the state of Maine for the term of the license. We conclude that the proposed easements would provide adequate protection of the natural and aesthetic qualities of the shoreland areas within the Ripogenus Project area for the term of the license.

GNP proposes no project boundary expansion or protection zones within the Penobscot Mills Project area, relying on LURC's regulations to control land use throughout the project area. The existing project boundaries generally extend to the high water marks of the project impoundments and streams; therefore, shorelands adjacent to project waters and affected by potential development lie outside the designated project boundaries. LURC's current regulations provide means to control the amount and location of shoreline development; however, for the Penobscot Mills Project area, these provisions are not directly tied to the project license and could change over the term of the license. Potential shoreland development could adversely affect recreational and aesthetic resources within the project area. We conclude that the proposed boundaries for the Penobscot Mills Project area are inadequate.

4.9.1.4 Cumulative Impacts

Potential shoreline development within the Penobscot Mills Project area would contribute to the loss of the natural character and could adversely affect recreational and aesthetic resources.

4.9.2 Alternative 1

This alternative proposes expanding project boundaries to allow for building setbacks and vegetative buffers, which would protect aesthetics, water quality, riparian wildlife habitat, and related backcountry recreational opportunities. Proposed setback areas would extend 500 feet from the highwater marks of the impoundments and would prohibit residential or commercial development. The recommended width is consistent with the guidelines established for river and lake protection on the adjacent Allagash River wilderness waterway and with the existing GNP conservation easement along the West Branch of the Penobscot River. The designated areas for vegetative buffers would extend 250 feet from the high water marks of the impoundments and permit no clear-cutting or other removal of vegetation. Other regulations would include standards established in the Resource Protection Plan for the Penobscot Waterway (LURC, 1981).

4.9.2.1 Shoreline Development

The proposed building setback would restrict future development within 500 feet along the impoundments' shorelines and prohibit development of piers, boat docks, and ramps. The setbacks would not prohibit the creation of waterfront lots or significantly reduce the amount of development, but they would require all buildings to be set back 500 feet from the shoreline. A 500-foot building setback exceeds nearly all recommended minimum setbacks for protecting water quality. We conclude that a 500-foot building setback would protect the natural and aesthetic resources within the project areas.

4.9.2.2 Timber Harvesting Practices

The proposed vegetative buffer would prohibit clear-cutting within 250 feet of the impoundment and stream shorelines, which is an additional 150 to 200 feet of buffer beyond LURC's current requirements (see section 3.11.1.1). The staff concludes that the additional buffer requirement would provide increased protection against nutrients, sediment, animal-derived organic matter, and some pesticides reaching water bodies, thereby reducing pollution and providing further protection of surface water quality.

4.9.2.3 Expansion of Project Boundary

This alternative proposes a 500-foot horizontal expansion of the project boundary from all impoundment shorelines. GNP would need to acquire fee simple interest in this land or

secure conservation easements. We estimate that GNP would need to secure easements on approximately 3,500 acres within the Ripogenus Project and approximately 3,000 acres within the Penobscot Mills Project. The proposed expansion would increase setbacks along the shoreline areas up to 400 feet, limiting potential negative impacts on the natural environment and preserving the aesthetic resources of the project area.

4.9.2.4 Cumulative Impacts

The CI's proposed conservation easements would provide cumulative environmental benefits through further regulation and mitigation of development and timber harvesting in critical areas, such as wetlands and shorelines, within the project areas.

4.9.3 Alternative 2

In the DEIS this alternative recommended a 200-foot boundary expansion for the Ripogenus Project and a 200-foot boundary expansion on GNP-owned lands for the Penobscot Mills Project. We revised this recommendation based on comments received during the DEIS comment period, including GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information (see section 4.12.2.2), and revised estimates of development potential (see section 4.9.1.1).

As a result of this assessment, we revised this alternative and propose two options for protecting the Ripogenus Project shoreline area. Option One would be adoption of the proposed 250-foot conservation easements as defined by the MOU (see section 4.9.1). Option two would incorporate a 200-foot boundary expansion on GNP-owned lands for the Ripogenus Project. Under both options, this alternative proposes expanding project boundaries for the Penobscot Mills Project to generally extend 200 feet from the high water mark of the impoundments within the project area, only on land currently owned by GNP. For both projects, the proposed boundary expansion would incorporate a 200-foot building setback and a 100-foot vegetative buffer.

4.9.3.1 Shoreline Development

The proposed easements or boundary expansion along approximately 73 shoreline miles of the Ripogenus Project area would limit development of commercial and residential structures within 200 feet (boundary expansion) to 250 feet (MOU) of the shoreline. The staff concludes that the proposed easements or boundary expansion would provide adequate shoreline protection for the Ripogenus Project area (See section 4.9.1.1).

For the Penobscot Mills Project area, no development would be allowed within the proposed 200-foot boundary expansion. This proposed boundary expansion would limit potential negative effects of development on the aesthetics and recreational resources of the shoreline areas. See section 4.9.3.3 for further discussion of the proposed boundary expansion.

4.9.3.2 Timber Harvesting Practices

This alternative proposes that timber harvesting practices within the proposed 250-foot easement (MOU), or 200-foot boundary expansion for the Ripogenus Project area and the
200-foot boundary expansion for the Penobscot Mills Project area would continue to follow existing state of Maine regulations (MOU) or incorporate a 100-foot vegetative buffer and those stipulations as established under the Shoreline Management Plan (boundary expansion). The effects of this alternative would be similar to Alternative 1 (see section 4.9.2.2).

4.9.3.3 Expansion of Project Boundary

Option One

Under this option the proposed easements for the Ripogenus Project area would not be included within the project boundaries. The proposed 250-foot wide easements for 73 shoreline miles of Chesuncook and Ripogenus Lakes and Brandy and Black Ponds would be conveyed to the State of Maine for the duration of the license under the terms of the MOU (letter from Angus King, Jr., Governor of Maine, and Donald McNeil, GNP, February 16, 1995). The existing project boundaries would not be altered except in the event that the recreation monitoring plan would warrant additional recreational facilities within the project area. In this event, the recreational enhancements, (i.e., a boat launch area) would be incorporated into the project boundaries. Under this proposed alternative, GNP would remain responsible for recreational enhancements or mitigation associated with project operations or license conditions as required by the Commission for the term of the license.

For the Penobscot Mills Project area, this option proposes expansion of the project boundary generally 200 feet from the high water mark of project impoundments to include a 200-foot building setback and a 100-foot vegetative buffer. The Commission can require an applicant to acquire and include within the project boundary enough land for project purposes, such as public recreation, shoreline control, or protection of environmental resources [18 CFR § 4.51 (8) (2) (B)]. The proposed 200-foot boundary expansion would establish buffer zones for such purposes along the GNP-owned portions of the project impoundments and waterways.

The Commission can require the applicant to include an analysis of costs and constraints in developing a protection zone around the impoundments.^{11,12} We recognize that the most practical and cost-effective way to establish a protection zone is to develop criteria for selecting shoreline widths according to land use and resources and not by indiscriminately selecting a baseline width (FERC, 1993a). Under this option, GNP would

^{11 18} CFR § 4.51 (f) (6) (iv) - which requires the applicant to provide a statement "including an analysis of costs and other constraints, of the applicant's ability to provide a buffer zone around all or any part of the impoundment, for the purpose of ensuring public access to project lands and waters and protecting the recreational and aesthetic values of the impoundment and its shoreline."

^{12 18} CFR § 2.7 (a) - the Commission expects the licensee: "to acquire in fee and include within the project boundary enough land to assure optimum development of the recreational resources afforded by the project. To the extent consistent with the other objectives of the license, such lands to be acquired in fee for recreational purposes shall include the lands adjacent to the exterior margin of any project reservoir plus all other project lands specified in any approved recreational use plan for the project."

establish a Shoreline Management Plan (SMP) in consultation with LURC, CI, Interior, the town of Millinocket, and the town of East Millinocket for the proposed 200-foot boundary expansion on GNP-owned lands at the Penobscot Mills Project area that would establish guidelines and define policies and regulations concerning the management of shoreline lands. The goals and objective of the SMP would focus on protecting and maintaining aesthetic, natural, and cultural resources of the impoundment area and balancing the protection of these resources with permitted private and public use of the area.

The SMP would include maps delineating the shoreline protection zone on GNP, criteria used for selecting widths, and incorporate the following provisions to:

- develop a management plan for the shoreline protection zone areas under LURC guidelines for Resource Protection Plan;
- maintain the prescribed minimum width protection zones around the project impoundments and waterways, public roads, and private property;
- identify and substantiate determinations for designating widths less than the prescribed recommended protection zone width;
- plan timber harvesting activities within and adjacent to the protection zones to follow guidelines established by LURC before timber harvesting;
- minimize openings in shoreline vegetation and provide additional vegetative screening as needed where future recreational facilities and project operation development requires construction closer to the shoreline than the prescribed minimum-width protection zone;
- maintain project transmission line rights-of-way in a manner that minimizes adverse aesthetic effects caused by clearing vegetation;
- screen adverse visual features (i.e., storage buildings, parking areas), particularly
 prominent features visible from the shoreline impoundment, or other adjacent critical
 viewpoints;
- conduct periodic inspections of project lands to identify and mitigate changes in land use that adversely affect the provisions stated in this management plan; and
- review and update the plan every 10 years to meet aforementioned provisions.

Option Two

For the Ripogenus Project area, this option proposes a 200-foot boundary expansion on all GNP-owned lands, and would also propose the same 200-foot boundary expansion for the Penobscot Mills Project area as Option One. The proposed boundary expansion would include a 200-ft building setback and a 100-foot vegetative buffer. Under Option Two, GNP would establish a Shoreline Management Plan (SMP) in consultation with LURC, and Interior for the proposed 200-foot boundary expansion on GNP-owned lands at the Ripogenus Project area and in consultation with LURC, Interior, the town of Millinocket, and the town of East Millinocket for the 200-foot boundary expansion on GNP-owned lands at the Penobscot Mills Project area. The SMP would establish guidelines and define policies and regulations concerning the management of shoreline lands, as described under Option One.

Within the proposed boundary expansion areas, existing structures would be grandfathered, and the Commission would authorize GNP to review and approve proposed actions as established under the Standard Land Use Article and in the SMP. Such authority would be for proposed actions consistent with LURC's regulations and would incorporate LURC's review and approval. Any proposed variances to LURC's regulations would warrant FERC's review and approval. The project boundary expansion would enhance aesthetic and recreational resources and mitigate potential negative effects of development along shoreline areas.

4.9.4 No Action

The No-action Alternative would maintain current project operations and land use regulations. No project-specific protection zones would be established to mitigate potential negative effects of future development within the project region. These potential negative impacts would occur primarily along the projects' shoreline areas and would affect the water quality, aesthetic, and recreation resources in the respective project region.

4.10 AESTHETIC RESOURCES

Potential aesthetic impacts of the proposed projects could be caused by fluctuations in impoundment elevations, shoreline development, and forestry practices. These potential impacts stem from project operations, not from new project construction.

4.10.1 Applicant's Proposal

4.10.1.1 Water Level Fluctuations

GNP proposes measures to reduce impoundment draw-downs in North Twin Lake to enhance recreational use. Reduced draw-downs also would enhance aesthetics, although GNP proposes no specific measures to enhance aesthetics. GNP contends that additional flows for aesthetic purposes would be incompatible with its WUP and would jeopardize flows required at other locations throughout the project areas (GNP, 1993b, 1993c).

FWS, SOSLA, CI, and others are concerned about impacts of impoundment fluctuations and draw-downs on aesthetics at Ripogenus, Millinocket, and North Twin impoundments (letter from D. Sosland, CI, September 3, 1993).

Under the Applicant's Proposal, the amount of exposed shoreline at the Ripogenus impoundment would remain an average of 1,400 acres based on average pool elevation. An additional 300 acres would be exposed during dry years. The areas of exposed shoreline at the North Twin impoundment would decrease by 300 acres from the 750 acres currently exposed during average years, and by 700 acres from the 2,050 acres currently exposed during dry years. Millinocket Lake fluctuations would increase slightly from existing conditions, but the increase cannot be quantified based on available information.

Visual elements of the project area contribute to the quality of the recreational experience. Exposed rocky shorelines and streambeds can detract from the overall visual experience, particularly when associated with recreational use of the area. We find that the proposed draw-downs would have minimal effects on the aesthetic environment during average years, resulting from minor changes in the amount of exposed shoreline. Significantly reducing the amount of exposed shoreline during dry years would positively affect the North Twin impoundment. The additional 300 acres of exposed shoreline at the Ripogenus impoundment would have minimal negative effect on visual elements of the environment.

4.10.1.2 Development Along Shorelines

GNP contends that current LURC regulations provide adequate means to control the kinds and amount of development along the impoundment shorelines within the project boundaries (GNP, 1993b). LURC regulations restrict shoreline development to one structure per 100 feet to 1 mile, setbacks range from 75 feet to 150 feet, and vegetative buffers from 75 to 100 feet. Permanent piers require a review process and permit by LURC based on the applicable zoning district. Temporary piers and docks require no review or approval by LURC (1991).

NPS and CI are concerned about the aesthetic impacts of development along impoundment shorelines (letter from D. Sosland, CI, February 29, 1992). Development along the impoundment shorelines can significantly affect the aesthetic environment. A recent study on northern forestlands conducted by the Forest Service stated: "The existing qualities of the northern forest will not persist if the forces driving change are allowed to run their course ... Unplanned growth, commercial sprawl and development incompatible with the surroundings will erode the visual quality of the community" (FS, 1990).

For the Ripogenus Project area, the proposed 250-foot easement would protect the aesthetic resources of the shoreline areas for the term of the license. For the Penobscot Mills Project area, GNP proposes no easements, and LURC's existing regulations would apply. LURC's regulations, however, are not directly tied to the project license and are subject to change separately from license requirements and project operations. For the Penobscot Mills Project, over the long term, future development could affect the aesthetics of the project shoreline areas.

4.10.1.3 Influence of Forestry Practices

GNP proposes no additional restrictive or regulatory measures to control existing forestry practices beyond the control exercised under LURC. During scoping meetings, SOSLA indicated that any clear-cutting visible from the lakes would affect the aesthetic environment. LURC land use regulations restrict the size and location of clear-cutting operations along lake and stream shorelines (see section 3.11.1.1). Clear-cut harvesting is prohibited in vegetative buffers ranging from 50 to 100 feet from the high-water mark. Clear-cut openings beyond these vegetative buffers also are regulated to minimize negative effects on water quality and forest resources.

LURC regulations provide measures to limit the extent of clear-cut openings along lake and stream shorelines, which would negatively affect the aesthetic environment. These regulations are subject to change separately from license requirements and project operations. The staff maintains that forestry practices have been a part of the landscape in northern Maine for more than 100 years and, assuming that proper buffers and regeneration are provided, do not represent a significant aesthetic impact.

4.10.1.4 Cumulative Impacts

The proposed relative stabilization of impoundment shoreline elevations would have minor, short-term cumulative visual benefits. Potential development along the shoreline areas, however, would have negative cumulative impacts on aesthetic resources in the project areas.

4.10.2 Alternative 1

Modifications proposed in Alternative 1 that would influence aesthetic qualities of the project area include decreased shoreline fluctuations and the creation of conservation easements to restrict future development.

4.10.2.1 Shoreline Fluctuations

Under this alternative, the amount of exposed shoreline at the Ripogenus impoundment would not change during an average year but would increase by 1,600 acres during a dry year compared with existing conditions. The amount of exposed shoreline at North Twin impoundment would be reduced by 300 acres from the 2,500 acres currently exposed during both dry and normal years. Fluctuations of Millinocket Lake would increase more under Alternative 1 than under other alternatives, but cannot be quantified.

The proposed stabilization of shoreline fluctuations would result in minimal reduction of exposed shoreline acres during average years, and the effects of the exposed shorelines would remain virtually unchanged. During dry years, the proposed stabilization measures would cause significant negative aesthetic impacts at the Ripogenus impoundment because the amount of exposed shoreline would be substantially increased.

4.10.2.2 Development Along Shorelines

This alternative would include establishing conservation easements 500 feet from the high water mark to prevent development along the shorelines. This would have beneficial effects because it would limit the kinds and extent of development along the shorelines and promote and protect the visual quality of the natural environment.

4.10.2.3 Influence of Forestry Practices

Alternative 1 proposes a 250-foot vegetative buffer, which would prohibit timber harvesting within this buffer area. This alternative would have beneficial aesthetic effects because of larger vegetative barriers between the viewer and forestry practices and protect the visual quality of the natural environment.

4.10.2.4 Cumulative Impacts

This alternative would have minor, short-term visual effects and relatively few long-term visual effects.

4.10.3 Alternative 2

Alternative 2 would delay and modify fall draw-down at storage impoundments and restrict excessive shoreline development.

4.10.3.1 Shoreline Fluctuations

Draw-downs proposed in Alternative 2 would maintain the existing amount of exposed shoreline in the Ripogenus impoundment during average years (1,400 acres) and would expose an additional 800 acres during dry years. The number of exposed acres at the North Twin impoundment would decrease by 300 acres during average years and by 400 acres during dry years. Millinocket Lake fluctuations would increase slightly.

We find that the proposed actions would have minor effects on the visual environment during average years. During dry years, the increased amount of exposed shoreline at the Ripogenus impoundment would negatively affect the aesthetic qualities of the natural landscape.

4.10.3.2 Development Along Shorelines

The proposed 250-foot easement or 200-foot boundary expansion in GNP-owned lands for the Ripogenus Project area would provide protection of aesthetic resources from potential negative effects of shoreline development within the project area. For the Penobscot Mills Project area, the proposed 200-foot boundary expansion would provide additional protection of the aesthetic resources for the term of the license along the shoreline areas of GNP-owned property.

4.10.3.3 Influence of Forestry Practices

The effects of this alternative would be the same as those of the Applicant's Proposal (see section 4.10.1.3).

4.10.3.4 Cumulative Impacts

Regarding potential for cumulative impacts from proposed actions, this alternative would have the same impacts as Alternative 1 (see section 4.10.2.4).

4.10.4 No-action Alternative

The No-action Alternative would not enhance the existing project areas or protect against potential future negative impacts of shoreline fluctuations and development along the shorelines.

4.11 CULTURAL RESOURCES

The primary cultural resource issues within the Ripogenus Project and Penobscot Mills Project areas are potential effects on archeological and historical sites caused by erosion of shorelines; potential cultural effects on the PIN, including claims to the West Branch islands; and potential cumulative effects on PIN tribal, riverine traditions.

4.11.1 Applicant's Proposal

4.11.1.1 Archeological and Historical Sites

Archeological surveys and consultation with the SHPO identified 7 archeological sites within the Penobscot Mills Project and 15 sites within the Ripogenus Project potentially eligible for inclusion in the National Register of Historic Places (NRHP). GNP developed Cultural Resource Management Plans (CRMPs) to mitigate potential effects on the eligible archeological and historical sites within the Ripogenus and Penobscot Mills projects. The CRMPs include descriptions of significant sites, alternatives for avoiding effects, data recovery plans for designated sites, preliminary recommendations for site access and conservation easements, and provisional data recovery schedules (GNP, 1992a, 1992b).

The SHPO found that the proposed project construction and operations would not adversely affect the archeological and historic resources in the area (letters from E. Shettleworth, Jr., SHPO, to J. Carson, GNP, November 2 and November 3, 1992). For the Penobscot Mills Project, these findings assumed full implementation of the CRMP, proper maintenance of the Ambajejus Boom House, and no alteration of the present level of the impoundments. The SHPO found that relicensing the Ripogenus Project would not adversely affect archeological site significance, Chesuncook Village, or Ripogenus dam if the CRMP was fully implemented (letter from E. Shettleworth, Jr., SHPO, to J. Carson, November 3, 1993).

The staff recommends that GNP revise the CRMPs to include measures to address the protection of previously unidentified archeological properties within the project area, and updated schedules and costs for implementation of the CRMPs. Previously undiscovered properties within the project areas could be adversely affected by project construction or operation. The existing CRMPs do not provide sufficient measures to protect such properties. The revised CRMPs would be developed in consultation with the SHPO and PIN (see section 4.11.1.2), and submitted to the Commission and the Advisory Council on Historic Preservation (ACHP) as outlined in the Programmatic Agreement (PA).

The staff prepared a PA that provides stipulations for necessary elements of the CRMPs, measures for implementing the CRMPs; measures for interim treatment of historic properties; and measures for resolving disputes and amending and terminating the PA. The PA was executed on July 1, 1996 with FERC, the ACHP, and SHPO as signatory parties and GNP as a concurring party.

4.11.1.2 PIN Claims to Islands in the West Branch

PIN claims islands and traditional fishing rights in the West Branch of the Penobscot pursuant to terms of the 1980 Maine Indian Claims Settlement Act (letter from B. Ott, Director,

Eastern Area Office, Bureau of Indian Affairs, September 8, 1993). GNP refutes PIN's claim to the islands within the Penobscot tributaries and proposes no actions related to this claim (GNP, 1993c, 1993d).

The Maine Indian Claims Settlement Act of 1980 attempted to resolve outstanding claims to land in the Maine, including claims of the PIN (letter from C. Wilson, Associate Solicitor, Bureau of Indian Affairs, to B. Ott, Eastern Area Director, Bureau of Indian Affairs, June 5, 1992). PIN bases its claim on the Act to Implement the Maine Indian Claims Settlement, 30 Maine Revised Statutes Annotated, 6201 et seq. (25 U.S.C. § 1722 (i), citing 30 M.R.S.A. § 6203.8), which describes the Penobscot Reservation as:

"the islands within the Penobscot River reserved to the Penobscot Nation by agreement with the States of Massachusetts and Maine consisting solely of Indian (or Old Town) Island and all islands in said river northward thereof that existed on June 29, 1818, excepting any island transferred to a person or entity other than a member of the Penobscot Nation subsequent to June 29, 1818."

Upon review of the PIN's claim to islands in the West Branch, the Maine State Department of the General Attorney found: "The 1818 Treaty reserved to the Penobscots only the islands in the Penobscot River. Nothing in the Treaty even remotely suggests that any land or islands in any branches or tributaries of the Penobscot River were being reserved. Indeed, exactly the opposite is true, since, the Tribe, by the Treaty, was unequivocally ceding any claim to land in the branches of the Penobscot River" (letter from Michael Carpenter, Attorney General, State of Maine, December 16, 1993). Furthermore, the Maine State Department of the Attorney General concluded, " It is the opinion of this Office that the definition of 'Penobscot Indian Reservation' in 30 M.R.S.A. § 6203(8) does not include islands in the Branches of the Penobscot River and the Implementing and Settlement Acts did not expand the Penobscot Indian Reservation from what is was in 1980 " (letter from Michael Carpenter, Attorney General, State of Maine, December 16, 1993).

The Department of Interior has examined the 1980 Maine Indian Claims Settlement Act (25 U.S.C. 1721, et. seq.) and the Maine Implementing Act (30 M. R. S. 6201, et seq.) to determine if islands in the West Branch are included within the Penobscot Indian Reservation. Interior determined the conditioning authority of Section 4(e) would not be pursued for the Ripogenus or Penobscot Mills projects (letter from Kerry O'Hara, Department of Interior, Office of the Solicitor, March 3, 1994). Interior states that the Commission has neglected to consider tribal interests in issues such as water quality, toxics, stream flows, fishery resources, and cultural significance of the project area (letter from Andrew Raddant, Interior, to Lois Cashell, FERC, February 21, 1995).

The staff reviewed available information regarding the PIN's claims to lands and rights within the branches of the Penobscot River and concurs with the decision of the Maine State Department of the Attorney General (see section 4.11.1.2). The staff, therefore, finds that the consideration of the Penobscot Indian Nation's traditional practices within the project area are outside the scope of Section 106 review.

Based on these findings, the staff does not recommend including PIN as a concurring party to the PA associated with the Ripogenus and Penobscot Mills projects. The staff

acknowledges, however, the interest of PIN in the management of historic properties potentially eligible for listing in the National Register of Historic Places within the project areas. Accordingly, the staff recommends that GNP consult PIN during the development of the revised Cultural Resource Management Plans for the Penobscot Mills and Ripogenus Projects.

4.11.1.3 Cumulative Impacts

Potential cumulative impacts involve effects on religious and social aspects of the PIN and cumulative impacts of cultural resource mitigation efforts. The potential impact of project operations on the religious and social beliefs and values of PIN are best described by a statement issued by the PIN Governor (Sappier, 1990):

"It is our belief that the Penobscot, is a living, breathing being, yielding and sustaining life in all forms. In our thousands of years of existence, we have lived in harmony with the earth, the air, and the waters. The river is the life-blood of our tribal existence. All of our families use the river and islands for gathering food, hunting, fishing, trapping, collecting medicines, in fasting, prayer lodges, vision quests, and to meditate alone or with others. Any project that disrupts the spiritual and natural balance of the river and the people of the Penobscot Nation is considered an attempt to harm the people of Penobscot Nation."

Past studies have focused on coastal Maine locations, resulting in limited information on aboriginal riverine and lacustrine zones and populations. Data recovery proposed in the CRMPs can provide information to formulate models of interior aboriginal adaptations over time, making the mitigation measures significant beyond local applications and contributing to regional studies throughout Maine (GNP, 1991b). We find that the proposed cultural resource mitigation measures would provide data recovery and potentially important information about the aboriginal riverine tribes of central Maine.

4.11.2 Alternative 1

The effects of this alternative would be the same as those of the Applicant's Proposal (see section 4.11.1).

4.11.3 Alternative 2

The effects of this alternative would be the same as those of the Applicant's Proposal (see section 4.11.1).

4.11.4 No-action Alternative

The No-action Alternative would require no protection of cultural sites within the Ripogenus and Penobscot Mills project areas. Impoundment fluctuations would continue to erode potentially significant cultural resource sites.

4.12 SOCIOECONOMIC RESOURCES

Potential effects of GNP's proposed actions include changes in employment, fiscal conditions, and GNP's manufacturing and maintenance operations. Each of these impacts and potential cumulative impacts are discussed below.

4.12.1 Applicant's Proposal

4.12.1.1 Employment and Fiscal Impact

The proposed projects would not affect population or economic conditions in the project areas, nor would they affect the demand for local government services or government revenues.

The proposed minimum flows through Upper Gorge and McKay station would increase recreational use of the West Branch of the Penobscot River between Ripogenus dam and Ambajejus Lake. Commercial whitewater boating is the principal recreational use of this river reach. Rafting participation has declined in recent years from a peak of nearly 19,000 customers per season (1985) to 16,441 commercial rafters in 1990 (see section 3.10.4.1). According to a survey by GNP, the average rafter spends \$400 (1,990 dollars) per trip in Maine (GNP, 1991a). Assuming that increased minimum flows in the West Branch would attract repeat rafting customers sufficient to maintain the average 18,600 yearly participants, this would contribute approximately \$863,000 annually to the local economy in terms of purchase of goods and services.

GNP also proposes to construct facilities to support recreation (privies, changing rooms, etc.). GNP did not quantify the construction costs and labor requirements for these enhancements, but they are minor in scale and would contribute little to the local economy in terms of construction expenditures.

Fishing in the West Branch is extremely popular, averaging about 27 percent of visitordays within the West Branch district (see section 3.10.4.1). Most of the anglers surveyed traveled 100 to 400 miles primarily from locations in Massachusetts and Maine. Average spending by anglers in Maine was \$421 (1990 dollars) per trip. At current participation levels, approximately 30,000 users per season would contribute approximately \$1.26 million annually to the local economy. We conclude that the proposed fishing enhancements would not attract significantly more anglers to the area, which is already widely recognized as a world-class salmon fishery (see section 4.8).

4.12.1.2 Effect on GNP Operations

GNP contends that the flows proposed in its WUP would sustain current operations and would not cause further layoffs; however, the company could not rule out additional layoffs from other causes. The annual cost to GNP of this alternative would be \$877,600; the company's operations would not be adversely affected by this minimal increase in the cost of doing business.

4.12.1.3 Cumulative Impacts

The proposed actions would have no significant cumulative impacts. The project area would maintain current levels of tourism and employment.

4.12.2 Alternative 1

4.12.2.1 Employment and Fiscal Impact

Required flows to the Back Channel could affect GNP operations and indirectly jeopardize population and employment within the project area. Approximately one-third of workers in the project area are associated with mill operations; therefore, any decrease in mill productivity or employment would significantly affect workers and residents. Between 1985 and 1992, GNP reduced its paper and allied products work force by approximately 980 jobs in the Millinocket/East Millinocket labor market area (letter from J. Haskell, Town Planner, town of Millinocket, to J. Kotredes, Town Manager, town of Millinocket, August 28, 1993). Applying the 1.61 employment multiplier used by the Maine Department of Labor, this resulted in a total loss of 1,578 jobs in the area. The annual payroll decline due to those lost jobs could total \$65.7 million (average wage \$800.19 per week). The town of Millinocket concludes that "any additional losses as a result of the FERC relicensing process clearly would generate catastrophic results in the local economy given the downsizing of GNP that we have already experienced" (letter from J. Haskell, Town Planner, town of Millinocket, to J. Kotredes, Town Manager, town of Millinocket, to J. Kotredes, Town Manager, town of Millinocket, to J. Kotredes, Town

GNP contends that flows in excess of 50 cfs in the Back Channel would result in the loss of approximately 238 mill jobs (GNP, 1991b). We have no way to independently evaluate this estimate, but we assume that it is a high-end estimate. Using the Maine Department of Labor employment multiplier, the indirect effect of these lost jobs is 383 jobs, representing a lost payroll of more than \$15 million.

Any decline in productivity or employment at these sites because of increased minimum flows in the Back Channel would significantly adversely affect local fiscal revenues and service levels. Also, during dry and worst-case years, reduced flows and reservoir elevations could adversely affect recreation and fisheries resources. Both the towns of Millinocket and East Millinocket rely heavily on GNP for operating revenues. Millinocket receives 72 percent of its property tax revenues from GNP, and the company contributes 88 percent of East Millinocket's property tax revenues. The assessed value of GNP's facilities, including real and personal property, within the town of Millinocket is approximately \$228 million. Approximately 17 percent of this amount is the value of real property and generates \$1.004 million annually in revenue to Millinocket. The greatest proportion of assessed value (\$4.902 million in revenues, or 83 percent) is personal property, which includes equipment, furnishings, and vehicles. During 1993, East Millinocket received \$3,690,295 from GNP; \$520,331 for real property and improvements and \$3,140,441 in personal property. FERC received comments from various Chambers of Commerce, labor unions, and the local delegation of the Maine Assembly that supported granting a new license that would not exacerbate declining economic conditions for GNP or the project area.

The proposed 500-foot conservation easements could affect property tax revenues of local governments. The proposed conservation easements within the watershed would restrict future development within a 500-foot setback of the high water mark. In areas where the conservation easement is applied, lot owners still would be able to sell their property or to assemble sufficient acreage to build one residence set back 500-feet from the shoreline. This would result in the same number of shoreline lots as with other project alternatives.

We expect that lot values would adjust to the revised setbacks and that county valuations and property tax revenues would not be adversely affected.

4.12.2.2 Effect on GNP Operations

GNP has operated its hydropower system according to the WUP recommendations since May 1991. GNP determined that further flow requirements would result in system instabilities, tightly constrained maintenance schedules, and unacceptable constraints on GNP's generating reserves. These restraints would directly affect the transmission system and paper-making machinery. Increased Upper Gorge flows under this alternative would add an estimated \$233,300 to GNP's internal generation costs and \$315,200 in purchased power costs. Increased the Back Channel flows would add an estimated \$906,000 (1991 dollars) to GNP's internal generation costs and \$1.73 million (1991 dollars) to purchase supplemental power. Increased minimum flows in Millinocket Stream would result in \$31,000 (1991 dollars) in lost power costs.

GNP contends that providing flows of 350 cfs in the Back Channel would require retiring machines 7 and 8 and the blade coater (lightweight coated paper complex) at Millinocket. GNP estimates that these losses would total 93,700 tons of paper production annually, approximately 238 mill jobs, and \$100 million in annual state output. GNP maintains that continued economic viability of the lightweight coated paper complex is critical to future capital investments planned through 1997. The staff estimates this alternative would cost GNP \$4,487,600 annually (1996 dollars).

To estimate the potential effect of GNP acquiring shoreline conservation easements, we reviewed various cost estimates and appraisals (Bangor Real Estate, 1995; Thompson Appraisal Company, Inc., 1995; the testimony of Robert Fiske, January 25, 1995; and LURC, 1990a). Based on this review, the staff estimated costs (per waterfront footage) for obtaining easements for the non-GNP owned shorelines in the Ripogenus and Penobscot Mills Project areas. The staff estimated the approximate costs to be \$8.316 million (1996 dollars) for the Ripogenus Project area and \$16.35 million for the Penobscot Mills Project area.

4.12.2.3 Cumulative Impacts

Alternative 1 would have cumulative economic impacts in addition to the significant, project-specific adverse effects. The project area is heavily dependent on GNP, directly and indirectly, for employment and retirement income, purchases, and tax revenue to support local government services. Economic conditions outside the Commission's licensing proceeding have caused mill layoffs, a higher local unemployment rate, a reduction of production and corresponding reduction in the purchase of goods and services, and a decline in population

from limited employment opportunities. This project alternative would create significant cumulative adverse effects on the socioeconomic resources of the region.

4.12.3 Alternative 2

4.12.3.1 Employment and Fiscal Impact

Alternative 2 considered increased minimum flows in Upper Gorge, Millinocket Stream, and the Back Channel and other enhancements in addition to the measures proposed by GNP. GNP asserts that providing minimum flows in excess of 50 cfs in the Back Channel would jeopardize 238 mill jobs.

This alternative also would restrict excessive shoreline development in the Ripogenus Project area through the 250-foot conservation easement or the 200-foot boundary expansion on GNP-owned lands and in the Penobscot Mills Project area through a 200-foot boundary expansion and through forestation and buffering requirements along the project impoundments and rivers. The buffer requirements would maintain the same number of shoreline lots and would not affect the maximum residential development of the area. The buffer requirements affect site development standards, not the development potential or indirectly assessed and market values of the lots; therefore, no adverse impact on the tax valuation or property tax revenues would accrue.

4.12.3.2 Effect on GNP Operations

Flows considered in Alternative 2 would affect mill operations by increasing flows in Millinocket Stream and the Back Channel. GNP estimates that average annual losses of GNP's internal generation would be \$16,000 and an additional \$32,510 in purchased power (1991 dollars). For the Back Channel flows of 150 to 165 cfs, the range of internal generation costs to GNP would be \$392,000 to \$431,200. Purchased power needed to account for these increased flows would cost between \$749,000 and \$823,900 (1991 dollars; GNP, 1994). The total cost to GNP for increased minimum flows under Alternative 2, therefore, would range from \$408,000 to \$447,200 in internal generation costs and from \$781,510 to \$856,410 in purchased power costs.

Under this alternative, the proposed conservation easement and boundary expansion would be on GNP-owned properties and would require no direct acquisition costs to GNP. The staff estimates the annual cost to GNP of this alternative would range from \$959,500 to \$1,465,500 (1996 dollars).

4.12.3.3 Cumulative Impacts

With increased minimum flows to Millinocket Stream and the Back Channel, GNP would incur an estimated \$408,000 to \$823,900 in average annual costs. GNP states that any minimum the Back Channel flows greater than 50 cfs would directly jeopardize 238 mill jobs (GNP, 1991a) in addition to the estimated 980 jobs lost to the Millinocket/East Millinocket Labor Market Area between 1985 and 1992 (letter from J. Haskell, Town Planner, town of Millinocket, to J. Kotredes, Town Manager, town of Millinocket, August 28, 1993). This is a

potentially significant cumulative impact on employment and socioeconomic resources of the region.

4.12.4 No-action Alternative

Potential impacts of the No-action Alternative, which are similar to the Applicant's Proposal, include minor impacts on employment, population, fiscal conditions, and tax revenues.

4.13 UNAVOIDABLE ADVERSE IMPACTS

Construction associated with the proposed recreation facilities, wetlands enhancements, and the Holbrook Stream nursery area would create some short-term, unavoidable adverse impacts from increased dust, noise, heavy equipment traffic, and increased water turbidity.

4.14 IRREVERSIBLE AND IRRETRIEVABLE COMMITMENT OF RESOURCES

Continued operation of the Ripogenus and Penobscot Mills projects would perpetuate the commitment of lands and waters previously developed for energy production.

4.15 RELATIONSHIP BETWEEN SHORT-TERM USES AND LONG-TERM PRODUCTIVITY

The projects are expected to continue to provide an average of 108 MW of energy production capacity for GNP's paper mills. This long-term productivity would extend at least as long as the duration of the licenses (30 years). The recommended alternative is designed to provide significant long-term enhancement of biological and recreational resources of the system, while meeting energy and economic needs of GNP and the local population.

If the projects were to operate solely to maximize hydroelectric generation, long-term productivity of the impoundment and river fisheries would decrease because increased drawdowns during spawning and incubation periods and reduced minimum flows in reaches such as Millinocket Stream and Upper Gorge would reduce habitat suitability. With the recommended alternative and appropriate enhancement at each site, however, there should be significant, long-term enhancement of existing aquatic resources.

5.0 STAFF'S CONCLUSIONS

5.1 INTRODUCTION

GNP proposes no major new construction or modifications of the Ripogenus or Penobscot Mills projects but seeks to operate them essentially as it has over the past 50 years, except for several resource enhancements. Several intervenors and agencies seek additional enhancements to further improve existing fisheries, recreational opportunities, and other environmental conditions. Negotiations between GNP and some intervenors preceding preparation of this FEIS resolved several issues, including access fees and flow releases for whitewater recreation. Other parties expressed concerns that measures proposed for resource enhancement would have adverse socioeconomic consequences on the local region.

One primary issue in this licensing process is whether GNP should be required to release flows substantially higher than it proposes into the west branch of the Penobscot River downstream of Ripogenus Dam, in Millinocket Stream, and in the Back Channel to enhance fisheries resources and recreational opportunities. GNP contends that any increase in nongenerating flows that would increase its hydropower costs might force the company to reduce paper production at local mills or to purchase more expensive power from regional suppliers. GNP also contends that this would put the company at a disadvantage in the paper market and cause workforce reductions and a decline in GNP's economic activity. Such declines could affect the local economy severely because GNP is the primary local industry, accounting for one-third of all jobs and a wage and benefit input of \$121 million annually, all concentrated in the Millinocket area. We evaluated the validity of these positions in section 4 and summarize our findings in our balancing discussion presented in this section.

GNP also claims that it cannot provide flow releases of the magnitude sought by intervenors because of constraints on impoundment water level management (i.e., draw-down limits to protect lake trout in North Twin, preserve recreational opportunity on the project impoundments, and maintain storage capacity for flood control). Intervenors question the validity of the model that GNP used to support its position and suggest that additional, upstream storage impoundments should be considered in this FEIS because they might provide water needed to meet all enhancement goals simultaneously. We analyzed the feasibility of various flow regimes in section 4.2 and summarize our conclusions here.

A second major issue addressed is the extent to which protection from development around project impoundments is necessary to preserve existing aesthetics, recreation and habitat. In this FEIS we evaluated two enhancement measures that GNP did not propose or adopt and that would impose significant additional cost on GNP: enhancing wetlands and expanding the boundaries of the projects around the impoundment shorelines to preserve regional aesthetics and protect water quality and the terrestrial ecosystem. We balanced the cost of the enhancements against the benefits in the same way that we addressed flow releases to determine the appropriate level of enhancement.

5.2 PROJECT ALTERNATIVES

We evaluated three alternatives, using the No-action Alternative as a baseline for comparison. The Applicant's Proposal includes minimum flows, recreational facilities, further studies, and strict schedules for impoundment draw-downs. Alternative 1 includes reasonable flow enhancements to benefit fisheries and other measures approaching those sought by conservation intervenors and some agencies. In defining Alternative 2, we considered several enhancement measures intermediate between those proposed by GNP and those included in Alternative 1 that would achieve some of the enhancement and mitigation benefits sought by conservation intervenors and some agencies but would provide a better balance among developmental and nondevelopmental values, eventually selecting only a few measures beyond those proposed by GNP. Table 5-1 summarizes the enhancement measures included in each alternative.

5.3 ENVIRONMENTAL COMPARISON OF ALTERNATIVES

The key issues in relicensing the Ripogenus and Penobscot Mills projects are the extent to which greater flow releases would enhance fisheries and the degree to which such releases would affect the economic status of GNP's paper production facilities. Adverse economic effects on GNP could result in loss of jobs and economic activity in communities in the project areas. Greater achievement of fisheries objectives would have a greater economic effect on GNP. The alternatives we defined allowed us to identify the maximum range of possibilities for balancing these conflicting objectives and to establish a basis for identifying an equitable licensing solution. We also evaluated if flow regimes described in the Applicant's Proposal and alternatives could be sustained in the context of overall water management within the project area; therefore, we first determined the feasibility of providing all flows defined in the various alternatives while operating within existing and future constraints on impoundment draw-downs and mandatory flow releases.

During scoping, we indicated our intent to evaluate the need for additional flows to enhance opportunities for whitewater recreation. Because of agreements negotiated between GNP and whitewater recreation intervenors and our own analysis of the potential benefits of additional flow enhancements for recreation, we concluded that no flow enhancements beyond those proposed by GNP are necessary for whitewater recreation. We did not include additional recreational flows in alternatives 1 or 2, and we do not view flows for recreation as a major issue.

Two other major issues relate to the benefits and costs of enhancing wetlands and expanding the projects' boundaries. Table 5-2 summarizes the effects or costs of each alternative for each resource. Below we summarize our comparison of alternatives for each of the five major licensing issues: (1) streamflow (i.e., the feasibility of providing flows proposed in all alternatives), (2) fisheries enhancement, (3) wetlands mitigation, (4) land use (i.e., habitat preservation), and (5) economic and socioeconomic effects.

Table 5-1. Summary of enhancement measures in the Applicant's Proposal and alternatives for the Ripogenus and Penobscot Mills projects. Measures listed under the No-action Alternative describe existing conditions; only measures that differ among at least some of the alternatives are included (Source: Staff).				
Enhancements	Applicant's Proposal	Alternative 1	Alternative 2	No-action Alternative
Upper Gorge Flows	100 cfs summer, leakage (12 cfs) and spillage rest of year	100 cfs summer; 50 cfs (and spillage) rest of year	100 cfs summer; leakage (12 cfs) and spillage rest of year	Leakage and spillage year round
West Branch Flows	Increased flows for rafting and incubation	g, during outages, and for en	hancement of salmon spawning	200 cfs minimum flow
North Twin Water Level	"Relatively stable" water minimum lake level for la lake trout incubation	levels for wildlife, wetlands, reke trout spawning, and great	ecreation, and aesthetics; er than minimum lake level for	Less stable reservoir levels
Back Channel Flows	Leakage and spillage	350 cfs	leakage and spillage	Leakage and spillage
Millinocket Stream Flows	60 cfs spring-fall 20 cfs fall-spring	60 to 80 cfs year round	60 cfs year round	20 cfs year round
Conservation Easement/ Vegetative Buffer	250-foot conservation easement for 73 shore- tine miles at the Ripo- genus Project area. None at Penobscot Mills Project.	500-foot building setback 250-foot vegetative buffer	250-foot conservation easement or a 200-foot boundary expansion on GNP- owned lands (Ripogenus) and 200-foot boundary expansion (Penobscot Mills)	None
Wetlands Enhancement	3 sites at North Twin impoundment	2 sites at Ripogenus 2 sites at North Twin	1 site at Ripogenus 2 sites at North Twin	None
Terrestrial Enhancement	Back Channel wildlife habitat management	None	Back Channel wildlife habitat management and maintenance	None
Threatened and Endangered Species	None	Aquatic bird nesting platforms	Aquatic bird nesting platforms	None
Fish Passage	Repair/modify North Twin ladder		No improvement of North Twin ladder	
Recreational Fisheries	Stocking brook trout in Millinocket Stream	None	Stocking brook trout in Millinocket Stream	None
Recreation Facilities	Enhance vehicle parking	areas and boat access		None

a canagina.

- Callab-

Table 5-2. Summary comparison of environmental and economic consequences of the Applicant's Proposal and alternatives for the Ripogenus and Penobscot Mills projects (Source: Staff)				
Resource Area	Applicant's Proposal	Alternative 1	Alternative 2	
Annual Net Benefit (1996 dollars)	-\$877,600	- \$4,487 ,600	-\$959,500	
Streamflow (both projects)	Water available for all enhancements <u>even</u> during very dry years	Water available for all enhancements <u>except</u> for several weeks during very dry years	Water available for all enhancements <u>even</u> during very dry years	
Fisheri es				
Ripogenus Project	Salmon stock enhanced below McKay station; angling opportunity increased in Upper Gorge (1 percent increase in stream length), but no increase in salmon stock.	Additional salmon habitat in Upper Gorge, but no sustained population increases.	Salmon stock enhanced below McKay station; angling opportunity increased in Upper Gorge (1 percent increase in stream length), but no increase in salmon stock.	
Penobscot Mills Project	Modest increase in salmon spawning and nursery habitat in Millinocket Stream and a small increase in regional salmon abundance. Enhancement of lake trout spawning in North Twin. No change in Back Channel fisheries habitat.	Potential doubling of adult and spawning habitat in Millinocket Stream and a small increase in the regional salmon stock (larger than under the Applicant's Proposal). North Twin lake trout spawning habitat jeopardized in dry years. Establishment of self-sustaining Back Channel salmon stock unlikely; if established, optimistic maximum population approximately several hundred legal-sized salmon.	Potential doubling of adult and spawning habitat in Millinocket Stream than under the Applicant's Proposal and a small increase in regional salmon abundance (approximately equal to that under the Applicant's Proposal). Enhancement of lake trout spawning in North Twin. No change in Back Channel fisheries habitat.	

Table 5-2. (Continued)				
Resource Area	Applicant's Proposal	Alternative 1	Alternative 2	
Wetlands (acreage of enhancements)				
Ripogenus Project	None	Quaker Brook: 100 acres Black Pond: 250 acres	Black Pond: 250 acres	
Penobscot Mills Project	Deep Cove West: 15 acres Deep Cove East: 15 acres Wadleigh Brook: 15 acres	Deep Cove West: 15 acres Deep Cove East: 15 acres	Deep Cove West: 15 acres Deep Cove East: 15 acres	
Total	45 acres	380 acres	280 acres	
Land Use	Protection against land use changes and potential aesthetic and terrestrial effects within the 250-foot conservation easement for 73 shoreline miles at the Ripogenus Project area. No change at Penobscot Mills Project.	Protection against land use changes and potential aesthetic, water quality and terrestrial impacts within 500 feet of all project impoundments.	Protection against land use changes and potential aesthetic, and terrestrial effects within the 250-foot conservation easement along 73 shoreline miles or the 200-foot boundary expansion on GNP-owned lands of the Ripogenus Project area and within the 200-foot boundary expansion along GNP-owned lands of the Penobscot Mills Project area.	
Socioeconomics (job losses)	No change	Potential loss of 238 mill jobs (383 jobs total including multiplier effects)	No change	
Cultural Resources	Protection of cultural resources through CRMP & PA	Protection of cultural resources through CRMP & PA	Protection of cultural resources through CRMP & PA	

μ.

Table 5-2. (Continued)				
Resource Area	Applicant's Proposal	Alternative 1	Alternative 2	
Terrestrial				
Ripogenus Project	No significant impact (NSI) for terrestrial resources. Protection of terrestrial resources within the 250-foot easement at the Ripogenus Project area.	Future protection of all impound- ments with 500-foot building setback	Future protection at Ripogenus impoundment with 250-foot conservation easement or 200-foot boundary expansion	
Penobscot Mills Project	Slight enhancement of wildlife in Back Channel	Minor negative effects on wildlife (beaver) in Back Channel because of increased flows	Future protection on GNP-owned land around impoundments with 200-foot project boundary expansion; slight enhancement of wildlife in Back Channel	
Threatened and Endangered Species (both projects)	NSI for bald eagles or other rare species	500-foot building setback could provide enhanced habitat protection for bald eagle	Proposed conservation easement and boundary expansion provide enhanced habitat protection for bald eagle	
Recreation (both projects)	Improved fishing (Upper Gorge in summer and Millinocket Stream in summer and fall, with brook trout stocking) and whitewater boating	Improved fishing and whitewater boating (both slightly greater than Applicant's Proposal, including opportunities in Back Channel)	Same as Applicant's Proposal except slightly greater increase in salmon angling opportunity in Millinocket Stream	

5.3.1 Streamflow

GNP contends that availability of water is a significant constraint on releasing the higher flows sought by the CI (Alternative 1). Using documentation provided by GNP we determined that the model is a suitable tool for evaluating the feasibility and consequences of implementing the various flow regimes. (Appendix D contains responses to all concerns raised by the CI about the model and our detailed evaluation of the model.)

We conclude that flows and impoundment-level constraints proposed in all alternatives could be met simultaneously during average and wet years without additional releases from storage impoundments upstream of the Ripogenus Project. All flow and impoundment-level objectives could not be met simultaneously under Alternatives 1 and 2 during dry years unless flows through the mill were reduced. Even then, however, those objectives would have to be compromised only slightly and for only a brief portion of the year to attain the desired flow releases.

We disagree with GNP's contention, therefore, and conclude that sufficient water is available to attain flow releases within other water management constraints (with some slight modification) under all alternatives.

5.3.2 Fisheries Enhancement

The Ripogenus and Penobscot Mills projects encompass several interconnected impoundments and segments of riverine waters. Although some fish species inhabit most or all bodies of water within the project area (see section 3.5), in nearly all cases those populations do not interbreed or depend on each other to sustain their abundance. For example, the status of the landlocked salmon stock below McKay station on the West Branch is not affected by the status of landlocked salmon in Millinocket Stream or the Back Channel. Project waters, therefore, do not support a single stock of any fish species that could be affected by the cumulative operations of all developments. The status of the recreational fisherv for some species (e.g., landlocked salmon and smallmouth bass) within the project region of Maine, however, would be affected by the collective abundance of those species in all waters in the area. For this reason, our comparison of the effects of the three project alternatives on fisheries focuses on the degree to which enhancement measures in all project waters would affect the overall abundance of key fish species, in particular Atlantic salmon, in the project areas. Only the migratory eel population supports what may be considered a single stock throughout project waters. We conclude that the differences among the alternatives in impact to eels are minor.

Under the Applicant's Proposal, the presently abundant landlocked salmon stock below McKay station would be further enhanced by eliminating emergency termination of flow, stabilizing flows during the spawning season, and creating a small amount of new spawning and nursery habitat. Providing a summer flow of 100 cfs in Upper Gorge would increase potential recreational fishing opportunity (about 1 percent increase in the length of stream supporting landlocked salmon in the upper West Branch) but would not contribute to an increase in salmon abundance. Providing a 60-cfs flow in Millinocket Stream between May 1 and October 15 would create only a modest amount of additional spawning and nursery habitat for salmon (25 to 30 percent increase for juveniles and adults). The existing population in Millinocket Stream is small and supported primarily by hatchery fish passing downstream over Millinocket dam; therefore, the enhancement would contribute little to regional landlocked salmon abundance and recreational fisheries. GNP proposes no additional flow in the Back Channel; therefore, that area would not contribute to enhancement of existing regional fisheries resources. Because water use objectives would not conflict during dry years under the Applicant's Proposal, North Twin impoundment levels could be managed to benefit lake trout.

Although the higher flows in Upper Gorge, Millinocket Stream, and the Back Channel proposed in Alternative 1 probably would enhance the aquatic ecosystems of those reaches significantly (e.g., enhancing benthic invertebrate populations by increasing available aquatic habitat), those flows would cause only a very limited increase in regional landlocked salmon stocks or other gamefish. There may also be slight benefit to American eel.

Even though year-round minimum flows in Upper Gorge would create a small amount of additional winter habitat for landlocked salmon, sustained population increases would be unlikely because high spillage flows during spring would continue to adversely affect the production value of fisheries habitat in that river reach.

The salmon population in Millinocket Stream is small (number of spawners ranged between 9 and 20 fish over 3 years, probably of hatchery origin), and the enhanced flows proposed in Alternative 1 would only double the small amount of available adult and spawning habitat (with potential doubling of the existing small population).

Even though a 350-cfs flow in the Back Channel would provide maximum habitat for nursery and fry life stages of landlocked salmon, a larger, self-sustaining salmon stock probably would not be established under this flow regime because the 2.7-mile reach in which those fish would have to remain throughout their life cycle probably could not support significant numbers of all life stages of salmon year-round. High-volume spillage during spring would displace early life stages and juveniles downstream significantly. Juveniles dropping downstream into Shad and Dolby ponds would encounter predation and competition from smallmouth bass and pickerel in those waters. Fish that survived to adulthood could be delayed at Grand Falls when attempting to return to their upstream natal spawning area under some flow conditions. With no production above Stone Dam, no juveniles would be recruited into the Back Channel from upstream. For these reasons, the Back Channel probably could not be established as a production area that would enhance salmon abundance downstream.

Even if the Back Channel could sustain a population, it probably would not be of substantial size. Population studies have documented approximately 200 legal-sized (16 inches) salmon per mile in the 21-mile-long reach of the West Branch below McKay station, one of the premier landlocked salmon areas in the country. The Back Channel from Stone dam to Grand Falls is 2.7 miles long and would provide less adult habitat per mile than the West Branch, even with the 350 cfs flow. At best the Back Channel might support several hundred legal-sized landlocked salmon.

As discussed in section 5.3.1, the flow regimes associated with Alternative 1 also may preclude simultaneous attainment of all water use objectives during dry years. Although minor modifications of some water use objectives would allow all to be met, we accounted for the

benefits of some of those objectives for fisheries in our evaluation of the alternatives. Under Alternative 1, maintaining 350-cfs flows in the Back Channel (in addition to 2,000 cfs through the mill at Millinocket) and the North Twin draw-down limits specified to protect lake trout spawning habitat may not be possible during a low-flow year. Although establishing a reproductively viable lake trout population in North Twin seems feasible, a self-sustaining landlocked salmon stock probably could not be established in the Back Channel, even with the 350-cfs flow. Implementing Alternative 1, therefore, could jeopardize an attainable fisheries goal (i.e., establishing a lake trout population in North Twin) in an attempt to achieve a less attainable and less valuable goal (i.e., establishing a self-sustaining landlocked salmon stock in the Back Channel).

Several minimum flows less than 350 cfs in the Back Channel would provide more salmon habitat than the flows proposed by GNP, but those increases would not result in significant increases in the regional availability of landlocked salmon to recreational fisheries. Based on those evaluations and accounting for the cost of flow releases (as discussed below), we included only the leakage flow in the Back Channel proposed by GNP in Alternative 2. The enhanced fall/winter flows in Millinocket Stream included in Alternative 2 would be the same as under Alternative 1 and would provide some enhancement of landlocked salmon (a substantial increase in juvenile habitat and protection of redds from freezing), but the benefit to regional fisheries resources would be small because of the small size of the salmon stock in that stream. Water use objectives would not conflict during dry years under Alternative 2.

In summary, the flow releases in the Applicant's Proposal would enhance regional fisheries resources compared with the baseline condition, particularly landlocked salmon in the West Branch downstream of McKay station and lake trout in the North Twin impoundment. Increased flows considered in Alternative 1 would provide greater enhancement of the aquatic ecosystems of Upper Gorge, Millinocket Stream, and the Back Channel but only small increases in regional availability of landlocked salmon, the most important and valuable regional fish species. Flows under this alternative also could have a slight negative effect on lake trout in North Twin during dry years. Fisheries benefits under Alternative 2 would be the same as under the Applicant's Proposal, except for a somewhat greater enhancement of salmon habitat and populations in Millinocket Stream.

5.3.3 Wetlands

Under the No-action Alternative, fluctuating water-levels would continue to adversely affect dispersed areas of wetlands (totaling about 325 acres) along the shorelines of project impoundments that experience significant water level fluctuations, particularly Ripogenus and North Twin lakes. These habitats would continue to provide only minimal wetlands functions and values compared with other project wetlands that are not affected by fluctuating water levels.

Under the Applicant's Proposal, wetlands enhancement would be implemented at two sites along North Twin impoundment and one site along Ripogenus impoundment, totaling 45

acres.¹³ Increased seasonal flows in Upper Gorge and Millinocket Stream would provide some mitigation of functions and values for the small amount of wetlands along the main course of those stream segments.

Alternative 1 includes developing four of the five potential areas for wetlands enhancement identified by GNP: Deep Cove East and Deep Cove West at North Twin, and Quaker Brook and Black Pond at Ripogenus, which comprise about 380 acres. These enhancements would maintain permanent hydroperiods within these degraded shoreline areas. Enhancement at the Wadleigh Brook site at North Twin could significantly degrade adjacent existing wetlands; therefore, we conclude that enhancement activities should be avoided there.

The minimal wetlands vegetation in Millinocket Stream and Upper Gorge would benefit from the increased hydroperiod provided by the year-round flows proposed under Alternative 1. High flows in the Back Channel would alter existing wetlands but could create additional wetlands. Alternative 1 offers the greatest benefit to wetlands within the project areas.

Alternative 2 includes wetlands enhancement at Black Pond at the Ripogenus impoundment, where the functions and values of about 250 acres of degraded shoreline wetlands would be enhanced by increasing water retention time during draw-downs. Alternative 2 also would provide wetlands enhancement at the Deep Cove East and Deep Cove West sites at North Twin. The enhancements would significantly improve the functions and values of about 30 acres of degraded shoreline wetlands. This alternative would not affect wetlands along the Back Channel but would enhance hydrology in Millinocket Stream slightly.

In summary, Alternative 1 provides wetlands enhancement of an acreage substantially greater than the estimated affected acreage without action, whereas Alternative 2 includes enhancement acreage almost equal to the existing affected acreage. Alternative 2 would provide greater flows in Millinocket Stream wetlands than would the Applicant's Proposal (60 cfs from October 16 to May 1 instead of 20 cfs). The Back Channel wetlands would retain their current status under Alternative 2 and the Applicant's Proposal but could experience some changes (both losses and gains) under Alternative 1.

5.3.4 Land Use

The Applicant's Proposal includes a 250-foot conservation easement for approximately 73 shoreline miles of GNP-owned lands for the Ripogenus Project area. The land would be conveyed to the state of Maine and maintained for the duration of the license period as put forth in the terms of the MOU (see section 4.9). These easements would provide adequate shoreline protection for the term of the license. The Applicant's Proposal includes no easements or project boundary expansion for the Penobscot Mills Project and LURC's current regulations would apply (see section 4.9). Although LURC's current regulations adequately manage shoreline development, these regulations are not directly tied to the license and

¹³ Although GNP's application does not include proposed wetlands enhancement, Maine's WQC Summary of Findings indicates that GNP agreed to implement this enhancement.

would not guarantee long term protection of the aesthetic and natural recreational experience for the term of the license.

Alternative 1 includes a 500-foot expansion of the project boundaries (including a 500-foot building setback, and a 250-foot vegetative buffer zone) around all impoundments of both projects, as recommended by CI (letters from Daniel Sosland, CI, September 3, 1993 and November 2, 1993). The building setback would protect aesthetic and recreational resources within the project area. The 250-foot vegetative buffer would provide further protection of water quality from the effects of logging operations and exceed buffer widths recommended by the Forest Service (FS, 1992) and the state of Maine (DOC, 1992; see section 4.9). The buffer zone also would protect existing riparian terrestrial habitat and benefit bald eagles in the project area (see section 4.7).

Alternative 2 includes a 250-foot conservation easement along approximately 73 shoreline miles of GNP-owned lands or a 200-foot boundary expansion on GNP-owned lands within the Ripogenus Project area and a 200-foot boundary expansion for all GNP-owned lands within the Penobscot Mills Project area. This alternative would protect the aesthetics, water quality and recreational use of the project area nearly as well as the larger 500-foot project boundary expansion and would provide further protection of the Penobscot Mills Project area than the Applicant's Alternative.

The No-action Alternative would not change current project boundaries or create vegetative buffers along impoundment shorelines. Current regulations would allow for development along the shoreline areas (see section 4.9.1). Although little development has occurred along the project impoundment shorelines to date, this is primarily because of GNP's moratorium on granting new leases. A recent LURC study found that more than 53 percent of all recent residential development in the project area occurred along shorelines of lakes with high scenic value, such as the Ripogenus impoundment. GNP does not guarantee that it will continue its current leasing moratorium. Potential development under either alternative could adversely affect aesthetics and recreational use of the project area (see sections 4.3, 4.8 and 4.10).

For the Ripogenus Project area, the conservation easements proposed under the Applicant's Proposal and Alternative 2 and the boundary expansions proposed under Alternative 1 would protect the natural character of the project areas for the 130,000 people who visit the West Branch region each year for recreation or sightseeing for the term of the license. Under the Applicant's Proposal, no conservation easement or boundary expansion is proposed for the Penobscot Mills Project area, and long term protection of the aesthetic and natural recreational experience would not be guaranteed for the term of the license.

Alternative 1 and Alternative 2 provide protection of the aesthetic, recreational, and terrestrial resources within both the Ripogenus and Penobscot Mills Project areas. The difference in the degree of protection provided by the two alternatives cannot be quantified rigorously; however, our qualitative evaluation of benefits suggests that the additional protection of the 500-foot expansion does not merit the much higher cost of that alternative.

5.3.5 Economic and Socioeconomic Effects

The pulp and paper products industry is highly competitive and energy-intensive. GNP's competitive position depends on the availability of a reliable source of inexpensive electric power. Imposing additional production costs for company-generated power could adversely affect the economic viability of the Millinocket and East Millinocket paper mills. GNP indicated that any changes of its proposed water management plan (including flow releases and impoundment water level management) that would add incremental costs beyond those associated with the plan would threaten the long-term viability of the Millinocket and East Millinocket mills. The confidentiality of GNP's financial status prevents an independent assessment, but the company suggests that environmental enhancements that would reduce generation more than reductions associated with its proposal would require it to curtail paper production rather than purchase replacement power to maintain existing production levels.

Section 2.4 defines the assumptions we used to analyze the economic effects of each of the proposed alternatives. Each alternative includes a distinct combination of average annual changes in generation, power values, annual project costs, and annual net benefits. Because none of the alternatives involves new construction for power generation, the annual net benefit (ANB) is the sum of the costs of replacement power and environmental enhancements.

Table 5-3 shows the incremental ANBs associated with the Applicant's Proposal and the three alternatives. The table includes the incremental ANBs associated with two different Back Channel flows (leakage and 165 cfs) that we considered in the process of defining Alternative 2. The ANB of the Applicant's Proposal is negative at a cost of \$877,600; the incremental ANB of Alternative 1 is five times more negative, at a cost of \$4.487 million. To define Alternative 2, we considered several minimum flows in the Back Channel for the Penobscot Mills Project, ranging between leakage (proposed by GNP) and the 350 cfs included in Alternative 1. A 165-cfs flow makes the ANB more negative by \$436,100 and would provide only a minor enhancement of fisheries resources in the project. The 350-cfs the Back Channel flow would make the ANB more negative by \$916,300. Based on project economics and fisheries benefits, we concluded that no intermediate flow is warranted; therefore, Alternative 2, our final recommendation, includes only leakage in the Back Channel. The incremental ANB of Alternative 2 is -\$959,500, about nine percent more costly than the ANB with the Applicant's Proposal.

Hydroelectric power from the Ripogenus and Penobscot Mills projects is an important component of GNP's electricity supply. The combined capacity of the Ripogenus and Penobscot Mills projects constitutes 34.1 percent of GNP's sustainable generation capacity. Based upon historic generation trends, hydroelectric power from the two projects is expected to provide 620,400 megawatt hours, or nearly one-half of the company's 1996 energy needs.

Except for the No-action Alternative, all alternatives would reduce annual power generation from the Ripogenus and Penobscot Mills projects by diverting flows to improve aquatic habitats, enhance wetland areas, and enhance recreation and lake aesthetics. Alternatives that would release more than leakage flows through the Back Channel would

Applicant's Proposal ^{®)}	Alternative 1 ^(c)	Alternative 2: ^(d) Final Recommendation	Alternative 2: ^(a) Intermediate Recommendation		
Ripogenus Project					
\$-145,300	-\$145,300	-\$145,300	-\$145,300		
-\$96,900	-\$235,900	-\$96,900	-\$166,800		
\$0	\$-105,600	-\$57,700	-\$57,700		
\$0	-\$1,700	-\$1,700	-\$1,700		
	-\$817,500	\$0	\$0		
-\$242,200	-\$1,306,000	-\$301,600	-\$371,500		
-\$224,700	-\$224,700	-\$224,700	-\$224,700		
\$0	-\$3,900	-\$3,900	-\$3,900		
\$0	-\$1,607,400	\$0	\$0		
-\$15,400	-\$34,000	-\$34,000	-\$34,000		
\$0	-\$916,300	\$0	-\$436,100		
-\$240,100	-\$2,786,300	-\$262,600	-\$698,700		
-\$200,100	-\$200,100	-\$200,100	-\$200,100		
-\$195,200	-\$195,200	-\$195,200	-\$195,200		
-\$395,300	-\$395,300	-\$395,300	-\$395,300		
-\$877,600	-\$4,487,600		-\$1,465,500		
	Proposat ^{es)} \$-145,300 -\$96,900 \$0 \$0 \$0 -\$242,200 -\$224,700 \$0 -\$20,100 -\$240,100 -\$2395,300 -\$877,600 ative.	Proposal** \$-145,300 -\$145,300 -\$96,900 -\$235,900 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-105,600 \$0 \$-1,306,000 \$0 \$-\$1,306,000 \$0 \$-\$1,306,000 \$0 \$-\$1,306,000 \$0 \$-\$1,306,000 \$0 \$-\$3,900 \$0 \$-\$3,900 \$0 \$-\$1,607,400 \$0 \$-\$34,000 \$0 \$-\$34,000 \$0 \$-\$16,300 \$0 \$-\$200,100 \$2200,100 \$2200,100 \$395,300 \$395,300 \$877,600 \$4,487,600 ative.	Proposal ^(%) Recommendation \$-145,300 -\$145,300 -\$145,300 -\$96,900 -\$235,900 -\$96,900 \$0 \$-105,600 -\$57,700 \$0 -\$105,600 -\$57,700 \$0 -\$11,700 -\$1,700 \$0 -\$11,700 -\$1,700 \$0 -\$11,700 -\$1,700 \$0 -\$11,306,000 -\$11,700 -\$242,200 -\$1,306,000 -\$301,600 -\$224,700 -\$224,700 -\$224,700 -\$224,700 -\$224,700 -\$224,700 \$0 -\$1,607,400 \$0 \$0 -\$1,607,400 \$0 \$0 -\$1,607,400 \$0 \$0 -\$1,607,400 \$0 -\$100 -\$24,700 \$0 -\$240,100 -\$2,786,300 -\$34,000 -\$200,100 -\$262,600 -\$195,200 -\$200,100 -\$200,100 -\$262,600 -\$200,100 -\$200,100 -\$200,100 -\$200,100 -\$200,100 <		

Includes Upper Gorge minimum flow of 100 cfs 7/1-9/30; leakage (~ 12 cfs) flow 10/1-6/30; Millinocket Stream flow of 60 cfs 5/1-10/15 and leakage (~ 20 cfs) 10/16-4/30.

Includes 350 cfs in Back Channel; Upper Gorge minimum flow of 100 cfs 7/1-9/30 and 50 cfs from 10/1-6/30; Millinocket Stream minimum flow of 60 cfs year round.

⁽⁶⁾ Includes Upper Gorge minimum flow of 100 cfs 7/1-9/30 and leakage (~ 12 cfs) 10/1-6/30; Millinocket Stream minimum flow of 60 cfs year round; leakage in Back Channel.

(a) Includes Upper Gorge minimum flow of 100 cfs 7/1-9/30 and 30 cfs 10/1-6/30; Millinocket Stream minimum flow of 60 cfs year round; 165 cfs in Back Channel.

an and the first

result in the greatest reduction in power generation, reducing it from 5.0 to 7.0 percent below the anticipated 1996 power production from the two projects. Alternative 2 (Final Recommendation), which releases only leakage flows through the Back Channel, would reduce power production from its level under the No-action Alternative by about 3 percent.

Hydroelectric power is GNP's least expensive source of electricity; it can be produced at about half the cost of the least-cost alternative of \$73.92/MWh. Any reduction of power production for environmental enhancement, therefore, would impose additional power costs upon GNP. The staff has estimated that GNP's power production costs would increase by between 0.5 percent (Applicant's Proposal) and 13.4 percent (Alternative 1).

We estimate that the power cost of the alternative impoundment levels and minimum flows, mitigation, and enhancement of other resources would range from 34.3 mills/kWh (Applicant's Proposal) to 38.67 mills/kWh (Alternative 1). Alternative 2 (Final Recommendation) would result in a cost of power of 34.41 mills/kWh.

We agree with GNP that the Millinocket and East Millinocket paper mills are high cost producers in a highly competitive market. We concur that environmental enhancement alternatives that would add significant replacement power costs could adversely affect the economics of the least competitive paper production processes at the Millinocket and East Millinocket Mills, resulting in production cutbacks. GNP indicated, however, that although annual costs of replacement power would place the coater complex in a long-term break-even economic position, substituting replacement power for hydroelectric generation losses caused by flow enhancements included in its own proposal would not affect the long-term operation of the coater. On the basis of this information, we conclude that the generations losses associated with Alternative 2 (Final Recommendation) would not adversely affect the longterm operation of the coater because they are similar in magnitude.

Environmental enhancement alternatives that cause production cutbacks at Millinocket would affect both GNP and the regional economy. GNP estimates that curtailing production at the coater complex would eliminate 238 jobs at Millinocket, reduce purchases of goods and services from Maine businesses by \$20 million annually, and jeopardize the company's plans to invest \$550 million in lightweight coated paper production at Millinocket (its least profitable production units) by 1997. Because GNP's data about corporate economic status is proprietary, we could not verify GNP's estimates of the economic effects of production cutbacks; however, our assessment of the paper production industry confirms GNP's contention that its present costs are high and that further cost increases could reduce the company's competitiveness. Multiplier effects from GNP job cutbacks and reduced expenditures for goods and services could result in the total loss of as many as 621 jobs and \$15.0 million in annual wages in the Maine economy. GNP's failure to invest could represent a significant lost opportunity to the region in terms of job and income growth.

The staff finds that there is no conclusive evidence that either Alternative 2 (both variations) or the Applicant's Proposal would adversely affect the competitiveness of the two mills. Both alternatives would produce only small increases in annual power costs, and small declines in annual power production. As a result, a significant increase in the total cost of production would not occur at the two mills. In contrast, Alternative 1 probably would

adversely affect the economic competitiveness of the two mills because of its high capital cost and high energy losses.

5.4 SUMMARY OF FINDINGS AND RECOMMENDATIONS

Sections 4(e) and 10(a)(1) of the FPA require the Commission to give equal consideration to all uses of the waterway on which a project is located. When the Commission reviews a proposed project, the fish and wildlife resources, recreational, and other nondevelopmental values of the involved waterway are considered together with power and other developmental values. In determining whether and under what conditions a hydropower license should be issued, the Commission must balance various economic and environmental considerations.

Based on a review of comments from the agencies and intervenors filed during these proceedings and on our independent analysis pursuant to Sections 4(e), 10(a)(1), and 10(a)(2) of the FPA, we conclude that issuing licenses for the Penobscot Mills and Ripogenus projects, with the required enhancement measures defined in Alternative 2 and other special license conditions, would permit the best comprehensive development of the subject portion of the West Branch of the Penobscot River. We recommend this option for the following reasons:

(1) Alternative 2, which includes some enhancement of flows in Millinocket Stream beyond those proposed by GNP but otherwise specifies flow releases and impoundment- level restrictions identical to those included in the Applicant's Proposal, would substantially enhance existing fish stocks (particularly landlocked salmon) in the project region. Flows beyond those included in Alternative 2 in Upper Gorge and the Back Channel probably would not further enhance regional fish stocks.

(2) The minimal enhancement of regional fish stocks associated with higher flow releases specified in Alternative 1 and considered as options for defining Alternative 2 would not cause significant increases in fisheries resources or recreational angling in the region and, thus, would contribute little to the regional economy.

(3) High-quality recreational fishing waters are abundant in the project region; our estimates indicate that creating a limited amount of additional, year-round, riverine habitat in Upper Gorge, and the Back Channel (Alternative 1), is not required to meet increasing recreational fishing demand, even if the additional habitat enhanced regional fish stocks.

(4) Flows for recreational boating negotiated between GNP and some whitewater boating groups and included in Alternative 2 offer some enhancement of existing recreational opportunities and would not adversely affect existing landlocked salmon populations in the West Branch.

(5) The wetlands enhancements (affecting about 280 acres) included in Alternative 2 are sufficient to enhance the estimated acreage of wetlands that are adversely affected by impoundment draw-downs and would eliminate a long-standing adverse effect of the projects on the regional ecosystem.

(6) The proposed 250-foot conservation easements or 200-foot boundary expansion on GNP-owned lands for the Ripogenus Project area, and the proposed 200-foot boundary expansion for GNP-owned land within the Penobscot Mills Project area (Alternative 2) would control potential future development and ensure protection of aesthetic and recreational resources within the project area. The proposed easements and boundary expansions would also protect against habitat modifications that could adversely affect terrestrial wildlife and vegetation in valuable riparian habitat around the impoundments.

(7) Continued limitation of nongeneration flow releases, particularly in the Back Channel, would ensure GNP's access to relatively inexpensive hydroelectric power necessary to maintain its paper production capability cost-effectively, thereby not placing greater economic stress on GNP and providing some protection for local municipalities and citizens against significant adverse effects on regional employment and socioeconomics.

(8) The fisheries and recreational enhancements that would result from the higher flows associated with Alternative 1 would be limited in magnitude, are not needed to meet recreational demand, and would have little benefit to the local economy. In contrast, the wetlands enhancements, conservation easement and project boundary expansion included in Alternative 2 would permanently improve and protect habitat elements critical to the aesthetics and environment of the project areas, habitats that would otherwise be vulnerable to continued or future degradation. The cost of all enhancements included in Alternative 2 is less than the cost of just the flow releases specified in Alternative 1.

(9) The 500-foot expansion of the project boundary around project impoundments under Alternative 1 would provide resource benefits only marginally greater than the conservation easement and boundary expansion included in Alternative 2, at a much higher cost (estimated at \$24.6 million).

Table 5-4 provides the levelized net annual costs of our recommended Alternative 2.

5.5 STATUTORY REQUIREMENTS

5.5.1 Water Quality Certificate Conditions

On April 15, 1993, Maine LURC issued a Maine Waterway Development and Conservation Act Permit and Water Quality Certification for the Ripogenus Project as proposed by GNP, subject to special conditions (see section 2.2.1.1). Maine DEP issued the certifications for the Millinocket Lake Storage impoundment (a part of the Penobscot Mills Project) and the other Penobscot Mills Project developments, subject to special conditions (see sections 2.2.1.2 and 2.2.1.3), on April 22, 1993. Tables 5-5, 5-6, and 5-7 lists the WQC conditions specified for the Ripogenus and Penobscot Mills projects, the staff's position regarding whether or not the conditions are related to water quality, and indicate the staff's conclusions regarding whether adopting each condition is warranted under various sections of the FPA.^{14,15} The Maine WQC conditions represent the final recommendations of all state agencies. Section 4 presents the detailed analyses upon which the staff's technical conclusions are based.

Table 5-4. Levelized net annual costs of the staff's final recommendation (Source: Staff)			
Project/Mitigation	Levelized Annual Costs (1995\$)	Net Benefits* (mills/kWh)	
Ripogenus Project Maintain relatively stable flows in the West Branch of the Penobscot River below McKay station Minimum flow in Upper Gorge of 100 cfs (Jul 1-Sep 30), leakage (Oct 1-Jun 30) Wetlands enhancements	\$145,300 \$96, 9 00 \$57,700	-7.5 -5.0 -3.0	
Holbrook Stream fisheries enhancement	\$1,700	-0.1	
Penobscot Mills Project Maintain relatively stable water levels in the North Twin impoundment during the summer season Wetlands enhancements Minimum flow in Millinocket Stream of 60 cfs year round Leakage flow in Back Channel	\$224,708 \$3,900 \$34,000 0	-11.5 -0.2 -1.7 0.0	
Hydroelectric operating efficiency losses Steam generation losses	\$200,100 \$195,200	-10.3 -10.0	
TOTAL	\$959,500	-49.3	
* Mills/Kwh based on average annual generation loss of 19.471 GWh.			

For the Ripogenus Project (table 5-5), we will recommend that the Commission not adopt Condition No. 5, consistent with its holding in *Tunbridge Mill*. We will also recommend that the Commission not adopt Condition No. 6. This condition is so broadly worded and vague that we cannot tell what it means or how it could be enforced. Condition No. 8 does not appear to be a condition, but a statement that the WQC for the Ripogenus Project applies only to the Ripogenus Project. However, the proposition is self-evident and we see no potential conflict with the Commission's public interest determination, so we propose to include

5-17

¹⁴ As stated in *Tunbridge Mill Corporation*, 68 FERC ¶ 61,078 (1994), under Section 401(d), states may lawfully impose only conditions related to water quality.

¹⁵ Commission staff is aware of the U.S. Supreme Court's decision in <u>PUD No. 1 of Jefferson County v.</u> <u>Washington Dept. of Ecology</u>, (No. 92-1911, May 31, 1994). The issue of how that decision affects past Commission practice will be addressed in any order issuing a license in this proceeding by the Commission or its delegate.

it in the license. For the Penobscot Mills Project (table 5-7), we will recommend that the Commission not adopt Condition No. 3 because it is not related to operation of the project. We will also recommend that the Commission not adopt Condition No. 10, consistent with its holding in *Tunbridge Mill*. We will also recommend that the Commission not adopt Condition No. 11, This condition is so broadly worded and vague that we cannot tell what it means or how it could be enforced. Other conditions we considered to be unrelated to water quality, as noted in Tables 5-5, 5-6, and 5-7, were considered under 10(a) and will be included as license conditions.

Та	Table 5-5. Summary of the staff's conclusions about Maine 401 Water Quality Certificate conditions for the Ripogenus Project (Source: Staff)			
	WQC Condition	Water Quality Related	Conclusion	
1.	Minimum flows in Upper Gorge and below McKay station	yes	Include	
2.	Monitoring of project flow releases	yes	Include	
3.	Study of mercury concentrations in aquatic life in project waters	yes	Include	
4.	Submit plan for implementing all enhancement measures for LURC approval	yes	Include	
5.	Any variance from project as proposed must be reviewed and approved by LURC	no	Do not include	
6.	Requires compliance with all applicable federal, state, and local licenses, permits, etc.	yes	Do not include	
7.	Effective date	yes	Include	
8.	No element of this WQC shall constitute approval of any operational element of upstream GNP projects	no	Include	

Table 5-6.Summary of the staff's conclusions about Maine 401 Water Quality Certificate
conditions for the Millinocket Lake Storage dam of the Penobscot Mills Project
(Source: Staff)

	WQC Condition	Water Quality Related	Conclusion
1.	Minimum flows in Millinocket Stream	yes	Include
2.	Maintenance of Millinocket Lake elevations	yes	Include
3.	Stock brook trout in Millinocket Stream and monitor fishery	yes	Include
4.	Submit plan for wetlands mitigation and implement	yes	Include
5.	Study impoundment level fluctuation influence on mercury levels in aquatic life	yes	Include
6.	Study need for removal of navigation hazards	no	Include

Tal	Table 5-7.Summary of the staff's conclusions about Maine 401 Water Quality Certificate conditions for the North Twin, Millinocket, Dolby, and East Millinocket developments of the Penobscot Mills Project (Source: Staff)			
	WQC Condition	Water Quality Related	Conclusion	
1.	Run-of-river operation of Millinocket, Dolby, and East Millinocket developments and 2,000 cfs minimum flow to the West Branch at Millinocket	yes	Include	
2.	Maintenance of North Twin draw-down to benefit lake trout	yes	Include	
3.	Study of project effects on Dolby Lake DO levels	yes	Do Not Include	
4.	Study impoundment level fluctuation influence on mercury levels in aquatic life	yes	Include	
5.	Repair North Twin fishway	yes	Include	
6.	Monitor North Twin lake trout spawning success	yes	Include	
7.	Submit plan for and implement wetlands mitigation	yes	Include	
8.	Study need for removal of navigation hazards	no	Include	
9.	Recreational access at various project sites	no	Include	
10.	Limit approval to plans and proposals contained in the application and supporting documents	no	Do No Include	
11.	Requires compliance with all applicable federal, state, and local licenses, permits, etc.	yes	Do Not Include	

5.5.2 Section 18 Prescriptions

Interior reserves authority to prescribe fishways for both projects pursuant to Section 18 of the FPA (letter from J. Deason, Interior, May 24, 1993; letter from A. Raddant, Interior, February 21, 1995).

5.6 SECTION 10(J) RECOMMENDATIONS

5.6.1 Fish and Wildlife Agency Recommendations

Under the provisions of the FPA, as amended by the Electric Consumers Protection Act of 1986, each hydroelectric license issued by the Commission shall include conditions based on recommendations provided by federal and state fish and wildlife agencies for the protection, mitigation, and enhancement of such resources affected by the project.

Section 10(j) of the Act states that whenever the Commission believes that any fish and wildlife agency recommendation is inconsistent with the purposes and the requirements of the Act or other applicable law, the Commission and the agency shall attempt to resolve any such inconsistency, giving due weight to the recommendations, expertise, and statutory responsibilities of such agency.

By executive order of the Governor of the state of Maine, the terms and conditions contained in Maine's 401 Water Quality Certificates represent the state's official recommendations on all issues including fish and wildlife regarding the application, and supersede all preliminary recommendations by individual state agencies. The state did not submit any formal 10(j) recommendations. Thus, in this section we deal only with 10(j) recommendations.

Interior filed initial 10(j) recommendations in response to the REA notice (letters from J. Deason, Interior, May 24, 1993) and submitted revised 10(j) recommendations with their comments on the DEIS (see Appendix E). We determined that eight of the revised recommendations filed by Interior could be inconsistent with the purpose and requirement of the FPA and applicable law. For the Ripogenus Project (numbered as in Table 5-8) these were items 3, 6, and 7; for the Penobscot Mills Project (numbered as in Table 5-9) these were items 1, 2, 3, 6, and 7.

Tal	Table 5-8. Ripogenus Project: Summary of Interior's 10(j) recommendations and their associated costs (Source: the Staff)				
	Recommendation	Annual Cost of Environmental Measures	Within Scope of § 10(j)	Adopted	
1.	Provide minimum flow of 100 cfs at Ripogenus Dam from 7/1 to 9/30	None; proposed by applicant	Yes	Adopt	
2.	Make habitat improvements at Holbrook site	None; proposed by applicant	Yes	Adopt	
3.	Provide minimum flows at McKay Station of 400 cfs outage, 1422 cfs or inflow 10/15 to 6/7 and 711 cfs or inflow 6/8 to 10/14	Moderate	Yes	Adopt	
4.	File plan for complying with instream flow requirements	Minor	Yes	Adopt	
5.	Monitor recreational use	Minor	Νο	Adopt (under 10(a)	
6.	Develop a plan for monitoring contaminants	Moderate	Yes	Adopt	
7.	Develop a plan for monitoring effectiveness of enhancement measures	Moderate	Yes	Adopt	

Under Section 10(j)(2) of the Act, whenever the Commission believes that any recommendations of Federal and state fish and wildlife agencies may be inconsistent with the FPA or other applicable law, the Commission shall attempt to resolve such inconsistencies.

Τa	Table 5-9.Penobscot Mills Project: Summary of Interior's 10(j) recommendations and their associated costs (Source: the Staff)				
	Recommendation	Annual Cost of Environmental Measures	Within Scope of § 10(j)	Adopted	
1.	Provide a minimum flow of 945 cfs or inflow at Stone Dam	\$2.53M	Yes	Not adopt; not cost- effective relative to environmental benefits under 4(e) and 10(a)	
2.	Provide a minimum flow of 120 cfs or inflow to Millinocket Stream between 10/15 and 6/7 and 60 cfs or inflow between 6/8 and 10/14	\$77,000	Yes	Not adopt; not biologically justified; year-round flow of 60 cfs agreed to by Interior and FERC staff at 10(j) meeting	
3.	Operate Millinocket, Dolby and East Millinocket developments in a run-of- river mode	Minor	Yes	Adopt	
4.	File plan for complying with instream flow requirements	Minor	Yes	Adopt	
5.	Monitor recreational use	Minor	No	Adopt (under 10(a)	
6.	Develop a plan for monitoring contaminants	Moderate	Yes	Adopt	
7.	Develop a plan for monitoring effectiveness of enhancement measures	Moderate	Yes	Adopt	

5.6.2 Preliminary Determination of Inconsistency

By letter dated May 1, 1995, we informed Interior of the inconsistency and requested that they consider other options that would adequately protect fish and wildlife consistent with other project purposes. The staff requested that Interior submit these options or additional evidence to support their recommendations to the Commission within 45 days of the date of our letter.

Interior responded by letter dated June 12, 1995. For the Ripogenus Project, Interior stated that they: maintained their recommendation for minimum flows below McKay Station (item 3), agreed that contaminant monitoring would be limited to mercury (item 6), and wished to further discuss monitoring of enhancements (item 7). For the Penobscot Mills Project, Interior stated that they: maintained their recommendation on minimum flows below Stone Dam to the Back Channel (item 1), maintained their position on minimum flows for Millinocket Stream (item 2), agreed to modified run-of-river operation at the Millinocket, Dolby, and East Millinocket developments, although requesting to review proposed language for the license articles(s) dealing with this recommendation (item 3), recommended contaminant monitoring

include mercury, zinc, and copper (item 6), and wished to further discuss monitoring of enhancements (item 7). Interior offered no alternatives or new information in support of their position.

5.6.3 10(j) Meeting

The staff held a 10(j) meeting on February 8, 1996 in a further attempt to resolve these issues. The staff discussed minimum flows below McKay station and the apparent inconsistencies between Interior's 1422 cfs or inflow from October 15 and June 7 and GNP's and the state of Maine's requirement of 1300 cfs during this time period. Both GNP and the state of Maine were concerned about the ability to provide this flow under extreme hydrologic conditions. The staff resolved the issue by indicating the license article for this condition would specify the 1422 cfs for this time period but would allow deviations under extenuating circumstances (low or high flow periods) following consultations between the licensee, state agencies, and the Fish and Wildlife Service. Flows less than 1300 cfs will not be allowed, as required by the 401 WQC.

The staff discussed the issue of contaminant monitoring at both projects . The 401 WQC condition 3 for the Ripogenus Project requires GNP to conduct a study of mercury concentrations on aquatic life at the project and to cooperate in studies on toxic metals in project waters. The 401 WQC condition 5 for the Millinocket Lake Storage Dam and condition 4 for the Penobscot Mills Project also requires GNP to cooperate in studies on toxic metals in project waters. We will recommend that the license include a condition requiring GNP to cooperate on toxic metal studies. The staff stated that effectiveness monitoring for both projects would be limited to those enhancement measures recommended by staff and required in the license. GNP would be required to develop effectiveness monitoring plans in consultation with resource agencies.

As noted above, conditions based on fish and wildlife recommendations submitted pursuant to Section 10(j) must be included in the license unless the Commission determines that the recommendations are inconsistent with the purposes and requirements of the FPA or other applicable law. If the Commission does not adopt a recommendation submitted pursuant to Section 10(j), it must explain, pursuant to Section 10(j)(2), how the recommendation is inconsistent with applicable law and how the conditions selected by the Commission adequately and equitably protect, mitigate damages to, and enhance fish and wildlife. In doing so, we first determine whether the recommendation is supported by substantial evidence in the record, that is, whether there is evidence in the record adequate to support a conclusion. If not, the recommendation is inconsistent with the requirement of Section 313(b) of the FPA that Commission orders be supported by substantial evidence. ¹⁶ Next, we determine whether a substantiated recommendation is inconsistent with the FPA or other applicable determinations under the equal consideration/comprehensive development standards of FPA Sections 4(e) and 10(a)(1), in that the recommendation conflicts unduly with

¹⁶ See IV FERC Statutes and Regulations, supra, 30,921 at p. 30,157.
another project purpose or value (including the project's economic benefits). ¹⁷ In short, we determine whether the recommendation would have a significant, negative impact on a valuable project purpose or beneficial use.

The staff discussed the minimum flow recommendation for the Back Channel. We consider this recommendation to be within the scope of 10(j) but did not adopt it because the magnitude of flow stipulated is unlikely to result in significant fisheries benefits (optimistically, several hundred legal-sized salmon; see section 4.4) at a cost of over \$2.5 million annually (see table 5-10). Thus, we concluded that the resource benefits to be derived are not worth the cost in lost generation and the recommendation is inconsistent with the public interest standards of Section 4(e) and 10(a) of the FPA. Therefore, the staff were not able to resolve the inconsistency regarding the Back Channel flows. Interior responded by letter dated March 25, 1996, stating that they disagreed with the staff's assessment of flows to the Back Channel. However, Interior offered no alternatives or new information in support of their position.

Table 5-10. Relative cost comparison for flows to the Back Channel.									
0 cfs 165 cfs		350 cfs	500 cfs	945 cfs					
0	9900	20800	30700	57231					
\$0	\$0 \$436,100 \$916		\$1,356,000	\$2,521,000					
	0 cfs 0 \$0	Cost comparison for flows 0 cfs 165 cfs 0 9900 \$0 \$436,100	cost comparison for flows to the Back C 0 cfs 165 cfs 350 cfs 0 9900 20800 \$0 \$436,100 \$916,300	cost comparison for flows to the Back Channel. 0 cfs 165 cfs 350 cfs 500 cfs 0 9900 20800 30700 \$0 \$436,100 \$916,300 \$1,356,000					

Does not include any additional hydro efficiency or steam generation losses which could result from additional flow to the BC

The staff discussed the minimum flow recommendation for Millinocket Stream and stated that the IFIM study conducted by GNP indicated a year-round flow of 60 cfs would provide fisheries enhancements beyond those proposed by GNP and the staff in the DEIS. Interior agreed to a year-round flow of 60 cfs or inflow.

We recommend GNP provide the following enhancements at the Ripogenus Project;

- additional flows in the upper gorge;
- additional flows below McKay Station;
- habitat improvements at the Holbrook site;
- plans for monitoring contaminants and effectiveness of enhancement measures;
- wetlands enhancements; and
- conservation easements.

We recommend GNP provide the following enhancements at the Penobscot Mills Project;

plans for monitoring contaminants and effectiveness of enhancement measures;

¹⁷ See Mead Corporation, Publishing Paper Division, 72 61,027 (1995). We also consider whether the application should in fact be denied, on the basis that the resources the project would adversely affect are more valuable than the benefits it would confer.

- wetlands enhancements;
- boundary expansion;
- additional flows in Millinocket Stream to enhance fisheries; and
- lake level management in North Twin to enhance lake trout.

These measures provide resource enhancements throughout both project areas at considerably less cost than providing additional flows to the Back Channel.

Thus, we conclude that our recommendations adequately and equitably enhance fish and wildlife resources affected by the project and that the fish and wildlife measures recommended for the license would comply with the requirements of section 10(j) of the FPA.

5.7 CONSISTENCY WITH COMPREHENSIVE PLANS AND OTHER RESOURCE PLANS

Section 10(a)(2) of the FPA requires the Commission to consider the extent to which a project is consistent with comprehensive plans (where they exist) for improving, developing, or conserving a waterway or waterways affected by a project. Consistency with comprehensive plans is one of several factors considered by the Commission in its licensing decision. Under Section 10(a)(2), federal and state agencies filed five applicable comprehensive plans which are listed and described in section 3.9. We found no conflicts between the projects as they would be licensed under Alternative 2 and all elements of those plans.

6.0 LITERATURE CITED

- ATSDR (Agency for Toxic Substances and Disease Registry). 1988. Toxicological profile for mercury. ATSDR, Atlanta, GA.
- American Rivers, American Whitewater Affiliation, Appalachian Mountain Club, Conservation Law Foundation, Maine Audubon Society, and Trout Unlimited. 1992. Motion to intervene, filed August 21, 1992.
- Atkins, C.G. 1987. The river fisheries of Maine. *In:* The Fisheries and Fishery Industries of the U.S., 1987. G.B. Goode and Associates, eds. pp. 673-728. [As cited in Baum 1983].
- Baum, E.T. 1983. The Penobscot River: An Atlantic salmon river management report. Atlantic Sea-Run Salmon Commission.
- Bangor Real Estate. 1995. Consulting Report, DEIS Cost Estimate Conservation Easements 82.5 Miles of Waterfront Ripogenus Dam Impoundment.
- Barbour, M.G., J.H. Burk, and W.D. Pitts. 1980. Terrestrial plant ecology. The Benjamin/Cummings Publishing Company, Inc., Menlo Park, CA.
- Boulter, D.E. 1993. Commission Ripogenus Project Decision in the Matter of Great Northern Paper, Inc. Findings of Fact and Decision. Water Quality Certification for the Ripogenus Project. Maine Land Use Regulation Commission. April 15, 1993.
- Brown, T.C., J.G. Taylor, and B. Shelby. 1991. Assessing the direct effects of streamflow on recreation: A literature review. *Water Resources Bulletin*. 27(6): 979-989.
- Budd, W.W., P.L. Cohen, and P.R. Saunders. 1987. Stream corridor management in the Pacific Northwest: I. Determination of stream corridor widths. *Environmental Management*. 11(5): 587-597.
- Bureau of the Census. 1980, 1990a. Census of population and housing. Government Printing Office, Washington, DC.
 - _____. 1990b. County Business Patterns. Government Printing Office, Washington, DC.
- COE (Corps of Engineers). 1994. John H. Kerr dam and reservoir shoreline management plan. Department of the Army, Wilmington District Corps of Engineers, Wilmington, ME. February 10, 1994.
- Cutting, R.E. 1959. Penobscot River salmon restoration. Maine Atlantic Sea-Run Commission.
- DEP (Maine Department of Environmental Protection). 1991. Mandatory Shoreland Zoning Act. Title 38, M.R.S.A., Sections 435-449.

_____. 1992. DEP issue profile. Mandatory Shoreland Zoning Act.

_____. 1993a. Natural Resources Protection Act. Title 38, M.R.S.A., Sections 480-A through 480-U.

. 1993b. DEP Issue Profile, Natural Resources Protection Act (NRPA).

DIFW (Maine Department of Inland Fisheries and Wildlife). 1991. Planning for Maine's inland fish and wildlife (1991-1996) Vol. II.

DOC (Maine Department of Conservation). 1982. Maine rivers study.

. 1988. State comprehensive outdoor recreation plan.

____. 1991. Best Management Practices, Field Handbook.

- . 1993. State comprehensive outdoor recreation plan.
- Dahl. T.E. 1990. Wetlands losses in the United States 1780's to 1980's. U.S. Department of the Interior, Fish and Wildlife Service, Washington, DC.
- Dewan & Associates. 1990a. Penobscot Mills Project, FERC No. 2458, Report on visual resources. Prepared for Great Northern Paper, Inc. Millinocket, Maine.
- Dewan & Associates. 1990b. Penobscot Mills Project, FERC No. 2572, Report on visual resources. Prepared for Great Northern Paper, Inc. Millinocket, Maine.
- EPA (Environmental Protection Agency). 1992. Motion to intervene, filed August 20, 1992. Environmental Protection Agency, Region I.

______. 1993. Guidance for assessing chemical contaminant data for use in fish advisories. EPA 823-R-93-002.

East Millinocket. 1993. Draft Comprehensive Plan.

- Elson, P.F. 1975. Atlantic salmon rivers, smolt production and optimal spawning: an overview of natural production. I.A.S.F. Spec. Pub. 6, New England Atlantic Salmon Conference. [as cited in FWS 1989]
- Environmental Science and Engineering. 1992. Mercury in surface water, sediment, and fish from selected lakes of northern Maine. Prepared for Great Northern Paper/Bowter, Millinocket, ME.
- FERC (Federal Energy Regulation Commission). 1993a. Upper Androscoggin River Basin Hydroelectric Project, New Hampshire, Draft Environmental Impact Statement FERC 2422, 2287, 2326, 2327, 2311, 2288, 2300, May 1993.
- _____. 1993b. Transcript of Town of Millinocket Public Hearing, August 26, 1993.
- ______ 1993c. Scoping Document 2 for Penobscot Mills/Ripogenus Projects.

FS (Forest Service). 1990. USDA Forest Service and Governor's Task Force on Northern Lands. Northern forest lands study. Second printing.

______. 1992. Riparian forest buffers, function and design for protection and enhancement of water resources. Department of Agriculture, Northeastern Area Forest Service, Radnor, PA.

FWS (U.S. Fish and Wildlife Service). 1989. Restoration of Atlantic Salmon to New England Rivers. Final Environmental Impact Statement 1989-2021. Department of the Interior, Region 5. U.S. Fish and Wildlife Service, Newton Corner, Massachusetts.

____. 1992. Motion to intervene, filed February 26, 1992.

Fin and Feather Club. 1991. Motion to intervene, filed December 31, 1994.

- Fin and Feather Club. 1993. Settlement Agreement between GNP and Fin and Feather Club, August 19, 1993.
- Flagg, L.N. undated. The American eel: its life history and fishing in Maine. Maine Dept. of Mar. Res., Augusta, ME.
- Frenette, M., C. Rae, and B. Tetreault. 1972. The creation of artificial salmon pools. Department of Civil Engineering, Laval University, Quebec. [as cited in FWS 1989]
- Fry, F.E.J. 1947. Temperature relations of salmonids. In: Proceedings of the 10th Meeting of the National Committee on Fish Culture, 1947. Fisheries Resources Board of Canada. Appendix D. [as cited in FWS 1989]
- GNP (Great Northern Paper, Inc.). 1991a. Ripogenus Project application for new license for major existing dam. FERC No. 2572.
 - _____. 1991b. Penobscot Mills Project application for new license for major existing dam. FERC No. 2458.
 - ______. 1992a. Cultural Resources Management Plan for Archaeological Resources. Penobscot Mills Project, FERC No. 2458. Prepared by Archaeology Research Center, University of Maine at Farmington and Kleinschmidt Associates, Consulting Engineers, August 1992.
 - ______. 1992b. Cultural Resources Management Plan for Archaeological Resources. Ripogenus Mills Project, FERC No. 2572. Prepared by Archaeology Research Center, University of Maine at Farmington and Kleinschmidt Associates, Consulting Engineers, August 1992.

_____. 1992c. Additional information for the application License for the Ripogenus Project, FERC No. 2573 and Penobscot Mills Project, FERC No. 2458. December 7, 1992.

_____. 1992d. Application for new license for Ripogenus Project FERC No. 2572. Vol XIII.

_____. 1993a. Settlement Agreement, Penobscot Mills/Ripogenus between PRO and GNP. September 2, 1993.

_____. 1993b. Application for Relicensing. Ripogenus Project. Vol. XIV: Response to Agency and Intervenor Comments. July 1993.

______. 1993c. Application for Relicensing. Penobscot Mills Project. Vol. XIV: Response to Agency and Intervenor Comments. July 1993.

to October 28, 1993 Additional Information Request. November 1993.

_____. 1994. Supplement to Vol. XV, Penobscot Mills and Ripogenus Projects. Response to Additional Information Request of January 14, 1994.

- Gilmour, C.G., and E.A. Henry. 1991. Mercury methylation in aquatic systems affected by acid deposition. *Environmental Pollution*. 71:131-169.
- Harper, S.C., L.L. Falk, and E.W. Rankin. 1992. The northern forest lands study of New England and New York. Department of Agriculture, Forest Service.
- J. Sewall Co. 1993. Northern forest land conversion study. Prepared by James W. Sewall Company of Old Town, Maine and Market Decision, Inc. of South Portland, Maine for the National Forest Lands Council.
- Jordan, R.M. 1981. Atlantic salmon spawning survey and evaluation of natural spawning success. Final Report, AFS-20-R. [as cited in FWS 1989]
- Joubert, L.B. 1985. Watershed protection strategies for the Sciuate Reservoir. Journal of the American Water Works Association . 99(2): 136-156.
- Knight, A.E. 1981. Unpublished data, Mad River, NH U.S. Fish and wildlife Service, Laconia, NH. [as cited in FWS 1989]
- LURC (Maine Land Use Regulation Commission). 1981. Resource protection plan and recreation management plan for the Penobscot waterway.

_____. 1990a. Fact sheet, land use plan and rule amendments regarding the development and conservation of lakes in Maine's unorganized areas.

_____. 1990b. Amendment of the comprehensive land use plan regarding the development and conservation of lakes in Maine's unorganized area. June 7, 1990.

_____. 1991. Land use districts and standards. Maine Department of Conservation.

_____. 1993. Proceedings of the public hearing for the Ripogenus project: application of Great Northern Paper, Inc. for water quality certification pursuant to Section 401 of the Clean Water Act. Volume 1. February 23, 1993.

- Land & Water Associates. 1993. Location of new development in the wildlands. Prepared for the Maine Land Use Regulation Commission by Land & Water Associates, Hallowell, Maine.
- MSPO (Maine State Planning Office). 1987. Maine Comprehensive Rivers Management Plan. State of Maine, Executive Department, Augusta, Maine. May 1987. 4 Volumes.
- Maine Department of Labor. 1990, 1991, 1992. Maine Employment and Earnings Statistical Handbook.
- Marriott, O. 1993. Department order on the matter of Great Northern Paper, Inc. Water Quality Certification for Penobscot Mills Project and Millinocket Lake Storage Development. April 22, 1993.
- MEPRO. 1993. Settlement Agreement between Great Northern Paper and Maine Professional River Outfitters, Sepember 2, 1993.
- Millinocket. 1992. Draft Comprehensive Plan.

- Mitnick, P. 1991. Penobscot River Basin waste load allocation. Maine Department of Environmental Protection.
- NERC (North American Electric Reliability Council). 1993. Supply and demand 1993-2002. Princeton, New Jersey. June 1993.
- NPCC (Northeast Power Coordinating Council). 1993. Regional Reliability Council long range coordinated bulk power supply programs, OE4411. New York, New York. April 1993.
- Nieswand, G.H., R.M. Hordon, T.B. Shelton, B.B. Chavooshian, and S. Blarr. 1990. Buffer strips to protect water supply reservoirs: A model and recommendations 26(6): 959-966.

PIN (Penobscot Indian Nation). 1992. Motion to intervene, filed 20 August 1992.

- Peterson, R.H. 1978. Physical characteristics of Atlantic salmon spawning gravel in some New Brunswick streams. Fisheries and Marine Service Tech. Rept. 785. [as cited in FWS 1989]
- Peterson, R.H., H.C.E. Spinney, and A. Screedharan. 1977. Development of Atlantic salmon (*Salmo salar*) eggs and alevins under varied temperature regimes. *Journal of the Fisheries Research Board of Canada.* 34:31-34. [as cited in FWS 1989]

Pierce, R.S., J.W. Hornbeck, C.W. Martin, L.M. Tritton, C.T, Smith, C.A. Federer, and H.W. Yawney. 1993. Whole-tree clearcutting in New England: Manger's guide to impacts o soils, streams, and regeneration. U.S. Department of Agriculture, Forest Service, Northeastern Forest Experiment Station. NE-172.

Rolde, N. 1990. Maine, a narrative history. Harpswell Press pp. 243-245.

- SOSLA (Save Our Scenic Lakes Association). 1992. Motion to Intervene, June 30, 1992.
- Sappier, J. 1990. Concluding statement of the Governor, Penobscot Indian Nation. November 13, 1990.

Smith, C.L. 1985. The Inland Fishes of New York State. Albany, NY: New York State Department of Environmental Conservation.

- State of Maine. 1989. Maine Forest Practices Act.
- Stokes, P.M. and C.D. Wren. 1987. Bioaccumulation of mercury by aquatic biota in hydroelectric reservoirs: A review and consideration of mechanisms. *In*: Lead, Mercury, Cadmium, and Arsenic in the Environment, 1987. T.C, Hutchinson and K.M. Meema, eds. John Wiley & Sons Ltd., New York. pp. 255-277.
- Symons, P.E.K. 1979. Estimated escapement of Atlantic salmon (*salmo salar*) for maximum smolt production in rivers of different productivity. *Journal of the Fisheries Research Board of Canada*. 35:175-183. [as cited in FWS 1989]
- Thompson Appraisal Company, Inc. 1995. Appraisal of approximately 2,000 acres of waterfront land with 82.5 miles of frontage on Chesuncook, Ripogenus and Caribou Lakes.
- Tiner, R.W. 1984. Wetlands of the United States: Current status and recent trends. Department of the Interior, U.S Fish and Wildlife Service.
- WHO (World Health Organization). 1990. Environmental health criteria 101: Methyl mercury. World Health Organization, Geneva.
- Warner, K. 1963. Natural spawning success of landlocked salmon, Salmo salar. Transactions of the American Fisheries Society. 92(2):161-164. [as cited in FWS 1989]
- Welch, L. 1991. Contaminant burdens and reproductive rates of bald eagles breeding in Maine. A research proposal submitted to the faculty of the graduate school of the University of Maine at Orono, in partial fulfillment of the requirements for the degree of Master of Science in Wildlife Management.
- Welch, L.J. 1994. Contaminant burdens and reproductive rates of bald eagles breeding in Maine. Master's thesis, University of Maine, Orono.

- Wiemeyer, S.N., T.G. Lamont, C.M. Bunck, C.R. Sindelar, F.J. Gramlich, J.D. Fraser, and M.A. Byrd. 1984. Organochlorine pesticide, polychlorobiphenyl, and mercury residues in bald eagle eggs - 1969-1979 - and their relationship to shell thinning and reproduction. *Archives of Environmental Contamination and Toxicology*. 13 (5): 529-549.
- Wiener, J.G., R.E. Martini, T.B. Sheffy, and G.E. Glass. 1990. Factors influencing mercury concentrations in walleyes in northern Wisconsin lakes. *Transactions of the American Fisheries Society*. 119:862-870.

7.0 LIST OF PREPARERS

7.1 FEDERAL ENERGY REGULATORY COMMISSION STAFF

Edward R. Meyer (B.S., Zoology; M.S., Fisheries Biology; Ph.D., Ecology) - Task Monitor

7.2 VERSAR STAFF

William Richkus -- Fisheries Resources, Deputy Project Manager (B.S., Zoology; M.S., Oceanography; Ph.D., Oceanography)

William Burton -- Water Quality(B.S., Biology; M.S., Ichthyology)

Carol DeLisle -- Technical Writer/Editor (B.A., Biology)

Steven Harriott -- Terrestrial Resources, Endangered Species, Wetlands (B.S., Biology; M.S., Environmental Studies)

Stephen Kartalia -- Fisheries Resources, Water Quality (B.S., Biology; M.S., Fisheries Biology)

Stephen Schreiner -- Geology and Soils, Streamflow (B.A., Biology; M.S., Zoology; Ph.D., Zoology)

Foster Stroup -- Terrestrial Resources, Endangered Species (B.S., Biology)

7.3 ERM STAFF

David Blaha -- Recreation Resources (B.S., Biology; M.S., Environmental Management) Sue Edwards -- Socioecomonics (B.S., Mass Communications; M.R.P., Urban and Regional Planning)

Karen Hardy -- Cultural Resources, Aesthetics, Land Use (B.S., Parks and Recreation; M.R.P, Urban and Regional Planning)

Robert Klosowski -- Facilities Operation (B.S., Electrical Engineering; M.S., Civil Engineering/Water Resources)

Peter Hall -- Economic Comparison of Alternatives (B.S., Geography; M.A., Economic Geography; Ph.D., Civil Engineering, Transportation)

8.0 LIST OF RECIPIENTS

*Ayotte, Allan H. United Paperworkers International Union Local #12 138 Lincoln Street Millinocket, ME 04462

*Banks, John S. Director Penobscot Indian Nation 6 River Road; Indian Island Old Town, ME 04468

*Barash, Mark U.S. Department of the Interior One Gateway Center Suite 612 Newton Corner, MA 02158

*Beaupain, Dean A. Town of Millinocket 4 Hill Street Millinocket, ME 04462

*Blake, Toni J. Katahdin Regional Development Corp. 58 Main Street P.O. Box 449 East Millinocket, ME 04430

*Bouchard, James United Paperworkers International Union Local No. 37 P.O. Box 292 East Millinocket, ME 04430 *Brown, David America Outdoors P.O. Box 1348 Knoxville, TN 37901

*Burrowes, Todd R. Maine Audubon Society P. O. Box 6009 Falmouth, ME 04105

*Callen, Jed Z. Attorney at Law Shedd Road New Boston, NH 03070

*Carson, Jr., Esq., James, B. Great Northern Paper Company Engineering & Research Bldg. One Katahdin Avenue Millinocket, ME 04462

*Connelly, John M. Maine Professional River Outfitters P.O. Box 109 Greenville, ME 04441

*Conte, Anthony R. U. S. Department of the Interior One Gateway Center, Suite 612 Newton Corner, MA 02158-2868

*Deveau, John United Paperworkers International Union Local No. 37 P.O. Box 292 East Millinocket, ME 04430

*Farrington, Robert
The International Brotherhood of Fireman and Oilers
Local #261
HCR 86, Box 195
Medway, ME 04460

*Gauvin, Charles F. Trout Unlimited 800 Follin Lane, Suite 250 Vienna, VA 22180-4959

*Givens, Eric Maine Sporting Camp Association 36 Minuteman Drive Millinocket, ME 04462

*Haines, Vernon Chariman Fin and Feather Club P.O. Box 123 Millinocket, ME 04462

*Hamilton, P. Andrew Eaton, Peabody, Bradford, & Veague P. O. Box 1210 Bangor, ME 04402-1210

*Heath, William United Paperworkers International Union Local No. 37 P.O. Box 292 East Millinocket, ME 04430

*Hill, Richard E. Maine Professional Guides Association P.O. Box 847 Orono, ME 04332-0847 *Huntington, Esq., Matthew H. American Rivers, Inc. Suite 720 1025 Vermont Avenue, N.W. Washington, DC 20005

*Kimball, Kenneth D. Appalachian Mountain Club P. O. Box 298 Gorham, NH 03581

*Kleinschmidt, R. Stevens Green Lake Waterpower Company RR 4, Box 134A Ellsworth, ME 04605

*Lane, Gordon C. Katahdin Regional Development Corporation 58 Main Street, P. O. Box 449 East Millinocket, ME 04430

*LaPointe, Dan Maine Farm Bureau P. O. Box 430 Augusta, ME 04332-0430

*Levasseur, Rick Katahdin Region Tourism Council P.O. Box 59 Millinocket, ME 04462

*Malone, Gail Fisk Rudman & Winchell 84 Harlow Street, P. O. Box 1401 Bangor, ME 04402-4501

*Manfredonia, Esq., Ronald G. U.S. Environmental Protection Agency Water Quality Branch (WQB) John F. Kennedy Federal Building Boston, MA 02203-2211

*Mark, Thomas E. LeBoeuf, Lamb, Leiby, & MacRae 125 West 55th St. New York, NY 10019-4513

*McBreairty, Milton International Brotherhood of Electrical Workers-Local #567 Unit I 240 Old Gray Road Falmouth, ME 04105

*McGillicuddy, Dan International Brotherhood of Carpenters and Joiners Local #658 40 Orchard Street Millinocket, ME 04462

*McTeague, Patrick N. McTeague Higbee, Libner, Macadam, et. al. Four Union Park P.O. Box 5000 Topsham, ME 04086

*Miller, Fred President United Plant Guard Workers of America P. O. Box 65 Brownville, TX 04415 *Minsky, Leonard E. Superior Paper Products 120 Rice Street Bangor, ME 04401

*Nannery, Thomas C. Sportsman's Alliance of Maine P. O. Box 2783 Augusta, ME 04330

*O'Hara, Esq., Kerry U.S. Department of the Interior Office of the Solicitor 1849 C Street, NW Washington, DC 20240

*O'Leary, John A. Great Bay Hydroelectric Associates 76 Thatcher Street Westwood, MA 02090

*Pardilla, Jerry Governor Penobscot Indian Nation 6 River Road: Indian Island Old Town, ME 04468

*Pray, Peter L. Pray's Big Eddy Campground, Inc. P. O. Box 548 Millinocket, ME 04462

*Reed, William United Paperworkers International Union Local No. 37 P.O. Box 292 East Millinocket, ME 04430

*Rouleau, John E. Chairman Town of East Millinocket (ME) 53 Main Street East Millinocket, ME 04430-1199

*Roundy, Charles G. Eastern Maine Development Corp. One Cumberland Place, Suite 300 P.O. Box 2579 Bangor, ME 04402-2579

*Rutherford II, Esq., Floyd R. Paper Industry Information Office 15 Western Avenue Augusta, ME 04330

*Silkman, Richard H. Maine State Planning Office State House Station #38 184 State Street Augusta, ME 04333

*Smart, Michael D. President International Association of Plumbers and Pipefitters 344 Katahdin Avenue Millinocket, ME 04462

*Sosland, Daniel L. Conservation Law Foundation, Inc. 119 Tillson Avenue Rockland, ME 04841-3632 *Tanous, Esq., Wakine G. Tanous & Snow P. O. Box 246 29 Main Street East Millinocket, ME 04430

*Tapley, George United Paperworkers International Union Local #37 P. O. Box 292 East Millinocket, ME 04430

*Tuckerman, Steven Maine FLOW 208 Broadway Bangor, ME 04401

*Veyhl, Erich Maine Conservation Rights Institute 24 Orchard Road Concord, MA 01742

*Voight, Robert President Maine Conservation Rights Institute P. O. Box 220 Lubec, ME 04652

Albert, Jr., Willie 38 Central St. Millinocket, ME 04462

Allender, David SHS #2 Augusta, ME 04333

Intervenor

8-4

Andrews, Thomas H. U.S. House of Representatives Washington, DC 20515-3223

Assistant Director U.S. Environmental Protection Agency (MA) Water Quality Branch (WQB) John F. Kennedy Federal Building Boston, MA 02203-2211

Assistant Secretary Department of Commerce National Oceanic & Atmospheric Adm. Ecology & Conservation Office 14th & Constitution Ave. N.W. Rm. 6222 Washington, DC 20230

Assistant Secretary Dept. of the Interior Main Interior Building, MS 2340 1849 C Street, NW Washington, DC 20240

Assistant Secretary U.S. Department of the Interior Office of Environmental Project Review 1849 C Street Washington, DC 20240

Athanassiadis, Kiki Paul F. Wilkinson & Associates, Inc. 5800, Avenue Monkland Montreal, Quebec, Canada CN H4A1G-1

Attorney General Maine Office of the Attorney General State House Station 6 August, ME 04330 Awaye, William 84 E. Terrace Millinocket, ME 04462

Ayoob, William Millinocket, Town of 207 Penobscot Avenue P.O. Box 959 Millinocket, ME 04462-0959

Bacon, Erwin Katahdin Region Tourism Council P.O. Box 59 Millinocket, ME 04462

Baretey, Jayne H. 105 Canyon Dr. Millinocket, ME 04462

Barker, John 94 New Hampshire St. Millinocket, ME 04462

Bearor, Esq., Edmond J. Rudman & Winchell 84 Harlow Street P.O. Box 1401 Bangor, ME 04401

Bernier, Robert 60 Canyon Drive Millinocket, ME 04462

Biologist Maine Department of Inland Fisheries & Wildlife P.O. Box 551 Greenville, ME 04441

Blake, Frank Katahdin Regional Development Corp. 58 Main Street P.O. Box 449 East Millinocket, ME 04430

Blood, Jim P.O. Box 959 Millinocket, ME 04462

Board of Selectmen Medway, Town of School Street HCR 86 Box 320 Medway, ME 04460-9512

Boland, John Regional Fisheries Biologist RR#1, 328 Shaker Road Gray, ME 04039

Bowers, Richard J. American Whitewater Affiliation 1430 Fenwick Lane Silver Spring, MD 20910

Boynton, Jr., Harold S. 87 Knox St. Millinocket, ME 04462

Bragdon, Bruce Bragdon Bus Service 86 Pumola Millinocket, ME 04462 Bryan, Rod Director Maine Audubon Society Gilsland Farm -- 118 U.s. Route One P.O. Box 6009 Falmouth, ME 04105-6009

Cables, Rich White Mountain National Forest U.S. Forest Service 719 North Maine Laconia, NH 03246

Campbell, Ray A. Fin & Feather Club 15 Delaware Place Millinocket, ME 04462

Carlisle, David Prentiss & Carlisle P.O. Box 637 Bangor, ME 04402-637

Chairman Board of County Comm. County of Piscataquis 51 East Main Street Dover-Foxcroft, ME 04426-1395

Chairman Penobscot, County of Board of County Commissioners 97 Hammond Street Bangor, ME 04401

Chairman St. Lawrence-Eastern Ontario Comm. (NY) 317 Washington St. Watertown, NY 13601

Chairman State of New York New York Fish & Wildlie Management BD. 50 Wolf Road, Room 524 Albany, NY 12233-4750

Chief Department of Agriculture U.S. Forest Service 201 14th Street, S.W. P.O. Box 96090 Washington, DC 20013-6090

Chief Department of the Army U.S. Army Corps of Engineers Office of Chief of Engineers 20 Massachusetts Ave., N.W. Washington, DC 20314-1000

Chief National Oceanic & Atmospheric Adm. Ecology & Conservation Office 14th & Constitution Ave., N.W. Rm.6222 Washington, DC 20230

Chief U.S. Army Corps of Engineers New York District 26 Federal Plaza New York, NY 10007 Chief U.S. Bureau of Land Management 310 West Michigan Avenue, Suite 225 P.O. Box 631 Milwaukee, WI 53201-0631

Chief U.S. Department of Agriculture P.O. Box 2890 Washington, DC 20013

Clark, Herb 180 Bowdon St. Millinocket, ME 04462

Clement, Jay U.S. Army Corps of Engineers Kennebec Market Place RR 2, Box 1855 Manchester, ME 04351

Commandant U.S. Department of Transportation U.S. Coast Guard (G-MEP-3) Water Resources Coord., 2100 2nd St, SW Washington, DC 20593-0001

Commander U.S. Army Corps of Engineers (NY) Buffalok District 1776 Niagara Street Buffalo, NY 14207

Commander U.S. Army Corps of Engineers (NY) North Atlantic Division 90 Church Street New York, NY 10007 Commissioner Maine Dept. of Conservation State House Station 22 August, ME 04333

Commissioner Maine Dept. of Environmental Protection Ray Building - Hospital Street Station House Station #17 Augusta, ME 04333

Commissioner Maine Department of Inland Fisheries & Wildlife 284 State Street Station #41 Augusta, ME 04333

Commissioner Maine Department of Marine Resources State House Station 21 Augusta, ME 04333

Commissioner New York Office of Parks, Recreation & Historic Preservation 2 Empire State Plaza Albany, NY 12223-1253

Commissioner State of Maine Maine Dept. of Agri, Food, & Rural Res State House Station #28 Augusta, ME 04333 Commissioner State of Maine Maine Department of Marine Resources 98 Winthrop Street Hallowell, ME 04347

Commissioner State of New York New York Office of Environ. Conservation Office of General Counsel 50 Wolf Road Albany, NY 12233-0001

Concord Field Office U.S. Fish & Wildlife Service 400 Ralph Pill Marketplace 22 Bridge Street Concord, NH 03301-4901

Coordinator U.S. Department of Transportation U.S. Coast Guard (G-MEP-3) 2100 2nd St, SW Washington, DC 20593-0001

Corin, Leonard P. Field Supervisor U.S. Fish & Wildlife Service (NY) 3817 Luker Road Cortland, NY 13045-9385

Cormico, Joseph Y. Box 7520, Rt. 1 Oakland, ME 04963

County Clerk Concord Township P.O. Box 652 Bingham, ME 04920 A state of the second s

County Commissioners Piscataquis, County of 51 East Main street Dover-Foxcroft, ME 04426-1395

County Commissioners

Crawford, Roland Timberland Trucking Co., Inc. P.O. Box 38 Wytopitlock, MC 04497

Crowfey, Eddie P.O. Box 341 Millinocket, ME 04462

Cyr, Gerard S. 843 Central St. Millinocket, ME 04462

Cyr, Greg Maine Forest Products Council 146 State Street Augusta, ME 04401

Dalgle, Terry 54 New Hampshire St. Millinocket, ME 04462

Dept. of the Interior U.S. Department of the Interior Office of Environmental Project Review 1849 C St. Washington, DC 20240 Dicentes, John International Assn of Machinists 64 Congress Street Millinocket, ME 04462

Director Maine Bureau of Land Quality Control Department of Environmental Protection State House Station 17 Augusta, ME 04333

Director Maine Bureau of Parks & Recreation Department of Conservation State House Station 22 Augusta, ME 04333

Director Maine Department of Marine Resources State House Station 21 Augusta, ME 04333

Director Maine Emergency Management Agency State House Station 72 Augusta, ME 04330-0072

Director Maine Historic Preservation Commission 55 Capitol Street State House Station 65 Augusta, ME 04330

Director Maine Land Use Regulatory Commission State House, Station #22 Augusta, ME 04333 Director Maine State Extension Service 102 Libby Hall University of Maine Orono, ME 04469

Director Natural Resource Consulting Services 31-A Fayette Street Concord, NH 03301-3712

Director New York Dept. of Law Environ. Protection - Science Sect. Justice Bldg. - The Capital Albany, NY 12224

Director New York Dept. of Transportation Syracuse State Office Building 333 E. Washington Street Syracuse, NY 13202

Director New York Dept. of Transportation Region 4 1530 Jefferson Road Rochester, NY 14623

Director New York Dept. of Environ. Conservation 50 Wolf Road Albany, NY 12233-0001

Director Penobscot Valley Council of Governments 1 Cumberland Place, Suite 300 P.O. Box 2579 Bangor, ME 04401 Director State of New York New York Dept. of Health Tower Building Empire State Plaza Albany, NY 12237

Director State of Maine Maine Office of Energy Resources State House Station #53 Augusta, ME 04333

Director State of New York Marine Sciences Research Center (NY) State University of New York Stony Brook, NY 11794

Director State of New York New York Dept. of Public Service Office of Energy Conservation and Envir. Planning 3 Empire State Plaza Albany, NY 12223

Director U.S. Department of Transportation Control & Planning Division 1220 Washington Avenue Albany, NY 12232

Director U.S. Department of Transportation (DC) U.S. Coast Guard (G-MEP-3) 2100 2nd St., SW Washington, DC 20593-0001 Director

U.S. Environmental Protection Agency Off. of Govt. Relations & Envir.-RGR2203 John F. Kennedy Federal Building Boston, MA 02203

Director

U.S. Environmental Protection Agency Water Quality Branch (WQB) John F. Kennedy Federal Building Boston, MA 02203-2211

Director U.S. Army Corps of Engineers New England Division 424 Trapelo Road Waltham, MA 02154-9194

Director U.S. Army Corps of Engineers (IL) North Central Ofice 111 N. Canal Street, Suite 1411 Chicago, IL 60606-7205

Director U.S. Bureau of Land Management U.S. Department of the Interior 1849 C Street, N.W. Washington, DC 20240

Director U.S. Department of Agriculture P.O. Box 2890 Washington, DC 20013 Director U.S. Department of Energy (NY) Room 3206 26 Federal Plaza New York, NY 10007

Director U.S. Department of Energy One Congress Street, Room 1101 Boston, MA 02114-2021

Director U.S. Environmental Protection Agency Water Quality Branch John F. Kennedy Federal Building Boston, MA 02203-2211

Director U.S. Fish and Wildlife Service (NY) 3817 Luker Road Cortland, NY 13045

Director U.S. Geological Survey Mail Stop 405 423 National Center Reston, VA 22092

Director Vermont State Historic Preservation Off. 135 State Street, 4th Floor Drawer 33 Montpelier, VT 05633-1201

District Chief U.S. Geological Survey 26 Ganneston Drive Augusta, ME 04330 Dysart, Edward R. Dysart's Service Inc. 124 Manning Mill Rd. Hampden, ME 04444

Elwell, Reginald Dead River P.O. Box 1427 Bangor, ME 04402-1427

Erickson, Anne R. 61 Westwood Ave. Millinocket, ME 04462

Erickson, Sally P.O. Box 581 Millinocket, ME 04462

Executive Director Delaware River Basin Commission (NJ) 25 State Police Drive Post Office Box 7360 West Trenton, NJ 08628

Executive Director Maine Land & Water Resources Council State Planning Office State House Station 38 Augusta, ME 04333

First Selectmen East Millinocket, Town of 34 Beech Street East Millinocket, ME 04438-1199

Fish, Kelsie L. 26 Winter St. Millinocket, ME 04462 Fiske, Bob J.W. Sewall Company P.O. Box 433 Old Town, ME 04468

Fitzgerald, Catheain 66 Somerset St. Millinocket, ME 04462

Flagg, Lewis Maine Department of Marine Resources 98 Winthrop Street Hallowell, ME 04347

Fowler, Albert L. P.O. Box 568 Millinocket, ME 04462

Frachella, John 103 Texas Ave. Bangor, ME 04401

Gagnon, George 130 Lincoln St. Millinocket, ME 04462

Geologist State of New York Cultural Education Center (NY) Geological Survey Albany, NY 12230

Gibb, Ken City of Bangor 73 Harlow Street Bangor, ME 04401 Given, Terry 106 Prospect St. Millinocket, ME 04462

Goding, Jeffrey G.E. Goding R.R. 1, Box 395 Lincoln, ME 04457

Goodger, Tim National Marine Fisheries Service Oxford Laboratory 904 South Morris St. Oxford, MD 21654

Goody, Ray 150 Highland Ave. Millinocket, ME 04462

Gould, Dick HCR 76 Box 260, East Road Greenville, ME 04441

Greenlaw, Art 132 New Jersey St. Millinocket, ME 04462

Hagan, Christopher J. Natural Heritage Institute 114 Sansome Street Ste. 1200 San Francisco, CA 94104

Haines, Keith 154 Kelley M.H.P. Millinocket, ME 04462

Hall, Stephen E. RFO Box 331 Millinocket, ME 04462

Ham, Jack P.O. Box 506 East Millinocket, ME 04430

Hargrove, Robert Chief U.S. Environmental Protection Agency Region 2 290 Broadway, 28th Floor, Room 500 New York, NY 10007-1866

Harley, Bob P.O. Box 1028 Greenville, SC 29602

Haskell, Jim Millinocket Planning Department P.O. Box 959 Millinocket, ME 04462

Hautt, Delmont E. RFD#1 Carmel, ME 04419

Haynes, Herb H.C. Haynes P.O. Box 96 Winn, ME 04495

Higgins, Betsy U.S. Environmental Protection Agency Water Quality Branch (WQB) John F. Kennedy Federal Building Boston, MA 02203-2211 Hinton, Esq., Carol D. Bowater, Inc. 55 East Comperdon Way P.O. Box 1028 Greenville, SC 29602

Honorable Cohen, William S. U.S. Senate Washington, DC 20510

Honorable D'Amato, Alfonse M. U.S. Senate Washington, DC 20510

Honorable Mitchell, George J. U.S. Senate Washington, DC 20510

Honorable Moynihan, Daniel P. U.S. Senate (DC) Washington, DC 20510

Honorable Snowe, Olympia J. U.S. Senate (DC) Washington, DC 20510

Honorable Governor of Maine State House Station 112 Augusta, ME 04330 Hutchings, Leroy K. Bradley Road RFD 2, Box 343 Brewer, ME 04412

Ingerson, Brad 157 Prospect St. Millinocket, ME 04462

Irwin, Bruce R. New York Dept. of Transportation Dulles State Office Bldg. 317 Washington Street Watertown, NY 13601

Janopaul, Mona M. Trout Unlimited (VA) 1500 Wilson Boulevard Suite 310 Arlington, VA 22209

Jermano, John R. New York State Canal Corporation New York St. Thruway Athy-Off. of Canals 200 Southern Blvd. P.O. Box 189 Albany, NY 12201-0189

Johnson, Paul Fish and Wildlife Headquarters P.O. Box 551 Greenville, ME 04441

Johnston, Edward I. Maine Forest Products Council 146 State Street Augusta, ME 04330 Johnstone, Roger SHS #22 Augusta, ME 04333

Kaiser, Bill Appalachian Mountain Club P.O. Box 331 Woolwich, ME 04579-0331

Keene, Elery North Kennebec Regional Planning Comm. 17 Main Street Fairfield, ME 04937-1119

Kolar, Margaret T. U.S. Fish & Wildlife Service Room 400 Arlington square 18th & C Streets, N.W. Washington, DC 20240

Kotredes, James P.O. Box 959 Millinocket, ME 04462

Lander, Philip Central Equipment Co. P.O. Box 261 Stillwater, ME 04489

Lawless, William F. U.S. Army Corps of Engineers New England Division 424 Trapelo Road Waltham, MA 02154-9194 Leader Maine State Extension Service . 102 Libby Hall University of Maine Orono, ME 04469

Leader State of New York New York Dept. of Natural Resources New York Cooperative Fish and Wildlife Research Unit, Cornell University Ithaca, NY 14853

Leavitt, David W. Cianbro RR3 Box 696 Newport, ME 04953

Leonard, Richard Central Equipment Co. P.O. Box 261 Stillwater, ME 04489

Lewis, Fred Millinocket Foundry & Machine Co. P.O. Box 120 Millinocket, ME 04462

Libby, John W. Director Maine Emergency Mgmt Agency State House Station 72 Augusta, ME 04333-0072

Limatainen, Gary 13 Davis St. Pittsfield, ME 04967 Ludwig, Michael National Marine Fisheries Service 212 Rogers Avenue Milford, CT 06460

Lujan, Manuel U.S. Dept. of the Interior Office of Environmental Affairs 1849 C Street, N.W. - Room 2353 Washington, DC 20240

MacArthur, Sandra J. 108 Highland Ave. Millinocket, ME 04462

MacDonald, Michael 134 Minuteman Dr. Millinocket, ME 04462

Manager U.S. Bureau of Land Management 310 West Michigan Avenue, Suite 225 P.O. Box 631 Milwaukee, WI 53201-0631

Marita, Floyd J. U.S. Forest Service Suite 500 310 West Wisconsin Avenue Milwaukee, WI 53203

Martin, Jeffrey M. Manager Great Northern Paper, Inc. Engineering and Research Bldg. One Katahdin Ave. Millinocket, ME 04462 Martin, Jr., George 112 Morgan Lane Millinocket, ME 04462

Mason, Steve Steven O. Mason, Inc. P.O. Box 201 Greenville, ME 04441

McDevitt, John H. President Save Our Scenic Lakes Association 5 Pleasant Street Burlington, MA 01803

McFalls, Michael W. Haskell Lumber, Inc. P.O. Box 236 Lincoln Center, ME 04458

McHatten, Matthew R. Key Bank of Maine 187 State St. P.O. Box 847 Presque Isle, ME 04769

McLaughlin, Rick 333 Mass Ave. Millinocket, ME 04462

Meagher, Don Penobscot Valley Council of Governments 1 Cumberland Place, Suite 300 P.O. Box 2579 Bangor, ME 04401 Mendik, Kevin U.S. National Park Service North Atlantic Region 15 State Street Boston, MA 02109

Merrill, Paul Merrill Industries P.O. Box 739 Portland, ME 04104

Michoud, Michael 111 Main St. East Millinocket, ME

Mike

Hardwood Products P.O. Box 149 Guilford, ME 04443

Mingo, James A. 23 Forest Ave. Millinocket, ME 04462

Montagna, Michael Maine State Planning Office 184 State St. Augusta, ME 04333

Morrison, Thomas Maine Department of Conservation State House Station 22 Augusta, ME 04333

Morrow, Darrell 65 Waldo St. Millinocket, ME 04462 Nelson, Eric National Marine Fishereis Service Northeast Regional Office - DOC/NOAA 1 Blackburn Drive Gloucester, MA 01930-2237

Normandeau, Donald New Hampshire Dept. of Fish & Game 2 Hazen Drive Concord, NH 03301

Office of Project Review Advisory Council on Historic Pres. The Old Post Office Building 1100 Pennsylvania Avenue, N.W., Ste. 809 Washington, DC 20004

Office of the Secretary U.S. Department of the Interior Office of the Secretary/Environ. Affairs 408 Atlantic Avenue, Room 142 Boston, MA 02210-3334

Officer Regional Environmental U.S. Dept. of the Interior (MA) Office of the Secretary/Environ. Affairs 408 Atlantic Ave., Suite 142 Boston, MA 02210-2209

Oliver, Diane & Doug AWA P.O. Box 1198 E. Holden, ME 04424

Olsen, Bill Casco Northern Bangor, ME 04401

8-17

Ouellette, David P. P.O. Box 10 Fort Cent Mills, ME

Outcalt, Jr., George V. Adirondack Park Agency P.O. Box 99 Ray Brook, NY 12977

Overlock, Charlotte A. Katahdim Times 34 Orchard Street Millinocket, ME 04462

Overlock, Warren Dead River Company 795 Main St. Millinocket, ME 04462

Patterson, William P. U.S. Department of the Interior Office of the Secretary 408 Atlantic Avenue, Suite 142 Boston, MA 02210-2209

Pelletier, Beverly Penobscot Medical Associates 899 Central St. Millinocket, ME 04462

Perroz, Roger Carpenter International 1115 Lisbon Street Lewiston, ME 04240

Perry, George 57 Colony Pl. Millinocket, ME 04462 Perry, Stephen C. United Brotherhood of Carpenters P.O. Box 402 Raymond, ME 04071

Peters, George 36 New Hampshire St. Millinocket, ME 04462

Peters, Nadia and Mark A. 428 Katahdin Ave. Millinocket, ME 04462

Phillips, Daniel 235 Forest Ave. Orono, ME 04473

Pierce, Leonard R.R.2, Box 401 Carmel, ME 04419

Plourde, Kevin P.O. Box 33 Lincoln, ME 04457

Pobert, Bruce Sprague Energy P.O. Box 495 Searsport, ME 04974

Polstein, Esq., Matthew Maine Professional River Outfitters P.O. Box 21 Caratunk, ME 04925 Pottle, Barry Pottle's P.O. Box 877 Bangor, ME 04401

President Great Northern Nekoosa Corporation 401 Merritt Seven Norwalk, CT 06856

Raddant, Andrew L. U.S. Dept. of the Interior (MA) Office of the Secretary/ Environ. Affiars 408 Atlantic Avenue, Suite 142 Boston, MA 02210-2209

Radel, Bill 13 Main St. East Millinocket, ME

Rains, Michael T. U.S. Department of Agriculture State & Private Forestry 100 Matson Ford Road - Suite 200 Radnor, PA 19087

Regional Administrator U.S. Environmental Protection Agency Region 2 290 Broadway, 28th Floor New York, NY 10007-1866

Regional Director Federal Energy Regulatory Comm. Suite 400 19 West 34th Street New York, NY 10001 Regional Director National Marine Fisheries Service (MA) Northeast Regional Office -- DOC/NOAA 1 Blackburn Drive Gloucester, MA 01930-2237

.

Regional Director National Park Service (PA) 143 South Third Street. Philadelphia, PA 19106

Regional Director U.S. National Park Service North Atlantic Region 15 State Street Boston, MA 02109

Regional Director U.S. Department of the Interior Bureau of Land Management Region 7450 Boston Boulevard Springfield, VA 22153

Regional Director U.S. Fish & Wildlife Service 300 Westgate Ctr Road Hadley, MA 01035-9589

Regional Officer National Park Service, (PA) 143 South Third Street Philadelphia, PA 19106

Reider, Mike 150 Lincoln St. Millinocket, ME 04462 Richert, Evan D. Director Maine State Planning Office 184 State Street State House Station 38 Augusta, ME 04333

Ricker, Lynn Oscar & Reuben Lumbra Box 337 Milo, ME 04463

Roach, Joseph Asset 200 Somerset Street Millinocket, ME 04462

Robinson, Michael P.O. Box 30 Sherman Station, ME 04777

Russell, Cheryl Harrington Bros. HCR 62 Box 53 Kingman, ME 04451

Russell, Gordon U.S. Fish & Wildlife Service 1033 South Main Street Old Town, ME 04468

Ruta, Gwen U.S. EPA JFK Federal Bldg. Boston, MA 02203 Sanders, Charles J. 75 Cottage Rd. Millinocket, ME 04462

Sasseville, Dennis ESE, Inc. 5 Overlock Dr. Amherst, NH 03031

Savage, David Maine Building Trades Council 20 Ayer Street Brewer, ME 04412

Scain, Leonard Bowater Damien, CT

Scarano, Theresa and Wayne 43 Katahdin Ave. Millinocket, ME 04462

Schneider, Richard W. H.O. Bouchard 67 Mass. Ave. Millinocket, ME 04462

Secretary Maine Public Utilities Commission 242 State Street State House Station 18 Augusta, ME 04333-0018

Sherwood, Ted Bangor Real Estate Bangor, ME 04401 Shettleworth, Earle G. Comissioner Maine Historic Preservation Commission 55 Capitol Street State House Station 65 Augusta, ME 04330

Shields, Donald V. 45 Grant Street Bangor, ME 04401

Sidoti, Anton J. Federal Energy Regulatory Comm. Suite 400 19 West 34th Street New York, NY 10001

Sims, Andrew E. Kleinschmidt Associates Consulting Engineers P.O. Box 576 Pittsfield, ME 04967

Sinclair, Esq., Mark A. Conservation Law Foundation (VT) Suite 301 Montpelier, VT 05602

Smart, Thomas 46 Aroostook Ave. Millinocket, ME 04462

Smith, Deborah S. Conservation Law Foundation 62 Summer Street Boston, MA 02110 Smith, Michael Maine Department of Inland Fisheries & Wildlife - Regional Biologist Region F - Box 66 Enfield, ME 04433

Smith, Tim 51 New Hampshire Millinocket, ME 04462

Snitzler, Vicky U.S. Environmental Protection Agency Region 2 290 Broadway, 28th Floor New York, NY 10007-1866

Spitzer, William T. Chief U.S. National Park Service Recreation Resources Assistance Div. P.O. Box 37127 Washington, DC 20013-7127

St. Cyr, Gerard 8 Field St. Millinocket, ME 04462

St. Mary, Ken U.S. Fish & Wildlife Service Room 400 Arlington Square 18th & C Streets, N.W. Washington, DC 20240

State Director U.S. Bureau of Land Mgmt. Oregon State Office 1515 S.W. 5th Ave. P.O. Box 2965 Portland, OR 97208-2965 State of Maine Maine Department of Conservation State House Station 22 Augusta, ME 04333

State of Maine Maine Dept. of Agri, Food, & Rural Res. Augusta, ME 04333

State of Maine Maine Department of Inland Fisheries & Wildlife 284 State Street Station #41 Augusta, ME 04333-0041

State of Maine Maine Statement Soil & Water Conserv'n Comm. Deering Bldg., AMHI Complex Station #28 Augusta, ME 04333

State of New York New York Dept. of Transportation State Capus 1120 Washington Avenue Albany, NY 12232

State of New York New York Dept. of Transportation Division of Canals - Sate Campus 1220 Washington Avenue Albany, NY 12232 Steadman, Eric C. Sociotechnical Research Appls., Inc. Energy & Environment Division 1100 Wilson Blvd., Suite 1700 Arlington, VA 22209

Stetson, Brian Great Northern Paper, Inc. (ME) Engineering and Research Bldg. One Katahdin Ave. Millinocket, ME 04462

Stevens, Craig 28 Maine Ave. Millinocket, ME 04462

Stockinger, Paul Lands, Watershed & Minerals Regional Forester, Mgt. Staff, Room 500 Federal Plaza, 310 West Wisconsin Ave. Milwaukee, WI 53203

Stolfo, Esq., Judith M. U.S. Dept. of the Interior (MA) Office of the Solicitor One Gateway Center - Suite 612 Newton Corner, MA 02158

Ten-Broeck, Craig Maine Emergency Management Agency State House Station 72 Augusta, ME 04330-0072

Thompson, Robert Androscoggin Valley Council of Government 125 Manley Road Auburn, ME 04210 Timpano, Steven Maine Department of Inland Fisheries and Wildlife 284 State Street State House Station #41 Augusta, ME 04333-0041

Town Clerk Brownville, Town of Town Hall Brownville, ME 04414

Town Clerk Town of Millinocket, 207 Penobscot Avenue P.O. Box 959 Millinocket, ME 04462-0959

Town Manager Millinocket, Town of 207 Penobscot Avenue P.O. Box 959 Millinocket, ME 04462-0959

Town Official Town of Colonie (NY) Colonie Memorial Town Hall Newtonville, NY 12128

Town Supervisors Sullivant County Government Hartwood 100 N. Street Monticello, NY 12701 Towsend, H. Anderson Shp Officer 135 State Street, 4th Floor Drawer 33 Montpelier, VT 05633-1201

Turin, David U.S. Environmental Protection Agency Water Quality Branch (WQB) John F. Kennedy Federal Bldg. Boston, MA 02203-2211

Tyson, John 80 Highland Ave. Millinocket, ME 04462

. .

Valley, Duane E. 52 Water St. Millinocket, ME 04462

Vamos, Ivan P. New York Office of Parks & Recreation & Historic Preservation Empire State Plaza Albany, NY 12223

Varney, Robert New Hampshire Dept. of Envir. Protection 6 Hazen Drive Concord, NH 03301

Vetter, A.S. New York Dept. of Transportation Syracuse State Office Bldg. 333 E. Washington Street Syracuse, NY 13202 Vose, Jim 11 William St. Lincoln, ME 04457

Waldrep, Marvin Georgia-Pacific Corporation P.O. Box 105605 Atlanta, GA 30348

Ward, Stephen G. Maine Office of Public Advocate State House Station 112 Executive Department 7th Floor Augusta, ME 04333

Warren, Henry E. Maine Dept. of Environmental Protection Ray Building - Hospital Street Station House Station #17 Augusta, ME 04333

Waters, Barbara M. 202 Penobscot Ave. Millinocket, ME 04462

Welch, Patrick H. 24 New Hampshire Street Millinocket, ME 04462

Wentworth, Roderick Vermont Department of Fish & Wildlife 103 South Main Street, 10 South Waterbury, VT 05676 Wieting, Donna Office Dir. National Oceanic & Atmospheric Adm. Ecology & Conservation Office - HCHB SP 14th & Constitution Ave. N.W. Rm. 6117 Washington, DC 20230

Willey, Lloyd Canteen P.O. Box 1207 Bangor, ME 04401

Wilson, Pat 66 Somerset St. Millinocket, ME 04462

APPENDIX A

_

____ .__ .__

DAMS IN THE PENOBSCOT RIVER BASIN

Table	A-1. Dams within the Pen	osbscot River Basin					
		· • • • • • • • • • • • • • • • • • • •				FISH PASSAGE STATUS	
MAP ID #	DAM NAME	RIVER	FERC #	STATE ID #	CAPACITY (KW)	TYPE	YEAR CONSTRUCTED
1	Ulmer Brook Dam	Ulmer Brook	NJ	901			
2	Orland Village Dam	Orland River	NJ	710			
З	Alamoosook Dam	Narramissic River	NJ	715			
4	Toddy Pond Dam	Powerhouse Brook	NJ	717			
5	Craig Pond Dam	Craig Brook	NJ	716			
6	Williams Pond Dam	Stubbs Brook	NJ	711			
7	Silver Lake Dam	Silver Lake Outlet	NJ	718			
8	Şearsport Water District	Halfmoon Stream	NJ	719			
9	Frankfort Dam	Marsh Stream	6618	720	550		
10	West Winterport Dam	Marsh Stream	6132	758	150		
11	Town Marsh Dam	N. Br. Marsh Stream	NJ	766	i i		
12	Brooks Dam	Marsh Stream	NJ	3807			
13	Samuel Foss Dam	Marsh Stream	7979	895	15		
14	Ellis Dam	Marsh Stream	NJ	3802			
15	Randall Dam	Marsh Stream	NJ	3803			
16	Sanborn Pond Dam	Marsh Stream	NJ	3806			
17	Swetts Pond Dam	Mill Brook	NJ	725			
18	Souadabscook Stream Dam	Souadabscook Stream	4727	726	200		
19	Temple Mill Dam	Souadabscook Stream	NJ	900			
20	Lovely's Dam	Souadabscook Stream	NJ	907			
21	Mill Street Dam	Sedgeunkedunk Stream	NJ	729			
22	East Orrington Dam	Sedgeunkedunk Stream	NJ	731			
23	Brewer Lake Dam	Sedgeunkedunk Stream	NJ	732			
24	Garland Pond Dam	Kenduskeag Stream	NJ	737		··· •	
Table /	A-1. Continued						
-------------	-----------------------------	---------------------------	--------------	---------------	---------------------	---------------	---------------------
					FISH PASSAGE STATUS		
MAP ID #	DAM NAME	RIVER	FERC #	STATE ID #	CAPACITY (KW)	ТҮРЕ	YEAR CONSTRUCTED
25	Bangor Dam	Penobscot River	P3986	700		Breached	
26	Veazie Dam	Penobscot River	2403	701	8400	Vertical Slot	1970
27	Gilman Mill Dam	Blackman Stream	NJ	742			
28	Orono Water Works Dam	Blackman Stream	NJ	743			
29	Leonards Mills Dam	Blackman Stream	NJ	906			
30	Parks Pond Dam	Parks Pond Brook	NJ	746			
31	Eddington-Holbrooks Dam	Mill Brook	NJ	745			
32	Orono Dam	Stillwater River	2710	702	2332	None	
33	Stillwater Dam	Stillwater River	2712	703	1950	None	1
34	Great Works Stream Dam	Great Works Stream	NJ	747	4680	[
35	Great Works Dam	Penobscot River	2312	705	7655	Denil (2)	1968
36	Milford Dam	Penobscot River	25 34	706	6400	Denil	1968
37	Gilman Falls Dam	Stillwater River	2534	704		None	
38	Pushaw Lake Dam	Pushaw Stream	NJ	749		}	1
39	Hatchery Dam	Cold Stream	NJ	761			
40	Stanhope Mill Dam #2	Cold Stream	NJ	762	1		1
41	Round Pond Dam	Cold Stream	IJ	763	1		
42	Lowell Tannery Dam	Passadumkeag River	4202	760	875		
43	Eskutassis Stream Dam	Eskutassis Stream	NJ	764	1		1
44	Eskutassis Pond Dam	Eskutassis Stream	NJ	765	1		1
45	NO. 3 Pond Outlet	W. Br. Passadumkeag River	NJ	772		1	1
46	Nicatous Stream Dam	Nicatous Stream	NJ	767		1	1
47	Howland Dam	Piscataquis River	2721	773	1800	Denil	1965
48	Old Farm Pond Water Control	Trib. to Seboeis Stream	NJ	888	1	1	
49	Seboeis Lake Dam	W. Br. Seboeis Stream	NJ	778		1	1
50	Schoodic Lake Dam	Schoodic Stream	L NJ	779	1	Ţ_ _	1

Table A	A-1. Continued							
						FISH PASSAGE STATUS		
MAP ID #	DAM NAME	RIVER	FERC	STATE ID #	CAPACITY (KW)	ТҮРЕ	YEAR CONSTRUCTED	
51	Brownville Dam	Pleasant River	P10664	781		Breached		
52	Milo Dam	Sebec River	5647	790	600			
53	Sebec Lake Dam	Sebec River	7253	791	1100			
54	Lake Hebron Dam	Hebron Lake Brook	NJ	798				
55	Lower Dam (Brown's Mill)	Piscataquis River	5613	774	550	Denil	1973	
56	Upper Dam (Moosehead)	Piscataquis River	5912	775	300	Denil	1973	
57	Davee Brook #1	Davee Brook	NJ	903				
58	East Davee Brook Site #2	Davee Brook	NJ	904				
59	Dunham Brook Site #2	Dunham Brook	NJ	887				
60	Branns Mill Dam	Black Stream	NJ	801				
61	Haley Dam	Carlton Stream	NJ	804				
62	Mahanock Pond Dam	Carlton Stream	NJ	805				
63	Gordon's Dam	Gordon's Pond Outlet	NJ	788	1			
64	Pingree Pond Outlet	Pingree Stream	NJ	806		I	1	
65	Guilford Dam	Piscataquis River	P8316	776		Denil	1972	
66	Bennett Pond Dam	Gales Brook	NJ	807				
67	Shirley Pond Dam	E. Br. Piscataquis River	NJ	809]]	
68	Kingsbury Pond Dam	Kingsbury Stream	NJ	808				
69	Piper Pond Dam	Piper Pond Outlet	NJ	782				
70	West Enfield Dam	Penobscot River	2600	707	13000	Vertical Slot	1987	
71	Runaround Dam	Merrill Brook	2600	777	1	None		
72	Roberts Dam	Mattamiscontis Stream	NJ	810		Dam Removed		
73	SO. Branch Lake Dam	South Branch Stream	NJ	813	1	1	1	
74	Mill Pond Dam	Mattanawcook Stream	NJ	814	1		1	
75	Mattanawcook Pond Dam	Mattanawcook Stream	NJ	815	1	1	1	

Table /	A-1. Continued						
						FISH PASS	AGE STATUS
MAP ID #		RIVER	FERC #	STATE	CAPACITY (KW)	TYPE	YEAR CONSTRUCTED
76	Folsom Pond Dam	Mattanawcook Stream	NJ	816			
77	Long Pond Dam	Mattanawcook Stream	NJ	817			
78	Mill Pond Dam	Cambolasse Stream	NJ	819			
79	Stump Pond Dam	Cambolasse Stream	NJ	820			
80	Cambolasse Pond Dam	Cambolasse Stream	NJ	821			
81	Long Pond Dam	Cambolasse Stream	NJ	822			
82	Dwinal Pond Dam	Mattakeunk Stream	NJ	824			
83	Mallett's Mill Dam	Mattakeunk Stream	NJ	825			
84	Village Dam	Mattakeunk Stream	NJ	827			
85	Silver Lake (Mattakeunk Pond)	Mattakeunk Stream	NJ	828			
86	Sherman Lumber Dam	West Branch Molunkus Stream	NJ	860			
87	Danforth Dam	Baskahegan Stream	NJ	831			
88	Dyer Brook Dam	Dyer Brook	NJ	840		1	
89	Rockabema Outlet Dam	W. Br. Mattawamkeag River	NJ	844			
90	Mill Pond Dam	Fish Stream	NJ	841			
91	Peasley Brook Dam	Peasley Brook	NJ	842			
92	Mattaceunk Dam	Penobscot River	2520	708	19200	Pool & Weir	1980
93	Swift Brook Dam	Swift Brook	ЦИ	848			
94	Sawtelle Deadwater Dam	Sawtelle Brook	NJ	852	1	}	
95	Medway Dam	W. Br. Penobscot River	2666A	862	3440	}	1
96	East Millinocket Dam	W. Br. Penobscot River	2458C	863	9600		1
97	Dolby Dam	W. Br. Penobscot River	2458B	864	20988		
98	Millinocket Lake Dam	Millinocket Stream	2458E	868	1	1	
99	Quakish Dam (Stone)	W. Br. Penobscot River	2458A	865	14880		
100	North Twin Dam	W. Br. Penobscot River	2458D	866	9840		1

Table	A-1. Continued		<u> </u>				
						FISH PAS	SAGE STATUS
MAP ID #	DAM NAME	RIVER	FERC #	STATE ID #	CAPACITY (KW)	TYPE	YEAR CONSTRUCTED
101	Abol Pond Dam	Abol Stream	NJ	873			
102	Rainbow Lake Dam	Rainbow Lake Outlet	IJ	872			
103	Sourdnahunk Lake Dam (Nesowadnehunk Dam)	Nesowadnehunk Stream	ЦИ	874			
104	Telos Dam	Webster Brook	NJ	859			
105	Lock Dam	Chamberlain Lake Outlet	NJ	2311			
106	Ripogenus Dam	W. Br. Penobscot River	2572	867	37530		
107	Harrington Lake Dam	Ripogenus Stream	NJ	875			
108	Ragged Lake Dam	Ragged Stream	2634A	876			
109	Umbazooksus Lake Dam	Umbazooksus Stream	NJ	878			
110	Cauoomgomoc Dam	Caucomgomoc Stream	2634B	879			
111	Loon Lake Dam	Loon Stream	NJ	880			
112	Seboornook Dam	W. Br. Penobscot River	2634C	869			
113	Dole Pond Dam	Dole Brook	LN	882			
114	Long Pond Dam	Long Pond Outlet	IJ	883			
115	Canada Falls Dam	S. Br. Penobscot River	2634D	885			
116	Penobscot Lake Dam	Penobscot Brook	NJ	886		••••	

APPENDIX B

-

FISHERIES RESOURCES

,

		Impoundments						
Common Name	Scientific Name	North Twin	Millinocket Lake	Ripo- genus	Quakish Lake	Dolby Pond	East Millinocket	
Land-locked salmon	Salmo salar	xª	x ^a	x	 x	×	x	
Brook trout	Salvelinus fontinalis	x	x	x	0	x	О	
Lake trout (togue)	Salvelinus namaycush	xa	xa	x	x	0	0	
Splake	S. fontialis x S. namaycush	0	ο	0	0	0	о	
Lake whitefish	Coregonus clupeaformis	x	x	x	ο	0	0	
Round whitefish	Prospium cylindraceum	0	ο	0	0	0	ο	
Rainbow smelt	Osmerus mordax	x	x	x	x	x	×	
Burbot (cusk)	Lota lota	x	x	x	0	0	о	
Smallmouth bass	Micropterus dolomieui	o	0	0	О	x	×	
Chain pickerel	Esox niger	x	×	0	x	x	×	
White perch	Morone americana	x	x	x	x	x	x	
Yellow perch	Perca flavescens	x	x	x	x	x	×	
Redbreast sunfish	Lepomis auritus	x	×	0	x	x	×	
Pumpkinseed	Lepomis gibbosus	x	x	0	x	x	x	
Brown bullhead (hornpout)	Ictalurus nebulosus	x	x	x	x	x	x	
American eel	Anguilla rostrata	x	x	x	×	x	x	
White sucker	Catostomus commersoni	x	x	x	x	x	x	
Longnose sucker	Catostomus catostomus	x	x	x	x	x	x	
Fallfish	Semotilus corporalis	x	x	x	x	x	×	
Pearl dace	Semotilus margarita	o	x	0	0	0	0	
Blacknose dace	Rhinichthys atratulus	x	x	x	x	x	x	
Lake chub	Couesius plumbeus	x	x	x	x	x	0	
Common shiner	Notropis cornutus	x	x	x	x	x	×	
Golden shiner	Notemigonus crysoleucas	x	x	x	x	x	x	
Creek chub	Semotilus atromaculatus	x	x	x	x	x	x	
Fathead minnow	Pimephales promelas	x	0	0	0	0	Ö	
Ninespine stickleback	Pungitius pungitius	x	x	o	x	ŏ	ō	
Slimy sculpin	Cottus cognatus	x	×	0	x	x	x	
Banded killifish	Fundulus diaphanus	×	x	0	x	Ŷ	Y Y	

ſ

				Riverin	e Reaches			
Common Name	Scientific Name	W. Branch N. Twin Dam to Quakish L.	W. Branch Millinocket Development Tailrace to Dolby Pond	W. Branch E. Millinocket Development Tailrace	Millinocket Stream	Back Channel	Upper gorge	W. Branch Ripogenus to N. Twir
Landlocked salmon	Salmo salar	x	x	×	x	x	x	x
Brook trout	Salvelinus fontinalis	x	x	o	x ^a	0	x	×
Lake trout (togue)	Salvelinus namaycush	0	o	0	x	0	x	x
Splake	S. fontinalis x S. namaycush	0	0	o	0	ō	0	×
Lake whitefish	Coregonus clupeaformis	о	0	0	0	0	0	x
Round whitefish	Prospium cylindraceum	О	0	0	o	0	0	x
Rainbow smelt	Osmerus mordax	x	x	x	x	0	x	x
Burbot (cusk)	Lota lota	о	o	0	×	0	0	×
Smallmouth bass	Micropterus dolomieui	0	х	x	×	x ^b	0	0
Chain pickerel	Esox niger	x	x	0	×	0	0	×
White perch	Morone americana	x	x	x	×	õ	x	x
Yellow perch	Perca flavescens	x	x	0	x	Ő	x	x
Redbreast sunfish	Lepomis auritus	o	o	0	×	×	0	×
Pumpkinseed	Lepomis gibbosus	0	0	0		0	Ő	0
Brown bullhead (hornpout)	Ictalurus nebulosus	o	x	0	0	0 0	ů	Ň
American eel	Anguilla rostrata	x	x	x	x	ů 0	Y	×
White sucker	Catostomus commersoni	x	x	x	×	Y Y		
Longnose sucker	Catostomus catostomus	x	x	0	x	Ô	Ŷ	
Fallfish	Semotilus corporalis	x	x	Ő	×	õ	Ŷ	
Pearl dace	Semotilus margarita	0	0	0	0	0		
Blacknose dace	Rhinichthys atratulus	×	x	0	×	ů O	×	×
Lake chub	Couesius plumbeus	0	0	Ő	0	ů O		Ŷ
Common shiner	Notropis cornutus	0	0	ů O	v v	×		ÛÛ
Golden shiner	Notemigonus crysoleucas	o i	ő	ů	Ô	â		Ŷ
Creek chub	Semotilus atromaculatus	0	ů O	õ	y l	v v		Î Î
Fathead minnow	Pimephales promelas	x	0 0	õ	Ô	0		
Ninespine stickleback	Punaitius punaitius	x	x	0	0	0		
Slimy sculpin	Cottus cognatus	×	x i	õ		v		Ŷ
Banded killifish	Fundulus diaphanus	0	Ô	õ	â	Ô		

^b Below Grand Falls only

Б

APPENDIX C

WETLANDS AND TERRESTRIAL RESOURCES



Figure C-1. Existing wetlands in the vicinity of Ripogenus impoundment, Ripogenus Project (modified after figure E3.2-2, GNP, 1991a)



Figure C-2. Existing wetlands bordering the West Branch of the Penobscot River, in the vicinity of the Ripogenus Project (modified after figure E3.2-3, sheet 1, GNP, 1991a)



Figure C-3. Existing wetlands bordering the West Branch of the Penobscot River, in the eastern section of the Ripogenus Project (modified after figure E3.2-3, sheet 2, GNP, 1991a)

С-5



Figure C-4. Existing wetlands in the vicinity of North Twin impoundment, Penobscot Mills Project (modified after figure E3.2-2, GNP, 1991b)

C-6



Figure C-5. Existing wetlands in the vicinity of Millinocket Lake, Penobscot Mills Project (modified after figure E3.2-3, GNP, 1991b)

line and a second s



Figure C-6. Existing wetlands in the vicinity of Quakish Lake, Penobscot Mills Project (modified after figure E3.2-4, GNP, 1991b)



and the second second



C-9

in for 2 min

Table 1.	Plants occupying softwood-dom Penobscot Mills projects, Piscat (original data from GNP Vol. X,	ninated mixed forest in the Ripogenus and aquis and Penobscot Counties, Maine 1991).
	Scientific Name	Common Name
	Саг	пору
	Picea rubens Abies balsamea Betula lutea B. papyrifera Fagus grandifolia Quercus borealis Sorbus americana Pinus strobus Thuja occidentalis Fraxinus pennsylvanica var. subintegerrima Populus balsamifera Picea mariana Acer rubrum Ostrya virginiana Picea glauca Acer saccharum	Red spruce Balsam fir Yellow birch Paper birch Beech Northern red oak Mountain ash White pine Northern white cedar Green ash Balsam poplar Black spruce Red maple Hophornbeam White spruce Sugar maple
	Acer pensylvanicum Acer spicatum Corylus cornuta Vaccinium angustifolium Lonicera canadensis Rubus idaeus var. strigosus Viburnum alnifolium Ribes glandulosum Vaccinium vacillans Gaultheria hispidula Amelanchier laevis Diervilla lonicera Rubus sp. Kalmia angustifolia Gaultheria procumbens Nemopanthus mucronata Viburnum lentago Hamamelis virginiana Gaylussacia baccata Salix sp. Ledum groenlandicum	Striped maple Mountain maple Beaked hazel Lowbush blueberry Fly honeysuckle Red raspberry Hobblebush Gooseberry Lowbush blueberry Bristly wintergreen Shadbush Bush honeysuckle Blackberry Sheep laurel Wintergreen Mountain holly Nannyberry Witch hazel Black huckleberry Willow Labrador tea

Table 1.	Continued	
	Scientific Name	Common Name
	Herbace	ous Layer
	Cypripedium acaule	Pink Lady's Slipper
	var. albicaulis	(white form)
	Trientalis borealis	Starflower
	Dryopteris spinulosa	Spinulose wood fern
	Maianthemum canadense	Canada mayflower
	Oxalis montana	Wood sorrel
	Cornus canadensis	Bunchberry
	Trillium erectum	Purple trillium
	Clintonia borealis	Clintonia
	Aralia nudicaulis	Wild sarsaparilla
	Pteridium aquilinum	Bracken fern
	Osmuinda claytoniana	Interrupted fern
	Athyrium filix-femina	Lady fern
	Smilacina racemosa	False Solomon's Seal
	Dryopteris disjuncta	Oak fern
	Streptopus roseus	Twisted stalk
	Polypodium vulgare	Rock polypody
	Linnaea borealis	Twinflower
-	Lycopodium lucidulum	Shining clubmoss
-	Lycopodium clavatum	Trailing clubmoss
	Coptis trifolia	Goldthread
	Actaea rubra	Dolls' Eyes
	Medeola virginiana	Indian cucumber
	Poa languida	Bluegrass
	Polygonatum biflorum	Solomons' Seal
	Chrysosplenium americanum	Golden saxifrage
	Trillium undulatum	Painted trillium
	Mitella nuda	Miterwort
	Mitchella repens	Partridgeberry
	Oryzopsis asperifolia	Mountain rice
	Dryopteris phegopteris	Long beech fern
	Cypripedium acaule	Pink Lady's Slipper
	Carex arctata	Sedae
	Brachyelytrum erectum	Brachvelvtrum
	Onoclea sensibilis	Sensitive fern
	Thalictrum polygamum	Meadow rue
	Osmunda cinnamomea	Cinnamon fern
	Epigaea repens	Trailing arbutus
	Viola sp.	Purple violet
	Sphagnum sp.	Sphaonum moss
	Streptopus amplexifolius	Twisted stalk
	Luzula accuminata	Wood rush
	Equisetum fluviatile	Horsetail

Table 1.	Continued				
	Scientific Name Common Name				
11 - 11 - 11 - 11 - 11 - 11 - 11 - 11	Herbaceous La	ayer (Cont'd)			
	<i>Melampyrum lineare Monotropa uniflora Solidago flexicaulis Dalibarda repens Veronica scutellata</i>	Cow wheat Indian pipe Goldenrod False violet Speedwell			
	Trisetum spicatum Circaea alpina	Three-awned grass Enchanter's nightshade			

Scientific Name	Common Name				
Сапору					
Fagus grandifolia	Beech				
Fraxinum americana	White ash				
Fraxinus pennsylvanica	Green ash				
var. subintegerrima					
Betula lutea	Yellow birch				
Acer saccarum	Sugar maple				
Picea rubens	Red spruce				
Tsuga canadensis	Hemlock				
Abies balsamea	Balsam fir				
Quercus borealis	Northern red oak				
Thuja occidentalis	Northern white cedar				
Ostrya virginiana	Hophornbeam				
Ulmus americana	American elm				
Betula papyrifera	Paper birch				
Pinus strobus	White pine				
P. resinosa	Red pine				
Populus tremuloides	Trembling aspen				
P. grandidentata	Bigtooth aspen				
Acer rubrum	Red maple				
Sorbus americana	Mountain ash				
Underst	tory				
Viburnum alnifolium	Witch hobble				
Amelanchier laevis	Shadbush				
Alnus rugosa	Speckled alder				
Lonicera canadensis	Fly honeysuckle				
Vaccinium angustifolium	Lowbush blueberry				
Corylus cornuta	Beaked hazel				
Acer pennsylvanicum	Striped maple				
Viburnum lentago	Sheep berry				
Salix spp.	Willow				
A. spicatum	Mountain maple				
Cornus alternifolia	Alternate-leaved dogwood				
Diervilla lonicera	Mountain-fly honeysuckle				
Herbaceous	s Layer				
Cypripedium calceolus	Yellow Ladys' Slipper				
Adiantum pedatum	Maidenhair fern				
Osmunda Cinnamomea	Cinnamon fern				
Dryonteris disjuncta	Oak fern				

Table 2. Plants occupying hardwood-dominated mixed forests in the Ripogenus and

Table 2.	Continued							
	Scientific Name	Common Name						
	Herbaceous Layer (Cont'd)							
	Trillium erectum Brachvelvtrum erectum	Purple trillium Brachvelutrum						
	Polvaanatum biflorum	Solomons' Seal						
	Drvopteris spinulosa	Spinulose wood fern						
	Osmunda claytoniana	Interrupted fern						
	Medeola virginiana	Indian cucumber						
	Actaea rubra	Dolls' Eyes						
	Pyrola rotundifolia	Shinleaf						
	Athyrium filix-femina	Lady fern						
	Dryopteris phegopteris	Long beech fern						
	Aralia nudicaulis	Wild sarsaparilla						
	Pteridium aquilinum	Bracken fern						
	Arisaema triphyllum	Jack-in-the-pulpit						
	Veratrum viride	False hellebore						
	Trientalis borealis	Starflower						
	Cypripedium acaule var. albecaulis	Pink Lady's Slipper (white form)						
	Uvularia sessilifolia	Bellwort						
	Viola oubescens	Violet						
	Smilacina racemosa	False Solomon's Seal						
	Oxalis montana	Wood sorrel						
	Pteretis pennsylvanica	Ostrich fern						
	Cornus canadensis	Bunchberry						

Table 3.	Plants occupying alder thicket wetlands at the Ripogenus and Penobsco Mills projects, Piscataquis and Penobscot Counties, Maine (original data from GNP Vol. X, 1991).				
	Scientific Name	Common Name			
	Unde	rstory			
	Alnus rugosa	Speckled alder			
	Acer rubrum	Red maple			
	Myrica gale	Sweet gale			
	Rubus sp.	Blackberry			
	Spiraea latifolia	Meadowsweet			
	llex verticillata	Winterberry			
	Viburnum cassinoides	Witherod			
	Salix spp.	Willow			
	Herbaced	bus Layer			
	Carex spp.	Sedge			
	Onoclea sensibilis	Sensitive fern			
	Calamagrostis canadensis	Blue-joint grass			
	Osmunda cinnamomea	Cinnamon fern			
	Dryopteris spinulosa	Spinulose wood fern			

Table 4.Plants occupying conifer swamp/wet forest wetlands at the Ripogenus and Penobscot Mills projects, Piscataquis and Penobscot Counties, Maine (original data from GNP Vol. X, 1991).		
Scientific Name	Common Name	
Can	юру	
Thuja accidentalis Betula Lutea Acer rubrum Abies balsamea Fraxinum pennsylvanica var. subintegerrima F. americana Picea rubens P. mariana	Northern white cedar Yellow birch Red maple Balsam fir Green ash White ash Red spruce	
Tsuga canadensis Larix laricina	Hemlock Larch	
Unde	rstory	
Alnus rugosa Rubus pubescens Ribes glandulosum Cornus rugosa Acer pensylvanicum Lonicera canadensis Salix sp. Cornus sericea	Speckled alder Dwarf raspberry Skunk Current Round leaved dogwood Striped maple Fly honeysuckle Willow Red osier dogwood	
Herbaceous Layer		
Impatiens capensis Thalictrum polygamum Athyrium filix-femina Dryopteris disjuncta D. phegopteris D. cristata D. spinulosa Cornus canadensis Osmunda cinnamomea Chrysosplenium americanum Nasturtium aquaticum Arisaema triphyllum Coptis groenlandica Mitella nuda Osmunda claytoniana Aralia nudicaulis Linnaea borealis Dalibarda repens	Jewelweed Meadow rue Lady fern Oak fern Longbeech fern Crested shield fern Spinulose wood fern Bunchberry Cinnamon fern Golden saxifrage Watercress Jack-in-the-pulpit Goldthread Miterwort Interrupted fern Wild sarsaparilla Twinflower False Violet	

Table 4.	Continued	
	Scientific Name	Common Name
	Herbaceous	Layer (Cont'd)
	Fragaria virginiana	Wild strawberry
l	Carex trisperma	Three-seeded sedge
	Cardamine pensylvanica	Bitter cress
	Moneses uniflora	One-flowered wintergreen
	Pteretis pennsylvanica	Ostrich fern
	Cares arctata	Sedge
	C. foliculata	Sedge
	Oxalis montana	Wood sorrel
	Circaea alpina	Enchanter's nightshade

•

Table 5. Plants occupying wet meadow/emergent marsh wetlands at the Ripogenus and Penobscot Mills projects, Piscataquis and Penobscot Counties, Maine (original data from GNP Vol. X, 1991).		
Common Name		
Dry		
Willow Steeplebush Meadowsweet Speckled alder Dewberry Red raspberry White pine Black spruce Red spruce Leatherleaf Northern white cedar		
Layer		
Blue-joint grass Horsetail Sensitive fern Meadow rue Wild strawberry Sedge Bedstraw Wool grass Brome grass Manna grass Manna grass Manna grass Boneset Bedstraw Lance-leaved goldenrod Bentgrass Three-awned grass Sedge Marsh bluebell Crested shield fern Greenwoodland orchis Reed canary grass Fowl meadow grass Water parsley Sedge Marsh St. Johns wort		

Table 5.	Continued	
	Scientific Name	Common Name
	Herbaceous	Layer (Cont'd)
	Carex leptalea Dulichium arundinaceum Habernaira psycodes Nuphar variegatum Lysimachia terrestris	Sedge Three-way sedge Small purple fringed orchis Yellowpond lily Swamp candles
<i>.</i>	Drosera rotundifolia Lycopus virginicus Carex lurida Eleocharis obtusa Sobagnum sp	Sundew Water horehound Sedge Spikerush Sphagnum moss
	Carex vulpinoidea Scutellaria lateriflora Polygonum sagittatum Chelone glabra	Sedge Skullcap Thumb tear
	Iris versicolor Elymus virginicus Glyceria borealis Sparganium chlorocarpum Typha latifolia	Blue flag Wild rye Manna grass Bur-reed Cattail
	Potentilla palustris	Marsh cinquefoil

Table 6.Plants occupying open bog/wet meadow wetlands at the Ripogenus and Penobscot Mills projects, Piscataquis and Penobscot Counties, Maine (original data from GNP Vol. X, 1991).		
Scientific Name	Common Name	
Und	erstory	
Chamaedaphne calyculata Kalmia angustifolia Kalmia polifolia Ledum groenlandicum Larix laricina Rhododendron canadense Andromeda glaucophylla Picea marina Vaccinium oxycoccos Alnus rugosa Viburnum cassinoides	Leatherleaf Sheep laurel Bog laurel Labrador tea Larch Rhodora Bog rosemary Black spruce Cranberry Speckled alder Witherod	
Herb	Layer	
Eriophorum virginicum E. spissum Carex trisperma Carex oligosperma Carex stricta Scirpus cyperinus Sarracenia purpurea Sphagnum sp. Calamagrostis canadensis	Cotton grass Hare's tail Three-seeded sedge Few-seeded sedge Tussock sedge Woolgrass Pitcher plant Sphagnum moss Blue-joint grass	

Table 7.	able 7. Plants occupying deep marsh/aquatic bed wetlands at the Ripogenus and Penobscot Mills projects, Piscataquis and Penobscot Counties, Maine (original data from GNP Vol. X, 1991).		
	Scientific Name	Common Name	
	Herba	aceous Layer	
	Glyceria borealis	Manna grass	
	Scirpus subterminalis	Bulrush	
	Nymphaea odorata	Pond lily	
	Brasenia schreberi	Water shield	
	Potamogeton epihydrus	Pondweed	
a.	Sparganium sp.	Bur-reed	
	Nuphar variegatum	Yeilow pond lily	
	Utricularia vulgaris	Bladderwort	
	Sparganium fluctuans	Bur-reed	

Table 8.Mammals with typical ranges that overlap the Ripogenus and Penobscot Mills project boundaries, Piscataquis and Penobscot Counties, Maine. An "R" or a "P" after a common name indicates the documented presence of a species within or adjacent to the Ripogenus or Penobscot Mills projects (adapted from original data in GNP Vol. X, 1991).		
Family	Scientific Name	Common Name
Soricidae	Sorex cinereus Sorex palustris Sorex fumeus Sorex dispar Sorex hoyi Blarina brevicauda	Masked shrew Water shrew Smoky shrew Long-tailed shrew Pygmy shrew Northern short-tailed shrew (R,P)
Talpidae	Parascalops breweri Condylura cristata	Hairy-tailed mole Star-nosed mole
Vespertilionidae	Myotis lucifugus Myotis keenii Lasionycteris noctivagans Eptisicus fuscus Lasiurus borealis Lasiurus cinereus	Little brown myotis Keen's myotis Silver-haired bat Big brown bat Red bat Hoary bat
Leporidae	Lepus americanus	Snowshoe hare (R,P)
Sciuridae	Tamias striatus Marmota monax Sciurus carolinensis Tamiasciurus hudsonicus Glaucomys sabrinus	Eastern chipmunk Woodchuck Gray squirrel Red squirrel (R,P) Northern flying squirrel
Castoridae	Castor canadensis	Beaver (R,P)
Cricetidae	Peromyscus maniculatus Clethrionomys gapperi Microtus pennsylvanicus Microtus chcrotorrhinus Ondatra zibethicus Synaptomys cooperi Synaptomys borealis	Deer mouse (R,P) Southern red-backed vole (R,P) Meadow vole (R,P) Rock vole Muskrat (P) Southern bog lemming Northern bog lemming
Muridae	Rattus norvegicus Mus musculus	Norway rat House mouse
Zapodidae	Zapus hudsonius Napaeozapus insignis	Meadow jumping mouse (R,P) Woodland jumping mouse (R,P)
Erethizontidae	Erethizon dorsatum	Porcupine

Table 8. Continued		
Family	Scientific Name	Common Name
Canidae	Canis latrans Vulpes vulpes	Coyote (R,P) Red fox (R,P)
Ursidae	Ursus americanus	Black bear (R,P)
Procyonidae	Procyon lotor	Raccoon (R,P)
Mustelidae	Martes americana Martes pennanti Mustela erminea Mustela frenata Mustela vison Mephitis mephitis Lutra canadensis	Marten (R,P) Fisher Ermine Long-tailed weasel Mink (R,P) Striped Skunk (P) River otter (R,P)
Felidae	Felis lynx Felis rufus	Lynx Bobcat (R,P)
Cervidae	Odocoileus virginianus Alces alces	White-tailed deer (R,P) Moose (R,P)

.....

Table 9. Birds with typical ranges that overlap the Ripogenus and Penobscot Mills project boundaries, Piscataquis and Penobscot Counties, Maine. An "R" or a "P" after a common name indicates the documented presence of a species within or adjacent to the Ripogenus and Penobscot Mills projects (adapted from original data in GNP Vol. X, 1991).		
Family	Scientific Name	Common Name
Gaviidae	Gavia immer	Common loon (R,P)
Podicipedidae	Podilymbus podiceps	Pied-billed grebe
Phalacrocoracidae	Phalacrocorax auritus	Double-crested cormorant (R,P)
Ardeidae	Botaurus lentiginosus Ardea herodias Butorides striatus	American bittern Great blue heron (R,P) Green-backed heron
Anatidae	Branta canadensis Aix sponsa Anas crecca Anas rubripis Anas platyrhynchos Anas discors Aythya collaris Bucephalaclangula Lophodytes cucullatus Mergus merganser Mergus serrator	Canada goose Wood duck Green-winged teal American black duck (R,P) Mallard (R,P) Blue-winged teal Ring-necked duck Common goldeneye Hooded merganser Common merganser (R,P) Red-breasted merganser (R,P)
Accipitridae	Pandion haliaetus Haliaeetus leucocephalus Circus syaneus Accipiter striatus Accipiter cooperii Accipiter gentilis Buteo lineatus Buteo platypterus Buteo jamaicensis Buteo lagopus	Osprey (R,P) Bald eagle (P) Northern harrier (R) Sharp-shinned hawk Cooper's hawk (R,P) Northern goshawk Red-shouldered hawk Broad-winged hawk (R,P) Red-tailed hawk (R) Rough-legged hawk
Falconidae	Falco sparverius	American kestrel (R)
Phasianidae	Dendragapus canadensis Bonasa umbellus	Spruce grouse Ruffed grouse (R,P)
Rallidae	Rallus limicola Porzana carolina	Virginia rail Sora
Charadriidae	Charadrius vociferus	Killdeer (R,P)

Table 9. Continued		
Family	Scientific Name	Common Name
Scolopacidae	Actitis macularia Gallinago gallinago Scolopax minor	Spotted sandpiper (R,P) Common snipe (R,P) American Woodcock (R,P)
Laridae	Larus delawarensis Larus argentatus Larus marinus Sterna hirundo	Ring-billed gull (R) Herring gull (R,P) Great black-backed gull (R) Common tern (R)
Columbidae	Columba livia Zenaida macrour	Rock dove Mourning dove
Cuculidae	Coccyzus erythropthalmus	Black-billed cuckoo
Strigidae	Bubo virginianus Nyctea scandiaca Strix varia Asio otus Asio flammeus Aegolius funereus Aegolius funereus Aegolius acadicus	Great horned owl Snowy owl Barred owl Long-eared owl Short-eared owl Long-eared owl Short-eared owl Boreal owl Northern saw-whet owl
Caprimulgidae	Chordeiles minor Caprimulgus vociderus	Common nighthawk Whip-poor-will
Apodidae	Chaetura pelagica	Chimney swift
Trochilidae	Archilochus colubirs	Ruby-throated hummingbird
Alcedinidae	Ceryle alcyon	Belted kingfisher (R,P)
Ficidae	Sphyrapicus varius Picoides pubescens Picoides villosus Picoides tridactylus Picoides arcticus colaptes auratus Dryocopus pileatus	Yellow-bellied sapsucker (R,P) Downy woodpecker (R,P) Hairy woodpecker (R,P) Three-toed woodpecker Black-backed woodpecker Northern flicker (R,P) Pileated woodpecker (R,P)
Tyrannidae	Contopus borealis Contopus virens Empidonax flaviventris Empidonax alnorum Empidonax minimum Sayorniss phoebe Myiarchus crinitus Tyrannus tyrannus	Olive-sided flycatcher Eastern wood-pewee (R,P) Yellow-bellied flycatcher Alder flycatcher Least flycatcher (R,P) Eastern Phoebe Great crested flycatcher (R,P) Eastern kingbird (P)

Table 9. Continued		
Family	Scientific Name	Common Name
Alaudidae	Eremophila alpestris	Horned lark
Hirundinidae	Tachycineta bicolor Riparia riparia Hirundo pyrrhonota Hirundo rustica	Tree swallow (R,P) Bank swallow Cliff swallow Barn swallow (R,P)
Corvidae	Perisoreus canadensis Cyanocitta cristata Corvus brachyrhynchos Corvus corax	Gray jay Blue jay (R,P) American crow (P) Common raven (R,P)
Paridae	Parus atricapillus Parus hudsonicus	Black-capped chickadee (R,P) Boreal chickadee
Sittidae	Sitta canadensis Sitta carolinensis	Red-breasted nuthatch (R,P) White-breasted nuthatch (R,P)
Certhiidae	Certhia americana	Brown creeper
Troglodytidae	Troglodytes aedon Troglodytes troglodytes Cistothorus palustris	House wren Winter wren (R,P) Marsh wren
Muscicapidae	Regulus satrapa Regulus calendula Sialia sialis Catharus fuscescens Catharus minimus Catharus ustulatus Catharus guttatus Hylocichla mustelina Turdus migratorius	Golden-crowned kinglet Ruby-crowned kinglet (R,P) Eastern bluebird Veery (R,P) Gray-cheeked thrush Swainson's thrush (R,P) Hermit thrush (R,P) Wood thrush (R,P) American robin (R,P)
Mimidae	Dumetella carolinensis	Gray catbird (R,P)
Bombycillidae	Bombycilla garrulus Bombycilla cedrorum	Bohemian waxwing Cedar waxwing (R,P)
Laniidae	Lanius excubitor	Northern shrike
Sturnidae	Sturnus vulgaris	European starling (R,P)
Vireonidae	Vireo solitarius Vireo gilvus Vireo philadelphicus Vireo olivaceus	Solitary vireo (R,P) Warbling vireo Philadelphia vireo Red-eyed vireo (R,P)

Pamily	Scientific Name	Common Name
Emberizidae	Vermivora peregrina	Tennessee warbler (R,P)
	Vermivora ruficapilla	Nashville warbler (R,P)
	Parula americana	Northern parula (R,P)
	Dendroica petechia	Yellow warbler (R,P)
	Dendroica pennsylvanica	Chestnut-sided warbler (R,P)
	Dendroica magnolia	Magnolia warbler (R,P)
	Dendroica tigrina	Cape May warbler (R,P)
	Dendroica caerulescens	Black-throated blue warbler (R,P)
	Dendroica coronata	Yellow-rumped warbler (R,P)
	Dendroica virens	Black-throated green warbler (R,P)
	Dendroice fusca	Blackburnian warbler (R,P)
	Dendroica pinus	Pine warbler
	Dendroica palmarum	Palm warbler
,	Dendroica castanea	Bay-breasted warbler (R,P)
	Dendroica striata	Blackpoll warbler (R,P)
	Mniotilta varia	Black-and-white warbler (R)
	Setophaga ruticilla	American redstart (R)
	Seiurus aurocapillus	Ovenbird (R,P)
	Seiurus noveboracensis	Northern waterthrush (R,P)
	Oporonis philadelphia	Mourning warbler
	Geothlypis trichas	Common yellowthroat (R,P)
	Wilsonia pusilla	Wilson's warbler
	Wilsonia canadensis	Canada warbler (R,P)
	Piranga olivacea	Scarlet tanager (R,P)
	Pheucticus Iudovicianus	Rose-breasted grosbeak (R,P)
	Passerina cyanea	Indiao bunting
	Snizella nasserina	Chipping sparrow
	Popecetes gramineus	Vesper sparrow
	Passerculus sandwichensis	Savannah sparrow (R)
	Melospiza melodia	Song sparrow (R.P)
	Melospiza lincolnii	Lincoln's sparrow
	Melospiza georgiana	Swamp sparrow (R.P)
	Zonotrichia albicollis	White-throated sparrow (R.P)
	Junco hvemalis	Dark-eved junco (R.P)
	Calcarius Iannonicus	Lapland longspur
	Plectrophenax nivalis	Snow bunting
	Dolichonyx aryziyorus	Bobolink
	Agelaius phoeniceus	Red-winged blackbird (R.P)
	Sturnella manna	Fastern meadowlark
	Funhanue carolinue	Rusty blackbird
	Lupnayus caronnus Ouiecalus quiecula	Common grackle (R)
	Molothrup stor	Brown-beaded cowhird
	WUUUIIIUS BLEI	DIVATITIÇALGU CUANTIL

1

Table 9. Continued			
Family	Scientific Name	Common Name	
Fringillidae	Pinicola enucleator Carpodacus purpureus Loxia curvirostra Loxia leucoptera Carduelis flammea Caduelis hornemanni Carduelis pinus Carduelis tristis Coccothraustea vespertinus	Pine grosbeak Purple finch (R,P) Red crossbill White-winged crossbill Common redpoll (R,P) Hoary redpoll Pine siskin (R,P) American goldfinch (R,P) Evening grosbeak (R,P)	
Passeridae	Passer domesticus	House sparrow	

.

Table 10. Reptiles and amphibians with typical ranges that overlap the Ripogenus and Penobscot Mills project boundaries, Piscataquis and Penobscot Counties, Maine. An "R" or a "P" after a common name indicates the documented presence of a species within or adjacent to the Ripogenus and Penobscot Mills projects (original data from GNP Vol. X, 1991).			
Family	Scientific Name	Common Name	
Ambystomatidae	Ambystoma jeffersonianum Ambystoma laterale Ambystoma maculatum	Jefferson salamander Blue-spotted salamander Spotted salamander (R,P)	
Salamandridae	Notophthalmus v. viridescens	Red-spotted newt	
Plethodontidae	Desmognathus f. fuscus Plethodon cinereus Eurycea b. bislineata	Northern dusky salamander (R,P) Redback salamander (R,P) Northern two-lined salamander	
Bufonidae	Bufo a. americanus	Eastern american toad (R,P)	
Hylidae	Hyla c. crucifer	Northern spring peeper (R,P)	
Ranidae	Rana catesbeiana Rana clamitans melanota Rana septentrionalis Rana sylvatica Rana pipiens Rana palustris	Bullfrog (R,P) Green frog (R,P) Mink frog Wood frog (R,P) Northern leopard frog (R,P) Pickerel frog (R)	
Chelydridae	Chelydra s. serpentina	Common snapping turtle	
Emydidae	Clemmys insculpta Chrysemy p. picta	Wood turtle Eastern painted turtle	
Colubridae	Storeria a. occipitomaculata Thamnophis s. sirtlis Thamnophis sirtalis pallidula Diadophis punctatus edwardsi Opheodrys v. vernalis	Northern redbelly snake Eastern garter snake (R) Maritime garter snake Northern ringneck snake Eastern smooth green snake	

APPENDIX D

ī.

Ę.

GNP'S WATER-USE MODEL
APPENDIX D.

GNP'S WATER-USE MODEL

D.1 DESCRIPTION OF THE MODEL

GNP's hydropower system historically has been operated to maximize power for mill operations. GNP used a historically derived rule curve (figure D-1) to determine reservoir operations for this purpose. Based on comments received from agencies and other interest groups during the consultation stage of the relicensing process, GNP determined that a wateruse model of its hydro system would be needed to evaluate various proposed flow enhancements. The purpose of the model was to assist in evaluating the various flow requests and aid in determining how conflicting requests could best be accommodated. The model was also used to simulate impoundment levels and possible release flows under major alternative flow scenarios. GNP developed the model to evaluate the following original flow requests as listed in table D-1.

GNP's water-use model includes the major storage reservoirs of its system, including the upper storage ponds, Ripogenus impoundment, and North Twin impoundment. The upper storage ponds are listed in table 3-1 and are grouped as one compartment in the model. These projects are used solely as storage reservoirs and do not produce power. Because they are not included in the current relicensing process, GNP used the historical operation of these projects as input to the model but did not simulate any changes in their operation. Other elements of the model include the inflow, storage, elevation, and outflow from Ripogenus and North Twin impoundments. Millinocket Storage Lake is not explicitly included in the model, although it supplies a small amount of storage and can be used to supplement North Twin as needed. The other impoundments within GNP's system have no usable storage and, therefore, are not included in the model; see figure 2-3 and table 3-2 for a complete diagram and listing of the impoundments included in the Ripogenus and Penobscot Mills projects. Figure D-2 shows a diagram of elements included in GNP's water-use model. Elevation/storage curves of the Ripogenus, North Twin, and Millinocket Lake Storage impoundments are shown in figures D-3 through D-5.

D.1.1. Period of Record and Timestep

GNP chose the 15-year period from 1976 to 1990 to provide historical operating records representing hydrologic conditions within which a water-use plan would need to operate. Prior to 1973, when log driving operations occurred on the West Branch, GNP's reservoir and river system were operated differently than they are now. The hydrologic system continued to be influenced by those operations for several years, so GNP started with 1976. Within this 15-year period, GNP defined very wet and very dry years based on total available water, which consisted of water available in storage and outflow at Ripogenus (table D-2). GNP believed the weekly timestep to be appropriate for the system because it does not have peaking operations or other short-term fluctuations that might require a smaller timestep. Using operating records of impoundment elevations (for the upstream projects collectively,



Figure D-1. Rule curve for GNP's integrated power system on the West Branch of the Penobscot River, based on a flow of 2400 cfs at Millinocket with a 1% probability of depleting storage (from figure B-3, GNP, 1991a)

0-4 4

Table D-1. Initial flow requests by agencies and interest groups to be evaluated for the Ripogenus and Penobscot Mills projects							
Water Body	Request	Reason	Time Period				
Ripogenus	"relatively stable"	wildlife	1 May - 15 July				
Impoundment	specifically defined)	wetlands, recreation, aesthetics	1 May - 30 Sept.				
Upper Gorge	100 cfs	salmon fishery	1 July - 30 Sept.				
	50 cfs	salmon fishery	1 Oct 30 June				
West Branch	2400 - 3600 cfs	rafting	1 May - 15 Oct.				
Station	ABF or IFIM	salmon fishery	15 June - 30 Sept.				
	maintain flow	salmon spawning salmon incubation	15 Oct 15 Nov. 16 Nov 7 June				
North Twin and	"relatively stable"	wildlife	1 May - 15 July				
Impoundments	water levels	wetlands, recreation, aesthetics	1 May - 30 Sept.				
	draw-down limit	lake trout spawning lake trout incubation	15 Oct 5 Nov. 6 Nov 1 May				
Back Channel	350 - 500 cfs	fishery habitat	year round				
Millinocket Stream	60 - 80 cfs	fishery habitat	year round				
West Branch downstream of Millinocket Mills	2,000 cfs instantaneously (required by state law, 401 WQC, and existing FERC license.	water quality	year round				
Aquatic baseflow (ABF) or Instream Flow Incremental Methodology (IFIM) study requested to determine minimum flow							



Figure D-2. Diagram of the major elements included in GNP's water use model of the Ripogenus and Penobscot Mills projects (Source: Staff)

÷



Figure D-3. Area-capacity curve for the Ripogenus impoundment (dashed line represents the combined surface area of Chesuncook Lake at elevation 910.49 ft., and Ripogenus Lake (from figure B-2, GNP, 1991a)

P7



Figure D-4. Area-capacity curve of the North Twin development (from figure B-2, GNP, 1991b)



Figure D-5. Area-capacity curve of the Millinocket Lake Storage development (from figure B-3, GNP, 1991b)

0-9

a seesilii a

Table D-2.Hydrologic rankings of water years (April through March) used in GNP's water-use model (based on historical records of total available water, defined as average storage and discharge volumes at Ripogenus impoundment)									
YEAR	Total Available Water (bcf) ^(c)	Rank (1 = lowest water; 15 = highest water)	GNP's Hydrologic Type						
1976	34.8	15	wet						
1977	31.9	13							
1978	27.9	10							
1979	25.0	7							
1980	14.5	1							
1981	28.5	11							
1982	25.6	8	average						
1983	29.3	12							
1984	25.0	6	· · · · · · · · · · · · · · · · · · ·						
1985	16.1	2	dry						
1986	26.3	9							
1987	21.1	4							
1988	17.4	3							
1989	24.4	5							
1990	34.7	14							
'typical' average ^(a)	25.2	•							
'typical' worst-case ^(b)	13.6	-							

^(a) Based on compilation of weekly average values of total available water from the period of record (1976-1990).

^(b) Based on compilation of weekly lowest values of total available water from the period of record (1976-1990).

 c_i bcf = billion cubic feet.

Ripogenus, and North Twin) and outflow records, GNP constructed a spreadsheet model to represent weekly inflow, storage, and outflow through the major project elements. GNP used the entire 15-year period of record to establish the historical baseline of project operations, although individual years also can be evaluated. Because the lowest amount of available water typically occurred toward the end of March, the model years run from the end of March one year, to the end of March in the following year. GNP also created synthetic 'typical' years to represent average conditions and worst-case, or minimum available water, conditions by using the average and driest individual weeks, respectively, within the period of record. Because it consists of the data from the driest weeks within the 15-year period of record, the worst-case year represents more extreme conditions than the single driest year within that record. It is important to note that the model is not predictive but allows simulations of changes in water use based on historically available water.

GNP simulated additional flows to the BC by adding flow to the 2000 cfs minimum flow required at Millinocket, plus an approximate 10% buffer to account for variations in flow regulation, gate settings, and control equipment sensitivities between North Twin and Millinocket Mills. The required amount of flow buffer capacity has not been quantitatively evaluated. The model runs attempted to release the combined minimum flow while maintaining desired flows in the West Branch and North Twin impoundment levels (North Twin outflow minimum was 2610 cfs to simulate 350 cfs to the BC and 2420 cfs to simulate 165 cfs to the BC). In a scenario with insufficient water within the simulation year, the model operates as follows. The simulations continue with the required minimum flows from North Twin until there is no longer enough water to continue at that rate without depleting historically available storage from the following year. Flows are then reduced to a rate small enough to prevent storage depletion. This reduced flow may or may not be sufficient to maintain the BC flow and the 2000 cfs minimum at Millinocket. Other flows and impoundment level goals may be jeopardized as well. If BC flows were made as a permanent enhancement measure, all flowrelated enhancements would have to be prioritized to list those which would be sacrificed first. Such a scenario might for instance involve maintaining the minimum flow at Millinocket Mills, reducing BC flows, and relaxing North Twin impoundment management for lake trout; alternatively, flows through the mill could be reduced.

D.1.2 Model Results

GNP determined which of the requested enhancements could be met at all times without compromising others (GNP 1991a, 1991b, 1992d, 1993d, 1994; letter from J. Carson, GNP, April 5, 1994). The final list of enhancements proposed by GNP are known collectively as the water-use plan (WUP) and are listed in table 4-1. Simulations comparing historic project operations with WUP operations (in terms of impoundment elevation and outflow at Ripogenus and North Twin) are illustrated in figures D-6 through D-9 and table D-3, for the wet, average, dry and worst-case years, respectively. These results are described in Section 4.2.1; in summary, all of the enhancements proposed by GNP can be accommodated in all years.



Figure D-6. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under historic conditions and the Applicant's Proposal during a wet year (1976)



Figure D-7. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under historic conditions and the Applicant's Proposal during an average year (1982)



Figure D-8. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under historic conditions and the Applicant's Proposal during a dry year (1985)



Figure D-9. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under historic conditions and the Applicant's Proposal during a worst-case year (lowest available water)

Table D-3. Comparison of weekly average impoundment levels and outflows at Ripogenus and North Twin under historical conditions and the Applicant's Proposal for a wet year, average year, dry year, and worst-case year, based on GNP's water-use model									
		Wet Year (1976)		Average Year (1982)		Dry Year (1985)		Worst-Case Year	
		Hist.	Applicant's Proposal	Hist.	Applicant's Proposal	Hist.	Applicant's Proposal	Hist.	Applicant's Proposal
Ripogenus	Minimum	925.7	925.6	925.0	926.5	915.7	915.6	913.8	914.0
(ft, USGS)	Maximum	942.1	941.2	941.6	941.6	932.4	932.0	930.3	930.1
	Average	938.3	936.3	934.0	934.9	926.6	925.8	924.2	924.6
	Range	16.4	15.5	16. 6	15.1	16.6	16.4	16.5	16.1
North Twin	Minimum	482.2	478.9	478.6	478.7	477.4	478.4	475.3	472.3
Elevation (ft. USGS)	Maximum	492.2	491.9	491.9	491.8	483.8	488.2	483.4	484.5
	Average	487.7	485.5	485.7	484.6	481.9	483.6	481.0	478.4
	Range	10.0	13.0	13.3	13.1	6.4	9.8	8.1	12.1
Ripogenus	Minimum	1974	2000	385	2000	1295	1800	795	1700
Outflow (cfs)	Maximum	6523	7500	7153	7000	2695	2150	2680	2100
	Average	3583	3585	2696	2688	1907	1906	1817	1817
North Twin Outflow (cfs)	Minimum	2802	2400	2023	2000	2003	2000	2003	2240
	Maximum	9106	9500	5924	6500	3457	4250	3489	4000
	Average	4549	4566	3339	3334	2592	2591	2530	2530

GNP provided simulations of two major flow alternatives - a conservation intervenororiented alternative (Alternative 1) with 350 cfs flow to Back Channel year round (figures D-10 through D-13; table D-4), and a staff alternative (Alternative 2) of 165 cfs to Back Channel year round (figures D-14 through D-17; table D-5). GNP provided these simulations for a wet, average, dry, and worst-case year, based partially on a FERC Additional Information Request. We explain these results in Sections 4.2.2 and 4.2.3 and summarize overall results in Section 4.2.5. The simulations for BNP's proposed Alternatives 1 and 2 were in addition to those already conducted by GNP, including a draw-down limit at Ripogenus and 100 cfs in Back Channel, separately and combined with the draw-down limit.

D.2 CRITICISMS OF MODEL

The Conservation Intervenors (CI) proposed terms and conditions for the Ripogenus and Penobscot Mills projects (letter from D. Sosland, CI, September 3, 1993). The CI criticized GNP's water-use model on the following grounds: 1) It fails to include watershed scale processes; 2) it uses uncalibrated data; 3) it uses an inadequate timestep; 4) it excludes over-year storage; 5) it does not include reservoirs other than North Twin and Ripogenus; 6) it is unable to consider conservation measures in the economic analysis; and 7) it lacks a habitat evaluation module (letter from D. Sosland, CI, November 2, 1993). In addition, CI believes that GNP did not use the model to evaluate different flow alternatives. Staff evaluation of these concerns are as follows.

D.2.1 Watershed Scale Processes

The CI and others want the water use model to include "watershed scale processes." These processes would seem to include the effects of timber management practices and other land use practices. These effects are not caused by project operations and, therefore, are outside the scope of this EIS. The CI also want all of the projects in at least the West Branch portion of the watershed included in the water use model. The water-use model does include all the major storage reservoirs, including the upstream storage ponds (the Great Northern Storage Project and unlicensed storage projects - see table 3-1), and Ripogenus and North Twin impoundments. Although the model treats the upstream storage projects as one compartment, staff does not believe it necessary to model them as separate entities. These projects do not generate power and are not included in the current license process, so their operation could not be modified at this time. In addition, their total storage represents less than 20 percent of GNP's total system storage, and their operation would be unaffected by downstream processes. Their current operation is similar to that of Ripogenus, with a minimum storage during the fall and winter months, a filling period starting in April with spring runoff, and a gradual draw-down during the summer months; the exact timing of draw-down depends on water availability during a particular year (figures D-18 through D-21). Although site-specific impacts could warrant a change in their operation, staff believes that it is unlikely that additional downstream enhancements would be possible by changing the current mode of operation. Any operational changes to increase downstream water availability probably would result in large draw-downs and related impacts in the upper ponds. Changes to decrease draw-downs in the upper ponds would probably reduce the ability to meet downstream enhancements. In addition, GNP has agreed to a reopener clause in its licenses for Ripogenus and Penobscot Mills, which would allow for a change in its proposed WUP if changes proposed during relicensing for the upper projects require it (GNP, 1993b, 1993c).



Figure D-10. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under the Applicant's Proposal and Alternative 1 (350 cfs to the Back Channel) during a wet year (1976)



Figure D-11. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under the Applicant's Proposal and Alternative 1 (350 cfs to the Back Channel) during an average year (1982)



Figure D-12. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under the Applicant's Proposal and Alternative 1 (350 cfs to the Back Channel) during a dry year (1985)



Figure D-13. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under the Applicant's Proposal and Alternative 1 (350 cfs to the Back Channel) during a worst-case year (lowest available water)

Table D-4. Comparison of weekly average impoundment levels and outflows at Ripogenus and North Twin under the Applicant's Proposal and Alternative 1 (350 cfs in the Back Channel), for a wet year, average year, dry year, and worst-case year, based on GNP's water-use model									
		Wet Year (1976) Average Year (1982)		Dry Year (1985)		Worst-Case Year			
		Applicant's Proposal	Ait . 1	Applicant's Proposal	AH. 1	Applicant's Proposal	Alt, 1	Applicant's Proposal	Alt. 1
Ripogenus	Minimum	925.6	925.6	926.5	926.0	915.6	915.4	914.0	912.7
Elevation (ft. USGS)	Maximum	941.2	941.2	941.6	941.6	932.0	930.1	930.1	929.3
•	Average	936.3	936.3	934.9	934.4	925.8	923.2	924.6	922.9
	Range	15.5	15.5	15.1	15.6	16.4	14.7	16.1	16.6
North Twin Elevation (ft. USGS)	Minimum	478.9	478.9	478.7	478.3	478.4	478.2	472.3	475.2
	Maximum	491.9	491.9	491.8	491.8	488.2	486.6	484.5	483.7
	Average	485.5	485.4	484.6	484.3	483.6	483.3	478.4	478.6
	Range	13.0	13.0	13.1	13.2	9.8	8.4	12.1	8.5
Ripogenus Outflow (cfs)	Minimum	2000	2000	2000	2000	1800	690	1700	400
	Maximum	7500	7500	7000	7000	2150	2610	2100	2500
	Average	3585	3585	2688	2694	1906	1907	1817	1817
North Twin	Minimum	2400	2610	2000	2610	2000	2272	2240	1300
Outflow (cfs)	Maximum	9500	9500	6500	6500	4250	3000	4000	3200
	Average	4566	4568	3334	3341	2591	2592	2530	2530

G



Figure D-14. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under the Applicant's Proposal and Alternative 2 (165 cfs to the Back Channel) during a wet year (1976)



Figure D-15. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under the Applicant's Proposal and Alternative 2 (165 cfs to the Back Channel) during an average year (1982)



Figure D-16. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under the Applicant's Proposal and Alternative 2 (165 cfs to the Back Channel) during a dry year (1985)



Figure D-17. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under the Applicant's Proposal and Alternative 2 (165 cfs to the Back Channel) during a worst-case year (lowest available water)

Table D-5.Comparison of weekly average impoundment levels and outflows at Ripogenus and North Twin under the Applicant's Proposal and Alternative 2 (165 cfs in the Back Channel), for a wet year, average year, dry year, and worst-case year, based on GNP's water-use model									
		Wet Year	(1976)	Average Year (1982)		Dry Year (1985)		Worst-Case Year	
		Applicant's Proposal	Alt. 2						
Ripogenus	Minimum	925.6	925.6	926.5	926.5	915.6	915.4	914.0	912.7
(ft. USGS)	Maximum	941.2	941.2	941.6	941.6	932.0	930.6	930.1	929.4
	Average	936.3	936.3	934.9	934.9	925.8	924.6	924.6	923.2
	Range	15.5	15.5	15.1	15.1	16.4	15.2	16.1	16.7
North Twin Elevation (ft. USGS)	Minimum	478.9	478.9	478.7	478.2	478.4	478.3	472.3	475.4
	Maximum	491.9	491.9	491.8	491.8	488.2	487.3	484.5	485.0
	Average	485.5	485.5	484.6	484.2	483.6	483.6	478.4	480.4
	Range	13.0	13.0	13.1	13.6	9.8	8.9	12.1	9.6
Ripogenus	Minimum	2000	2000	2000	2000	1800	1400	1700	400
Outflow (cfs)	Maximum	7500	7500	7000	7000	2150	2420	2100	2500
	Average	3585	3585	2688	2687	1906	1916	1817	1817
North Twin	Minimum	2400	2420	2000	2420	2000	2272	2240	2420
Outflow (cfs)	Maximum	9500	9500	6500	6500	4250	3000	4000	3300
	Average	4566	4566	3334	3340	2591	2592	2530	2530



Figure D-18. Water-use model results comparing weekly total impundment storage in the upper ponds (panel A), Ripogenus (panel B), Ripogenus and North Twin combined (panel C), and North Twin (panel D), under historic conditions and GNP's proposed Water Use Plan (WUP) during a wet year (1976)



Figure D-19. Water-use model results comparing weekly total impundment storage in the upper ponds (panel A), Ripogenus (panel B), Ripogenus and North Twin combined (panel C), and North Twin (panel D), under historic conditions and GNP's proposed Water Use Plan (WUP) during an average year (1982)



Figure D-20. Water-use model results comparing weekly total impundment storage in the upper ponds (panel A), Ripogenus (panel B), Ripogenus and North Twin combined (panel C), and North Twin (panel D), under historic conditions and GNP's proposed Water Use Plan (WUP) during a wet year (1985)



Figure D-21. Water-use model results comparing weekly total impundment storage in the upper ponds (panel A), Ripogenus (panel B), Ripogenus and North Twin combined (panel C), and North Twin (panel D), under historic conditions and GNP's proposed Water Use Plan (WUP) during a worst-year (lowest available water)

The model could be used to simulate collective changes in the flows from the Upper Storage ponds; however, no one has suggested such a modelling scenario or suggested a way to operate the upper projects to produce additional enhancements.

D.2.2 Uncalibrated Input Data

Staff is not certain about what CI means by this statement. GNP's water-use model uses actual (historical) measured impoundment levels and flows for the 15-year period of record as a baseline within which to evaluate various proposed water uses. The historical data provides information about the total available water within each year that can be redistributed to provide alternative water levels and flows within the system. GNP ran simulations for all 15 years of available historical data using spreadsheet files for each individual year. The ending point of the first year (in terms of water levels) was the starting point for the following year, and so on. Water could not be reallocated from the following year without affecting water availability in the subsequent year.

D.2.3 Inadequate Timestep

GNP used a weekly timestep, and CI requested a daily timestep to enable simulation of daily changes that may affect recreational flows and the influence of storm flows. Staff contends that this is not necessary for several reasons. Because GNP operates its system for continuous power production rather than for peaking operation, large changes in flow within a day do not occur due to project operation and thus do not require simulation to determine water usage. In addition, the storage capacities of Ripogenus and North Twin are so large that flows resulting from storm events occurring over a period of days would be absorbed; seasonal patterns in rainfall, flows, and storage are the dominant factors affecting water availability in the system. The requested enhancements also occur on a monthly to seasonal basis and do not require the detailed analysis that would be available with a daily timestep model. The only enhancements not specifically simulated with the weekly timestep model are the recreation flows, which occur on an hourly/daily scale. From an overall water-use perspective, however, the weekly timestep model is sufficient to characterize the water available for that need. Having the ability to simulate hourly/daily flow fluctuations is not required to determine water availability on a monthly/seasonal basis.

D.2.3 Over-year Storage

GNP used storage and inflow data for the upper ponds, Ripogenus, and North Twin for 15 consecutive years (1976 to 1990) as input for the water-use model. GNP conducted simulations for individual years and presented data for a dry year and a 'typical' average year in its application; however, GNP actually simulated all 15 years for the historical (existing conditions) and WUP cases. For each year, historically available water in storage at the end of the year was used as the starting point for the next year, regardless of the alternative being simulated. Other alternatives could also be simulated for any or all of the years, although staff believes that analysis of alternatives for a wet, average, and dry year is adequate to determine whether there is sufficient water to meet all of the requested flows.

D.2.4 Omission of Other Reservoirs

Staff addressed the inclusion of the upper ponds in the water-use model earlier. Other than Millinocket Lake Storage Development (MLS), the other impoundments in GNP's system (Quakish Lake, Ferguson Pond, Shad Pond, Dolby, East Millinocket) are operated as run-ofriver and have no significant storage. Their operation depends almost entirely on flows released from Ripogenus and North Twin; consequently, including them in the water-use model would serve no useful purpose. MLS has a storage capacity of two billion cubic feet, less than four percent of total storage capacity of GNP's system. It is used to supplement flows to North Twin via a pumping station, or flows may be released to Millinocket Stream. Water available from MLS is included in the model, although its storage is not. Including this storage in the model would allow a more direct determination of impacts of minimum flow releases on the impoundment elevation, although its storage is not significant enough to influence the probability of meeting other water use needs in the overall system. Impacts of minimum flows from MLS on its elevation can be evaluated independently of the model; see sections 4.2.2 and 4.2.3.

D.2.5 Economic Analysis

CI claims that the model is biased toward power production. The model itself is not used to evaluate the effects of flow alternatives on project economics. We used it solely to evaluate the feasibility of providing alternative flows using available water. We have not seen any other suggested management schemes that provide for additional enhancement while including the state's desired fishery management goals. There is nothing inherent in the model structure that would preclude its use in evaluating other project alternatives.

D.2.6 Habitat Evaluation Module

Staff is not aware of any other water-use model currently in use that contains a habitat evaluation module; habitat analyses are usually conducted with IFIM, independently of any hydrologic or hydraulic model. The HEC-5 model (suggested by CI as a more appropriate water-use model) does not contain such a module. Staff used the results of the IFIM analyses conducted by GNP to reach its own conclusions about minimum flows appropriate for various stream reaches within the project areas (see section 4.4).

D. 3 OTHER SCENARIOS EVALUATED BUT NOT CONSIDERED AS REASONABLE ALTERNATIVES

D.3.1 Draw-down Limits

D.3.1.1 Full-Pond, Run-of-River Operation

D.3.1.1.1 Flood Control Benefits of Current Operation

Current operations at the Ripogenus and Penobscot Mills projects are designed to maximize water storage during the spring snowmelt and runoff period for sustained power generation during the lower-flow summer and fall months. Such operation smooths out the natural fluctuations in river flow and creates a more constant and moderated flow in the West

Branch. These operating practices and the large storage capacity of the Ripogenus and North Twin impoundments significantly reduce the magnitude and frequency of flood flows in the entire Penobscot River basin (GNP, 1991a). Short-term storms and seasonal runoff are captured during wet periods and released later during periods of lower flow. The Corps of Engineers (Corps 1989 as cited in GNP, 1991a) stated that the West Branch has historically contributed very little to flood events due to the large reservoir storage capacity; the present mode of operation has greatly reduced the West Branch's contribution to main stem flood peaks. During the April 1987 flood for instance, significant flood damage occurred in the Penobscot River Basin, but no significant damage occurred in towns on the West Branch. Moreover, significant flow reduction and storage in the project impoundments of the West Branch resulted in a 98 percent decrease in potential discharge at Millinocket and more than a 35 percent decrease in potential flow in Bangor, Maine (GNP, 1991a). GNP's WUP would not significantly alter operations relative to flood control and, in fact, could slightly increase flood control benefits. During the wet year simulation (figure D-6), impoundment levels were slightly reduced over historic operations, and slightly more storage volume would have been available prior to spring runoff.

D.3.1.1.2 Impacts of Full-pond Operation

During the consultation phase of relicensing, several agencies requested consideration of a full-pond operating policy for the Ripogenus and Penobscot Mills projects to provide enhancements of impoundment recreation, fisheries, wildlife, and wetlands (GNP, 1991a). GNP evaluated this proposed policy and found that it would increase the magnitude and frequency of flooding in the basin because impoundment storage would no longer be available. The West Branch would experience flood flows that do not occur under current operating procedures. Flooding would also be increased downstream in the basin. GNP calculated that a 100-year flood (approximately 200,000 cfs at Bangor) would become a 50year event with a full-pond policy. There would also be an 80 percent increase in flood damage costs in the basin for a 500-year event, and a 300 percent increase in flood damage costs for a 10-year event, as compared with the current water management policy. GNP calculated the annual loss of flood control benefits to be \$1,000,000 (in 1994 dollars) or \$69 million over the 30-year license term.

A full-pond operation policy would also have adverse impacts on Bangor Hydro generation, James River Paper generation, whitewater rafting, water quality, and some aquatic resources; however, there would be some positive benefits for other aquatic resources and recreation opportunities. In balance, the adverse impacts would be much greater than the minor positive benefits of full-pond operation, and such a policy would be unacceptable. Staff asserts that such a policy should not be considered further.

D.3.1.2 Partial Ripogenus Draw-down Limit

GNP showed that a total limit would have unacceptable adverse impacts on downstream recreation, fisheries, water quality, and flood control. GNP used the water-use model to illustrate impacts on downstream water availability by simulating a partial draw-down limit on Ripogenus. This simulation was conducted for a dry year (1985) and a worst-case year by reducing the outflow from Ripogenus such that the maximum draw-down would not exceed 10.1 bcf. This amount of storage is roughly equivalent to limiting the change in elevation during the year to about half of the normal level (figures D-22 and D-23). The consequences of this limit can be seen in the outflow from Ripogenus (panel B in these figures), which is significantly reduced during the winter. This would prevent attainment of salmon spawning flows during this period. North Twin elevations also would be affected (panel C) because less water would be available downstream. Elevations would be reduced by several feet during the fall and winter period, and the lake trout incubation period would be affected by a draw-down that exceeds the minimum spawning level, which is targeted for the end of October in the WUP. Flood control benefits also would be affected because the Ripogenus impoundment elevations would be significantly elevated over the historical and WUP levels at the end of the winter. This would reduce storage in Ripogenus and could result in spillage and flood flows during a subsequent wet spring; however, this loss in flood control benefits has not been quantified. Staff agrees with GNP and other agencies that a draw-down limit on Ripogenus should not be imposed because even a partial limit would have unacceptable adverse effects.

D.3.2 Unregulated Flows in the West Branch

Trout Unlimited (TU) believes that flows in the West Branch are too high during summer and too low during spring (letter from C.F. Gauvin and M. M. Janopaul, May 24, 1993). TU suggests a ramped flow consisting of much lower flows from July through October (500 to 1000 cfs), higher flows from April through May (3000 to 6750 cfs), and 750 to 2000 cfs from November through March and during June. No other agency or group has suggested such flows. These recommendations do not account for the state's recommendations as specified in the 401 WQC and would render the goals of the state and GNP's WUP unobtainable. They also would create a large economic penalty to GNP and could result in a substantial loss in flood control benefits within the river basin; therefore, we do not consider these recommendations as part of a reasonable alternative.

D.3.3 Millinocket 2000 cfs Minimum Flow

EPA, PIN, and TU suggested reducing the 2,000-cfs minimum flow at Millinocket to provide additional flows to the Back Channel (BC) without altering other aspects of the proposed Water-Use Plan (letters from R.G. Manfredonia, EPA Region I, May 21, 1993; P. Biscula, PIN, May 21, 1993; C.F. Gauvin and M.M. Janopaul, TU, May 24, 1993). GNP is presently required by Maine legislation and its WQC to release a 2,000-cfs minimum flow at Millinocket (this release is currently passed through the generating station at Millinocket Mills). The outflow from Millinocket Mills combines with the outflow from Shad Pond (into which BC flows empty) a few miles downstream from Millinocket Mills. There is a legal question as to exactly where existing regulations and legislation require that the 2,000-cfs minimum flow be met. The state law says GNP "may use West Branch water, subject to the provision that day and night throughout the year the flow of water down the West Branch, so long as there shall be any stored water shall not be less than two thousand cubic feet per second, measured in the canal and at the stone dam of the Great Northern Paper Company, at Millinocket ... " (Chapter 174, Section 10 of An Act to Incorporate the West Branch Driving and Reservoir Dam Company, enacted by the Senate and House of Representatives, Approved March 13, 1901, as cited in USEPA's letter to FERC on Penobscot Mills, dated May 21, 1993).



Figure D-22. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under historic conditions GNP's proposed Water Use Plan (WUP), and with the drawdown limit for Ripogenus during a dry year (1985)



Figure D-23. Water-use model results comparing weekly average impoundment elevations (panels A,C) and outflows (panels B,D) at Ripogenus and North Twin under historic conditions GNP's proposed Water Use Plan (WUP), and with the drawdown limit for Ripogenus during a worst-case year (lowest available water)

EPA contends that this state law should be interpreted to mean that the combined flows at Millinocket Mills and BC (stone dam) must be maintained at 2,000 cfs. Thus, EPA claims that BC flow of any magnitude could be maintained without affecting any other water management measures in the project area by decreasing the flow through Millinocket Mills. GNP claims that the legislation cannot be interpreted in that way; diverting flow from Millinocket Mills to BC decreases power production at Millinocket Mills. The WQC states only that 2,000 cfs be passed "at Millinocket." Trading flows between Millinocket Mills and BC is the only sure way of providing significant flows in the BC without affecting other water management objectives and WQC condition requirements for the two projects under all hydrologic conditions. The only possible environmental consequence of a flow out of Millinocket Mills less than 2,000 cfs could be a reduction in dilution of discharge from the mill and waste water treatment plant over a short stretch of Millinocket Stream upstream of where the BC enters Shad Pond. We do not address this legal issue in the EIS because flow release in the BC is not warranted due to lack of biological/fisheries benefits and the cost in lost power benefits.




Office of Hydropower Licensing

September 1996

FERC/FEIS-0075

FINAL ENVIRONMENTAL IMPACT STATEMENT Upper Penobscot River Basin Maine

Volume 2 - Responses To Comments on DEIS



Ripogenus Hydroelectric Project (FERC No. 2572) – 05 Penobscot Mills Hydroelectric Project (FERC No. 2458) – 00

FEDERAL ENERGY REGULATORY COMMISSION OFFICE OF HYDROPOWER LICENSING

FINAL ENVIRONMENTAL IMPACT STATEMENT VOLUME 2 APPENDIX E RESPONSES TO COMMENTS

LICENSING TWO EXISTING HYDROELECTRIC PROJECTS IN THE UPPER PENOBSCOT RIVER BASIN

FERC Project Nos.

2572 Ripogenus 2458 Penobscot Mills

Applicant:

Great Northern Paper, Inc.

Additional copies of the FEIS are available from:

Federal Energy Regulatory Commission Public Reference and Files Maintenance Branch 888 First Street, N.E., Room 2A Washington, DC 20426

September 1996

FEDERAL ENERGY REGULATORY COMMISSION Washington, DC 20426

TO THE AGENCY OR INDIVIDUAL ADDRESSED

Attached is the Final Environmental Impact Statement (FEIS) for relicensing the Ripogenus (FERC No. 2572) and Penobscot Mills (FERC No. 2458) hydroelectric projects, Maine. The FEIS is contained in two volumes. Volume 1 includes the text of the FEIS through Appendix D. Volume 2 is Appendix E, on the DEIS and Commission staff's responses to those comment. This FEIS was prepared pursuant to requirements of the National Environmental Policy Act (NEPA) and the Commission's regulations implementing NEPA (18 CFR Part 380).

The FEIS documents the views of government agencies, nongovernmental organizations, affected Indian tribes, the public, the license applicant, and the Commission's staff. It contains staff's recommendations about licensing the Ripogenus and Penobscot Mills projects in the upper Penobscot River Basin.

Any Commission order issued pursuant to this document will be subject to the Commission's rehearing process under 18 CFR Section 185.713. Requests for rehearing must be filed within 30 days of the date of issuance of the subject order.

Before the Commission makes a decision on relicensing these projects, it will take into account all concerns relevant to the public interest. This FEIS will be part of the record from which the Commission will make its decision.

Attachment

LETTERS OF COMMENT ON THE DRAFT ENVIRONMENTAL IMPACT STATEMENT AND FERC STAFF RESPONSES

The Notice of Availability of the draft environmental impact statement (DEIS) was published in the Federal Register on December 15, 1994. The DEIS was mailed to federal, state, and local agencies, and individuals for comments immediately prior to the public notice date.

All timely letters of comment that address specific analyses in the DEIS were reviewed by the FERC staff. Suggestions for correcting text or data and requests for further discussion of a subject have been given consideration. Those editorial changes and suggestions which were practicable, reasonable, and which improved the quality of the final environmental impact statement (FEIS) are incorporated herein. With some exceptions, as appropriate, attachments to comment letters have not been reproduced in the FEIS (Volume 1) because they don't provide specific commentary on the DEIS.

Constructive criticism presenting a major environmental point of view or one in opposition to the staff's, when persuasively supported, is treated by making revisions in the appropriate part of the FEIS. When the major point of view is not persuasive, reasons are given why the staff did not change its point of view in the space opposite the comment. The sections of the FEIS that have been modified as a result of comments that raised no questions concerning treatment of subject matter in the environmental impact statement have been identified in our responses.

The respondents and the page on which their letter occurs are:

RESPONDENT	DATE	<u>SYMBOL</u>	<u>PAGE</u>
Agency Commentors			
Conservation Intervenors	2/21/95	CI	E-1
American Rivers			
American Whitewater Affiliation			
Appalachian Mountain Club			
Conservation Law Foundation			
Maine Audubon Society			
Conservation Intervenors	3/15/95	CIMOU	E-49
Department of Interior	2/21/95	DOI	E-59
Environmental Protection Agency	2/21/95	EPA	E-101
Great Northern Paper	2/22/95	GNP	E-117

Angus King, Governor, and Donald McNeil, GNP	2/16/95	AKDM	E-187
Penobscot Indian Nation	2/21/95	PIN	E-193
Penobscot Indian Nation	3/10/95	PIN	E-219
Save Our Scenic Lakes Association	1/25/95		E-221
State of Maine, Planning Office	1/25/95		E-223
State of Maine, Planning Office	2/17/95	SPO	E-227
Town of Millinocket and ASSSET	2/21/95	T&AS	E-239
Trout Unlimited	2/22/95	τu	E-339
Non-agency Commentors			
Susan Dolan John Frachella Charles Gadzik Jim Haskell Nancy Johnson Lance Rogers James W. Sewall Sprague Energy Steve Tuckerman	1/31/95 2/1/95 2/9/95 1/25/95 2/16/95 2/6/95 1/31/95 1/25/95 1/31/95		E-349 E-350 E-351 E-354 E-361 E-369 E-369 E-371 E-373
Group - A			
John Benoit Gerald Bouffard Richard Campbell David Carpenter Dean Clukey	2/1/95 2/1/95 2/1/95 1/27/95 1/28/95		E-378 E-379 E-380 E-381 E-382
Norman Ferguson Charles Fisher Albert Gamache Ernest Greenlaw Dan Gwadosky	1/25/95 2/1/95 1/20/95 2/27/95 1/20/95 2/1/95		E-383 E-384 E-385 E-386 E-387 E-388
Charles Heino Robert Keam Priscilla Lane Willis Lord Rodney McElroy Judy Paradis	1/24/95 1/30/95 1/26/95 1/17/95 1/26/95 1/20/95		E-390 E-391 E-392 E-393 E-394 E-395
Thomas Poulin	1/30/95		E-396

Chester Rice	1/23/95	E-397
Robert Spear		E-398
Richard Stone	1/25/95	E-399
Joseph Taylor	1/20/95	E-400
Verdi Tripp	1/17/95	E-401
Robert Tufts	2/3/95	E-402
Marc Vique	1/17/95	E-403
Julie Winn	1/27/95	E-404
Croup - B		
<u>Group - p</u>		
Alex Agnew	1/25/95	E-407
Ann Badham	1/23/95	E-408
H.M. Bliss	1/23/95	E-410
Margery Blonder	2/1/95	E-412
Peter Boehmer	1/23/95	E-413
Philip Booth	1/23/95	E-414
Eileen Burnell	2/14/95	E-416
D.L. Caldwell	1/23/95	E-418
David Carle	2/2/95	E-420
Nancy Chandler	2/14/95	E-422
Colby Environmental Council	2/14/95	E-424
Richard and Lois Cole	1/26/95	E-425
Steven Corman	2/20/95	E-427
Leverett Davis	1/23/95	E-430
Sharon Drake	1/23/95	E-433
Stephen Drane	2/10/95	E-435
Frances Dunn	1/23/95	E-437
Robert Ewing	2/3/95	E-439
Virginia Hammond	1/26/95	E-441
Francis Hatch	1/30/95	E-442
Booth Hemingway	1/23/95	E-446
Horace Hildreth	1/25/95	E-448
Howard Jones	1/23/95	E-450
Charles Kitdin	1/23/95	E-450
Julie Khorana	2/7/95	E-452
Ernest and Coralie Kinney	2/21/95	E-434
	2/10/95	E-400
	1/23/95	E-409
Greenville Lloyd, Jr.	1/23/95	E-401
Kelly McClintock	2/10/95	E-403
Paul McPheters	1/25/95	E-404
Uenn Maguire	1/23/90	E-400 E 167
nugh wontgomery	1/23/90	E-40/ E-460
Edward Wyers Poter Nesson	1/24/30	E-409 E-470
Coordo O'Connoll	1/20/00	E-470 E_470
	1/20/30 2/15/05	E=4/2 E_170
Lucia anu James Uwen	2/10/50	E-4/3 E-475
VV dyne Fersons Cappia Pataraan	1/20/90 2/10/05	E-4/3 E_176
	2/13/33	C-4/0

Glenn Rampe	3/2/95	E-478
Barbara Rappaport	1/24/95	E-479
Norman Sims	2/20/95	E-480
Wickman Skinner	1/23/95	E-481
Eleanor Stenson	1/23/95	E-483
Philip Thompson	1/25/95	E-485
Kathy Winslow	2/10/95	E-487

.

Group - C

John Anderson	2/16/95	E-493
Nancy Anderson and Richard Rockefeller	1/25/95	E-494
Thad Austin	2/3/95	E-495
Bruce Bailey	1/29/95	E-496
Woodruffe Bartley	1/29/95	E-497
David Beaulieu	1/16/95	E-498
Linea Belmont	2/16/95	E-499
Mary Bernmier	2/16/95	E-500
Bernard Bienkowski	1/29/95	E-501
Frank Bolton	1/17/95	E-502
W. Bragton	1/31/95	E-504
Shelia A. Brain	2/16/95	E-505
Marjorie Briggs	1/17/95	E-506
Mrs. Frank Browning	2/18/95	E-507
Laura Burch	1/29/95	E-508
Douglass Campbell	1/17/95	E-509
David Carlisle	1/31/95	E-510
Peter and Norma Cesare	1/25/95	E-516
Kendall Chevalier	1/29/95	E-518
Michael S. Coffman	1/29/95	E-519
Randall Comber	1/17/95	E-521
Janet Comeau	2/16/95	E-522
Marian Comstock	1/27/95	E-532
Robert Cressey	1/29/95	E-525
Stephen Cullin	2/16/95	E-526
Timothy Cullin	1/29/95	E-527
Barbara Curtis	2/16/95	E-528
Alyce Cusson	1/29/95	E-529
Dayle Rollin Heating, Inc.	2/16/95	E-624
John and Charlotte Dilworth	1/29/95	E-531
Robert Dishon	1/27/95	E-532
Richard Doane	1/29/95	E-533
John Downing	1/29/95	E-534
Robert Erickson	2/16/95	E-535
Anthony Filauro	1/26/95	E-536
Eric Givens	2/13/95	E-537
David Guernsey	1/27/95	E-540
William Haggerty	1/23/95	E-542

Vernon Haines	1/23/95	E-543
Chester Hamm	2/16/95	E-544
Arthur Hansen	2/16/95	E-545
Karan Hansan	2/16/95	E-546
Kenneth Hansen	2/16/95	E-547
Melania Hansen	2/16/95	E-548
Michael Harney	1/29/95	E-549
Walter Heal	1/27/95	E-550
Dan and Dahorah Hoffses	1/29/95	E-553
Dorathy Jamieson	2/16/95	E-554
Ion Jamieson	2/16/95	E-555
Carolyn Johnson	2/16/95	E-556
Depaid Johnson	2/16/95	E-557
Donad Johnson	1/26/95	E-558
Dora Kelly	1/27/95	E-559
Kathy Kappasan	2/16/95	E-560
	2/16/95	E-561
Tem Lambert	1/29/95	E-562
Colon Lander	2/3/95	E-563
	1/20/95	E-565
Lawrence Lankhorst, Jr.	2/16/95	E-568
	2/16/95	E-569
	2/16/95	E 505
	2/10/95	E-571
Paul Iviain	1/22/05	E-577
Louisa P. & Joseph P. Malizia	1/23/95	E-572
Grace Maloney	1/29/95	E-573
Nichael Maloney	1/29/95	E-574 E-575
Richard and Lenore Maloney	1/29/95	E-575
	1/29/95	E-570
Harold Mason	2/2/05	E-572
Bert WCBurnie	2/2/90	E-570
	2/16/95	E-379
Maurice McLean	1/20/95	E-500 E-581
Robert and Charlene Weininger	2/16/05	E-583
John Michaud	2/10/95	E-503
	3/11/95	E-004
Roger Morin	2/16/95	E-360
Robert Mosca	1/27/95	E-367
Charles Paul	1/29/95	E-368
Beverly Pelletier	2/16/95	E-589
Thomas Pelletier	1/25/95	E-590
Suzanne Peneau	2/16/95	E-591
Susan Prait	2/16/95	E-592
Philip and Carol Ramu	1/31/95	E-593
Conrad Rice	2/14/95	E-594
John Rouleau	1/25/95	E-596
A.L. de Saint-Rat	2/1/95	E-597
Patrick Santerre and Celeste Bard	1/21/95	E-598
Tom Sawyer	3/2/95	E-599

•

Wayne Scarano	2/16/95	E-601
Charles Shorter	1/21/95	E-603
Raymond Skoski	1/29/95	E-604
Doc Simmons	1/29/95	E-607
Thomas Simmons	1/29/95	E-608
Carolyn Simone	1/31/95	E-605
Fred and Marianne Smith	1/29/95	E-609
Richard and Jean Smith	1/29/95	E-610
Sandra Smith	1/29/95	E-611
Robert Speed	2/20/95	E-612
Thomas and Patricia St. John	2/3/95	E-613
Betty Tessman and Sandra Davenport	1/26/95	E-623
Glen Van Deventer	1/23/95	E-615
Robert Van Deventer	1/23/95	E-614
William Van Deventer	1/23/95	E-530
Robert Wells	1/29/95	E-616
Perley Wheaton	2/16/95	E-617
Richard Wheaton, Jr.	1/31/95	E-618
JoAnn Whitehead	2/16/95	E-620
David Whorf	1/29/95	E-621

UNITED STATES OF AMERICA

Penobscot River Basin EIS:

Project Nos. 2458 and 2572

In re Applications of Bowater/) Great Northern Paper, Inc. for) New Hydropower Licenses,) Ripogenus and Penobscot Mills) Projects)



CONCENTS OF AMERICAN RIVERS, AMERICAN WHITEWATER AFFILIATION, APPALACHIAN NOUNTAIN CLUB, CONSERVATION LAW FOUNDATION AND MAIME AUDUBON SOCIETY ON DRAFT ENVIRONMENTAL INFACT STATEMENT

These comments on the Draft Environmental Impact Statement ("DEIS") are submitted by and on behalf of American Rivers, American Whitewater Affiliation, Appalachian Mountain Club, Conservation Law Foundation and Maine Audubon Society.¹

Preliminary Statement

While FERC deserves credit for preparing an EIS as required under NEPA and for appropriately expanding project boundaries as required by law, many aspects of the DEIS fail to comport with the requirements of NEPA and the Federal Power Act and render the draft EIS defective as an information tool in making important resource decisions. Recommended Alternative 2:

CI-2

CI-1

Cont

CI-1

- * allows for continued degradation of resources;
 * fails to give equal consideration to power and non-power
- values; * fails to address impacts in the basin; and
- * uncritically relies on GNP for its conclusions.

A major defect of the DEIS is its reliance on the applicant's financial and socioeconomic claims to limit meaningful

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-1 We have corrected the EIS and refer to the following organizations collectively as the Conservation Intervenors: American Rivers, American Whitewater Affiliation, Appalachian Mountain Club, Conservation Law Foundation and Maine Audubon Society.

No response required. We provide detailed responses to your comments below.

¹ The DEIS refers to our organizations in various ways. We assume that the references to "CLF" are intended to reflect the joint positions submitted by our five organizations. However, the listing of acronyms does not reflect the coalition nature of our intervenor group. For clarity, we suggest that the references in the text to CLF be modified to reflect the fact that these positions are on behalf of the Conservation Intervenors, and that the definitions set forth on page xi be amended to include definitions for AMC (Appalachian Mountain Club), AWA (American Whitewater Affiliation) and Conservation Intervenors (American Rivers, AMC, AWA, CLF and Maine Audubon Society).

environmental improvements. By any objective measure, the flow modifications and other requests reviewed in Alternative 1 are: reasonable, constitute no more than 5-7% of the value of these licenses in the aggregate and, notwithstanding the unsubstantiated claims by the applicant to the contrary, are affordable by this applicant without detrimental socioeconomic impact.

In making claims of economic doom, Bowater faces a credibility gap which undermines its claims. Bowater acquired Great Northern in 1991/1992 to fulfill a long term strategic investment plar valued at nearly \$1 billion. Yet, it asks the public to believe that energy cost increases equivalent to less than 1% of the price Bowater paid for GNP threatens its investment plan and could cause 15% of GNP's work force assigned to marginally profitable machines to be discharged.

These threats do not stand the test of common sense. Bowater is not some small, struggling company, as it would have the Commission believe:

* Bowater is one of the major pulp and paper companies in the US and Canada. It is the largest supplier of newsprint in the United States. It's net assets approach \$3 billion. Bowater's financial capability enabled it to recently install a \$100 million recovery boiler at its Calhoun, Tennessee mill, to construct a \$63 million deinking facility at its East Millinocket mill, to announce, last month, a \$3.8 million investment in a new pulp process in Millinocket -- and to spend millions of dollars in legal and consulting fees contesting virtually every legitimate request of resource agencies and intervenor groups.

* While like almost all paper companies, Bowater endured a rough recession in the last three years, the recession is defiantly over. Bowater's most recent quarterly report in October states that "after nearly three years of recessionary conditions, industry recovery in newsprint, pulp and coated groundwood products appears to be well underway." The quarterly reports net income of over \$10 million and pulp price increases of over 60%. The price of newsprint, Bowater's major product line, has escalated so dramatically since December that newspapers such as the Portland Press Herald are increasing prices. Bowater is raising prices for all its major products.

* These economic improvements are reflected in the nearly 60% increase of Bowater stock over the last year -- from

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-3

The staff finds, based on our most recent economic analysis, that there is no conclusive evidence that either the Applicant's Proposal, or both versions of Alternative Two, would adversely affect the competitive position of the GNP's mills. This is because these alternatives would produce small percent increases in annual power costs, small percent declines in annual power output, and ultimately, that these small changes would not translate into a significant increase in the cost of production at the two mills. The staff also concludes that the negative annual net benefits under Alternative One would be large enough such that the competitiveness of the two mills is likely to be adversely affected.

CI-3

CI-2

Cont

CI-3 Cont

CI-4

CI-5

\$20 share in early 1994 to over \$31 share this month.²

The inconsistencies in the company's assertions loom large. For example, in the area of energy efficiency as a replacement power source for flow modifications:

* GNP now claims to be a conservation leader: yet, has failed to capture the overwhelming bulk of conserved power at its facilities; only 1.5% of the conservation potential of the most likely area for efficiency improvements -- variable speed motor drives -- has been captured at the GNP facilities.

* GNP claims that "modernization" will increase electrical demand need, yet GNP has only invested in small amounts of selective demand side reduction; vast amounts of savings remain untapped; the company's claims conflict with the fact that merely in the replacement of groundwood pulp with recycled pulp, energy savings occurred.

* Bowater/GNP uses an arbitrary, internal corporate hurdle rate of return leaving most of the cost-effective conservation potential in the mills uncaptured. This may be fine for GNP's own purposes, but it is wholly inappropriate when the result is to exclude a costeffective source of replacement power when making decisions over public resources: FERC cannot allow GNP to waste the very hydropower it claims is so essential to employment and future investments.

On socioeconomic issues, Bowater/GNP claims that large job losses will occur if flows are provided in the Back Channel -- yet claims as confidential and refuses to share with any party including FERC the information which purportedly justifies this claim. On land issues, Bowater/GNP claims to be an outstanding steward of its land, yet in the last few months agreed to pay a penalty for some of the most egregious violations of state land use regulations seen in recent years.

Bowater/GNP has not hesitated to politicize the process by turning to state and federal legislators to change state laws or to

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-4 Energy conservation was identified as an alternative during the scoping process. However, we concluded in the DEIS that energy savings gained through conservation have largely been offset by increased energy demand from GNP's plant modernization efforts. This conclusion was based upon our review of GNP's conservation and modernization programs, as documented in Exhibit H, and the memorandum of Owen Merrill of GNP, both of which are based upon actual plant data. The claim that there is an enormous conserved power potential in the GNP facilities by Conservation Intervenors, on the other hand, is based upon generic or theoretical evidence, which staff did not accord the same weight. The staff also notes that many of the arguments posed by the intervenors have been raised throughout this proceeding and have been addressed by the Applicant.

CI-5 Opinion noted.

2

μ

² The turnaround at Bowater is consistent with other paper companies doing business in Maine, including Boise Cascade (first quarterly profit in several years; stock price up nearly 60%); Champion International (stock price up 50% in one year); Scott Paper (stock price almost doubled in the past year); and International Paper (stock price up 33% in the past year). The long-term resource decisions addressed in the DEIS should not be determined on the basis of transitory economic conditions now past.

influence FERC's environmental review. Bowater's threats have exacerbated job anxieties in the local community arising from years of corporate ownership changes; it has not acted when it could to stop the flow of incorrect information about many of the issues raised in this proceeding, including most recently shoreland protection. The result has been to create a needlessly emotional environment over these licenses, particularly at the public meetings.

The FEIS must test and correct for Bowater's economic claims before the Commission can make informed, fair decisions on the future of the West Branch.

COMMENTS ON SPECIFIC PROVISIONS OF THE DEIS

I. Geographic Scope

As originally set forth in Scoping Document I, FERC proposed that the DEIS examine the impacts of the Ripogenus and Penobscot Mills project on the West Branch basin. The geographic scope of the EIS was narrowed in SD II to the area surrounding the Ripogenus and Penobscot Mills project. This has undermined the usefulness of the EIS in examining the environmental impacts of various alternatives on the basin. The GNP system has basin wide impacts: GNP is the sole owner and operator of the dozen or more unified hydropower projects in the West Branch. The refusal to examine the basinwide impacts of this basinwide hydropower system means that the EIS will fail to assess the true environmental impacts of these projects.

Specifically, the DEIS wholly fails to consider the impact on the upper basins that the proposed operations of these projects will have. The upper basin are drawn down so as to provide water quantity's to enable the applicant to operate Ripogenus and Penobscot Hills. Without the water quantity provided from upstream impoundments, the operations proposed by the applicant, as well as those set forth in Alternative 2, would not be possible. The impoundment draw downs and operations have significant environmental impact in dewatering wetlands and littoral zones and impacting fish and wildlife values. Yet, the DEIS is silent on the environmental impacts of the drawdowns at the upper basins.

The DEIS asserts that a cumulative impact review upstream is not needed because the upper basins provide 19% of the total water quantity in the hydropower system. (DEIS at 3-5.) We submit that this contribution is significant. Without this water, power generation at the Ripogenus and Penobscot Mills projects would not be possible.

The omission of review of the upper storage area also undermines the range of reasonable alternative management modes in meeting the power and non-power needs of the system. By excluding

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

2

CI-6

The upper storage projects are included in GNP's water use model and the alternatives evaluated with that model. including additional model runs requested of GNP by FERC staff. These projects are not up for relicensing now but are included in the water use model as a combined input. This input was kept constant in developing the water use plan although the input could be varied in the model. We saw no need to do this however, since no one has suggested alternative management schemes for the upstream projects that would provide additional downstream benefits. In addition, GNP has agreed to inclusion of a reopener to allow modification of the water use plan, should analysis of the upper projects during relicensing result in unanticipated findings that desired changes could be made. We recommend the orders for both the Ripogenus and Penobscot Mills projects include an article containing a reopener clause for consideration of changes in water use in the West Branch when the upper storage projects are evaluated for relicensing.

CI-5 Cont

C1-6

5

consideration of the upper basins, the DEIS treats the Ripogenus CI-7 and Penobscot Mills projects in isolation for purposes of water quantity and power generation. Carving up the system in this manner denies the real world nature of the basin's ecosystem and the unified operations of the hydrosystem.

In lieu of a basin analysis, SD II declared FERC's intent to prepare a rigorous analysis of direct and cumulative impacts of specific resource issues. This approach provides far less value than a basin analysis. Yet, the DEIS fails to examine properly cumulative and direct impacts as outlined in SD II. For example, the DEIS fails to examine the decline of the brook trout fishery in the region as a cumulative impact of hydro operations. Nor does the DEIS examine the cumulative impacts of the hydropower system on landlocked salmon and Atlantic salmon at upstream <u>or</u> downstream projects excluded from the review in the DEIS, even though these species are of great importance to the region.

Finally, the DEIS does not fulfill the requirement that FERC examine past as well as present and future impacts. There is little discussion of the losses of indigenous species, such as brook trout, in the project area and cumulatively in the basin. In discussing flow needs in areas like the Back Channel, the DEIS compares the potential fishery with flow improvements to existing conditions, even though existing conditions are the result of damage to the ecosystem caused by hydropower operations, such as flow diversions. This failure to examine past conditions should be corrected in the FEIS.

The DEIS fails to review the past, present and future impacts from the proposed operations of the Ripogenus and Penobscot Mills projects on the upper basins. The DEIS is therefore inadequate in examining the cumulative and direct impacts of the operations of these projects on the upper storage basins. The FEIS needs to examine these impacts in order to fulfill NEPA requirements.

II. Expansion of Project Shoreland Boundaries

The DEIS appropriately recognizes that project boundaries must be expanded to include shoreland set back and vegetation zones in fee or easement form to protect non-power values in this extraordinary area of Maine. The need for shoreland buffer zones within project boundaries is essential in fulfilling the commission's statutory obligation to protect the aesthetic, recreational and ecological values of the project areas.

A. Penobscot Mills

Alternative 2 calls for 200' non-development zones and 100' vegetation zones on GNP land only in the project area. Alternative 2 also allows Bowater/GNP to develop a Shoreland Management Plan ("SMP") in lieu of the set backs.

Opinions noted. FERC has not changed its definition of baseline and no-action. Baseline conditions continue to be existing conditions, not pre-project conditions.

The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff proposes, for the Ripogenus project area, two options: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion on GNP-owned lands. For the Penobscot Mills project area, the staff recommends a 200 foot expansion of the project boundaries on GNP owned lands. The staff recommends that existing structures would be grandfathered under any of the project boundary expansion alternatives. See section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.

CI-6

Cont

CI-7

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

6

We agree that the buffer and vegetation zones should only apply to Bowater/GNP land in the Penobscot Mills project. However:

1) The SMP should not apply to all areas of the Penobscot Mills project. The SMP option should apply to only those areas of the project area which are already developed and where proposed allowance for future additional development should be clustered. Areas in the project area which are currently entirely undeveloped and exhibit back country characteristics which should be preserved. These areas should have the set back and vegetation zones installed without allowing for diminution of the zones.

2) The geographic scope of the area covered by the zones should be clarified. Attachment A sets forth recommended boundaries.

B. <u>Ripogenus</u>

Unlike portions of the Penobscot Mills project, the Ripogenus project area is marked by only minimal existing development outside of Chesuncook Village. It is the absence of development on the overwhelming majority of shoreland that makes the non-power values of this project so high. For that reason, there is a need and opportunity, as expressed in the DEIS, to protect the entire shoreland of the project area that is currently undeveloped. In making this distinction between the Penobscot Mills and Ripogenus Projects, the DEIS has carved a rationale approach to the somewhat differing characteristics of the two projects.

The need to protect as much of the undeveloped shoreline of the Ripogenus Project area is clear. If the 1/3 of shoreland not owned by Bowater/GNP is developed over the next 30 years, many of the non-power values of the entire area will be severely degraded. Accordingly, the DEIS recommendation that Bowater/GNP protect its own land and, in addition, seek easements or fee ownership on land it does not own reflects the needs of this project area.

The DEIS is relatively silent in specifying where project boundaries would be extended. Attachment A describes recommended boundaries.

A number of concerns have been raised about the recommendation to acquire easements or fee interest on land not owned by Bowater/GNP around Ripogenus. Some of these concerns are unfounded, others can be adequately addressed and still maintain the integrity of the set back and vegetation zones.

1) A representative of the State expressed concern that State of Maine land in the Ripogenus project area not be included in the boundary expansion. We believe that it is clear from the DEIS itself, as well as the entire record, that public lands are not part of the recommended project boundary expansion or buffer zones.

CI-8

Cont

The FEIS should, apparently, make this clearer.

2) Concerns were raised at the public hearing that some shorefront owners may be unwilling to sell in fee or nondevelopment easements to Great Northern. An unwilling seller, however, can only be known if an offer is made and then refused. There are many reasons why a shoreland owner would be a willing seller of an easement of even fee: the long-term interests of the owners in a protected reservoir would be maintained and the owner would received monetary compensation.

Bowater should be required to make offers to non-GNP landowners in the project area. If owners appear who prove unwilling sellers, than Bowater/GNP should document the refusals. Bowater/GNP should be required to establish an escrow account to hold monies for future acquisition in the event the shorefront land becomes available.

Acreage Estimates and Land Valuation

We are unable to verify the estimates in the DEIS that 2000 acres of land not owned by the applicant would need to be acquired or protected through conservation easements in order to expand the project boundaries. At 200' setbacks, this estimate is equivalent to 82.5 miles of shoreline. By contrast, the Ripogenus application at Table A-7 states that the shoreline of the project is 118.3 miles. If correct, and Bowater/GNP owns 67% of the shoreland (DEIS at 4-57), then 33% or approximately 39 miles, of non-GNP land is involved. A 200' project boundary expansion on 39 miles would incorporate approximately 945 acres of land, not 2000 acres. A 500' project boundary expansion would require the project boundaries to incorporate approximately 2,364 acres of land, roughly consistent with the amount of acreage deemed affordable by the DEIS in Alternative 2.

In terms of land values, using the Rangeley Lakes example of \$1000/acre is a reasonable surrogate but may result in higher cost estimates than actual market purchase or easement agreements due to the fact that the Rangeley Lakes region, as with the Moosehead Lake area, has experienced significantly greater development pressures and real estate activity than the Upper Penobscot Region. Nonetheless, the valuations estimated in the DEIS appear to inflate the value of backland over shoreland. The DEIS assumes that all acreage, whether in the 200' zone or the 500' zone, is equivalent in value. The DEIS estimates that a 200' project boundary set back would cost \$2 million for 2000 acres and that a 500' boundary set back would cost an additional \$5 million for Ripogenus, based on 5000 additional acres, and \$4.5 million for Penobscot Mills, based on 4500 additional acres. (DEIS at 4-60.) These values indicate that the land from the 201 foot mark to the 500 foot mark would cost \$7.5 million, or nearly four times as much as the first 200 feet. Shorefront land is generally worth more than back land.

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

- CI-9 Under the revised land use alternatives, GNP would obtain conservation easements on non-GNP owned land and incur easement acquisition costs only under Alternative 1 (see section 4.9.2).
- CI-10 The staff reviewed various cost estimates and appraisals to determine the potential effect of GNP acquiring shoreline conservation easements for the non-GNP owned lands under Alternative 1. Based on this review, staff estimated the 'potential costs for acquisition of conservation easements to be approximately \$24.6 million (see section 4.12.2).

CI-8 Cont

CI-10

CI-10 Cont Accordingly, it is likely that the estimate in the DEIS inflated the cost of a 500' buffer zone.

C. Issues Applicable to Both Projects

Easement Terms

The terms set forth in the Upper Androscoggin EIS referred to in the DEIS would allow certain activities to occur in the zones which would frustrate the purpose of the zones. Instead, terms along the lines of those set forth in the Deerfield River Offer of Settlement filed with FERC, Docket No. LP 2323-012.

Buffer Zone Widths

The DEIS rejects 500' set back widths for two reasons: cost for non-Bowater land in the Ripogenus Project and the conclusion that the additional ecological protection gained with 500' zones is not worth the cost.

We disagree with that view. First, as noted above, the costs for 500' zones should not be significantly higher than for 200' zones. Second, it is clear that the 500' zones provide a minimum level of protection for non-power values. That is why other nondevelopment zones in the West Branch area and elsewhere in northern Maine have been set at 500'. This includes the 1981 500' wide easement given by GNP to the State of Maine on the West Branch itself. The Allagash non-development buffers are also 500', reflecting the minimum widths that are considered necessary to provide scenic and ecological protection.

Accordingly, the 500' width identified in Alternative 1 should be adopted in the FEIS, with the exception of lands which fall within the SMP option in the Penobscot Mills project.

Responding to Concerns Raised at the Hearing

Several concerns regarding the recommendation for appropriate buffer and vegetation zones have been raised at the public hearing in Millinocket. Despite the assurances given by FERC personnel at the hearing, we expect that written comments from certain entities will continue to express these concerns.

Clarification on Exemptions

Concerns have been expressed by certain entities regarding the impact of the set back and vegetation zones on existing campowners. FEIS should incorporate the clarifications made by FERC personnel at the start of and during the public hearing, namely, that:

- i) Existing structures are grandfathered;
- ii) There will be no need for campowners to obtain FERC

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

- CI-11 The staff revised the location and terms of the proposed easements in the FEIS from those proposed in the DEIS. These revisions reflect assessment of GNP's proposed easements and further review of available information pertaining to this issue (see section 4.9).
- CI-12 Our analysis (see section 4.9) indicates that the additional protection of the 500-foot expansion does not merit the much higher cost of that alternative (\$24.6 million versus no direct costs in the other alternatives).
- CI-13 The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). Within the Ripogenus Project area, the recommended alternative proposes two options: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion GNP-owned lands. Within the Penobscot Mills Project area, the recommended alternative proposes expansion of the project boundary on GNP-owned lands to 200-feet from the high water mark of the impoundments. We also recommend development of a Shoreline Management Plan for the boundary expansion areas in consultation with various resource agencies. See section 4.9.3 of the FEIS for further discussion regarding the land use regulations and management of lands within the proposed boundary expansion area.

m

÷

CI-11

CI-12

CI-13 Cont

CI-14

approval for bedroom expansion or other modifications to existing structures;

iii) There is no prohibition on Bowater/GNP electing to sell leased land; if a structure exists on the leased land to be sold, it will be grandfathered. If no structure exists, then the set backs would apply.

Economic Development

It is most likely that the expansion of project boundaries will act to maintain the current aesthetic and environmental quality of the area and protect the attractiveness of the region. Claims that the modest boundary expansion will harm the economic development of the area are unfounded: the zones do not prohibit additional development, but merely establish minimum set back requirements. The SMP for Penobscot Mills can allow for additional development in appropriate areas. All of the land in which the boundaries are to be expanded in the Penobscot Mills projects are owned by GNP; GNP has had a non-binding moratorium in place for several years. It is difficult to understand how the project expansion would be economically harmful compared to the non-binding moratorium on any additional leasing in place over the last years.

Landowner Stewardship

Some have expressed the view that existing landowners have been good stewards of the shoreland resources and have been responsible for the lack of extensive development on these waters.

We note that even as it prides its land management, Bowater/GNP recently admitted to a series of land use regulation violations spanning some 14 years, including on Third Debsconeag, a Class 1-A lake. (Information about these violations is attached.) The violations include construction of roads without permits; construction of roads in non-development zones; violations of timber harvesting rules; violations of regulation intended to protect high value lakes; and other violations. Bowater/GNP has acknowledged its embarrassment at these violations, and has agreed to take some remedial action, but the fact is that GNP has admitted to some of the most egregious violations in recent memory.³

More importantly, the public interest in protecting non-power values is not fulfilled by relying on the inclinations of landowners, including Bowater/GNP. Without long term protection, economic and other pressures could act to change the minds of landowners. Without project boundary expansion, Bowater could

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-14 See responses CI-12 and CI-13.

0

CI-15 Staff acknowledge that land use regulations and land ownership could change over the term of the license. The recommended alternative (Alternative 2) includes measures to ensure protection of shoreland areas for the term of the license. For the Ripogenus Project area, Alternative 2 proposes the adoption of GNP's proposed conservation easements for the term of the license or 200-foot boundary expansion on GNP-owned lands for the proposed easement area. For the Penobscot Mills Project area, the recommended alternative proposes the expansion of the project boundary along approximately 70% of the project's shorelines (see section 4.9.3).

³ Importantly, the violations were brought to light by a third party; due to budget cutbacks, LURC's enforcement staff has been virtually eliminated, underscoring one of the uncertainties of relying on a zoning body for long-term protection.

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

10

elect to sell additional lots or land (as it has recently sold significant amounts of land in Maine and Canada); Bowater could sell GNP; Bowater could be acquired through a hostile tender offer. No more proof of the uncertainties of the future exists than the fact that there have been three owners of these assets in the last few years alone, the Bowater acquisition from Georgia-Pacific being the most recent change in corporate ownership; the hostile tender of Great Northern Nekoosa by G-P is evidence of the ownership changes that can occur.

LURC Sufficiency and Federal/State Jurisdiction

Many comments reiterated Bowater/GNP's position that state zoning regulations are sufficient to provide long-term protection as required by the Federal Power Act and implementing regulations. Without repeating our prior arguments, it is important to note that no party or commentator provided any rebuttal of the factual information submitted by the Coalition that thousands of units could theoretically be built; that the areas most susceptible to development are the shoreland zones of high value waters; that the purpose of LURC zones is to control and monitor development over time, not provide the same level of long-term protection that expanded project boundaries provide.

These comments also confuse the issues. If zoning was adequate for long-term protection, then there was no need for GNP to grant easements to the state on the West Branch, East Branch and portions of Lobster Lake in 1981. There would be no need for the important efforts of land trusts to acquire high value land in northern Maine. There would be no need for the Land for Maine's Future Board and the Forest Legacy program. All of these activities exist in areas where there is current LURC zoning, but there is an obvious recognition that easements and fee ownership provide a much more significant and necessary form of long-term protection.

As for references to State rights, the question here involves longstanding federal obligations on Bowater/GNP as a hydropower developer and owner. The State makes its positions known through comprehensive plans filed with FERC and through water quality certification. Maine has an interest in public use of waterways and there is no inconsistency between the expansion of project boundaries and any Comprehensive Plan.⁴

CI-16 Further staff review of LURC land use regulatory requirements resulted in revised land use assessment and recommendations (see section 4.9). The staff proposes, for the Ripogenus project area, two options: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion on GNP-owned lands. For the Penobscot Mills project area, the staff recommends a 200-foot expansion of the project boundaries on GNP owned lands. The staff recommends that existing structures would be grandfathered under any of the project boundary expansion alternatives. Under the recommended alternative, the proposed conservation easements and boundary expansion would provide long term protection of valuable shoreland resources (see section 4.9.3).

CI-16

⁴ Although the applicant now relies on LURC regulations to avoid its shoreline protection obligations, Bowater/GNP opposed consideration of LURC shoreland standards during the Ripogenus water guality certification process and the State explicitly refused to consider shoreland issues in the Ripogenus Project. (See LURC Decision, Dec. 23, 1992 at 3.)

Bowater/GNP Alternative Shoreland Proposals

Perhaps Bowater/GNP will accept the recommendations of Alternative 2 in its comments. We would expect, however, that Bowater/GNP requested an extension of time to comment in order to prepare an alternative to the recommendations in the DEIS. We would expect Bowater/GNP to file comments, and organize comments by others, claiming that the easement acquisition in the Ripogenus Project will cost a multiple of the estimate in the DEIS, and suggest a less costly mechanism to ostensibly achieve the goals of the boundary expansion.

FERC should look upon such information submitted and any alternative proposed by Bowater with great skepticism. The recommendation in the DEIS is extremely low in cost: there is no cost for the boundary expansion in the Penobscot Mills project. The only cost in the Ripogenus Project is on non-GNP lands, and is modest. Most importantly, expansion of project boundaries through easements and fee is the only legally <u>binding</u> mechanism that will protect non-power values. Commitments to maintain the moratorium on new leases, for example, are not binding. Nor should FERC accept some expansion of the SMP concept to the Ripogenus Project. The SMP option should be strictly limited to portions of the Penobscot Mills project: the SMP has the potential to undermine the goals of project boundary expansion.

II. Water Use Issues and Replacement Power Costs

As the DEIS points out, the central issue in this proceeding is the question of flows in the Back Channel. In addition, there are continuing concerns by our organizations with regard to the recommendations in Alternative 2 on summer recreational flows in the Upper Gorge, winter flows in Millinocket Stream and late summer evening flows below McKay Station.

1. Flow Recommendations

a) Back Channel: Habitat and Recreational Potential

The staff recommends that the flow recommendation of Alternative 1, 350cfs in the Back Channel, be rejected for three reasons: that the habitat value will not be significant; that the costs of replacement power will be high; and that the socioeconomic impacts on employment in the local area could be significant and would outweigh any habitat gain. The DEIS's view on each of these points is erroneous and based on inadequate information.

The DEIS should have considered a range of flow options beyond 350cfs. The DEIS excluded consideration of the environmental and recreational values of higher flows for economic reasons. That exclusion is unwarranted. Data in the record indicates that flows as high as 945cfs could provide significant benefits. The

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

11

CI-17 See response CI-8.

CI-18 Responses to specific comments are provided below.

CI-19 We did not evaluate flows greater than 350 cfs to the Back Channel since there would be few fish produced in these areas even with this level of flow, relative to the costs due to lost power and within the context of fisheries within the region. The 945 cfs recommended by Interior is the aquatic baseflow value to be used when site-specific information is not available. Since a site-specific IFIM study was conducted with resource-agency participation, the ABF flow is irrelevant. The IFIM study indicated a maximum habitat value in the range of 350-500 cfs. There is no evidence in the record suggesting that flows greater than this range would provide greater benefit. We used the lower end of that range to examine flow feasibility and economic consequences relative to the environmental benefit. We found that the environmental benefit was not worth the cost in terms of lost power and therefore did not consider higher flows, since most of the environmental benefit is obtained in the first 350 cfs.

CI-18

CI-17

COMMENTS FROM CONSERVATION INTERVENORS

ON UPPER PENOBSCOT RIVER BASIN DEIS

12

CI-19

CI-20

Cont -

ecological values of flows in ranges higher than 350 cfs should be reviewed and considered in the FEIS.

The DEIS argues that the habitat which might be created by minimum year round flows of 165 or 350 cfs would be marginal and that there is ample existing fishery and recreational opportunities in the West Branch. The view set forth in the DEIS conflicts with that of expert resource agencies, particularly the U.S. Fish & Wildlife Service, private fishery advocates and conservation and recreational organizations. Alternative 2 sacrifices the potential to restore both the aquatic integrity and cold water fisheries that historically thrived, the cultural value of a restored Back Channel to the Penobscot Indian Nation, and the recreational opportunities in this reach. PIN has requested flows sufficient to meet tribal cultural needs for canoe passage into the headwaters of the West Branch of the Penobscot, as well as to protect the future possibility of the return of anadromous fish runs to which they have treaty rights.

The Back Channel Has Significant Habitat Potential

Because of its current flow regime, it is true that the Back Channel contains very little fish habitat. But as the DEIS points out, the potential habitat is extensive. (DEIS §3.5.2.12.) Studies have indicated that there is significant potential for improving fisheries by enhancing year round flows. An IFIM study indicated that enhanced flows could significantly increase habitat quantity for all life stages of landlocked salmon except adults. Additionally, as noted in the DEIS, "species other than salmon, such as brook trout and a variety of sucker species, would benefit from the flow increase as would the macroinvertebrate community." (DEIS §4.4.2.4.)

The US FWS, US EPA, Penobscot Indian Nation. Trout Unlimited and the Conservation Intervenors have all requested that flows sufficient to meet fishery needs be provided in the Back Channel. The Maine Department of Inland Fisheries and Wildlife original position was that increased flows in both the Millinocket Stream and the Back Channel would significantly enhance salmon production in the Millinocket Stream-Back Channel-Dolby Reservoir reach of the river, which could be managed as a unit. No information has yet been presented to disprove this original position. There are no obvious biological or physical habitat conditions that would preclude a healthy salmon population from being established. (Penobscot Mills Application, Vol. IX at 193.)

The DEIS correctly acknowledges that there is sufficient water to provide a fishery flow in the Back Channel:

We [FERC] disagree with GNP's contention, therefore, and conclude that sufficient water is available to attain flow releases within other water management constraints

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-20 The staff reviewed available information regarding the Penobscot Indian Nation's claims to lands and rights within the Branches of the Penobscot River and concurs with the decision of the Maine State Department of the Attorney General (see section 4.11.1.2). The staff finds that consideration of the Penobscot Indian Nation's traditional practices within the project area are outside the scope of Section 106 review. The staff, therefore does not recommend the Penobscot Indian Nation to be included as a concurring party to the Programmatic Agreement associated with the Ripogenus and Penobscot Mills projects. The staff acknowledges, however, the interest of Penobscot Indian Nation in the management of historic properties potentially eligible for listing in the National Register of Historic Places within the project areas. Accordingly, the staff recommends the Penobscot Indian Nation be consulted during the development of the revised Cultural Resource Management Plans for the Penobscot Mills and Ripogenus Projects. Regarding anadromous fish, no state or federal plans identify restoration in the West Branch as being a goal or objective anytime in the future. However, Interior has reserved authority to issue Section 18 prescriptions for the projects. This issue can be addressed in the event that anadromous fish restoration is planned for the project area.

CI-21 We agree that there is habitat potential in the Back Channel. However, within the context of similar available habitat in the region, there is little unique habitat in the Back Channel that warrants the cost in lost power and socioeconomic benefits. that would result in requiring minimum flows there. See also our response to PIN-35.

E-12

13

(with some slight modification) under all alternatives.

(DEIS at 5-7.) The DEIS, however, then dismisses the US FWS recommended minimum flow of 500 cfs in the Back Channel on the unsubstantiated grounds of cost without properly evaluating the effectiveness of a 500cfs flow to protect and improve aquatic resources.

The DEIS asserts that the US FWS recommended flow of 350 to 500 cfs would not provide sufficient habitat for a meaningful salmon fishery because:

- (1) Grand Falls impedes fish passage, allowing for limited habitat:
- (2) predation of young salmon by bass and pickerel; and
- (3) the warm waters in Shad and Dolby Pond are sub-optimal for salmon.

At very low flows it is true that Grand Falls would impede fish passage. But the DEIS's conclusion that Grand Falls is impassable to salwon is false, is not supported by the US Fish and Wildlife Service, is contradicted by the historic record; and rather is based on field observations during unnaturally low flow conditions.

Historically, natural flows over Grand Falls provided sufficient flow to permit salmonid passage and migration well upriver of the Back Channel. Records depict accounts of salmon traversing Grand Falls and migrating to the tributary spawning streams of the Ripogenus Impoundment. (See "US Commission of Fish and Fisheries Report of the Commissioner for 1872 and 1873, An Inquiry into the Decrease of the Food-fishes with Supplementary Papers," Washington: Government Printing Office at 394 (1874). attached.) This report notes that "the West Branch gradually changes its character...and the uniformity of its current is interrupted by numerous falls and extensive lakes; but there is no natural obstacles to the ascent of salmon throughout the entire length; and the dams at North Twin and Chesuncook do not wholly prevent salmon reaching the upper waters. At both these dams they are frequently seen and sometimes caught." In fact, Grand Falls offers the best of both worlds in that sufficient flows would provide cold water fish passage, while inhibiting the passage of warm water species such as the smallmouth bass.

The assumption that competition with smallmouth bass would prevent a successful cold water fisheries below Grand Falls is also unproven. Smallmouth bass and salmonid fisheries co-exist in the same river reaches in many areas in Maine, including on the mainstem of the Kennebec River, where 37% of the salmonid catch in the reach between the Wyman Reservoir and the Williams Projects are brook trout; smallmouth bass inhabit this reach. (Wyman Application, FERC No. 2329, v. 1 §E3.1-2.) Smallmouth bass were

CI-21 Cont

F-13

14

illegally introduced into Moosehead Lake in 1974: this has not eliminated the salmonid fisheries in the East and West Outlets of Moosehead Lake which have high quality brook trout and landlocked salmon fisheries. (Kennebec River Resource Management Plan, Maine State Planning Office, Feb. 1993.)

The amount of spawning and incubation habitat in the Back Channel is underestimated in the application. The IFIM model used by the applicant treats substrate as a static parameter (Penobscot Mills Vol. X, App. Q, p. 24) and assumes that no changes in substrate would occur when discharges from Stone Dam increased. Almost 100 years of dewatering have degraded potential spawning areas, with finer particles in the dry riverbed being currently bound in the root systems of encroaching riparian vegetation. Provisions for continuous, adequate flows in the Back Channel would eliminate the encroaching riparian vegetation in the riverbed and these finer substrates would eventually be released, thereby opening up far more salmon spawning habitat in the Back Channel reach. The applicant and the DEIS ignore this reality which needs to be remedied in the FEIS.

The assertion that only limited spawning and incubation habitat could exists for a salmonid fishery is also flawed for it assumes that downstream immigration would not be a significant factor. Downstream passage of salmon into the Back Channel already occurs when water passes over Stone Dam during high flow periods, which could be greatly enhanced with modification of the Stone Dam to permit downstream passage during minimum flow releases.

Moreover, the FEIS also needs to correct the unproven assumption in the DEIS that sub-optimal summer temperature conditions would inhibit the establishment of a guality cold water fisheries in Shad Pond, Dolby Pond and the Back Channel. Sufficient flows over Grand Falls would permit salmon to access pools in the upper 2.7 miles of the Back Channel, where cooler water temperatures with better oxygenation could provide summer riverine refuges. In fact the application (Penobscot Mills, Vol. XIII. AIR response #2. Table 3) shows that during test flow releases at Stone Dam going from 7 to 50 cfs, the water temperature in the Back Channel dropped from 20° C to a more optimal cold water fisheries 16.3° C, while the DO level increased by over 0.5 ppm for a period mean of 8.47 ppm (ibid., Table 1). At just 7 cfs the summertime dissolved oxygen content exceeded 7.2 ppm, well above the threshold for a cold water fisheries during the non-spawning season. The temperature reduction and DO enhancement effect should be even greater using the higher flows requested by the US FWS.

The DEIS also acknowledges that a rich forage base of smelt exists in Dolby Pond for salmon and brook trout: "In 1986, for example, the total smelt drift at Dolby between March and December was estimated at 20,370 pounds." (DEIS at 3-16.) Smelt drift below the Ripogenus Project where a very high quality salmon

CI-21

Cont

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

15

- CI-22 State and federal resource agencies had the opportunity during initial stage consultation to recommend species to be considered. The state and Interior chose not to consider brook trout in the Back Channel. Restoration of native species was not a goal established by MDIFW, which is the agency with management responsibility for the state's two inland waters. FERC is not a fisheries management agency but seeks only balance. We view among various public interests MDIFW management plans as reflecting the public fisheries interest, thus we did not consider brook trout in our analysis.
- CI-23 We stand by our statement that the project meets or exceeds the numerical standards, based on available information.

The legal status of the Back Channel relative to Maine's 401 WQC will be addressed in the order for the project.

CI-21 Cont fishery exists averages 28,085 pounds. Therefore from a food source perspective, in the Back Channel/Dolby Pond reach the potential for the production of a salmon and trout fishery harboring large sized, quality fish could be very substantial.

In addition to the potential for a salmon fishery, the DEIS acknowledges that "Although not addressed in the IFIM study. species other than salmon, such as brook trout and a variety of minnow and sucker species, would benefit from the flow increase, as would the macroinvertebrate community..." (DEIS at 4-28.) The FEIS should include a review and flow alternative for the Back Channel based on a brook trout rather than a salmon fishery. brook trout fishery would require less flow than an Atlantic salmon fishery. Brook trout are documented to already spawn in the tributaries including Constock Brook (Penobscot Mills Application Vol. IX, p. 160) to the Back Channel. To select the needed flows for a brook trout fishery, the IFIM field data already exists. and the computer simulations can be run using brook trout Habitat Suitability Indices (HSI), with little additional burden in time. Assumptions and corrections of improved spawning and incubation habitat due to improved substrate conditions should be made in such analyses, which were not done in the original IFIM runs for salmon.

The brook trout fishery in the region was the native fishery until it was decimated by the development and operation of the hydro facilities on the West Branch watershed. It is estimated that a flow range of approximately 165 cfs might meet the needs of a brook trout fishery. This flow range is slightly higher than the historic 7010 flow rate, hardly an economic burden on the applicant. The study should also verify if this flow range would meet the cultural needs for canoe passage by the Penobscot Indian Nation. Furthermore as noted in the DEIS, providing 165 cfs in the Back Channel is easily achievable: "Results of 165 cfs in Back Channel show little or no difference from the GNP Water Use Plan during the wet and average years (Appendix D, figures D-14 and D-15), and all other enhancements could be achieved as proposed" (DEIS at 4-6) and "Under Alternative 2, available water would be sufficient to provide all proposed flow enhancements , including flows up to 165 cfs in the Back Channel, except during very dry years." (DEIS at 4-8.)

The DEIS also admits that water quality in the Back Channel would also be improved by enhanced flows. "Any flow higher than leakage would increase flushing rate and aeration in Back Channel and, thus, protect against water quality degradation during worstcase summer and early fall conditions." (DEIS §4.3.3.3.)⁵

⁵ The DEIS accepts the Applicant's false conclusion (at page 2-14 of the DEIS) that the Back Channel does not have to meet water quality standards because it is not classified by the State. The DEIS incorrectly states (at 3-9): "Riverine segments of the

In short what prevents the entire Back Channel reach from providing spawning habitat and serving as a refuge from warm summer temperatures in Dolby and Shad Pond is the lack of water due to project operations. The FEIS needs to incorporate the full record and not selectively use information to draw false conclusions.⁶

Penobscot Mills Project all met or exceeded the respective water quality standards." This is incorrect. The DEIS fails to present the record and acknowledge that the state waived its rights to certify the Back Channel for water quality (Letter form Dean Marriott of ME DEP to Lois Cashell of FERC, April, 22, 1993), in large part because the leakage flow recommended by the Applicant would not meet Class C water quality standards, which require that

Discharges to Class C waters may cause some changes to aquatic life, provided that the receiving water shall be of sufficient quality to support all species of fish indigenous to the receiving waters and maintain the structure and function of the resident biological community.

One cannot rationally argue that a dryway can support indigenous fish or maintain the structure and function of the resident biological community. Furthermore the ME DEP's 1994 Water Quality Assessment for Maine lists the four mile stretch below Stone Dam as not attaining the aquatic life standard for classification (Class C) because of dewatering due to hydroelectric power generation. The State "passed the buck" due to pressure from Bowater/GNP; FERC, however, has a responsibility to give equal consideration to providing sufficient flows to meet aquatic and fishery needs in the Back Channel in the FEIS.

Furthermore, resolution of the dissolved oxygen (DO) deficit problem in Dolby Pond would greatly enhance the overall mix of habitat available in the Back Channel and Dolby Pond for brook trout and salmon. The DO problem Dolby Pond currently experiences due to excessive BOD loads from the mill, compounded by the presence of this impoundment used for hydroelectric generation, is a solvable problem during the term of the license. Both the State of Maine Water Quality Certification and the DEIS acknowledge the issue of a DO problem in Dolby Pond and its requirement to be remedied (see DEIS at page 4-12). The Company's pollution control facility is located just above Shad Pond and the effects of the treatment are basically not realized by the river until Dolby Pond. The ME DEP 401 Certification for this project acknowledges that corrective action may be needed with the next renewal of the Waste Discharge License for the Millinocket Mill. An option being the possible relocation of the applicant's minimum flow and discharge compliance site to avoid flow or discharge violations. This potential solution during the course of this hydro license would greatly enhance the water quality of Shad and Dolby Pond, providing additional quality habitat for a successful cold water fisheries.

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

16

CI-24 There is currently no evidence to show that project operations affect DO in Dolby Pond. DO problems resulting from BOD loads from the mill are not within FERC jurisdiction; this issue will be discussed further in the license order for the project.

CI-24

CI-23

Cont

17

High Recreational Demand Exists in the West Branch

The DEIS's conclusions that a fishery that may be created in the Back Channel would not be world class and therefore is of little value conflicts with other findings that the recreational resources of the West Branch region are reaching saturation point. The DEIS erroneously concludes that "High quality recreational fishing waters are abundant in the project region and that riverine habitat in Upper Gorge, Millinocket Stream and Back Channel (Alternative 1), is not required to meet increasing recreational fishing demand, even if the additional habitat CI-26 enhanced regional fishing stocks." (DEIS at xvii.)

This statement is inaccurate and contradicts the current condition where fishing pressure (and other recreational uses of the area) on existing resources is high and demand is growing. In fact, fishing pressure on existing resources is so high that the Maine DIFW is evaluating implementation of revised catch and length limits in the West Branch to protect existing fisheries from existing fishing pressure. Attached is ME DIFW Quality Fisheries Initiative Public Hearing testimony in Millinocket, ME, 12/2/94, which addresses the problem of overfishing and demand control.

The DEIS also fails to acknowledge and estimate fishery needs in the area throughout the term of the license. The evidence supports the view that increased pressure will be substantial. In 1993, use on the West Branch between Ripogenus Dam and Abol exceeded 10,000 days of fishing during the summer season, up from fewer than 5,000 days per summer in 1975 - a doubling in 18 years. (ME DIFW Quality Fisheries Initiative Public Hearing Testimony at 8.) This rate of increase over the term of the license would put the fishing pressure at upwards to 40,000 days in just that reach alone. The FEIS should describe the projected increased fishing pressure in the region over the term of the license and address ways to meet increased pressure through other fisheries, such as restore fish habitat degraded by hydro operations in the Back Channel. A Back Channel fishery, even if only of regional attraction, would be a significant enhancement to the area, one that may be essential given future demands on existing resources.

Finally, the DEIS erroneously states that no study of whitewater boating potential was requested on the Back Channel. despite the request having been made by several groups years ago. Specifically, the DEIS claims that no whitewater boating organization requested a study. That is incorrect. In our coalition's August 20, 1992 Motion to Intervene, American Whitewater Affiliation together with other members of the Conservation Coalition requested that such a study be performed. Moreover, in February 19, 1992, the letter of deficiency filed by AMC, American Rivers, CLF and Maine Audubon Society clearly requested that a study be performed on the recreational potential of the Back Channel which FERC rejected. These organizations

- CI-25 GNP's proposed actions would improve recreational fisheries in the Upper Gorge, West Branch and Millinocket Stream (see section 4.8.1.3). Staff determined that additional flows of 350 cfs in the Back Channel would provide no significant improvement to recreational fisheries, and that significant flows in the Back Channel could jeopardize other environmental enhancements (see section 4.8.2.3 and 4.4).
 - Back Channel would only provide an approximately 4.5-mile-long boating trip in a less aesthetically pleasing setting than along the West Branch. We conclude in Section 4.8.1.1 that neither commercial nor local private whitewater boating groups expressed interest in expanding whitewater boating opportunities in Back Channel, that significant opportunities exist within the project area to meet recreational boating demand, and that the significant flows that would be required to enable recreational boating activities would ieopardize other environmental enhancements.

E-17

18

include recreational users among their members; the application, record and now DEIS is deficient due to the failure to prepare a whitewater boating study in the Back Channel.

There is ample evidence to support the view that increased CI-27 recreational boating demand will occur on the West Branch over the course of the license period. The DEIS appears to accept Bowater/GNP's conclusions that recreational boating demand has peaked. (DEIS at 3-38 and 3-40.) We dispute the contention that canceing and whitewater recreational boating will decline in the future. National data show that there is a steady growth in these activities. According to the President's Commission on the Outdoors, paddlesports have grown 515% between 1960 and 1987. More recently, a 1992 <u>Cance Magazine</u> survey calculates a 33% growth in CI-28 paddlesports (cances, kayaks, rafts) since 1988. The growth rate is particularly strong among private boaters.

Applicant data the DEIS apparently relies upon to support the prediction of level demand is also questionable because the survey data took place during a deep and difficult recession in New England. As the national data cited above shows, rating in other US markets and in specialty niches continues to grow. The State Bureau of Parks and Recreation has identified the West Branch as the most intensively used, multiple use river in Maine.

With adequate summer flows, a whitewater boating opportunity could be developed in the Back Channel. The applicant has acknowledged the presence of Class III and IV rapids on this river stretch (Penobscot Mills Application at E5-32). The Appalachian Mountain Club Maine River Guide reports that Class IV rapids occur in the Back Channel during spillage at Stone Dam. The applicant's own studies indicate that "improvements in recreation opportunity" would occur with an increase in flows to the Back Channel to meet aquatic habitat requirements. (Penobscot Mills Application at E5-69.) Though neither Bowater/GNP or the DEIS adequately or fairly examine the recreational and cultural boating potential of the Back Channel, the FEIS should.

b) Millinocket Stream

As documented in our previous filings, the brook trout fishery in the West Branch of the Penobscot region has declined dramatically due to hydro operations. It is also a resource dwindling both in the state and region. The loss will only be felt more in the future as the demand for this resource continues to expand over the course of the license.

The DEIS analysis for the 7.9 mile Millinocket Stream is focused on providing salmon habitat (DEIS at 4-28) yet acknowledges that brook trout is the dominant and native game fish. (DEIS at 3-20.) The DEIS recommends that flows of 60 cfs and 30 cfs be maintained in Millinocket Stream from March 1 to September 15 and

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

•27 We agree that recreational boating demand will increase on the West Branch during the license period. The State of Maine's cap of 560 commercial passengers per day on the West Branch will limit recreational use. We do not believe, however, that the Back Channel would serve as an alternative to the West Branch.

-28 See response CI-26 and CI-27.

CI-29 Based on the IFIM study for Millinocket Stream, which included brook trout, a year-round flow of 60 to 80 cfs would provide optimal habitat for this species. Following a 10(j) meeting with the Department of the Interior, we have adopted a year-round flow of 60 cfs or inflow, whichever is less, for Millinocket Stream.

CI-28

CI-26

Cont

CI-27

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

July 1 through September 30.

Comment noted. Both the state of Maine in its 401 WQC for

the Bipogenus Project and the Department of the Interior have

accepted a seasonal flow of 100 cfs in the Upper Gorge from

19

from September 16 to April 30, respectively. Although this recommendation would provide some enhancement beyond that provided by GNP's proposal, the flows remain inadequate. In fact, the DEIS acknowledges that habitat values would be improved with higher, year round flows. As indicated in the DEIS, the 60 cfs winter flow would likely enhance overwintering populations of juvenile and adult life stages of salmon. The higher winter flow also "would protect redds from ice formation and, thus, improve the reproductive success of salmon in Millinocket stream." (DEIS 54.4.2.3.) The ABF that has been determined for Millinocket Stream is 37 cfs. (DEIS at §4.4.1.3.)

The DEIS recognizes that the Water Use Plan cannot differentiate the impact of winter flows for Millinocket Stream and concludes that a year round flow of 60 cfs "is feasible within the context of the of overall water use." (DEIS at 4-4.) The DEIS falsely assumes that the impact of increased flows for Millinocket Stream would be increased drawdowns in the reservoirs to supply equivalent amounts of power, but fails to acknowledge that energy conservation coupled with the shutdown of part of the Millinocket Mill in 1994 could provide for the needed year round flows in Millinocket Stream without further impacting the reservoirs. Year round flows of 60 to 80 cfs to reduce desiccation and freezing should greatly enhance the reproduction capabilities of this stream over existing conditions.

The rationale that summer water temperatures eliminate the possibility of a year round native brook trout population is entirely questionable. Salmon, which have a lower threshold for warm water temperatures and require higher flows than do brook trout, currently survive and reproduce in Millinocket Stream, as documented in the application. In fact the Bowater/GNP application defines insufficient water depths in Millinocket Stream as a major cold water fishery limiting factor. The increase in water depth from higher flows should also lower summer temperatures to more suitable levels. This is more than conjecture; in a Back Channel test when the flow was increased from 25 to 50 cfs, the water temperature dropped from an average 18.60 to 16.30 C. (Penobscot Mills, Vol. XIII, Air Response #2, Table 3.) The current flow in the Back Channel is 20 cfs, and less when water is "not available", while projected summer flows are now 60 cfs, indicating that similar drops in temperature are to be expected.

The FEIS should conduct IFIN analysis of flow needs for brook trout in Millinocket Stream, including spawning and incubation flow needs over the non-summer period in the range of 60 cfs or higher. The computer simulations can be easily run as the field data has already been collected. The final management of Millinocket Stream should be based on a year round native fishery, not a sub-standard and very seasonal put and take fishery.

c) <u>Upper Gorge</u>

CI-29 Cont

The Upper Gorge, the historical river channel between CI-31 Ripogenus Dam and McKay station, extends approximately 3,900 feet and now receives either minimal flows (very low flows) or spillage (very high flows). Under this flow management regime habitat is marginal; many of its pools provide suitable habitat for salmon and trout only at certain flows. (DEIS at §3.5.2.2.) Although a winter flow greater than leakage would protect these values, the proposed Holbrook spawning area as mitigation for inadequate winter flows will provide ecological benefits.

The summer flows recommended in the DEIS are adequate for habitat protection, but insufficient for recreational use, particularly private, whitewater boating. Alternative 2 recommends two weekends in May (9 hours/day) of flows between 1000 and 2000 ofs be provided. Optimal flows for whitewater boating cannot be determined based on existing information. The DEIS states that the "Upper Gorge has been rafted successfully at flows as low as 1100 ofs." (DEIS at 4-48.) During spillage events in 1994, however, hard-boats were able to navigate the gorge at flows thought to be well below the 1100 ofs range, as low as 400 ofs. This discrepancy could provide for additional days in which boating releases could be accommodated.

In order to determine the actual flows needed to provide boating flows in the gorge, the FEIS should require a postlicensing study, in conjunction with AWA and other interested parties. The recommendation in Alternative 2 of two weekend release events should be subject to change depending on the outcome of the post-licensing study, including dates and hours for flow releases. All such release events would be coordinated to avoid conflicts with fishery and habitat management goals.

d) <u>McKay Station</u>

The DEIS refers to the settlement agreement between GNP and the Maine Professional River Outfitters approving the agreement's flow regime below McKay Station. Although the minimum flows set forth in the agreement are acceptable to private boaters, the agreement addresses the needs of commercial boaters, not the private boating community. More varied flows, however, would provide opportunities for private boaters with diverse skills to utilize the resource.

Under the recommendation in the DEIS, flows are only provided for between 8am and 5pm, to provide optimal flows for commercial trips. Private boaters utilize this resource well past 5pm during summer months. In order to provide diverse flows for these users, the FEIS should require flows to be maintained at 1800 cfs by 5 pm until 9pm on weekdays, weekends and holidays throughout the whitewater season. These flows will provide several benefits. First, private boaters will be able to experience less crowded conditions, more diverse boating experiences and manageable river

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

20

- 31 See section 4.8.3.1. The staff recommends that GNP consult with whitewater interest groups regarding appropriate flows for the recommended releases in the West Branch during two weekends in May.
- CI-32 We are recommending that GNP notify AWA and local whitewater boating interests seven days in advance of planned spillage events during the month of May.
- CI-33 As stated in section 4.8.1, GNP would provide minimum flows from McKay Station during daytime hours. GNP usually maintains flows at or above 1800 cfs throughout the year, therefore, evening flows may be reduced or varied. The reduced or varied evening flows would provide additional opportunities for less experienced paddlers or those desiring a diversified experience.

CI-32

CI-30

Cont

21

1 levels (particularly at the Cribworks). This flow will allow short evening runs on a very popular section of the river. The DEIS states that "increased minimum flows for recreation are not likely to attract significant new numbers of customers." (DEIS at 4-48.) This statement may be true for commercial rafting, but not for private boaters. AWA members, who reflect a dramatically growing segment of the recreational boating population on the West Branch, would be greatly attracted to these manageable flows.

2. Replacement Power Costs for Flow Modifications

Scoping Document II stated that the DEIS would examine alternative source of replacement power to purchased power from Bangor-Hydro Electric Company, included energy conservation and other alternatives. Yet, the DEIS utterly fails to examine these alternatives in any reasonable degree of detail.

Squarely put, the issue is this: Bowater claims that the cost of replacing hydropower lost to year round flows in the Back Channel would be so costly that several paper machines which produced lightweight coated paper may be subject to closure, requiring the termination of 238 jobs. The cost estimate is based on purchased power from Bangor Hydro Electric at \$83/HWA, for a total estimated cost of \$900,000 to \$1.8 million year over the proposed water use plan at a year round flow of 350cfs, and an increment from \$1.4 to \$2.6 million at a flow of 500cfs.

Yet, the cost estimates used by Bowater/GNP are inflated. There are other power sources that could provide replacement electricity for less than half the cost estimated by Bowater and relied upon in the DEIS. FERC cannot balance power and non-power values with respect to the cost of flows in the Back Channel unless accurate information is obtained on the true <u>least-cost</u> source of replacement power. The DEIS simply fails to examine the least-cost alternative sources.

Energy Conservation

Energy conservation is a significant source of power available to utilities and large industrial facilities. Energy conservation is more than load management. It is a power source that can be obtained to avoid new power plant or hydroelectric construction; to avoid additional purchases of electricity; to meet growing demand; to bring off line existing power sources.

The question of conserved power as a replacement source is not limited to regional electric grid planning or whether an industrial facility such as GNP is investing at all in conservation. The question is one of replacement power needs when examining a flow recommendation sought by intervenor organizations and expert federal resource agencies.

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-34 The staff's analysis indicated that purchasing firm power from BHE is the least-cost source of alternative power. Table 2-9 of the FEIS shows that the cost of power production under the three alternatives would range between 34.1 mills/kWh under the Applicant's Proposal up to 38.67 mills/kWh under Alternative #1. We used a value of \$73.92/MWh in our analysis, which includes purchases of energy and capacity. Based on these figures, the preferred Alternative Two (final recommendation) would have a cost of power about half that of the least-cost alternative.

CI-35 See response CI-4.

CI-34

Ē

-2

CI-33

Cont

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

22

The amount of power that can be obtained through conservation or efficiency upgrades depends upon the current efficiency level of the industrial facility (or service territory, in the case of a utility) and avoided costs estimates. As explained in the attached affidavit of Paul Chernick, a key factor involved in assessing whether conserved power can be obtained in a cost-effective manner is the payback or "hurdle" rate used to estimate the time return on capital spent on the efficiency upgrade.

.

The evidence in the record in this proceeding indicates two key points:

* there is an enormous conserved power potential in the GNP facilities; and

* Bowater/GNP has failed to examine whether the efficiency potential could be captured to replace energy lost to flow improvements. Bowater/GNP greatly limits the amount of conserved power it chooses to obtain by requiring efficiency investments to pass an arbitrarily high payback rate which forecloses most of the costeffective conserved power potential in the facilities. As a result, Bowater's claim that modernization will only result in increased electrical demand -- a claim repeated without analysis in the DEIS -- is incorrect.

Amount of Conserved Power Potential

In 1985, when Great Northern sought to construct a new 40MW dam on the West Branch, expert testimony submitted by CLF indicated that the mills were so inefficient that as much power could be obtained at a fraction of the cost of the new dam if the company implemented an energy conservation program. In other words, conserved power was a viable, least-cost source of replacement to the proposed hydroelectric facility.⁷ Expert testimony identified

Ś

⁷ As Amory Lovins testified, "My analysis reveals that Great Northern has far more effective and higher-return investment opportunities than the Big "A" dam, and that these opportunities can meet the Company's present and long-term electrical needs at lower cost and higher reliability than the proposed dam. I shall show that the systematic and comprehensive use of proven, practical, cost-effective electricity-savings devices in Great Northern's mills...can save more electricity, at far lower cost, than Big "A" could provide. I believe that this "best-buys-first" approach can not only make it unnecessary and uneconomic to build the Big "A"; it can also improve Great Northern's competitiveness and profitability and hence contribute to the sound economic development of central Maine." Lovins Testimony at 6 (1985), excerpts of which are attached. Mr. Lovins' testimony was not rebutted successfully during those proceedings.

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

23

from 29.2 to 58.4 MW of conserved power potential at a cost of about 1 cent/kwh. (See Affidavit of Paul Chernick.) Bowater/GNP claims that it has invested in efficiency improvements at the East Millinocket mill, although as we have repeatedly pointed out, only about 6.5MW of power is identified as being captured. <u>CNP has</u> <u>identified virtually no conserved power at the Millinocket mill</u>.

The overwhelming use of electricity by GNP is to drive electric motors making that end use one of the single largest opportunities for obtaining conserved power is installing variable speed drives on alternating current (AC) motors. (See Affidavit of Steven Buchsbaum, attached.) After reviewing the memorandum of Owen Merrill of GNP, one point is crystal clear: GNP has 320,000 horsepower of connected AC motor load. Only 4.200 horsepower of this AC motor load has variable speed drives -- about 1.5%. The remaining motor load represents a <u>vast</u> source of conserved power potential. By all accounts in the record, the amount of Conserved power potential in the GNP facilities <u>now</u> vastly exceeds any power requirement for Back Channel flows. (See Affidavit of Steven Buchsbaum.)

The Question of Payback Rates

Determining the amount of the conserved power potential in the GNP facilities that is cost-effective to capture is dependent upon the payback criteria and costs of avoided power used; an investment is cost-effective if the energy saved pays for the investment in an acceptable period of time. If purchased power from Bangor-Hydro is the power source avoided through the conservation investment, then the avoided costs used in the calculation would be the cost of purchased power. We have repeatedly requested that FERC obtain information on the payback period used by Bowater/GNP; FERC has not obtained that information.

In the absence of specific information, we assume that Bowater/GNP is using a payback rate typical of other large industrial facilities, which is about 1-2 years. (See Affidavit of Paul Chernick.) Generally, corporations for internal purposes will not approve a capital investment in energy efficiency unless the cost of the investment can be recaptured in a brief period of time. The effect of these short payback periods leaves uncaptured substantial amounts of conserved power, cost-effective at a 3. 4 or 5 year payback period. Based on experience at other industrial facilities, it is likely that using a payback formula of even only four years, would capture at least 5% of the mills' electrical demand -- and more likely 10-20% of the demand, for more than is needed to not only implement a year round flow in the Back Channel, but to meet virtually all of the flow changes set forth in Alternative 1, including those GNP is willing to make in its proposed water use plan. (See Affidavit of Steven Buchsbaum.)

Without information from GNP on payback criteria, it is not

The staff is not able to comment on the financial criteria used by GNP in evaluating the feasibility of investing in conservation. We agree that longer analysis periods may be needed to capture all of the cost savings produced by a conservation project. GNP, operating as a private for-profit enterprise, would have a strong economic incentive to maximize savings from conservation and not waste electric power, particularly as it manufactures an energy-intensive product, one of whose major cost components is electric power. We agree that the proper economic approach to evaluating a conservation improvement is to compare the cost of making it with the value of the power saved as measured by the cost of obtaining it from the least-cost alternative. This is precisely the approach that the staff used in determining the annual net economic benefits of the four alternatives.

The reference to conservation as a "nonexclusionary" alternative to hydropower does not mean that conservation was not considered. Since the entire output from the two projects supplies only a portion of GNP's total annual power needs, the remaining needed power is obtained from sources that use fossil-fuels. As a result, conservation is used to displace higher-cost fossil fuel power. Conservation projects can be implemented in addition to obtaining power from a hydroelectric facility, as sufficient economic incentives would still exist to obtain conservation savings. This is particularly true where the least-cost power being displaced by conservation has a cost of \$73.92/MWh. Since conservation is not an alternative to hydropower, the least-cost source for GNP in evaluating conservation projects is purchasing replacement power from BHE.

CI-35 Cont

possible to assess accurately the cost of conserved power. Typically, though, conserved power can cost as little as \$10 or \$20/MWh. Even at a cost of \$35 - \$50/MWh, conserved power would be far less costly than purchased power from Bangor-Hydro. Equally important, these costs would not be incurred annually but rather only over the period of the payback -- 3, 5 or 7 years. If utilized, these source could replace lost power in the Back Channel at a cost at far less than half the estimate used by Bowater/GNP.

2:

The failure in the DEIS to examine energy conservation as a source of replacement power is troubling for several reasons. By not examining this power source, the DEIS is basing public resource decisions on inaccurate information. If accepted by the Commission, then FERC will have allowed the internal capital return requirements of a private corporation to control the decisionmaking process over an important public resource. FERC will be allowing an applicant to waste electricity through inefficient end use while claiming that it cannot afford to lose 2% of its current hydropower generation without severe socioeconomic impacts. Decisions on a public resource cannot be limited by a private corporations' internal capital investment formulas which waste electricity that could be captured cost-effectively. If GNP is allowed to waste electricity by maintaining a highly inefficient plant, then the hydropower produced from the GNP system is simply being wasted.

Inaccurate Predictions of Effect of "Modernization"

The DEIS admits that energy conservation is a possible option for replacing lost power, but then rejects this alternative on the assumption that modernization results in higher energy costs. This assertion is flawed for several reasons. Bowater/GNP's discussion of energy conservation at the mills reflects the company's historic efficiency investments. These investments have been selective; the company has simply not attempted to capture vast amounts of untapped efficiency potential in the mills. In fact, it is likely that the overwhelming amount of conserved power potential at the mills remains wholly untapped. No trend can be drawn from the selective efficiency investments made by GNP, which excludes the bulk of the conserved power potential, are flawed; an accurate trend would need to be based on total treatment of the facility. Moreover, a great deal of conservation potential is cost-effective without modernization and can be captured cost-effectively now. FERC is making decisions now, based on the current status of power needs and resource values. Finally, although Bowater has claimed that it will modernize, there is nothing in the record detailing what the company's modernization plans are, when it might modernize

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-37 See response CI-36. The staff used current market-based financial parameters to conduct its evaluation; criteria specific to the applicant's likely source of funds were used since the applicant will be the one providing the capital to pay for the improvements.

CI-38 The staff identified five alternative power sources in the DEIS: improvements in the efficiency of existing hydropower generation, wind generation, reduction of conservation voltage, refuse-fueled generation, and fuel cells. In evaluating these alternatives, the staff considered the first three to be non-exclusionary because these sources have low marginal costs and, as such, would displace higher cost fossil fuel generation. As non-exclusionary resources, they are not considered to be reasonable alternatives to hydropower and therefore we eliminated them from the analysis. Refuse-fueled generation and fuel cells were rejected for both operational and environmental reasons (see section 2.0). The staff determined the only other power source currently available to GNP to be Bangor Hydro-Electric; GNP does not have access to wholesale power in NEPOOL.

> The staff does not have evidence of large amounts of potential energy savings that could be obtained at GNP's mills. To the contrary, the applicant has submitted documentation describing the conservation measures they have implemented. As noted above, there are strong economic incentives for GNP to economize on energy use rather than waste it, particularly when energy is a major cost input. The staff finds that more conservation projects are likely to be found cost-effective if their savings are valued at the cost of obtaining replacement power from BHE, than if they are valued at the much lower cost of hydropower. We agree that conservation under some circumstance may be the least-cost source of power other than hydropower.

CI-36 Cont

CI-37

E-24

C1-38

⁸ GNP's overall demand for electricity has actually <u>declined</u> from 1.410 million MWh/yr in 1985 to 1.2 million MWh/yr in 1994. (1994 figure is prior to closure of the Millinocket grinder facility.)

25

and how much of the plant will be modernized; there is no commitment by Bowater to actually invest in modernization.⁹

The DEIS also appears to accept GNP's view that any source of power that "does not burn fossil fuels" is "non-exclusionary" and therefore "eliminated from further analysis". (DEIS at 2-30.) The term "non-exclusionary" is not defined in the DEIS. The DEIS on the Lower Penobscot states that "non-exclusionary energy sources" are those that "would be constructed or implemented in addition to hydropower rather than replace hydropower, and vice versa, because all such energy sources have low marginal costs and would be used to displace higher cost fossil fueled generation rather than to displace ach other in actual practice. Nonexclusionary resources, therefore, are not reasonable alternatives to each other..." (DEIS, Lower Penobscot River Basin, FERC/DEIS-0082, at 2-25.)

This explanation relies upon an inaccurate assumption and also asks the wrong question. The inaccurate assumption is that nonexclusionary sources will be captured in addition to hydropower development. GNP has proven that this assumption is not true; the fact is that vast amounts of conserved power exist untapped in the GNP system because Bowater enforces an arbitrarily high payback rate. As in the case of the Big A Dam proposal, energy conservation, which would be characterized as a "nonexclusionary resource" under this definition, was not captured in addition to hydropower production. (See attached affidavits of Amory Lovins.)

Nore fundamentally, however, FERC's position is based on the wrong question. The question with regard to the Back Channel is whether the costs estimated by GNP for replacement power are accurate as a least-cost source of power. We submit that they are not, that the enormous conserved power potential provides a much less expensive source of replacement power. The question is, therefore, is not simply whether hydropower can be displaced by conservation, but whether conservation is the least-cost source of power <u>other</u> than hydropower, and whether it is cost-effective to capture the conserved power based on fairer payback periods. Conservation may displace oil in some cases; it may displace purchased power in other cases; it may displace hydropower in some cases. But when it is cost-effective, it is the least cost source.

Other Power Sources

Much has and is happening in the wholesale and retail power markets in the four years since GNP estimated that purchased power

CI-38 Cont

⁹ Modernization, of course, is the fastest route to substantial employment reductions due to intensive reliance on mechanization rather than labor, as can be seen from the example of modernized pulp and paper plants in the US, as well as other manufacturing sectors such as steel and automobiles.

CI-38 would cost \$83/MWh. Currently, market costs are in the \$30 for short term power and \$50 for long term power. (See Affidavit of Paul Chernick.) There is no evidence that GNP has attempted to capture these benefits.

Grinder Closure

In the summer of 1994, Bowater/GNP closed its grinder pulp facility at the Millinocket mill. This facility had been in use for 90 years. The diversion of water out of the Back Channel and into the Millinocket Mill was done to provide hydromechanical power to turn these now closed grindstones. In the past decades, electric motor/generators were installed to turn the grindstones and, when not in use, to generate electricity. As best as we can determine, the closed grindstone pulp facility consumed some 203 of all of GNP's total electrical needs.

Clearly, the closure of the grindstones raises a host of fundamental questions with respect to flows in the Back Channel. GNP has reduced its power demand by 20% and, instead of consuming electricity at this facility, is now <u>producing</u> electricity. On August 19, 1994, the Conservation Intervenors filed a Motion with FERC raising questions about this closure and its implications for issues under review in the DEIS. FERC has not acted upon our motion. On September 2, 1994, GNP responded to our motion but did not address any of the substantive points or resolve any of the questions raised by the motion and rather invited comments on this issue during the DEIS process.

A 20% reduction in overall demand, with increased generation, clearly affects the issues surrounding the ability and costs to GNP of provided year round flows in the Back Channel and GNP's need for power. The closure could provide more than sufficient water to meet the requested flow needs for fisheries, recreation and cultural needs in the Back Channel, which can now be diverted from the Millinocket Mill over Stone Dam at no additional cost to the applicant. The DEIS itself fails to incorporate this major and fundamental change in the GNP system. Instead, the DEIS incorrectly and inappropriately treats the grinder facility as if it were still on line. (DEIS at 1-3, 2-3, 2-6, 2-10.) By treating the grinder facility as if it were still on line, the DEIS is based on a fundamentally flawed assumption which affects the heart of many issues in this proceeding. The FEIS must amend the DEIS to reflect the changes wrought by the closure of the Millinocket grinder facility and examine the impact of the closure on these issues. Without an accurate analysis of this closure, the DEIS is fundamentally flawed.

3. Socioeconomic Implications

O The DEIS recommends against implementation of certain environmental and recreational improvements, most prominently year

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

26

CI-39 Closure of the Millinocket grinder room has not released water flows for non-generational purposes since the existing synchronous grinder motors are now used to generate electric power for the modernized East Millinocket grinder room. The staff does not consider these changes in operations to be either major or fundamental, nor is the analysis flawed because of them.

> The re-diversion of water from the Millinocket mill into the Back Channel would impose a power loss on GNP. The existing turbines formerly produced power that was used on-site for the grinder pulp facility. These turbines are still producing the same amount of power, but it is now used in the modernized East Millinocket Grinder room. GNP's need for hydroelectric power has not declined as you stated. The diversion of water into the Back Channel is not cost-free, and the resulting decrease in power production should be valued at the cost of obtaining it from the least-cost alternative source. This is the approach that was used in our economic analysis.

CI-40 The staff agrees that the preferred Alternative Two (final recommendation) would not adversely affect the competitive position of the mills, and would not translate into a significant increase in the cost of production at the two mills. The staff also concludes that the negative annual net benefits under Alternative One would be large enough such that the competitiveness of the two mills is likely to be adversely affected.

E-26

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

round flows in the Back Channel, because of the claimed impact on employment in the region and Bowater's threat not to invest in GNP if flows are provided in the Back Channel. (DEIS at 5-15.)

No parties, including our organizations, support measures which would cause such dire employment reductions. But neither is it acceptable or responsible for a federal licensing agency to accept the unsubstantiated threats of this company. Bowater/GNP, although willing to play the "jobs card," has been unwilling to provide information to support its claims of 238 job losses should flows be restored to the Back Channel. Rather, common sense and the history of this company tell that it is guestionable whether flow changes would indeed cause such dire socioeconomic impacts:

* A flow of 350cfs in the Back Channel would create power losses equivalent to 1.8% of GNP's total electrical demand and 3% of GNP's total hydropower production. This hardly seems sufficient to require the loss of 15-20% of GNP workers -- or to forestall any planned investments by Bowater in the GNP facilities. Clearly, as noted below, there is abundant conservation potential in the mill systems to provide 2% more power at a fraction of the cost used by GNP to estimate job losses and relied upon by FERC.

* If these machines are so marginal, then it is most likely that they will be shut down by Bowater, for reasons having nothing to do with hydroelectric generation. (See Affidavit of Paul Chernick.)

* If Bowater/GNP is devoting low cost hydropower to the most marginal machines, then the company is failing to maximize the profitability of its best machines. In other words, a rational manager would devote the least cost power source to the most profitable machines; it would not be economically rationale to send high cost power to machines with high profit margins and low cost hydropower to machines with low profit margins. (See Affidavit of Paul Chernick.)

It is even more puzzling that Bowater would claim that the need to replace 2% of hydropower would jeopardize a claimed planned investment in the mills of \$600 million. Bowater claims that the reason it purchased the Great Northern assets from Georgia-Pacific was to diversify its product mix from dominant reliance on newsprint to more profitable products such as lightweight coated magazine paper. For that reason, it spent \$322 million to acquire the Millinocket and East Millinocket mills; the largest privately used hydropower system in the United States; and 2.1 million acres of Maine land. Bowater has already proceeded to invest over \$60 million in a deinking plant (with State subsidized loans) and, in January 1995 announced plans to invest an additional \$3.8 million

Cont
in a peroxide bleaching system.

It defies common sense to claim that an increase of \$1-2 million in energy costs -- less than 1% of the GNP asset purchase price -- would cause Bowater to back off from its billion dollar strategic acquisition and investment plan for GNP.

There is no viable evidence supporting claimed job losses

The "evidence" that (i) existing, marginal paper machines will be closed due to flow changes and (ii) that Bowater may refuse to follow through on planned investments in the mills consists of the following:

* Proprietary, "confidential" information, known only to Bowater/GNP, information not in FERC's possession and not supplied to any parties;

* Excerpts -- selected by Bowater -- from a proprietary consultant report paid for by Bowater/GNP (Jaakko-Poyry Consulting, Inc.) regarding overall costs of productions at GNP facilities; GNP has refused to even supply FERC or parties to this docket with the full report.

The DEIS admits that "Because GNP's data about corporate economic status is proprietary, we could not verify GNP's estimates of the economic effect of production cutbacks." (DEIS at 5-15.) The DEIS also states that:

GNP contends that flows in excess of 50cfs in the Back Channel would result in the loss of approximately 238 jobs. (GNP, 1991b). We have no way to independently evaluate this estimate, but we assume that it is a high and estimate. Using the Maine Department of Labor employment multiplier, the indirect effect of these lost jobs is <u>383 jobs</u>...

(DEIS at 4-72, emphases added.) Yet, the DEIS later inconsistently adopts Bowater/GNP contention that 238 mill and 1200 regional jobs could be lost. (DEIS at 5-15.)

FERC did request that Bowater/GNP substantiate its assertions that 238 jobs would be lost at the lightweight coated paper complex in Additional Information Request 5 of October 28, 1993. AIR 5 asked for specific information: the operating costs for paper machines 7 and 8 and the blade coater at Millinocket; current and anticipated profitability of these machines; and current and anticipated profitability of the Millinocket pulp and paper operations compared to other Bowater facilities. Bowater/GNP refused to supply answers to any of these questions. Rather, in its response to AIR 5, the company submitted selected excerpts from a report by its paid consultant. These excerpts were intended to

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

28

CI-41 The staff carefully examined the evidence submitted by GNP and determined it was sufficient to conclude that, relative to other producers, GNP's costs are high and that further cost increases could reduce the company's competitiveness. The staff trusted GNP's estimate of job losses as a worst case estimate (as detailed in Exhibit H). The staff revised the multiplier effects (see section 5.3.5).

CI-41

CI-40

Cont

show that GNP experiences higher costs in the aggregate than other mills in North America. No information specific to the lightweight coated paper machines as requested in AIR 5 was provided. Bowater/GNP refused to even provide the full Jaakko-Poyry report to the Commission or intervenors under a claim of confidentiality. (See Letter from Thomas E. Mark to Daniel L. Sosland, February 1, 1994 at 2: the J-P document "is a highly sensitive business analysis that was prepared for Bowater under a condition of

29

analysis that was prepared for Bowater under a condition of confidentiality. It has not been made available to the public, the Commission or its consultants. Due to its confidential and sensitive nature, Great Northern will not provide copies to CLF.") General information on GNP's competitive position does not begin to address issues related to the threat to close specific, individual machines due to a 2% increase in power costs.

The DEIS also states that "our assessment of the paper industry confirms GNP's contention that its present costs are high and that further cost increases could reduce the company's competitiveness." (<u>Ibid</u>.) No citation to any portion of the record is made for this assessment and no analysis or summary of FERC staff's assessment is provided. Any independent staff assessment of employment losses should be provided together with source material. If the assessment referred to are excerpts from the Jaakko-Poyry report, then the assessment is fundamentally flawed. It is not appropriate and is, we believe, a violation of NEPA and the Federal Power Act for FERC to base a resource decision on confidential information to which FERC has not had access.

FERC must do more than simply accept and repeat Bowater's threats. The DEIS uncritically accepts the assertions of the applicant that economic doom could be caused by implementing year round flows in the Back Channel. These assertions have been accepted equally uncritically by the local communities¹⁰ and business interests which depend upon and have transactions with Bowater/GNP. Bowater/GNP's forecast of economic doom do not pass the straight face test and the DEIS is faulty for relying on them.

4. <u>Resource Values</u>

CI-43

CI-4 2

Cont

In reviewing each alternative and in finalizing

¹⁰ At the public hearing on the DEIS, the official representative of the Town of Millinocket, a party to this proceeding, made a statement regarding conversations with staff over the Town's views of the socioeconomic implications of year round flows in the Back Channel. Such conversations from a party consistent inappropriate ex parte contacts. To the extent that these conversations took place, FERC should enumerate the dates and contents of such conversations and provide an opportunity for response by any other party to this proceeding, in order to ensure that the FEIS is fairly prepared.

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-42 Staff is not aware of any ex parte communications with any party to this proceeding. All communications have been made through official and open channels available to all parties.

CI-43 The staff's resource balancing in Chapter 5.0 of the FEIS notes the recreational benefits that would accompany the increase in recreational opportunities at the two projects. GNP's survey-based information was the most accurate data available to us, and our knowledge of other recreational use studies indicated to us that its values were appropriate for use in the FEIS. We conclude in Section 4.8.1.1 that neither commercial nor local private whitewater boating groups expressed interest in expanding whitewater boating opportunities in Back Channel and that high-guality recreational fishing waters are abundant in the project region. We also conclude that increasing flows in Millinocket Stream and Back Channel to make it navigable could jeopardize other environmental enhancements and would not result in any significant benefits to recreational fishery resources.

Ci-41 Cont

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

30

recommendations, the DEIS fails to model accurately the economic value of increased fishing and recreational opportunities in the West Branch. (DEIS at §4.12.1.1.) The DEIS relies on GNP survey for its conclusions on recreational demand and economic values. As pointed out in our Letter of Deficiency, Motion to Intervene and proposed Terms and Conditions, we believe that GNP's figures are low in assessing how fishing and whitewater recreation on the West Branch benefits the local economy. The data relied upon is limited to commercial rafting. Commercial rafting operations represent only one section of the boating public. Private boaters represent a users segment which make multiple trips to the area, sometimes weekly throughout the boating season. In fact, most trips on the Penobscot are day trips. An accurate assessment of the economic value of these users must be included in the economic estimates.

As we have repeatedly pointed out since our February 1992 filing, the economic potential of controlled, additional recreational resource in the Back Channel and elsewhere is high. The DEIS fails to consider the potential economic value of these enhancements to the regional and state economies.

5. Water Use Model

As the DEIS notes, our coalition submitted extensive criticisms of the Bowater/GNP water use model. Our criticisms go to three general main concerns. First, that the claim by GNP that there is insufficient water quantity in the system is erroneous and that, in fact, the model is too simplistic to accurately predict the implications of various flow changes. Second, that the model excluded methods of forecasting energy losses, making FERC and the parties unable to substantiate Bowater/GNP's estimates of power losses. Third, that by excluding upper basins which are operated so as to provide water quantities to support the proposed and existing operations of the Ripogenus and Penobscot Mills projects, environmental impacts upstream directly created by these project operations are absent from basin water models.¹¹

We agree with the conclusions set forth in the DEIS that the question of fulfilling agency and intervenor flow requests is not limited by water availability or quantity. In doing so, the DEIS rejects the applicant's position that water quantity constrains additional flow improvements. However, we disagree with the conclusions in the DEIS that GNP's water use model is appropriate for evaluating streamflow issues at the Ripogenus and Penobscot Mills projects because the model developed by GNP is based on

CI-44 Opinion noted. We provide responses to specific comments on the water use model below.

CI-44

CI-43

Cont

¹¹ The Conservation Coalition provided a discussion of other deficiencies of the GNP water use model during the scoping process for the EIS. FERC responded to some of these criticisms in Appendix D of the DEIS. We have responded to the DEIS' description of our concerns in Attachment B.

problematic assumptions and is too simplistic to represent the complex system being studied. (DEIS §4.2.) Given the scale and scope of the project, and the benefits that would accrue to all parties from optimizing operational flexibility, the existing model cannot be used to make long-term resource decisions.

* The model is intrinsically biased toward power production.

The model developed by the GNP is a simulation of the way that the hydropower system has been managed in past years. GNP's system has historically been operated to maximize power for mill operations, and GNP used a historically derived rule curve to determine reservoir operations for this purpose. In this way, the assumptions which underlie the historic rule curve are embedded in the model. Maximizing energy production is a key element of the model. In effect the model answers the question "What opportunities for changes in water use exist given the current energy production management plan," instead of "What opportunities for changes in water use exist independent of any initial value judgments?"

* The model excludes reservoirs other than North Twin and Ripogenus and is therefore incapable of evaluating the cumulative effects of the project on the basin.

GNP's operations on the upper Penobscot River involve the operation of multiple impoundments and generating systems. Yet, GNP's water use model represents the entire system as only three elements. This approach <u>a priori</u> forecloses upon undetermined and potentially significant operational flexibility. In order to model and route flows through the basin, the upstream reservoirs should be considered individually so that it is possible to determine how changes in the management of these reservoirs could influence the basin-wide water balance.

GNP dismisses the need to more accurately portray the system by asserting that upstream impoundments have no "significant storage". Yet GNP provides no explanation of what constitutes "significant storage." The upstream impoundments in question represent nearly 20% of the total storage of the system. (DEIS at D-27.) We assert that this storage is "significant." The DEIS should address this issue. Additionally, GNP asserts that the operation of the upstream impoundments is "unaffected by downstream processes." This claim requires explanation. If the operation of the Ripogenus impoundment were changed, there would certainly be opportunities to change the operation of, and therefore the environmental effects of the project on, the upstream ponds.

FERC's acceptance of GNP's assertion that upstream impoundments need not be studied or modelled is troubling. The scope of this project is very significant and the precedent set by

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

31

CI-45 We disagree that GNP's water use model is biased toward power production. As we used the model for evaluation of alternatives in the EIS, energy production was not a factor at all. The model is simply a water-accounting model, developed from historical impoundment elevations and flows. There is nothing inherent in the model that prevents it from being used to evaluate any flow and lake level management alternatives.

CI-46 The upper storage projects are included in GNP's water use model and the alternatives evaluated with that model. including additional model runs requested of GNP by FERC staff. These projects are not up for relicensing now but are included in the water use model as a combined input. This input was kept constant in developing the water use plan although input could be varied in the model. We saw no need to do this however, since no one has suggested alternative management schemes for the upstream projects that would provide additional downstream benefits. In addition, GNP has agreed to inclusion of a reopener to allow modification of the water use plan, should analysis of the upper projects during relicensing result in unanticipated findings that desired changes could be made. We recommend the orders for both the Ripogenus and Penobscot Mills projects include an article containing a reopener clause for consideration of changes in water use in the West Branch when the upper storage projects are evaluated for relicensing.

Ci-45

CI-44

Cont

this decision will be important. An incomplete study is not acceptable. Certainly GNP has the data necessary to include additional elements of the complex system in its model. These elements should be included individually and their worth should be quantitatively evaluated.

32

In the DEIS, FERC indicates that these system elements need not be explicitly considered because they are not included in the current license proceedings. In fact, as explained by the Conservation Coalition in many previous filings, FERC has the authority and the responsibility to review the cumulative effects of the entire system.

* The model uses an inadequate timestep.

The weekly timestep used in GNP's model does not adequately represent system operations. The record indicates that there is often a tremendous amount of variability over the course of a week or day that would affect GNP's ability to provide certain benefits.

Meeting a minimum flow average over the course of a week is no guarantee that the minimum is met during every day of that week. For example, in 1984 within the week from April 13 to April 19 the "Natural Inflow" at Ripogenus ranged from 5,062 cfs to 22,777 cfs, a range of 17,715 cfs. The average for the week is 10,729 cfs. The large range of flows is lost in the average. It would, therefore, be impossible to tell from such an average whether there were significant low flow days or floods during that week. Without at least a daily time step -- which is commonly used in modelling operations such as this -- it is not possible to accurately predict what benefits are provided or compromised at any given time.

Knowledge of daily fluctuations may be critical to the evaluation of some aquatic resource issues. The goal of sustaining minimum flows is to provide habitat in which aquatic species can survive. If minimum flows are sustained only on average and not continuously, this goal can not be achieved. In effect, modeling minimum flows on a weekly timestep does not ensure that those flows can be met continuously and, thus, severely undermines the goal of the effort. Additionally, since white water boating usually takes place for several hours a day, weekly average data is not sufficient for evaluating the impacts of various flow alternatives on this activity.

In the DEIS, FERC stated that "Because GNP operates its system for continuous production rather than for peaking operation, large changes in flow within a day do not occur due to project operation and thus do not require simulation to determine water usage." (DEIS at D-27.) This statement is misleading. Large changes in flow which occur within a week or within a day may not be due to GNP's operation, but they occur nevertheless and should influence GNP's operations. Furthermore, the system is already being managed

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-47 We disagree that GNP's model uses an inadequate timestep. We used the model to evaluate whether there was sufficient water storage and flow within the river system to provide various flow and lake level enhancements on a seasonal to annual basis. The smaller timesteps you suggest would allow evaluation of within week and diurnal changes in flows, but this is not necessary for the broad flow-related alternatives we evaluated in the EIS.

CI-46

Cont

33

for variation within each day of week since there are diurnal fluctuations in releases below McKay Station.

The water use model should operate on an hourly or daily timestep. The coarse weekly time step does not resolve existing variabilities on the time scale of a day and does not permit proper evaluation of optimization measures which could allow for appropriate enhancement flows. Certainly GNP has all of the data necessary to make this important change in the model.

The DEIS does not adequately evaluate the accuracy of the GNP water use model.

Nowhere in the DEIS is the accuracy of GNP's water use model evaluated. The DEIS should include a comparison of model predicted power generation and actual power generation for every time period modeled, a discussion of model error and accuracy, and a complete sensitivity analysis. Without such an evaluation, model results can not be responsibly accepted because there is no proof that the scenario being studied actually accurately portrays the system. If, for instance, the model underpredicts power generation, then the model results are seriously misleading and there is yet undiscovered operational flexibility in the system. Applicant estimates of energy losses at the Back Channel alone vary by at least 10%. (See AIR response 4 (at 28), which indicates that energy losses at Stone Dam approximate 22,559 MWh/yr; Penobscot Mills application states that the losses would be 20,800 MWh/yr, about 10% less.)

For these reasons, the water use model developed by GNP does not provide an adequate basis on which to make long-term resource decisions.

IV. <u>Water Quality: Toxins</u>

The DEIS acknowledges that elevated levels of mercury exist in project waters but then concludes, mirroring the position of the applicant, that "all available information leads us to conclude that project operations probably are not the cause of elevated mercury concentrations." (DEIS at 4-11.) The DEIS's analysis and conclusions are not supported and in fact are contrary to data in the record.

* The application treats elevated mercury levels in sediment at Dolby Pond as not significant -- although these levels are almost 5 times greater than in control reference lakes. (See Penobscot Mills Application, Vol. XIII, § 8). The DEIS fails to acknowledge this anomaly and fails to review obvious links between hydropower operations and elevated sediment levels. For example, the applicant's mills and other operations likely emitted mercury as part of its industrial processes. The hydroelectric reservoir created at Dolby Pond would be the immediate and obvious sediment

- CI-48 We did not use the water used model to predict power generation or losses of generation under various alternatives.
- CI-49 Elevated levels of mercury were only found in the sediments of Dolby pond (which does not experience periodic drawdowns) and in Lake Trout in the draw-down reservoirs. All other samples (sediments, water column and tissues of mussels and other fish) were similar between project waters and control lakes. Statistical tests were conducted for top predatory species since bioaccumulation would be expected for these higher organisms. After adjusting the data for differences in fish length, significantly higher mercury concentrations in draw-down reservoirs were found for Lake Trout between draw-down and non-draw-down lakes using Scheffe's Multiple Comparison test. However, because similar mercury concentrations were found for other predatory fish, bottom feeding fish, and freshwater mussels between project lakes and reference lakes, we concluded that the higher concentrations in Lake Trout was not related to reservoir draw-down. Due to Lake Trout's top predator status and the abundance of rainbow smelt prey in the project impoundments we agreed with GNP's explanation that the higher concentrations in the Lake trout in project water relative to control lakes was due to heavy predation on rainbow smelt.
- CI-50 Higher levels of mercury in Dolby pond sediments are probably the result of historic discharges from the mill operations (which discharges into the upper portion of the Dolby impoundment) and not a result of the fluctuating water levels in the reservoir. Controls of mercury concentrations in the mill's effluent is established in the NPDES permitting system.

CI-47 Cont

m

င္မ်ာ

CI-48

CI-50

CI-50 Cont

CI-51

CI-52

trap for any mercury laden effluent from the mill.

* The DEIS fails to acknowledge and examine errors in study protocols. For example, the applicant did not sample the number and size of fish as outlined in the original mercury study design. The smaller sample makes it more difficult to detect differences between reservoirs and control lakes. At Tables 3-7 and 4-3 the DEIS compares average mercury concentrations found in fish from reservoirs and reference lakes. Comparisons based on lake averages have questionable validity as it is well known that mercury bioaccumulation in fish is related to the size of the fish. The DEIS never make the necessary fish size to mercury fish tissue concentration adjustments, considered a standard protocol for any such comparisons. When Terry Haines of the US FWS made such a size adjustment by weight, the results show statistically higher mercury levels in the impoundments compared to control lakes for lake trout and smelt. (Letter of 12/30/92 from T. Haines to K. Kimball.)

Protocol errors call into question mercury levels in sampled waters reported in these applications. Again, when reviewed in an independent analysis by Terry Haines of the USFWS, an acknowledged expert in mercury, Dr. Haines concluded that the GNP/Bowater "mercury data in water are analytical artifacts, which result because trace metal-free collection and analysis protocols were not used." The DEIS fails to acknowledge such data inadequacies.

* The DEIS accepts almost verbatim the applicant's conclusion that project operations are not the cause of elevated mercury concentrations. (DEIS at page 4-11.) In doing so, the DEIS ignores the rich scientific literature base which shows that comparative studies of mercury bioaccumulation between lakes should be based on top predators. Organisms lower on the food chain, such as the mussel, are inappropriate for such purposes. The applicant has provided data on top predatory game fish, the lake trout. Mercury levels in lake trout from drawdown reservoirs are statistically much higher than from similar fish from control lakes. This was verified by independent, recognized experts and been reported by both our coalition and natural resource agencies. The same holds for rainbow smelt, a species intermediate on the food chain and the principal forage species for lake trout.

The absence of significant differences of mercury levels in species low on the food chain, such as the white sucker and mussel, has little bearing on the question of the contribution of hydropower operations to mercury contamination as it is well known that mercury which bioaccumulates will likely not show such differences at lower levels in the food chain. Similarly in previous filings the US Fish and Wildlife Service has shown that in fish eating birds, such as the endangered species the bald eagle, mercury levels in both feather and blood samples of eaglets within the project area exceeded mean levels reported for other lacustrine nesting eagles in Maine and for other parts of the country. It is

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

34

CI-51 The number and types of fish outlined in the study plan was the target for the study. We recognize that field conditions (i.e., the type and number of fish available) ultimately dictate the number and suite of species that are included in testing of this nature.

> GNP's study included adjustments based on fish size for the statistical tests with predatory fish. This analysis indicated that mercury concentrations were significantly higher in drawdown reservoirs for Lake Trout.

- CI-52 The field and laboratory methods used in the GNP study were developed in consultation with the resources agencies involved with the project, including the detection limits for the various media to be tested.
- CI-53 The selection of fish species (predatory, forage, and bottom feeders), invertebrates (mussels), sediment and the water column was developed in consultation with the resources agencies involved with the project.

Statistically higher levels of mercury were found in Lake Trout in draw-down reservoirs (approximately 2.5 times higher). However, given that there were no differences in the mercury concentrations in mussels, sediment, and tissues of bottom feeding fish species, project operations do not appear to be the cause of the higher levels in Lake Trout. Differences in mercury concentrations in rainbow smelt were not evident in the GNP study, and you presented no quantitative results indicating that size adjusted rainbow smelt concentrations were statistically higher in project waters.

E-34

ER PENUBSCUT RIVER BASIN DER

CI-53 Cont also well known that reservoir operations greatly accelerate the CI-54 bioaccumulation of mercury, particularly in newer reservoirs.

The DEIS, while relying on the conclusion that there is no reservoir connection to elevated mercury levels, then attempts to rely on state ordered studies requirements. (Table 5-6, 5-7, and 5-8). These studies, however, are not intended to address the specific issue of whether the GNP hydropower operations contribute to elevated mercury levels in impoundment waters. The mercury studies requested by the State:

* only require the applicant to "cooperate in a study to be conducted by the Department and the Environmental Protection Agency to determine the interrelationship and impacts of atmospheric deposition and water level fluctuations on concentrations of mercury, cadmium, lead and other metals on aquatic life in the project waters." (ME DEP 401 Water Quality Certification at page 18). This condition shifts the responsibility for determining the contribution of hydropower operations from GNP, where it properly belongs, to the State. It leaves unanswered what will happen if the State does not find the moneys needed to conduct such studies. Most importantly, the potential study relies in part on the EPA and State of Maine Environmental Monitoring and Assessment Program (EMAP) -- which EPA has acknowledges was not designed to answer the question of whether reservoir drawdowns enhance the rate of bioaccumulation of mercury.

In short, there is substantial indicia in the record that mercury levels in top predators in hydropower impoundments exhibit elevated mercury levels compared to non-hydropower control reservoirs. The DEIS fails to examine the link between hydropower operations and elevated mercury levels. The FEIS and final licenses should be conditioned upon the filing within five months after the date of issuance of the license with the Commission of a plan to analyze the severity of the mercury problem in a predatory game fish, the bald eagle and the common loon in their reservoirs and riverine stretches downstream of the reservoirs that are part of the West Branch of the Penobscot. The plan should be designed by an independent group of mercury experts agreed to by FERC, the US Fish & Wildlife Service, the Penobscot Indian Nation, Maine DEP and Maine DIFW. The plan shall include a description of species and the desired size of the specimens to be sampled and the analytical methods used. All sampling protocol should meet US EPA and US Fish & Wildlife Service quality control and assurance criteria for mercury testing. The study plan should be designed to determine the role reservoir operations have on the mobility and bioaccumulation of mercury in the West Branch as influenced by the GNP/Bowater water use plan; whether human health is at risk; and whether the reproductive success of bald eagles and loons which are predatory fish eaters are at risk from mercury contamination.

The mercury study should be completed within five years

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

35

It is true that the 401 WQC study stipulated for the Penobscot project only requires cooperation with studies to determine the interrelationships and impacts of atmospheric deposition and water level fluctuation on concentrations of metals. However, the 401 WQC studies for the Ripogenus projects also requires that the applicant "conduct a study similar in scope to that conducted by the applicant for Millinocket Lake in 1992 to assess mercury concentrations in aquatic life in the Ripogenus impoundment and the West Branch.... A work plan and proposed schedule for the study shall be submitted to the Commission [FERC] for its review and approval within 90 days of the effective date of this certification". Therefore, additional studies designed to investigate the effects of periodic draw-downs on mercury accumulation will be conducted in the project area.

The cooperation with the atmospheric deposition study and the additional study to be conducted in the Ripogenus project waters should be sufficient to further investigate whether fluctuating reservoir levels are related to bioaccumulation in fish. If the state does not fund the atmospheric study, the mercury contamination study in the Ripogenus project water will still be conducted at GNP's expense.

Neither the record for the Ripogenus and Penobscot Mills projects nor the scientific literature contain substantial evidence linking hydroelectric reservoirs with increased mercury levels. Although the potential link is currently unproven, additional research has been adequately mandated by the 401 WQC for both projects.

However, we are not requiring that terrestrial wildlife be studied under either of the plans. As we stated in the DEIS, there is no conclusive evidence to date that the fecundity of the eagle populations at the Ripogenus and Penobscot Mills projects is affected by project operations. No data is currently available in the record indicated that project operation could affect the reproductive success of the common loon.

36

following relicensing. Provisions in the FEIS and license should specifically provide for changes in reservoir operations if the mercury study concludes that they need to be altered to reduce any verified mercury contamination problems by the study.

| v. <u>Wetlands</u>

Although the DEIS acknowledges that the applicant's proposed operations will degrade wetlands in project areas, the DEIS fails to adequately estimate the correct number of wetlands impacted by the proposed operations. The DEIS also fails to specify the means by which wetlands mitigation will be implemented with sufficient certainty.

In describing the number of wetlands impacted by proposed project operations, the DEIS adopts verbatim without any apparent additional review the applicant's estimates of acres of wetlands detrimentally impacted.¹² The DEIS fails to acknowledge that the acreage estimated by the applicant represents only marginal wetlands which have survived a regime of extensive annual drawdowns for decades. Annual drawdowns, however, expose vegetation to desiccation along much larger portions of the shoreline than would occur under non-fluctuating conditions. During reflooding, these drawdowns resuspend and winnow away the finer sediments that provide more suitable substrate for wetland vegetation. Fluctuations reduce the diversity and quality of existing wetland communities and prevent the establishment of wetland vegetation in other areas now lacking it and therefore affect far more wetlands acres than estimated by the applicant and accepted by the DEIS.

The extensive impact of large drawdowns on wetlands has been recognized and documented in other licensing proceedings now underway in northern Maine. Central Maine Power Company's study on the Brassau Reservoir, which has a similar size and annual drawdown (31 feet) as the reservoirs in these licensings. In its report "Monitoring and Assessment of Lake Level Fluctuation Effects on Fish, Wildlife and Wetland Resources at Brassau Reservoir" (1994), CMP recognized that major annual drawdowns diminish wetland development; cause scouring, freezing and desiccation; reduce waterfowl and furbearers due to nest flooding, dewatering and

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-55 Our estimate of the acreage of potentially impacted wetlands at Ripogenus in the DEIS was based on data presented by the applicant in Volume II of the Ripogenus Project license application and from a site visit in August 1993. No other data have been presented to date by the applicant or the intervenors that would refute these figures. The FEIS includes an estimate of the quantity of affected shoreline wetlands at North Twin.

The means and costs of the wetlands enhancements were presented as part of the record in response to AIR #2, October 28, 1993. We subsequently determined that this information was sufficient to make decisions concerning the enhancements.

CI-56 The wetlands acreages presented in the DEIS, i.e., 1,251 acres at Ripogenus, 296 acres at North Twin, and 709 acres at Millinocket Lake, are the total acreages of wetlands at each of the reservoirs, including both affected and unaffected wetlands.

> There is no data currently existing in the record that suggests more wetlands are affected by impoundment fluctuations at the projects than has been indicated by the applicant and also observed by Staff during a field visit in August 1993.

CI-54

Cont

CI-55

¹² The DEIS concludes that there are only 1,251 acres of wetlands associated with the Ripogenus impoundment, 296 acres with the North Twin Reservoir and 709 acres with the Millinocket Lake (DEIS at Section 3.6). The DEIS then narrows the scope of affected acres of wetlands by arbitrarily estimating that 20% of the Ripogenus wetlands, or 250 acres, are at least hydrologically dependent on impoundment levels. The DEIS is even more vague about the number of acres of wetlands impacted in the North Twin and Millinocket Reservoirs due to annual water drawdowns. (DEIS at Section 4.5)

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

37

CI-56 We considered most of the water level fluctuation-related Cont factors to which you refer, i.e, diminished wetlands development, freezing and thawing, and effects on wildlife in the DEIS. Scouring and desiccation, which were not mentioned as possible effects in the DEIS, have been included in the FEIS.

CI-57 No other definitive data currently exists in the record that estimate wetlands impacts at the projects. There is also no data in the record that indicates far more wetlands would exist at the projects under a regime of less dramatic drawdowns. Site visits by Staff during a dry period in August 1993 helped to determine that roughly 80% (250 acres) of the wetlands around the Ripogenus impoundment are at least partially hydrologically independent of the impoundment, owing to deep peat wetland substrates and/or inflow from adjacent tributaries.

The FEIS includes an estimate of the quantity of impacted shoreline wetlands at the North Twin impoundment. Based on site visits by Staff during a dry period in August 1993, we estimate that approximately 75% (222 acres) of the wetlands at the North Twin impoundment are partially or wholly independent of the impoundment fluctuations, owing to deep peat wetland substrates and/or inflow from adjacent tributaries. From the same field visits, and the information presented by the applicant, we estimate that virtually all of the wetlands at Millinocket Lake are independent of the impoundment fluctuations. As we indicated in the DEIS, continued run-of-river operation at the Millinocket Lake development would maintain *status quo* wetlands conditions.

CI-58 Approximately 250 acres of wetlands are affected by drawdowns at Ripogenus impoundment, and a smaller amount are affected at the North Twin Development. Wetlands enhancements recommended for the Ripogenus and Penobscot Mills projects would result in enhancement of approximately 280 acres of wetlands. We emphasize that shoreline wetlands at the projects still function under existing operations as wetlands prior to any enhancements. No wetlands are proposed to be filled or otherwise destroyed by the proposed

causal factors recognized as severely impacting wetlands on Brassau Reservoir are treated as virtually non-existent by the DEIS on the Upper Penobscot River wetlands. The DEIS fails to address questions over the scope of wetlands

The DEIS fails to address questions over the scope of wetlands impacts in the data submitted by the applicant. The US FWS has questioned the company's wetland calculations and assumptions in numerous filings. Far more wetlands would exist in project waters with a less dramatic drawdown regime. (See Conservation Intervenors Motion to Intervene, August 20, 1992, at 25-28.) The DEIS fails to document how it determined that 80% of the Ripogenus Impoundment's 1,251 acres of wetlands are hydrologically independent of the reservoir and therefore not impacted by the extensive annual drawdowns experienced annually. Similarly, the DEIS fails to provide evidence as to how the DEIS calculated which wetlands on Millinocket Lake and North Twin Reservoir are and are not impacted by annual drawdowns.

increased predation; and kill benthic organisms. The kinds of

The DEIS (at 5-10) boldly goes even further to point out that the Alternative 1 wetland mitigation package of approximately 380 acres (at 5-9) would be in excess of those negatively impacted by the drawdowns, and on this basis eliminates the wetlands mitigation recommendation set forth in Alternative One. The assumption that 380 acres would be sufficient wetland mitigation is too low. Such conclusions are neither warranted nor documented; they need to be corrected and addressed in the FEIS.

The DEIS accepts the development of 280 acres of wetlands at Black Pond, Deep Cove East and West as providing approximately 1:1 mitigation. At the same time, the DEIS recognizes that the necessary federal and state permits may not even be obtained to create these "mitigated wetlands." (DEIS at 4-40.) If the necessary permits are not obtained, the DEIS is unclear as to what provisions should be implemented to ensure that an adequate wetland mitigation is accomplished.

The FEIS needs correct the defects in the DEIS by: (i) seriously considering modifications to water level fluctuations to better manage project wetlands; (ii) developing more appropriate estimates of the amount of acres of wetlands impacted and create a more representative wetland mitigation package; and (iii) reviewing potential off-site wetland mitigation, including wetland and buffer zone protection around other important wetlands in the immediate watershed of the reservoirs, for example Pine Stream Flowage and Passamagamet Lake.

VI. Access and Access Fees

The settlement agreement reached between Bowater/GNP and the CI-61 | Fin & Feather Club is a positive step in the right direction of ensuring free access to Maine residents to these public waters. To

E-37

CI-56

Cont

CI-57

CI-58

CI-59

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

- CI-58 operation of the projects. In addition, wetlands are an extremely
- cont abundant resource in the area of the project. For these reasons, we maintain that Alternative 2 is the most appropriate alternative.
- CI-59 We recommend that GNP prepare the restoration plans after consultation with the Corps of Engineers, the Fish and Wildlife Service, and the Maine Department of Inland Fisheries and Wildlife. No restoration activities will begin until the plans are approved and GNP is notified by the Commission.
- C1-60 Modification of water level fluctuation was considered as part of the planning for balancing of several issues at the projects. In essence, the required wetlands enhancements will serve to moderate water level fluctuations in strategic areas of project wetlands. The wetlands to be enhanced are adjacent to existing higher quality wetlands, with the purpose of expanding areas of such higher quality wetlands. It is most appropriate to provide onsite enhancements because these sites could increase the total acreage of project wetlands that possess high functional and wildlife values.
- CI-61 Opinion noted. This comment will be addressed in any order issued for the project.

fulfill the need for free access by all persons, however, the FEIS should include a license term which provides for the removal of access fees for all persons regardless of state residency.

38

CI-61 Cont

First, imposition of fees act to restrict access. The DEIS in effect acknowledges this by describing the "number of visitor-days dronmed in 1987 after GNP instituted access fees." (DEIS at 3-34.) Second, basing the fee structure on state residency is inappropriate. The waters of the Penobscot River are public waters to be used and enjoyed by all citizens. Restrictions based on state residency conflicts with the need to provide nondiscriminatory access to those involved in river recreation. Discrimination based on state residency violates Constitutional protections. Finally, without providing for free access regardless of state residency, the applicant is in the position of setting a fee structure that requires non-Maine residents to subsidize the activities of others. The FEIS should prevent this fundamentally unfair result by extending the concept underlying the access agreement to cover access by residents of all states.

The DEIS is vague in defining the responsibilities that the applicant has in providing information through a toll-free telephone flow information service. At a minimum, the applicant has a responsibility to inform the public free of charge about daily flow release schedules for all boatable stretches in the West Branch where flows are controlled by the applicant. The areas that should be included in the toll-free service include all river stretches affected by the proposed water use plan, not limited to the section below McKay Station. In addition, user safety is dependent upon access to flow gauges and on private boater take-out and put-in access.

The FEIS should address these issues and include in the recommended alternative, the following conditions. (We note that some of these provisions may have been incorporated in the GNP/MEPRO agreement; however, that agreement has not been circulated to the parties and it does not appear to be part of the record. We therefore rely upon the representations in the DEIS.)

* Ripogenus Dam. An accessible, safe put-in point and parking facility for private boaters running the Upper Gorge;

* McKay Station. A drive-in location for boat unloading adjacent to the river at McKay Station for private boaters, and parking within a reasonable walking distance. Installation of CI-64 rustic, environmentally compatible, gender separate changing rooms and toilet facilities to accommodate the significant numbers of existing users and protect the health and sanitary conditions of the surrounding area.

* Flow information. Require, at the applicant's expense, CI-65 installation of a toll-free telephone answering service to provide

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

- CI-62 We recommend that GNP establish a telephone message system that would be updated twice daily to provide information about flow conditions at several points along the West Branch and scheduled releases from Ripogenus dam.
 - CI-63 We believe the existing parking and boat unloading areas are adequate at this time. The recommended alternative would require monitoring of recreational facilities every 6 years and would allow for potential improvements to recreational facilities as recreational demand so warrants.
 - CI-64 We believe that the existing parking and boat unloading areas are adequate. GNP proposes two concrete vault privies at McKay Station.
 - CI-65 As part of the settlement agreement with MEPRO, GNP would establish and maintain a telephone message system to provide information about flow conditions along the West Branch and scheduled releases from Ripogenus dam. GNP would also notify a designated representative of MEPRO of any unscheduled releases from Ripogenus Dam.

39

CI-65 Cont updated information on flows, access, notification of deep gate releases and other relevant information. The discussion and recommendation in the DEIS needs to be made more specific and clear that the obligation to provide the service rests on the applicant and that the information provided be more extensive than merely flows below McKay Station. (DEIS at 2-16.)

* Staff gauges. Provide readily accessible and visible staff gauges calibrated to river flow levels at the put-in below Ripogenus Dam, at McKay Station, the cribworks, Abol Bridge and the Back Channel.

| VIII. Water Quality Certification

The DEIS recommends that conditions 5 and 6 of the Section 401 Water Quality Certificate for the Ripogenus Project be omitted from the final license on the authority of Tunbridge Nill Corp., 68 FERC 461,078 (1994). (DEIS at 5-18.) The Tunbridge Mill decision is currently under review for rehearing. The decision reverses a longstanding FERC policy that the Commission does not have authority to reject specific conditions set forth in 401 certificates. FERC's decision to reverse its own policy is contrary to CWA §401(d) which provides that a certification "shall become a condition of a Federal license." See Escondito Mut. Water Co. v. La Jolla Indians, 466 US 765,772 (1984). Jurisdiction for a challenge to a CWA \$401 condition lies in state court. Roosevelt Campobello International Park v. US EPA, 604 F.2d 1041, 1056 (1st Cir. 1982). Accordingly, FERC does not have authority to reject conditions set forth in the Ripogenus 401 certificate.

Conclusion

The Draft EIS should be modified so that (i) the recommended Alternative includes expansion of project boundaries to 500', as detailed in this filing: that a year round flow of 60cfs is provided in Millinocket Stream; that additional summer flow releases for whitewater boating are provided for in the Upper Gorge; that a year round minimum flow of at least 350-500cfs is provided for in the Back Channel; that the cumulative impacts of the proposed action on the upper basins should be assessed and of specific resource up and downstream of the projects; that the amount of wetlands acres to be mitigated approaches the number impacted by the project operations; that a post-licensing study of the potential connection between mercury contamination in project waters and project operations is prepared; that access fees be removed for all users; and for such other and further modifications as set forth in this filing.

ON UPPER PENOBSCOT RIVER BASIN DEIS As part of the settlement agreement with MEPRO, GNP would

RESPONSES TO CONSERVATION INTERVENORS

- CI-66 As part of the settlement agreement with MEPRO, GNP would provide readily visible staff gauges calibrated to river flow levels at McKay Station, the cribworks, and Abol Bridge. We do not recommend gauges below Ripogenus dam or in Back Channel.
- CI-67 Opinion noted. The legal status of conditions contained within Maine's 401 WQC will be addressed in the orders for the projects.
- CI-68 Opinion noted.

CI-67

E-40

CI-66

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

Dated: February 21, 1995 Rockland, ME

Respectfully submitted,

Daniel L. Sosland, Esg. Solution Conservation Law Foundation 119 Tillson Avenue Rockland, ME 04841 (207) 594-8107

Michael Cline, Ph.D. Control Maine Audubon Society, PO Box 6009 Falmouth, ME 04105 (207) 781-2330

Richard J. Bowers Cry Met-American Whitewater Affiliation 1430 Fenwick Lane Silver Spring, MD 20910 (301) 589-94532

Kenneth D. Kimball, Ph.D. 2

Kenneth D. Kimball, Ph.D. Appalachian Mountain Club PO Box 298, Rt. 16 Gorham, NH 03581 (603) 466-2721

Margaret Bowman, Esq. 77 American Rivers 801 Pennsylvania Ave., SE Suite 400 Washington, DC 20036 (202) 547-6900

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

CI-A Comment noted. See CI-8.

ATTACHMENT A -- Recommended Project Boundaries Ripogenus Project

a) Lands on the Ripogenus impoundment not owned by GNP/Bowater: These areas include the northern tip of Black Pond in Township T6 R14; a small segment (ca. 0.5 miles) of the shorefront approximately 1 - 2 miles south of the State of Maine Chesuncook Village property on the west shore of Chesuncook Lake (in Township T4 R13); portions of the west shoreline of Chesuncook Lake and the north end of Caribou Lake in Township T3 R13; lands on the south end of Caribou Lake; lands on the south end of Chesuncook Lake and on Ripogenus Lake; and lands on Moose Pond/Chesuncook Lake in Township T5R12 as identified in Exhibit G, Sheet 2 of 5 in the Ripogenus Application. For these areas, the applicant should:

- to make a fair market offer to the owner for either a 500 conservation easement or the land in outright purchase;
- to set aside a dedicated buffer zone trust fund sufficient to purchase a 500 foot buffer on these shorefront properties, should an owner be unwilling to sell at the time of the offer, to be used in the event that the shoreland property becomes available during the during the course of the license.

b) On the portion of Black Pond, which Bowater has divided interest on the shoreline property (Exhibit G, sheet 2 of 5 in Ripogenus application), the applicant should place a 500 foot shoreline conservation easement, or purchase such rights from the co-owner.

c) Gero Island and the western shorefront of Chesuncook Lake immediately south of the West Branch of the Penobscot River inlet to approximately the border of Townships T5 R13 and T4 R13 (the Chesuncook Village region) are owned and managed by the State of Maine and need no further shorefront buffer zone protection responsibilities by the applicant. The private holdings within Chesuncook Village should also be exempt from GNP/Bowater responsibilities.

d) Existing commercial and private leases with buildings (primarily in the LURC zoned D-RS portions of Caribou Lake and the commercial lease in the LURC zoned D-GN on Ripogenus Lake identified in Fig. E6-2 in Ripogenus Application) for the Ripogenus Project as identified in the DEIS on page 3-47 should be honored and be renewable, subject to LURC zoning regulations. Bowater/GNP should repurchase leases which do not have buildings on them. The sale of new leases or private shorefront properties by Bowater/GNP should be specifically disallowed during the course of the license.

e) For all lands owned by Bowater on the Ripogenus impoundment and not noted previously in this section, a 500 foot no development conservation easement, with no timber harvesting in the first 100 foot zone would apply.

CI-A

2

Penobscot Mills Project

1) Millinocket Lake Shoreline

a) For land on the Millinocket Lake Storage Development not owned by GNP/Bowater, identified as an approximately 1/2 mile stretch on the southwest shoreline (Exhibit G, Sheet 4 of 12, Penobscot Mills application), and zoned by LURC as D-RS (see Figure E6-3, Penobscot Mills application), the applicant should not be required to institute a shoreline conservation easement.

b) On the north and northeast shoreline of Millinocket Lake where GNP/Bowater has divided interests on the shoreline property (identified in Exhibit G, sheet 4 of 12 in the Penobscot Mills Application), the applicant should put in place a 500 foot shoreline conservation easement with no timber harvesting in the first 100 foot zone, or purchase such rights from the co-owner.

c) GNP's existing commercial and private leases with buildings, as identified in the DEIS at page 3-48, should be honored and be renewable subject to LURC zoning regulations. These are mostly in the LURC zoned D-RS and D-GN shorefront in the vicinity of the Golden Road on the west shoreline and the LURC zoned DR-S in the vicinity of the Millinocket Lake Daw at the Millinocket Stream outlet, identified in Fig. E6-3 in the Penobscot Mills Application. The LURC zoned development shoreline represents 13% of the shorefront as listed in Table 3-9 of the DEIS. The selling of new leases or private shorefront properties by GNP/Bowater would not be permitted during the term of the license.

d) For all lands owned by GNP/Bowater on the Millinocket Lake Storage Reservoir and not noted previously in this section, a 500 foot no development conservation easement, with no timber harvesting in the first 100 foot zone, would apply.

Elbow, North and South Twin, Penadumcook and Ambaiejus Lakes ("Upper Penobscot Nills")

a) For all lands not owned by GNP/Bowater (identified as the south shoreline on Elbow Lake in Exhibit G, sheet 3 of 12 in the Penobscot Mills Application), the applicant would not be required to provide for a shoreline conservation easement.

b) All existing commercial and private leases issued by Bowater (primarily in the LURC zoned D-RS and D-GN shorefront and islands of these five connected lakes and identified in Figure E6-2 of the Penobscot Mills Application) as identified in the DEIS at 3-48 should be honored and renewed, subject to LURC regulations.

c) For all lands owned by Bowater on the Upper Penobscot Mills Reservoir and not noted previously in this section, a 500' no development conservation easement, with no timber harvesting in the

CI-A Cont

COMMENTS FROM CONSERVATION INTERVENORS

ON UPPER PENOBSCOT RIVER BASIN DEIS

first 100 foot zone would apply. To permit limited new development while still preserving the unique backcountry resources, an exception could allow for the sale of additional leases in those areas currently zoned D-RS and D-GN by LURC as defined in Figure E6-2 of the Penobscot Mills Application, which have not been build out to their maximum development under current LURC zoning. This represents 60% of Ambajejus, 33% of North Twin, 10% of Pemadumcook, and 15% of the South Twin Lake shoreline (see Table 3-9 in the DEIS). In this way the conditions of the Shoreline Management Plan in the licensing process are defined and appropriately focus new development in areas where it has already occurred while not encouraging its spread. Any new development in these currently zoned development areas (D-RS and D-GN) could not be used as a catalyst to rezone adjacent lands for further development under the "adjacency" rezoning LURC regulations.

3) Ouakish Lake and Ferguson Pond

a) Exhibit G Sheet 3 of 12 of the Penobscot Mills Project application indicates that GNP/Bowater owns all of the shorelands on Quakish Lake and Perguson Pond. In the LURC zoned portion of Quakish Lake (Township T3 Indian Purchase) any existing leases should be honored and be renewable. The selling of new leases or private shorefront properties by Bowater would not be permitted during the course of the license in this township. GNP/Bowater would place a 500 conservation easement no timber harvest within the first 100 feet restriction on the LURC zoned properties within Township T3 Indian Purchase, with appropriate Rights of Way for Route 11. The shorefront on Quakish Lake and Perguson Pond in the Town of Millinocket would be subject to the zoning and development regulations of the town.

4) Dolby Pond, Shad Pond and East Millinocket River Reservoir

a) The applicant owns lands principally on the southeastern shoreline of Dolby Pond (Exhibit G, Sheet 5 of 12, Penobscot Mills application). All existing private or commercial leases with buildings on GNP/Bowater Dolby Pond property would be honored and be renewable, but no new leases would be permitted during the course of the license. Existing leases with no buildings would be repurchased by GNP/Bowater. The applicant would not have responsibility for additional shoreline conservation easements on non-GNP/Bowater properties on Dolby Pond.

b) The applicant owns all lands on Shad Pond and the East Millinocket River Reservoir (Exhibit G, Sheet 5 of 12, Penobscot Mills Project application). All existing development and private or commercial leases on these two waterbodies would be honored and be renewable. A 500 foot conservation easement with no timbering within 100 feet of the shoreline would apply to the south shoreline of the East Millinocket Reservoir and all of the Shad Pond shoreline, with exceptions being made for rights of ways.

з

CI-A

Cont



CI-B

Attachment B Water Use Model Limitations

In addition to the limitation noted in the text of our comments, the water use model developed by GNP has additional limitations.

CI-B 1. The model fails to include watershed scale processes.

The Conservation Coalition has repeatedly stressed the importance of watershed scale processes. In the DEIS, FERC stated that staff is uncertain what this means.

GNP controls not only the system of dams and impoundments in the Penobscot River basin, but the basin itself. This is important in the context of the relicensing procedure because timber operations have indirect effects which should be included when considering non-power values. Evapotranspiration and runoff rates, very important elements of the water balance, are directly affected by timber management.

In order to truly analyze the implications of system management on the non-power values of the basin, it is necessary to discuss flows in terms of all impoundments and flow routes through the basin (in a manner more specific than that of current model) and to consider the relation between timber management practices and water availability. PERC has addressed the second of these two needs to some extent by requiring land protection in the form of building setbacks. As already discussed, however, by accepting GNP's water use model, FERC fails to consider the importance of all of the waterbodies in the system and, thus, still fails to adequately consider watershed scale processes.

2. The model uses uncalibrated data.

The Conservation Coalition has expressed concern about the data used by GNP to develop its model. In the DEIS, FERC stated that staff is unsure what this means.

GNP created synthetic 'typical' years to represent average conditions and worst-case, or minimum available water, conditions by using the average and driest individual weeks, respectively within the period of record. FERC addressed this issue in the DEIS when it stated that "Because it consists of the data from the driest weeks within the 15-year period of record, the worst-case year represents more extreme conditions than the single driest year within that record." DEIS D-11.

The scenarios modelled by GNP are interesting but are without context since is impossible how often annual flows like the ones they 'constructed' will occur. The use of a probability or flow duration analysis would be more informative. A flow duration curve for the period of record and for each year would show the

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

We already provided responses to these concerns in Appendix D of the DEIS. In response to these comments and others, we have revised Appendix D to further clarify our responses. We have concluded that GNP's water use model is not a central issue, since water availability is not a central issue. The main issue in this proceeding is whether additional flows to the Back Channel are worth the cost in lost power and socioeconomic benefits as compared with the relatively small environmental benefits that would be gained.

RESPONSES TO CONSERVATION INTERVENORS ON UPPER PENOBSCOT RIVER BASIN DEIS

2

probability of a given flow being equalled or exceeded. To establish the feasibility of an alternative with regard to the typical, wet and dry years it is important to know the probability of these conditions occurring and therefore to know the probability of being subjected to stress as a result of the water use option under consideration.

3. The model excludes over-year storage.

The model considers each year independently, and outflows are adjusted to provide the same storage at the end of the year as existed at the beginning. Single-year simulations of this sort are overly restrictive and preclude the possibility of over-year storage which could be used to optimize operations.

FERC acknowledged this criticism of the model in the DEIS, but failed to directly address it. DEIS D-32. As FERC noted, GNP did, in fact, simulate all fifteen years individually as part of their study. This does not, however, change the fact that there has been no simulation that encompasses the fifteen year period in a continuous manner.

Reservoir operations modeling typically involve continuous modelling over the period of record in order to make use of overyear storage. The fifteen-year data set should be used for a continuous model instead of simulating each year in isolation.

4. The model is unable to consider conservation measures in the economic analysis.

FERC responded to this concern in the DEIS by stating that "The water-use model itself is not used to evaluate the affects [sic] of flow alternatives on project economics, so this concern is not relevant." DEIS D-32.

The economic analysis is based projected diminished power generation flows and head as 'predicted' by the simulation model. In this way, the water use model is an integral part of the economic analysis.

If valuable water savings can be achieved by updating equipment or implementing new water conservation practices, then such improvements would have to be accounted for somehow in the flow information upon which the economic analysis is based. These possibilities must be fully explored.

5. General Comments on the Nodel's Limitations

It is important to recognize the significant limitations of the current model. Even if it were refined in the ways suggested above important limitations would remain.

m

46

CI-B

The GNP water use model is a simulation model. It is designed to simulate reality based on past operations -- not to predict and not to optimize benefits. The model allows for an evaluation of what was -- in terms of water availability in the West Branch of the Penobscot -- but not of the possibilities of what could be under other, new circumstances.

ъ

CI-B Cont Any discussion of the limitations of the GNP water use model raises the question of whether a more sophisticated model would produce different results. The appropriate answer to that question is twofold. First, refinement of the existing model, along the lines suggested above, may produce different results. There is any opportunity to increase the specificity of the model with out increasing the uncertainty of the model since many of the refinements suggested would involve data sets already being used in the model and would introduce no new error. Changing the time step would, for instance, involve using the daily flow data for elements of the hydropover system instead of what is used now- the weekly average of that daily flow data.

Second, the model should be refined whether or not such a refinement will produce different results. This project involves the largest private hydropower system in the country. The value of the licenses to Bowster/GNP and the worth of the natural resources involved to the general public are extremely significant. A decision of this sort should not be based in any way, upon an overly simplistic model. FERC should be wary of setting such a precedent.

The existing model should be significantly refined or a new model should be developed if the model is to be used as a basin upon which to make long-term licensing decisions.



Lois D. Cashell Secretary, Federal Energy Regulatory Commission 825 North Capitol Street, N.E. Washington D.C. 20426

Panobscot Basin EIS, FERC Nos. 2572 and 2458 Riponemus and Panobscot Hills Projects

Dear Secretary Cashell:

On February 23, 1995, the day after the close of the comment period on the draft EIS in the above proceedings, Bowater/GNP and the Governor of Maine held a joint press conference to announce an agreement negotiated in secret during the comment period between the State and the applicant on the issue of shoreland protection and project boundaries in the above-referenced proceeding. In conjunction with the press conference held in Maine, Bowater issued a national press release on Pebruary 23 in which it announced the "donation" of conservation essements to the State of Maine, together with a program to repurchase or call \$200 million of outstanding corporate bonds and the election of the company's president as its Chief Executive Officer. (See attached Bowater wire service version of the Bowater press release.)

Of course, the need to expand project boundaries is a key issue in these proceedings and is discussed at length in the draft EIS. This is an issue that intervenors Conservation Law Foundation, Appalachian Hountain Club, Maine Audubon Society and American Whitewater Affiliation have commented on extensively to FERC. By this letter, these organizations wish to provide FERC with comments on the substance of the agreement announced at the press event (the "HOA"). Because we are commenting on the terms of the public announcement and NOA made public only after the close of the comment period, we do not believe that leave is required to include our comments in the record.

Nonetheless, if the Commission determines that leave is required, we respectfully request leave to include our comments into the record.' Good cause exists to grant our request. We do not think it incidental that the announcement of the MOA occurred the day after the closure of the comment period. Both the State and the applicant were well aware of the terms of their agreement

B Maine Othop: 119 Tillson Avenus. Rochland, Maine 04841-3632 + (207) 584-8107 + FAZ (207) 596-7708 ☐ Headquarters: 62 Summer Sheet, Boston, Massachuens 02110-1008 + (817) 350-0980 + FAX (817) 350-0080 ♡ Yammor (2016a; 21 Estis State Sheet, Monitosier: Vammor (26102-2152 + (802) 223-5992 + FAX (802) 223-0080

RESPONSES TO CONSERVATION INTERVENORS ON MEMORANDUM OF UNDERSTANDING ON UPPER PENOBSCOT RIVER BASIN DEIS

CIMOU-1 No response needed.

CIMOU-2 Opinion noted.

¹ We do not seek leave to respond to the comments submitted by the State, the applicant or others. In fact, as of the date of this letter, the State had not served any of our organizations with its comments.

-

7

CIMOU-3

such sarlier than February 23. The letter from the governor summarizing the agreement to FERC is dated February 16; the NOA itself is dated February 17, 1995. Because neither the State nor the applicant announced the agreement when it was reached on Feb. 16 and 17, did not notify other parties to this docket of their discussions but rather chose to delay the announcement until after the comment period ended, the applicant and the State removed the ability of our parties in these proceedings to comment on the MOA during the official comment period. We believe that it is important that the Commission have available to it the comments of our organizations that have advocated for the appropriate expansion of project boundaries to include shoreland areas. The timing of the MOA announcement should not be allowed to preclude comment by our organizations.

The State-Applicant MOA is Fundamentally Flawed and Cannot Be Accepted by FERC in Lieu Of Alternatives 1 or 2

The purpose of the MOA is to provide a substitute for the expansion of project boundaries as recommended in the DEIS. Under the MOA, the applicant would "donate" 250' wide easements (500' easements, on cartain water bodies), on (i) land it currently owns and (ii) only in the Ripogenus Project. The MOA excludes the entire Penobscot Mills Project; shoreland not currently owned by GMP; and does not include vegetation zones or other provisions set forth in the DEIS.

As a result, the MOA negotiated by the applicant fails to provide a substitute for the expanded project boundaries recommended in Alternatives 1, which we continue to support, or even Alternative 2. Rather, the MOA is a transparent attempt by the applicant to avoid the more rigorous mandates of either Alternative 1 or 2 and effect an end-run around the obligations the company bears to protect these shoreland. As more fully discussed below, FERC should not and cannot rely on the MOA as a substitute for the expansion of project boundaries.

1. The Agreement Excludes Critical Areas from Protection

A. There is No Factual or Legal Basis for Excluding the Penobscot Mills Project from Shoreland Protection

The MOA excludes the entire Penobscot Nills Project from the benefits of shoreland protection. Yet, there is no basis on which to exampt wholesale the Penobscot Hills Project from expanded project boundaries. Nost of the Project exhibits back country, relatively remote, undeveloped, high value characteristics similar to the Ripogenus Project, including largely undeveloped Class 1-A lakes. Recreational use by the public on these lakes is high and the Project includes areas which support unique, multiday, back country canceing. The Project includes miles of undeveloped, high value shoreland and important fisheries. The need for shoreland

Comment noted. While the Penobscot Project area is more developed than the Ripogenus Project area and the same level and type of protection may not be appropriate, staff determined that some protection was necessary to protect the aesthetic and recreational resources of the shoreland areas directly adjacent to the project impoundments. Staff reviewed comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. We still recommend protection measures for the Penobscot Mills shorelines. See section 4.9 for further discussion regarding shoreland protection zones.

RESPONSES TO CONSERVATION INTERVENORS

ON MEMORANDUM OF UNDERSTANDING

ON UPPER PENOBSCOT RIVER BASIN DEIS

E-50

CIMOU-2

CIMOU-1

Cont

protection in these undeveloped portions of the Penobscot Mills Project is just as high as in the Ripogenus Project.

Yet, the MOA excludes entirely protection of these high value lakes. There simply is no logical basis on which to draw a wholesale distinction between the Ripogenus and the Penobscot Wills Project. The same aesthetic, wildlife, water quality, recreational values that GMP has finally agreed should be protected at Ripogenus as a part of these proceedings also need to be protected at the Frondemot Wills Project. Nor is there a legal basis for drawing a wrolesale distinction between projects. FERC shoreland protection regulations and the obligation to ensure an appropriate balance between power and non-power values apply equally to the Ripogenus and Penobscot Wills Projects.

It is true that the Penobscot Mills Project includes areas with developed shorelands, more so than Ripogenus. As the DEIS acknowledges, and as our organizations agree, it is appropriate to permit additional, clustered development in these locations. The Shoreland Management Plan, properly structured, would allow for additional development in these areas.

 The Agreement Avoids GNP's Obligation to Adequately Protect Lakes in the Rivogenus Protect

One obvious goal of Bowater/GNP in entering into the MOA is to avoid any financial obligations associated with protecting the Ripcgenus Project. To that end, the MOA is glaringly transparent in its effort to avoid the need for the applicant to obtain from willing sellers, in fee or easement, shoreland property not currently owned by Bowater/GNP.

Bowater's major objection to the requirement to protect non-GNP land appears to be cost, an objection the State has accepted as a limiting factor. There is no basis, however, on which to -unclude that the cost of obtaining protection in the Ripogenus :- ject would do more than add a nearly incalculably small cost to two company. As the DEIS makes clear, the \$2-4 million estimate would increase the cost of generating electricity by a tiny fraction. This is a reasonable estimate and imposes a lenient obligation on the applicant. We fail to understand why these costs would be so crippling.²

RESPONSES TO CONSERVATION INTERVENORS ON MEMORANDUM OF UNDERSTANDING ON UPPER PENOBSCOT RIVER BASIN DEIS

CIMOU-4

The staff revised potential cost estimates for the shoreline easements upon review of comments received during the DEIS comment period and updated land valuation information. Staff determined the potential costs, based on waterfront footage for the easements proposed in Alternative 1, approximately \$24.6 million, would be greater than previously estimated in the DEIS. Our qualitative evaluation of benefits suggests that the additional protection of the 500-foot expansion does not merit the much higher cost of that alternative as compared to the recommended alternative (\$24.6 million versus no direct costs). See section 4.9 for further discussion.

Cont

CIMOU-3

² There is a fundamental inconsistency reflected in the State/applicant announcement. On the one hand, the State and the applicant assert that the land does not face a significant risk of development; on the other, they limit the amount of shoreland protection in the MOA largely due to an assertion that the <u>current</u> development value of the lands is so high that GMP cannot incur the cost of protecting the land without some dire, unspecific socioeconomic impacts.

II. The NOA is Unilaterally Voidable By the Applicant

Under the express terms of the MOA, the applicant retains the <u>unilateral</u> right to void the agreement if the Commission orders project boundaries other than those accepted by GNP in the Agreement.

This places FERC in an untenable position. In the case where FERC adheres to the legal obligation to expand the project boundaries at Penobscot Hills, GNP retains the right under the NOA to void its Ripogenus shoreland protection agreement at its unilateral discretion.

Because there is no factual or legal basis to exclude Penobscot Hills or non-GHP land from the need for project boundary expansion, the effect of the MOA is to create two, rather stark choices. One, which the applicant and the State propose, is to accept shoreland protection only as provided in the MOA, even though that would violate the Federal Power Act and FERC regulations because of the omission of the Penobscot Hills Project or non-GHP land. Or, two, as we suggest, maintain the appropriate recommendation to expand project boundaries at both projects and on non-GHP land as set forth in the DEIS. Because the Bowater/State HOA alternative violates FERC's own mandates and regulations, only option 2 is viable and legal.

III. The Terms of the Shoreland Protection are Vague And Do Not <u>Provide the Level of Protection that FERC Regulations Would</u>

A. The NOA Fails to Provide An Equivalent Level of Protection as the DEIS for Vegetation Iones

The MOA proposes to establish as the terms of the easements the provisions of the 1981 easements on the West Branch. The terms of the 1981 easements do not provide for any vegetation zones. Under P-SL1 and P-SL2, ch. 10 of LURC's rules and regulations, timber harvesting is permitted to the watermark with certain restrictions, most importantly a preclusion sgainst clearcutting within 50' of the high water mark in the shoreland zone. In the remaining 200' back, small clearcuts otherwise known as patchcuts are permitted as long as they are smaller than 14,000 square feet.

Alternative 2 in the DEIS includes a no-cut, vegetation zone of 100' within the expanded project boundaries. Although we continue to believe that the 250' vegetation (no-cut) zone in Alternative 1 is preferable, even the terms of Alternative 2's 100' no-cut zones are far more protective of non-power values than the MOA and as the DEIS correctly notes, the 100' zone is more protective of non-power values than LURC zoning. Moreover, incorporation of standardized no-cut zones in project boundaries provides, as with non-development zones, an entirely different and needed level of protection for these resources than zoning.

RESPONSES TO CONSERVATION INTERVENORS ON MEMORANDUM OF UNDERSTANDING ON UPPER PENOBSCOT RIVER BASIN DEIS

CIMOU-5

4

- In the Applicant's Proposed Alternative, staff incorporated GNP's proposed conservation easements for the Ripogenus Project area under the terms described in the MOU (see section 4.9.1). In Alternative 2 (recommended alternative), staff proposed for the Ripogenus Project area two options: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion on GNP-owned lands (see section 4.9.3). For the Penobscot Mills Project area, State recommended a 200-foot boundary expansion on GNP-owned lands. The staff determined the recommended alternative provides long term protection of valuable shoreland resources (see section 5.3.4).
- CIMOU-6 See comment above. In the recommended alternative, GNP would be responsible for recreational facilities and potential enhancements or mitigation of recreational resources in association with project operations or license conditions (see section 4.9.3).

CIMOU-5

RESPONSES TO CONSERVATION INTERVENORS ON MEMORANDUM OF UNDERSTANDING ON UPPER PENOBSCOT RIVER BASIN DEIS

5

In addition, under the terms of the 1981 easements, the State is required to bear the costs of any recreational improvements, relieving Bowater/GNP of any financial obligation to provide for recreational needs, a critical non-power value, even as GNP retained the right to approve recreational uses. The MOA clearly contemplates the same arrangement. The 1981 easements also allow for a multitude of uses inconsistent with the goals of shoreland protection, such as mineral extraction, new access roads and new hydroelectric dams.

In sum, the 1981 easements do not impose any additional restrictions on timber harvesting and relieve GNP of responsibility for recreational management in the shoreland zones, shifting the financial obligation to the State. Bowater/GNP is clearly seeking the same arrangement under the MOA. These terms are inadequate and fail to attain the goals that the DEIS establishes.

B. The Limitations in the MOA Are Based On False Assumptions

The parties to the MOA accept uncritically Bowater's assertions as a basis on which to limit the kind and scope of protection for non-power values. In addition to claims of economic doom, the State appears to adhere to the view that FERC is proposing that Maine be required to sell State land against its will to GMP, so that GMP can then protect the land. The State repeats its threat of litigation on behalf of all unwilling sellers.

FERC has attempted to clarify this issue. The State appears to prefer to maintain this strawman, a strawman that provides no basis on which to reject the requirement that Bowater seek to obtain, from willing sellers, shorefront land. As Bowater now accepts the legitimacy of protecting shoreland outside of project boundaries, FERC should require the company to seek to protect nonproject lands if landowners in the Ripogenus Project prove unwilling to sell.

IV. The Accessent and its ferms are Hon-enforceable

There is a fundamental distinction from an enforcement perspective between including shoreland areas in project boundaries and the terms of the ensements as proposed in the MOA. The former provides much improved opportunities for public input if concern arises about potential violations of the shoreland areas by the applicant. Under the MOA, only the State and the applicant have enforceable legal obligations to each other: the public, resource segncies and FERC have no standing and are excluded from any forum within which to raise concerns.

V. The Bowater/GMP Credibility Gap Grows

CIMOU-9

Although it is clear that FERC cannot rely on the NOA as a

The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). Under the recommended alternative, the proposed conservation easements and project boundary expansion would be solely on GNP owned land and purchase of non-GNP owned properties would not be required (see section 4.9.3).

CIMOU-8 Opinion noted.

CIMOU-7

CIMOU-9 Opinion noted.

E-53

CIMOU-7

CIMOU-6

Cont

6

CIMOU-10

Opinion noted.

RESPONSES TO CONSERVATION INTERVENORS

ON MEMORANDUM OF UNDERSTANDING

ON UPPER PENOBSCOT RIVER BASIN DEIS

substitute for either Alternative 1 or 2 as set forth in the DEIS, it is equally clear that the Bowater/State NOA demonstrates yet again the inconsistencies and credibility gaps of this applicant. For example, the company has aggressively argued against the need for any form of term or persanent shoreland protection, relying on LURC zoning. Under the MOA, however, it accepts the need for more significant forms of protection than zoning provides. In the past, Bowater/GNP argued that LURC non-development zones of 75' or 100' were adequate and that even 200' zones were burdensome. Now, it proposes 250' easements for much of the Ripogenus Project area and 500' zones easements elsewhere -- far more protection than LURC somes provide and greater widths than even Alternative 2 contemplates. In the past, Bowater/GNP has lashed out at our organizations for proposing mitigation and enhancement measures outside of strictly defined project boundaries -- and has opposed review by FERC of environmental impacts outside of project boundaries. Under the NOA, the company itself now proposes shoreland protection outside of project boundaries.

It is telling that even as the public hearing was taking place on January 25, Bowater/GNP evidently conceived of the proposal contained in the MOA, even as it railed against shoreland protection at the public hearing in Millinocket. Within days of the hearing, on January 30, Bowater requested an extension of time to comment, so that it could finalize its pending agreement with the State. This pattern of behavior, consistent with the company's past practices, continues to provide grounds to call into question the validity of its assertions and credibility.

<u>Conclusion</u>

It is perhaps not difficult to understand why the applicant would rush to the State to propose an agreement along the terms of the MOA. In contrast to either shoreland protection proposal reviewed in the DEIS, the MOA excludes from protection the entire Penobacot Hills Project, including remote, undeveloped shoreland of high value; would relieve Bowater/GNP of any obligation to seek to protect shoreland it does not currently own in the Ripogenus Project; is voidable -- at Bowater/GNP's unilateral discretion; is vague on the terms governing the easements; omits the any vegetation zone; and relieves Bowater/GNP of any concern that violations of the terms of the easements would be enforceable as part of its hydropower licenses. Because of these defects, the NOA, by its terms it fails to provide the same level and quality of protection as either Alternative 1 or 2 in the DEIS.

Yet, despite these crippling deficiencies, the HOA does signify several important elements. One, the applicant has validated our view that Zoning regulations do not provide the same level of long-term protection as fee or essements. Two, by proposing essement zones of 250' to 500', the HOA supports our view that 200' wide Zones are not adequate for protection. Three, by

п-54

CIMOU-9

Cont

proposing to protect shoreland <u>outside</u> of project boundaries, the parties to the MOA accept the concept that mitigation and enhancements outside of project boundaries are acceptable and legitimate.

Properly constituted easements held by the State may be a fine way to protect shoreland values. But "easements" negotiated in secret as an end run around FERC provisions, which fail to provide an equivalent or adequate level of protection as Alternatives 1 and 2, do not meet the goals set forth in the DEIS and the protection of non-power values.

FERC should reject the applicant's attempt to once again write its own permit conditions with the assistance of the State, -- as it has in changing the water quality law applicable to Ripogenus and in inducing the State to valve water quality standards at Penobacot Hills. FERC should accept Alternative 1 with the clarifications set forth in our comments filed in a timely manner on Pebruary 22.

Thank you for your attention.

Respectfully submitted,

Daniel L. Sosland, Reg

Conservation Law Foundation 119 Tillson Avenue Rockland, ME 04841 (207) 594-8107

Kenneth D. Kimball, (603) 466-2721

Michael Cline, M.D. Naine Audubon Society.

PO Box 6009 Falmouth, ME 04105 (207) 781-2330

1 Q. 600 Richard J// Bowers American Whitewater Affiliation 1430 Ferwick Lane Silver Spring, HD 20910 (301) 589-9453

ce: Service List

Ph.D. Appalachian Nountain Club PO Box 298, Rt. 16 Gorban, NH 03581

- TOnno

Margaret Bowman. Lag. American Rivers **BO1 Pennsylvania Ave.** SE suite 400 Washington, DC 20036 (202) 547-6900

CIMOU-10 Cont

Copyright 1995 Business Wire, Inc. Business Wire

February 23, 1995, Thursday DISTRIBUTION: Business Editors LENGTH: 591 words HEADLINE: BOWATER ELECTS MEMIROW CHIEF EXECUTIVE OFFICER: DEBT TENDER OFFER AND CONSERVATION GRANT PROPOSAL ANNOUNCED DATELINE: GREEVVILLE, S.C. BODY:

Feb. 23, 1995 -- Bowater Incorporated (NYSE:BOW) today announced the election of Arnold H. Nemirow, 51, as Chief Executive Officer, effective March 1, 1995. Mr. Nemirow already holds the title of President. In keeping with a previously announced transition, Mr. Memirow will assume the duties of Chief Executive from Anthony P. Gammie, 60, who has held that position since January 1983. Mr. Gammie will remain as Chairman until March 1996. Nemirow has been President and Chief Operating Officer of Bowater Incorporated since joining the company in September 1994, from Wausau Paper Hills Company, where he was President and Chief Executive Officer. Earlier, he served in senior positions with Great Morthern Nekoosa Corporation, including Chairman, President and Chief Executive Officer of Mekoose Papers Inc. from 1988 to 1990.

Additionally, Donald G. McHeil, 44, was elected Senior Vice President. Mr. McHeil serves as President of Bowater's Great Northern Paper Division based in East Hillinocket, Maine. Earlier, Mr. McHeil held the position of President and General Menager of Bowater Mersey Paper Co., Ltd., in Liverpool, Nova Scotia, Canada.

The company also announced the commencement of an offer to purchase its outstanding 8.5% Notes due 2001 having a face value of \$200 million. The offer begins on February 24, 1995, and will expire on March 7, 1995. The purchase price for the 8.5% Notes will be determined by using a yield to maturity of the 7.5% Treasury Note due Hovember 15, 2001 at the time the holder agrees to tender the bonds plus a fixed spread of .375 percent. Herrill Lynch & Co. will serve as the exclusive dealer manager of the offering and Morrow & Co., Inc. will serve as information agent.

Bowster additionally stated that it yesterday filed with the Federal Energy Regulatory Commission (FERC) a Hemorandum of Understanding between Bowater's Great Northern Paper Division and the State of Haine wherein the company proposes to convey to the state a conservation easement protecting approximately 80 miles of shoreline along the West Branch of the Penobscot River. The essement would be conveyed upon successful FERC relicensing of Great Northern Paper's Ripogenus and Penobscot Hills hydroelectric projects and is intended to expand the shoreline protection and expedite relicensing. Memirow said, "I am confident that Bowater's continued cost reduction efforts and balance sheet improvements will position us for higher levels of financial performance. I look forward to the opportunity to lead Bowater." Bovater Incorporated is a manufacturer of virgin and recycled fiber newsprint, coated and uncoated virgin fiber and recycled groundwood publication papers, bleached kraft market pulp and lumber. It is

RESPONSES TO CONSERVATION INTERVENORS ON MEMORANDUM OF UNDERSTANDING ON UPPER PENOBSCOT RIVER BASIN DEIS

COMMENTS FROM CONSERVATION INTERVENORS

ON MEMORANDUM OF UNDERSTANDING ON UPPER PENOBSCOT RIVER BASIN DEIS

RESPONSES TO CONSERVATION INTERVENORS ON MEMORANDUM OF UNDERSTANDING ON UPPER PENOBSCOT RIVER BASIN DEIS

also a converter of communication papers for computer forms and other business applications. Rowater Incorporated common stock is listed on the New York Stock Exchange, U.S. regional exchanges, the Londom Stock Exchanges and the Swiss Stock Exchanges.

CONTACT: Bowater Inc. Media contacts: Robert D. Leahy 803/282-9571 (Office), 803/234-0941 (Home) William B. Kearney 803/282-9593 (Office), 803/458-9817 (Home) Analyst contact: SuAnne B. Aune 803/282-9560 (Office), 803/879-1877 (Home) LANGUAGE: ENGLISH LOAD-DATE-HDC: Pebruary 24, 1995

.



The following comments are provided in three parts: first, we give an overall assessment of the adequacy of the DEIS in addressing issues of concern to this Department; second, Attachment A is a section-by-section analysis of the DEIS; and, third, Attachment B contains updated and/or modified recommendations and fishway prescriptions, that have been previously provided by the Department pursuant to Sections 10(j) and 18 of the Federal Power Act (FPA).

GENERAL COMMENTS

DOI-2 The DEIS fails to give equal consideration to fish and wildlife resources, potential recreational needs, and to tribal issues that have been raised during these licensing proceedings by the Department and the Penobscot Indian Nation (PIN). As discussed in greater detail in Attachment A, the Commission has given undue weight to the economic interests of the applicant in the analysis of impacts to fish and wildlife resources, including endangered species, for which this Department is responsible. The Commission has also arbitrarily dismissed important alternatives to the proposed action, such as conservation. Some of these alternatives could help achieve the applicant's stated energy capacity needs, while at the same time avoiding and reversing environmental harm to local and regional natural resources.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

No response needed.

The staff reviewed available information regarding the Penobscot Indian Nation's claims to lands and rights within the branches of the Penobscot River and concurs with the decision of the Maine State Department of the Attorney General (see section 4.11.1.2). The staff finds that consideration of the Penobscot Indian Nation's traditional practices within the project area are outside the scope of Section 106. The staff, therefore, does not recommend including the Penobscot Indian Nation as a concurring party to the Programmatic Agreement associated with the Ripogenus and Penobscot Mills projects. The staff acknowledges Penobscot Indian Nation's interest in the management of historic properties potentially eligible for listing in the National Register of Historic Places within the project areas. Accordingly, the staff recommends that GNP consult the Penobscot Indian Nation during the development of the revised Cultural Resource Management Plans for the Penobscot Mills and Ripogenus Projects.

Energy conservation was identified as an alternative during the scoping process; however, we concluded in the DEIS that energy savings gained through conservation have largely been offset by increased energy demand from GNP's efforts to modernize its plants. This conclusion was based upon our review of GNP's conservation and modernization programs, as documented in Exhibit H. and the memorandum of Owen Merrill of GNP, both of which are based on actual plant data. The Conservation Intervenors' claim that there is an enormous conserved power potential in the GNP facilities is based upon generic or theoretical evidence, which the staff did not accord the same weight. The staff also notes that many of the arguments posed by the intervenors have been raised throughout this proceeding and have been addressed by the applicant. Detailed responses to additional comments. on economics and effects on fish and wildlife resources are provided below.

DOI-1

Honorable Lois D. Cashell

In the DEIS, the Commission fails to recognize and to properly account for its trust responsibility to the Penobscot Indian Nation (PIN), a Federally recognized Indian Tribe. In dealing with Indian Tribes, the Federal government is judged by the most exacting fiduciary standards. <u>Seminole Nation v. United States</u>, 316 U.S. 286, 297 (1942). As an agency of the Federal government, the Commission is charged with the United States' fiduciary responsibilities towards Indian Tribes. <u>Covelo Indian Community v. FERC</u>, 895 F.2d 581, 586 (9th Cir. 1990). Moreover, any Federal action is subject to this fiduciary responsibility towards Indian Tribes. <u>Nance v. Environmental Protection Agency</u>, 645 F.2d 701, 711, <u>gett. denied</u>, 454 U.S. 1081 (1981). This duty must be fulfilled both procedurally and substantively.

While the DEIS briefly mentions the PIN in its section dealing with Cultural Resources (Section 4.11), the Commission fails to recognize and protect the PIN's rights and trust resources, including its property interest in the fishery resource of the Penobscot River. The Commission has ignored and disregarded the comments proffered by the BIA and the PIN during the scoping process, and thus neglects to consider tribal interests in issues of importance in this DEIS, such as water quality, toxics, stream flows, fishery resources, and cultural resources. These specific issues are addressed in Attachment A.

The Department further points out that the Commission's treatment of tribal rights and interests in these proceedings is contrary to the provisions of the President's Memorandum dated April 29, 1994, entitled "Government-to-Government Relations With Native American Tribal Governments." Pursuant to this Memorandum, each executive department and agency "shall assess the impact of Federal Government plans, projects, programs, and activities on tribal trust resources and assure that tribal government rights and concerns are considered during the development of such plans, projects, programs, and activities." The proposed issuance of Federal licenses for hydroelectric developments is certainly the type of activity for which such assessments and consideration of tribal interests are required. However, in this situation, the Commission has failed to do so.

The shortcomings in the DEIS must be remedied prior to the Commission reaching final licensing decisions on these projects. The Department is prepared to assist the Commission and its staff in producing an improved EIS, and looks forward to additional consultation on the issues discussed herein.

Thank you for the opportunity to comment on this DEIS.

Sincerely,

Andrew L. Raddant Regional Environmental Officer

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-4 See response to DOI-2

2

DOI-4

REF: FERC # 2572, 2458

ATTACHMENT A: SECTION BY SECTION ANALYSIS OF THE PENOBSCOT RIVER BASIN DEIS

PURPOSE AND NEED

The Council on Environmental Quality's (CEQ) National Environmental Policy Act (NEPA) Regulations require that an environmental impact statement (EIS) contain a statement on the underlying purpose and need to which the agency is responding in proposing the alternatives, including the proposed action (40 CFR Sec. 1502.13). This portion of the document is crucial in that it forms the foundation upon which the alternatives requirement of NEPA can be met.

The Penobscot River Basin DEIS does not contain a clear description of the underlying "purpose and need," as required by NEPA. Although the DEIS describes the "action" as the potential issuance of new long-term licenses for the two projects, and discusses how the document will be used to "evaluate potentially significant environmental and development effects of the projects," the overall/basic project purpose(s) and need(s) are not clearly defined.

The DEIS does cite to Section 10(a) of the FPA regarding the Commission's requirement to license only projects that are best adapted to a comprehensive plan for improving or developing a waterway for various beneficial public uses in addition to waterpower development: interstate commerce (i.e., tourism); protection, mitigation and enhancement of fish and wildlife (including spawning grounds and habitat); and, recreation. Section 10(a) establishes that a variety of needs specific to the Penobscot River Basin can be identified in the EIS, and that the purpose of the Federal action can be to satisfy those needs as much as possible in order to ensure that comprehensive river basin development is achieved. However, this implies that there may be a variety of project purposes" fill needs, specific to the Penobacot River Basin. This portion of the DEIS does not elaborate on what the specific needs and purposes are that should be addressed in order to ensure that comprehensive river basin development is achieved.

The issue is further obscured because, in the Purpose and Need section (1.0), the DEIS only identifies a <u>single</u> need, that being the applicant's (Great Northern Paper, Inc.) need to remain competitive in the pulp and paper industry by having a reliable source of inexpensive electric power. A secondary need for economic prosperity and continued employment in the project locale and elsewhere in northern Maine is expressed in subsequent portions of the document.

A clear statement on underlying, basic project purpose is also critical in the context of subsequent action by the U.S. Army Corps of Engineers (Corps) under Sec. 404 of the Clean Water Act (CWA). The Corps will need to authorize the filling of wetlands and impacts associated with the enhancement measures recommended in the DEIS. In accordance with CWA regulations (40 CFR 230), the project purpose must be clearly identified prior to analyzing the proposed action and alternatives.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-5

We disagree. The EIS clearly states in section 1.1 that the project purpose is the relicensing of two hydroeleclectric projects on the West Branch of the Penobscot River. Section 10(a) of the FPA provides the context within which the Commission bases its decision, but the beneficial public uses described in section 10(a) do not represent the "project" purpose or need.

The project "need" is the applicant's need for power, which is described in section 1.2. We evaluated need in the context of the competitive nature of the pulp and paper industry and the applicant's need for inexpensive electricity, as well as its requirement for 40-hertz energy.

DO1-5

REF: FERC # 2572, 2458

DOI-5 Cont We recommend that the Commission staff revise this portion of the DEIS to identify the purpose(s) and need(s) to which they are responding in these licensing proceedings. The DEIS should also state whether the applicant's basic project purpose (e.g., competitive pulp and paper operations) is different from that of the Commission under the FPA (e.g., comprehensive river basin development accommodating multiple beneficial public uses).

We also recommend that needs beyond increased generating capacity and economic growth be included in this portion of the DEIS in order to satisfy the comprehensive development requirement (multiple beneficial public uses) contained in the FPA. As stated in the Department's comments during scoping for this DEIS, the need to protect, restore and enhance fish and wildlife resources, and the need to protect Penobscot Indian Nation rights should also be identified at the outset of the document. (See the Fish and Wildlife Service's (FWS) September 2, 1993 letter to the Commission (hereafter FWS scoping comments); and the Bureau of Indian Affairs' (BIA) September 8, 1993, letter to the Commission (hereafter BIA scoping comments).)

In addition, it is unclear how the Federal Energy Regulatory Commission (FERC) staff reached the conclusion at page 1-3 that Great Northern Paper's (GNP) competitive position, which is dependent upon a reliable source of inexpensive power, is related to the fact that GNP's capacity outstrips its demand for products in most of its markets. It would seem logical that if GNP were to reduce its level of production, and consequently its need for electric power, the company would benefit economically by reducing ongoing losses caused in part by producing a product for which there is limited demand.

SCOPE OF THE EIS

The DEIS states that the Commission was requested to combine into one environmental document all licensing actions involving the Penobscot River Basin, including its West Branch, that are currently pending or will soon be before the Commission. This includes applications for relicensing the Great Northern Storage Project, located upstream from Ripogenus, and the hydroelectric facilities that are being addressed in the Lower Penobscot River Basin DEIS, issued in November 1994.

DOI-8 As stated in the DEIS the Commission has decided not to address the upriver storage projects, which had been recommended by several intervenors. Instead, the Commission opted to deal with cumulative impact issues that may be germane throughout the West Branch, such as the effects of hydropower development on fishery resources. We believe that this approach is short-sighted, and will limit the Commission's ability to arrive at the best comprehensive plan for the basin. The applicant currently operates 12 storage facilities on the West Branch of the Penobscot, aside from those at Ripogenus and Penobscot Mills. Given the absence of generating facilities at any of these projects, the applicant could alter current plant operations in order to utilize the upriver facilities for power generation. This change in operations could address the basic project purpose of providing a reliable source of inexpensive electric power to ensure competitive pulp and paper operations, while allowing for greater allocation of water resources for habitat protection and other environmental values at the Ripogenus and Penobscot Mills Project.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-6 See response to DOI-2.

2

- DOI-7 The staff finds that Alternative 2 (final recommendation) would increase the cost of power from the two projects by about one percent. We conclude that the competitive position of the two mills would not be adversely affected as their total costs of production would increase only minimally (i.e., less than one percent depending on the proportion of total production costs that are accounted for by electric power). The current market supply and demand conditions are the dominant influences on GNP's competitive position.
- DOI-8 GNP has agreed to the inclusion of a reopener to allow modification of the water use plan, if analysis of the upper projects during relicensing produces unanticipated findings that require changes. We recommend the orders for both the Ripogenus and Penobscot Mills projects include an article containing a reopener clause for consideration of changes in water use in the West Branch when the upper storage projects are evaluated for relicensing.

DOI-6

DOI-7

3

REF: FERC # 2572, 2458

DOI-9 We also note that the Commission has not succeeded in the DEIS in its intent to fully address cumulative impacts to natural resources within the West Branch. The document does not address the impacts to landlocked salmon, brook trout and anadromous species (principally sea run Atlantic salmon) at the upstream and downstream projects, including those on the main stem, that were intentionally excluded from the DEIS. Further, the Commission fails to consider the cumulative impacts of the West Branch projects on the rights and Federal trust resources of the Penobscot Indian Nation.

> Regarding the Commission's rationale for using separate NEPA documents for the lower basin projects and those on the West Branch, we believe that the linkage and similarities may be greater than is presented in the DEIS. First, the upper basin projects regulate and augment water flows throughout the remainder of the river, such that the feasibility of continued operation of existing, as well as construction of any new, hydroelectric facilities downstream would likely be compromised if more natural river flows were to return. (This is a possible outcome, based on agency and PIN recommendations for restored aquatic habitats at the West Branch Projects.) Second, the West Branch historically supported runs of anadromous Atlantic salmon, and provided spiritual and physical sustenance as well as a means of transportation and trade for native peoples inhabiting the basin. Both are important issues to the Department in the licensing proceedings in the lower river basin.

> The Commission's charge under Sec. 10(a) of the FPA is in terms of a single comprehensive plan for the waterway(s), which in this case includes the entire Penobscot River Basin. Accordingly, decisions reached in each DEIS should not be made in a piecemeal fashion, but rather reflect a consideration of cumulative impacts, resource needs and balance throughout the basin.

PROPOSED ACTION AND ALTERNATIVES

In the DEIS, the Commission considers a narrow range of alternatives in evaluating whether to issue new licenses for the Ripogeaus and Penobscot Mills projects on the West Branch. This includes the applicant's proposal, and operational alternatives that have been recommended by the Commission staff, Penobscot Indian Nation (PIN), resource agencies, and conservation intervenors. Absent from the analysis are any alternative actions that could be taken by the applicant to improve competitiveness and preserve jobs in its paper mills (presumably the underlying and basic project purposes) that do not involve hydropower operations at the two projects. By doing so the Commission staff could have identified additional alternatives that met the basic project purpose(s) and also improved environmental conditions at the projects.

As was stated in the FWS scoping comments, additional alternatives should have been considered in the DEIS that involve installation of generating facilities at the applicant's dams on the West Branch of the Penobscot River. This consideration should include the licensed and unlicensed dams mentioned above, and also several storage/diversion dams at the Penobscot Mills and Ripogenus projects, specifically Ripogenus Dam, Stone Dam, and Millinocket Lake Dam. Retrofitting these dams with generating facilities could offset some of the power losses associated with the environmental measures evaluated elsewhere in the DEIS.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-9 See response to DOI-8, with respect to cumulative effects and the upstream projects. Restoration of anadromous species to the West Branch of the Penobscot River has not been proposed in any current plans and is therefore not a reasonably foreseeable action that can be considered in this EIS. See response to DOI-2.
- DOI-10 We have included analysis of the effects of relicensing of the projects in the West Branch of the Penobscot River on the projects in the lower Penobscot River. See also our responses to DOI-9 and DOI-2.
- DOI-11 See response to DOI-8. Installation of new generation facilities at the upstream projects is not a reasonably foreseeable action at this time but could be considered during relicensing of those projects. The feasibility of additional generating facilities at other developments within the Ripogenus and Penobscot Mills projects has already been evaluated by the applicant. Specifically with respect to Stone Dam, a new generation facility to provide flows to the Back Channel would not be economically feasible (see GNP's responses to FERC's October 28, 1993 Additional Information Request No. 4).

DOI-10

DOI-11

4

REF: FERC # 2572, 2458

DO1-12 AL

We also believe that the Commission should have more fully considered alternative water level regimes in the West Branch, including an independent modeling of streamflow. Although the applicant prepared a Water Use Plan (WUP), based on its data and perceived demands for water, and used this as a basis for subsequent streamflow/lake level recommendations is the application for new licenses, the FWS also recommended additional analyses. (See FWS scoping comments.) The Commission addressed some of the issues raised by the FWS in Appendix D of the DEIS, but did not pursue any of these as alternatives, nor did it include an evaluation of whether the applicant's basic project purposes could be met.

We note that in the Lower Penobscot River DELS, the Commission made an independent analysis of alternatives using the FWS' Atlantic salmon model, ASAL, after modifying inputs that had been developed by a team of interagency, tribal and other interests. It appears that the Commission has been less willing to attempt to replicate the applicant's WUP, or to modify it to address other issues, as has been recommended by the FWS and others. (In contrast to the ASAL model runs, the WUP was entirely applied by the applicant without assistance from the agencies or other interests.)

Other Issues Involving Alternatives

 Use of economic analyses in evaluating alternatives. Under Alternatives 1 and 2 the Commission considers, but dismisses a number of potential environmental improvements due largely to economic consequences. The Commission is reminded that:

Reasonable alternatives include those that are practical and feasible from the technical and economic standpoint and using common sense, rather than simply desirable from the standpoint of the applicant.¹

It appears that the Commission has given undue weight to inappropriate economic factors in deciding whether a particular alternative should be considered. The Commission should provide a more thorough explanation of the methodology and rationale it applied in factoring project economics into its alternatives analysis, and should illustrate that the methodology is consistent with the FPA and a revised definition of project needs and purposes.

 <u>No Action Alternative</u>. Under NEPA the Commission must include a "No Action" alternative in this and other EIS documents. According to the DEIS for the Penobscot River Basin, "no action" is considered to be the issuance of annual licenses for the next 30 to 50 years, with no change in current operations, including modifications to benefit fish and wildlife and other environmental values.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-12 We have used the applicant's water use model to evaluate additional alternatives beyond those proposed by the applicant. Some of these model runs were made by the applicant at FERC's request; however, we have independently evaluated the model and concluded it is adequate to evaluate flow needs in the areas affected by the projects. We have fully addressed issues raised about the model in Appendix D and we have revised that appendix to address additional comments by various parties on the DEIS.
- DOI-13 The Commission has changed its approach to economic analysis to better reflect the changing conditions in the electric power market brought about by deregulation. We have adopted a current cost approach under which a project's cost of producing power and the value of its output (i.e., the cost of acquiring a project's output from the least-cost alternative) are based on current market costs and prices. We identified the most feasible least-cost source of alternative power for GNP; it was found to be firm purchases from Bangor Hydro-Electric (BHE).

The staff assumes that there will be no inflation and escalation of capital costs fuel prices, operating and maintenance costs, and power prices over the 30 year analysis period. Current year capital costs are amortized over the 30 year financing period at the cost of capital appropriate for the applicant. No future inflation or escalation is assumed beyond the license issuance date, and no long-term levelized values for energy and capacity are used (i.e., their use implicitly incorporates assumptions about long-term inflation and escalation rates into an analysis). The staff's position is that our revised approach to economic analysis explicitly includes the appropriate economic factors, and accurately reflects the current conditions in the electric power industry. All the economic

DOI-13

DOI-14

E-64

CEQ Memorandum: Questions and Answers About the NEPA Regulations

5

REF: FERC # 2572, 2458

This view of the "No Action" alternative is incorrect. The issuance of an interim, annual license for the pending projects represents an "action" by the Commission. If the Commission's proposed "No Action" alternative was the preferred federal action, the Commission would also be carrying out other actions on these projects including its safety and compliance inspections, collection of annual charges, and other administrative duties. If threats to public safety are discovered in the inspection process, including possible dam safety problems, or violations of the existing terms and conditions in the licenses become apparent, the Commission staff would not hesistate to take a variety of actions, including ordering major repairs or implementation of enforcement proceedings.

The "No Action" alternative is clearer when the licensing action involves a proposed new data or modification of an existing facility; that is, the application for an original license or modification is denied, the new project does not go on-line, and associated environmental impacts do not occur. The same is not true for hydroelectric projects that are reauthorized by the Commission on a yearly basis while relicensing decisions are pending. Impacts due to impeded fish passage, degraded habitst, and impaired public access continue, and are sanctioned through the annual licensing process. As a result, the project is effectively provided regulatory relief and shelter from the amended FPA while an improved environment and other benefits are denied the public.

The Commission must redefine the "No Action" alternative when relicensing is involved. An analysis similar to that used for new projects must apply for applications for a new license for an existing project; <u>i.e.</u>, the requested action, licensing, does not occur. The "No-Action" alternative means denying the license.

In the case of relicensing, the Commission could exercise its authority to decommission the project through the "No-Action" alternative. All hydrotelectific operations would terminate. The Commission would remain involved to overset site restoration until decommissioning is accomplished at the licensee's expense.

Accordingly, we recommend that the Commission modify its description of the "No Action" alternative in the DEIS as discussed above. Issuance of annual licenses (over the 30- to 50- year license term) is not a reasonable action for the Commission to consider under NEPA. Annual licenses may only be issued under Section 15 of the FPA on an interim basis, until the Commission takes action on the application, either through issuance or denial of a new license, or the Federal government takes over the project.

The DEIS also discusses (but does not seriously consider in terms of meeting basic project purposes) additional structural and non-structural alternatives to the applicant's proposal, including conservation and windpower. Use of other technology to satisfy future demand for energy, as considered in this part of the DEIS, could result in a variety of significant "actions" for the existing dams and project works, including decommissioning and removal.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-13 analyses in Sections 2.4 and 5.1 of the FEIS have been redone using our revised methodology. The staff's position is that our revised economic methodology is consistent with the FPA.
- DOI-14 FERC has not changed its definition of baseline and noaction. Baseline conditions continue to be existing conditions, not pre-project conditions. No-action refers to continued project operation under the existing license.

DOI-14

Cont

DOI-15
REF: FERC # 2572, 2458

3. Dismissal of "ponexclusionary" alternatives, including conservation. The DEIS eliminates from consideration a variety of alternatives, including conservation and improvements at other generating facilities, because such measures do not involve the burning of fossil fuels. (Sec. 2.4.2.1) We find this reasoning to be somewhat arbitrary, given that all of the other alternatives in the DEIS involve measures that are "nonexclusionary" (i.e., do not burn fossil fuels) and are used in addition to the applicant's other hydropower facilities in the Penobscot River Basin. We believe that the major criterion that should be used in determining whether to include any alternative in the EIS is whether it addresses the basic project purpose (which is presumably to maintain the applicant's competitiveness in the pulp and paper industry and protect local employment).

6

The alternatives eliminated by the Commission because they are "nonexclusionary" are reasonable and should have been included in the DEIS because they could be used to produce, or reduce the demand for, electricity. The failure to give serious consideration to conservation and other such measures in this DEIS is contrary to the criteria set forth in Sec. 10(a) of the FPA for deciding whether a project is best adapted to a comprehensive plan for the waterway.

4. Failure to consider denial of license applications. The DEIS states that no one has recommended license denial, decommissioning of facilities and removal of the dams at the Ripogenus and Penobscot Mills projects, and therefore these were not considered as reasonable alternatives in the DEIS.

In the review of a license or permit application, the possibility for denial should always be considered. Relicensing is not to be taken as a given under the FPA; hydropower licenses are a privilege "justified only on the theory of resulting benefit to the public," and are thus revocable and not vested rights. Niazara Mohawk Power Corporation v. Federal Power Commission, 379 F.2d 153, 155 (D.C. Cir. 1967). Otherwise the objectivity of the application process becomes questionable, as the Commission strives to determine under what terms a new license should be given, rather than first addressing the fundamental issue of whether it is environmentally acceptable to do so. The Commission, on December 14, 1994, issued a Policy Statement in which it concluded that it has the "legal authority to deny a new license at the time of relicensing" if no license can be fashioned that will comport with the statutory standards, particularly that of Section 10(a) of the FPA. (See, Policy Statement - Project Decommissioning at Relicensing, 69 FERC (61,336). While the Commission may not choose to implement a denial option, consideration of this option would allow for a cumulative impact review. In this way, a complete review of dam impacts on the Penobscot River, its associated wetlands, and plant and wildlife resources could be done. Accordingly, we recommend that license denial for all pending applications be a required alternative in the EIS.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-15 The staff identified five alternative power sources in the DEIS: improvements in the efficiency of existing hydropower generation, wind generation, reduction of conservation voltage, refuse-fueled generation, and fuel cells. In evaluating these alternatives, the staff considered the first three to be non-exclusionary because these sources have low marginal costs and, as such, would displace higher cost fossil fuel generation. As non-exclusionary resources, they are not considered to be reasonable alternatives to hydropower and therefore we eliminated them from the analysis. Refuse-fueled generation and fuel cells were rejected for both operational and environmental reasons (see section 2.0). The staff determined the only other power source currently available to GNP to be Bangor Hydro-Electric; GNP does not have access to wholesale power in NEPOOL.

> The staff does not have evidence of large amounts of potential energy savings that could be obtained at GNP's mills. To the contrary, the applicant has submitted documentation describing the conservation measures they have implemented. As noted above, there are strong economic incentives for GNP to economize on energy use rather than waste it, particularly when energy is a major cost input. The staff finds that more conservation projects are likely to be found cost-effective if their savings are valued at the cost obtaining replacement power from BHE, than if they are valued at the much lower cost of hydropower. We agree that conservation under some circumstances may be the least-cost source of power other than hydropower. See response for DOI-3.

E-66

DOI-16

DOI-15

Cont

.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-16 We disagree. No party recommended license denial and we have no basis for considering it as a project alternative. See Section 2.3.4 of the FEIS.

7

REF: FERC # 2572, 2458

5. <u>Consideration of an alternative which describes conditions without the existing projects</u>. In order for cumulative impacts to be considered, an alternative which describes conditions without the dams must be considered. Such a comparison would allow for the most accurate accounting of unavoidable impacts to wetlands, plants, fish, wildlife, and their habitats which would continue or occur as a result of relicensing.

On page 2-23, it is stated that the "No-action" alternative was used to "establish baseline environmental conditions to compare with other alternatives." The Department advocates that the Commission adopt the without-or no-project condition as the baseline paradigm most in keeping with customary impact assessment, the need to consider license denial, and its authority to decommission projects at relicensing.

The PIN has used the Penobscot River, including the West Branch, since time immemorial and bases the condition of the resource as it existed without the dams. The existing dams were contributory to the decline of the runs of Atlantic salmon and other anadromous fish in the West Branch, and therefore the only appropriate baseline for impact assessment and comparison of alternatives is consideration of the river without these obstructions. The Department therefore requests that the Commission redo its analysis of benefit:costs and resource tradeoffs between the different alternatives such that they are based in the final EIS on the without-project condition.

AFFECTED ENVIRONMENT

Geology and Soils

The DEIS identifies the geologic significance of the portion of the West Branch of the Penobscot River immediately below Ripogenus Dam, known as the Upper Gorge, noting the steep bedrock outcrops and surface boulders. The document should also mention the similar geologic features in the river segment located downstream below Stone Dam (Penobscot Mills Project), particularly in the vicinity of Grand Falls.

Streamflow

The DEIS describes the streamflow regulation that occurs in the Penobscot River Basin due largely to the applicant's hydropower operations at Penobscot Mills and Ripogenus. In order to accurately describe cumulative impacts to streamflow in the basin, the staff should identify and discuss the magnitude and timing of a natural discharge regime in the basin. We note, however, that this has been done for the so-called 7Q10 flow (lowest flow that would be expected to occur over a 10-year period for a duration of 7 days). The DEIS states that a natural (unregulated) 7Q10 flow would be 81 cfs at Ripogenus and 126 cfs at Dolby (Penobscot Mills project) (p. 3-6). Although a discussion of natural 7Q10 may seem academic in light of the long history of waterflow regulation in the basin, in which normal low flows have been greatly augmented using reservoir storage, the Department emphasizes that would have occurred during what many consider to be drought conditions (i.e., 7Q10).

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-17 See our response to DOI-14. Although we have not considered pre-project conditions as a project alternative, we have considered impacts to specific resources in relation to flows or other conditions which may have existed in the absence of the projects (e.g., minimum flows or stabilized lake levels).

- DOI-18 We concluded that dam removal is not a reasonable alternative for the Penobscot Mills and Ripogenus projects. We determined that no highly significant environmental value or public use is rendered inviable by the existing developments in this project. The power lost as a result of removal of the project dams would place a significant economic impact on the licensee and the local communities. In addition, existing valuable fisheries for landlocked salmon and important recreational use of both the project impoundments and the riverine segments between impoundments would be eliminated by removing the dams. The staff presented this decision in Scoping Document 2 (November 1993). See also response DOI-2.
- DOI-19 We have revised the EIS to remove the description of the Upper Gorge. Specific information regarding various areas of both projects (including Upper Gorge and the Back Channel) are contained in GNP's application and need not be repeated in the FEIS.
- DOI-20 Comment noted. As you stated in your comment, the DEIS already states the unregulated 7Q10 flow for the project area.

DOI-18

m

ဗ္ဗ

DOI-17

DOI-19

8

REF: FERC # 2572, 2458

Water Ouslity

DOI-21

The DEIS states that riverine segments of the Penobscot Mills Project meet or exceed State of Maine water quality standards (p. 3-9). This finding is presumably based on the MDEP's water quality certification for the project, issued in 1993. However, the DEIS fails to acknowledge that the MDEP waived certification authority for the segment of the West Branch below Stone Dam (Back Channel). In addition, according to the MDEP's 1994 Water Quality Assessment for Maine, the four-mile section of the West Branch below Stone Data does not attain the aquatic life standard for its classification (Class C) because of dewatering due to hydroelectric power generation.

The state's 1994 Water Quality Assessment also found that two other portions of the Penobacot Mills Project area are in son-attainment of water quality standards. These include a three-mile segment of Millinochet Stream (untrested residential wastewater causing violation of bacterial standard) and a 0.5-mile river reach within Dolby Pond (dissolved oxygen below 5.0 ppm due to industrial discharges and the presence of an impoundment used for hydroelectric power generation).

We recommend that the final EIS correctly report the current water quality conditions in the West Branch of the Penobscot River as described in the state's 1994 Water Quality Assessment. This report was issued after the applicant submitted their application and subsequent to the water quality certification process.

Toxics

The DEIS focuses on mercury contamination in the project area and its implication for certain fish and wildlife resources (e.g., hald engles). While the occurrence of mercury and its relation to project operation is an important issue in these relicensing proceedings, the DEIS should also mention other contaminant investigations that have been conducted in the project area and list other potential point and non-point contaminant sources. This is particularly important in the context of discussing cumulative impacts to water quality due to contamination.

One such study that deserves mention is the State's dioxin monitoring program. In the 1988-1990 review of the program (Mower 1990), small amounts of dioxin (0.94 pg/g) were reported in sludge from the wastewater treatment plant in Millinocket. Mower's report also summarized a fish study in which furans were detected in smallmouth bass and white suckers taken in the West Branch below East Millinocket. While we are not suggesting that dioxins and furans are major contaminants of concern in the project area, these omissions in the DEIS suggest that not all of the potential contaminants or contaminant sources in the project area were identified.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-21 We have revised section 2.2.1.2 of the EIS to include MDEP's waiver of water quality certification for the Back Channel. We have revised section 3.4 to include information in MDEP's 1994 report that became available after the DEIS was published.

DOI-22 We have revised the EIS to include information from Mower (1990) as you suggest.

REF: FERC # 2572, 2458

Fishery Resources

Anadromous Fish

The DEIS identifies the fact that anadromous fish, including sea run Atlantic salmon and clupeids (American shad and/or river herring) were present in the West Branch of the Penobscot River (p. 3-13). The Commission notes that historic runs of these species were substantially reduced as a result of timber-related industries, dam building and pollution.

The DEIS fails, however, to discuss any plans for restoring anadromous fish runs to portions of their historic habitat in the West Branch. The Maine Atlantic Sea Run Salmon Commission's (ASRSC) Strategic Plan for the Management of Atlantic Salmon in the State of Maine (Strategic Plan) issued in 1984 identifies as its goal the restoration of selfsustaining salmon runs in historical habitat. Although current restoration and management efforts do not include the West Branch of the Penobscot, it should be expected that the ASRSC will at some time consider restoring salmon in autable habitat to this part of the drainage (recognizing that any potential conflicts with resident fisheries management would have to be addressed).

The restoration of anadromous fish runs to the West Branch of the Penobscot River is a critical issue to the PIN, which historically depended upon such fish for sustenance. As requested in the BIA's scoping comments, the EIS must examine the impact of continued operation of the projects on the probability of restoring Atlantic salmon, American shad, and river herring populations, as well as effects on resident fish populations. The barrier to passage for anadromous species posed by the West Branch projects must be resolved.

We also note that anadromous fish populations that existed in the West Branch were subjected to the same pressures as those fish in the lower Penobscot River basin, and were similarly extirpated as a result of human activities. This establishes a clear Hakage With the Commission's ongoing DEIS for the Lower Penobscot River Basin, in contrast to the position taken in Sec. 1.3.3 of the Penobscot River Basin DEIS.

Resident Species

DOI-24

The DEIS focuses on several resident fish species, including landlocked Atlantic salmon, brook trout, rainbow smelt, lake trout, and smallmouth bass. However, it should be noted that both landlocked salmon and smallmouth bass are not native to the West Branch. While we recognize that landlocked salmon are well-established and support a substantial fishery in the West Branch, we believe that the Commission should give adequate consideration to the needs of native fish species, both game and non-game, particularly in reference to cumulative impacts and biodiversity (see below).

The DEIS notes that the fishery for native brook trout declined in the West Branch in the 1950's, indicating that this was likely a result of the increasingly popular fishery for landlocked salmon. We suspect that an equally causative factor was the change in flow regulation that occurred in the West Branch after the McKay powerhouse was completed (1953). The higher than natural flows below McKay are more suitable for landlocked salmon than brook trout, as has been demonstrated elsewhere using instream flow studies.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-23 See also response DOI-2 and DOI-9.

9

DOI-24 Comment noted. Existing conditions form the baseline against which impacts are assessed. Furthermore, DIFW has emphasized management of landlocked salmon and lake trout within the project area. Therefore, we will not expand the FEIS to include a cumulative impact analysis for other species.

10

REF: FERC # 2572, 2458

(See for example, license application for the Mooschead Lake Project, FERC No. 2671, located nearby in the Kennebec River Basin.)

We note that the DEIS acknowledges a relatively depauperate fish community in portions of the Penobscot Mills project area, compared to upstream portions of the river. We believe that this lack of species richness is due in part to hydropower operations, including water diversions, and diminished water quality, as discussed above. These cumulative impacts to the natural agustic community should be further evaluated in the EIS.

Wetlands

The DEIS describes the types and amount of wetlands occurring in the area occupied by the Ripogenus and Penobscot Mills projects. Insofar as some of these wetlands are proposed to be altered in the recommended Alternative 2, it would be useful to have more information on specific functions and values of the affected areas. This information will also be needed by the Corps in its review of any subsequent applications for permits needed before the applicant initiates construction. In order to expedite the permitting process, and to avoid subsequent delays, we recommend that the final EIS include all necessary information that may be required by the Corps regarding modification of wetlands in the ornject area.

As part of its investigation of project-related impacts on wetlands, the applicant evaluated the quantity and composition of riparian areas bordering the West Branch below McKay Station. As reported in the DEIS, the applicant concluded that wetlands along the West Branch closely resembled those in unregulated lakes in remote parts of Maine. The final EIS abould specifically identify the location of those unregulated lakes that were assessed by the applicant, as this was an issue that was not fully resolved during the consultation process. (See for example the Department's comments to the Commission, dated May 24, 1993, regarding the Rigogenus relicensing application.)

River and Land Management Plans

The DEIS describes several comprehensive plans that have been filed by State of Maine agencies, and that are to be considered in the licensing process, pursuant to Sec. 10(a)(2) of the FPA (pp. 3-30 - 3-31). Given the historical occurrence of Atlantic salmon in the West Branch, and the long-term restoration goal discussed above, we recommend that the final EIS include the following additional comprehensive plans:

Strategic Plan for the Management of Atlantic Salmon in the State of Maine. Maine Atlantic Sea Run Salmon Commission, 1984.

Atlantic Salmon Restoration in New England. Final Environmental Impact Statement 1989-2021. U.S. Fish and Wildlife Service. 1989

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-25 Section 404 Clean Water Act wetlands permit review is performed by the Corps, not the Commission. Hence, it would not be appropriate for us to detail what information the Corps may require for its review (if any) of the wetlands permitting.
 - DOI-26 This particular description of these wetlands in the DEIS is subjective and has no direct bearing on the licensing process of the project.
 - DOI-27 Restoration of anadromous species to the West Branch of the Penobscot River is not currently planned by either state or federal agencies. The West Branch is not included in either of the plans you cite. Thus, our recommended alternative does not consider those plans.

DOI-24

DOI-25

DOI-26

DOI-27

Cont

REF: FERC # 2572, 2458

Threatened and Endangered Species

Bald cagle

As reported in the DEIS, the endangered bald engle (<u>Haliacetus leucocephalus</u>) occupies the West Branch of the Penobacot River Basin throughout the year (p. 3-29). There are seven pairs of engles (one more than reported in the DEIS) that are known to nest within the project area. The West Branch also provides some wintering habitat, although such use is limited by availability of open water. (The DEIS incorrectly states that the area is significant for wintering birds.)

Candidate Species

The DEIS also mentions the presence of several species in the project areas that are considered to be candidates for listing under the Endangered Species Act (long-tailed shrew, North American lynx, and Orono sedge). Of these the long tailed shrew does not need to be considered because it has been reclassified as a category 3C species, indicating that it is more abundant than previously believed.

Four additional candidate species may potentially occur within the project area, and should be addressed in the DEIS, including the yellow lampmussel (<u>Lampsilis cariosa</u>), brook floater (<u>Alasmidonta varicosa</u>), Extra striped snaketail dragonfly (<u>Ophiogomphus anomalus</u>), and the Midget snaketail dragonfly (<u>Ophiogomphus howeii</u>). Information pertaining to the distribution of these species is severely limited, although both species of freshwater mussels and both dragonflies have been recorded in the main stem of the Penobscot River.

DOI-30 The two federal candidate species of freshwater mussel, and the Squafoot mussel (Strophitus undulatus) have been proposed as state endangered or threatened species. All three species are restricted to flowing waters of unpolluted rivers and streams. Exuvia (exoskeletons shed during molting) of <u>Q</u>. howei were observed in the both the East (1993) and West (1994) Branches of the Penobscot River. <u>Q</u>. anomalus and <u>Q</u>. howeii have been proposed as state listed threatened and endangered species, respectively.

Current and historic project operations may limit the distribution and abundance of these species within this watershed. Restoration of appropriate flow rates to the West Branch of the Penobscot River below Stone Dam may allow for these species, as well as numerous other aquatic resources, to repopulate this area. Although candidate species are not formally protected under the Endangered Species Act (ESA), the FWS encourages their consideration during project planning. Protecting species when they are listed as candidates may reduce the need for formal protection in the future.

The DEIS also mentions the presence of common loons in both project areas. Although this is a bird species of concern in Maine, it is not on the Federal or State endangered species list.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-28 We have added a sentence to the FEIS stating that, according to FWS, there were seven pairs of bald eagles known to nest within the project area in 1995. We have also changed the FEIS to indicate that although the West Branch is one of the most important nesting areas for bald eagles, a relatively small proportion of bald eagles winter in Maine's interior.
- DOI-29 We have added a sentence to the FEIS stating that we did not consider it an important species in terms of our review.
- DOI-30 In the FEIS we indicated that these species could potentially occur in the project area.
- DOI-31 Opinion noted.

11

DOI-29

DOI-28

12

REF: FERC # 2572, 2458

Recreation

Although the channel downstream of Stone Dam does not at present support any significant fishery due to lack of habitat, the "potential habitat is extensive," (DEIS at page 3-20). The West Branch supports a significant amount of recreational fishing and is "recognized as one of the world's premier landlocked salsmon fisheries...." (DEIS at page 3-38). The applicant claims that it cannot afford to put sufficient flows into this 4.5 mile segment of the original river to redevelop a high quality fishery and provide additional opportunity for Class III-IV boating. However, data supporting this claim are not provided; the applicant has claimed that such information on its corporate consult at its proprietary. Such critical assertions must be fully supported by data if they are to be considered valid. The Commission can not make an informed decision without this supporting data.

Although commercial rafting use apparently peaked in 1985, this is likely due to the combination of increased user fees, the State's cap on commercial passengers per day and the increasing popularity of the Kennebec and Dead Rivers. However, the West Branch contains other recreational opportunities which are currently underutilized due to lack of water, such as the channel below Stone Dam and Millinocket Stream. Again, the applicant's assertions of financial limitations on providing flows to enhance recreational opportunities must be justified by data in order to be considered valid. Even if found to be valid, those claims must be balanced against the value of providing additional recreational opportunities to this nationally significant resource area.

DOI-34 The applicant often releases enough water into the Upper Gorge to support white water boating, but due to the lack of notification or predictable scheduling, these releases are not available to the public. A formal notification system, with a dedicated phone line should be implemented, and releases associated with acheduled maintenance should also be incorporated into a published achedule.

- DOI-35 General recreational use in the project area peaked in 1986, and has remained relatively constant. This is not consistent with recreational use nation-wide, but can be attributed to the applicant's institution of access fees at its gates in 1987.
- DOI-36 GNP has reached an agreement with the Fin and Feather Club to allow Maine residents free access. However, such an agreement should extend to all users regardless of domicile.

Land Use

DOI-37 While the applicant owns most of the land surrounding the project area, it asserts that there is a cost associated with protecting a 200 to 500 foot buffer around the impoundments and river sections. The Commission should state how this cost is determined; whether it is considered to be an opportunity cost associated with protection rather than sale of lands or its caused by purchase of additional lands or interests therein. The Commission estimates that the 15 percent of the impoundment shorelands currently unprotected that would be part of the proposed protection plans in Alternatives 1 and 2 and not owned by the applicant represent approximately 2,000 and 9,500 acres respectively. The cost figures associated with this land acquisition appear to be highly inflated at \$1,000 per acre for conservation

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-32 Increased minimum flows in the Back Channel would primarily create nursery areas for fry and juvenile salmon. Little adult habitat would be created. Young salmon reared in the Back Channel might contribute to adult populations downstream, but spawning habitat is limited downstream, and the shallow, warm water of the downstream impoundments represent marginal salmon habitat. We have estimated that the Back Channel might produce no more than several hundred adult salmon under optimal conditions. We do not support provision of increased minimum flows into the Back Channel because the benefits to recreational fishery resources are minimal and not commensurate with the costs. See Section 4.8.1.3.
 - DOI-33 See Section 4.8.1.1. Both the Back Channel and Millinocket Stream would require substantial flows to be navigable, and such flows would jeopardize other environmental enhancements. The Back Channel would only provide an approximately 4.5-mile-long boating trip in a less aesthetically pleasing setting than along the West Branch. Millinocket Stream would provide a longer trip (approximately 7.8 miles), but with only Class I and II whitewater. Our analysis suggests neither of these stream segments would attract commercial boating or significant levels of private boating activity.
 - DOI-34 Comment noted. Staff's recommended alternative includes provisions for whitewater boating releases in the Upper Gorge.
 - DOI-35 Opinion noted. This comment wil be addressed in any order issued for the project.
 - DOI-36 See Section 4.8.1.5.

REF: FERC # 2572, 2458

easements. Similar lands in like portions of northern New England are normally available for \$300 to \$400 per acre for fee acquisition. The Commission should retain an independent appraiser to determine the actual value of these lands and interests therein in order to allow staff to make an accurate determination of the financial impacts associated with their protection.

Given the presence of the conservation ensement referred to at page 3-47, it is unclear as to how either Alternative 1 or Alternative 2 would cause an adverse financial impact on the applicant in the area already covered by the easement.

Cultural Resources

The Commission in the DEIS disregards the interests of the PIN in the cultural resources of the West Branch. While the historic use of the project area by the PIN is acknowledged (p. 3-51), the Commission fails to go beyond that acknowledgment to pursue a clear understanding of the cultural significance of this area to the PIN. The PIN traditionally used the West Branch for travel and trade, in addition to depending upon its resources, including fish, wildlife, and plants, for sustemance. Tribal members travelled up the West Branch, including the now de-watered stretch below Stone Dam, by cance to reach Mt. Katahdin (a spiritually significant site) and the Debsconeag area, as well as to associate with other tribes in the interior. These uses and values are ignored in the DEIS. The Department requests that these uses and values be a major issue in the final EIS, particularly in regard to the effect that hydropower has on these values and uses.

ENVIRONMENTAL IMPACTS

The DEIS states that the Ripogenus and Penobscot Mills projects are so closely linked in terms of their operations, that the Commission has conducted some of their impact analyses on a cumulative basis (p. 4-1). While we agree that the projects have been integrated into a single system by the applicant, additional facilities on the West Branch (e.g., upstream storage projects) fall into the same category. The Commission should have incorporated the impacts of these facilities into the DEIS, as they are also inextricably linked to the projects covered in the DEIS.

The Commission's collective approach to impact assessment on the West Branch also raises a question regarding decision-making in connection with comprehensive development. In the past the Commission routinely judged whether a project was "best adapted" (i.e., the requirement in Sec. 10(a) of the FPA), based in part on the applicant's proposed mitigation and enhancement measures. Each project typically stood on its own merits, and did not depend upon environmental protection measures being taken elsewhere in the drainage.

For Ripogenus and Penobscot Mills, the applicant has dealt with environmental needs in a collective manner for both projects using a streamflow allocation model (WUP). One result has been to concentrate mitigation and enhancement measures at Ripogenus while recommending fewer environmental measures at the Penobscot Mills Project. The Commission has appeared to accept this reasoning (particularly with regard to fishery resources and recreation), departing from its usual practice of requiring each project to

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-37 The staff revised potential cost estimates for the shoreline easements upon review of comments received during the DEIS comment period and updated land valuation information. Staff determined the potential costs, based on waterfront footage for the easements proposed in Alternative 1, approximately \$24.6 million, would be greater than previously determined in the DEIS. Our evaluation of benefits suggests that the additional protection of the 500-foot expansion does not merit the much higher cost of that alternative as compared to the recommended alternative (\$24.6 million versus no direct costs). See section 4.9 for further discussion.
- DOI-38 See response to DOI-2.

13

DOI-39 The upper storage projects are included in GNP's water use model and the alternatives evaluated with that model. including additional model runs requested of GNP by FERC staff. These projects are not up for relicensing now but these projects are included in the water use model as a combined input. This input was kept constant in developing the water use plan although the model could be used to vary the input. We saw no need to do this however, since no one has suggested alternative management schemes for the upstream projects that would provide additional downstream benefits. In addition, GNP has agreed to inclusion of a reopener to allow modification of the water use plan. should analysis of the upper projects during relicensing result in unanticipated findings that desired changes could be made. We recommend the orders for both the Ripogenus and Penobscot Mills Projects include an article containing a reopener clause for consideration of changes in water use in the West Branch when the upper storage projects are evaluated for relicensing.

DOI-38

DOI-39

DOI-37

Cont

REF: FERC # 2572, 2458

DOI-39 Cont

DOI-40

independently meet the FPA's "best adapted" standard. This approach is inappropriate when the Commission intends to license these projects separately. As separate and individual projects, each should stand alone in terms of the mitigation and enhancement measures required to proceed with licensing.

14

Streamflow

The Commission relied largely on the applicant's streamflow model (WUP) to evaluate alternative water regimes in the project area. However, a number of recommended alternatives were not addressed by the Commission, even though they were raised during acoping or in comments on the final license applications. These include minimum flows higher than 350 cfs below Stone Dam and diminished pumping from Millinocket Lake to North Twin impoundment. (See Department's letter to the Commission, dated May 24, 1993, and FWS scoping comments.)

Furthermore, it appears that the WUP was based on the applicant's full use of water for industrial purposes. As stated in the DEIS, the applicant manages its storage system on the West Branch to maximize sustained power generation for its mills in Millinocket and East Millinocket. The absence of explicit inputs for power generation in the WUP (see for example Tuble 4-1 in the DEIS) leads to the conclusion that continued full use of the water by the applicant for power generation is a given. While exclusive use of water resources for hydroelectric power generation may have been acceptable when these projects were unlicensed, or during their initial license terms, the FPA now contains an "equal consideration" standard to address other needs. The WUP as described in the DEIS should have more equitably displayed the competing demands for water in the West Branch, including the power needs of the applicant. This would have given a clearer indication of whether the WUP actually achieves balance among competing needs.

As indicated in the BIA's scoping comments, sufficient flows should be provided below Stone Dam to support viable fisheries, to meet existing water quality standards, and to permit the use of this river channel for transportation. PIN members historically travelled by cance up this river channel and through the West Branch to reach sites of cultural significance. These interests in flows are ignored by the Commission in the DEIS. Consideration and accommodation of these uses must be addressed in the EIS.

Water Quality

The DEIS should have more fully addressed the non-attainment of water quality standards in the Penobscot Mills project area, as discussed above. The Commission has overlooked the problems in Millinocket Stream, deferred consideration of the dissolved oxygen deficiency in Dolby Pond until after licensing, and has only dealt with the non-navigability of the West Branch below Stone Dam. As mentioned previously, the West Branch below Stone Dam does not meet the aquatic life standard, which reads:

"Discharges to Class C waters may cause some changes to aquatic life, provided that the receiving waters shall be of sufficient quality to support all species of fish indigenous to the receiving waters and maintain the structure and function of the resident biological community."

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-39 Enhancement measures appear to be concentrated at the Cont Ripogenus Project because the measures proposed there are less costly for the environmental benefit produced, as compared with some measures at the Penobscot Mills Project (e.g., additional flows to the Back Channel). We considered enhancement measures and made balancing decisions for each project separately.
- DOI-40 You appear to have confused the applicant's water use plan (WUP) with the model they developed to evaluate various flow alternatives. The model can be used to evaluate various flow alternatives, including the Back Channel flows much greater than 350 cfs; we decided that flows to the Back Channel greater than 350 cfs were too costly to consider as a reasonable alternative. However, the model can simulate any desired flow condition. It can also be used to simulate water elevations of the two largest reservoirs in the system, Ripogenus and North Twin lakes. Your request for including explicit inputs for power generation in the water use model is irrelevant and has no bearing on the applicability of the model for evaluating alternatives.

DOI-41 See response DOI-2.

D01-42

DOI-41

.75

15

REF: FERC # 2572, 2458

Although the MDEP waived certification authority for this segment of the West Branch, the Commission has an opportunity and an obligation to ensure attainment of water quality standards in the project area. As discussed in the state's 1994 Water Quality Assessment, and in the DEIS, this could be accomplished by eliminating the dewatered condition of this portion of the West Branch of the Penobscot River.

Toxics

The DEIS notes the elevated levels of mercury in fish and bald eagles sampled in the project area by the applicant and others, including the FWS. As mentioned in the FWS' scoping comments, the applicant's contaminants study did not follow agency-prescribed methods, including aampling within the Ripogenus project area. As discussed below, the applicant's findings do not entirely agree with other work that has been done in the area, nor were the statistical analyses replicated by independent reviewers. Accordingly, the Commission has been too hasty in accepting the applicant's conclusions about the relation of project operations to mercury levels in fish and wildlife populations.

Fish tissue data collected by the applicant for the mercury study do not compare well with what was collected for the Regional Environmental Monitoring and Assessment Program (REMAP) (Stafford 1994). For example, lake trout from Debsconeag Lake that were analyzed in the REMAP Program (0.47 ug/g) contained higher levels of mercury than what was reported by the applicant (0.27 ug/g). Similarly, smallmouth bass from Molunkus Lake were also higher in the REMAP Program (1.12 ug/g) than in the applicant's study (0.68 ug/g). These data discrepancies may be related to differences in sampling methodologies, but they may also suggest problems in analytical procedures and data quality. The applicant's mercury study report did not include standard quality assurance/quality control information (e.g., spike recoveries, duplicates, certified reference material). Therefore, it is difficult to judge the validity of the applicant's data without this quality assurance information.

DOI-45 The applicant's information on sediment, tissue residue, pH, and water color suggest that Dolby Pond may not be a representative "control" location. The mercury concentrations in sediment from Dolby Pond are nearly three times higher than the project area with the next highest concentration (Millinocket Lake) and five times higher than some of the other project areas sampled. The mercury levels in Dolby Pond may indicate different, sitespecific biological conditions than the project areas or the presence of a local source of contamination upstream from the pond. Unless no biological factors or local mercury sources can be identified that would explain the elevated concentrations, inclusion of Dolby Pond in the comparison to other project areas does not seem appropriate.

DOI-46 Detection limits used in the analysis of project water samples were higher than generally accepted limits for mercury investigations. To accurately measure the concentration of mercury in water, ultra-clean sampling and analytical procedures are required. The detection limits used in the project study (0.20 ug/l) would be adequate for identifying grossly contaminated waters, but low-level detection limits (0.05 ng/l) would be required to adequately measure mercury concentrations in project waters.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-42 We have not overlooked water quality problems in the project area that are related to project operations. We have no information that any problem exists in Millinocket Stream or Dolby Pond that are caused by the operations of GNP's hydroelectric facilities. We have nevertheless concurred with the 401 WQC requirement that GNP conduct studies to determine the cause and extent of any dissolved oxygen problem in Dolby Pond. The legal status of the Back Channel relative to Maine's 401 WQC will be addressed in the order for the project.
- DOI-43 FWS, MDEP, PIN, MDIFW reviewed the field and laboratory study methods and GNP revised its study plan accordingly (see Volume XIII, Application for New License for Penobscot Mills Project, Appendix A, Work Plan/Correspondence). We have reviewed the comment letters and other correspondence on the draft study plan and have determined that the comments of the reviewing agencies were adequately addressed in the revised plan.

Differences in mercury between GNP's study and REMAP could be due to different methods, analytical labs, collection years (REMAP was conducted in 1993 while GNP's study was conducted in 1992), different sizes of fish, and natural variation.

We carefully reviewed the existing information on mercury levels in fish and sediments, including other generic studies of mercury mobilization before concluding that elevated levels are not associated with project operations.

DOI-44 See response above. The QA/QC results for the tissue analysis were obtained from GNP and placed in the public record. The testing included analysis of duplicates, spiked samples, mercury reference standards, and blanks. The QA/QC results indicate that the data obtained from the mercury study were of good quality.

E-76

DOI-42

Cont

DOI-43

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-45 Dolby Pond was inadvertently labeled a control pond in the DEIS; this has been corrected in the FEIS. However, higher levels of mercury in Dolby pond sediments are probably the result of historic discharges from the mill operations (which discharges into the upper portion of the Dolby impoundment) and not a result of the fluctuating water levels in the reservoir. Control of mercury concentrations in the mill's effluent is established in the NPDES permitting system.
- DOI-46 To obtain detection limits of 0.05 ng/l (i.e., 0.05 parts per trillion) would require ultra-clean sampling and ultra-clean laboratory analyses, which is extremely expensive to implement and unobtainable by most standard methods. The detection limits for the study were developed in consultation with the commenting agencies and clearly stipulated in the revised work plan (see Volume XIII, Application for New License for Penobscot Mills Project, Appendix A, Work Plan/Correspondence).

REF: FERC # 2572, 2458

The applicant's conclusion (endorsed by the Commission in the DEIS) that periodic drawdowns do not promote mercury accumulation requires further investigation. Analysis of the applicant's data by Terry Haines (currently with the National Biological Service) showed that lake trout and rainbow smelt had higher levels of mercury in project waters with drawdowns when compared to non-fluctuated control lakes. An additional review of the data that was performed for the U.S. Environmental Protection Agency (currently available as a draft report by Metcalf and Eddy Associates) concluded that impoundment and/or project operations are at least partially responsible for the elevated concentrations of mercury in fish tissue sampled in Millinochet Lake, North Twin impoundment and Dolby Pond. Moreover, although the applicant did not include landlocked salmon in the study, we suspect that they would also exhibit elevated mercury levels, due to their relative long life and reliance on smelt as forage. If this is true, it carries serious implications for current and protocosed fishery management practices that favor salmon in the West Branch.

We recognize that the mechanism for increased mercury levels in some of the project waters and their associated fauna may not be entirely understood. Regarding the effects of drawdowns, we suspect that if the water were lowered during the latter part of the summer (especially when the impoundments do not entirely fill in the spring), the rate of methylation could increase, leading to heightened bioavailability of mercury in project reservoirs. Methylation of mercury generally increases in shallow waters with warmer temperatures (Bodaly et al 1993). If there is a dieback of exposed vegetation during drawdowns, this could also contribute to increased methylation rates and the bioavailability of mercury.

Because of the lack of agreement on the cause of elevated mercury levels in the project area, we recommend post-licensing studies be carried out by the applicant in consultation with State and Federal resource agencies and PIN. (We note that Terry Haines is currently conducting studies on the relation between drawdowns and mercury level in Maine, but we are uncertain whether any of the applicant's project areas are included.) As indicated in BIA's scoping comments, these studies should include the effect of storage impoundment fluctuations and current and past pulp and paper mill practices, and should investigate the past use of mercuric slimicides and production of chlorine gas or caustic soda via the continuous mercury cathode method. The Commission should require the applicant to investigate and examine past records of mill practices and operations at least as far back as the early 1940's. Any licenses issued by the Commission for these projects should contain reopener provisions which permit the modification of project operations should the findings indicate a causal relationship of such operations with mercury levels in fish and wildlife resources.

Similarly, the Department is concerned about the low Dissolved Oxygen concentrations in Dolby Pond, which the Commission indicates are "not explained" but "probably" natural phenomena (p. 4-12). The Department prefers that the studies of the relations between mill discharges and DO concentrations precede licensing; however, if the Commission continues to require post-licensing studies, the Department requests a reopener provision similar to that discussed above, which would permit the modification of project operations should the study findings indicate a causal relationship between mill discharges and DO concentrations.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-47 There is no evidence in the scientific literature that periodic draw-downs in established reservoirs increase the mobilization of mercury. In newly formed reservoirs, increases in mercury concentrations in fish have been observed, but within a few years levels, decline to pre-impoundment (Johnston et al. 1991¹).

The information above reviewed by Terry Haines and U.S. EPA (draft report by Metcalf and Eddy Associates) relied on data that was slightly in error. GNP reissued the data entitled "Addendum Report: Correction in Mercury Tissue Concentrations of Fish Taken From Selected Lakes in Northern Maine". The data error resulted in only a 5% difference in the actual mercury concentrations (sample bag weights were not subtracted from the tissue sample weight). However, this error did not change the overall results of the study which indicated that increased mercury concentrations were not strongly linked to water level changes.

Target species to include in the GNP study were developed in consultation with the resource agencies. Landlocked salmon was not included as a target species in the revised work plan.

- DOI-48 The link between draw-downs and increased mercury methylation in established reservoirs statement is speculation and is not supported by the results of the sitespecific study nor the scientific literature.
- DOI-49 Staff concluded that the relationship between draw-down and mercury contamination in sediment, mussels, and fish tissue has been adequately addressed in the Penobscot Mills Project studies conducted by the applicant. As part of the State of Maine's water quality certification for the project, a study on the effects of reservoir draw-down on mercury levels in fish at the Ripogenus project is being required, similar to the study conducted for Penobscot.

¹ Johnston, T.A., R.A. Bodaly, and J.A. Mathias. 1991. Predicting fish mercury levels from physical characteristics of boreal reservoirs. <u>Can. J. Fish. Aquat.</u> Sci., Vol. 48:1468-1475.

16

DO1-48 E-78

DOI-47

DOI-49

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-50 There is currently no evidence to show that project operations affect DO in Dolby Pond. DO problems resulting from BOD loads from the mill are not within FERC jurisdiction; this issue will be discussed further in the license order for the project.

REF: FERC # 2572, 2458

Fishery Resources

The DEIS focuses on a number of issues affecting fishery resources, including streamflow below dams and powerhouses, water levels in impoundments and fish passage. Although some reference is made to non-game species, the emphasis is clearly on species that are being managed by the Maine Department of Inland Fisheries and Wildlife (MDIFW). Although we defer to the MDIFW's management decisions and priorities on the West Branch, we have in some cases advocated increased mitigation and enhancement to provide for restoration of habitat and/or additional protection of fishery resources.

Recreation

DOI-52 Adequate notice should be provided for boatable flows occurring in the Upper Gorge as a result of spillage or maintenance events. The time required for notification and an appropriate methodology, such as a dedicated phone line, should be included in the license conditions.

The DEIS states at page 4-48 that neither commercial nor private boating groups expressed any interest in expanding whitewater boating opportunities in Millinocket Stream or the channel below Stone Dam. This is not the case. Both the American Whitewater Affiliation (AWA) and the Maine Professional River Outfitters (MEPRO) requested flows suitable for white water boating in these areas, and a white water suitability study was conducted during the summer of 1993, which found that high quality recreational boating could be accommodated in both reaches. However, the applicant has claimed that releasing water into these areas would cause a severe economic impact, although they have not provided the data (again based upon proprietary claims) to justify those assertions.

DOI-54 While an additional week of stable impoundment levels at North Twin is an improvement over current operations, a stable impoundment should be maintained until after Labor Day, to allow for the high level of use that typically occurs during this end of the summer holiday. Otherwise, users during this weekend would be subject to unpredictable water levels which can adversely affect recreational opportunities.

DOI-55 Eliminating access fees for in-state users will result in proportionally higher fees for out-ofstate users. The use of a federally licensed facility (which is also a public resource) should be provided free of charge to all users. If a fee is to be charged to any users it must be charged to all users.

DOI-56 The applicant's proposal and that preferred by staff implies that existing access and facilities are adequate to meet existing demand. However, the level of existing demand is artificially low due to the recent imposition of access fees. It is likely that there is a higher demand for use than presently occurs. The proposed reevaluation every ten years is not adequate as there will be a different operational regime in place upon the issuance of the license. This regime will provide for increased recreational opportunities, and therefore, a reevaluation should occur at a minimum of every five years to adequately keep pace with changing use and demand patterns.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-51 No response required.

17

- DOI-52 Comment noted. Staff's recommended alternative includes provisions for whitewater boating releases in the Upper Gorge.
- DOI-53 Staff determined that the Back Channel would only provide an approximately 4.5-mile-long boating trip in a less aesthetically pleasing setting than along the West Branch. We state in Section 4.8.1.1 that significant opportunities exist within the project area to meet recreational boating demand, and that the significant flows that would be required to enable recreational boating activities would jeopardize other environmental enhancements.
- DOI-54 See Section 4.8.3.2. Our recommendation meets the reservoir's maximum drawdown rate and satisfies the request of SOSLA, which represents many of the waterfront property owners.
- DOI-55 Comments noted. See License Order.
- DOI-56 We recommend relocation monitoring studies be conducted every 6 years in consultation with various resource agencies (see section 4.8.3.4).

DOI-51

REF: FERC # 2572, 2458

Land Use

DOI-57

The applicant has stated that they may change their policy of not leasing any more camps, therefore land uses in the area could change dramatically. There may also be significant shifts in land use patterns if recreational uses become more profitable than timber harvesting, and given the high debt load of the applicant's parent company, Bowater, the sale of timber lands for other uses is not unforeseeable.

At pages 4-57 and 4-58, the staff provides estimates of potential residential development. Staff acknowledges that these estimates "do not account for development limitations such as steep slopes, poor soils, wetlands, or access." These limitations often have a significant impact upon any development, and to include figures as to a theoretical buildout without evaluating development limitations serves no purpose. In order to properly evaluate the actual threat of development, and thus determine exactly how much land is actually unprotected, the Commission should do a thorough analysis of development limitations, allowing for an accurate estimate and valuation of lands in need of conservation easements. From this, the Commission could come up with valid figures as to costs to the applicant for providing the project boundary increases deemed necessary by staff, and those deemed necessary by intervenors. It appears as though cost is the primary factor in the Commission's conclusion to not support the land protection measures evaluated in Alternative 1. Yet without accurate valuation information and with important economic data withheld by the applicant, the Commission is not likely to have reached a valid conclusion as to which alternative best meets the comprehensive development needs of the West Branch.

The Commission's proposal to allow for the establishment of a Shoreline Management Plan (SMP) does not provide any details, nor does it include any specified implementation date or methodology. The Commission would presumably develop an SMP that would represent the lowest cost to them, without the proper balancing of resource interests. Outright protection of critical lands is essential. If such a SMP is mandated, the FERC should require that those entities that have intervened in these proceedings be allowed to review and provide meaningful input on such plans as a condition to implementation. In addition, such a plan should be implemented in addition to outright land protection specified in Alternative 1 or 2.

Aesthetic Resources

DOI-60

The potential for shorelands development under any SMP scenario is not evaluated in the DEIS, and therefore, represents a potentially significant impact. The outright protection of critical riparian lands is the preferred methodology for preventing the kind of aesthetic impacts and intrusions that will erode the visual quality of the project area. Any SMP should include specific prohibitions on clearcutting within winter sight lines of any project impoundment or river section. In addition, any conservation easements developed should also contain land management plans developed with the input of the conservation intervenors and appropriate resource agencies.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-57 Opinion noted.

18

The staff revised its land use assessment and DOI-58 recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff proposes, for the Ripogenus project area, two options: (1) accepting the conservation easement proposed by GNP and the State of Maine: or (2) a 200-foot boundary expansion on GNP-owned lands. For the Penobscot Mills project area, the staff recommends a 200 foot expansion of the project boundaries on GNP owned lands. The staff recommends that existing structures would be grandfathered under any of the project boundary expansion alternatives. See section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.

DOI-59 Under the preferred alternative, the staff recommends the development of a shoreline management plan for those lands to be incorporated into the shoreline boundary expansion in consultation with resource agencies and interested parties (see section 4.9.3).

DOI-60 The staff considered aesthetic resources within the project areas and these considerations were incorporated into land use recommendations (see section 4.9 and 4.10). In the preferred alternative, the proposed conservation easements or 200-foot boundary expansion on GNP-owned lands for the Ripogenus Project area and the Penobscot Mills Project area would provide protection of shoreline aesthetic resources. We also recommend development of an SMP for both projects in consultation with various resource agencies (see section 4.9.3). See also responses DOI-58 and DOI-59.

DOI-58

E-81

REF: FERC # 2572, 2458

DOI-60 While there is a need to improve or otherwise minimize aesthetic impacts on resources used by most people, there is a tangible value associated with preserving the wilderness quality Cont of those areas least impacted by human intrusion.

Ripogenus Project

DOI-61 We concur with the applicant's proposed discharge regime at Ripogenus Dam, consisting of seasonal releases during the fishing season, provided the off-site habitat improvement described in the application (Holbrook site) is carried out and monitored for its effectiveness. We recognize that additional flows may be needed outside of the fishing season to maintain the aquatic community below the dam. Such additional flows will not interfere with the fishery management and enhancement measures that have been agreed to by the resource agencies.

DOI-62 We also defer to the MDIFW regarding management of the Ripogenus impoundment for iandlocind Atlantic salmon, although a reduced drawdown may allow for a more diverse fish fauna (e.g., lake trout and lake whitefish). Given that the drawdowns also affect wetlands and associated wildlife, as discussed below, and could be a factor in augmenting mercury levels, including those in gamefish, a moderated drawdown should be seriously considered.

> We also generally concur with the proposed flow regime below McKay Station, as being adequate to meet the needs of landlocked Atlantic salmon. However, as we have stressed repeatedly in these relicensing proceedings, additional instream flow studies should have been done by the applicant to evaluate whether flows below McKay could be reduced without endangering landlocked salmon in order to moderate drawdowns in the impoundment. In contrast to the findings by the Commission staff in Appendix D of the DEES, incorporating the results of instream flow studies below Ripogenus into the WUP would have been a way to address habitat conditions under alternate flow regimes.

Penobscot Mills

We also defer to the MDIFW's management objectives for the Penobscot Mills project area, although note that during the initial phases of consultation, the agency was interested in restoring habitat in the West Branch below Stone Dam, and in Millinocket Stream for landlocked salmon. The record is unclear as to why the MDIFW abandoned its interest in these riverine segments, opting instead to focus on the North Twin and Millinocket Lake impoundments. However, it does not appear that their reasons are consistent with the conclusions provided in the DEIS, as discussed below.

The applicant's WUP calls for a revised drawdown regime in North Twin Lake to protect and enhance lake trout populations. We believe that it will be necessary to monitor the population after the new regime goes into effect. Because the <u>magnitude</u> of the drawdowns in North Twin would remain unchanged, lake trout populations (and other aquatic resources) may continue to suffer due to reduced productivity. If post-licensing studies do not show an improvement in the lake trout population, additional measures, including a reduced drawdown will be needed.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-61 No response required.

19

DOI-62 GNP, the state (during the 401 WQC proceedings), and FERC staff considered drawdown limits at the major impoundments. As we stated in Appendix D of the DEIS, limiting drawdowns of Ripogenus is not a reasonable alternative, for at least 2 reasons: 1) downstream flow needs could not be met; and 2) some flood control benefits would be lost, depending on how much of a limit was imposed.

DOI-63 MDIFW agreed with the applicant that an IFIM study for the West Branch below McKay Station was not necessary and we concurred, as there was sufficient site-specific information to indicate appropriate salmon spawning and incubation flows for this area. Approximately the minimum flows recommended by FWS will be provided, except during unplanned outages when 400 cfs would be provided for no more than 3 days. At other times, the minimum would be 1422 or 1000 cfs or inflow, at times when FWS recommends 1422 or 711 cfs or inflow. We discuss flows below McKay Station more fully in section 4.2.

> The water use model is not an appropriate tool to address habitat conditions, as we explained in Appendix D. It could of course be used to evaluate any desired flow regime on overall water use throughout the project area.

с. 8-1

DOI-63

.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-64 We recommend that GNP conduct post-licensing studies on lake trout in North Twin, relative to the revised drawdown regime. We also recommend GNP propose changes to this regime if study results indicate that the revised regime does not result in desired levels of enhancement.

COMMENTS FROM US DEPARTMENT OF THE INTERIOR

20

ON LIPPER PENOBSCOT RIVER BASIN DEIS

REF: FERC # 2572, 2458

With regard to the riverine segments below Stone Dam and Millinocket Lake Dam, we believe that additional streamflows are required to restore and maintain aquatic productivity. We also agree with the MDIFW's original position that both riverine segments could significantly enhance salmon production in the project area. Although neither segment contains significant amounts of adult salmon habitat, MDIFW's initial conclusion was that Doiby Pond would support larger fish. With adequate instream flows below Stone Dam and in Millinocket Stream, free access between Dolby and riverine spawning and rearing habitat would be assured. (Contrary to the statement in the DEIS, Grand Falls would not be a barrier to salmon returning from Dolby Pond.) The West Branch below Stone Dam in the vicinity of Grand Falls also contains adult holding areas that were not acknowledged by the Commission staff in the DEIS.

We also note that the staff's dismissal of the production of salmon below Stone Dam as inconsequential is inconsistent with what is expected to be produced by habitat improvement at the Holbrook site below Ripogenus. In both cases the potential exists to produce several hundred adult salmon according to the DEIS; however, the staff recommends enhancement only at the Holbrook site below Ripogenus, where a substantial population of salmon already exists. Contrary to the position taken by the staff in the DEIS, we believe that significant potential exists for salmonid angling below Stone Dam. That demand for angling in this reach does not now exist is not surprising, given the lack of permanent flows and restricted access by the applicant.

Because of the documented presence of brook trout and other native, indigenous species, including invertebrates, in Millinocket Stream and below Stone Dam, it is also clear that restored flow regimes would have far ranging benefits beyond landlocked salmon. This includes use by anadromous fish, should they be restored to this portion of the project area in the future.

Instream flow studies by the applicant only addressed the needs of landlocked Atlantic salmon below Stone Dam and in Millinocket Stream. These instream flow studies were used to formulate the recommendations for 500 cfs at Stone Dam and 60 cfs in Millinocket Stream that were contained in the Department's May 24, 1993, letter to the Commission. Because the MDIFW has abandoned its previous management objectives for landlocked salmon in these river segments, the results of the applicant's previous studies should not be used to determine instream flow needs.

As an alternative we recommend that the Commission use the FWS' established procedure for determining instream flow needs at hydroelectric projects in New England. Published as the Interim Regional Policy for New England Stream Flow Recommendations (Flow Policy) in 1981, this instream flow methodology relies on historical discharge date to determine an Aquatic Base Flow (ABF) and seasonal adjustments for fish spawning and egg incubation. (The need for spawning and incubation flows is determined by the FWS on a case by case basis.) The ABF is an estimate of the historic, unregulated median flow for the August, typically when base flows occur in New England watersheds. Actual stream gaging records (collected and published by the U.S. Geological Survey) (USGS) for the month of August and for the periods when spawning and incubation occur are used to determine instream flow needs at a project.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-65 The DEIS stated that there would be few fish produced in the Back Channel even with additional flows, relative to the costs due to lost power and within the context of fisheries within the region. We considered enhancements of existing resources within the context of balancing power and non-power uses of the project area.

> Also, because DIFW has placed management priority on salmon and lake trout population which already exist, we recommend enhancements to benefit those populations rather than unknown benefits to currently non-existent populations. See also our response to PIN-35.

Based on the IFIM study for Millinocket Stream, which included brook trout, a year-round flow of 60 to 80 cfs would provide optimal habitat for this species. Following a 10(j) meeting with Interior, we recommend adopting a yearround flow of 60 cfs or inflow, whichever is less, for Millinocket Stream.

.

E-84

REF: FERC # 2572, 2458

In the event that USGS gaging data are unavailable, or if their records indicate that the natural streamflows in the project area have been regulated as a result of upstream dam releases or water diversions, the FWS' Flow Policy calls for using regional estimates of unregulated ABF and spawning/incubation flow. According to the Flow Policy the estimated unregulated ABF in New England is equivalent to 0.5 cubic feet per second per square mile of drainage area (cfsm). Base spawning/incubation flows are estimated to be 1.0 cfsm for the fall/winter period used by salmonids, including handlocked salmon and brook trout. In all cases recommendations under the FWS' Flow Policy allow for a reduction below the ABF or spawning/incubation standards if inflow to the project area is less than the specified amount in order to prevent lowering of the water level in the impoundment.

Given the extent of flow regulation on the West Branch of the Penobscot River, the regional estimates of ABF and spawning/incubation flows apply. In accordance with the FWS Flow Policy, the AVG at Stane Dam is 945 cfs (0.5 cfsm) or inflow, whichever is less. At Millinocket Lake Dam the ABF is 60 cfs (0.5 cfsm). A spawning/incubation flow of 120 cfs (1.0 cfsm) is needed in Millinocket Stream given the documented spawning activity that is described in the DEIS. This higher flow should apply for the period October 15 through June 7, as has been recommended by the applicant for the West Branch below McKay Station at the Ripogenus Project. Spawning in the West Branch below Stone Dam is limited due to the lack of suitable substrate, and likely takes place in tributary streams. Accordingly, the FWS does not find the need for spawning/incubation flows below Stone Dam.

Weilands

As discussed in the DEIS impoundment fluctuations have a continuing impact on wetlands, a phenomenon that is not restricted to the West Branch projects. As was reported by Central Maine Power Company in their report, Monitoring and Assessment of Lake Level Fluctuation Effects on Fish. Wildlife, and Wetland Remources at Brassua Reservoir, issued in 1994, drawdowns resulted in diminished wetland development, scouring, freezing and desiccation of substrates, stress or mortality of aessile or torpid organisms, and reductions in wildlife populations due to nest flooding, dewatering, and increased predation. Bald eagles nesting in the project area also exhibited elevated levels of mercury, based on FWS studies. The maximum drawdown in Brassua Reservoir is 31 feet, comparable to what can occur at Ripogenus and North Twin.

We believe that many of the wetland impacts described for the Brassua Project also occur at the applicant's impoundments on the West Branch that are used as storage facilities. Although the Commission staff is recommending wetland mitigation in the form of protected subimpoundments and loon nesting platforms, we believe that aerious consideration should be given to moderating water level fluctuations to improve the viability of wetlands and enhance their use by fish and wildlife. (The Corps of Engineers will undoubtedly be interested in the applicant's efforts to minimize and avoid impacts prior to issuing permits for the Commission staff's proposed wetland enhancement measures.) At a minimum the new licenses for the two projects should require long-term studies to evaluate how well wetland functions and values are restored and maintained at the projects.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

21

DOI-66 A total of 280 acres of affected wetlands would be improved by increasing water retention times. Such improvements would sufficiently ameliorate long-standing effects to project wetlands. As part of the Penobscot Mills and Ripogenus Licenses, we recommend that final design details, monitoring plans, contingency plans, and schedules be prepared after consultation with the Corps, FWS, and MDIFW. Final plans are to be approved by the Commission.

DOI-65

Cont

22

REF: FERC # 2572, 2458

DO1-66 With regard to the Commission staff's concern about impacts to riparian areas, including wetlands inhabited by beaver, along the West Branch below Stone Dam, as a result of minimum flow releases, this consequence is far outweighed by the impacts due to drawdowns in the impoundments. It is also not unreasonable to expect new wetlands to develop in this segment of the river as a result of permanent water flows, as is documented in the DEIS for the stretch of the West Branch below McKay Station.

Terrestrial Resources

As discussed in the DEIS the applicant proposes to manage a 2,300-acre parcel of land adjacent to the West Branch below Stone Dam for wildlife, including waterfowl. While we agree with this enhancement measure (and with the need for buffer strips and other land protection measures, as identified elsewhere in the DEIS), we recommend that detailed plans be developed by the applicant in consultation with the MDIFW and FWS. However, we do not consider the wildlife management plan as a trade-off for water flows below Stone Dam. Enhancement for terrestrial wildlife was never recommended by the agencies in contrast to the expressed need for restored streamflows below Stone Dam. If the wildlife management plan is required in the license for Penobacot Mills, the applicant should be required to monitor its use and assess its effectiveness in consultation with the resource agencies.

Threatened and Endangered Species

Baid cagic

During consultation with the applicant, the FWS reached the initial finding that continued operation of the project would not likely adversely affect bald eagles using the project area. (See FWS letter to the applicant, dated June 5, 1990.) Following subsequent studies by the FWS on contaminant levels in bald eagles in Maine, the FWS advised the Commission that its previous finding could be revised pending further analysis of the results. (See Department's letters to the Commission, dated May 24, 1993, regarding the applications for new licenses.) The FWS also requested additional investigations, particularly on contaminant levels in fish and wildlife resources, including bald eagles, and project operations. (See FWS scoping comments.)

As was found in the FWS' eagle investigations in 1991 and 1992, mercury levels in both blood and feathers collected from 6-8 week old eaglets nesting along Maine lakes were significantly higher (P < 0.001) than recorded for the other habitat types. In particular, mercury levels within the project areas in both feather and blood samples exceeded mean mercury levels reported for other lacustrine nesting eagles. The biological implications of these elevated mercury concentrations are not understood at this time. However, mercury is a neurotoxin and exposure to elevated concentrations may reduce the eaglets ability to fledge from the nest successfully and survive until adulthood. The concentrations observed in the eaglets from the project area exceed concentrations observed in adult eagles sampled from the Great Lakes.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-67 Monitoring plans will be required as part of the Penobscot Mills License.

DOI-68 No evidence exists in the record that would suggest DDE and PCBs are linked to operation of the projects. Data contained in Welch (1994) indicates that both DDE and PCBs in eaglet blood and feathers were found at levels just as high and sometimes higher at other lakes outside the project boundaries.

m

80

REF: FERC # 2572, 2458

Eaglets at Dolby Pond also had elevated levels of Polychlorinated biphenyls (PCBs) and DDE, a metabolite of DDT, as compared to other eaglets sampled from lacustrine nest sites. Absence of these compounds upstream of Dolby Pond suggests a possible point source discharge of these contaminants into the system.

The presence of extremely elevated levels of mercury in the eaglets indicates that mercury is bioaccumulating in the system. Much information exists correlating increased mercury concentrations in fish with recent impoundment of water. However, little if any research has been conducted on mercury concentrations in impoundments undergoing yearly fluctuations, such as those within the project area. Current operational procedures may be influencing the availability of mercury to organisms within the foodchain, including the bald eagle, as discussed above.

Because of these uncertainties regarding the effects of project operations on contaminant levels in fish and wildlife resources, the FWS is not yet prepared to concur with the Commission staff's assessment of no effect on bald eagles. The FWS also reserves the right to request additional consultation with the staff pursuant to Sec. 7 of the ESA, pending, the results of additional contaminant investigations in the project area, including monitoring studies conducted by the applicant and others.

Alternative 2 in the DEIS suggests that selective cuts in the vicinity of the bald eagle nest trees could provide potential foraging and perching areas for the eagles. The FWS recommends that no cutting occur in the immediate area of the nest trees. (The exact extent of the no-cut zone will need to be determined on a case-by-case basis in consultation with the FWS and MDIFW.) Selective cutting may result in increased exposure of the nest tree to high winds and other elements, resulting in damage to the nest. (The availability of snags is also not considered to be a limiting factor for eagles in this area.)

Biodiversity

The DEIS does not deal with cumulative impacts on biodiversity, even though the issue was raised during the scoping process. (See FWS scoping comments.) As is stated in the DEIS, a total of 137 dams have been constructed on the West Branch of the Penobscot alone, with more being built elsewhere in the basin. (See for example, the Commission staff's DEIS on the Lower Penobscot River.)

As was discussed in the FWS' scoping comments, the Commission staff should have identified the extent to which riverine habitats have been altered by dam construction and flow regulation. We agree with the conceptual approach in the Lower Penobscot River DEIS, where the staff recognized the importance of taking a conservative approach to biodiversity that includes protecting communities and ecosystems, promoting native species, protecting rare and ecologically important species, as well as unique or sensitive environments, and maintaining the natural processes and structural diversity of ecosystems. These issues are as germane on the West Branch as they are elsewhere in the basin and should have been included in the DEIS.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

23

- DOI-69 Comment noted. In the FEIS we have removed the recommendation relating to selective cutting for creating potential foraging and perching areas for eagles under Alternative 2.
- DOI-70 Comment noted. The DEIS for the lower Penobscot River considered biodiversity more fully because that DEIS included a new, unconstructed project. In the upper Penobscot River DEIS, the projects have existed for as long as 100 years and there are no unique or unusual habitats in the area that appear to require additional enhancements beyond those proposed by staff.

DOI-68

Cont

REF: FERC # 2572, 2458

Cultural Resources

DOI-71

The Commission's main consideration of PIN concerns is contained within Section 4.11. However, in this section, there is no provision for PIN's inclusion or participation in the development of the Programmatic Agreements or the Cultural Resource Management Plans which the applicant proposes to implement in order to mitigate impacts on the eligible archeological and historical sites with the project areas. Such provision must be made. Further, it fails to discuss whether the islands in the west branch could be of cultural significance pursuant to the National Historic Preservation Act (NHPA). BIA's scoping comments informed the Commission that cultural resources of significance to the PIN were present in the project areas, and that consideration should be given to the potential applicability of Section 106 of the NHPA. There is no indication in the DEIS that the Commission did so. To ensure that tribal resources and interests are considered, the Department again recommends that consultation with tribal leaders and other appropriate consulting parties be initiated immediately.

Unavoidable Adverse Impacts

As stated previously it may be possible to significantly reduce unavoidable adverse impacts to fish and wildlife resources by modifying the applicant's existing and/or proposed WUP for the projects covered in the DEIS. Pursuit of additional alternatives, such as retrofitting other dams owned by the applicant with generating facilities so as to accommodate modification of flow regimes and water levels at the Ripogenus and Penobscot Mills projects, could also reduce unavoidable adverse impacts. Additional mitigation consisting of dam removal in the West Branch drainage would greatly offset unavoidable adverse impacts. (See FWS scoping comments, calling for consideration of a "restoration alternative".)

Irreversible and Irretrievable Commitment of Resources

We concur with the conclusion in the DEIS that licensing of the projects would result in a commitment of the area to energy production. However, this single occupation of the river for waterpower development (particularly in the river segment below Stone Dam where total diversion of streamflow is proposed by the Commission staff) would appear to be contrary to the comprehensive planning mandate in Sec. 10 of the FPA, where other beneficial public uses, such as fish and wildlife, are to receive equal consideration.

Relationship between Short-term Uses and Long-term Productivity

NEPA requires that an EIS consider the "relationship between short-term uses of man's environment and the maintenance and enhancement of long-term productivity" (40 CFR 1502). However, the DEIS appears to consider production of 87.7 MW of energy over the next 30 to 50 years to be a form of "long-term productivity", rather than a "short-term use of man's environment". In addition the staff concludes that the proposed mitigation that would be implemented during the term of the licenses would result in significant, long-term enhancement of existing aquatic resources.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-71 See response to DOI-2.

24

- DOI-72 You suggest additional generating facilities be considered to meet GNP's need for power. No economically feasible facilities have been identified for the developments included in this EIS. GNP has agreed to inclusion of a reopener to allow modification of the water use plan, should analysis of the upper projects during relicensing result in unanticipated findings that desired changes could be made. We recommend the orders for both the Ripogenus and Penobscot Mills Projects include an article containing a reopener clause for consideration of the upper storage projects. We do not consider dam removal to be a reasonable alternative to provide additional enhancements for these projects.
- DOI-73 Opinion noted.
- DOI-74 Opinion noted.

DOI-73

DOI-72

DO1-74

REF: FERC # 2572, 2458

We view the 30- to 50-year license terms that would be in effect for the projects covered in the DEIS to be a relatively short-term phenomenon. The staff's proposed mitigation for losses in long-term productivity of aquatic life deals is not evenly balanced between the two projects, with resource needs at Penobscot Mills being enhanced less than those at Ripogenus. This does little to ameliorate impacts to, and maintain and enhance long-term productivity of, other fish and wildlife that have historically used all portions of the West Branch of the Penobscot River.

STAFF'S CONCLUSIONS

As discussed above, the staff has failed to give equal consideration to fish and wildlife resources in the DEIS, as is required by the FPA. The staff's proposed alternative will perpetuate the manipulation of the West Branch of the Penobscot River, for what has largely been a single purpose, the operation of its pulp and paper mills. Major impacts to aquatic ecosystems, resulting from large-scale water diversions and manipulations would continue virtually unabated under the proposed new licenses.

This lack of balance is reflected not only in the large number of dams that have been, and continue to be, used for hydropower development in the Penobacot River Basin, but on the meager reduction in the applicant's historic use of water flows, as proposed by Commission staff. As illustrated in Table 5-4 of the DEIS, restoration of 350 cfs (less than twice the historic 7Q10 drought flow) to the West Branch below Stone Dam (Back Channel) diminishes power production by only 3.35 percent, leaving the applicant with almost 97 percent of historic access to water for power production purposes. Contrast this with the fact that the staff's proposed alternative would result in the total elimination of instream uses in this natural river segment by aquatic life, including fishery resources. The staff's failure to recommend adequate instream flows at the Penobscot Mills Project, in light of the relatively insignificant impact in power production, shows a clear disregard for the equal consideration requirements in the FPA.

Important alternatives have been disariased by the staff, even though they could help achieve the applicant's basic project purpose of having dependable supplies of electricity for use in its mills, while at the same time minimizing or reversing environmental harm to local and regional natural resources. The applicant owns and operates numerous dams on the West Branch of the Penobscot River, and would have the opportunity to modify its project facilities and operations to more equitably accommodate fish and wildlife, tribal interests and concerns, and other environmental needs.

DOI-76 The Commission has failed to discharge its trust responsibility to protect the rights and Federal trust resources of the Penobscot Indian Nation. In its conclusions, as well as throughout the DEIS, the Commission has failed to consider PIN interests and disregards the concerns which were raised in BIA's scoping comments. Specifically, the Commission fails to require sufficient flows in the West Branch below Stone Dam to permit viable fisheries, to meet water quality standards, and to permit use of this channel for transportation. Tribal concerns regarding mercury contamination and dissolved oxygen concentrations are ignored. PIN's cultural beritage in the West Branch is disregarded, and there is no provision for PIN inclusion or participation in the development of the proposed Programmatic Agreements or Cultural Resource Management Plans.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-75 We have given equal consideration to fish and wildlife and human resources in the EIS. The DEIS stated that there would be few fish produced in the Back Channel even with additional flows, relative to the costs due to lost power and within the context of fisheries within the region. We considered enhancements of existing resources within the context of balancing power and non-power uses of the project area. See also response to comment DOI-72 above.

DOI-76 See response to DOI-2.

25

DOI-75

DOI-74

Cont

REF: FERC # 2572, 2458

Project Alternatives

DOI-77 As discussed previously the DEIS dismisses conservation and other alternatives that could belo meet the applicant's needs and satisfy the Commission's requirement for comprehensive river basin planning. The DEIS also fails to examine a number of design alternatives at the projects that could lessen impacts to aquatic habitat.

The continued operation of the Ripogenus and Penobscot Mills projects, without modifications, is not a "No Action" alternative. This scenario is also inappropriate as an environmental baseline, as it represents a highly imbalanced state of hydropower development on the West Branch of the Penobscot River. A more reasonable benchmark would be the "without-project" baseline.

Environmental Comparison of Alternatives

As mentioned above the staff has based much of its analysis of alternative operating regimes on the applicant's WUP. Because the applicant's power needs are not explicitly included in the WUP, it is unclear to what extent developmental and non-developmental (i.e., environmental) demands have been balanced. We suspect that the WUP was prepared to totally meet the applicant's energy and waste assimilation needs before addressing the environmental measures that were recommended by resource agencies and others (who were given no role in developing the WUP). A more equitable approach would have involved all parties from the outset.

It is also clear that the Commission staff's evaluation of alternatives was significantly tempered by the applicant's economic constraints (the extent of which are not revealed in the DEIS due to the proprietary nature of the information). For example, the Department's originally recommended discharge for the West Branch below Stone Dam (500 cfs) was not evaluated for its effectiveness in protecting and improving aquatic resources, but was dismissed at the outset because it was considered to be too costly for the applicant.

As discussed above we believe that the Commission staff is acting outside of the explicit licensing mandates in Sec. 10(a) of the FPA, when economic consequences for the applicant are allowed to sway decisions on whether to consider environmental protection measures. Instream flow recommendations by the Department for the Penobscot Mills Project are fully consistent with what the Commission has already required from the applicant at the downstream Mattaceunk Project, and are also less than what the applicant has proposed for the West Branch at the Ripogenus project.

We also suspect that the staff is inconsistently applying its "economic filter" to the licensing process. We are not aware of similar considerations when the staff addresses a safety issue at an operating project. For example, if during a routine inspection, the project dam is considered to be unsafe, thereby risking human safety downstream, we suspect that the licensee would be asked to comply with modifications prescribed by the Commission staff, with little regard to cost, or face a requirement for dam removal to alleviate the threat. In

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

- DOI-77 See response to DOI-3. The existing project is the appropriate No-Action alternative for this EIS. If none of the alternatives considered in the EIS are implemented, then the existing operations would continue. The current environmental conditions along the river, as they have been affected by the existing operation, would also continue.
- DOI-78 FERC has not changed its definition of baseline and noaction. Baseline conditions continue to be existing conditions, not pre-project conditions. No-action refers to continued project operation under the existing license.
- DOI-79 Opinion noted. We have responded above in DOI-40.
- DOI-80 The staff estimates that a flow of 500 cfs in the West Branch below Stone Dam would result in power replacement costs approximately three times larger than those for the 165 cfs flow. We also conclude that increasing the Back Channel flows would not produce significant environmental and recreational benefits.
- DOI-81 See response DOI-75.

26

E-90

DOI-78

DOI-79

DOI-81

REF: FERC # 2572, 2458

contrast when the safety and health of aquatic ecosystems are placed at risk, the staff does not heaitate to first consider the "cost effectiveness" of environmental protection. Just as it is totally inappropriate to weigh the cost of human lives when dealing with a dam safety problem, the expense of protecting the rest of the ecosystem should not enter into the discussion. Environmental protection must be a cost of doing business in order for the greater public interest to be served.

We also note inconsistencies in the staff's treatment of fishery resource needs in the DEIS. The improvements proposed below Ripogenus (Holbrook site) would result in an estimated 245 adult sulmon in this segment of the West Branch, where according to the DEIS there are already roughly 200 fish per mile and an excellent sport fishery. In contrast flow improvements below Stone Dam could result in "several hundred" adult salmon, in an river reach where there are very few fish of any species, due to the long-standing flow diversions by the applicant. Clearly, the addition of a new salmon stock (and numerous other species of fish and invertebrates) would be more noticeable below Stone Dam than at Ripogenus. Yet this finding is not made by the staff, based largely on the economic implications of returning only a portion (350 cfs) of normal flows to the natural river channel below Stone Dam.

The staff has concluded in the DEIS that high-quality recreational fishing waters are abundant in the project region, and therefore believes that additional mitigation, as has been proposed by the Department, is not required. Following this logic would eliminate virtually all of the proposed fishery enhancement measures in the DEIS, including those proposed by the applicant and recommended by the MDIFW to achieve their management goals for the West Branch. Although we believe that it is important to consider cumulative impacts within a multi-project context, it is inappropriate, and inconsistent with other relicensing actions by the Commission, to waive the need for mitigation and enhancement at one project, based on measures required at another. We believe that this would in fact occur if the Ripogenus and Penobacot Mills projects are relicensed as proposed in the DEIS.

White Water Recreation

Staff states at page 5-2 that "we do not view flows for recreation as a major issue." Such flows are clearly a major issue for the NPS, for white water intervenors and for the many tens of thousands of recreational users of one of the nation's most significant white water boating resources.

Page 5-2 further states that the due to agreements between the applicant and white water recreation intervenors, no flow enhancements beyond those proposed by GNP are necessary. This assumes that all white water recreation intervenors not only participated in those negotiations, but supported the result. In addition, Commission staff assumes that those agreements have been finalized and remain unchanged from their initial provisions. This is not the case. The Maine Professional River Outfitters (MEPRO) negotiated certain user fees for commercial boaters. The fees worked out did not apply to private boaters, and have been changed or withdrawn since that agreement reached in the summer of 1993.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-82 Opinion noted.

27

DOI-83 See response DOI-53.

DOI-81

Cont

REF: FERC # 2572, 2458

Land Use

The section at pages 5-10 to 5-11 relating to staff's conclusions on land use do not adequately explain the implications of a proposed building setback and vegetative buffer. Those elements

would not affect owners or lessees of existing properties, which would be grandfathered. Further, those provisions would be offset by the benefit to those visitors who value this area for its undeveloped character. Given current land use projections, much of the shoreline of the project areas is unprotected from development, and large numbers of dwelling units could conceivably be constructed. Considering the applicant's indication that it will remove its current moratorium on new leases, it is imperative that the Commission, as a condition of granting new licenses for these projects, take adequate steps to ensure the undeveloped character of this resource.

The Commission has recommended adopting a 200 foot expansion of project boundaries, and states that the environmental benefits between that degree of expansion and those afforded by the 500 foot expansion "cannot be quantified rigorously." It seems inconsistent that Commission staff can, however, attribute an exact financial cost to the different alternatives. Furthermore, the Commission's policy contained in Commission Order 31 (amending Section 4.41 of the Commission's regulations under the Federal Power Act) states that project boundaries are expected to be extended beyond 200 feet from reservoir shorelines. While staff has not accepted the conservation intervenors recommendations, they have not followed Commission policy, but have set an artificially low level of protection. The compromise should not have been between no additional protection and 500 feet, but between some amount beyond 200 feet and the 500 foot recommendation.

Socioeconomic

At page 5-15, staff acknowledge that GNP's claims regarding estimates of the economic effects of production cutbacks caused by environmental enhancement alternatives have not been verified because GNP's data about corporate economic status is proprietary." Commission staff state at page 5-11 that "the confidentiality of the applicant's financial status prevents an independent assessment." Such claims have no basis in this proceeding. The Commission is entitled to all relevant financial information that could reasonably be expected to assist staff in forming an adequate factual basis for a licensing decision.

A 350 cfs year-round flow in the channel below Stone Dam represents less than 2 percent of the energy used by GNP, but they claim that to lose this available energy would require layoffs of over 200 people, which represents 10-15 percent of their total work force. These claims are not only inconsistent, but can not be substantiated as they are derived from what the applicant claims is proprietary economic information.

A critical component of the Commission's functions is to balance economic and environmental factors when making a licensing decision. To fail to evaluate or even request an applicant's economic data cannot lead to a true balancing as required by the FPA.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

28

DOI-84 The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff's recommendations (see section 5.3.4) provide measures to protect shoreland resources within the project areas.

Within the Ripogenus Project area, the recommended alternative proposes two options: (1) adoption of the proposed 250-foot conservation easements for GNP owned land as defined by the MOU; or (2) a 200-foot boundary expansion on GNP-owned lands (see section 4.9.1). For the Penobscot Mills Project, the recommended alternative proposes expanding project boundaries to generally extend 200 feet from the high water mark of the impoundments within the project area, only on land currently owned by GNP.

Within the proposed boundary expansion areas, existing structures would be grandfathered and GNP would have the authority to review and approve proposed actions as established by the Commission under the Standard Land Use Article or SMP. See Section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.

REF: FERC # 2572, 2458

Statutory Requirements

Section 10(i) of the FPA

The FPA requires that licenses issued by the Commission contain conditions intended to adequately and equitably protect, mitigate damages to, and enhance, fish and wildlife (including related spawning grounds and habitat) affected by the development, operation and management of the project. Section 10(j) of the FPA further requires that these conditions be based on recommendations received pursuant to the Fish and Wildlife Coordination Act (FWCA), 16 U.S.C 661 et ang., from the National Marine Fisheries Service, the FWS, and State fish and wildlife agencies, unless the Commission determines that the agency recommendations are inconsistent with the purposes and requirements of Part I of the FPA or other applicable law.

The FWS has provided initial Sec. 10(j) recommendations in its comments on the license applications for the Ripogenus and Penobscot Mills projects. The Commission staff has rejected a number of these recommendations, finding that they are inconsistent with the purposes and requirements of the FPA. Specifically, the staff concluded that the FWS recommended provisions would be inconsistent with Secs. 4(e) and/or 10(a) of the FPA due to economic implications (i.e., the resource benefits are in the staff's estimation not worth the cost in lost generation) and therefore, not in the "public interest".

We believe that the staff may be reading more into Secs. 4(e) and 10(a) than actually exists. Both sections of the FPA do make explicit reference to preservation of environmental quality, including the protection, mitigation of damage to, and enhancement of fish and wildlife (including related spawning grounds and habitat). The Sec. 10(j) recommendations by the FWS for these two projects that were subsequently rejected by the Commission staff were entirely within the letter of the FPA.

The staff has also given undue deference to the Maine Department of Environmental Protection (MDEP) under Sec. 10(j) of the FPA. The MDEP is not a State fish and wildlife agency, nor do they have any legislative authority to act on behalf of Maine's three, independent fish and wildlife agencies in carrying out obligations under the FWCA. Although the Commission must adhere to the MDEP's (in the case of the Ripogenus Project, the Land Use Regulation Commission) conditions in the State water quality certification (under Sec. 401 of the CWA), the staff is incorrect to consider these requirements as fish and wildlife agency recommendations.

It should also be noted that the MDEP performs a balancing function in its certification process, similar to what is done by the Commission in the Federal licensing process. This is because hydropower is considered to be a "designated use" in Maine's water quality standards. The MDEP believes that to fully maintain the hydropower designated use at a project, it often must temper the recommendations from the State fish and wildlife agencies, resulting in less than full protection and enhancement for fish and wildlife and associated public uses. Accordingly, the recommendations from the agencies, formulated in accordance with the FWCA to "prevent loss of, or damage to" fish and wildlife resources,

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-85

20

The staff finds, based on our most recent economic analysis, that there is no conclusive evidence that either the Applicant's Proposal, or both versions of Alternative Two, would adversely affect the competitive position of the GNP's mills. This is because these alternatives would produce small percent increases in annual power costs, small percent declines in annual power output, and ultimately, that these small changes would not translate into a significant increase in the cost of production at the two mills. The staff also concludes that the negative annual net benefits under Alternative One would be large enough such that the competitiveness of the two mills is likely to be adversely affected.

DOI-86

E-93

REF: FERC # 2572, 2458

30

Cont are typically compromised by the MDEP before they are passed on to the Commission. While this did not specifically occur in the case of the Ripogenus and Penobscot Mills projects (although MDIFW objectives were changed for the latter project prior to the issuance of the Water Quality Certification), the Commission staff should not view the MDEP or LURC as fish and wildlife agencies for purposes of Sec. 10(j) of the FPA.

Consistency with Comprehensive and Other Resource Plans

As discussed previously anadromous fish, including sea run Atlantic salmon, may eventually be restored to historic habitat in the West Branch of the Penobscot River. Should that occur, it will be critical for all salmon habitat to be adequately maintained through flow releases. Because the staff's proposed discharges for Millinochet Stream and the West Branch below Stone Dam are not sufficient for salmon and other species, this licensing would not be consistent with the comprehensive salmon restoration plans that have been received and accepted by the Commission.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-86 We note that subsequent to the DEIS, we have determined Cont that the state of Maine has not submitted any 10(j) recommendations. The Commission's final determination with respect to Interior's 10(j) recommendations will be addressed in the orders for the projects.

DOI-87 We do not consider restoration of anadromous fish to the West Branch of the Penobscot River a reasonably foreseeable action that could be considered in this EIS. We note, however, that Interior has reserved the right to prescribe fishways under section 18 of the FPA and this reservation will be included in the orders for the projects.

ATTACHMENT B: REVISED SEC. 10(1) RECOMMENDATIONS AND SEC. 18 FISHWAY PRESCRIPTIONS

The Commission's regulations (18 CFR Subchapter B Part 4) allow for modification of recommendations and prescriptions previously provided by resource agencies pursuant to Sections 10(j) and 18 of the FPA, when the licensing proceeding involves preparation of a DEIS. Accordingly, we are providing the following modified recommendations and prescriptions for the projects covered in the Penobscot River Basin DEIS.

DOI-88 Ringermus Project

Section 10(i) Recommendations

The Section 10(j) meansmendations for the Ripogenus Project contained in the Department's May 24, 1993 letter are modified as follows:

- 1. The Licensee shall discharge at the Ripogenus Dam an instantaneous flow of 100 cfs during the period July 1 through September 30 each year.
- 2. The Licensee shall, within six (6) months of the date of issuance of this license, file a plan for making habitat improvements for landlocked Atlantic salmon at the Holbrook site. The plan shall be prepared following consultation with the Fish and Wildlife Service, Maine Department of Inland Fisheries and Wildlife, and Penobscot Indian Nation, and shall include a description of construction activities to be used to alter and maintain juvenile salmon habitat. The plan shall also describe how the use of the modified habitat will be monitored.

The Licensee shall arek comments from the fishery agencies and tribe on the plan prior to its being filed with the Commission, allowing a minimum of 30 days for responses.

 The Licensee shall provide a minimum flow at McKay Station of at least 400 cfs for short-term outages lasting three days or less. Beyond the "short-term outage" period, the following instream flow schedule shall apply:

October 15 - June 7: no less than 1,422 cfs or inflow, whichever is less.

June 8 - October 14: no less than 711 cfs or inflow, whichever is less.

4. The Licensee shall, within three (3) months after the date of issuance of the license, file with the Commission a plan for complying with all instream flow requirements at the Ripogenus Project. The plan shall include a description of the mechanisms and structures that will be used, the level of automatic or staffed facility operation, the methods to be used for recording data on minimum flows, and a plan for maintaining these data for inspection and filing with the Commission and resource agencies. The Licensee shall consult with the Fish and Wildlife Service, U.S Geological Survey, Maine Land Use Regulation Commission, the Maine Department of Inland Fisheries and Wildlife, and Penobscot Indian Nation in developing this

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-88 Section 10(j) recommendations are addressed in section 5.6 of the FEIS and in the orders for the projects.

2

REF: FERC # 2572, 2458

plan and shall respond to agency/tribal comments and include their correspondence in future filings with the Commission. The Licensee shall provide the resource agencies and tribe a minimum of 30 days to respond to a draft plan before it is filed for Commission approval.

5. The Licensee, after consultation with the Fish and Wildlife Service, National Park Service, Maine Department of Inland Fisheries and Wildlife, Maine Department of Conservation, and Penobscot Indian Nation shall monitor recreational use of the project area to determine whether existing access facilities are meeting demands for public use of fish and wildlife resources. Monitoring studies shall begin within 6 years of issuance date of this license, and shall include, at a minimum, the collection of annual recreation use data.

Every 6 years during the term of the license, the Licensee shall file a report with the Commission on the monitoring results. The report shall include: (1) annual recreation use figures; (2) a discussion of the adequacy of the Licensee's recreation facilities at the project site to meet recreation demand; (3) a description of the methodology used to collect all study data; (4) if there is a need for additional facilities, a recreation plan proposed by the Licensee to accommodate recreation needs in the project area; (5) documentation of agency/tribal consultation and agency/tribal comments on the report after it has been prepared and provided to the agencies; and (6) specific descriptions of how the agency/tribal comments are accommodated by the report.

The Licensee shall allow a minimum of 30 days for the agencies and tribe to comment and to make recommendations prior to filing the report with the Commission.

- 6. The Licensee shall within 90 days of issuance of this license file a plan for monitoring the level of contaminants in fish and wildlife resources at the Ripogenus Project. The plan shall be prepared in consultation with the Fish and Wildlife Service, Maine Department of Inland Fisheries and Wildlife, Maine Department of Environmental Protection, Environmental Protection Agency, and Penobscot Indian Nation, and shall provide for the regular and periodic sampling of fish and wildlife resources and habitats for the presence of contaminants in the project area. The Licensee shall also file annual reports on contaminant levels in the project area with the Commission, consulting first with the resource agencies and PIN, and allowing a minimum of 30 days for review and comment.
- 7. The Licensee shall within 90 days of issuance of this license file a plan for monitoring the effectiveness of all fish and wildlife enhancement measures at the Ripogenus Project, including instream flow releases, lake level manipulation, wetland and foresty management, loon nesting platforms, and shoreline protection/buffer zones. The plan shall be prepared in consultation with the Fish and Wildlife Service, Maine Department of Inland Fisheries and Wildlife, Maine Department of Environmental Protection, Environmental Protection Agency, and Penobscot Indian Nation, and shall provide for the regular and periodic sampling of fish and wildlife resources to determine the level of resource protection resulting

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-88 Cont

REF: FERC # 2572, 2458

3

from the enhancement measures. The Licensee shall also file reports on the fish and wildlife monitoring studies in the project area with the Commission, consulting first with the resource agencies and PIN, and allowing a minimum of 30 days for review and comment.

Section 18 Prescription

The information and analysis presented in the DEIS do not indicate a need to modify the Secretary of the Interior's prescription pursuant to Section 18 of the FPA, as contained in the Department's May 24, 1993 letter to the Commission.

Penobacot Mills Project

DOI-88 Cont

E-97

Section 10(i) Recommendations

The Section 10(j) recommendations for the Penobscot Mills Project contained in the Department's May 24, 1993 letter are modified as follows:

- The Licensee shall discharge at the Stone Dam into the natural channel of the West Branch of the Penobacot River an instantaneous flow of at least 945 cfs or inflow, whichever is less, throughout the year.
- 2. The Licensee shall discharge at the Millinocket Lake Dam into Millinocket Stream an instantaneous flow of at least 120 cfs or inflow whichever is less, between October 15 and June 7. Between June 8 and October 14 an instantaneous flow of at least 60 cfs or inflow, whichever is less shall be maintained.
- The Licensee shall operate the Millinocket, Dolby and East Millinocket Developments in a run-of-river mode, whereby outflows equal inflows on an instantaneous basis, and water level fluctuations are minimized.
- 4. The Licensee shall, within three (3) months after the date of issuance of the license, file with the Commission a plan for complying with all instream flow requirements at the Penobscot Mills Project. The plan shall include a description of the mechanisms and structures that will be used, the level of automatic or staffed facility operation, the methods to be used for recording data on run-of-river operation and minimum flows, and a plan for maintaining these data for inspection and filing with the Commission and resource agencies. The Licensee shall consult with the Fish and Wildlife Service, U.S Geological Survey, Maine Land Use Regulation Commission, the Maine Department of Inland Fisheries and Wildlife, and Penobscot Indian Nation in developing this plan and shall respond to agency/tribal comments and include their correspondence in future filings with the Commission. The Licensee shall provide the resource agencies and tribe a minimum of 30 days to respond to a draft plan before it is filed for Commission approval.

COMMENTS FROM US DEPARTMENT OF THE INTERIOR

ON UPPER PENOBSCOT RIVER RACHEDRIN

4

REF: FERC # 2572, 2458

5. The Licensee, after consultation with the Fish and Wildlife Service, National Park Service, Maine Department of Inland Fisheries and Wildlife, Maine Department of Conservation, and Penobscot Indian Nation shall monitor recreational use of the project area to determine whether existing access facilities are meeting demands for public use of fish and wildlife resources. Monitoring studies shall begin within 6 years of issuance date of this license, and shall include, at a minimum, the collection of annual recreation use data.

Every 6 years during the term of the license, the Licensee shall file a report with the Commission on the monitoring results. The report shall include: (1) annual recreation use figures; (2) a discussion of the adequacy of the Licensee's recreation facilities at the project site to meet recreation demand; (3) a description of the methodology used to collect all study data; (4) if there is a need for additional facilities, a recreation plan proposed by the Licensee to accommodate recreation needs in the project area; (5) documentation of agency/tribal consultation and agency/tribal comments on the report after it has been prepared and provided to the agencies; and (6) specific descriptions of how the agency/tribal comments are accommodated by the report.

The Licensee shall allow a minimum of 30 days for the agencies and tribe to comment and to make recommendations prior to filing the report with the Commission.

- 6. The Licensee shall within 90 days of issuance of this license file a plan for monitoring the level of contaminants in fish and wildlife resources at the Penobscot Mills Project. The plan shall be prepared in consultation with the Fish and Wildlife Service, Maine Department of Inland Fisheries and Wildlife, Maine Department of Environmental Protection, Environmental Protection Agency, and Penobscot Indian Nation, and shall provide for the regular and periodic sampling of fish and wildlife resources and habitats for the presence of contaminants in the project area. The Licensee shall also file annual reports on contaminant levels in the project area with the Commission, consulting first with the resource agencies and PIN, and allowing a minimum of 30 days for review and comment.
- 7. The Licensee shall within 90 days of issuance of this license file a plan for monitoring the effectiveness of all fish and wildlife enhancement measures at the Penobscot Mills Project, including instream flow releases, lake level manipulation, wetland and foresty management, loon nesting platforms, and shoreline protection/buffer zones. The plan shall be prepared in consultation with the Fish and Wildlife Service, Maine Department of Inland Fisheries and Wildlife, Maine Department of Environmental Protection, Environmental Protection Agency, and Penobscot Indian Nation, and shall provide for the regular and periodic sampling of fish and wildlife resources to determine the level of resource protection resulting from the enhancement measures. The Licensee shall also file reports on the fish and wildlife monitoring studies in the project area with the Commission, consulting first with the resource agencies and PIN, and allowing a minimum of 30 days for review and comment.

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

DOI-88

Cont

RESPONSES TO US DEPARTMENT OF THE INTERIOR ON UPPER PENOBSCOT RIVER BASIN DEIS

REF: FERC / 2572, 2458

ŝ

Section 18. Prescription

The information and analysis presented in the DEIS do not indicate a need to modify the Secretary of the interior's prescription pursuant to Section 18 of the FPA, as contained in the Department's May 24, 1993 letter to the Commission.

DOI-88 Cont

COMMENTS FROM ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

UNITED STATES ENVIRONMENTAL PROTECTION AGENCY REGION 1 J.F. KENNEDY PEDERAL BUILDING, BOSTON, MASSACHUSETTS 02203-2211

February 21, 1995

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, N.E. Washington, D.C. 20426

re: Draft Environmental Impact Stavement for the Penobscot Riv Basin, Maine; Ripogenus and Penobscot Mills Hydroelectr Projects (FERC Projects #2572 and #2458)

Dear Secretary Cashell:

The Environmental Protection Agency-New England, in accordance wi our responsibilities under the National Environmental Policy A (NEPA), Section 309 of the Clean Air Act, and Section 404 of t Clean Water Act has reviewed the draft Environmental Impa Statement (dEIS) prepared by the Federal Energy Regulatc Commission (FERC) for the above referenced hydroelectric projec on the West Branch of the Penobscot River in Maine.

GENERAL CONNENTS

EPA believes the dEIS does not adequately consider fish and wildlife resources affected by the proposed licenses as it simply endorses the applicant's proposal to continue current operations of the Ripogenus and Penobscot Mills facilities. EFA believes these projects, as proposed, are inconsistent with Maine's water quality standards, especially regarding the issue of flows in the Back Channel. Although the dEIS discusses facility operational changes that would result in substantial aquatic habitat improvements with relatively small reduction in power production, FERC does not recommend implementing these changes.

Section 4(e) of the Federal Power Act requires equal consideration of multiple potential uses of a public waterway. In the case of the Penobscot River dEIS, EPA believes that appropriate application of the Section 4(e) requirement to give equal consideration to fish, wildlife and recreation purposes would have resulted in recommendations that restore flows to the "Back Channel" (the 4.5 mile reach of the West Branch dewatered by the construction of the Penobscot Mills development). Additionally, EPA believes the equal consideration requirement would have resulted in a FERC endorsement of other recommendations of the resource agencies to restore stabilized flows to Millinocket Stream and the Upper Gorge.

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

EPA-1 No response needed.

EPA-2 We have not simply endorsed the applicant's proposal but have recommended several additional enhancement measures. With respect to the Upper Gorge, fish and wildlife agencies did not recommend additional flow enhancements beyond those proposed by GNP. Following a 10(j) meeting with the U.S. Department of the Interior, we recommend a year-round flow of 60 cfs in Millinocket Stream. In the Back Channel, we have given equal consideration to fish and wildlife and human resources in the EIS. The DEIS stated that there would be few fish produced in the Back Channel even with additional flows, relative to the costs due to lost power and within the context of fisheries within the region. We considered enhancements of existing resources within the context of balancing power and non-power uses of the project area.

EPA-2

COMMENTS FROM ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

2

WATER QUALITY

Back Channel

The state water quality certification confirms EPA's position that the Back Channel is a classified water of the United States. As such, this waterway must meet water quality standards attributed to its class, including the "habitat for fish and other aquatic life" designated use and the criteria necessary to support that use. Other protected uses include, but are not limited to, recreation in and on the water, fishing, navigation, and hydroelectric power.

According to Maine's water quality standards, the Back Channel and other waters of the West Branch "from the outlet of Ferguson and Quakish Lakes to its confluence with the East Branch of the Penobscot River, including all impoundments" are designated Class C waters.¹ The aquatic life designated use criteria for Class C waters require that discharges "support all species of fish indigenous to the receiving waters and maintain the structure and function of the resident biological community."^{2,3} As the state certificate notes, the West Branch of the Penobscot River generally supports both cold water and warm water fish species. Hajor West Branch fisheries are for salmon, lake trout, smelt, burbot, and white perch, while minor fisheries exist for lake whitefish, pickerel, brook trout, and smallmouth bass. Additionally, the river supports a typical Northern Maine assemblage of non-sport species.

The dEIS acknowledges that because of the limited amount of habitat under current flow conditions, the fish community in the Back Channel is restricted (dEIS page 3-20). Likewise, the relicense application to FERC recognizes that the "Back Channel has a sparse fish population and does not support coldwater species with current flow management." (Great Northern Paper Final Application, EJ.1-51). Failure of the Back Channel to support indigenous aquatic life under existing leakage flow demonstrates nonattainment of Maine's water quality standards, as confirmed by Maine's 1994 water quality report:

"Water quality sampling indicates that this waterbody segment

¹38 M.R.S.A. §467(7)(C)(1)(f)

²38 M.R.S.A. §465(4)(C)

³Maine's water quality standards define "indigenous" as "supported in a reach of water or known to have been supported according to historical records" and "resident biological community" as "aquatic life expected to exist in a habitat which is free of the influence of the discharge of any pollutant." 38 M.R.S.A. \$466(8) {(1).

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

EPA-3 We have revised section 2.2.1.2 of the EIS to include MDEP's waiver of water quality certification for the Back Channel. We have revised section 3.4 to include information in MDEP's 1994 report that became available after the DEIS was published. The legal status of the Back Channel relative to Maine's 401 WQC will be addressed in the order for the project.
3

does not attain the aquatic life standard of its classification. The cause of non-attainment is the dewatering of this segment due to hydroelectric power generation."⁶

In approving relicensing of these hydropower facilities, FERC must ensure that state water quality standards are met. EPA believes these standards would be met only by restoring flows to the Back Channel sufficient to support an indigenous aquatic community.

Throughout the NEPA and FERC application process, EPA and the United States Fish and Wildlife Service (USFWS) have maintained that with sufficient flows the Back Channel could provide fisheries habitat for Atlantic salmon and other coldwater fish species. An Instream Flow Incremental Method (IFIM) study conducted by the applicant showed that flows between 350 and 500 cubic feat per second (cfs) would provide babitat for all lifestages stages of salmon. According to FERC's analysis in the dEIS, flows of up to 500 cfs would increase habitat for juvenile salmon, a target lifestage, by 460 percent (dEIS page 4-28), while providing habitat for "several hundred" adult salmon (dEIS page 5-25). However, FERC concluded that while restoring flows to the Back Channel would "definitely increase total available habitat for many elements of the aquatic ecosystem (including the macroinvertabrate community and brook trout), the landlock salmon stock would not be substantially increased" (dEIS page 4-31). Based on its determination that there will only be minor enhancements to the game fish resource, coupled with the anticipated costs to the applicant of providing the increased flows, FERC determined restoring flows to the Back Channel was not warranted.

EPA disagrees with the dEIS assertion that the Back Channel cannot be managed for landlocked salmon. Based on information provided by USFWS, EPA believes Grand Falls would not be a barrier to salmon movement through the Back Channel: it is our understand that anadromous runs of sea run Atlantic salmon extended considerably further upstream beyond Grand Falls before dam construction blocked their passage. EPA also does not believe that seasonal high flows and droughts would prevent establishment of the fishery as FERC concludes; these are natural conditions even in unregulated rivers to which fish and fishery populations naturally adapt. Additionally, FERC's conclusion that restoring flows to the Back Channel would not significantly enhance the fishery value of this reach is inconsistent with FERC's endorsement of the benefits of the applicant's proposal to develop spawning and nursery habitat near Holbrook Pool. As the dEIS states, both habitats would support the same number of adult fish (dEIS pages 4-19 and 4-53).

In considering the environmental benefits to be derived from increased flows to the Back Channel, EPA believes the dEIS errs in

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

EPA-4 The DEIS stated that there would be few fish produced in the Back Channel even with additional flows, relative to the costs due to lost power and within the context of fisheries within the region. We considered enhancements of existing resources within the context of balancing power and non-power uses of the project area.

> Furthermore, fisheries enhancements elsewhere in the project area, such as North Twin drawdown limits or increased minimum flows in Millinocket Stream, could be compromised if flows are enhanced in the Back Channel. Because MDIFW's management goals give priority to these other enhancements, we have not changed our position regarding flows in the Back Channel. We have, however, added text to the FEIS clarifying the extent to which Grand Falls is a barrier to fish movement. See also our response PIN-35.

EPA-4

EPA-3 Cont

> EPA-3 Cont

focusing exclusively on game fish. As discussed above, Maine's water quality standards protect the full spectrum of aquatic life. For these reasons, EPA recognizes the benefits to fisheries of flows from 350 - 500 cfs, and supports restoration of flows to the Back Channel.

EPA is concerned that the dEIS may overestimated the burden on the applicant of restoring flows to the Back Channel. According to Table 5-4 (dEIS page 5-14), flows of 350 cfs into the Back Channel would result in a 3.35 percent reduction in hydroelectric production from the Ripogenus facility. As hydropower production from these two projects provides less than 50 percent of Great Northern's 1994 energy needs, this reduction in actual power generation represents less than 1.7 percent of Great Northern's total power needs. EPA believes improvements in energy generation efficiency or conservation have the potential to offset this loss in energy production. Additionally, even were Great Northern to replace the 20,800 MM of lost power at the prices cited in the dEIS (\$33 per MMh), additional cost for a project alternative does not in itself render an alternative impracticable for the purposes of consideration as a reasonable alternative under NEPA.

<u>Navigation</u>

As noted above, Maine's water quality standards include "navigability" as a designated use for Maine's Class C waters. The dEIS does not adequately address navigability in the Back Channel, except to state that "neither commercial nor private whitewater boating groups have expressed any interest in expanding whitewater boating in (Millinockst Stream and the Back Channel)" and that the "significant flows that would be required to make these streams navigable would jeopardize other environmental enhancements" (dEIS page 4-48). EPA believes the dEIS analysis fails to consider the Penobscot Indian Nation's (PIN) interest in establishing sufficient flows to the river restored to support its traditional practice of navigating the Back Channel by cance.

FERC, in its request to the applicant for additional information, recognized this navigation use with its request that the applicant "determine the minimum flow required for cance passage through the Back Channel" (June 8, 1992 letter from Dean Shumway to James Carson, Great Northern Paper). The applicant's response that the Back Channel is not navigable by cance (application Vol. XIII) contradicts PIN's historical accounts of up and downstream cance passage with one or two short portages (see PIN's "Comments and Recommendations" to FERC dated May 21, 1993). EPA believes the navigability issues should be addressed in the final EIS and before issuance of any operating license for this development.

Mercury

EPA-7 In response to observations by EPA and other parties regarding

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

EPA-5 We disagree. The relatively meager environmental benefits that would be derived from restoring flows to the Back Channel are not worth the cost in terms of lost power and socioeconomic benefits that would occur.

Energy conservation was identified as an alternative during the scoping process. However, we concluded in the DEIS that energy savings gained through conservation have largely been offset by increased energy demand from GNP's plant modernization efforts. This conclusion was based upon our review of GNP's conservation and modernization programs, as documented in Exhibit H, and the memorandum of Owen Merrill of GNP, both of which are based upon actual plant data. The claim that there is an enormous conserved power potential in the GNP facilities by conservation intervenors, on the other hand, is based upon generic or theoretical evidence, which staff did not accord the same weight. The staff also notes that many of the arguments posed by the intervenors have been raised throughout this proceeding and have been addressed by the Applicant. Detailed responses to additional comments on economics and impacts to fish and wildlife resources are provided below.

FPA-6 The staff reviewed available information regarding the Penobscot Indian Nation's claims to lands and rights within the branches of the Penobscot River and concurs with the decision of the Maine State Department of the Attorney General (see section 4.11.1.2). The staff finds that consideration of the Penobscot Indian Nation's traditional practices within the project area are outside the scope of Section 106. The staff, therefore does not recommend including the Penobscot Indian Nation as a concurring party to the Programmatic Agreement associated with the Ripogenus and Penobscot Mills projects. The staff acknowledges the interest of Penobscot Indian Nation's interest in the management of historic properties potentially eligible for listing in the National Register of Historic Places within the project areas. Accordingly, the staff recommends that GNP consult the Penobscot Indian Nation during the development of the revised Cultural Resource Management Plans for the Penobscot Mills and Ripogenus Projects.

EPA-5

EPA-6

EPA-4

elevated levels of mercury in project impoundments, FERC required additional information from the project applicant regarding mercury concentrations in fish and invertebrates taken from project waters. The results of this study demonstrate that mercury levels in lake trout and other predatory fish were higher in project impoundments than at background sites; however, the study concluded that project operations were not responsible for these elevated mercury levels. EPA believes that this conclusion requires further analysis and FERC's acceptance of these conclusions is premature.

5

At the request of FERC, the applicant did an additional analysis of mercury contamination. Reviews of the applicant's analysis by EPA's technical consultant, Hetcalf and Eddy, Inc. (M&E), 5 and by the U.S Fish and Wildlife Service (USFWS) indicated a number of potential problems with both the applicant's study procedures and conclusions. The applicant's study failed to comply to USFWS prescribed study methods or provide quality assurance and quality control (QA/QC) information. These omissions are significant due disagreement between the study's reported fish tissue to concentrations for lake trout from Debsconeag Lake and smallmouth bass from Molunkus Lake and fish tissue mercury concentrations reported for these lakes in the Regional Environmental Monitoring Additionally, the and Assessment Program (REMAP) study." applicant's study reports generally higher than expected water column mercury concentrations. The validity of the study results cannot be determined without QA/QC information. Finally, EPA is concerned that selection of Dolby Pond as a "control" lake for its study is inappropriate given the elevated sediment mercury concentrations at this location.

Despite the problems with the applicant's study, the dEIS concludes that "project operations probably are not the cause of elevated mercury concentrations" (dEIS page 4-11). EPA is troubled by FERC's assertion that "higher rates of smelt consumption by lake trout in project impoundments could cause faster and greater accumulation of mercury" (dEIS page 4-11). While this may be a viable explanation, no information has been provided to substantiate the claim.

M4E's draft review of the applicant's study states there are significantly higher levels of mercury concentrations in fish tissue from the project impoundments than in the control lakes. The M&E report states that atmospheric deposition, as the applicant contends, is not sufficient to account for this difference. M&E concludes that while there is insufficient data "it appears that

⁵"Review of Great Northern Paper Mercury Study" (Draft), EPA-8 Metcalf and Eddy, May 1994. Cont

⁶REMAP is a joint EPA and state study of water quality in randomly selected Maine lakes.

FPA-7 Statistical tests were conducted for top predatory species since bioaccumulation would be expected for these higher organisms. After adjusting the data for differences in fish length, significantly higher mercury concentrations were detected in Lake Trout from draw-down relative to non-drawdown lakes using Scheffe's Multiple Comparison test. Given that mercury concentrations in bottom feeding fish, and freshwater mussels were similar between project lakes and reference lakes we concluded that the higher concentrations in Lake Trout were not directly related to reservoir draw-down. Due to Lake Trout's top predator status and the abundance of rainbow smelt prey in the project impoundments we concurred with GNP's explanation that the higher concentrations in Lake trout from project water relative to control lakes was due to heavy predation on rainbow smelt.

> No evidence (e.g., published scientific studies) exist linking changing water levels to increased mercury methylation. Atmospheric inputs are the most likely source of mercury to the project area as there are numerous reports of relatively. high mercury concentrations in fish taken from remote lakes throughout North America.

EPA-8 According to GNP, the methodologies used for the collection of samples and measurements of mercury were "performed according to the terms negotiated and conditions outlined in consultation correspondence with the consultation agencies for the work plan. The quality assurance and quality control with the laboratory were rigorous" (GNP 1995¹). In fact, GNP used one of the three laboratories that FWS was under contract with at the time the GNP study was conducted. Although it is true that the QA/QC results were not included in GNP's report, the fact that a FWS approved laboratory was selected to perform the tissue analysis and the fact that the agencies took an active role in the design of the study suggest that agency approved methods were followed. Furthermore. the QA/QC results for the tissue analysis were obtained from GNP (and placed in the public record) which included analysis

¹ GNP Letter dated March 17, 1995 to FERC providing reply comments to agency comments on the DEIS.

EPA-10

EPA-9

EPA-7

EPA-8

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

EPA-8 of duplicates, spiked samples, mercury reference standards,

Cont and blanks. The QA/QC results indicate that the quality of the data obtained from the mercury study were of good quality.

The reported disagreement between the GNP study and REMAP samples mentioned in this comment refers to the differences cited in DOI comments (see comment DOI-44). This comparison indicates that tissue concentration of mercury in lake trout in Debsconeag Lake and smallmouth bass in Molunkus Lake were about 1.7 times higher in the REMAP study relative to GNP's study. Given the variable nature of mercury tissue concentration, and differences between the project lakes and reference lakes, it is not surprising that an independent study found difference is not excessive and yearly differences in mercury availability could account for the observed differences; the REMAP collections were conducted in 1993 while the GNP collections were taken in 1992.

The mercury concentration in surface water taken from project lakes was over 5 times lower than EPA's freshwater acute criteria of 2.4 ppb and was typically near or below the detection limits reported for the study (work plan stipulated detection limit of 1 ppb and 0.2 ppb was achieved). This work plan was reviewed by EPA, FWS, PIN, and MDEP and revisions to the plan were made based on agency suggestions (with the exception of conducting mercury analysis on sediment cores to investigate historical mercury loadings). Although the reported mercury concentrations in surface water may be higher than those reported for other systems, the concentrations in draw-down lakes were similar to nondraw-down takes.

While it would be appropriate to include the QA/QC results, the fact that a FWS approved laboratory was used for the tissue analysis lends credibility to the study results.

£

EPA-10 impoundment and/or project operations are at least partially responsible for the elevated concentrations of mercury in fish tissue from Millinocket Lake North Twin impoundment and Dolby Pond."⁷

While the mechanisms linking hydropower facility operations with elevated mercury levels are not entirely understood, there is evidence for increased mercury methylation rates to be associated with project operations. EPA recommends FERC require postlicensing studies to evaluate the relationship between project operations and mercury levels in project fish and vildlife as a condition of hydropower relicensing of these projects. These studies should be conducted in consultation with state and federal resource agencies, utilize appropriate analytical methodologies, and be coordinated with ongoing REMAP studies to enlarge the data base. EPA recommends that any hydropower license should also contain reopener clauses sufficient to address operational changes necessary to eliminate or reduce any project-operation caused increased mercury levels.

Dissolved Oxvden

EPA supports the need for post-licensing studies to evaluate the relationship between low dissolved oxygen (DO) in Dolby Pond, project operations and mill discharges. As historic discharges may have resulted in accumulations of organic wastes with high biological oxygen demand, EPA believes that studies should be sufficiently broad to include a range of alternatives including dredging the impoundment to reduce sediment oxygen demand. Any license issued during the interim period should include reopener clauses sufficient to address remedial actions identified by postlicensing studies.

Millinocket Stream

EPA-13 EPA believes that year around flows of 60 to 80 cfs are necessary in Millinocket Stream to meet Maine's aquatic life criteria. EPA does not believe that FERC's recommendation to increase the minimum flow from 20 to 30 cfs provides sufficient protection to overwintering redds, or for the rest of the aquatic community expected to exist in Millinocket Stream.

Upper Gorde

EPA-14 As is noted in the dEIS, EPA recommended that flows of 50 cfs be provided from October 1 through June 10 each year. As EPA has stated in previous comments to FERC, these flows are necessary to meet the aquatic life criteria in the Upper Gorge on an annual

> ⁷"Review of Great Northern Paper Mercury Study" (Draft), Metcalf and Eddy, May 1994, p.11.

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

- EPA-8 Dolby pond was incorrectly labeled a control pond in Table
- Cont. 4-2. Mattamiscontis and Molunkus Lake were designated as control lakes for Dolby pond in the revised task plan. This has been corrected in the FEIS.
- EPA-9 Given that mercury concentrations were similar among bottom feeding fish and mussels inhabiting draw-down reservoirs, the data do not support the hypothesis that changing water levels increase mercury mobilization. In addition, average mercury concentration in chain pickerel in Dolby Pond (which does not experience water level fluctuations) had higher mercury levels than chain pickerel collected in the Matamiscontis and Molunkus reference lakes. Although there are no smelt population data for the reference lakes (only the project lakes were extensively sampled for fish abundance in the license application), the fisheries studies conducted in the project impoundments indicate that rainbow smelt populations were high.
- EPA-10 In the Metcalf and Eddy report only Lake Trout was cited to have significantly higher concentrations of mercury (this was also indicated by the analysis conducted by GNP). For all other fish species, mussels, water column measurements, and sediments mercury concentrations were similar between project waters and control lakes (except Dolby Pond which had higher sediment concentrations). With the exception of Lake Trout, elevated concentration of mercury in project impoundments is not indicated by the data collected by GNP, It is commonly accepted among experts on mercury contamination that atmospheric sources from the burning of fossil fuel in power plants and municipal incineration is the major source of mercury in surface waters particularly in remote areas similar to the project area. The statement in the Metcalf and Eddy report that atmospheric deposition is not sufficient to account for the differences observed is not supported by quantitative information.

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

- EPA-11 The study conducted by GNP indicate that there is little evidence linking periodic reservoir draw-down to increased mercury methylation. The possible connection cited in this comment is based on a theory which has not been substantiated by any scientific study. The Metcalf & Eddy report citing reservoir draw-downs potentially causing bioaccumulation in fish tissues references personal communications among several researchers, which is not evidence that a connection exists. Nevertheless, in accordance with the State of Maine's water quality certification, we recommend a study on the effects of reservoir draw-down on mercury levels in fish at the Ripogenus Project, similar to the study conducted for Penobscot Mills. We also recommend GNP cooperate with MDEP and other agencies on additional mercury and other toxic metal studies conducted in the project areas.
- EPA-12 There is currently no evidence to show that project operations affect DO in Dolby Pond. DO problems resulting from BOD loads from the mill are not within FERC jurisdiction; this issue will be discussed further in the license order for the project.
- EPA-13 Based on the IFIM study for Millinocket Stream, which included brook trout, a year-round flow of 60 to 80 cfs would provide optimal habitat for this species. Following a 10(j) meeting with Interior, we recommend a year-round flow of 60 cfs or inflow, whichever is less, for Millinocket Stream.
- EPA-14 Both the state of Maine in its 401 WQC for the Ripogenus Project and the Department of the Interior have accepted a seasonal flow of 100 cfs in the Upper Gorge from July 1 through September 30.

7

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

EPA-14 Cont

ont

WETLANDS

basis.

EPA-15 No response required.

More than 1,251 acres of wetlands are associated with the Ripogenus impoundment. These include 118 acres of alder thickets, 282 acres of wet meadow/smergent marsh, 66 acres of wet meadow/open water, and 785 acres of bog/wet meadow. Areas of conifer swamp/wet forest and deep marsh/aquatic bed have not been measured or estimated. Major wetland systems associated with the impoundment include the 437 acres Brandy Pond system (bog/wet meadow and wet meadow/emergent) and the 183 Quaker Brook system (wet meadow/emergent marsh and bog/wet meadow).

Below Ripogenus impoundment wetlands occur along the West Branch and in the Upper Gorge area between Ripogenus Dam and McKay Station. About 528 acres of riparian wetlands occur along the West Branch in the vicinity of Ripogenus. Most of these wetlands are associated with deadwater areas. The various wetland types include: alder thickets, conifer swamp/wet forest, wet meadow/emergent marsh, wet meadow/open water, and open bog/wet meadow/emergent marsh, wet meadow/open water, and open bog/wet meadow. The surrounding areas are dominated by softwood and hardwood mixed forests. Wetlands in the Upper Gorge area are limited due to the rocky substrate and steep topography.

15. Numerous wetlands are associated with the Penobscot Mills Project and the associated impoundments. Morth Twin impoundment has 296 acres of wetlands. Of this 296 acres, 180 acres are wet meadow/emergent marsh, 114 acres are open bog/wet meadow, and 2 acres are wet meadow/open water. Millinocket lake is connected to 709 acres of wetlands consisting of 383 acres of wet meadow emergent marsh, 316 acres of open bog/wet meadow, 8 acres of alder thickets, and 2 acres of wet meadow/open water. Quakish Lake is associated with 159 acres of primarily wet meadow/emergent marsh. Dolby Pond is connected to 219 acres of wetlands of predominantly wet meadow/emergent marsh. East Millinocket impoundment is associated with very few wetlands due to the paper mill complex and associated mill yards.

> Riparian wetlands associated with Penobscot Mills occur along Millinocket Stream and the West Branch. Millinocket Stream has small areas of emergent and scrub/shrub wetlands which have developed along several deadwater areas along the stream. Portions of the West Branch in this area contain similar wetland types. The Back Channel also contains narrow areas or riparian scrub/shrub wetlands along it's shores.

Section 404(b)(1) Guidelines

The proposed action and alternatives would not require the placement of fill, an activity regulated under Section 404 of the Clean Water Act. However, the proposed action and project

A

alternatives would alter water levels in the Ripogenus and North Twin impoundments and thus could have an adverse effect on the functions and values of the existing wetlands. Additionally, wetlands would also be affected by the diversion of water from natural stream channels such as the Upper Gorge, the Back Channel, and Millinocket Stream.

According to the dEIS, impacts to wetland functions and values at the Ripogenus and North Twin impoundments are similar for all the alternatives considered in the dEIS. Approximately 250 acres of wetlands hydrologically connected to Ripogenus impoundment will continue to be affected adversely by the impoundment drawdowns.⁴ It is unclear how many acres of wetlands will be impacted at North Twin impoundment. The dEIS states that this impoundment is associated with 296 acres of wetlands but does not state how many acres would be affected by the proposed action. Additionally the dEIS does not evaluate what impacts the drawdowns have on aquatic beds and species utilizing the drawdown zone. The final EIS should to quantify the amount of wetlands which will be impacted at North Twin impoundment and evaluate the impacts associated with aquatic beds and species utilizing the drawdown zone.

The dEIS states that wetland enhancements at Black Pond and the Deep Cove sites should be included in the final FERC license to offset watland impacts from impoundment drawdowns. EPA believes these enhancements have the potential to compensate for the drawdown impacts, however, additional information is necessary to allow us to finalize this determination. EPA recommends the final EIS quantify the impacts associated with North Twin impoundment so that the appropriate mitigation acreage can be set. If impacts at North Twin impoundment are large, enhancement at Quaker Brook and other sites should be included in the mitigation plans presented in the final EIS.

While the dEIS includes a brief description of potential wetland enhancement measures at Black Pond, Quaker Brook, and the Deep Cove sites, the final EIS should include a more detailed analysis of these enhancement opportunities. The final EIS should include a description of the wetlands, the enhancement measures proposed, how the enhancement will be executed, and the benefit to wetland functions and values from these measures. As these enhancements are likely to involve construction of dikes or berms to hold back stream flows, these measures likely would require a Section 404 of the Clean Water Act permit from the Army Corps of Engineers (Corps). EPA recommends the applicant contacted the Corps as early as possible to determine compliance with the requirements of the Section 404(b)(1) Guidelines.

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

- EPA-16 The FEIS includes an estimate of the quantity of affected shoreline wetlands at North Twin under the proposed action. Aquatic beds were included in the estimate of affected shoreline wetlands.
- EPA-17 The FEIS includes an estimate of the quantity of affected shoreline wetlands at North Twin under the proposed action. Such effects to project wetlands will be compensated by the wetlands enhancements under the proposed action.
- EPA-18 A description of the affected wetlands was provided in the DEIS. Detailed analyses of constructing the potential enhancement sites was provided by GNP as part of the record in response to AIR #2, October 28, 1993. A qualitative analysis of how wetlands functions and values could be benefitted at the sites is provided in GNP's response to AIR #6, December 7, 1992, also part of the record. These analyses were subsequently verified by Staff during field visits to the sites on August 26, 1993. Your comment concerning Section 404 of the Clean Water Act is noted.

E-110

○ EPA-17

EPA-15

EPA-16

Cont

⁶Impoundment drawdowns affect these hydrologically connected wetlands primarily due to continuing cycle of winter drawdowns exposing these areas to freeze-thaw conditions.

9

Wildlife Impacts

FERC's flow recommendation for the Back Channel would continue to provide only leakage flows of about 12 cfs from Stone Dam. The dEIS concludes that restoring flows to the Back Channel would provide little banefits to wetlands due to the area's rocky substrate. While the restoration of flows to this area would initially disrupt some current wetland systems, EPA agrees with the determination in the dEIS that additional areas of wetlands in peripheral areas would be created and that aquatic vegetation within the channel would be enhanced.

The dEIS states that restoring flows to the Back Channel would disrupt existing beaver population; however, these animals would be able to relocate or adjust to the higher flows. Currently, beaver are being affected adversely by seasonal releases of excess flow over Stone Dam; Stone Dam can pass flows of up to 109,000 cfs. EPA recommends the final EIS document these seasonal releases and quantify flow volumes from Stone Dam to the Back Channel.

The dEIS concludes that drawdowns at Ripogenus and North Twin impoundments have detrimental effects on waterfowl and shorebirds such as the common loon, ring-billed gull, and common term by increasing predation of these species. FERC has recommended establishing artificial, floating mast structures at both impoundments to help increase survival of common loons and other aquatic birds utilizing the area. EFA recommends FERC coordinate with the USFWS on this mitigation proposal and that these measures be committed to in the final EIS.

FERC recommends establishing a buffer zone around the Ripogenus impoundment and GNP-owned property along the Penobscot Mills Project. This buffer zone would establish a 200 foot building set back and a 100 foot vegetative buffer. EPA concurs with this recommendation as it would provide benefits to wildlife resources in the area and would help to maintain or improve water quality.

Additionally, EPA recommends that the final EIS address the following aquatic resource issues:

EPA-23 i. Impacts to wetland functions and values from the drawdowns at Ripogenus and North Twin impoundments should be mitigated. Currently, drawdowns at Ripogenus impact about 250 acres of wetlands. The impacts at North Twin impoundment were not detailed in the dEIS and should be presented in the final EIS so that appropriate mitigation plans can be developed.

The final EIS should include an expanded evaluation of mitigation sites. Specifically, a functions and values assessment of the wetlands to be enhanced should be provided in the final EIS in addition to a discussion of the proposed enhancement measures. The final EIS should discuss how these enhancements will be achieved to

EPA-19 No response required.

- EPA-20 There is no data currently in the record that suggests beaver might be adversely affected by existing seasonal excess flows over Stone Dam. Table 3-3 in the DEIS shows that the peak discharge at Stone Dam is 64,800 cfs, not 109,000 cfs, as EPA suggests. Further, it is misleading to suggest that flows of this magnitude are seasonally passed over Stone Dam. Spillage releases on the order of 29,000 cfs have been recorded at the project by the applicant. There is no evidence currently in the record for much greater regular spills (i.e., order of 109,000 cfs) occurring at Stone Dam. We see no purpose in further documenting these flow volumes in the FEIS.
- EPA-21 We recommend GNP prepare the floating nest structure plans after consultation with FWS, Maine Department of Inland Fisheries and Wildlife, and other interested entities. The plan would be required within 12 months of license issuance.

EPA-22 No response required.

- EPA-23 The estimated effects to wetlands functions and values would be adequately compensated as a result of our recommended license conditions. The FEIS includes an estimate of the quantity of impacted wetlands at North Twin. Wetlands enhancements recommended for the Penobscot Mills Project would adequately compensate for affected wetlands functions and values at North Twin.
- EPA-24 The applicant provided an adequate qualitative functions and values assessment in their response to AIR #6, December 7, 1992. A detailed discussion of the proposed enhancement measures was also provided as part of the record in response to AIR #2, October 28, 1993.

EPA-20

EPA-19

[] EPA-21 그

EPA-22

COMMENTS FROM ENVIRONMENTAL PROTECTION AGENCY	1
ON UPPER PENOBSCOT RIVER BASIN DEIS	-

	10	
EPA-24 Cont	permit full evaluation of the value of the mitigation measures. EPA recommends the applicant contact the Corps to determine whether permits will be needed for these enhancement projects.	EPA-25
FDA 05	2. Restoration of minimum flows to the Upper Gorge, Millinocket Stream, and the Back Channel would provide the substantial benefits to the aquatic ecosystem in these areas. Consistent flows would enhance and extend without in these areas. Idditionally disburged	EPA-26
EPA-25	and other wildlife would also likely benefit from the restoration of flows to these areas. EPA recommends FERC adopt these restoration flows as part of its proposed action.	EPA-27
EPA-26	3. EPA recommends that the placing of artificial nesting structures on Ripogenus and North Twin impoundments, as well as the establishment of buffer zones around Ripogenus impoundment and land owned by the applicant along the Penobscot Mills Project, be	
	committed to in the final EIS and incorporated as a condition of FERC's license.	
	On the basis of the comments above, we have rated this project "Environmental Concerns - Insufficient Information" (EC-2). Please see the attached sheet for a full explanation of this	

Thank you for the opportunity to review and comment on this draft EIS. EPA is available to work with FERC and the applicant to assure that our concerns are adequately addressed in the final EIS. If you have any questions about our comments, please contact Steven John of my Environmental Review team at 617/565-3426.

Sincerely,

rating.

John P. Devillars

Regional Administrator

RESPONSES TO ENVIRONMENTAL PROTECTION AGENCY ON UPPER PENOBSCOT RIVER BASIN DEIS

- See responses to EPA-4, EPA-5, EPA-13 and EPA-14
 - See response to EPA-21 and CI-8
 - No response required.

SUMMARY OF RATING DEFINITIONS AND FOLLOW-UP ACTION

Environmental Impact of the Action

LO--Lack of Objections

The EPA review has not identified any potential impacts requiring substantive changes to the proposal. The review may have disclosed opportunities for application of mitigation measures that could be accomplished with no more than minor changes to the proposal.

EC--Environmental Concerns

The EPA review has identified environmental impacts that should be avoided in order to fully protect the environment. Corrective measures may require changes to the preferred alternative or application of mitigation measures that can reduce the environmental impact. EPA would like to work with the lead agency to reduce these impacts.

EO-Environmental Objections

The EPA review has identified significant environmental impacts that must be avoided in order to provide adequate protection for the environment. Corrective measures may require substantial changes to the preferred alternative or consideration of some other project alternative (including the no action alternative or a new alternative). EPA intends to work with the lead agency to reduce these impacts.

EU-Environmentally Unsatisfactory

The EPA review has identified adverse environmental impacts that are of sufficient magnitude that they are unsatisfactory from the standpoint of public health or welfare or environmental quality. EPA intends to work with the lead agency to reduce these impacts. If the potential unsatisfactory impacts are not corrected at the final EIS stage, this proposal will be recommended for referral to the CEQ.

Adequacy of the Impact Statement

Category 1--Adequate

EPA believes that draft EIS adequately sets forth the environmental impact(s) of the preferred alternative and those of the alternatives reasonably available to the project or action. No further analysis or data collection is necessary, but the reviewer may suggest the addition of clarifying language or information.

Category 2-Insufficient Information

The draft EIS does not contain sufficient information for EPA to fully assess environmental impacts that should be avoided in order to fully protect the environment, or the EPA reviewer has identified new reasonably available alternatives that are within the spectrum of alternatives analyzed in the draft EIS, which could reduce the environmental impacts of the action. The identified additional information, data, analyses, or discussion should be included in the final EIS.

Category 3-Inadequate

EPA does not believe that the draft EIS adequately assesses potentially significant environmental impacts of the action, or the EPA reviewer has identified new, reasonably available alternatives that are outside of the spectrum of alternatives analysed in the draft EIS, which should be analyzed in order to reduce the potentially significant environmental impacts. EPA believes that the identified additional information, data, analyses, or discussions are of such a magnitude that they should have full public review at a draft stage. EPA does not believe that the draft EIS is adequate for the purposes of the NEPA and/or Section 309 review, and thus should be formally revised and made available for public comment in a supplemental or revised draft EIS. On the basis of the potential significant impacts involved, this proposal could be a candidate for referral to the CEQ.

COMMENTS FROM GREAT NORTHERN PAPER, INC. ON UPPER PENOBSCOT RIVER BASIN DEIS GREAT NORTHERN PAPER. INC. One Kalandin Aver Millino Aet ME CAAGE 3ar (2017) 23 (13) Subsidiary of Bowarei Incorporation
1-8603/1

•

• :

February 22, 1995

Honorable Lois D. Cashell. Secretary Federal Energy Regulatory Commission 825 North Capitol Street, N.E., Room 3110 Washington, DC 20426

RE: Comments on Draft Environmental Impact Statement Penobscot River Basin Docket Nos. 2458 and 2572

Dear Secretary Cashell:

In November of 1994, the Commission distributed the Draft Environmental Impact Statement (DEIS) for the Ripogenus and Penobscot Mills Projects. By Notice dated December 15, 1994, the Commission invited all participants to submit comments. Pursuant to that Notice, Great Northern Paper, Inc. ("Great Northern") hereby respectfully submits these comments on the DEIS.

Enclosed for filing with the Commission on behalf of Great Northern licensee and applicant for new licenses for Project Nos. 2458 and 2572 are an original and eight (8) copies.

Copies of this filing have been served on all parties listed on the Official Service List for this proceeding.

Very truly yours,

3min R Stite

Brian R. Stesson Manager, Environmental Affairs

BRS/blw Enclosure

123-2664Y

cc: FERC Service List

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

CERTIFICATE OF SERVICE

I hereby certify that I have this day served the forgoing document upon each person listed on the official service list compiled by the Secretary in these proceedings.

G Xilis

Brian R. Stetson Manager, Environmental Affairs Great Northern Paper, Inc. One Katahdin Avenue Millinocket. Maine 04462 (207) 723-2664

Dated: February 22, 1995 Millinocket, Maine

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

INTRODUCTION

GNP-1 Opinion noted.

Great Northern. FERC staff, state, local and federal officials, private interest groups and hundreds of interested persons have all participated in this relicensing process over the last eight years. Voluminous information has been collected and submitted to Commission by Great Northern over the past eight years and more recently as part of the scoping process under the National Environmental Policy Act (NEPA). The DEIS's role in the NEPA process is clearly intended to be a comprehensive review of that substantial record which has been developed for the Ripogenus and Penobscot Mills Projects. Great Northern appreciates the efforts and work of FERC staff and many of these parties as it has sought to develop a water use plan and project enhancements which will best balance competing uses and preserve and protect the many important multiple uses of the project area and its waters.

As stated in the DEIS, Great Northern has worked throughout the eight-year period of this process to develop a balance for competing uses of the water resource. The Water Use Plan (WUP), which was developed through consultation with agencies and interest groups, clearly represents the best balance. It establishes a firm basis by which to manage this resource on the priorities of public safety, environmental protection, and recreation while still providing adequate power to the company's manufacturing facilities. This lowest cost source of electric power is essential to Great Northern's ability to compete in the pulp and paper marketplace, and the difficult economic conditions in recent years further exemplify the importance of this power to Great Northern. Although some have tried (and likely will continue) to criticize the WUP, no viable alternative has been proposed, even though the model has been under intense scrutiny for the past three years.

An essential component of the WUP is the maintenance of leakage flow in the back channel. The DEIS reached the correct conclusion noting that there is no biological justification for additional flow in this channel. Great Northern strongly agrees with this conclusion and urges the Commission to continue to support this critical position in the Final Environmental impact Statement (FEIS).

GNP-1

There is, however, one significant problem raised by the conclusion included in the DEIS. It concludes that the project boundaries be expanded to include all company-owned land within two hundred feet of the high water mark. Furthermore, on the Ripogenus impoundment the DEIS recommends that Great Northern be required to purchase an estimated 2,000 acres of shoreline not presently owned by the company. While Great Northern agrees with the goal of assuring shoreline protection, the Commission has greatly underestimated the costs and impacts of this recommendation and has not given appropriate consideration to Maine's lake protection regulatory structure which has been developed over the last twenty years and sharply refined in recent years. This is a structure which has been developed with full, open participation of the people of Maine as required by the State's legislative and regulatory processes. It is incorrect for intervenors to assert that Maine's Land Use Regulation Commission (LURC) can act independently to amend its regulations or abolish them through an act of the Legislature. As the regulatory and legislative processes demand, changes in State standards can only occur in a public forum. Thus, the resulting laws and regulations are a reflection of the people of the State of Maine and not a small coalition of national environmental organizations.

GNP-2

In recommending ordering project boundary expansion, the Commission has significantly underestimated the complexity and cost of obtaining the protection as prescribed (DEIS, page 4-62). These requirements would force, through a questionable, and at the very least bitterly contentious eminent domain process, the acquisition of lands around Ripogenus impoundment from the present owners, most of whom have publicly stated their unwillingness to sell. Also, the State of Maine presently owns approximately nine miles of the shoreline which the Commission recommends be obtained by Great Northern and can only transfer ownership through an act of the Maine Legislature. Finally, for Great Northern to assume the responsibility for regulating shoreline activities, under FERC's direction, in lieu of State and local agencies is an unreasonable burden and clearly inappropriate for a private company. This point was repeatedly supported by comments made at the hearing on January 25, 1995.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

- GNP-2 Opinion noted. The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period. GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC's land use regulations. The staff proposes two options for the Ripogenus Project area: (1) accepting the conservation easement proposed by GNP and the state of Maine: or (2) a 200-foot boundary expansion on GNP-owned lands. For the Penobscot Mills Project area, the staff recommends a 100-foot expansion of the project boundaries on GNP owned lands. The staff recommends that existing structures would be grandfathered under any of the project boundary expansion options. Any proposed variances to LURC's regulations would warrant review and approval by FERC (see section 4.9.3).
- GNP-3 The staff revised potential cost estimates for the shoreline easements upon review of comments received during the DEIS comment period and updated land valuation information. The staff estimated that the potential cost of approximately \$24.6 million, based on waterfront footage for the easements proposed in Alternative 1, would be greater than previously determined in the DEIS. Our analysis indicates that the additional protection of the 500-foot expansion does not merit the much higher cost of that alternative as compared to the recommended alternative (\$24.6 million versus no direct costs). See section 4.9 for further discussion.

GNP-3

Even if the property could be obtained, the cost is over \$10 million and not the \$2 million estimated in the DEIS. This higher cost is consistent with locally available valuation information, testimony presented at the January 25 hearing, and, most importantly, has been confirmed by the Thompson Appraisal Company. Inc. which developed the Rangeley appraisal upon which FERC inappropriately selied for its conclusions with respect to the buffer zone costs. This cost assumes that the Commission does not intend that Great Northern acquire existing camps and humans on project impoundments. If existing developments are to be included, the costs would ingrease substantially. Since both mills are located within 200 feet of project impoundments, the direct impact on company operations of including, in any way, existing developments much be enormous. Furthermore, the limits FERC places on development of areas in the Dentificot Mills Project would take from Great Northern the opportunity to realize value faux fault specifically directed by the State of Maine towards properly accomplished development. This lost value would exceed many millions of dollars.

Great Northern believes that there is no junification for such drastic measures in substituting the Commission's judgment for that of the state agencies charged with managing these resources. As the DEIS states and the antironmental studies performed in this relicensing process demonstrate, the resources of the West Branch region have flourished under Great Northern's stewardship within the State of Maine regulatory framework. In reviewing the impacts of proposed license conditions. Great Northern asks that the Commission thoroughly consider the need for each putposal and not underestimate or ignore the impacts of the measures required.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-4 Opinion noted.

GNP-3

Cont

GNP-4

- 3 -

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS



RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

.

.

LAND USE

Impoundment Buffer Zone/Setback Proposal

Despite the fact that the lands surrounding project impoundments are highly controlled by a thorough, publicly developed regulatory program implemented by the State of Maine. the DEIS proposes that the project boundaries for both the Ripogenus and Penobscot Mills Projects impoundments be expanded to include a 200 foot building setback and a 100 foot vegetative buffer. One aspect of this proposed imposition of setbacks/buffer zones would require Great Northern to obtain title to, or conservation easements for, significant land areas bordering the Ripogenus impoundment owned by other persons, including the State of Maine, who have publicly stated that they would not willingly sell their land for this purpose, The DEIS also proposes the imposition of the various restrictions (200 foot setback/100 foot buffer) on shoreland owned by Great Northern within the Penobscot Mills Project.

As the hearing testimony and the following comments make clear, this proposal is unnecessary and unsupportable as a matter of public policy, environmental protection and fairness, not only to Great Northern and the State of Maine, but also to those who use the resources the most and live and work in the Millinocket area. In brief summary, Great Northern's basis for objecting to this proposal, discussed in great detail in this section of its comments, is as follows:

1. The State of Maine, through its LURC, has developed and implemented a Lakes Protection Plan. This plan, as well as other aspects of LURC's regulations, provides protection for the values FERC is concerned with, but in a balanced manner based upon a long-term public process which considered the rights and needs of a wide range of interests. The DEIS proposal is both unnecessary in light of LURC's action and contrary to State policy.

2. The DEIS's projections of possible new shorefront development are not possible as a matter of law, fact, practicality and economic reality. Furthermore, some areas where the DEIS seeks to severely limit development potential have been specifically designated by the State of Maine as appropriate areas to which development should be directed. The DEIS's proposal would be contrary to Maine's need to wisely direct tourism and take development to certain already developed, accessible areas,

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-5 See resoonse GNP-2.

- GNP-6 See response GNP-2. The staff's review of LURC's land use regulations resulted in revised land use assessment and recommendations (see section 4.9). Under the recommended alternative, the proposed conservation easements and/or boundary expansion would provide long-term protection of valuable shoreland resources in accordance with LURC's regulations (see section 4.9.3).
- Comment noted. The staff revised potential build-out GNP-7 estimates to reflect LURC's subdivision regulations. As stated in the DEIS, the staff acknowledges these estimates do not account for development limitations such as steep slopes, poor soils, wetlands, or access (see section 4.9.1.1).

GNP-5

GNP-6

E-122

GNP-7

- 3. The setback/buffer zone proposal would seriously impair, or at the very least. raise significant confusion as to the ability of leaseholders to use their lands in the proposed expanded boundary. The proposal raises a host of questions. for example: can leases that expire each year be renewed? Can unbuilt lots subject to one year leases be built upon anytime in the future? Can existing GNP-8 camps be enlarged or replaced? Can new docks or piers be built? Will existing campowners be subject to LURC standards or new FERC rules on their property? Can camps be freely sold or otherwise change ownership without any new requirements? Are commercial leaseholders treated exactly as residential campowners?
- 4. Great Northern would be placed in the impossible position, without legal authority, of policing and enforcing all activities by leaseholders and the GNP-9 general public within the expanded project boundaries covering well over 200 miles of shorefront.
- 5. Aesthetic, water quality and recreational resources in the project areas are not in jeopardy and do not need the imposition of draconian, divisive and confusing conditions to protect them for the future. Water quality is meeting all required standards and recreational and aesthetic opportunities are the subject of high praise by state and federal agencies, as well as resource users. GNP-10 Furthermore, FERC's expressed concern that LURC's timber harvesting standards are not as stringent as U.S. Forest Service (USFS) criteria is misplaced. The 1972 USFS reference supplied by FERC is apparently outdated and unavailable for review. In any event, more recent USFS guidelines to control erosion and sedimentation from logging in the Northeast are virtually identical to the current LURC standards.
 - 6. Even if one were to assume that expanded project boundaries are needed, the DEIS's projected land or easement acquisition costs are a fraction of what the costs would be. The conclusion that estimated costs are completely out of line with what the true easement or fee acquisition costs would be is supported by:
 - a. The appraisal firm which the DEIS cited as its authority Thompson Appraisal Company. Inc.:

 - One of Maine's leading independent appraisers Norman Gosline; A knowledgeable expert from Maine's leading forestry and с. woodlands consulting firm - Robert Fiske of J. W. Sewall Co.;
 - Current Maine Bureau of Taxation land valuations; d.
 - The public testimony of knowledgeable local real estate e. professionals.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-8 Opinion noted. The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). See response GNP-2.

> Within the proposed boundary expansion areas, existing structures would be grandfathered, and GNP would have the authority to review and approve proposed actions as established by the Commission under the Standard Land Use Article. See Section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.

GNP-9 See response GNP-8.

GNP-10 The staff reviewed LURC's timber harvesting regulations and determined that they provide adequate controls for vegetative clearing and vegetative filter strips in association with timber harvesting practices. The staff revised the FEIS accordingly (see sections 3.11.1.1 and 4.9).

> See response GNP-8. The proposed boundary expansions and/or conservation easements would provide adequate protection.

GNP-11

As all of these sources confirm, even if the land could be acquired, the true

cost is between four and ten times that of the DEIS's estimate. This cost (or even the \$2 million suggested by the DEIS), if imposed on Great Northern, would seriously affect its ability to improve the mills' competitive position. In addition, because the DEIS's proposal would so severely limit remaining use of the land, the experts concur (and common sense agrees) that the purchase of an easement, as opposed to fee, of the shorefront buffer would not meaninefully reduce the cost.

COMMENTS FROM GREAT NORTHERN PAPER. INC.

ON UPPER PENOBSCOT RIVER BASIN DEIS

7. Implementing, or attempts to implement, the DEIS's proposal is virtually certain to result in contention, divisiveness and litigation. The State of Maine and landowners, including some who have owned their property for over 100 years, have clearly stated that they will not willingly sell their land to comply with this condition. This means that the ability of FERC to require, and Great Northern to utilize, the Federal Power Act's (FPA's) eminent domain power will be tested to its limits in challenges by landowners and state officials to both the license and the taking process.

Existing Requirements of Maine Law Provide all Necessary Protection for Shoreline Areas

Great Northern agrees that it is appropriate to assure reasonable protection of shoreline areas and to assure that inappropriate activities do not take place which would jeopardize those values required to be protected by the license. The question, however, is how to achieve this protection in the most cost-effective way which does not unreasonably intrude on the authority or goals of the State of Maine or unnecessarily affect the existing leaseholders and users of the land. The State of Maine has developed and implemented a regulatory structure which Great Northern believes meets FERC's obligations and objectives in a manner more consistent and cooperative with State of Maine policy and practices while being more predictable and reasonable in its cost and impact on Great Northern, users of the project resources and the people of the State of Maine. A brief description of the recent extraordinary efforts undertaken by the State to protect its lakes is necessary to fully understand Great Northern's position that the proposed conditions on buffer zones are unnecessary and inappropriate and unwisely jeopardize the State's land use goals.

> FERC has cited 18 C.F.R. § 2.7(A) as support for the expectation that the Scenare acquire in fee enough land to assure optimum development of excessional resources. Section 2.7 stato states that such expenditures shall be "reasonable" and that action under Section 2.7 shall "encourage and planning — In addition to raising the issue, discussed elsewhere, or whether the extended hoffer zones for water quality and aesthetic reasons can be accomplished using FERC's eminent domain authority for other than "recreational resources." this provision makes clear the week for cooperation with the Same

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-11 The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). Under the revised recommendations. GNP would have to acquire conservation easements on non-GNP-owned land only under Alternative 1. The staff revised potential cost estimates for the shoreline easements upon review of comments received during the DEIS comment period and updated land valuation information. The staff estimated that the potential cost of approximately \$24.6 million, based on waterfront footage for the easements proposed in Alternative 1, would be greater than previously determined in the DEIS. Under the recommended alternative (Alternative 2) the proposed conservation easements and project boundary expansion would be only on GNP-owned lands. Our evaluation of benefits suggests that the additional protection of the 500-foot expansion does not merit the much higher cost of that alternative as compared to the recommended alternative (\$24.6 million versus no direct costs). See section 4.9 for further discussion.

GNP-12 Opinion noted.

GNP-13 Opinion noted.

E-124

GNP-13

GNP-11

GNP-12

Pursuant to Maine law, LURC is charged with designating areas within its jurisdiction where development would jeopardize significant natural, recreational and historical resources, as well as areas which are appropriate for commercial forest product uses and areas appropriate for residential, recreational, commercial or industrial development (12 M.R.S.A. § 685-A). Once LURC designates protection, management or development districts, it must then promulgate land use standards regulating uses of these areas. Pursuant to the laws of the State of Maine, these land use standards must, among other things:

Encourage the most appropriate and desirable uses of tand resources consistent with LURC's comprehensive plan;

Protect and preserve significant natural, scenic and historic features:

Encourage minimal adverse impact of one use upon the use of surrounding areas; and

Protect against water pollution and other environmental intrusions (12 M.R.S.A. § 685-A.3).

m GNP-13 → Cont LURC's law also requires that it adopt a Comprehensive Land Use Plan to guide it in developing land use standards and fulfilling its planning and environmental protection mandates (12 M.R.S.A. § 685-C). The first LURC Comprehensive Land Use Plan was adopted in 1976 and was significantly revised in 1983 and most recently on June 7, 1990. This latest amendment to LURC's Comprehensive Land Use Plan incorporated two major State planning initiatives, the Wildlands Lakes Assessment and the Lakes Action Program. both dealing exclusively with lake protection issues and the relationship between land use and water quality, precisely the issues the DEIS seeks to deal with through its setback/buffer zone proposals. Furthermore, even though LURC is presently in the process of a comprehensive review and update of its Comprehensive Land Use Plan, it chose to proceed and implement an immediate amendment to its standards to assure the existence of measures designed to improve lake management. According to LURC.

"these measures, which will dramatically improve the Commission's [LURC's] ability to wisely manage take resources, are needed immediately to deal with rapidly increasing development pressure on lake shorefront" (see amendment of the Comprehensive Land Use Plan, June 7, 1990),

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

The Wildlands Lakes Assessment and Lakes Action Programs were initiated to study all lakes with a surface area of ten acres or more in LURC's jurisdiction. rank the resource values for each lake in terms of fisheries, scenic quality, shoreline character, wildlife and other resources, and place lakes of significant or outstanding resource values into resource classifications. In the process, LURC collected information pertaining to land and water uses including such things as zoning, access and shoreline development (id at 3), all in a very public and thorough fashion which considered a wide range of interests and objectives.

Following completion of the Wiklands Lakes Assessment, LURC appointed a Lakes Policy Committee consisting of members of the public, statewide environmental and sportsmen organizations, major landowners, university scientists and public officials. The committee was charged with developing a proposal for a policy that might guide future LURC lake management decisions and identify specific actions that should be taken to implement LURC's policies. The Lakes Policy Committee was specifically directed to develop a balanced approach utilizing innovative regulatory and non-regulatory lake management techniques. The actions identified by this committee were consolidated into a proposed plan. "An Action Program for Management of Lakes in Maine's Unorganized Areas," which was accepted by LURC in January 1989.

During this process LURC also recognized the need to update its approach for reviewing impacts on lake water quality. For this reason, the LURC staff developed a systematic approach to understanding the impacts of clearing and development on water quality and worked closely with the Lake's Division of the Maine Department of Environmental Protection (DEP) and the Maine Department of Inland Fisheries and Wildlife (DIFW) to minimize the impacts of these activities.

Following its Wildland Lakes Assessment, Lakes Action Program, water quality review and other initiatives, on June 7, 1990 LURC revised its Comprehensive Land Use Plan and amended its rules dealing with development and conservation of lakes. The new regulatory program dealing with lakes was incorporated into the revised Comprehensive Plan with the following policy statement by which LURC committed to:

"Guide take development based on identified land use characteristics and natural resource values, conserving important values and directing development toward those lakes or lake areas most capable of absorbing new development."

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-13 Cont

Pursuant to this policy guidance, and as required by LURC's statutory review criteria (12 M.R.S.A. § 685), LURC established specific review criteria for any shoreland permit to further assure that any shoreline development would fit "harmoniously into the existing natural environment". Among other things, LURC's revised Comprehensive Land Use Plan states that:

"The Commission will give specific consideration to the affect that a proposed development will have on lake water quality. For a proposed development on lakes, the Commission will require a finding regarding the probable effect of the proposed action on lake quality . . . If unacceptable water quality degradation will require soft additional measures, the Commission will deny the application."

Review criteria for natural character were also contained in the revised Comprehensive Land Use Plan by which LURC stated that it would:

"... seek to maintain the natural character of lakes by encouraging: visual screening of larger developments and non-conforming structures: consolidated use of recreation facilities such as boat docks and access ramps: and provisions for long-term protection of undeveloped shoreland as part of subdivisions and commercial, industrial and other non-residential proposals." (id at 5).

3 LURC made a couple of key changes to its regulations as part of this process specifically in order to address lake water quality and aesthetics. The minimum shoreline frontage was increased to 200 feet and the minimum setback for structures was increased to 100 feet for great ponds (a body of standing water ten acres or greater in size) (LURC Regs. Chap. 10.17.B, 1.b(2) and d(1)(b)). Nearly all of the structures built along the Ripogenus and Penobscot Mills impoundments were built well before these changes were adopted when minimum shore frontage was 150 feet and setbacks were 75 feet or nonexistent. Therefore.

development.

LURC also has adopted clearing standards that require a vegetative buffer for new structures. These standards were revised in June of 1990. Within 100 feet of the high water mark of great ponds, LURC regulates the size of cleared openings, footpath width and configuration, vegetation removal, soil disturbance and which species may be planted in natural openings (LURC Regs. Chap. 10.17.A.2a (3)(a-e)). Also within 100 feet, no more than 30% of the volume of trees may be removed in any ten year period; between 100 and 250 feet, no more than 40% may be removed (LURC Regs. Chap. 10.17.A.2b).

the appearance or impact of existing development will not be the same as future

After establishing comprehensive land use policies and goals guiding lake development. LURC developed six specific lake classifications based on natural and other resource values. Over 50% of the Ripogenus impoundment was placed in the Management Class 2 category (P-AL) which was limited to takes having less than one development unit per shore mile within 250 feet of the normal high water mark as of November 17, 1988 and which have high resource values but are relatively accessible. If no dwellings presently exist in the Lake Protection Sub-district (P-AL), the zoning only allows one single-family unit per shore mile in the future. The main portion of the Ripogenus impoundment, known as Chesuncook Lake, is zoned (P-AL) thereby providing strict protection for existing aesthetic. recreational and water quality values.

In contrast to the designation made for Chesuncook Lake, LURC designated the undeveloped parts of Ambajejus, Pemadumcook, North Twin and Elbow Lakes as Management Class 3 or "Most suitable for Development." In other words, LURC determined that when development takes place, these are among the relatively few lakes in its entire 10.5 million acre jurisdiction where development should be directed.

The above discussion makes it clear that very significant State of Maine legal protection measures exist to assure that only minimal shorefront development can take place. This development must meet the highest standards of environmental protection on virtually all of the land surrounding the Ripogenus and Penobscot Mills Projects impoundments. In addition, the program presently in place in the State of Maine is a reflection of an extensive and serious effort by the State of Maine to devise a meaningful takes protection policy with substantial public input. These requirements are firmly set in law and regulation in the State of Maine. a state which has a history of significant environmental protection. It is no more likely that the State of Maine will repeal its take protection measures than it is that Congress would repeal the FPA. Existing legal mechanisms must be considered by FERC, especially when they are as recent and comprehensive as those adopted by the State of Maine, when it determines whether adequate levels of protection exist to meet the needs of the FPA.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-14 Opinion noted.

GNP-14

GNP-13

If in fact FERC continues to be concerned that the requirements of Maine law could be changed or altered during the duration of the project license, it is not necessary to impose burdens which are so significant in cost and possibly incapable of compliance in a process which is bound to pit neighbors against each other and invoke a contentious and uncertain eminent domain process. In fact, the Conservation Law Foundation et al (CLF) has previously said that it should be unnecessary for Great Northern to have to secure additional land to establish buffer zones:

"Because Great Northern owns the land surrounding affected waters, FERC need not require the applicant to incur the costs of acquiring land in fee. Rather, FERC should require Great Northern as a condition of the final license to protect all undeveloped shorelands of the Ripogenus and Penobscot Mills project by implementing adequate buffer zones around all riverine and impoundment shorelines of waters utilized in the hydropower system."2

This observation was specifically contained in a request to FERC to condition final licenses upon a shoreland buffer zone which expressly exempts "currently leased lots in effect at the time the applicant filed final applications with FERC and around transmission and generating facilities." Thus, Great Northern, local citizens, officials of the State of Maine and even the intervenors have all expressed the view that it is not necessary to require costly, complex, divisive and perhaps impossible land acquisition burdens to obtain setbacks in buffer zones around the impoundments.

FERC's Projection of Possible Development is not Feasible

The DEIS expresses a concern that the potential residential development in the GNP-15 Penobscot Mills Project area, presumably the shoreline area, is approximately 5,000 dwelling units and in the Ripogenus Project area approximately 700 dwelling units (DEIS, page 4-57). Such development is impossible as a matter of fact, law, practicality and economic reality.

Pronound terms and conditions submitted by American Rivers, et al., May 21, 1993 (pages 34-35).

RESPONSES TO GREAT NORTHERN PAPER. INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-15 The staff revised potential build-out estimates to reflect LURC's subdivision regulations. As stated in the DEIS, the staff acknowledges that these estimates do not account for development limitations such as steep slopes, poor soils, wetlands, or access (see section 4.9.1.1).

129

GNP-14 Cont

As a legal matter under Maine law, any development consisting of more than two lots in a five year period constitutes a subdivision. As FERC has correctly pointed out (DEIS, page 4-57), the vast majority of the Penobscot Mills Project area is zoned P-GP and in the Ripogenus Project area, virtually all of the project boundaries are zoned as either P-AL or P-GP. Subdivisions are prohibited in P-GP or P-AL subdistricts. Thus, as a matter of law, no more than two units every five years may be constructed or an illegal subdivision will result. Even if one considers individual townships bordering the Ripogenus and Penobscot Mills Projects as separate parcels, neither project could have more than 144 single family homes (two every five years for the 30 year license term) constructed on hundreds of miles of shore frontage.³

There are a number of site restrictions, such as poorly drained or shallow soils. insufficient lot depth due to road location, poor or non-existent road access, poor boat access, etc., that would further limit potential development or make other areas more attractive than every "theoretically developable" lot on these impoundments. Over 10 miles of the Ripogenus shoreline is owned by the Bureau of Public Lands which has a policy prohibiting the issuance of new leases for private purposes.

In assessing potential shoreland development. FERC must base any projections of potential development on reasonable scenarios. The most relevant analysis of projected housing demand is a recent study done by LURC in connection with its ongoing review of the comprehensive plan. In a study completed only within the last few months, LURC estimated that the total number of new seasonal units within the <u>entire</u> 10.5 million acres of Maine's unorganized territory would be between 2.600 and 3.000 over the next ten years (see <u>Summary of the Commission's Current Land Use Policies and Their Net Effects</u>, August 1994, page 36). <u>However</u>, since most development under LURC's new Lake Management Classifications is likely to take place on Management Classes 3, 4 and 5, (i.e., areas to which LURC is directing development), the total number of homes likely (even if rezoning were allowed) to be constructed in the Ripogenus Project area is especially minuscule, certainly only a tiny fraction of those estimated by FERC as theoretically possible.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-15

- 12 -

¹ It is true that Great Northern could seek to change the zone of shear areas where subjivisions are prohibited and shearedcally, allow a greater number of single family houses. As noted earlier, however, LURC has specifically deals with the question of subdivisions proposed on land adjacent to lakes and, or Jane 7, 1990, substantially enhanced the normal development approval criteria found in 12 M.R.S.A. § 685-A-8 to further require flushings that there will be no undue adverse impact on water quality, traditional uses in natural and cultural resource values and that the natural character of the shoredard the more class. LURC rules. Chapter 10.13-8.2.

In short, the DEIS greatly overstates the amount of development that would be possible, especially along the Ripogenus impoundment. The strict zoning that already exists makes it inconceivable that the DEIS's estimates can ever be realized. The DEIS offers no evidence to support even a hypothesis that the LURC zoning might be changed. Under these circumstances, there is simply no basis to assume that state law and LURC zoning will not protect the Ripogenus and Penobscot Mills Project's shoreline areas for the life of the new license. Thus, the DEIS's goal of limiting new units for aesthetic purposes is clearly met.

A Shoreline Buffer Zone Within the Project Boundary Would Impair Enforcement Efforts

At present, LURC is legally responsible to enforce its rules governing activities along the Ripogenus Project. If the new license for the Ripogenus Project requires the creation of an expanded shoreline buffer zone, it will become the responsibility of Great Northern and of the Commission to develop and enforce the rules governing acceptable activities within that zone. The DEIS fails to evaluate the negative impact of this transfer of responsibility as well as the likelihood that private enforcement would be impossible.

During the course of this proceeding, no one has credibly suggested that LURC lacks the legal authority to enforce its rules. No one has suggested that the State of Maine lacks the necessary police and judicial authorities to support LURC's enforcement efforts. Indeed, the record demonstrates that ample effective legal authorities are already in place to govern activities in the Project's shoreline zone.

If the Commission were to create a 200 foot shoreline buffer zone, however, enforcement authority would shift to Great Northern and, ultimately, to the Commission itself.4 Unfortunately, neither Great Northern nor the Commission has any effective resources or authority to regulate activities in this zone. At best, Great Northern would have to resort to attempts at land restriction and civil trespass actions to accomplish what LURC can now do directly with the State's police powers.

> Given the Commission's apparent premise in creating a shoreline buffer zone is that LURC's regulations are ineffective and unpredictable, it is unlikely that LURC could or would assume any enforcement authority for the Commission.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-16 See responses GNP-2 and GNP-8.

GNP-15

GNP-16

Moreover, neither Great Northern nor the Commission have staff resources or expertise necessary to police the various activities that will occur on a daily basis in a shoreline buffer zone of this magnitude.s These activities include not only recreation and activities by camp owners or lessees, but also ongoing commercial forestry, a subject that the Commission has, as the DEIS demonstrates, little background in.

"Grandfather Clause", Impacts on Existing Leases

In comments published prior to and during the January 25 hearing, and in public statements on the record, various intervenors and Commission staff itself have stated that existing developments would be unaffected by the proposed buffer zone. These statements ignore Commission precedent and contradict the DEIS's rationale for creating a Commission-controlled buffer zone.

First, Great Northern is unaware of any mechanism by which the Commission could "grandfather" existing development. Indeed, less than a year ago, in <u>East Bay Municipal</u> Utility District, **166** FERC to 61, 199, the Commission stated:

"The Commission does not condone residential development and occupancy on project lands, and in fact requires license applicants to exclude residences unless the underlying property is needed for a project purpose..... Such long-term residency, essentially granting individuals exclusive use of a site, is not consistent with the Commission's policy of maximizing public recreational development. Such development undercuts the rational for expanding the project to create a buffer zone."

In light of these conclusions, it is not clear how the Commission could grandfather existing development. More importantly, however, the willingness to consider grandfathering existing development appears to be inconsistent with the intention to supersede LURC regulation. Thus, although the DEIS concludes that current regulations do not "guarantee long-term protection for the duration of the license" (DEIS, page 4-57), the current level of development, which the Commission now proposes to formalize, reflects the operation of the very rules and regulations the DEIS seeks to replace. If the current level of development is acceptable, however, there is no reason to supplement the regulations that produced it.

Indeed, when is created the standard license article governing permitted land uses within licensed projects, the Commission indicated that it was doing to. In part, because it was devoting too many staff resources to the consideration of routine land use matters. Breads River Authority, 11 FERC § 61.162 at page 61.347 (1990). The creation of a shoreline buffer some in this case would be inconsistent with the provident. The DEIS does not explain this departure from Commission procedent.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-16

Indeed, if the DEIS's true concern is with the remote possibility that LURC regulations might somehow disappear, then the Commission should explore options that deal with this problem, not by creating new problems by expanding the project boundary. As stated at the January 25 hearing, current lease holders have grave concerns about the impact from the Commission's proposal to expand the project boundaries. For example, will this requirement prohibit improvements, repairs or expansions of existing homes or camps? Would the construction of a new dock, garage or shed in conformance with state standards be prohibited? Would new construction on existing leases be prohibited?

The DEIS's Estimated Land Acquisition or Easement Costs are a Fraction of the Actual Costs Which Great Northern Would Incur

The DEIS estimates that the cost to Great Northern of acquiring shoreline in fee or conservation easements is approximately \$1,000 per acre. This figure was derived from an examination of a recent sale of approximately 1.200 acres acquired by Maine's Forest Legacy Program along the Rangeley Lake shoreline (DEIS, page 4-73). The value of the rights acquired in that transaction is not at all comparable to this situation and is inapplicable to land acquisition costs along the Ripogenus impoundment shoreline.

The purchase of a 200 foot strip of land requires that Great Northern purchase only the most valuable part of a parcel, i.e., shore frontage. In the case of the Rangeley transaction, however, much of the land (approximately 94%) was not shore frontage. The \$1,000 per acre appraisal covered a great deal of land, much of which was not shorefront land. In the case of the DEIS's proposal for the Ripogenus impoundment, all of the land would be shorefront land, and thus, the per acre cost would be far higher than the Rangeley Lake price.

To confirm this view. Great Northern contacted the appraisal firm which conducted the Rangeley Lake appraisal. Attached as Appendix A is an appraisal by Thompson Appraisal Company. Inc. indicating that the DEIS's use of a \$1,000 per acre figure for the Ripogenus impoundment shoreline is inaccurate and inappropriate. Their estimate of the fair market value of acquiring simple fee interest for the non Great Northern properties is \$10.9 to \$12.3 million. In addition, Great Northern has contacted one of Maine's leading appraisers, Norman Gosline, to provide an independent assessment of whether the DEIS's estimate of \$1.000 per acre is appropriate. Mr. Gosline concludes that the number is not accurate and the actual figure would be considerably higher (see Appendix A).

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-17 See response GNP-11.

GNP-17

GNP-16

In addition to the views of Mr. Thompson and Mr. Gosline. Great Northern has also estimated the value of the land interests the DEIS would require it to acquire based upon current Maine Bureau of Taxation valuations. According to these valuations, a typical lot with 200 feet of shore frontage is valued at \$16.000 on Chesuncook Lake and \$20,000 on Caribou Lake (both lake basins within the Ripogenus impoundment). Using these figures, the cost to acquire shoreline property not owned by Great Northern around the Ripogenus impoundment is over \$13 million. As well, testimony at the January 25 hearing presented by Mr. Bob Fiske of J. W. Sewall Company indicates that the real cost for acquiring by fee or easement is eight times higher than the estimate included in the DEIS.

In short, whether one uses the estimates of the appraisers which FERC initially cited. leading Maine appraisers, valuations based upon the Maine Bureau of Taxation approach, or comparable local real estate sales as testified to at the public hearing on January 25, the DEIS would actually be imposing a cost of at least \$10 million, rather than \$2 million for Great Northern to purchase a 200 foot fee or easement around Ripogenus impoundment.

GNP-17 Cont

E-134

The option of acquiring an easement as opposed to fee is not one which would result in any significant cost savings to Great Northern. The limitations on land use recommended in the DEIS are such that virtually no rights would remain with the landowner, who would, however, still face the burden of taxation and easement maintenance. Thus, for an easement to accomplish the DEIS's objectives, it would have to provide compensation equivalent to a purchase of a fee interest. Great Northern would also incur significant easement monitoring costs as well, in past due to the nature of the requirements that have been described for this acquisition in the DEIS. For example, it would be very time consuming to monitor and enforce the restrictions on temporary docks along hundreds of miles of shoreline. The specific means by which existing development may be "grandfathered" would have a significant bearing on ongoing, annual monitoring costs for Great Northern.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

Timber Harvesting Practices Do Not Impact Water Quality

It is clear that existing LURC regulations have fully protected water quality in project impoundments. This is recognized in the DEIS itself and by State agencies charged with issuing water quality certifications for the projects. Both LURC and the DEP have issued certificates which concluded that water quality standards are being met in project impoundments.6 In addition, the proposal to create a shoreline buffer disrupts the balance LURC has sought to achieve by recommending some areas as suitable for development and greatly restricting development in other areas like the Chesuncook Lake section of the Ripogenus impoundment.

In support of its conclusion that expanded buffer zones are necessary, the DEIS states that LURC's "50-foot buffer does not provide the recommended width calculated using the commonly accepted Forest Service criteria; therefore, a 50-foot vegetative buffer may not adequately protect aesthetics and water quality depending on existing slope" (DEIS, page 4-59). This conclusion is inconsistent with the DEIS's evaluation of water quality and aesthetics and is based on incorrect, inapplicable or outdated Forest Service criteria. As the DEIS itself notes, "water quality throughout all of Ripogenus and Penobscot Mills Projects waters is generally very good; nearly all waters meet State water quality issues were identified during the scoping process; mercury concentrations in the impoundments, dissolved oxygen (DO) concentration in Dolby Pond, and water quality in Back Channel" (DEIS, page 4-8). Furthermore, the state water quality certificates essentially found no water quality problems in impoundments, let alone problems related to project operations, forestry or other activities of Great Northern.

In short, statements in the DEIS to the effect that timber harvesting practices can degrade water quality by altering temperature, lowering DO concentrations, and increasing concentrations of nitrate and suspended sediments (DEIS, section 4.9.1.2) are purely theoretical and are inconsistent with the actual water quality in the Ripogenus and Penobscot Mills Projects.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-18 Staff determined the recommended alternative and the proposed conservation easements and/or boundary expansion would provide long-term protection of valuable shoreland resources (see sections 3.11.1.1 and 4.9). In the recommended alternative, GNP would be responsible for recreational facilities and potential enhancements or mitigation of recreational resources in association with project operations or license conditions (see section 4.9.3).

Possible DO deficits in the Dolby improvidement have been noted, but there is no indication that these deficits are caused by activities that would be affected by a shoreline buffer zone. Great Northers has agreed to examine whether mill operations have contributed to these deficits, but nothing in the record separate onclara that forestry operations have been the cause.

LURC regulations have been an effective means to limit negative impacts from timber harvesting on water quality, aesthetics and wildlife habitat for the Ripogenus and Penobscot Mills Projects. The LURC standards applicable to timber harvesting near lakes, rivers and streams have remained relatively unchanged for roughly 20 years. It is highly unlikely that these standards will be seriously weakened in the future. In addition, because these standards have been in place for such a long time, any inadequacies in these standards should have been evident in the results of recent water quality monitoring work.

FERC has made an erroneous interpretation of LURC regulations in its statement that the vegetative buffers applicable to timber harvesting are 50 to 100 feet (DEIS, pages 4-58 & 4-59). The width of the protection zone (P-GP) around takes over ten acres (includes all project impoundments) and rivers which drain over 50 square miles (includes all of the West Branch) is 250 feet (Maine LURC Regs., Chapter 10). The regulations which apply within this 250 foot zone are specifically designed to "regulate these areas so that development will not degrade waters, recreation potential, fishery habitat or scenic character".

The requirements which apply to timber harvesting in the P-GP Protection Sub-districts are duplicated below from the LURC Regulations. There are several specifications that apply within 50 feet of the high water mark (b.1. and b.4.), while the others apply between 50 and 250 feet or to the whole 250 feet for the Great Pond Protection Sub-district.

- "b. Timber harvesting operations in P-SL1 and P-GP Protection Sub-districts shall be conducted in the following manner:"
 - Within 50 feet of the normal high water mark, no clearcutting shall be allowed and harvesting operations shall be conducted in such a manner that a well-distributed stand of trees is retained so as to maintain the aesthetic and recreational value and water quality of the area and to reasonably avoid sedimentation of surface waters."
 - At distances greater than 50 feet from the normal high water mark, harvesting activities may not create single openings greater than 14,000 square feet in the forest canopy. In such areas single canopy openings of over 10,000 square feet shall be no closer than 100 feet apart.
 - 3. Harvesting shall not remove, in any ten year period, more than 40 percent of the volume on each acre involved of trees 6 inches in diameter and larger measured at 4 1/2 feet above ground level. Removal of trees less than 6 inches in diameter, measured as above is permitted if otherwise in conformance with these regulations. For the purpose of these standards, volume may be determined as being equivalent to basal area.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-18

- 4. No accumulation of slash shall be left within 50 ft. of the normal high water mark of surface water protected by the P-SLI and P-GP Protection Sub-districts. In such Sub-districts, at distances greater than 50 ft. from the normal high water mark of such waters, all slash larger than 3 inches in diameter shall be disposed of in such a manner that no part thereof extends more than 4 ft. above the ground.
- "c. Except as provided in subsection g of this section, skid trails and other sites, where the operation of machinery used in timber harvesting results in the exposure of mineral soil, shall be located such that an unscarified filter strip of at least the width indicated below is retained between the exposed mineral soil and the normal high water mark of surface water areas:"

Average Slope of Land Between Exposed Mineral Soil & Normal High Water Mark (%)	Width of Strip Between Exposed Mineral Soil & Normal High Water Mark (Feet Along Surface of the Ground)	
0	25	
10	45	
20	65	
30	85	
40	105	
50	125	
60	145	
70	165	

The provisions of this subsection c apply only on a face sloping toward the water; provided, however, no portion of such exposed mineral soil on a back face shall be closer than 25 feet; the provisions of this subsection c do not apply where skid roads cross such waters:

*i. Written notice of all timber harvesting operations shall be given to the Commission prior to the commencement of such activity. Such notice shall conform to the requirements of Section 10.20 of this Chapter and shall state whether or not such operations will be conducted according to the provisions of subsection g of this section; and"

"j. In addition to the foregoing minimum requirements, except as provided for in subsection g, provision shall otherwise be made in conducting timber harvesting operations in order to reasonably avoid sedimentation of surface waters."

It should be clear from the above that LURC's standards require a well managed. forested buffer that extends to 250 feet from the high water mark. Within the first 50 feet, harvesting is more strictly controlled, and within 25 feet or more, depending on slope, soil disturbance is prohibited. The types of provisions included in LURC's regulations are consistent with essentially all of the major research that has examined this issue and which are relevant to this region and situation. In a number of cases, LURC standards go beyond what is recommended.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

E-137

GNP-18 Cont

- 19 -

The DEIS references a "commonly accepted Forest Service criteria" in determining that the minimum buffer width should be 100 feet in which no removal of vegetation is allowed (DEIS, pages 4-58 & 4-59). The reference for this criteria is a 1972 USFS publication which cannot be found at the University of Maine library, USFS offices in Pennsylvania, nor by the consulting firm (ERM) who drafted this part of the DEIS. ERM was only able to locate an abstract of the publication which does not discuss what type of management is recommended within the buffer, nor any background regarding the recommendation.

The USFS has published at least three more current sets of guidelines for erosion and sediment control using buffers for the northeast since the "elusive" 1972 document. The applicable recommendations from these are duplicated below:

> "Woodlands of the Northeast: Erosion and Sediment Control Guides". USFS 1977:

"In order to maintain water quality, eroding material must be trapped between disturbed areas and stream system. This can be accomplished by leaving essentially undisturbed buffer strips between disturbed areas (road, construction sites, etc.) and water courses. Water from the disturbed areas should not only enter a buffer or filter strip, but should be spread in the buffer to be effective. This slows the velocity of the water so that the sediment load may be dropped and the water will soak into the soil. This may require placement of brush, rocks, and logs to supplement the natural vegetation and litter, in order for the filter strip to be effective. The following shows the recommended widths for filter strips on various slopes:"

Width of Slope of Land Between Road & Stream (%)	Fiker Strip for Common Logging Areas (Feet)	Width of Filter Strip in Municipal Watersheds and Critical Areas (Feet)
0	25	50
10	45	90
20	65	130
30	85	170
40	105	210
50	125	250
60	145	290
70	165	330
80	185	370
90	205	410
100	225	450

The Commission should note that the width of the filter strip for common logging areas recommended in the previous publication is exactly the same as LURC's requirement for the width of an unscarified filter strip.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DELS

GNP-18 Cont

 "Riparian Forest Buffers: Function and Design for Protection and Enhancement of Water Resources". USFS 1991:

This publication recommends two forested zones designed to serve as streamside buffers for cropland or pasture land. Zone 1 is immediately adjacent to the stream and Zone 2 is adjacent to Zone 1 away from the stream.

"Zone 1 (15 feet wide) - Undisturbed Forest. Tree removal is generally not permitted in this zone."

"Zone 2 (60 feet wide) - Managed Forest. Periodic harvesting is necessary in Zone 2 to remove nutrients sequestered in tree stems and branches and to maintain nutrient uptake through vigorous tree growth."

 "Whole-tree Clearcutting in New England: Manager's Guide to Impacts on Soils. Streams and Regeneration". USFS 1993.

"Prevention of erosion is the main approach to protecting water quality . . . Some additional guidelines more specific to streams are given below."

"Consider leaving a buffer strip of living trees along streams to trap sediment and shade the channel. We recommend a variable width of 15 to 30 m on each side of the stream. Actual widths depend upon factors such as: stream gradient (the steeper the gradient, the wider the buffer); slope gradient and length to stream channel (steep slopes and short distances will require wider buffers); and soil characteristics (clay or highly eradible soils should have wider buffers)."

These and other studies consistently indicate that vegetated buffer strips, and properly constructed roads and main skid trails. as required by LURC, are entirely consistent with USFS guidelines. Stream turbidity, temperature and chemistry can be maintained at close to pre-cut levels by applying procedures currently in place in existing regulations and practices. Even where buffer strips have not been left adjacent to small streams, dissolved nutrients in the streams generally do not exceed standards established for the protection of water use and aquatic life. In addition, surface water nutrient levels return to pre-cut levels within three to four years due to the rapid re-growth of vegetation.

Few studies have specifically examined the differences between managed and unmanaged buffers. Lowrance et al (1984) recommended that "the selective cutting of mature trees for timber or fuel wood will maintain the net annual nutrient uptake by vegetation." and that "proper streamside forest management requires both periodic harvest of trees to maintain nutrient uptake and minimum disturbance of soil and drainage conditions." This research was conducted on riparian buffers and agricultural fields in the southeastern U.S. and therefore represents a more demanding situation in terms of nutrient removal required by the buffer than for buffers for general forest management areas.

- 21 -
Corbett et al (1978), in a review of data from experimental watersheds in the northeastern U.S. (NH. PA, WV), concluded that "buffer strips, in which only light selection cutting is allowed, will help minimize sedimentation as well as nutrient leaching and stream temperature increase." Charles Gadzik, former LURC Commissioner, in his testimony at the January 25 hearing, also stated his belief that allowing some timber harvesting to take place within the buffer helped to maintain a healthy condition of the forest stands in that area.

Finally, in its conclusion, FERC assumed that slopes around Ripogenus impoundment average 12-15% (DEIS, page 4-58). Most of the land within 250 feet of the water that is operable for timber harvesting has very minor slopes, i.e., under 5%. Measurements on U.S. Geologic Survey topographic maps indicate that only 20% of the shoreline contains slopes in excess of 9%. A portion of these steep sections are cliffs (near Ripogenus Dam) and are not relevant to timber harvesting setbacks since they are inoperable and contain little to no timber volumes.

The real proof of the effectiveness of current regulations and practices is in the measurement of water quality parameters. The Ripogenus impoundment system has demonstrated excellent water quality throughout the study years and meets all numerical water quality criteria and trophic state assessments for its assigned classification of GPA.

Evidence supporting the exceptional water quality in the Ripogenus impoundment system was derived from extensive monitoring programs conducted from 1981 through 1983 and from 1986 through 1988. For a FERC license application submitted in March of 1984. Great Northern examined 32 water quality parameters, including nutrients, metals and pesticides. in Ripogenus Lake (FERC No. 3779, Volume VII, Exhibit E, Report on Water Use and Quality). A 48-hour diel survey was also conducted in August 1981 in Ripogenus Lake. Major conclusions concerning the Ripogenus impoundment system drawn from this extensive database included:

- 1. Low alkalinity, typical of weakly buffered New England lakes,
- Low conductivity, turbidity, biological oxygen demand, chloride, sulfate, metals and indicator bacteria.
- Absence of any target pesticides (ones used for timber management) in the water column.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-18

Cont

- 4. Low nutrient concentrations and low to moderate primary (algal) productivity; comparable to other Maine lakes such as Sebago and Rangeley Lakes which are utilized as major recreational water bodies.
- No unusual or unexplainable spatial or seasonal trends in water chemistry or thermal stratification patterns.
- 6. Low variability of water quality parameters during the diel survey.
- Low variability of any parameter measured among the three lake stations monitored.
- Moderate color typical of many northern New England lakes, especially those in the Penobscot River basin.

The 1986-1988 database (having multiple sampling locations within Ripogenus impoundment sampled on 14 occasions) further supports the earlier conclusions that this water body meets the criteria for its classification (GPA) and for supporting all designated uses of that classification. In the specific issue of sedimentation and water column turbidity, a comparison was made between the Ripogenus impoundment and 43 northern Maine drainage lakes or reservoirs surveyed by the U.S. Environmental Protection Agency (EPA) in 1986, including five within the Penobscot River drainage basin. A review of this data provides quantitative evidence that turbidity and secchi disc depth in the Ripogenus impoundment are typical for northern Maine lakes. Thus, speculation that timber harvesting practices are resulting in adverse water quality conditions in the Ripogenus impoundment is not supported by the data and is contradicted by comparisons with monitoring data from less developed watersheds.

Indeed, the DEIS reaches this conclusion by stating that the "staff maintains that forestry practices have been a part of the landscape in northern Maine for more than 100 years, and assuming that proper buffers and regeneration are provided, do not represent a significant aesthetic impact" (DEIS, page 4-66). In addition, the DEIS rejected the Department of Interior's (DOI's) request that the Commission require Great Northern to develop a watershed management plan. The DEIS's analysis is directly applicable here (DEIS, pages 5-23 and 5-24).

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-18

Cont

- 23 -

With respect to the Ripogenus impoundment, LURC, with review of the Maine DEP. Bureau of Land and Water Quality, concluded that the impoundment is attaining a stable trophic state (see Ripogenus Project Water Quality Certificate, section 11), fish populations are natural and self-sustaining in all relevant cases, and that fishing opportunities, according to the DIFW, are "among the finest and the most important in the State of Maine" (Ripogenus Project Water Quality Certificate, Section 24.A). LURC found that recreational uses are abundant and that fishing is the principal recreational use of the impoundment (Ripogenus Project Water Quality Certificate, Section 26.A). LURC concluded that continued operation of the project, as proposed by Great Northern with minor enhancements and changes, will result in water quality standards being met in all respects, including suitability for recreation, including fishing, and satisfaction of all criteria for attainment of designated uses of aquatic habitat and other aquatic life in the impoundment.

Similarly, the DEP found the project operation as proposed would meet all water quality standards for the Millinocket Lake impoundment. Quakish Lake and Ferguson Pond. as well as for the Dolby and East Millinocket impoundments (Millinocket Lake Storage Dam Water Quality Certificate, page 13: Penobscot Mills Project Water Quality Certificate, page 15). Since there are no present water quality problems in the project impoundments. additional setbacks and buffer zones are not necessary to protect water quality.

Exercise of FERC's Eminent Domain Power to Extend Project Boundaries is Uncertain, Will Clearly Result in Contention Between Great Northern, its Neighbors, and the State of Maine, and is Unnecessary to Accomplish its Objectives

By its proposal that Great Northern acquire the shorefront property of others along many miles of Ripogenus impoundment, the DEIS suggests a condition which may be impossible to comply with or, at the very least, is far more costly than estimated and clearly likely to result in litigation between Great Northern and its neighbors, including the State of Maine. This tremendous economic penalty which would be imposed on Great Northern assumes that the eminent domain power is available to acquire the land. The exercise of the eminent domain power would clearly be required given the testimony of applicable landowners at the public hearing.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-19 See responses GNP-2, GNP-8, and GNP-11.

GNP-19

GNP-18 Cont

During the January 25 bearing, landowners and the State of Maine made it clear that they will not willingly sell or make their land available to Great Northern, because they do not believe that the DELS's approach is necessary, fair or appropriate to meet the Commission's objectives legitimately. Thus, the DEIS makes it inevitable that Great Northern will have to resort to eminent domain proceedings pursuant to Section 21 of the FPA. Therefore, this proposed condition would essentially guarantee that landowners would have no choice but to challenge the license on the grounds of misuse of FERC's powers and the potential inapplicability of its eminent domain authority for Great Northern's use. This would increase the cost of any transaction and render its outcome totally uncertain.

No issue has received greater public attention since the DEIS was issued than the question whether the Commission can or should require Great Northern to acquire the shoreline buffer zone by use of the power of eminent domain pursuant to Section 21 of the FPA (16 U.S.C. § 814). Equal concern has been focused on the future of the recreational camps located within the shoreline zone identified in the DEIS.

If the Commission believes that Great Northern must gain ownership of the entire shoreline zone, a conclusion that Great Northern strenuously disputes, it is essential for the Commission also to explain how Great Northern is to accomplish this objective. If, despite ⁴ the DEIS's failure to justify adequately the need to create a shoreline buffer zone, the Commission still believes that the power of eminent domain will be applicable, it should clearly state whether and how that power will be available. This will, at least, enable Great Northern to begin to develop a more accurate assessment of the complete cost of acquiring the shoreline zone.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

2

•

THE SCOPE OF THE DEIS

Cumulative Impact Analysis

During the scoping process, the Commission rejected requests that its DEIS analysis encompass Great Northern's storage projects upstream of the Ripogenus Project. Great Northern supports this conclusion and believes that the FEIS should as fully discuss the basis for excluding upstream impoundments as the DEIS did for the downstream projects.

The exclusion of consideration of different upstream projects is well supported by Commission NEPA policy which provides: (1) absent extraordinary circumstances, the NEPA analysis should extend from the upstream limit of the Ripogenus impoundment to the tailwater of the East Millinocket development, and (2) the Commission will include a reopener in the licenses for the Ripogenus and Penobscot Mills Projects to permit the Commission to address issues raised in the relicensing of the Great Northern Storage Project. FERC Project No. 2634.

GNP-20

E-145

Given the circumstances of this relicensing, no other policy would make sense or be appropriate. This approach allows timely relicensing of the Ripogenus and Penobscot Mills Projects without limiting the Commission's ability to make minor modifications to these licenses which may be suggested as a result of the subsequent relicensing of FERC Project No. 2634. More importantly. Great Northern also believes that there are no extraordinary circumstances that would justify, at this time, the delay involved in further analysis of any upstream projects, whether licensed or unlicensed. As Commission staff itself noted at the Scoping Sessions, the Commission has made no determination as to the jurisdictional status of the unlicensed upstream impoundments and, without such a determination, has no authority over those projects. Because FERC Project No. 2634 will not be relicensed until 2000, any attempt to "synchronize" that relicensing with the pending proceedings will cause an excessive and unjustifiable delay in the pending relicensing. Moreover, any significant impacts actually caused by the operation of the Ripogenus and Penobscot Mills Projects have been fully discussed in their respective applications. There is absolutely no basis to support a contention that these Projects have any meaningful upstream impacts.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-20 We addressed the issue of including analysis of the upstream projects in Appendix D. We will recommend inclusion of a reopener article in the orders for the Ripogenus and Penobscot Mills projects to ensure that any currently unidentified relationships between these projects and the upstream licensed and unlicensed storage projects will be addressed at the appropriate time.

The Commission recently restated its policy with respect to upstream projects in the Final Environmental Impact Statement issued by the Office of Hydropower Licensing for the Upper Androscoggin River Basin Hydroelectric Projects. May 1993 ("Androscoggin FEIS"). This FEIS states two reasons why the Commission would not expand its cumulative impact assessment to upstream projects. Both of the Commission's reasons are directly applicable to Great Northern's projects.

First, the detailed information gathering and studies needed for a NEPA analysis are:

"[I]nextricably intertwined with license processing. If the licensing process for projects is not fairly closely synchronized, enhancement measures or developmental benefits could be unreasonably delayed for projects closest to license issuance.... We believe that we could not justify the 3 to 5 year delay in relicensing the projects considered in this EIS and the improvements to environmental resources that would result" (see Androscoggin FEIS, p. 1, 4).

in this case, there are only two projects (Ripogenus and Penobscot Mills) pending relicensing. The current license for the Great Northern Storage Project does not expire until the year 2000 and the license for the downstream Mattaceunk Project does not expire until 2018. As in the case of the Androscoggin FEIS, the remaining upstream projects are not licensed, and it would take at least three to five more years before they could be ready for environmental analysis.

Second, the Androscoggin FEIS notes that any cumulative effects analysis requires the Commission to consider the distribution of the target resource:

"The interaction of water uses and projects to cumulatively affect target resources diminishes with distance between them" (id at 1-5).

This is equally true in the case of Great Northern where the distances to the upstream projects are comparable to the Androscoggin case. Great Northern's hydro system includes ten storage ponds upstream of the Ripogenus impoundment. Four of these ponds - Seboomook. Canada Falls. Ragged. and Caucomgomoc Lakes - have been ticensed by FERC as the Great Northern Storage Project, the license for which expires in April 2000 and will be subject to review in connection with that relicensing. Four other unticensed impoundments - Penobscot. Long. Dole, and Loon - are located upstream of and drain into the Storage Project. The last two unlicensed impoundments. Umbazooksus and Harrington Lakes, are tributary to the Ripogenus impoundment.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-20

Cont

The capacity of the upstream unlicensed ponds and their role in the Great Northern hydro system is insignificant by comparison to the Ripogenus impoundment. Together, these six storage ponds have the maximum capacity to store only 3% of the system's water. Umbazooksus and Harrington Lakes, the only two that drain directly into Ripogenus, can only hold 1% of the system storage, even if full. By contrast, Ripogenus holds 50% of the total system's storage. Furthermore, flows from Harrington Lake are operated by agreement with the DIFW to meet its management goals for landlocked salmon.

In this respect, the argument by the CLF, that the storage ponds are affected by the operation of the Ripogenus Project, is factually incorrect. In fact, at the request of the DIFW, the WUP was designed so that it would not change the operation of the upstream ponds in any way. Upon implementation of the WUP, the storage ponds will continue to be operated as they always have been and in consultation with DIFW in order to further its management goals. Because nothing in the WUP will change the historic mode of operation or foreclose future changes that may be desired by the appropriate agencies, the relicensing of the Ripogenus and Penobscot Mills Projects will simply not affect the upstream ponds.

During the original licensing proceeding for the Storage Project. FERC and Great Northern agreed that only Seboomook, Ragged. Caucomgomoc and Canada Falls Lakes would be licensed as headwater storage reservoirs. It was also agreed that the other dams and lakes on tributaries to the West Branch of the Penobscot that were owned by Great Northern (Dole and Long ponds, Harrington, Loon, Penobscot, Rainbow, Nesowadnehunk and Umbazooksus Lakes) would not be licensed since their impact on Great Northern's hydro system was inconsequential. Nevertheless, by letter dated January 30, 1991, Great Northern responded to FERC's inquiries, prompted by the DEP, and committed to including these projects in the Storage Project if data collected during the relicensing of that project show that the ponds contribute in any meaningful way to net generation downstream:

"Great Northern respectfully submits that, upon receipt, during first-stage consultation, of such documentation as DEP possesses, it will examine this issue as part of the relicensing activities for the Storage Project. If data collected during the relicensing supports DEP's position that the ponds contribute to net generation downstream. Great Northern will include the ponds in the application for new license. Relicensing activities for Project No. 2634 will commence within two to five years."

Moreover, FERC has already rejected requests from CLF et al that is study the upstream ponds during the Ripogenus relicensing

"We agree that the Commission could decide to license these reservoirs as a separate storage project. If the Commission does decide that they should be licensed, they would be subject to a separate license. Therefore, fish and wildlife issues and impacts, as they relate to the management of the upstream reservoirs would be addressed under another licensing review" (FERC letter dated June 8, 1992 to G.E. Beckett, U.S. Fish and Wildlife Service (USFWS)).

FERC's decision was entirely appropriate. Investigation of the unlicensed ponds during the Ripogenus relicensing would be misplaced and futile.

Moreover, the Director of the Office of Hydropower Licensing (Director-OHL) has taken the position that there must be a nexus between a project impact and a study request. His refusal to require Alabama Power Company (APCO) to conduct a habitat-based instream flow study, two projects upstream of APCO's Yates and Thurlow Projects, was based on the grounds that a nexus is required between a project impact and a study request:

"Section 10(j) specifies that fish and wildlife conditions placed in licenses are for the purpose of protecting, mitigating damages to, and enhancing fish and wildlife affected by the project: there must be a nexus therefore between an agency's study request and a project's impact on fish and wildlife resources.

The project assessment area, as defined by the agencies, includes the Martin and Harris Projects (FERC Nos. 349 and 2628), which are located upstream of and well beyond the area directly affected by the operations at the Yates and Thurlow Projects. The only section of the Tallapoosa River that could be conceivably affected by the relicensing of the projects would be located downstream of the Thurlow Dam.

Based on the facts in this case. I must conclude that APCO has no obligation to undertake studies in areas upstream of the Yates and Thurlow Projects (letter of Fred Springer. Director-OHL. to Mr. John E. Dorsett, Alabama Power Company at 2 (Alabama Power Company, Project Nos, 2407 and 2408), dated May 10, 1990)."

During the NEPA process in this relicensing, no one has identified any upstream resource that is actually affected by the operation of the Ripogenus or Penobscot Mills Projects. Nor has anyone identified a way in which any proposed license term or condition would preclude any future options concerning the upstream projects. Indeed, the DEIS specifically concludes that it is not necessary to include the upstream storage plans in the WUP (DEIS, page D-27).

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-20

Cont

Finally, as previously noted. Great Northern has specifically acknowledged that, despite the lack of upstream impacts, this is an appropriate case for inclusion of a "reopener" in the new licenses to ensure that any currently unidentified relationships between these projects and the upstream storage impoundments can be addressed by the Commission in any future licensings or relicensings. Under these circumstances, the Commission's determination not to include upstream storage ponds in the scope of the DEIS was correct. The FEIS should explain the Commission's complete reasoning and justification, incorporating the analysis set forth above.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-20 Cont

.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

3

NEED FOR POWER

Background

Since its inception, one of the central issues in this proceeding has been Great Northern's need for low cost hydropower produced by the Ripogenus and Penobscot Mills Projects. In fact, it has often been suggested that Great Northern has overstated its need for low cost power. This is entirely inaccurate and the DEIS has quite properly rejected this argument. More importantly, during the last several years Great Northern has provided the Commission an enormous amount of information on energy conversation and utilization. However, the DEIS does not discuss much of the data that supports its conclusion regarding Great Northern's need for power. Great Northern believes that the FEIS should more fully reflect the record in this critical area.

Great Northern's Competitive Position

As noted repeatedly in various sections of the DEIS. "Great Northern's presence in certain markets and its competitive position in general, therefore, depends upon the availability of a reliable source of inexpensive power". This point is as valid today as it was when the original application was submitted to FERC in December, 1991. Great Northern's relative competitive position remains unchanged from years earlier, and hence, the need for a reliable source of inexpensive power is as critical today as it was then.

During the last four years in particular. the paper industry has suffered a downturn which is unparalleled in recent history. In the uncoated groundwood papers segment alone. 12 new newsprint machines, with a combined capacity of 2.25 metric million tons, were commissioned during the period from 1989-1991. This new capacity came on-stream during a very lengthy economic recession in the U.S. causing a massive, multi-year imbalance in supply and demand. The downturn itself caused 18 older and less cost-effective machines with a capacity of 1.26 metric million tons to be shut down, many of them on a permanent basis. Additionally, another 700,000 metric tons of newsprint capacity was in the process of being converted to uncoated groundwood specialty papers during the 1991-1994 period. The net effect of all these actions was to add a significant amount of cost-effective production in the newsprint and uncoated groundwood specialty markets in which Great Northern competes.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-21 The staff agrees that GNP's need for low-cost hydroelectric

power has not diminished in recent years, even as improvements have been made at its two mills. These improvements enable GNP to use electric power more efficiently (i.e., fewer kWh required per unit of output). The staff finds no conclusive evidence that either the Applicant's Proposal or either version of Alternative 2 would adversely affect the competitive position of GNP's mills because these alternatives would produce only small increases in annual power costs, and small declines in annual power production. Ultimately, these small changes would not translate into a significant increase in the cost of production at the two mills. Finally, the staff concludes that the negative annual net benefits under Alternative 1 would be large enough to adversely affect the competitiveness of the two mills.

Additionally, during this same time period, several new coated free sheet machines were commissioned which created a very turbulent market for all coated grades, including lightweight coated groundwood papers which Great Northern also manufactures. Coated paper imports were also at all time high levels, especially imports which were, in 1993, nearly twice as high as any recent year.

This excess supply of paper capacity, coupled with the weak economy during the 1989-1993 period, caused industry operating rates (expressed as a percent of installed capacity) to dip into the mid 80 to low 90 percent range. More normal rates would be in the range of 93-96%. As a consequence, the financial performance levels for the U.S. Paper and Allied Products Industry fell to historic lows. The industry's return on net worth and total capital in 1992, for example, fell to 2.5% and 2.4%, respectively, the lowest levels since 1960. During this same period. Bowater Incorporated, parent company of Great Northerm, reported losses in operating income for 10 consecutive quarters.

Simultaneously with these unprecedented markets swings and losses in operating income. Great Northern installed a new \$62 million wastepaper recycling plant in order to meet customer requirements for recycled papers. This plant was started up in May of 1993, just ahead of a rapid upswing in wastepaper prices. In the last 12 months alone, wastepaper prices have increased nearly 200%. An additional commitment by the company to hold Great Northern's relative competitive position, in the face of these increases, is demonstrated by the investment of nearly \$170 million at Great Northern in the midst of the industry's worst recession.

Great Northern's relative competitive position. in fact, has not improved over the last four years. This installation of new low cost manufacturing capacity in the industry is one of the primary reasons. Actions by the company, however, such as retirement of circa 1920 equipment in the Millinocket grinder room, have not lessened the company's need for low cost hydropower. These actions have only changed the way which power is used internally. For example, CLF incorrectly alleges, both in an August 1994 request for additional information and in its January 25 testimony, that Great Northern has made major changes in operations which have a direct and substantial impact on these proceedings. The recent changes in company operations have not lessened the company's need to fully utilize the hydroelectric system. In fact, the closure of the antiquated grinder room at the West Operation did not change Great Northern's energy balance or reduce the need for power.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-21

Cont

Raw wood supplies for the West Operation are now ground at the recently modernized East Operation grinding facility: energy needs for wood pulping, which as a whole are unchanged, have simply shifted from the West to the East Operation. More importantly, the closure of antiquated operations by the company is part of a well-considered effort to utilize and maximize all available resources as effectively as possible to sustain Great Northern's existence. The changes in operations at the time were viewed as absolutely essential to <u>retain</u> Great Northern's relative competitive position in the face of a depressed and unparalleled marketing environment. A key component of those efforts was the continued use of all available hydropower.

GNP-21 Cont

In summary, for the reasons noted above. Great Northern fully supports the DEIS's conclusion that Great Northern's continued presence in certain markets and its continued competitive viability depend upon the availability of a reliable source of lower cost hydropower.

Energy Conservation

As amended, the FPA requires the Commission to consider "energy conservation" when it determines whether, and on what terms, to issue a new license. The terms of the Act are general: they are not limited to <u>electrical</u> energy conservation. Consequently, the DEIS's rejection of electrical energy conservation as an alternative to hydropower in this case is correct and is fully supported by evidence in the record although not cited in the DEIS.

GNP-22 Furthermore. in its DEIS for the lower Penobscot River Basin. FERC clearly stated its position that energy conservation is a "nonexclusionary" alternative to hydropower. for the reason that it is an energy source which does not burn fossil fuels.

"Nonexclusionary energy sources are sources that would be constructed or implemented in addition to hydropower rather than replace hydropower, and vice versa, because all such energy sources have low marginal costs and would be used to displace higher cost fossil fueled generation rather than to displace each other in actual practice. Nonexclusionary resources, therefore, are not reasonable alternatives to each other, and we eliminate them from further analysis." (FERC, Lower Penobscot River Basin Draft Environmental Impact Statement, p. 2-25).

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-22 The staff agrees that there are two issues: (1) using conservation measures to increase the efficiency of use of electric power (i.e., using fewer kWh per unit of output); and (2) increasing total demand for electric power (i.e., more kWh used per year) because of the installation of production technologies that use electric power more efficiently. The staff agrees with GNP that, even with aggressive conservation measures, total annual use of electric power is likely to increase. This would increase their reliance on its lowest-cost source of electric power, which is hydropower. The staff agrees that GNP has many market-driven incentives to maximize its savings through conservation and that its need for low-cost hydropower is not eliminated through use of conservation.

Although Great Northern agrees that conservation is not an alternative or replacement for hydropower as an energy source for its mills. Great Northern does recognize that energy conservation could, theoretically, be less expensive than other sources of replacement power. As a result, Great Northern carefully analyzed the actual potential for energy conservation in its mills and determined that energy conservation measures at best would only have the potential to offset a small part of the increase in demand for power which would occur with modernization of equipment. Energy conservation would not reduce Great Northern's need for hydropower

Even if it were possible for the mills to reduce electricity demand, hydroelectricity is still the lowest cost source of power and must be maximized over more costly fossil fuel fired and purchased power in order for Great Northern to remain competitive. In fact, however, the record demonstrates that conservation measures which have been and could be implemented cannot offset the increased energy demand that will result from modernizing equipment to produce higher quality products required by Great Northern's customers. In addition, the ongoing need for steam conservation will reduce cogeneration potential and further increase the need for hydropower. The net result is that electrical conservation, which Great Northern is vigorously pursuing, is needed merely to offset significantly increased demand and cannot reduce the need for hydropower (see Penobscot Mills and Ripogenus Projects, Volume XIV, pages 175-190).

As stated in its license applications and in subsequent filings with the Commission, Great Northern has worked continuously to improve overall - thermal and electrical - energy efficiency at its mills. Great Northern's conservation efforts, both associated and unrelated to mill modernizations, are completely described in Sections 1.3 and 2.3.2.3 of Exhibit H, and in Volume XIV of the Application.

In short, the record in this proceeding demonstrates that Great Northern has fully analyzed the availability of further energy conservation measures and that energy conservation will not materially reduce Great Northern's cost of replacement power. In fact, as the DEIS correctly notes, electrical energy conservation cannot keep pace with the loss of cogeneration through steam conservation and can only partially offset increased energy demands that flow from improving pulp and paper quality to meet market requirements. In sum, the FEIS should reflect, as the record demonstrates, that the Commission has given equal consideration to energy conservation and has correctly determined that it does not affect Great Northern's need for hydropower in this case.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-22

Cont

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

COMMENTS FROM GREAT NORTHERN PAPER, INC. ON UPPER PENOBSCOT RIVER BASIN DEIS

4

WATER USE PLAN AND FLOWS

Water Availability for the Back Channel

Great Northern concurs with and supports the conclusion of the DEIS that flows in the back channel should not be required. Simply put, water in the back channel would provide virtually no environmental benefits and would only serve to divert water away from more important uses. Because this issue has been so frequently argued by the intervenors. Great Northern does believe that the FEIS should more clearly articulate the many strong environmental reasons which support FERC's conclusion. The FEIS should also eliminate the confusion between FERC's correct conclusion (DEIS, pages 4-7 and 4-8) that "during dry years, not all enhancements could be satisfied simultaneously" when flows of 165 or 350 cfs are discharged to the back channel, and the incorrect and contradictory conclusion that sufficient water is available under all alternatives with some minor effects on other water use constraints (DEIS, page 5-7). In fact, water used to maintain back channel minimum flows will adversely affect other enhancement goals established in the WUP.

GNP-23

m

The record clearly demonstrates the significant negative effects of mandated flows in the back channel. The most significant fisheries consequence of discharging flows to the back channel is the inability to maintain salmon incubation flows on the West Branch in late winter and early spring during dry years (DEIS. Appendix, Figures D-12 and D-16). Also, lake trout spawning and incubation levels cannot be maintained on North Twin impoundment during the driest (worst-case) years if flows are discharged to the back channel (DEIS, Appendix, Figures D-13 and D-17). For the 15 year period of record on which the WUP is based, water availability was below the normal range (dry) 21 percent of the time (see Penobscot Mills Project. Volume VII. Appendix E2 V 5.0). Back channel flows could result in losses of salmon eggs 21 percent of the time from the West Branch's exceptional fishery. This is significant and certainly not a minor effect, especially when one considers that these losses would occur during an attempt to establish an unsustainable back channel salmon fishery (see Fisheries. Section 6). Furthermore, since DIFW, the resource agency with management responsibility in these waters, requested that Great Northern give West Branch salmon and then North Twin lake trout the highest priority in the WUP modeling. Great Northern's proposal for leakage flow only in the back channel is consistent with DIFW's management goals.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOESCOT RIVER BASIN DEIS

GNP-23 We have clarified the apparent confusion between DEIS statements on pages 4-7, 4-8, and 5-7 that not all enhancements could be satisfied simultaneously if flows are provided to the Back Channel (BC). In Appendix D, we added discussion of the specifics of the effects of BC flows with regard to other objectives and more fully explained the assumption that the BC flow takes priority in the model runs conducted by GNP, even though it does not have to have first priority in actual operation. We also state that the model is simulating conditions after the fact, which we agree is much more feasible than actually attempting to use the model predictively.

FERC's comments in the DEIS also seem unaware of the need to manage the water actually available at a specified time rather than analyzing the availability of water after the fact. The FEIS must recognize the fact that Great Northern will be required to predict future water availability when the hydro system is managed under the constraints of the WUP. Once a West Branch spawning incubation flow. North Twin impoundment recreation level, or lake trout spawning level is selected. Great Northern is locked into that operational mode for many months. A significant drop in water availability can make maintenance of these flows or levels difficult or impossible. The FEIS should reflect the fact that since the West Branch and North Twin impoundment will likely be the first to be impacted, higher minimum flows at North Twin Dam (to provide back channet flows) would greatly reduce the available margin of error and can only exacerbate the problem of changing and predicting water availability and Great Northern's corresponding ability to maintain recreational and fisheries flows and water levels.

다 GNP-23 --- Cont

The 1984 water year provides an example of the difficulty in predicting water availability which must be recognized in the FEIS. In late May and early June of 1984, the volume of available water at Ripogenus Dam rose to nearly 47 billion cubic feet (a high for the 15 year period of record) and daily West Branch flows downstream of McKay Station peaked at nearly 15,000 cfs. However, by the end of December, water availability had dropped to under 15 billion cubic feet. Water conditions had changed from "wet" to "dry" in the same calendar year. Under the WUP, a spawning flow would have been established in October of 1984 while conditions were still "normal". With the WUP proposed by Great Northern, this spawning flow and the resulting subsequent incubation flows would have been set under an assumption that water availability would be normal through the winter. As can be seen from modeling for the 1984 water year (see Ripogenus Project, Volume XV, Additional Information Request No. 1). Great Northern is confident that sufficient water would have been available under the WUP to maintain West Branch incubation flows. North Twin impoundment levels, and minimum flows at North Twin through the winter, even with the sudden change in water availability. However, when minimum flows at North Twin are increased to discharge water (165 or 350 cfs) to the back channel, water availability is not sufficient to maintain these flows and elevations (see Penobscot Mills and Ripogenus Projects, Supplement to Volume XV, Additional Information Request No. 3).

The setting of the spawning and corresponding incubation flows will always be a "balancing act": the flows must be low enough to force West Branch salmon to spawn in deeper water, but high enough to maintain both the impoundment level at North Twin for lake trout egg incubation and flows at North Twin Dam necessary to pass the minimum flow at Millinocket. Since most of the winter inflow to the North Twin impoundment is the result of outflow from Ripogenus ("natural" winter inflow is minimal), water availability is not sufficient during dry years to discharge water to the back channel. Higher minimum flow requirements at North Twin Dam in turn require higher salmon incubation flows on the West Branch (to maintain North Twin impoundment levels for lake trout egg incubation) for an extended period (November through early June) thereby increasing the chances that water supplies will be exhausted.

Water availability is simply not sufficient during dry periods to discharge water to the back channet and satisfy all WUP enhancements simultaneously. The very questionable benefits derived from spreading the available water this thin are outweighed by the likely negative environmental consequences to other significant and attainable benefits proposed in the WUP. Perhaps even more importantly. FERC should recognize that Great Northern would be in violation with the conditions of the state water quality certificates if the WUP enhancements are not satisfied. And since FERC agrees that "during dry years, not all enhancements could be satisfied simultaneously" (DEIS, pages 4-7 and 4-8), it follows that Great Northern would inevitably be faced with future non-compliance since these certificates require Great Northern to maintain West Branch incubation flows for salmon and North Twin lake levels for incubating take trout in addition to other WUP enhancements. Great Northern's ability to meet these conditions would be threatened with a mandated discharge to the back channel.

In summary, Great Northern supports FERC's decision to maintain current leakage flow in the back channel. However, the real basis for this decision goes beyond the cost factors mentioned in the DEIS to the ability of Great Northern to satisfy the WUP enhancements and maintain compliance with the state water quality certificates.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

Minimum Flow Requirements at North Twin

In the DEIS the Commission also stated that "(1)he modeling results probably overestimate the outflow required from North Twin because Great Northern assumed that 2,610 cfs would be needed to provide 2,000 cfs at Millinocket. 350 cfs to Back Channel, and an additional buffer of 260 cfs to account for flow regulation, gate setting, and control equipment sensitivities". The DEIS provides no basis for this statement although the issue is likely moot so long as the FEIS continues to reject any requirement for water in the back channel. Nevertheless, it is important that the Commission understand that there is a very sound basis for Great Northern's conclusion that at least a 10% buffer should be available to ensure compliance with a Commission-established absolute minimum flow requirement. As the party responsible for compliance with the terms and conditions of the new license, Great Northern believes that the FEIS should be based upon realistic assumptions regarding the actions a licensee must take to assure compliance.

GNP-24

As noted in the WUP and required in the water quality certificates, the minimum flow of 2.000 cfs that must be maintained at Millinocket is not an average flow (see Penobscot Mills Project, Volume VII, pages 57-58 of the WUP). It is an instantaneous minimum that must be maintained at all times. In order to assure that this minimum flow is maintained at all times. Great Northern must release more than 2.000 cfs at North Twin, which is the control point for Millinocket. Moreover, Great Northern must also account for flow fluctuations between North Twin, Stone Dam. Ferguson Pond, and Millinocket, as well as fluctuations resulting from load shifts at the West and East Operations, gate leakage at Stone Dam, and the inevitable variations in sensitivities between the computer sensing equipment at North Twin and Millinocket.⁷

Historical flow records in the application clearly demonstrate that a minimum 10% safety factor or buffer is necessary to maintain the 2.000 cfs minimum instantaneous flow at Millinocket. These records show that normal fluctuations from a 2.200 cfs target flow can cause the flow at Millinocket to approach the 2.000 cfs minimum. If a higher instantaneous minimum requirement is established by putting flow down the back channel, the buffer would in fact need to be increased to ensure that both requirements are maintained. In this respect, Great Northerm's estimated minimum flow of 2.610 cfs at North Twin to pass 2.000 cfs at Millinocket and 350 cfs to the back channel may be conservative.

The drysson of Project Compliance and Administration has approved Great Northern's plan to install automatic drysses. Like those recently wishalled at McKay Station, to improve the ability to pass the 2.000 cts minimum flow as Milliancies in the event of interruptions of the power system.

- 38 -

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-24 GNP states that a 10% buffer is needed to ensure compliance

with the minimum flow requirements to account for uncertainty in flow regulation, gate settings, and control equipment sensitivities. Although GNP has not provided the information needed to determine exactly how much buffer is needed, the exact amount is irrelevant. The 2000 cfs minimum flow at Millinocket could include flows provided to the BC, as we explain more fully in Appendix D. The economic consequence relative to the environmental benefit of providing flow to the BC is the primary factor in determining its feasibility.

E-159

As the Commission evaluates comments on the DEIS it is important to assess accurately the true availability of water and the realistic capabilities of the WUP as it must be implemented in day to day operation under the terms and conditions of a new ficense. Viewed in this light, there is no basis for the conclusion that increased flows to the back channel would be consistent with the WUP.

GNP-24 Cont

As indicated in Penobscot Mills Project Volume VII (Appendix E2 V 5.0, page 58), the Commission should be aware that this 10% buffer requirement is not included in the original modeling work supporting Great Northern's proposed WUP (i.e., weekly flows as low as 2,000 cfs from North Twin were used in the models). However, Great Northern is confident that the requirements of its proposals can be met as evidenced by the successful modeling of the WUP during the theoretical worst-case year (DEIS, Appendix, pages D-15 and D-16).

Winter Flows in Millinocket Stream

Great Northern agrees with the Commission that "Millinocket Stream has little value as a salmon production area" (DEIS, page 4-30), and that "salmon production in Millinocket Stream would not increase significantly (DEIS under Alternative 1, page 4-28). However, these findings are inconsistent with the statement in the Executive Summary that a substantially enhanced salmon population would result from winter flows of 30 cfs.

GNP-25

E-160

The necessity of winter redd protection in Millinocket Stream is overstated since few (if any) salmon will spawn by October 15 (at 60 cfs). The WUP conservatively assumes a salmon spawning period of October 15 to November 15 on the West Branch (i.e., virtually all salmon would be expected to spawn during this period): there is no reason to believe that the timing of spawning on Millinocket Stream would differ. Therefore, protection of redds with higher winter flows is not an issue (i.e., eggs will incubate through the winter under the same flow regime as that which occurs during spawning in late October and early November).

Juvenile salmon habitat is already extensive in Millinocket Stream under the existing flow regime, i.e. 20 cfs (see Penobscot Mills Project. Volume 11, Section E3.1 and Appendix Q of Appendix E3.1 V-2), and yet a significant adult population has not resulted. Increasing winter flows on this stream to create additional juvenile salmon habitat would be a wasteful use of water resources, since the salmon population is bounded by other factors such as limited adult habitat, insufficient forage base (smelt), and/or high seasonal flows. Because of these overriding limiting factors, any increase in salmon stock size resulting from additional juvenile salmon habitat (resulting from a 30 cfs minimum flow in the winter) would be insignificant.

- 39 -

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-25 We revised our statements (in sections 4.4.2.3 and 4.4.3.3 and in the executive summary) concerning the benefit of additional minimum flows in Millinocket Stream. As explained in section 4.4.1.3, habitat for juvenile, adult, and spawning salmon increases steadily up to 80 cfs, whereas habitat for early and late salmon fry decreases slightly at flows greater than 20 to 30 cfs. Interior has agreed to a year-round flow of 60 cfs in Millinocket Stream as part of its 10(j) recommendations. Giving due deference to Interior's expertise in fish and wildlife matters, we are accepting this recommendation and recommending a year-round flow of 60 cfs at this location.

GNP-25 Cont

Great Northern continues to support DIFW's management of Millinocket Stream for stocking hatchery brook trout. Great Northern's proposal for 60 cfs in the summer and 20 cfs in the winter is consistent with this management strategy.

Upper Gorge Boating Flows

The DEIS concludes that "two weekend releases during May would not conflict with fisheries goals; would have a negligible cost to GNP; and would provide whitewater boating opportunities for more than 200 paddlers" (DEIS, page 4-55). The specific recommendation is for Great Northern to provide approximately 1.000 to 2.000 cfs during scheduled flow releases in the Upper Gorge for expert daylight kayaking on two weekends in May.

There are two ways to implement potential flow releases down the Upper Gorge. The water may either be diverted from McKay Station thereby impacting the production of hydroelectric power, or it may be in addition to flows from Ripogenus impoundment included in the WUP. In order to minimize the power generation and WUP impacts in accomplishing this goal. Great Northern would attempt to utilize natural spillage events from Ripogenus Dam to fulfill this requirement.

The Commission's conclusion that placing a bypass flow in the Upper Gorge (up to 57% of total station flow) would have a negligible cost can only be based on an assumption that natural spillage flows would be used in great part to meet this requirement. As can be seen in the 15 year period of record included in the WUP, natural spills are random, and it would be impossible to predict accurately the available water from the spring runoff. In fact, in some years the Ripogenus impoundment does not fill, i.e., there is absolutely no natural spillage from Ripogenus Dam. It is clear that the requirement to notify the whitewater boaters at least 30 days in advance will make it almost impossible to utilize the infrequent natural spillage events because of their random occurrences.

The Commission should also understand that if natural spillage events are not utilized. it is likely that water would be diverted from McKay Station to provide these boating flows to maintain the WUP flow regime. One concern in "splitting" the flow in this manner is the resulting possible impact on smelt drift and the downstream West Branch salmon fishery (see Fisheries. Section 6). Also, the FEIS must recognize that under some conditions, beyond the control of Great Northern, such as high spring flows and impoundment elevation, it will not be possible to reduce flows in the Upper Gorge to safe boating levels to comply with this requirement.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-26 Opinion noted. We revised the text of the EIS to reflect that GNP would notify whitewater interests at least seven days before the two scheduled weekend releases during May in the Upper Gorge.

GNP-26

- 40 -

To minimize costs and possible effects on other resources. Great Northern would hope to utilize natural spillage events for these boating flows whenever possible. Great Northern suggests that in the FEIS, the Commission revise this proposal to require that a plan for notifying the hoaters of natural spill events be developed in consultation with that group. From past experience, it is clear that a much shorter notification period is adequate so long as it is assured that the information is disseminated in a timely manner.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-26

Cont

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

.

5

.

WATER QUALITY

Back Channel - State's Waiver of Water Quality Certification

The Commission has rightly concluded that it should adopt virtually all relevant aspects of Maine's water quality certifications issued by the LURC and DEP. Thus, Great Northerm does not disagree with the DEIS's ultimate conclusions and findings with respect to water quality issues.

Nevertheless, the Commission has, in its discussion of environmental impacts, observed that there is "no compelling basis for challenging the State's decision to waive certification for back channel" (DEIS, page 4-13). This statement implies that FERC could, if it desired, challenge or override the State's water quality decision. Great Northern believes that not only is there no environmental basis for questioning the certification waiver, but there is clearly no legal basis. As the Commission knows, the water quality certificates were issued after a complete public process involving hearings, opportunity for public comment, draft certifications and ultimately final issuance by Maine's major environmental and land use commissions and boards. These water quality certificates were not appealed, nor was reconsideration sought, by any of the parties to this proceeding.

It is clear under the federal Clean Water Act that the NEPA process cannot be used to question the adequacy of any state certification decision under Section 401 (33 U.S.C. Section 1341(c)(2)). The State's findings must be conclusive on the issue of whether water quality standards are met or have been waived. Recently, the U.S. Supreme Court has confirmed the State's role in water quality standard implementation in a federal proceeding through the use of its Section 401 authority. See, Jefferson County P.U.D. v. Washington, 1994 U.S. Lexis 4271 (1994). The State's supremacy in this regard has been clearly endorsed in a long line of cases under the Clean Water Act. as well as other federal statutes which have confirmed that the certification process rests exclusively with the State. In essence, the water quality certification process is the State's opportunity to assert its views on water quality. Here, the State of Maine made a decision with respect to certification in the back channel. Whether that decision was to certify, not certify, or waive, it is not one which can be challenged by FERC through a contrary conclusion in the NEPA process. See, for example, Lake Erie Alliance v. Army Corps of Engineers, 526 F.Supp. 1063 (W.D. Pa. 1981); aff'd 707 F.2d 1392 (ed Cir. 1983) cert. den. 464 U.S. 915 (1983); Roosevelt Campobello Int. Park v. EPA, 684 F.2d 1041 (1st Cir, 1982),

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-27 Opinion noted. The status of Maine's 401 WQC conditions for the projects will be addressed in the orders for the projects.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

Back Channel Water Quality

Great Northern agrees with the Commission's finding that "using existing conditions as the baseline, we conclude that Great Northern's proposed projects would not affect water quality in back channel because Great Northern proposes no changes in the existing flow regime" (DEIS, page 4-13). However, Great Northern believes the FEIS should also reflect that evidence has been presented to the Commission which documents that Maine water quality and aquatic life standards are being met throughout the length of the back channel under current leakage conditions. In fact, this evidence was provided in studies performed in response to the Commission's request that additional water quality investigations be conducted in the back channel to determine compliance with its classification. The results of those studies, submitted to the Commission in December 1992, demonstrated that dissolved oxygen levels consistently met Maine Class C standards of five parts per million or 60% saturation, whichever is higher, and that E, coli bacteria were minimal and well below the required standard. Further chemical/physical indicators of water quality such as temperature, pH and conductivity all indicated healthy conditions typical of stream waters in the West Branch basin (see Penobscot Mills Project, Volume XIII, 2).

다. GNP-27 --- Cont

Aquatic life conditions were also well documented during the mid-1992 studies although not specifically requested by either the Commission or by any resource agency. Three stations were established within the back channel to evaluate the health of the aquatic macroinvertebrate community using DEP methodology. The results of this investigation demonstrated that a diverse macroinvertebrate community is present throughout the length of the back channel under leakage conditions, and that the community meets or exceeds the draft Class C aquatic life criteria for flowing waters currently under evaluation by the DEP. Furthermore, the studies required by the Commission clearly demonstrate that the lotic (flowing) characteristics of the back channel have not been adversely affected by the presence of Stone Dam (see Penobscot Mills Project, Volume XIII, 2).

Great Northern believes that the Class C designated use of the back channel as aquatic life habitat has been met, and is being met, under leakage conditions. This leakage flow has created its own well-established channel environment within the former riverbed over the past 90-plus years.

Mercury

Great Northern supports the Commission's conclusion that "based on our review and data presented by Great Northern, we conclude that current or proposed operation of the projects, including periodic drawdowns of the reservoirs, would not increase mercury levels in the impoundments or the production of methyl mercury" (DEIS, page 4-12). Nevertheless, because CLF et al continues, even as recently as the January 25, 1995 hearing, to assert some unspecified connection between drawdowns and mercury, it is important that the FEIS thoroughly support the fact that mercury in Northern Maine takes is a regional, not local issue.

Historical increases in mercury concentrations in sediments and surface waters of remote lakes across the northern hemisphere have been extensively documented. This increase is known to have occurred within the last 100-200 years and is generally associated with an increase in industrialization and contribution of fallout of airborne mercury. There is no evidence linking these mercury concentrations to impoundment operations. A number of surveys in Maine have shown that mercury levels are elevated in fish from some inland lakes. The same pattern has been reported in blood and feathers from nesting bald eagles reflecting the eagles' exposure to mercury via the consumption of fish, their primary food source. Concentrations of mercury are generally greater in tissues of eagles from lakes in eastern and northern Maine than in those from estuaries or coastal Maine. Interestingly, coastal eagles (and other birds) contain greater levels of chlorinated organic contaminants (pesticides, PCB) than do the inland (lake or river-dwelling) populations. Researchers are unsure of the reason for this geographic-based trend. While the reproductive success of Maine's bald eagle population has been lower than in most other U. S. areas, the primary cause of this impairment appears to be the continued exposure to the chlorinated organic compounds. Dr. Ian Nisbet, an acknowledged expert on this subject, testified at length to the eagle issue at the LURC water quality certification hearings (LURC 401 Hearings, February 23, 1993, Transcript Vol. 1, pgs. 145-150 and 235-238).

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-28 No response required.

Great Northern has performed extended studies on mercury in sediment, surface water, and fish tissue taken from its project waters. These efforts supplement those conducted by the Penobscot Indian Nation (PIN), the USFWS, and the University of Maine. Environmental Science and Engineering (ESE) has acquired many of the available public and private fish monitoring studies on mercury for all New England states (dating back to 1972) and compiled a computer database containing over 750 individual records. The weight of evidence collected to date indicates that mercury is at least a regional problem and likely a national or international issue. The majority of experts in this field agree that much of the mercury arrives as fallout deposited from pollution migrating along the trade winds of the "northeast corridor" (from targe scale industrial, fossil fuel power-generating, and municipal incineration activities in the midwest and along the eastern seaboard).

Mercury released from newly flooded soils has been documented to occur in the U.S. and Canada. Specific references to the phenomenon were detailed in Great Northern's December 1992 response to FERC's June 1992 Additional Information Request and in the written and oral testimony of Great Northern's four expert witnesses at the LURC water quality certification hearings. No documentation has been presented to indicate that impoundments of the age of Great Northern's (portions dating back to the 1840's) continue to release mercury as a result of original flooding. Nor is there a single study published in a referenced scientific journal that links mercury cycling in aged impoundments to annual water level management.

Great Northern's literature and data search. performed at the request of FERC. presented evidence that mercury is a common contaminant in fish throughout Maine and New England. It is found in river systems, remote natural lakes and even in water supply reservoirs. Large lake trout (trophy game fish rarely caught by anglers) sampled from Maine's waters have mercury concentrations comparable to the fish caught in Great Northern's project waters irrespective of the origin or the nature of watershed management.

GNP-28 Cont

E-167

The mercury levels observed in Great Northern's project waters are consistent with other Maine lakes as well as other northern lakes in Canada. Wisconsin, Minnesota, Sweden and Finland. No viable mechanism has been suggested that could relate the bioaccumulation of mercury to project operations. Acknowledgment that the disposition of mercury is a pervasive problem in the northeast came at a recent meeting of the American Water Resources Association (Northeast Chapter, Marlborough, MA, 1994). Discussions led by experts in the field (Russ Isaac, Terry Haines, John Colman) indicated that airborne migration and deposition of pollutants (especially coal-fired utilities) still appears to be the major source of mercury input to remote northeastern lakes.

In summary, the record supports the DEIS's conclusion that project operations do not increase impoundment mercury levels. Great Northern has demonstrated both in its applications for new licenses and in supplemental studies, and through the water quality certification proceedings, that its project operations are not adversely influencing the cycling of mercury in project waters. This evidence has been confirmed by the conclusions reached at recent scientific conferences that have addressed this subject.

Dolby Pond Dissolved Oxygen Levels

The DEIS refers to the continuing claim by CLF et al that Great Northern has not adequately investigated the cause(s) of low dissolved oxygen concentrations in bottom waters of Dolby Pond (DEIS, Section 4.3.1.2). Great Northern believes that it has more than adequately demonstrated that operation of its FERC licensed hydroelectric facilities do not cause or exacerbate dissolved oxygen conditions within the bottom waters of Dolby Pond (see Penobscot Mills Project. Volume II. sections 2.2.5.4(b) and 2.5.1.4).

Dolby Pond is a relatively shallow man-made impoundment with essentially stable water levels. A small area of the pond, spatially removed from the main channel area and Dolby Dam, exhibits low DO in its bottom waters during a portion of the summer months. This condition is well documented to exist only in restricted hypolimnetic waters (generally below 25 feet) affecting a small percentage (less than five percent) of the pond's total water volume.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-29 Information you present here confirms the DEIS conclusions that hydroelectric project operations do not result in low DO problems in Dolby Pond. The legal issue of requiring DO studies in Dolby Pond will be addressed in the order for the project.

GNP-28

GNP-29

Cont

Low DO concentrations in bottom waters of stratified natural and impounded lakes and ponds is a commonly documented phenomenon. The DEP in their 1992 water quality assessment report [305(b) report] documents some 188 Maine lakes with DO concentrations in bottom waters less than three mg/l due to natural causes. Table 4 of Mr. Dennis Sasseville's 1993 written hearing testimony to LURC, which has been provided for the FERC record, listed 11 lakes in Maine as examples of healthy water bodies with seasonally low DO in bottom waters due to natural causes (written testimony presented by Dennis R. Sasseville on February 1, 1993 and sent to LURC by Great Northern).

Sampling in Dolby Pond by Great Northern in 1994 confirmed that low DO conditions are observed in the "deep hole" but not in the mid-pond nor in the river channel areas that comprise the majority of this water body's surface area (Table 1). DO levels throughout Dolby Pond are consistently high and welf above Maine Class C standards of 5 parts per million or 60% saturation. Biological Oxygen Demand (BOD) samples collected by Great Northern in the summer of 1994 further document that oxygen demand in the water column is low throughout Dolby Pond (Table 2).

GNP-29 Cont GO

Table i

Dolby Pond Dissolved Oxygen Profiles August 17, 1994

Station	Deep Hole		Mid-Pond		River Channel	
Depth	DO	Temp	DO	Temp	DO	Temp
(Feet)	<u>(mg/l)</u>	<u>(C•)</u>	<u>(mg/l)</u>	(<u>C</u> •)	<u>(mg/l)</u>	<u>(C*)</u>
3.5	8.60	20.6	8.70	20.6	7.65	20.0
7.0	8.55	20.6	8.50	20.6	7.60	19.9
10.5	8.32	20.5	8.40	20.5	7.60	19.8
14.0	8.00	20.5	8.13	20.4	7.60	19.8
17.5	7.87	20.4	8.05	20.3	7.60	19.8
21.0	7.27	20.4	7.75	20.2	7.60	19.7
24.5	6.80	20.2	7.65	20 1	7.55	19.7
						(24.5')
28.0	3.90	17.0	6.00	19.5		
		•		(27.0)		
31.5	0.14	15.8				
34.0	0.12	13.9				
35.8	0.11	13.0				

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

Table 2

BOD Analysis of Dolby Pond Water

BOD Sample Location	07/14/94	08/17/94
Deep Hole - Location A. mid-depth	0.9	1.1
Deep Hole - Location A. 3' above bottom	2.3*	2.3*
Deep Hole - Location C. mid-depth	1.0	1.2
Deep Hole - Location C. 3' above bottom	1.0	1.8*
Mid-Pond - mid-depth	1.4	1.3
River Channel - mid-depth	0.6	0.9/0.6
		(2 samples)

BOD in mg/l (5 day test) *Settleable solids present which may have increased BOD.

These data support the position that Great Northern's effluent discharges in Millinocket are not impairing DO conditions in Dolby Pond, a position also supported by the DEP water quality modeling effort conducted in 1987 and reported in the Penobscot River Basin Waste Load Allocation Report of January. 1991. Rather, the low DO conditions observed in the Dolby Pond hypolimnion during summer months result from weakly stratified conditions (common to natural lakes and impounded waters) and the probable natural effects of Sediment Oxygen Demand (SOD) on these trapped deeper waters.

Great Northern's current and proposed method of hydropower operations clearly do not adversely affect the DO regime of Dolby Pond. However, as a condition to the water quality certification. Great Northern has agreed to further investigate wastewater treatment conditions at its West Operation in Millinocket. These studies, and any National Pollutant Discharge Elimination System (NPDES) permit renewals for this wastewater discharge, should not impact the Commission's schedule for relicensing the Penobscot Mills Project.

GNP-29 E-1 70

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

6

.

FISHERIES

Back Channel: Upper Gorge: Millinocket Stream

The DEIS concludes that the "Back Channel probably could not be established as a production area that would enhance salmon abundance downstream". Moreover, the DEIS also determined that "implementing Alternative 1, therefore, could jeopardize an attainable fisheries goal (i.e., establishing a lake trout population in North Twin) in an attempt to achieve a less valuable goal (i.e., establishing a self-sustaining landlocked salmon stock in back channel" (DEIS, page 5-8).

Great Northern believes that flow commitments contained in the WUP, as is currently proposed, meet all environmental and recreational goals sought during consultation. There is no reason to release higher flows in the Upper Gorge. Millinocket Stream or the back channel. Indeed, Great Northern has documented that the proposals to increase these flows either confer no benefit, or in fact, have a negative impact upon the existing fish communities (and recreational groups that rely on them) in the West Branch watershed. In summary, these proposals will not result in additional fish, and could, by reapportioning water, actually be detrimental to existing fish resources.

The principal reasons not to increase flows are the world-class West Branch salmon fishery, the smelt "drift" that supports it, and the North Twin impoundment lake trout restoration program. The high base flows in the West Branch below McKay Station are the key to the unique year-long smelt drift. Reducing McKay Station flows raises the risk of altering the ongoing smelt drift, thus reducing and conceivably eliminating that salmon population. Less flow in the West Branch could affect North Twin impoundment winter lake trout egg incubation levels, thus compromising lake trout restoration. Winter low flow events in the West Branch could also negatively impact incubating salmon eggs.

Although back channel flows of 350 cfs would provide the physical conditions (depth and velocity) believed to be suitable to fry and parr life stages of landlocked salmon. actual colonization would not be realized. Any strategy that involves juvenile salmonid production in the back channel depends on contiguous adult habitat either in the Dolby impoundment or in the back channel itself. Despite intensive study of such adult habitat by Great Northern, observations of landlocked salmon were rare in the back channel/Dolby impoundment area. These studies confirmed that while some of the physical factors that favor adult salmon may exist, by using a presence of fish criterion adult salmon habitat does not exist (Penobscot Mills Project, Volume II, E3.1-106).

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-30 Your comments on the merits of increasing flows in the Upper Gorge, Back Channel, and Millinocket Stream are noted. In the FEIS, we recommend not increasing the minimum flow in the Back Channel, adopting GNP's proposed flow enhancement for the Upper Gorge, and providing a year-round minimum flow of 60 cfs in Millinocket Stream. We consider this combination of fisheries enhancements to be the most balanced use of project waters.

GNP-30

Factors that are limiting adult salmon use of potential habitat would not be ameliorated with a greater flow. These factors include lack of smelt drift, a fish species mix that favors non-salmonid species, and the occasional hot summer that produces elevated water temperatures. Great Northern's studies found low smelt abundance in Quakish Lake and low subsequent drift to downstream outlet areas. Smelt abundance is considered high in the Dolby impoundment: however smallmouth bass and pickerel provide competition and predation, and naturally-occurring hypolimnetic oxygen deficiencies further reduce limited salmon habitat during the summer. Additional problems associated with spill events, embedded spawning substrates, and downstream predation of juvenile salmon are minor when compared to the lack of spawning escapement, i.e., adult salmon are needed to complete the life cycle. In sum, the absence of contiguous adult salmon habitat is the major reason why flow in the back channel has no biological merit.

It is ironic that the conservation intervenors suggested at the January 25, 1995 DEIS hearing that a brook trout fishery be developed in the back channel. Back channel brook trout management strategies, similar to those ongoing in Millinocket Stream, were discussed with the agencies during consultation. The conservation intervenors refused to support DIFW's goals for a Millinocket Stream brook trout fishery, and yet they are now suggesting a brook trout fishery for the back channel. All of the environmental problems alleged for Millinocket Stream by the conservation intervenors are significantly more imposing in the back channel. Given higher thermal regimes and lack of overhead cover and thermal refuge areas, it is not surprising that the agencies rejected back channel brook trout management proposals five years ago. Great Northern assumes that FERC will take the conservation intervenors back channel request as an endorsement of the Millinocket Stream fisheries management plan. For the litany of reasons outlined above. Great Northern does not believe that brook trout or salmon fisheries are realistic back channel management objectives.

Moreover, the DEIS is correct when it points out that a back channel flow of 165 or 350 cfs would compromise other environmental or recreational goals during dry years. The severe impact of transferring water allotted for one goal to another cannot be underestimated. Flows below McKay Station are the most obvious example. While lower spawning and incubating flows can be manipulated in the WUP, the effect of reduced winter flows on the extent and duration of smelt drift is unknown. Smelt drift netting studies below McKay Station have shown that smelt drift is poor during the normal low flow periods of late March

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-30 Cont

E-173

and early April. If Ripogenus impoundment "runs out" of water during the winter month and smelt drift is reduced earlier than normal, the effect on the West Branch salmenpopulation, especially mending kelts (post-spawning salmon), could be devastating. Any further reduction in West Branch flows during this late winter/early spring period will impact the health of the world famous salmon population. This in turn would affect the Upper Gorge and Holbrook side channel enhancement measures sought by Maine fisheries and environmental agencies.

GNP-30 Cont

Finally, although additional storage could, of course, also be used from the North Twin impoundment, the use of this storage to meet back channel flow demands could jeopardize the DIFW's North Twin impoundment lake trout restoration program as lake levels are reduced below fall spawning levels. Again, back channel flows of 165 or 350 cfs will threaten other more realistic WUP goals during low water years.

Instream Flow Incremental Methodology (IFIM) below McKay Station

In the DEIS, the Commission restated its decision that Great Northern would not be required to undertake IFIM studies below McKay Station (DEIS, page 4-16). The original Commission decision found that "an IFIM would be an unnecessary academic exercise that would not materially improve the record in this proceeding" (Ripogenus Project, Volume XIV, page 69). Because this issue has been raised repeatedly by CLF et al even after rejection by FERC. Great Northern believes the FEIS should more fully explain FERC's IFIM decision by setting forth additional important considerations that were developed elsewhere in the record which deal with this issue (Ripogenus Project, Volume XIV, pages 69-75).

In addition to those set forth in the DEIS, there are four additional reasons for not undertaking an IFIM study below McKay Station: 1) IFIM is not a fisheries management tool: 2) better information (use, catch, distribution) needed to manage the fishery presently exists: 3) collecting transect data would involve significant flow alteration and a significant fisheries and aquatic impact could be expected: and 4) there is no method of integrating macroconsiderations (smelt drift) into the IFIM analysis.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-31 Opinion noted. We do not think that further explanation or justification for not requiring an IFIM study in the West Branch below McKay Station is necessary; therefore, we did not modify the FEIS as you suggest.

GNP-31

The following facts should be set forth in the FEIS to further support the Commission's decision to not require an IFIM below McKay Station:

- WUA, the measurement unit of IFIM, does not equate to fish biomass; it is fish that the resource agency (DIFW) is charged with managing. As was discussed at length during the LURC water quality certification hearing, DIFW does not rely on IFIM analysis and did not want one performed in this case.
- 2. With the exception of emergency outages, all of the WUP flow regimes exceed the Aquatic Base Flow (ABF) of 711 cfs. Indeed, in its initial discussions of this subject, the USFWS requested either that Great Northern release a minimum flow consistent with ABF policy or that it perform an IFIM study. Since the flows proposed by Great Northern are consistent with the USFWS's ABF policy, no IFIM study is needed.
- 3. IFIM cannot address macrohabitat considerations, e.g. temperature, DO, pH, and smelt drift. The record is replete with references that confirm the critical predatory/prey relationship that exists between West Branch salmon and smelt drift from the Ripogenus impoundment. Since the IFIM cannot analyze possible flow/smelt drift relationships. flow regime(s) based on IFIM analysis would inherently ignore critical behavioral components (smelt predation) of the West Branch salmon population. A flow regime based on IFIM could destroy this world-class fishery by not considering these other factors.
- 4. Great Northern is not proposing any change in system operation: all proposed enhancements fall within existing flow regimes. Conservation intervenors would like to use the IFIM to determine an optimum operating flow. Since measurements of WUA are not linked to biomass. Great Northern believes that this use is beyond the scope of IFIM on the West Branch.
- 5. The West Branch has a number of moderate gradient areas, and therefore, conducting an IFIM would present serious physical and logistical difficulties. Also, river flows would require significant time to stabilize because of the many brooks and backwaters. The simple act of collecting the necessary flow and elevation measurements would require numerous days of data collection. The temporal aspect of limiting West Branch flows (at 400, 600, 800 cfs) for a lengthy period would have a documented (FERC No. 3779, Volume IV, Exhibit E, Section E3, 2-52) negative effect on fish and aquatic communities and whitewater recreationists.
- 6. The West Branch has a number of high gradient areas. Conventional IFIM models do not work well when stream gradient exceeds 3%. In the West Branch, areas such as these are critical parr habitats. If these habitats cannot be included in the IFIM process, it makes little sense to conduct that study.
- 7. The West Branch has a number of large deadwaters (Debsconeag, Frozen Ocean for example) that have extremely low velocities during low/moderate flow events. In addition to providing some adult habitat, these areas also provide critical overwintering habitat. IFIM and Habitat Suitability Index (HSI) curves were not designed for large deadwaters and overwintering habitat.

GNP-31 Cont

E-175
RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

7

WETLANDS

Ripogenus Impoundment

GNP-32

GNP-33

Great Northern believes that the DEIS recommendation to provide wetlands enhancement at Black Pond on the Ripogenus impoundment as a mitigation measure is not appropriate and will have significant adverse environmental impacts outweighing any benefits gained.

As indicated throughout the DEIS (e.g. page xiii), the Commission has chosen the No-action Alternative as the baseline for comparison of proposed project changes since the projects and their operations have existed in essentially their current state for many years. Thus, any changes to the projects or their operations proposed by the Commission serve to enhance existing conditions rather than mitigate for any adverse effects. The proposed barriers at Black Pond (Alternatives I and 2) and Quaker Brook (Alternative I) on Ripogenus impoundment would increase (enhance) wetlands acreage in the Ripogenus Project: however, they do not mitigate for acreage affected by Ripogenus impoundment drawdowns since these drawdowns are considered in the baseline operating conditions.

More importantly, the construction of any barriers that would develop wetlands would limit fish and boat access from the impoundment to the tributary areas associated with this proposal. Fish passage is of particular concern. While a number of fish species (cusk, white perch, salmon) may use these tributaries for some phase of their life history, smelt reportedly use both the Black Pond outlet (Caucomgomoc Stream) and Quaker Brook for spawning. For the Ripogenus Project, smelt are the most critical forage fish species and the key component to agency fisheries goals for the Ripogenus impoundment and its outlet area, the West Branch of the Penobscot River. Smelt are extremely poor swimmers; any significant hydraulic control structure for wetlands enhancement would prevent passage and eliminate all upstream reaches as production areas. The loss of critical spawning habitat would jeopardize production of young smelt and alter the predator/prey balance sought by the DIFW. Wetland flow control structures would also limit boating, primarily into Black Pond. The present principal access to Black Pond is across Brandy Pond and through the area proposed for wetland enhancement. Recreationists, often from Chesuncook Village, travel into this area to fish, hunt, canoe, sightsee and camp. Any flow control structure would curtail these recreational opportunities.

Although these barriers may enhance wetlands acreage in the Ripogenus impoundment, the environmental consequences of lost recreational boating access, lost smelt spawning areas and the corresponding risk to existing fish populations far outweigh any wetlands enhancement benefits.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

- GNP-32 GNP asserts that the DEIS recommendation to provide wetlands enhancement at Black Pond on the Ripogenus impoundment is not appropriate and will have significant environmental effects that outweigh benefits. As we stated in the DEIS, we estimate that approximately 20% (250 acres) of the 1,251 acres of wetlands at the Ripogenus impoundment would continue to be affected adversely under the proposed operation by exposure during impoundment drawdowns. It is appropriate to consider this measure to help minimize such effects. We address GNP's concerns on adverse environmental effects in response GNP-34.
- GNP-33 GNP has indicated that the proposed barriers at Black Pond would "...increase (enhance) wetlands acreage in the Ripogenus Project..." There may be a misunderstanding by GNP that the Commission is requiring creation of new wetlands in the vicinity of Black Pond and Caucomgomoc Stream. Our understanding is that the mitigation at Black Pond would be to enhance existing degraded wetlands; it would not necessarily increase wetlands acreage. This distinction is important because the affected existing wetlands are degraded due to project drawdowns and should be enhanced.
- GNP-34 It is unlikely that the placement of a structure at Black Pond would limit fish or boat/recreational access through Brandy Pond, except during impoundment drawdowns, when water on the Ripogenus side may fall below the top of the barrier. It is not certain at this time whether this situation would occur. If boat/recreational access to Black Pond were temporarily blocked through Brandy Pond during drawdowns, another access alternative may exist through Little Scott Brook. It is also unlikely that construction of the barrier structure at Black Pond would "prevent passage and eliminate all upstream reaches as production areas" of smelt. First, no information exists in record that definitively indicates that smelt spawn in

GNP-34

.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-34 Caucomgomoc Stream. Second, as designed, the barrier would

Cont'd only temporarily affect the movement of fish during drawdowns, when the water on the Bipogenus reservoir side falls below the barrier. Most importantly, numerous other tributaries in the vicinity of Black Pond provide unrestricted access to upstream fish spawning areas; therefore, we view the Black Pond wetland enhancement as beneficial overall.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

8

.

THREATENED AND ENDANGERED SPECIES

Bald Eagle Population Enhancement

The DEIS recommends that "GNP confer with DIFW and FWS to preserve existing eagle perching areas and to investigate the appropriateness of creating new perching areas around the periphery of the impoundments." The DEIS also proposes that "a five year plan for limited monitoring of the enhancements should be developed in consultation with DIFW and FWS to observe use by eagles, and a report of findings should be submitted annually" (DEIS, page 4-45).

GNP-35

GNP-36

Although bald eagles are currently on the federal endangered species list. a proposal is now being considered to downgrade their federal status to threatened because of increasing numbers within the United States. Maine's breeding eagle population has increased slowly but steadily since the 1960's (see Table 3 reproduced from the DIFW's 1994 Research and Management Report). DIFW (1994) states that the primary hindrances to eagle reproduction today are environmental contaminants and changing land use (human disturbances) near occupied eagle nests. In response to the latter problem, in 1990 the DIFW began designating protective "Essential Habitats" around eagle nests.

A lack of "perching areas" is not indicated by DIFW (1994) as a problem currently facing Maine eagles. Furthermore, the DEIS correctly concludes that "operation of the Ripogenus Project as proposed by GNP would have no demonstrable adverse effects on the bald eagle population in the project area" (DEIS.page 4-43). Thus, FERC's recommendation to preserve and enhance perching areas is unnecessary and unwarranted.

It should be noted that Great Northern already has an active and effective program in place to track the locations of eagle nests on its lands and to protect those sites. In fact, the company has a 25 year history of managing these sites cooperatively with the DIFW and the USFWS and is currently considering a voluntary program to enhance the viability of existing and future sites. Great Northern would be willing to formalize this program, which places emphasis not on perching sites as described in the DEIS, but on future nest trees. In the West Branch region of Maine, bald eagle nests are nearly always constructed in super-dominant white pine located in the riparian zone. Therefore, a management plan will be developed to meet the objectives of a) maintaining the overmature and mature white pine component in stands close to the water and b) allowing for eventual replacement of the older white pine by retaining mid-story white pine in the vicinity of the overmature trees. If feasible and desirable, these trees could be released to a free-to-grow condition.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

- GNP-35 GNP indicates that "FERC's recommendation to preserve and enhance [bald eagle] perching areas [at Ripogenus] is unnecessary and unwarranted." We have specified the bald eagle perching areas to assist in attracting and retaining new eagles and potential subsequent nesting in the vicinity of the Ripogenus Project. We regard such measures as a relatively easy, inexpensive means to enhance the existing population of a federally listed threatened species in the vicinity of the project.
- GNP-36 We have no details of any former or existing GNP program to track and protect eagle nests on its lands and, therefore, cannot judge its merits. As stated in the DEIS, at a minimum, we recommend that GNP confer with the DIFW and FWS to preserve existing bald eagle perching areas and to investigate the appropriateness of creating new perching areas around the impoundments. The DEIS also states that large shoreline trees (such as mature white pines) could be preserved through selective cutting techniques. If GNP wishes to further enhance eagle habitat by preserving potential eagle nesting trees through other methods as proposed in these comments. we recommend they include such measures in discussions with the DIFW and FWS. If the DIFW and FWS indicate that such measures may be feasible, they should also be included in the 5-year plan and annual reports for limited monitoring of the perching area enhancements. As stated in the DEIS. recommendations developed through consultations with the DIFW and FWS should be finalized in the enhancement plan and filed with the Commission for approval.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

TABLE 3

Bald Eagle Nesting and Productivity in Maine-

1962 - 1970 and 1972 - 1993

Year		Successful Sites		No. Young		Occupied Nests				
	Occupied Sites				Young Fledged/Nest		Fledging # of Young			
		<u>N</u>	<u>*</u>	Fledged	Occupied	Successful	ō	1	2	3
1962	27	8	30	8	0.30	1.00	19	8	0	0
1963	32	9	28	12	0.38	1.33	23	6	3	0
1964	28	6	21	6	0.21	1.00	22	6	0	0
1965	33	4	12	- 4	0.12	1.00	29	4	0	0
1966	28	7	25	11	0.39	1.57	21	3	4	0
1967	21	4	19	6	0.29	1.50	17	2	2	0
1968	23	9	- 39	11	0.48	J.22	- 14	7	2	0
1969	29	11	- 31	15	0.52	1.36	18	7	- 4	0
1970	32	8	25	11	0.34	1.38	24	5	3	0
1972	29	8	28	8	0.28	1.00	21	8	0	0
1973	31	6	19	6	0.19	1.00	25	6	0	0
1974	36	12	33	12	0.33	1.00	24	12	0	0
1975	31	9	29	- 11	0.35	1.22	22	7	2	0
1976	41	12	29	19	0.46	1.58	29	6	ŝ	Ĩ
1977	50	24	48	35	0.70	1.46	26	16	ŝ	3
1978	62	20	32	32	0.52	1.60	42	9	tõ	Ĩ
1979	52	29	56	38	0.73	1.31	23	20	9	Ó
1980	56	29	52	40	0.71	1.38	27	19	9	ĩ
1981	63	34	54	49	0.78	1.42	29	iś	15	ò
1982	72	36	50	56	0.78	1.56	36	17	18	ĭ
1983	74	40	54	60	0.81	1.50	34	20	20	ò
1984	66	35	54	46	0.70	1.31	31	24	11	ŏ
1985	86	51	59	75	0.87	1.47	35	27	24	ň
1986	89	50	56	76	0.85	1.52	39	25	24	ĭ
1987	91	46	51	65	0.71	F 4 I	45	28	17	i
1989	109	45	- 4î	70	0.64	1.56	64	20	25	ó
1990	123	69	56	98	0.80	1.42	54	40	20	ň
1991	127	79	61	117	0.92	148	48	44	ž	ž
1992	140	77	55	113	0.81	1 47	63		12	5
1993	150	84	56	115	0 77	1 37	66	53	31	ā

 Data comparisons between the periods 1962-1967 and 1968-1989 are ravalid the to variations in survey methodology, regional emphasis, and intensity 1988 data were incomplete due to fack of labora.

.

- 55 -

Two types of areas will comprise the focus of this management plan. The first will be currently used sites. There are a number of sites in the plan area which have long-term historical use. It is clear that the conditions are suitable for supporting eagles at these locations and the implementation of the management plan described above can help to secure the long-term supply of nest trees. In addition, there would be no new development (structures) permitted on Great Northern lands within 1/4 mile of the areas containing promising future nest trees that are subject to the above management plan.

The second type of area will be highly-promising potential sites. These will be determined by reviewing the entire plan area to identify shallow water areas that are preferred by eagles for fishing. In these coves and deadwater areas, the shoreline will be searched for suitable nest trees, if the site is determined to have a high likelihood of suitability for hald eagle nesting based on food source and nest site conditions, then the management plan as described above will be developed with the best cluster of trees as the focal point. There would be no new development (structures) allowed near the shoreline that is within a 1/4 mile circle around these clusters of trees on Great Northern lands.

Great Northern would agree that this eagle population enhancement plan will be developed and formalized in a Memorandum of Understanding between DIFW and Great Northern by January 1, 1997. in addition, the effectiveness of the plan will be monitored cooperatively by DIFW and Great Northern. The plan will be adjusted if monitoring results indicate changes are needed.

The area subject to this plan is the Great Northern owned shoreland surrounding all the impoundments in the Ripogenus and Penobscot Mills Projects, plus the sections of the West Branch of the Penobscot River in between these impoundments.

shoreline of the impoundment. In fact, the setbacks proposed in the DEIS would likely be ineffective for the purpose of enhancing eagle populations. Therefore, Alternatives 1 & 2 do not provide any enhancements, and this item should be deleted from Table 5-2 if similar

Since this plan will prohibit development within a 1/4 mile zone around areas with realistic potential for future eagle nests, there is no benefit to bald eagles from a general project boundary expansion that would increase existing building setbacks on the entire

GNP-37

RESPONSES TO GREAT NORTHERN PAPER. INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-37 No plan currently exists in the record indicating areas with "realistic potential for future eagle nests." Hence, building setbacks required under Alternative 1 could enhance habitat protection. We revised Table 5-2 to reflect that the proposed conservation easement and boundary expansion could also enhance habitat protection.

GNP-36

comparisons are made in the FEIS.

E-182

Cont

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

o

r

•

CULTURAL RESOURCES

Penobscot Indian Nation Land Claims

Section 4.11.1.2 of the DEIS states that the claims of the PIN to islands in the West Branch of the Penobscot River "remain unsubstantiated" (DEIS, page 4-69). The DEIS further states that "(t)his issue and associated impacts remain unresolved". Great Northern agrees that the PIN claims are unsubstantiated but respectfully disagrees that this issue is unresolved.

In its Motion to Supplement Intervention filed with FERC on May 24, 1993, the DOI provided notice to the Commission of its intention to determine whether it would seek to exercise its Section 4(e) conditioning authority to the Penobscot Mills Project. In this Supplemental Motion, the DOI explained that, pursuant to its trust responsibility to the PIN, it was investigating the PIN's historical and statutory claims to lands on the West Branch of the Penobscot River:

"The Department of the Interior is currently reviewing the Nation's ownership claims pursuant to the terms of the Settlement Act. If it is determined that the Ripogenus project works are located upon reservation lands, as defined in 25 USC 1722 (i) and in the Maine Implementing Act. (30 Maine Revised Statutes Section 6201, et. seq.), the Department shall proceed with the development of conditions as authorized by Section 4(e) of the Federal Power Act. If so the Commission is advised that determination of annual charges for use of tribal lands under Section 10(e) may also be required. See Motion to Supplement Intervention in Opposition By Department of the Interior, 3 (May 25, 1993)."

DOI undertook a thorough review of this issue. including reviews of state information and information provided by Great Northern which included land ownership records of Great Northern land holdings on the West Branch and the legislative history of the Maine Indian Claims Settlement Act, codified at 25 U.S.C. 1721 et seq. ("the Settlement Act"), and the Maine Implementing Act, codified at 30 M.R.S.A. 6201 et seq..

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

GNP-38 The staff reviewed available information regarding the Penobscot Indian Nation's (PIN) claims to lands within the branches of the Penobscot River and concurs with the decision of the Maine State Department of the Attorney General (see section 4.11.1.2) that the PIN retains no ownership or title to lands within the branches of the Penobscot River. The staff finds consideration of the PIN traditional practices within the project area are outside the scope of Section 106. The staff, therefore, does not recommend including the PIN as a concurring party to the Programmatic Agreement associated with the Ripogenus and Penobscot Mills projects. The staff acknowledges, however, the PIN's interest in the management of historic properties potentially eligible for listing in the National Register of Historic Places within the project areas. Accordingly, the staff recommends consulting the PIN during the development of the revised Cultural Resource Management Plans for the Penobscot Mills and Ripogenus projects.

GNP-38

Because the issue of what islands are owned by the PIN under the Settlement Act is a question of State law, the Maine Attorney General's Office also undertook an extensive review of this matter. DOI has not disagreed with the legal and factual conclusions of the Maine Attorney General's review of this matter. The Attorney General of Maine summarized the results of this review in a letter to Michael J. Anderson. Associate Solicitor with DOI, dated December 16, 1993 (see Appendix B). The Maine Attorney General concluded that:

- "With the exception of the Indian reservation and Indian territorial lands set forth in the Settlement Act, the Settlement Act extinguished all claims to land and islands which the Maine Indians. Indian Tribes. and Indian Nations asserted prior to enactment of the 1980 Settlement Act;
- The islands in the branches of the Penobscot River are not part of the Penobscot Reservation defined in 30 M.R.S.A. 6203(8), and further, these islands historically were never part of the Penobscot Reservation; and
- Under the Settlement Act and the Maine Implementing Legislation. Maine law is explicitly required to be applied to resolve land disputes arising under these acts and prior treaties:" (see Letter of Michael E. Carpenter, State of Maine. Attorney General to Michael J. Anderson, Associate Solicitor, Division of Indian Affairs, U.S. DOI (Dec. 16, 1993)).

Notably, the PIN has previously taken the position that it held only the islands in the main stem of the Penobscot River. The islands in the main stem of the river between Old Town and Mattawamkeag were originally granted to the PIN in a 1796 agreement and further reserved for the PIN by a Treaty signed in 1818. The fact that the reserved islands do not include islands on the branches of the Penobscot River was evidenced by the admission of the PIN's former counsel. Harvard Law School Professor Archibald Cox. Professor Cox, in a letter dated March 22, 1977 and accompanying background documents sent to Judge William B. Gunter. President Carter's appointed representative to resolve the Indian's land claims. wrote (p. 29):

"In 1818 the Penobscots, who had failen on hard times, sent word to the State that they wished to sell an additional ten townships. The Commonwealth responded by appointing three commissioners to treat with the Tribes for the release of all its remaining lands. The Tribe reserved from the conveyance four townships near the point where the east and west branches of the Penobscot River converge. The Tribe also reserved the islands in the river which had been previously reserved."

The islands "previously reserved" were those in the 30 mile stretch of the main stem as indicated by Cox's conclusion that "today the Penobscot Tribe has only the islands in the Penobscot River between Old Town and Mattawamkéag". Research Summary at p. 30 (see Appendix B). Thus, the PIN's own counsel admitted that the 1818 Treaty only reserved those islands lying between Old Town and Mattawamkeag in the main body of the River.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS

m

GNP-38

Cont

- 58 -

Further. Professor Cox's view on what the Penobscots really owned was shared by the DOI which conveyed similar conclusions to the Justice Department in a letter dated January 10, 1977 (see Appendix B). This DOI letter describes the 1818 Treaty as follows:

"Most of the rest of Penobscot territory was lost as a result of the treaty of June 29, 1818 between the Penobscot Nation and Massachusetts. Reserved from an otherwise complete cession of all their lands above the thirty-mile tract lost in the 1796 transaction were four townships now identified as Mattawamkeag. Woodville, Indian Purchase, and Millinoctet. Those tewnships were purchased by the State of Maine in 1833. None of these transactions appear to have been executed in accordance with the Non-intercourse Act. As a result, the Penobscot Nation today holds only the islands in the Penobscot River between Old Town and Mattawamkeag. (emphasis added) (Senare Hearings, Vol. 1 at p. 261), "

This letter indicates that DOI has, in the past, recognized that the PIN claim to the islands in the branches was limited to the islands on the main stem of the river.⁸

Thus, after conducting a lengthy review of the relevant legal and historical documents relating to these island claims. DOI concluded last year that it would not seek to exercise its conditioning authority of Section 4(e) in connection with the Penobscot Mills Project. In its letter withdrawing the Supplemental Motion. DOI stated that:

"Upon review, the Department has concluded that it will not exercise this authority in connection with the Penobscot Mills Hydroelectric Project." See Letter from Kerry O'Hara, DOI, Office of the Solicitor, to Ms. Lois D. Cashell, Secretary, FERC re Penobscot Mills Hydroelectric Project. No. 2458-009 (March 3, 1994).

DOI reached this conclusion after a thorough review of the complete record, including its own documents and evidence, whatever support the PIN presented to DOI, and the documents and analysis offered by Great Northern and the State of Maine. Since DOI's year-long review of this issue failed to substantiate the PIN claims and DOI declined to exercise its Section 4(e) conditioning authority, this issue should be considered resolved in the FEIS.

⁶ It should be noted that the legal views of Professor Cov and DOI are supported by the actual, historical understanding of the Penobicots' fand ownership by the State of Manne. Numerous varyes, and reports to the Maine Legislature and the Manne Department of Health in the last century and as recently as 1952, indicate that the State of Manne consulted only the valuable in the main riser between Old Town and Matsawamkeng to be reserved to use PIN.

RESPONSES TO GREAT NORTHERN PAPER, INC. COMMENTS ON UPPER PENOBSCOT RIVER BASIN DEIS



The Honorable Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, N.E. Washington, D.C. 20426

> Re: Great Northern Paper, Inc.; Project Nos. 2572 and 2458 Memorandum of Understanding

and the State further describing the easements to be conveyed is attached.

Dear Secretary Cashell:

The State of Maine and Great Northern Paper, Inc. have carefully considered and discussed the draft DEIS and what we believe are FERC's goals and obligations to assure reasonable shoreline protection for these projects in a manner consistent with its mandate to balance environmental and energy needs. Our cooperative discussions have led to an agreement to present FERC with an approach to shoreline protection which we believe meets FERC's needs and is consistent with the land use policies of the State. In addition, we believe that the approach we propose also addresses the concerns expressed by landowners and local residents while relieving Great Northern of a potential economic burden which could have serious adverse consequences for it and the State's economy.

The State of Maine and Great Northern have concluded that all of these important objectives can be met through conveyance to the State by Great Northern of conservation easements which follow many of the successful concepts already in place on areas of the West Branch. This time tested approach of voluntary conservation easements with State recreation management of easement lands has been a major factor in the national acclaim which this area has achieved for its recreational, aesthetic, fishery, water quality and other significant values. A copy of a Memorandum of Understanding between Great Northern

As the FERC staff knows from public comments at the January 25, 1995, hearing on the DEIS, as well as the comments filed by the State of Maine and Great Northern, a key concern is the DEIS proposal to expand project boundaries surrounding the impoundments to 200 feet and impose buffer zone/setbacks to limit building and forestry

1.

E-187

RESPONSES TO GOVERNOR KING AND DONALD MC'NEIL ON UPPER PENOBSCOT RIVER BASIN DEIS

1.

The State revised its land use assessment and recommendations in the FEIS (see section 4.9). Within the Ripogenus Project area, the recommended alternative proposes two options: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion on GNP-owned lands. Within the Penobscot Mills Project area, the recommended alternative proposes expansion of the project boundary on GNP-owned lands to 200-foot from the highwater mark of the. impoundments. The staff recommends development of a Shoreline Management Plan for the boundary expansion areas in consultation with various resource agencies. The proposed easements and/or boundary expansion would provide for protection of valuable shoreland resources. See section 4.9.3 of the FEIS for further discussion.

activities in the expanded boundary. The agreements reached in the Memorandum of Understanding will more than achieve FFRC's objectives on 73 miles of shorefront for Ripogenus and Chesuncook I akes and in a manner which is not inconsistent with, or contradictory to State policy

Finally, since all major landowners, including the State of Maine, whose land would be affected by FERC's proposal, have indicated an unwillingness to sell their land or convey any interest in it to comply with the DEIS' proposed recommendations, we believe the Memorandum of Understanding provides very significant new protection while avoiding a potentially contentious situation in which landowners and leaseholders could be forced to engage in long, bitter and nonproductive proceedings to protect rights to their land.

We hope FERC will join with us in viewing the approach reflected in this letter and the Memorandum of Understanding as a positive and significant alternative to the land use recommendation contained in the DEIS. If FERC agrees, we believe a simple condition requiring the conveyance of the easements could be contained in the licenses to be issued. It is our intention that these easements would be granted in lieu of any further project boundary expansions or shoreline land use restrictions in the nature of those described in the DEIS on any of the impoundments in either project. To the extent licenses granted by FERC meet this objective, as you can see from the enclosed Memorandum of Understanding, Great Northern is committed to granting these easements.

Thank you for your attention and cooperation and thanks to the FERC staff for its long and diligent efforts during these relicensings.

Sincercly. Aneus S

Governor, State of Maine

Donald G. McNeil, President Great Northern Paper, Inc.

RESPONSES TO GOVERNOR KING AND DONALD MC'NELL ON UPPER PENOBSCOT RIVER BASIN DEIS

1.

Cont

MEMORANDUM OF UNDERSTANDING

MEMORANDUM OF UNDERSTANDING. DATED February $\frac{1}{27}$. 1995 between the STATE OF MAINE (the "State") acting through its Governor and Great Northern Paper. Inc., a Delaware corporation having an office in Millinocket. Maine ("Great Northern")

The State and Great Northern acknowledge the existence of valuable resources in the area adjacent to the West Branch of the Penobscot River and Chesuncook Lake and that protection of these resources can best be achieved by carrying out the provisions of this Memorandum of Understanding ("Memorandum").

The purpose of this Memorandum is to assist in protecting those resources within a framework of continued use of the river corridor for timber harvesting, recreational uses, other traditional uses of the region's forest lands, and hydroelectric power generation and transmission.

1. Great Northern agrees, upon fulfiliment of the terms of this Memorandum to contribute to the State a renewable Conservation Easement, on 73 miles, more or less, of shorefront land, within 250 feet of the normal high water mark of certain areas on. Chesuncook and Ripogenous Lakes and Brandy and Black Ponds, measured as horizontal distance landward of such high water mark as shown on attached Exhibit A and to be more specifically described in the aforementioned Conservation Easement. The purpose of said Easement shall be to prohibit additional commercial and residential structures and to grant to the State the right to manage recreational activity within the 250 foot zone. The term of this Easement shall be the same as the term of any license (including any renewal thereof) issued for the Ripogenous and Penobscot Mills projects by the Federal Energy Regulatory Commission ("FERC") or its successor entity and the covenants, restrictions, exceptions and reservations shall be consistent with those set forth in the Conservation Easements granted by Great Northern Nekoosa Corporation to the State of Maine on or about August 14, 1981 (the "1981 Easements").

2. Great Northern further agrees, upon fulfillment of the terms of this Memorandum, to contribute to the State a perpetual Conservation Easement, in order to prohibit additional commercial and residential structures and to grant to the State the right to manage recreational activity thereon. Said Easement shall cover the following real estate:

a 1/2 mile, more or less, of shoreline on Lobster Lake ENR14 within 500 feet of the normal high water mark thereof, measured as horizontal distance landward of such high water mark as shown on attached Exhibit. A and to be further described in said Conservation Easement. The purpose thereof being to convey an easement on those additional lands on Lobster Lake acquired since the grant of the 1981 Easements to the State of Maine.

b. 5 miles, more or less, of shoreline on the West Branch of the Penobscot River, T4R14, within 500 feet of the normal high water mark, measured as horizontal distance landward of such high water mark as shown on attached Exhibit A and to be further described in said Conservation Easement. The purpose thereof being to convey an easement of those additional lands on the West Branch of the Penobscot River acquired in fee since the grant of the 1981 Easements to the State of Maine.

The covenants, restrictions, exceptions and reservations of this Easement shall be consistent with those of the 1981 Easements,

3. All rights reserved or retained by Great Northern in connection with the donation of said Conservation Easements are and shall remain subject to applicable requirements, regulations, and laws of state and local governmental bodies having jurisdiction thereof.

4. The State will add the proposed Conservation Easement areas to its existing recreation management plan for the 1981 Easement area. The State recognizes that because Great Northern is the owner in fee of said easement lands, as well as lands adjacent thereto, the cooperation of Great Northern is essential to the successful operation of the aforementioned recreation plan. In order to facilitate recreation management. Great Northern will grant property leases to the State. *upon terms* satisfactory to Great Northern and the State, within the easement lands, in order that administrative structures and areas as defined in the Deed of Conservation Easement, may be erected, maintained and utilized.

5. It is the intent of Great Northern and the State that land uses within the proposed easement area shall be subject to requirements no less stringent than applicable zoning and land use standards of the Maine Land Use Regulation Commission in effect at the time of execution of this Memorandum.

6. The State and Great Northern understand and agree that conveyance of the Easements described herein is intended to provide for shoreline protection and multiple use in a manner consistent with the policy and goals of the State of Maine. The parties intend that

2

the Easements be conveyed as soon as practicable following issuance of licenses by FERC for the Ripogenus and Penobscot Mills projects. Provided, however, that if any order by FERC issuing a new license for the Ripogenus or Penobscot Mills projects imposes conditions or requirements with respect to land use or shoreline protection for any shoreline areas of the impoundments of the Ripogenus and Penobscot Mills projects which differ materially from those imposed by the Conservation Easements described in paragraphs 1 and 2 hereof, either party shall have the right, within 30 days of the date of said order, to provide written notice to the other that it is withdrawing from the obligations of this Memorandum. Upon such notification, this Memorandum and all obligations of either party hereunder shall be void and withdrawn.

IN WITNESS THEREFOF, the parties hereto have duly caused this Memorandum to be executed, as of the day and year first above written.

Bv

In the presence of:

GREAT NORTHERN PAPER, INC.

Donald G. McNeil, President

In the presence of:

STATE OF MAINE Angus .S. King, Jr.

Governor, State of Maine







RESPONSES TO PENOBSCOT INDIAN NATION **ON UPPER PENOBSCOT RIVER BASIN DEIS**

Opinion noted. We have also considered comments in your letter of March 9, 1995.

The staff reviewed available information regarding the Penobscot Indian Nation's claims to lands and resources within the West Branch region. The staff concurs with the Maine State Department of the Attorney General findings that the Penobscot Indian Nation retains no ownership or title to lands within the branches of the Penobscot River (see section 4.11.1.2). The staff finds consideration of the Penobscot Indian Nation's traditional practices within the project area are outside the scope of Section 106. The staff, therefore, does not recommend that GNP include the Penobscot Indian Nation as a concurring party to the Programmatic Agreement associated with the Ripogenus and Penobscot Mills projects. The staff acknowledges Penobscot Indian Nation's interest in the management of historic properties potentially eligible for listing in the National Register of Historic Places within the project areas. Accordingly, the staff recommends that GNP should consult the Penobscot Indian Nation during the development of the revised Cultural Resource Management Plans for the Penobscot Mills and Ripogenus projects.

to an extent even greater than with the Lower Penobscot Basin DEIS, for which PIN has submitted comments, FERC has ignored its responsibility to protect the natural and cultural resources of the PIN.

LAND OWNERSHIP

There is a conflict in the DEIS between the discussion of regional land use issues in Section 3.11.1 (page 3-41) and the section entitled "PIN Claims to Islands in the West Branch" (4.11.1.2, page 4-69). In 3.11.1, the DEIS states that the applicant "owns in fee or holds flowage rights to all the project lands and most land surrounding the project area." However, the DEIS also states, in Section 4.11.1.2, that the PIN "claims islands in the West Branch of the Penobscot pursuant to terms of the 1980 Maine Indian Claims Settlement Act...GNP refutes PIN's claim to the islands within the Penobscot tributaries but proposes no actions related to this claim." The DEIS goes on to state: "To date, PIN claims remain unsubstantiated. This issue and associated impacts remain unresolved."

In fact, the PIN's claims are substantiated. The PIN's original title to the islands in the Penobscot River, including those occupied by the Ripogenus and Penobscot Mills projects, was recognized as early as the 1796 and 1818 treaties with the State of Massachusetts. Though the treaties lacked congressional approval, PIN's title and rights under them were affirmed with the enactment of the Maine Indian Claims Settlement Act (MICSA, 25 U.S.C. § 1721 <u>et.geq.</u>) which approved the Act to Implement the Maine Indian Claims Settlement (30 MRSA § 6201 <u>et.geq.</u>). MICSA is a modern treaty; it is the product of government-to-government negotiations between PIN and the State which the United States approved as trustee. MICSA confirmed the Penobscot Indian Reservation which consists of Indian Island and islands in the Penobscot River upstream thereof. Among the rights reserved to the PIN by MICSA was the right to fish free of State regulation for sustenance purposes in the waters of the Penobscot River that lie within the Reservation boundaries (30 MRSA § 6207(4)). In consideration of those rights, PIN gave up its claims to a substantial portion of the State of Maine, including extensive land holdings occupied by the apolicant, except those lands reserved by treaty.

However, in this proceeding, FERC is barred from hearing or disposing of any challenge by the applicant to PIN's ownership of the Penobscot River islands occupied by the Ripogenus and Penobscot Mills projects, or PIN's fishing rights. Section 17(a) of Public Law 99-495 (Electric Consumers Protection Act) provides that, with respect to each license, permit, or exemption issued under the Federal Power Act after October 16, 1986, FERC shall have no authority under the Federal Power Act to "alter, amend, repeat, interpret, modify, or be in conflict with, the Treaty rights or other rights of any Indian tribe." The conference report on this 1986 act makes it clear that the act applies to all Indian rights and not just to the protections created by sections 4(e) and 10(e) of the Federal Power Act.

With respect to Indian rights, this legislation does not affect or modify any treaty or other right of an Indian tribe. Additionally, nothing in this legislation is intended to affect or modify any existing protections under sections 4(e) and 10(e) of this Act.

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-3 Section 17(a) of the Electric Consuments Protection Act (ECPA), P.L. 99-495, is a savings provision. It does not grant or deny the Commission any power or authority, but states that no provision of ECPA shall be considered to "alter, amend, repeal, interpret, modify, or be in conflict with, the Treaty rights or other rights of any Indian tribe."

House Conference Report No. 934, 99th Cong., 2d Sess. 33 (1986). reprinted in 1986 U.S.C.C.A.N. 2496, 2550.

Whatever claim the applicant asserts against PIN's rights under MICSA regarding the islands occupied by the Ripogenus and Penobscot Mills projects is derived from the settlement negotiated by the State on its behalf. As an entity created pursuant to State law and whose property rights are subject to State law, the applicant should be required by FERC in the first instance to present its dispute with PIN through the State representatives to the Maine Indian Tribal-State Commission (MITSC), established by Section 6212 of the Maine Implementing Act. Section 17(a)(5) of Public Law 99-495 requires that this threshold procedure, along with any other proceedings that may be authorized to resolve disputes over PIN's rights, be exhausted as a precondition to the insuance of a license (if one is to be issued at all). The Section 17(a) prohibition also creates a bar to FERC's insuance of a license for the purpose of enabling the applicant to assert eminent domain authority under Section 21 of the Federal Power Act (16 U.S.C § 814), against PIN's interest, because issuing a license under that circumstance would be "in conflict with, the Treaty rights or other rights of any Indian tribe."

The applicant has also contended that the Penobscot Reservation is not a reservation as defined in the Federal Power Act. To support this contention, it cites <u>Federal Power Complication v.</u> <u>Tuncarora Indian Nation</u>, 362 U.S. 99 (1960), in which it was held that lands purchased and held in fee simple by the Tuncarora Indian Nation were not a reservation under the Federal Power Act. We believe that there are substantial reasons why that precedent is inapplicable to the Penobacot Reservation, but also believe, in view of Section 17(a)(5) of Public Law 99-495, that they are not subject to adjudication by FERC. The PIN also adheres to its claim of ownership of the islands occupied by the Ripogenus and Penobacot Mills projects, and points out that Section 17(a)(5) of Public Law 99-495 prohibits FERC from interpreting PIN's rights in that regard as well.

It should also be noted that, since the applicant's submission on July 8, 1993 (FERC Reference: GNP 1993(c)), there have been numerous contacts between the applicant and the PIN. The PIN's ownership of the islands occupied by the projects need have no adverse effect on the applicant's interest, and it has communicated those views to the applicant. It would be a loss of tragic proportions for FERC to indulge the applicant's unfounded fears over its operational needs at the expense of PIN's Reservation islands in which the applicant appears to have no interest beyond their location within project boundaries. In view of the prohibitions in Section 17(a)(5) of Public Law 99-495, and the firm belief by PIN that reasoned negotiation with the applicant could produce a satisfactory resolution of this conflict, FERC should advise the parties that no license will be issued until the parties have either negotiated a settlement or adjudicated their differences in an appropriate forum. In the meantume, FERC can issue annual licenses for the projects pursuant to 16 U.S.C. Section 808(a)(1) until the matter is resolved.

FISHERY RESOURCES

Staff's approach to fishery resource management is to assume a static condition. As Staff correctly points out in Section 3.5.1, the upper Penobscot River, including the project areas

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

- PIN-4 See response to PIN-3. Section 17(a) of ECPA does not mandate any particular procedure for resolving disputes over title to land or treaty rights.
- PIN-5 See response to PIN-3
- PIN-6 Opinion noted.
- PIN-7 Opinion noted. We added text to the affected environment and impact sections that better describes the American eel resources in the project area and the effects of project operation on eels. We conclude that the enhancements we incorporated into Alternative 2 will benefit the American eel population to some degree.

PIN-4

PIN-5

before much of the habitat was destroyed and fish migrations were terminated by the construction of numerous dams without fish passage, historically supported populations of anadromous fish species. Restoration of these species in the Penobscot River drainage is a PIN goal, and habitat expansion in the upper basin, through restoration, rehabilitation, and enhancement, is considered necessary for the future realization of this goal.

The upper basin also represents a potentially significant habitat resource for the production of catadromous American eels, which, while present in the project areas, are in serious decline throughout the Penobscot River drainage. Historical declines in the numbers of young eels reaching upper production areas and the number of adults reaching the estuary, though to an unquantified level, are likely due in large part to the construction of numerous dams in the drainage without adequate upstream and downstream fish passage, and in the case of these projects, dam construction without any upstream (except North Twin) or downstream fish passage facilities. The DEIS inadequately addresses past and continuing adverse impacts to eel populations in the Penobscot River, as well as impacts resulting from Staff's recommended licensing alternatives for these projects. Mitigation in the form of modification of project structures and operations, possibly similar to that required for Great Bear Hydropower, Inc., at its Columbia Dam Project (FERC #8396-013), must be considered for the Ripogenus and Penobscot Mills projects.

INSTREAM FLOWS

The PIN lands contained within the main river channel below Stone Dam must be sufficiently watered on a permanent basis to permit travel by canoe (navigation), meet all other applicable Federal and State water quality standards, and support native and indigenous fish species on a sustainable basis, in habitat where they once flourished, and in sufficient abundance, size, and quality to provide for the opportunity for a sustained subsistence harvest by PIN tribal members.

The DEIS, in Section 3.5.2.12, recognizes that potential habitat in this reach of river is extensive. Elsewhere, the DEIS points out that Staff-recommended flow releases will not permit this reach to meet applicable water quality standards, including navigability and habitat for indigenous fish species. These overt violations regarding tribal fishing and navigation rights, and general water quality law, cannot be permitted.

We note that Staff has continued to utilize a water allocation scenario for the reach below Stone Dam which depends on bringing additional water from sources further up in the drainage. Staff then advances the possibility that such a scenario may adversely impact other water-use related enhancements proposed by the applicant or Staff in the project areas, and generally casts flow allocation to this reach in a negative light with respect to overall water-use related enhancement in the project areas. The alternative of providing flow by a direct split of that inflow which would otherwise arrive at Stone Dam, as a result of whatever other upstream water management practices that are in place, continues to be ignored, despite the PIN clearly pointing out this viable alternative in our scoping comments, and in other filings with FERC on these projects.

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

- PIN-8 See response to PIN-2.
- PIN-9 Opinion noted. The legal status of the Back Channel relative to Maine's 401 WQC will be addressed in the order for the project.
- PIN-10 We agree that the 2000 cfs minimum flow required at Millinocket by the WQC and the state charter could technically include flows provided to the Back Channel, as we explain more fully in revised Appendix D (section D.3.3) of the EIS. The minimal resource benefit of providing flow to the BC is the primary factor in determining its feasibility.

PIN-9

PIN-8

PIN-7

Cont

Until Staff seriously evaluates this water allocation alternative for the reach below Stone Dam **PIN-10** the EIS remains conceptually flawed and entirely inadequate with respect to this instream flow issue Cont WATER QUALITY The DEIS fails to adequately address the cumulative impact of project impoundments, project area thermal discharges, and the loss of radiational cooling due to water diversion away from **PIN-11** natural free-flowing river channels, on the summertime temperature of the water emerging at the lower end of the Penobscot Mills project area, despite the PIN requesting this analysis in our October 4, 1993 scoping comments. We originally requested the analysis because of the critical relevance of human-induced warming of the Penobscot River to the Atlantic salmon restoration program currently underway in the Penobscot drainage. This EIS must fully address these issues, and include recommended conditions for the mitigation of cumulative water temperature impacts due to continuing project operation. In addition, mitigative measures at the Dolby impoundment to enable this water body to meet

PIN-12 In addition, mitigative measures at the Dolby impoundment to enable this water body to meet applicable water quality standards have not been adequately considered in this DEIS.

MERCURY CONTAMINATION OF SEDIMENTS, WATER, AND BIOTA

The impact of historic use of mercuric compounds in the pulp and paper industry, and the impact of extensive annual impoundment drawdowns, or project operation in general, on mercury contamination levels and dynamics, have been irresponsibly diminished or dismissed in this DEIS. Beyond the critical pertinence of these issues towards the safe consumption of fish and wildlife resources from the project areas by tribal members, the impacts of mercury contamination in project sediment, water and biota may be detrimentally impacting bald eagles, which are, much like, Atlantic salmon, culturally significant to the PIN.

BIODIVERSITY

The DEIS does not address, nor even mention, the cumulative impacts of existing project operation, the applicant's or Staff's proposals for relicensing, or other licensing action alternatives identified by PIN and others, on aquatic biodiversity in the project areas, or in the Penobacot River ecosystem as a whole. The PIN requested this analysis in its scoping comments on this EIS, and this issue is considered pertinent in the recently issued DEIS for the lower Penobscot River project licensing actions. The issue of the extent to which dam construction, impoundment drawdowns, and flow regulation/diversion has adversely impacted aquatic biodiversity is just as important in the upper drainage as it is for the lower river basin, because the underlying objective of maintaining or restoring biodiversity is long-term ecosystem health (i.e. the whole river system). Until this issue is fully addressed within Staff's proposed licensing actions for these projects, including a discussion of potential mitigative measures, this DEIS is considered inadequate.

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

- PIN-11 Your request for the EIS to address the cumulative effect of the projects on water temperature of the Penobscot River downstream of the project area is not reasonable. FERC considers the baseline condition to be the existing projects; your request assumes that preproject conditions should be the baseline. You have suggested no specific reasonable alternative that could be evaluated to mitigate for a presumed effect on temperature. In fact, summer temperatures are probably cooler throughout much of the West Branch basin because of discharge of cooler hypolimnetic waters from the projects' larger impoundments.
- PIN-12 You did not suggest specific mitigation for the Dolby impoundment. No evidence shows that project operations affect dissolved oxygen (DO) in Dolby Pond. DO problems resulting from BOD loads from the mill are not within FERC jurisdiction; this issue will be discussed further in the license order for the project.
- PIN-13 FERC required additional studies to address whether project operations are causing an increase in mercury bioaccumulation in biota inhabiting the project waters. We reviewed the results of this study and concluded that the data do not support the contention that fluctuating water levels are affecting the bioavailability of mercury in drawdown reservoirs. See our response to EPA's comments EPA 9-11.
- PIN-14 Opinion noted. The DEIS for the lower Penobscot River considered biodiversity more fully because that DEIS included a new, unconstructed project. In the upper Penobscot River DEIS, the projects have existed for as long as 100 years, and there are no unique or unusual habitats in the area that appear to require additional enhancements beyond those proposed by staff.

PIN-14

PIN-13

E-197

5

CULTURAL RESOURCES

In Section 3.1, Staff correctly concludes that "The region is also home to the Penobscot Indian Nation (PIN), much of whose cultural heritage is closely associated with the river and the resources it provides." However, in conflict with this statement, the discussion in Section 3.13 conveniently but inaccurately writes PIN out of the history of the upper Penobscot River after the late 18th century. We have always used these lands and waters, still do, and will continue to do so in the future.

Sixty-one aboriginal sites are reportedly eligible for listing on the National Register of Historic Places, and we have not been consulted. These were our homes and hunting and fishing camps. A Programmatic Agreement for the management of these historic places is recommended in Section 4.11.1.1 and PlN is not included. The Programmatic Agreement must include the PIN.

<u>CLOSING</u>

This concludes our initial comments on the DEIS for the upper Penobscot River Basin, and serves to alert the Commission and its Staff as to the <u>primary</u> areas (but, not necessarily all areas) where we intend to provide additional or detailed comments, to be filed with the Commission by March 10, 1995. Please feel free to contact me (207-539-8219) should you have any questions.

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-15 Phase I and II archeological surveys and SHPO's review identified 7 archeological sites within the Penobscot Mills project area and 15 archeological sites within the Ripogenus project area potentially eligible for listing in the National Register of Historic Places (see section 4.11). Although the staff finds that consideration of the Penobscot Indian Nation's traditional practices within the project area are outside the scope of Section 106, the staff acknowledges the Penobscot Indian Nation's interest in the management of these historical properties. Accordingly, the staff recommends that GNP consult the PIN during the development of the revised Cultural Resource Management Plans. The staff, however, does not recommend that GNP include the PIN as a concurring party to the Programmatic Agreement.

PIN-16 No response required.

Sincerely.

Paul Bisulca

Asst. to the Governor on Environmental Affairs

Distribution

FERC Service Lists #2458; #2572

E-198

PIN-16

Office of the Governor and Council Richard H. Hamilton Governor Arnold E. Neptune

Li. Governor Priscilla Attean Representative

Cammunity Building Indian Island Old Team, Maine 64468 (207) 827-7776 FAX: (207) 877-8642

March 9, 1995

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, N.E. Washington, DC 20426

RE: FERC PROJECTS #2458 AND #2572; COMPLETION OF PENOBSCOT INDIAN NATION'S COMMENTS AND RECOMMENDATIONS ON THE DRAFT ENVIRONMENTAL IMPACT STATEMENT FOR THE PENOBSCOT RIVER BASIN IN MAINE

Dear Secretary Cashell:

The Penobscot Indian Nation (PIN) submits the following supplemental comments on the Draft Environmental Impact Statement (DEIS) for the Penobscot River Basin. Maine, issued by Federal Energy Regulatory Commission Staff ("Staff") to intervenors in December, 1994 (dated. however, November 1994). The original comment deadline for this DEIS was February 8. 1995. By letter dated January 18. 1995, the PIN requested a thirty (30) day time extension. until March 10. By notice dated February 2, FERC emtended the deadline to February 22, 1995. The PIN filed initial comments within this extended deadline by letter dated February 21, 1995. identifying major issues for which we would be submitting detailed comments, and indicating that we needed the additional time to complete this work and would file our completed comments with the Commission by March 10, 1995.

These comments are provided in <u>supplement to</u>, not in substitution for, the comments we filed with the Commission on this DEIS on February 21, 1995. These comments expand on our initial discussion contained within that filing of the higher priority natural resource issues of concern to the PIN in the project areas, including water quality standards and flows in the Back Channel, mercury contamination in project waters and biota, biodiversity, and anadromous and catadromous fisheries impacts, management and rehabilitation/restoration.

INTERVENORS "DISREGARD" FOR APPLICANT'S ECONOMIC VIABILITY

PIN-18 On page 2-21 (last paragraph), the DEIS states that "Alternative 1 includes measures proposed by the CI and other parties to <u>maximize</u> or substantially increase benefits for various resources, particularly fisheries. <u>without regard to economic effect for GNP</u>".

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-17 No response required.

PIN-18 Opinion noted.

E-199

PIN 17

We assume PIN is included among "others parties". While on the surface this may seem a rather innocuous statement, it is anything but that, and the PIN takes very strong exception to this characterization. First of all, the portion of Staff's statement indicating disregard for the applicant's economics demonstrates that Staff is unfamiliar with the objectives and content of the innumerable consultation meetings, negotiation sessions, comment filings, hearings, and other interactions that CI. PIN and others have had with the applicant over the past several years.

Obviously, if PIN and others were proposing changes in project operation "without regard to

economic effect", we would not be even remotely considering allowing the applicant to continue with several of the most substantial, and most adversely impactive, water use components of historical and proposed project operations. Included among these are its extensive impoundment

PIN-18 Cont

> drawdowns, artificially high flows in the West Branch below McKay, or diversion of the overwhelming bulk of available water (a public resource) at Stone Dam away from the natural river channel for private power production purposes. And if we were seeking to maximize fisheries and water quality enhancement without regard for the applicant, we would be calling for selective removal of some of the project dams, since this would be the most beneficial scenario towards our long term resource management goals in the Penobscot River drainage. Beyond this mischaracterization of the consultation process to date. Staff's statement also reflects

> an overall deficiency with respect to understanding the need to balance developmental and nondevelopmental uses of a waterway. The West Branch is currently entirely out of balance with respect to these competing uses. With the exception of the reach between McKay and North Twin impoundment, nearly the entire remaining length and gradient of the West Branch, between the head of Ripogenus impoundment and its confluence with the East Branch, at Medway is currently impounded. In addition, the small remaining reach of potential free-flowing habitat available in the Penobscot Mills Project area, the Back Channel, has been and continues to be relegated to a non-functional (ecologically) dryway.

What PIN and others seek pales in comparison to the privileges that the applicant has enjoyed in the past, and will continue to enjoy for the foreseeable future, even if every one of the most impactive (to the applicant) water use restrictions under consideration are implemented. What PIN and others hope to achieve is the restoration of the lost natural functions and values that the West Branch represented prior to the applicant assuming control of water use in the drainage.

ADEOUACY OF RESOURCE IMPACT ANALYSES

The PIN requested a number of cumulative and site-specific impact analyses in our EIS scoping comments. Staff has addressed only very few of these, and has omitted the remainder from further consideration. While it is cumbersome to identify each and every instance where these omissions have occurred, we would point out that several of the most important ones towards addressing our concerns in these proposed licensing actions were among those omitted. including:

1. Cumulative impacts of project existence and operation on water temperature in the West **PIN-22** Branch, particularly at the point where water leaves the project area and approaches its

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

- PIN-19 Opinion noted.
- PIN-20 Opinion noted.
- PIN-21 Opinion noted.
- PIN-22 See response PIN-11.

	confluence with East Branch waters
PIN-23	2. Cumulative impacts of project existence and operation on biodiversity in the West Branch.
PIN-24	3 Cumulative impacts of project existence and operation on mercury deposition and bioaccumulation in project area sediments, water, and biota (the DEIS does provide some <u>site-specific</u> discussion of this issue, but subsequently dismisses it from consideration within Staff-recommended licensing actions, and provides no <u>cumulative impact</u> analysis at all).
PiN-25	4. Cumulative impacts of project existence and operation on American cel populations in the West Branch and in the entire Penobscot River basin (see additional comments below under <u>AMERICAN EEL IMPACT ANALYSIS</u>).
PIN-26	We continue to recommend that these analyses be conducted and included in the EIS. We also recommend that Staff revisit our scoping comments and identify additional EIS requests which we made but which were omitted by Staff, and either include the requested analyses in the EIS, or provide a detailed explanation in the EIS as to why these were not included.
	PIN'S CHOICE NOT TO APPEAL STATE WATER OUALITY CERTIFICATIONS
e-20 Pin-27	We frequently see comments to the effect that PIN's (or others) failure to appeal a State Water Quality Certification reflects an acceptance of the conditions contained therein. At least in the case of PIN, this should not be assumed, and in terms of the specific projects at hand, this is absolutely not true. It should be readily apparent from our difficulty in meeting some of FERC's filing deadlines that PIN is of limited staffing and financial resources, and must carefully choose how to commit those resources. We also understand the political forces at play in State Water Quality Certifications and are unwilling to squander resources on lost causes.
	It should also be pointed out that the State has no fiduciary responsibility to PIN, whereas the FERC does. Therefore, our expectation is that FERC, not the State, will afford proper consideration of the impact of hydropower operations on a federally-recognized tribe such as PIN. In short, we believe that our time and effort is better spent presenting our arguments before FERC than before any State proceedings, which, in the case of these projects, are heavily

dominated by the Paper Industry.

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-23 See response PIN-14.

- PIN-24 Although we have found no evidence that project operations contribute to mercury mobilization in the project area, we nevertheless concurred with the 401 WQC requirement for the GNP conduct studies on mercury in the Ripogenus Project area and recommend GNP cooperate with MDEP and EPA in mercury studies that those agencies conduct in the areas of those projects.
- PIN-25 Our baseline for evaluating project alternatives is existing conditions. Thus, we have not addressed the issue of how the existence of the project may be affecting an eel stock that might exist in a no-project, no-dam scenario. We have, however, expanded our discussion of eels in section 3 of the FEIS.
- PIN-26 Opinion noted.
- PIN-27 Opinion noted.

3

SPECIFIC COMMENTS

1 BACK CHANNEL

A Instream Flow Requirements and Potential Natural Resource Values

Legal Requirements

in the BC should be challenged.

On page 4-13. Staff concludes that Class C water quality standards for designated uses are unlikely to be met under the leakage flows proposed for the Back Channel (BC) by the applicant and Staff. While navigation is identified as one designated use which would not be met, the DEIS fails to identify additional designated uses, such as habitat for indigenous fish species and other aquatic life, that will also not be met under leakage flows. We also doubt that the BC at leakage meets dissolved oxygen standards during the warm summer months. The applicant's additional study of this aspect was inadequate to determine this, and oxygen measurements that were made were not taken early enough in the morning (i.e. 0600-0700 hours), when DO levels are expected to be lowest.

Staff then incorrectly concludes that there is "no apparent demand for navigation in this river reach". As we discuss in more detail below, the PIN identified the need for navigable flows for tribal transportation in the BC as early in this process as November of 1990. Finally, Staff concludes that there is "no compelling basis for challenging the state's decision to waive certification for Backchannel".

While we strongly question why these direct violations of State and Federal water quality laws are not "compelling" enough for Staff to seek correction of DEP's poor judgement. Staff has also apparently ignored other major statutory requirements. These include:

1. FERC's fiduciary responsibility to protect the PIN Reservation and natural resources.

PIN-29 2. FERC's clearly defined responsibility, as contained in the ECPA amendments to the FPA Section 10(a), to consider the "recommendations (including fish and wildlife recommendations) of Indian tribes affected by the project" (i.e. proposed licensing actions).

3. FERC's equally binding requirement, contained in ECPA amendments to Section 4(e) of the FPA, to give "equal consideration to the purposes of energy conservation, the protection, mitigation of damage to, and enhancement of, fish and wildlife (including related spawning grounds and habitat), the protection of recreational opportunities, and the preservation of other aspects of environmental quality", in deciding whether to issue any license.

As we discuss in more detail below, Staff has also mischaracterized, underestimated, or ignored significant potential environmental benefits of increased flows in the BC. Had these aspects been comprehensively and objectively assessed and discussed, the resulting analysis would have added to the already strong foundation upon which DEP's abandonment of their legal responsibilities.

4

PIN-30

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-28. The staff reviewed available information regarding the Penobscot Indian Nation's claims to lands and rights within the branches of the Penobscot River and concurs with the decision of the Maine State Department of the Attorney General (see section 4.11.1.2). The staff finds that consideration of the Penobscot Indian Nation's traditional practices within the project area are outside the scope of Section 106. The staff, therefore does not recommend including the Penobscot Indian Nation as a concurring party to the Programmatic Agreement associated with the Ripogenus and Penobscot Mills projects. The staff acknowledges the interest of Penobscot Indian Nation's interest in the management of historic properties potentially eligible for listing in the National Register of Historic Places within the project areas. Accordingly, the staff recommends that GNP consult the Penobscot Indian Nation during the development of the revised Cultural Resource Management Plans for the Penobscot Mills and Ripogenus Projects.

PIN-29 See response PIN-3.

PIN-30 Opinion noted.

PIN-28

E-202

Navigation

The DEIS summarily ignores the repeated requests by the PIN for navigable flows in the BC. We know this to be the case, by comparing comments filed to date with the applicant and the Commission, and relevant consultation meeting minutes, with Staff's incorrect conclusion on page 4-12 that "there is no apparent demand for navigation in this river reach". The PIN was excluded by the applicant from the Initial Stage Consultation, including numerous agency consultation meetings on BC issues, and was excluded from participating in the instream flow/habitat study conducted on the BC. Had we been included in these early proceedings, we would have certainly raised this issue at those times.

In our filings of November 15, 1990 (to Carson, GP) and February 27, 1992 (to Cashell, FERC), we admittedly, and pethaps mistakenly, characterized our need for navigation flows as "recreational canoeing" flows, or, even more generally, flow allocation as it would relate to compliance with Class C water quality standards, including all designated uses. However, as early in this process as February, 1991, the PIN clearly, and unequivocally, identified to the applicant the importance of this issue to the PIN (see meeting minutes from consultation meeting held between PIN and the applicant on February 4, 1991). Subsequently, in filings with the Commission dated May 21, 1993 (Recommendations for Final Terms and Conditions) and October 4, 1993 (Comments on Scoping Document 1), the PIN again clearly raises the importance of this issue. Given the frequency at which we have raised this issue, we can only assume that Staff has chosen to ignore our comments in the preparation of this DEIS.

In addition, we find that Staff has confused legal mandates to provide for navigation with enhancement of whitewater boating, has failed to require the applicant to conduct the necessary navigation flow studies to the level of adequacy which PIN, and initially the Commission itself, had requested of the applicant, and leads the reader to believe that it has independently studied navigation flows as part of this DELS preparation, when it has not.

Staff's confusion with the meaning of navigation (i.e. not a whitewater rafting enhancement. but rather a requirement to meet water quality law and Federal fiduciary responsibility to the PIN) is clearly indicated by statements, on page 4-14, that "Flows as high as 350 cfs still would not provide sufficient water for mavigation of the BC (see section 4.8); consequently, this flow regime probably would not be sufficient to meet all Class C water quality criteria...", and on page 4-48, "FERC decided not to evaluate the potential for whitewater boating in these two streams [referring to Millinocket Stream as well] in this DEIS because the significant flows that would be required to make these streams <u>navigable</u> [emphasis added] would jeopardize other environmental enhancements (FERC, 1993c)".

Clearly, navigation/transportation by cance, and whitewater rafting, are relatively exclusive of each other, in terms of purpose, in terms of the legal basis for addressing them, and in terms of the amount of water that would be needed to accommodate each of them. While we disagree with Staff's conclusion that there would be no desire on the part of whitewater enthusiasts for expansion of raftable water into this reach of the West Branch, we are only concerned within this context with flows adequate to allow passage up and down by cance with a minimum of

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-31 See response PIN-28

PIN-32 See response PIN-28

portage Since neither the applicant nor Staff has adequately studied the flows that would be needed for <u>navigation</u> (see next paragraph), we find Staff's negative conclusion concerning whether 350 cfs, or any other reasonable flow allocation in the BC, would provide for <u>navigation</u>, as unfounded.

The applicant has failed to provide adequate study of the issue of navigation in the BC, despite PIN's request to do so, and despite the Commission's order of June 8, 1992 (see Item 2 of FERC's AIR to the applicant) requiring that the applicant conduct additional study on the BC. including an investigation as to how recreation and navigation criteria would be met under their flow allocation proposal. Unfortunately, while the Commission ordered these investigations, it never conducted the proper follow up to assure that the "study" done by the applicant met the objectives of the PIN or FERC. This failure to follow up and require adequate study occurred despite the PIN pointing out the inadequacy of the applicant's response to the AIR in our comment letter dated May 21, 1993. The resulting "assessment" of this issue by the applicant was limited to one sentence. "As indicated by the enclosed video, the BC is not navigable by cance" (applicant's response to AIR Item 2, dated 12/07/92). Since PIN was never provided the referenced video tape, despite requesting it from the applicant, we must assume that it was a tape of the BC at leakage, showing, obviously, that the reach was not canocable at that flow. We see no other evidence in the applicant's response, not in the application, that the objectives of PIN's repeated requests for mavigation flow study, or FERC's AIR, were ever met by the applicant. Certainly, the referenced video tape would not adequately address the pertinent objectives, since we know that the BC would become navigable by cance at some flow within the range that has been discussed for fisheries and other purposes in these proceedings.

Lastly, the DEIS leads the reader to believe that Staff conducted an independent study of this issue (although, as pointed out above, Staff has confused navigation with whitewater rafting). when in fact they have not. Referring to the statements quoted four paragraphs above, note that the first statement refers to section 4.8 (implying that more explanation is provided there). The only point in section 4.8 where this issue is mentioned is in the second quoted statement in the above-referenced paragraph, which cites "FERC 1993c", again implying that additional analysis or study is available from that citation. FERC (1993c) is Scoping Document 2 (SD2) for these projects. We reviewed SD2 in detail again. Page 21 (4.7) identifies whitewater boating in the BC, and page 20 (4.3) identifies compliance with water quality standards (which include designated uses) in the BC, as site-specific issues to be addressed in the DEIS. However, we note that, at the only other point in the document where whitewater boating is identified (FERC response under Item 6.10, page 32). Staff contradicts its obligation set forth in 4.7, by summarily dismissing the potential for whitewater boating enhancement, despite never having studied it. And in the only other section of SD2 where navigation flows could be addressed, namely, under 6.6 - Water Quality, we find that, despite the commitment that FERC sets forth to address PIN's and others' contention that leakage flows proposed for the BC are inadequate to meet applicable water quality standards (including designated uses), the DEIS fails to even mention, much less address to an adequate level, navigation flows in the BC.

Thus, the situation with respect to either navigation or recreation issues in the BC is that Staff has essentially no pertinent information to base an informed decision on, and has chosen instead

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-32

Cont

to neglect its responsibilities regarding these issues, and to falsely portray its conclusions as being based on some internal or independent analysis, which in fact was never conducted.

These inadequacies with respect to the issue of navigation in the BC must be fully addressed and resolved in the EIS.

Fisheries and Fisheries Management in Backchannel

On page 3-20, Staff recognizes that potential fish habitat in the BC is extensive. On page 4-28, Staff concludes that species other than salmon, including brook trout, would benefit from flow increase to the BC. Inexplicably, however, Staff fails to follow up these potential benefits to any meaningful level in the DEIS. Instead, Staff clings desperately to its contention that the only possible fisheries value from increased flows in the BC that it should have to address is that for landlocked salmon. This is a very narrow-minded approach, and demonstrates the strong tendency evident throughout this DEIS to avoid devoting any independent, insightful, or creative thought to these issues. In addition to mischaracterizing the potential for landlocked salmon management in this reach, the DEIS also fails to adequately consider other potential fishery management to fisheries management for a zone of river where direct prior experience or empirical data is unavailable. We address each of these issues separately, starting with general conceptual fisheries enhancements.

<u>General Concepts.</u> Both the applicant and Staff have relied heavily on the robust landlocked salmon fishery that is currently supported in the West Branch below McKay station to rationalize resisting change in the flow regime for that reach. The applicant routinely has utilized the quality of this fishery to exempt it from having to consider other flow/fisheries management options below McKay which could impact its hydropower generation. While we do not dispute the quality of this fishery, the DEIS fails to recognize that this fishery developed purely by chance, as a direct result of decades of operation of the applicant's projects without flow and drawdown restrictions, and their overriding objective of maximizing annual hydropower generation from West Branch waters. There was no State management plan in place nor any significant management activities or monitoring by either the State or the applicant until comparatively recent times.

While the situation below McKay appears to have worked out to the mutual benefit of both the State's and the applicant's respective desires for landlocked salmon fisheries and hydropower, this by no means indicates that the current flow regime is optimal for salmon populations and fisheries, only that this fishery is of relatively high quality when compared against the statewide average. The point here is, that there were no predetermined fisheries goals either sought or required when the applicant began its artificial manipulation of flows in the West Branch decades ago. Instead, the projects have been operated for decades to maximize hydropower generation, which has resulted in the sacrifice of most of the resident brook trout fisheries that would have been available in this reach, in lieu of benefits toward salmon populations. As the West Branch drainage became more accessible and anglers became more affluent, they began arriving in ever

7

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-33 Opinion noted.

PIN-34 Opinions noted. Staff considers the applicant's proposed management of project waters, including the West Branch below McKay Station and the Back Channel to be the most balanced, reasonable alternative; we also conclude that their proposed measures are entirely consistent with state fisheries management plans from which staff take guidance in establishing enhancement measures and supported by the state fisheries agency (MDIFW). Therefore, we have not recommended additional modeling of the West Branch or enhancements in the Back Channel.

PIN-33

PIN-32

Cont

increasing numbers. They "discovered" the high quality salmon fishery that had developed there. Today, at the first real opportunity for the public to evaluate past project operations in relation to environmental protection or enhancement, the existing salmon fishery in the West Branch below McKay becomes worthy of the utmost level of protection in the eyes of those users who have become accustomed to its high quality. We do not question the validity of this perception by others.

What we do question is the tendency in this DEIS and similarly in the DEIS recently issued for the lower Penobscot River hydro project licensings (which was prepared by some of the same consultants which prepared this DEIS, including the lead fisheries consultant and deputy project manager), for Staff to presume that fishery management goals and objectives, and even year-toyear population sizes, must be firmly established for the entire next licensing term (i.e. 30-50 years), before Staff can assess such potential within its proposed licensing actions. This approach does not provide for any flexibility, or opportunity for experimentation and reevaluation, and thus we find this approach unrealistic and contrary to the way fisheries management actually occurs.

We believe that the situation with the BC, having been relegated to dry way for so many decades, dictates that considerable flexibility be afforded to the managers in terms of developing a fishery management scheme for that river reach. This is particularly true considering that we (collectively) lack empirical water quality, aquatic life, or fisheries data collected under a long-term adequate flow regime in the BC, to use as a baseline. Instead, we are faced with an existing condition and historical operating scenario that greatly constrains the level of certainty as to what the fisheries management options in the BC could or would be, once a stable flow is established during periods of non-spill. Had Staff addressed the BC instream flow issue with objectivity, it should have realized this.

Furthermore, if Staff had devoted the level of thought to this issue that we believe is necessary to fulfill its "equal consideration" mandates, it would have discovered that these management "constraints", instead of representing obstacles to be pitted against potential but uncertain fishery management objectives, actually represent a rare opportunity for resource managers, the applicant, and the Commission to become innovative and creative in developing fisheries management options in this reach. It is indeed unusual for managers to come upon a situation where there are no concrete expectations; most would be excited about the opportunity to have such a situation at their disposal. Again, Staff must recognize that DIFW's abandonment of the instream flow issues and fisheries management in the BC during the latter stages of consultation was not a biological decision.

Landlocked Salmon. In trying to support its pessimistic prognosis regarding the potential for a landlocked salmon fishery, Staff has relied exclusively on the data from the instream flow study, the applicant's subjective interpretation of those data, and the applicant's subjective opinion of the overall fisheries potential of the BC. Staff has not conducted any independent onsite evaluation, and it is doubtful that Staff has even seen any part of the BC when it contained flows in the range that have been considered for fisheries development in that reach. In employing this approach. Staff has drawn several unsubstantiated or erroneous conclusions, has

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-35 in our evaluation of the fisheries production potential of the Back Channel, we took into account all available data and study findings, including the results of the IFIM study in which both state and federal fisheries agencies participated. Species selection for that study was done in consultation with the fisheries management agencies, and we note that brook trout was not one of the species designated for consideration. The results of that report were reviewed and commented on by those agencies. We do not interpret those study findings as representing simply the "...applicant's opinion ... " but as scientifically valid characterizations of the existing habitat. We have no data or information by which to validate your suggestions that the existing habitat may change in response to changes in flow regimes, and must thus consider it to be unsupported speculation. Our review of the record does not show that DOI or MDIFW raised that issue during the IFIM study. Our argument concerning the possible displacement impacts from spillage relate to the short length of the Back Channel above Grand Falls. While we have revised our text to reflect the fact that adult salmon might be able to ascend Grand Falls, it is not likely that young life stages would be able to do so. The Back Channel would also not benefit from any recruitment through displacement into the existing nursery habitat of young fish produced upstream, since no salmon production occurs above Stone Dam. Thus, the conditions in the Back Channel are guite different than those in the West Branch below McKay Station, where 27 miles of free flowing river occur, and movements of all life stages within this river reach are relatively unrestricted. Regarding predation, all existing data support our conclusion that Dolby Pond supports substantial populations of pickerel and smallmouth bass. Such population densities would not exist in a free flowing river system, and thus the level of potential predation downstream of Grand Falls is much greater than would occur in a natural riverine environment. Finally, we evaluated the need for flows in the Back Channel within the context of fisheries objectives established by the applicable fisheries management agency, MDIFW. FERC is not a fisheries management agency, but only seeks an appropriate balance among principal resource values within a waterway. We

PIN-34 Cont

failed to consider all viable alternatives for providing such a fishery, and has unjustifiably diminished the value of such fisheries.

Apparently. Staff defers to the applicant's opinion in contending that there is only a "very small amount" of potential adult salmon habitat in the BC (pages 4-29, 4-31). Based on the rather limited scope of the instream flow study, this might be a valid contention (PIN was not invited to participate in the flow study, and therefore cannot verify the applicant's conclusions from that study). However, the PIN participated in an on-site visit over the entire length of the BC at a flow of 50 cfs. While this flow was inadequate for adult salmon, we observed at least three long, deep ledge pools in the last quarter to half mile of the BC. We do not believe that any of these pools were included in the applicant's instream flow study area, however, with proper flow, these pools appear to have good potential for adult salmon readily move up and down throughout the West Branch below McKay, depending on season and food availability, spending some time in virtually every habitat type available (including deadwaters/flowages in the winter), we see no reason why the same naturalized scenario could not occur in the BC. Just because the preponderance of adult habitat for the life history purpose it would serve.

Similarly, Staff contends that spawning habitat is lacking in the BC. In doing so, Staff demonstrates that it is unfamiliar with where such habitat occurs, and what causes it to occur. First of all, spawning habitat for salmonids in general, at least in Maine, is almost always contained in localized pockets, where physical characteristics such as gradient, and where physical processes such as weathering and seasonal water flow patterns, come together to produce the exacting conditions that permit the appropriate substrate to accumulate, while simultaneously providing the microhabitat flow and depth characteristics necessary for the spawning substrate to successfully incubate salmonid eggs. In addition, it is well documented in the scientific literature that it takes only a relatively small amount (in proportion to other habitat types) of high quality spawning habitat to sustain a viable population. Because such habitat is localized, it is unlikely that the two study sites selected in the BC would produce representative data as to the true availability or adequacy of this specialized habitat in the BC. In addition, since the BC has only received spillage flows for many decades, whatever spawning habitat was there before the applicant assumed its control over the drainage would not have been historically maintained by other natural cleansing processes, and thus may have become embedded. The only way that true spawning habitat potential could be determined is by passing an adequate minimum flow during periods of non-spill for several years, to allow habitat characteristics and natural processes to stabilize under the new regime.

Even if it is true that existing spawning habitat is limiting or is embedded, we do not see how this precludes the possibility that new spawning habitat could be introduced, or that existing but embedded habitat could be physically cleaned to improve its suitability. The former option would be consistent with Staff's recent proposed action in the DEIS for the lower Penobscot River, where it recommended that the applicant place habitat in an existing project tailrace as mitigation for lost fishing and resting lies for Atlantic salmon.

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-35 establish those resource values within the context of

Cont management objectives of the agencies responsible for their stewardship and do not establish independent management objectives.

COMMENTS FROM PENOBSCOT INDIAN NATION

Statt contends that annual spillage at Stone Dam will displace any salmon that might occur in the BC under increased minimum flows, that these fish will experience heavy predation in downstream waters, and that displaced fish would not be capable of returning to the BC because of Grand Falls. Concerning displacement, we would contend that this is an artificially contrived obstacle. First of all, high spring spillage flows occur in hundreds of regulated and unregulated stream and river systems throughout Maine, many of which contain robust, self-sustaining populations of coldwater species, including salmon. The fact that salmon thrive in the West Branch below McKay, at summer flows far higher than would occur naturally, indicates that they are well adapted to cope with such flows. Granted that spillage at Stone Dam is greater at times than the regulated flow below McKay, we would note that spillage also occurs at Ripogenus. We believe that any fish that may be displaced could find temporary refuge in one of the three large ledge pools we identified in our discussion above on adult habitat, until flow conditions allowed them to redistribute through the BC.

Staff's contention that any displaced salmon would not be able to move back over Grand Falls if they chose to, is unfounded. Staff has apparently not examined Grand Falls at flows in the range that is being considered. PIN viewed Grand Falls at 50 cfs, and, while this flow appeared marginal in terms of the ability for salmon to negotiate the falls, it was obvious that additional flow would markedly improve this situation. Grand Falls, while a total of 15 feet in height, actually was series of several smaller drops at 50 cfs, each with a small holding pool in between, and in our view could be surmounted by at least adult and larger sub-adult salmon at the proper flow (which has yet to be determined because the instream flow study did not address this issue).

Regarding potential losses of displaced fish to downstream predation, this would occur whether the applicant managed the waterway as it does today or if it were entirely undammed and unregulated. Predation losses are simply a fact of nature, and are not pertinent to the true potential for development of a salmon fishery in the BC above Grand Falls.

In summary, Staff's choice to approach the potential for management of salmon in the BC in a confrontational and obstructional way, and draw conclusions which are unsubstantiated in the record or conceptually flawed, reflects a fundamental level of negligence with respect to the Commission's own mandate to give equal consideration to fish and wildlife benefits in its consideration of licensing actions. We recommend that Staff abandon this misguided approach and instead concentrate its efforts in identifying and pursuing positive ways under which such potential could be realized, an approach that would be more consistent with its equal consideration obligations.

<u>Brook Trout.</u> While landlocked salmon was a "key" species for consideration in relation to increased BC flows, inasmuch as it was identified as such early in the consultation with DIFW, it is by no means the only species for which management in the BC could be attempted. It is a logical starting point, since the species is well established in project waters upstream and is in high demand in the region by anglers. The PIN was not included in early consultations nor in the salmon habitat/flow study conducted by the applicant. If we had been, we certainly would have asked that other logical potential indigenous species for providing a fishery, such as brook trout, also be examined.

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-36 Opinion noted. See response PIN-35.

PIN-36

PIN-35

Cont

Staff admits on page 4-28 that flows in the range being considered would benefit brook trout fisheries in the BC as well as their preferred food supply, macroinvertebrates. Staff also correctly identifies that native brook trout are present in the three tributaries to the BC. However, instead of pursuing this highly viable option. Staff again shrouds itself in the inappropriate assumption that only landlocked salmon, the "key gamefish of concern", should be seriously considered. However, as far as the PIN is concerned, landlocked salmon was singled out as a "key" species only because it is the species which DIFW initially was considering for management. Since DIFW has abandoned its initial plans for fisheries management in the BC, we see no reason why landlocked salmon must be the only species under consideration. The fact that the State chose to abandon its obligations to provide fisheries where an opportunity clearly exists should not preclude others, which have greater or more genuine interest in the resource potential represented by the BC, from pursuing that interest.

Further support for the potential value of the BC for a brook trout fishery stems from the fact that the flows being considered for the BC, including the highest flow of 945 cfs contemplated by USFWS, are significantly lower than the applicant's historic and proposed flows in the West Branch below McKay, which heavily favor salmon at the expense of brook trout. Since it is widely recognized in Maine waters that, all other factors being equal, flows that would be optimum for brook trout are likely to be lower than optimum flows for salmon, we see the potential for brook trout are likely to be lower than optimum flows for salmon, we see the potential for brook trout fishery management as going hand-in-hand with lessening the burden of BC flows on the applicant's economic interests. In fact, this would appear to be a perfect opportunity to at least initiate movement towards "equal consideration", and more balanced use of a waterway, at least for that portion of the river represented by Stone Dam, the BC, and the Millinocket power station and canal.

In addition, PIN, USFWS, TU and others have raised concerns in relation to lost brook trout populations and fisheries in the project areas due to impoundment drawdowns and artificially high flows in the West Branch below McKay, which favor salmon at the expense of brook trout. These groups have repeatedly argued that instream flow studies be conducted in the West Branch below McKay to quantify potential brook trout habitat at different flows, and to quantify the loss of brook trout potential due to the applicant's proposed artificially high flow regime. Beyond the issue of whether the Commission's decision not to require these studies is proper, the potential for a brook trout fishery in the BC should be viewed by Staff as a prime opportunity to mitigate for some of the atmost certain diminialment of brook trout resources that has occurred historically, especially in the Ripogenus impoundment and the West Branch below McKay, due to the applicant's impoundment drawdowns and artificially high flow regimes.

Lastly, we would offer that a healthy mixture of both landlocked salmon and brook trout in the BC has just as much potential feasibility as a fishery dominated by either species, assuming an adequate and appropriate minimum flow is afforded.

American Eci. As reflected by our above discussion of brook trout potential in the BC. Staff's approach to BC fisheries has been unnecessarily restricted to a single species, and in essence. a single life stage (adult landlocked salmon) and purpose (sportfishing). While Staff routinely takes this approach in this DEIS, and similarly in the DEIS recently issued for the lower

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-37 Opinion noted. See response PIN-35.

PIN-36

Cont

Penobscot, this reflects a scientifically outdated, piecemeal approach to resource management issues. By failing to account for all potential benefits of <u>adequately watered riverine habitat</u>, and instead focusing entirely on a single resource entity within that habitat, the value derived from the flow allocation is grossly underestimated, and the cost/benefit relationship grossly overestimated. Staff fails to recognize that <u>physical habitat</u> is the ecological foundation for all biotic potential. One must have the habitat to work with before the full biotic potential of that habitat can be realized and developed. In contrast, Staff's approach of focussing on one species, usually one representing the top level of the ecological pyramid, and at the exclusion of all others, precludes the resource managers from ever having the opportunity to explore and develop the full ecological potential of the aquatic <u>habitat</u> that would otherwise be at their disposal.

A case in point concerns the American eel. While we discuss eels in more detail and broader context in a later section of our comments (see section "IV" below), increased flow in the BC stands to benefit American eel populations, and provide at least some initial mitigation for the extensive damages to these populations caused by the construction of numerous dams without fish passage, and in the specific case of the BC, due also to the disposal of this section of the West Branch to dryway for many decades. If additional flow were provided to the BC, it is likely that any subadult cels that do make it up as far as the Dolby impoundment would not only have the BC itself as additional potential production habitat, but will also have far better opportunity during the dry months to move further up into the drainage into areas where continuing impacts of hydro project operation have essentially eliminated viable populations from these waters. Upstream and downstream fish passage at Stone Dam would provide the best opportunity for this to occur. However, even without such passage, at least some additional cels will likely be able to enter upper project waters (by crawling over the face of, or around, Stone Dam), and thus enhance the total adult eel production potential of the drainage, particularly in comparison to the case under existing conditions, where the only passage route during no-spill periods would be up through the turbines at Millinocket Development (i.e. highly unlikely).

<u>Anadromous Fisheries Restoration</u>. Waters of the West Branch once provided high quality adult spawning and juvenile nursery habitat for all of the anadromous fish species native to the Penobscot River, except perhaps the sturgeons, prior to the construction of numerous dams without fish passage. The complete loss of these fisheries over a comparatively brief period of time greatly impacts present day opportunity for Penobscot Indians to harvest such fish for subsistence purposes. The significance of this loss is magnified by the fact that many <u>resident</u> fisheries of the West Branch (which could represent additional subsistence fishing resources), contain dangerous levels of contaminants such as mercury, causing them to be unfit for subsistence fishing purposes. Resident fisheries further downstream contain not only unsafe levels of mercury but also dangerous levels of dioxin and other complex chlorinated contaminants, also making them unsafe for subsistence consumption.

We recognize that the State ASRSC and DMR, respectively, identify no <u>currently active</u> plans to open up the West Branch for enhanced Atlantic salmon, or clupeid (alewife and shad), restoration. However, we believe, based on discussions with them, that this present-day approach reflects only their short term interest and management/restoration capability. We believe that both agencies would express interest in expanding these programs, particularly with

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-38 Any license issued for these projects would include an article that would ensure that fish passage facilities could be provided in the future, as deemed necessary by Interior.

> Restoration of anadromous species to the west branch of the Penobscot River has not been proposed in any current plans and is therefore not a reasonably foreseeable action that can be considered in this EIS.

PIN-38

PIN-37

Cont

the assistance of others. to a point at least as far upstream as Stone Dam, as restorations advance and fish passage is constructed. The PIN is very interested in this potential long term eventuality, and we know that USFWS is also interested in, and supportive of, such future efforts. Based on the available data from the instream flow study, which indicates an abundance of juvenile landlocked Atlantic salmon habitat at fairly low flows (compared to the applicant's generation flows), and based on the extremely close similarity in juvenile habitat requirements between landlocked Atlantic salmon and anadromous Atlantic salmon, we see the potential long term value of increased flows in the BC (along with Millinocket Stream), toward enhanced juvenile Atlantic salmon production, to be quite significant as well as attainable within the term of any new license issued.

PIN's long term goals in the BC, and in Millinocket Stream, do include this consideration of the potential for enhanced juvenile production of Atlantic salmon in these waters. In the shorter term, adequate minimum flows in the BC will provide the opportunity to examine physical, chemical and biological aspects of this river reach while it develops and supports resident fisheries and other aquatic life under a permanently flowed condition. Thus, when anadromous fish restoration moves into the West Branch drainage, we will have a solid information base upon which to develop restoration and management strategies for these species in these reaches.

Biodiversity

In the same light in which we have portrayed the diverse potential for both enhanced sport and food fisheries from increased flows in the BC, we rationalize the expansion of broad ecological benefits, to include the very core of healthy, functioning aquatic ecosystems, namely, biodiversity. The PIN has expressed its concern to the Commission regarding the loss of biodiversity in the Penobscot River drainage due to the construction of dams, which convert physically and biologically heterogenous free-flowing habitats into homogenous impoundments. The PIN requested that Staff conduct a cumulative impact analysis on biodiversity in this DEIS. Staff neither heeded our request nor provided any explanation as to why this critical issue was ignored. However, beyond the fact that Staff's negligence in this area has caused this DEIS to be inadequate in terms of addressing the cumulative environmental impacts of the applicant's operation of these projects. Staff does imply that increased flows in the BC would have value in restoring part of the lost biodiversity (at least, macroinvertebrates, page 4-28) in the West Branch subdrainage, and particularly in the Penobscot Mills Project area. Under existing conditions, there is virtually no significant portions of free-flowing, permanently watered riverine habitat in the Penobscot Mills Project area, as essentially every foot has been converted to impoundment. The cumulative impact on aquatic organisms that depend partially or entirely on this habitat type for their survival, while never quantified (though we requested that it be), is likely substantial.

Increased flows in the BC affords an ideal opportunity to regain some of the lost diversity due to the numerous dams in the West Branch subdrainage. Staff should seriously consider this broad, ecosystem-based value of increased flows in the BC, along with continued consideration of traditional multiple resource enhancements such as sport and food fisheries, in determining

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-39 See response to PIN-14.

PIN-39

PIN-38

Cont
the overall value and benefit of increased flow in the BC.

L BACK CHANNEL (continued)

B. Flow Allocation Options

PIN-40

Beginning with our comment letter to the Commission dated February 27, 1992 (Request for Additional Scientific Study), and continuing in filings dated May 23, 1993 and October 4, 1993, the PIN has recommended that Staff evaluate the allocation of flow to the BC by <u>splitting</u> the flow which would otherwise arrive at Stone Dam as a result of implementing the Water Use Plan for project waters above that point, rather than drawing <u>additional</u> flow from upstream waters (i.e. the applicant's long-standing approach). Despite these repeated requests, Staff has never addressed this highly pertinent alternative in this proceeding, even though there is absolutely no objective rationale for dismissing this option.

We find this omission very disturbing. Not only does it indicate, again, that our comments and participation in this process have been essentially ignored, but it also predisposes other portions of Staff's implied "independent" analysis, and "carefully considered" deliberation, to serious questioning in terms of the level of comprehensiveness, objectivity, accuracy, and responsibility provided in its proposed licensing actions.

In addition, the failure to consider this flow allocation alternative compounds itself in the DEIS, causing Staff's conclusions regarding the actual cost and indirect water availability impacts of BC flows to be inappropriate and indefensible. For example, all discussions contained on pages 4-5 through 4-8, and page 5-8 (and, perhaps elsewhere as well), where Staff attempts to cite adverse impacts of BC flows on other parts of the WUP, particularly those upstream of Stone Dam, are irrelevant, if flow is delivered in the way PIN recommends. The only project-related impact of a split-flow approach to BC flow allocation is the direct loss of hydro generation at Millinocket, which is currently undergoing plant rationalization with concurrent reduced energy requirements.

We continue to recommend that this flow allocation alternative for the BC receive equal consideration in this EIS, and if this does not occur, we will consider any subsequent EIS document issued by the Commission to be deficient and unacceptable, on this basis alone. And, in an effort to eliminate at the outset a potential obstacle to considering this flow allocation alternative, namely the 2000 cfs minimum flow requirement at <u>Millinocket Development</u>, we believe, based on discussions with DEP's Water Bureau in 1991, that they would agree to move the compliance point for this 2000 cfs requirement (which, according to them, is in place primarily, if not exclusively, for the purposes of diluting the waste water discharge from the applicant's Millinocket paper mill discharge), downstream to where flows from Millinocket Development and the BC would recombine, effectively removing this obstacle from the issue of providing BC flows via a split flow regime.

We can also identify other available options and innovations for providing increased flow to the BC, which could diminish the overall cost to the applicant, but which the DEIS fails to consider.

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-40 See response to PIN-2.

COMMENTS FROM PENOBSCOT INDIAN NATION

For instance, Staff could examine a seasonal flow regime, which could begin each spring with one flow allocation at the point when spill at Stone Dam falls below that allocation, but then could be seasonally varied to reflect what would happen, on a gross cyclical basis, if natural flows were occurring in this reach. In other words, flows to the BC could be gradually decreased from the spring starting point, over the course of the summer, until they reached a floor minimum in late summer, then increased slightly in the fall, and maintained through the winter, to protect any salmonid spawning that has occurred. This would save the applicant considerable amounts of water, particularly during the driest times of the year, and would be a functional approach regardless of the absolute numerical value of the spring flow allocation starting point.

Another viable option that could be considered, either independently or in concert with the above idea, would be to have specific minimum flow scenarios for dry, average, and wet years. We are unsure how predictable future monthly water availability during a given water year might be, however, it would seem that with one to two months of advanced predictability, subsequent monthly flow allocations to the BC could be adjusted to reflect the gross water storage availability for that water year at that time. If feasible, this would seem to be another option for providing adequate flows in the BC while cutting the cost of that allocation to the applicant, particularly in dry years when less total water would be available.

The feasibility of these flow allocation innovations should be studied and discussed in the EIS.

I. BACK CHANNEL (continued)

C. Cost of Flow and Cost Recovery Options

We put little stock in the applicant's contention that it will be forced to lay off significant numbers of workers if an increased minimum flow is allocated to the BC. We have observed in the recent past this same entity, during the proceedings related to their proposal to build a new dam at "Big A Falls" on the West Branch, making these same claims (in relation to the possibility that it would not get a license to build the project), only to have the claims proven false by intervenors in that process. We also suspect that the highly volatile and competitive industry and market for pulp and paper products is a much more significant driving force in terms of the employment at the applicant's mill than the comparatively small amount of lost generation (only 2.8% of annual generation at 350 cfs in the BC) that would occur with additional flow in the BC.

In addition, the likelihood that potential job losses due to increased flow in the BC are falsely inflated is born out in pertinent sections of the DEIS. On page 4-74 (4.12.3.3), the DEIS states that "GNP states that any minimum Back Channel flows greater than 50 cfs would directly jeopardize 238 mill jobs..." This would imply that job losses would go from none at 50 cfs, to 238 at, for example, 60 cfs. On page 4-73 (top paragraph), the DEIS states that "GNP contends that providing flows of 350 cfs in Back Channel...{will cost]...approximately 238 mill jobs". Since it is logical to conclude that the relationship between flow allocation and potential job losses, assuming any such losses would actually occur, is one of direct proportion between the

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-41 The staff agrees that Alternative 2 (final recommendation) would not adversely affect the competitiveness of GNP's mills, nor would it result in a loss of employment at the mills. We do not agree that there is a direct relationship between the amount of Back Channel flows and the potential number of jobs lost. A flow threshold exists that, if exceeded, would require shutting down an entire production line. It is difficult to identify the threshold at which an increase in the applicant's power costs due to increased flows in the Back Channel would be significant enough to cause production cutbacks and job loss. We conclude that the costs associated with flows in the Back Channel outweigh the few benefits that would result.

PIN-40

Cont

amount of flow and the number of jobs, then both of these statements cannot be true. Assuming that the applicant has carefully considered its interests in forwarding these statistics, we can only conclude that the 238 figure is relevant to the 350 cfs allocation, and that lesser flows would mean lower potential job losses. The DEIS fails to recognize this and thus overestimates the economic impact to the applicant of flows lower than 350 cfs, such as the 165 cfs figure that Staff considers in Alternative 2.

The DEIS also discusses the impact of BC flows on lost internal generation at the applicant's mills (about \$400K/year at 165 cfs and 900K/year at 350 cfs). Since FERC has <u>neither</u> <u>comprehensively assessed nor seriously considered</u> the adverse impacts of the applicant's Millinocket and East Millinocket mills on the environment of the West Branch below the mills, anywhere within the context of these hydro project relicensing proposals (although we would prefer such an approach, since there is little doubt that adverse impacts such as mercury contamination and anoxic hypolimnetic waters in Dolby Pond are a result of the combination of hydro project operation and mill discharges), we find it inappropriate to assign such costs against the environmental benefits that increased BC flows, an issue which is an integral part of these licensing actions, would provide.

Lastly, the DEIS fails to consider at least one available cost-recovery option for lost generation due to increased BC flows, namely, the placement of an appropriately sized turbine at Stone Dam. We recommended in our scoping comments that this be evaluated. We see no evidence in the DEIS that it has been. This option should be seriously considered in this EIS.

-21

II. MERCURY CONTAMINATION OF SEDIMENTS, WATER, AND BIOTA

The impact of historic use of mercuric compounds in the pulp and paper industry, and the impact of extensive annual impoundment drawdowns, or project operation/existence in general, on mercury contamination levels and dynamics, have been irresponsibly diminished or dismissed in this DEIS. Beyond the critical pertinence of these issues towards the safe consumption of fish and wildlife resources from the project areas by tribal members, the impacts of mercury contamination in project sediment, water and biota may be detrimentally impacting bald eagles, which are, much like Atlantic salmon, culturally significant to the PIN.

On pages 3-12, 3-13, and 4-8 through 4-11, the DEIS discusses the applicant's mercury contamination studies and basically reiterates the applicant's opinion and interpretation of the meaning and significance of the results. The discussion so provided demonstrates little if any independent or objective thought or analysis on the part of Staff. In choosing this casual approach. Staff has ignored pertinent information in the record supplied by PIN and others, and because of this shortcoming, draws erroneous or unsupported conclusions regarding mercury contamination issues in project sediments, water and biota. We provide several examples below. Subsequently, again compounding its initial errors in judgement, Staff dismisses mercury contamination issues entirely from its "Conclusions" section (5.0) and thus from its proposed licensing actions.

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-42 See response to PIN-13 and PIN-24.

PIN-41

Cont

In our first example, page 3-11 of the DEIS cites EPA (1993) in stating that "EPA's fish consumption guideline is 0.6 micrograms per gram of total mercury in edible fish tissue". We consulted EPA (1993) as it is listed in the literature cited. It contains nothing regarding mercury contamination guidelines and, instead, is Volume I of a pending four volume series by EPA on a myriad of fish contamination issues. Volume I only contains guidelines for how to collect, prepare, and analyze fish for contaminants. Volume II of this series, on the other hand, deals with risk assessment and consumption limits, is dated June, 1994, and has EPA reference number 823-B-94-004, which does not match the reference number in the DEIS citation. In addition, we cannot verify from Volume II, which contains EPA's consumption guideline.

We are unsure where Staff actually obtained its information, however, it is incorrect and must be rectified in the EIS. We therefore recommend that Staff obtain the correct document and consult tables 3-55 through 3-59. Note that these tables are for <u>methyimercury</u>, not <u>total</u> mercury, because EPA has determined that the overwhelming proportion of mercury contamination in biota is of this form, which is the toxic form. The tables contain consumption rate guidelines for several combinations of age/sex grouping, time period over which consumption occurs, and category of human health impacts. For evaluation of the impacts of mercury contamination in project area fish in relation to tribal subsistence fishing, Staff should use the 12-ounce meal size column (i.e. above <u>average</u> meal size), and the lowest mercury concentration at which the recommended consumption limit changes from unlimited (UNLIM) to anything lower, since subsistence fishers must be permitted to safely consume fish as a primary food source and at unlimited levels.

In a second example, PIN recommended in our scoping comments that FERC consult several references concerning the historical use of mercuric slimicides by the pulp and paper industry (generally pre-1965), and also to investigate the existence of, and obtain, any records of such usage available from the applicant. We see no evidence in the DEIS that either of these tasks were ever addressed at even a cursory level, even though Scoping Document 2 (section 6.6, page 29) indicates that they would be. These investigations are very pertinent to determining possible reasons for significantly higher sediment mercury levels in Dolby Pond sediments (i.e. below the Millinocket mill) when compared to sediment levels in impoundments above the mill.

In a third example, the DEIS correctly indicates that PIN and USFWS were involved in the consultation on the applicant's mercury contamination study plan, developed in response to FERC's June 8, 1992 AIR to the applicant, Item 8. However, the DEIS fails to discuss the fact that, despite this consultation, the applicant did not feel it was compelled to address many of PIN's and USFWS's study design recommendations, including one that was crucial to the determination of the extent that project operation has contributed to mercury contamination in project waters. Both PIN and USFWS strongly recommended that <u>aediment core</u> samples, rather than traditional <u>sediment surface</u>, <u>grab-type</u> samples, be collected, and portioned into time series segments to correspond to major changes in human activity which could have influenced the deposition rate of mercury in project waters. The applicant refused to collect core samples, and FERC apparently never followed up, despite the PIN discussing in detail, in our May 21, 1993 comment letter, the limitation that this study design shortcoming would have on gaining insight

into the relationship of past project operation to mercury contamination in project waters

PIN-42 Cont

In a last example, the DEIS concludes that, regardless of what levels of mercury were found in project area fish tissue, the differences in those levels between project and non-project lakes were not statistically significant, thus, any link to project operation can be ruled out. Since there is no indication in the DEIS that Staff conducted any independent statistical analysis, we must assume that Staff has again simply reiterated what the applicant's interpretation of their own statistical analysis conducted by Dr. Terry Haines of the University of Maine, one of the foremost mercury contamination experts in the Northeast, despite this document being a part of the record for this proceeding. In contrast to the applicant's analysis of the mercury study results, Dr. Haines' analysis indicated that lake trout and smelt from project lakes had statistically significantly higher mercury levels than fish from control lakes.

In summary, we disagree with Staff's conclusions regarding the effects of project existence and operation on mercury contamination in project area sediments, water, and biota. We recommend that the two licensing conditions expressed in the pertinent section of our May 21, 1993 filing with the Commission (Terms and Conditions) be incorporated into any new licenses issued for these projects.

III. DOLBY POND

We continue to assert that existing water quality conditions in Dolby Pond cause that water body not to meet all applicable water quality standards, especially those addressing dissolved oxygen requirements and habitat for indigenous fish and aquatic life. We reviewed Staff's analysis and conclusions regarding Dolby dissolved oxygen levels, and we are not convinced that the analysis presented is logical. or that the conclusions so drawn are defensible. In contrast, we continue to believe that the combination of mill wastewater discharge and operation of the Dolby impoundment are the primary factors responsible for the failure of this water body to meet its riverine classification standard, and also the primary factors exacerbating the accumulation in Dolby impoundment sediments and biota of contaminants such as mercury.

We recommend that the applicant be required to conduct post-licensing study in consultation with PIN, and others as appropriate, to provide further insight into the connection between project operation and these water quality and contaminant problems in Dolby Pond. In addition, we recommend that the applicant be required, based on the results of such studies and an analysis of available remedial alternatives, to develop and implement feasible remedial measures, in consultation with PIN, and others as appropriate.

IV. AMERICAN EEL IMPACT ANALYSIS

PIN-44

The eel is a catadromous species which is native to the Penobscot River drainage and which was abundant in virtually all waters of the drainage, including the West Branch, prior to the industrial revolution and the subsequent construction of numerous dams in the drainage without fish passage. The eel is of high importance to the PIN in relation to our sustenance fishing

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-43 See response to PIN-12.

PIN-44 We reviewed the information you refer to and have added text to the affected environment and impact sections of the FEIS pertaining to American eel.

PIN-43

rights in the Penobscot River, and is subject to growing commercial fisheries. This importance is identified in our scoping comments, where we asked the Commission to conduct a cumulative impact analysis in this DEIS (which was never done), and in the extensive comments on this resource in our recent filing with the Commission on the lower Penobscot River DEIS. Recent insight from lifetime eel fishers, one of which also operated a hydro power dam on the Kennebec River in Maine for over 20 years, indicate that impacts to downstream migrating adult eels forced to pass through turbines are extensive, and likely much greater than anyone ever anticipated.

One eel weir fisher on the Passadumkeag River, a tributary to the mainstem Penobscot, has developed an agreement with a hydro operator upstream wherein the operator shuts down the turbines during selected overnight periods when the peak eel migrations are expected to occur each year. The eel fisher maintains that eel catch in his tributary increased substantially in years subsequent to this agreement. Further evidence supporting the significance of turbine passage impacts on eels is available from the Commission's own recent actions with respect to motify project structures and operations in order to at least partially mitigate for the adverse impacts of project operation on long term sustainability and viability of eel populations in the river in question.

The West Branch basin represents a potentially significant habitat resource for the production of American cels, which, while present in limited numbers in the project areas (but in progressively decreasing numbers as each impassable dam is encountered), are in serious decline throughout the Penobscot River drainage and in other large Maine rivers. Declines in the numbers of young eels reaching West Branch production areas, and the number of adults reaching the estuary, have almost certainly occurred, and are continuing to occur, in large part due to the presence of numerous dams lacking unstream (except North Twin) and downstream fish passage. The DEIS inadequately addresses past and continuing cumulative adverse impacts to eel populations in the Penobscot River, as well as impacts resulting from Staff's recommended licensing alternatives for these projects. We continue to recommend that this EIS provide an adequate cumulative impact analysis on American eel within the context of existing project operation and Staff's proposed licensing actions for these projects. Furthermore, we recommend that, in addition to the site-specific mitigation that could result from increased flows in the Back Channel (see discussion under "BACK CHANNEL: American Eel"), Staff should evaluate and adopt additional appropriate mitigation measures, including any of those utilized at the abovereferenced Columbia Dam Project (FERC #8396-013) which could provide benefits here, in its licensing actions for these projects.

V. PRESENT AND FUTURE FISH PASSAGE NEEDS/ISSUES

On pages 4-25 and 4-26, the DEIS discusses fish passage issues and needs in the project areas. While we agree that anadromous species are currently absent from project waters, the DEIS fails to point out that the primary reason for this absence in the Penobscot Mills project area today is the total lack of fish passage facilities at the three lower dams on the West Branch, including the Medway Project Dam which is not operated by the applicant. In addition, the DEIS fails

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-45 See response to PIN-38.

PIN-44

Cont

PIN-45

to recognize that catadromous American eets require upstream, and especially downstream fish passage facilities, to minimize life history impacts of project operation on that species.

While we are not recommending fish passage facilities at Penobscot Mills Project dams at this time, we would alert FERC that it is likely that we will seek the construction of fish passage at one or more of the lower project dams within the term of any new license issued for this project. This foresight is consistent with our long term goals for restoration of native anadromous fisheries to the Penobscot River, including, eventually, waters of the West Branch where these species were historically present (see additional discussion under <u>1. BACK CHANNEL</u>; Anadromous Fish Restoration).

We expect that USFWS's reservation of Section 18 authority within any new license issued for this project would provide the mechanism for us to seek such facilities in the future.

Lastly, on page 4-26, under 4.4.1.7, the DEIS concludes that "impoundment and river reach populations [of smallmouth bass and pickerel] do not interact (i.e. they do not migrate between waters)". This implies that lack of movement between such reaches is a chosen behavior. Years of video tape data at the West Enfield Project upstream fishway window clearly indicate that this is not the case, at least not for bass. Countless numbers of bass move up and down through the fishway annually, indicating that, if given the choice (of behavior), at least some bass will likely choose to move between impoundment and river reach segments. The only reason that this movement may not occur in the project area now is because it is not physically possible for them to do so (i.e. no fish passage).

This concludes our supplemental comments on the DEIS for the upper Penobscot River Basin. Please feel free to contact me (207-539-8219) should you have any questions.

Sincerely yours,

Paul Bisulca

Asst. to the Governor on Environmental Affairs

Distribution

FERC Service Lists #2458; #2572

RESPONSES TO PENOBSCOT INDIAN NATION ON UPPER PENOBSCOT RIVER BASIN DEIS

PIN-46 We have changed the wording of the FEIS to reflect that the non-migratory behavior may not be a chosen behavior. However, staff believe that fish passage for bass and pickerel is unnecessary and, in certain situations, undesirable. The proliferation of these species since their introductions to the Penobscot basin is an indication that they are capable of thriving in discrete riverine segments or impoundments.

PIN-45

Cont

PIN-46



Dear Secretary Cashell:

By letter dated March 9, 1995, the Penobacot Indian Nation submitted its completion of comments to the Commission on the Draft Environmental Impact Statement for the Penobacot River in Maine (FERC Projects #2458 and #2572).

When our letter was in the final stage of preparation, an oversight occurred which caused two minor changes on the last page (p. 20) of our letter not to be executed. While we caught these emissions prior to serving copies of our letter on the Service List parties, unformately, the original and eight copies that were sent to your attention did not contain the corrections and had already been picked up by the overnight coaries.

One of the two intended alterations was cosmetic only; a missing "close parenthesis" and period. The other, however, was of minor substance, in that, the correction was necessary to allow an errant sentence to make sense within the context of the paragraph where it was located. The sentence we refer to is the second sentence in the last full paragraph of the last page of our March 9 letter. The uncorrected version reads: "This implies that movement between such reaches is a behavioral phenomena". The corrected version should read; "This implies that lack of movement between such such reaches is a chance behavior".

All copies sent to Service List parties, and the copy sent directly to the FERC Staff Project Task Officer for this EIS, Edward Meyer, contain the <u>corrected</u> sentence. Only the original and eight copies sent to your attention do not. We would ask that this letter of notification be filed in the appropriate project docket(s), such that there is no confusion in the future as to why this minor difference exists between the Commission's and others' copies of our March 9 letter. If the Secretary would prefer that we supply a corrected original and eight new corrected copies, we would be glad to do so. Please have the project officer contact my staff assistant, Clem Fay, at 207-827-7776, ext 230) at his convenience, if this is desired.

We have included a copy of this letter with the corrected copies of our March 9 letter sent to the Service List parties in this proceeding. We apologize for any inconvenience due to this oversight. Thank you in advance for your attention to this matter.

Sincerely yours.

Paul Bisulca

Assi. to the Governor on Environmental Affairs

COMMENTS FROM SAVE OUR SCENIC LAKES ASSOCIATION ON UPPER PENOBSCOT RIVER BASIN DEIS

RESPONSES TO SAVE OUR SCENIC LAKES ASSOCIATION ON UPPER PENOBSCOT RIVER BASIN DEIS



Lois Cashell, Secretary Federal Energy Regulatory Commission 823 Morth Capitol Street, ME Washington, DC 202426 RE: Application for <u>Meigr License</u> (Type of Application)

<u>Penobacot Mille Mydroelectric Project</u> (Project Name)

FERC No. 2454

Dear Secretary Cashell:

Mr. John McDevitt, Fresident of the Save our Scenic Lake Association announced today that his organization is withdraving their opportion to the re-licensing of the Great Korthern Paper Dams in the Penobecot River Basin and be intende to support the company's relicensing efforts at the January 23th hearing in Millinocket.

relicensing efforts at the January 23th hearing in Hillinocket. Trelicensing efforts at the January 23th hearing in Hillinocket. Telicensing because of the unrestricted water level fluctuation to the North Twin Impoundeent. The SOBLA was also opposed to the water use list computer model that set August 15 as the date to begin the drewdown of the North Twin impoundeent and argued in favor of August 22nd in order to provide work predicable water levels for those persons wing the North Twin impoundeent during the summer recreational sesson. Nov that the Federal Energy Regulatory Commissions's (FERC) Draft

the dravdown date, we now feel that the re-worked for the SOSLA and the company and nothing (pp. 4-55) and that Great Morthern Paper officials intend to accept these staff recommendations which establishes stable water levels and now feel that the reto the reagree's with both positions of SCSLA in oppomition recommendations which establishes stable acting SOGLA Environmental Impact Statement served by 22nd. as the licensing process has August licenaing. further

Dedicated to the Preservation of Recreational Water Levels

COMMENTS FROM SAVE OUR SCENIC LAKES ASSOCIATION ON UPPER PENOBSCOT RIVER BASIN DEIS

Hr. HcDevitt has been assured in conversations with Hr. Brian Stetson, manager of Environmental Affairs for Great Northern Paper that G.N.P. will accept the findings of the FERC staff that provides for stable water levels and establishes August 22 as the benchmark date for the beginning of the North Twin impoundment drawdown and that they will not comment in opposition to these recommendations at the January 25th hearing.

As a result of these recommendations and this accord with Great Northern Paper officials SOSLA is withdrawing its apposition to the relicensing of the dams.

Mr. McDevitt would like to thank all of the people in the Katahdin region that supported the efforts of SOSLA during these past five years and he would also like to thank the officials of Bowater Great Northern Paper for their patience, professionalism and villingness to listen and compromise for the public good.

Sincerely,

SKa HMA

Ar. John HcDevitt, President

RESPONSES TO SAVE OUR SCENIC LAKES ASSOCIATION ON UPPER PENOBSCOT RIVER BASIN DEIS



We believe that FERC staff have underestimated Maine's capacity and commitment to perserving this State's natural resources. Maine is very proud of its track record in developing creative and effective measures that balance the competing needs of landowners and recreational users and the imperative to protect our natural environment. Through a combination of regulation, public ownership, and cooperation, Maine has earned a national reputation as a leader in resource protection.

We also take exception to the recommendation that would require the licensee to purchase or otherwise take property in the project area to implement protections measures. The State of Maine owns Gero Island and shoreland in the vicinity of Chesuncook Village which is managed by the Bureau of Public Lands. This property is managed consistent with an approved management plan. The Bureau's policy is to issue no new residential or commercial leases on these lands. Moreover, timber harvesting on this property is subject to the Bureau's Integrated Resource Policy, developed after with public participation and subjected to interdisciplinary review.

The State of Maine has no intention of selling its property within the project area. We will strenuously oppose any effort to take it by emminent domain, and resent being placed in a position of potential conflict with a landowner with whom we have enjoyed fruitful cooperation in the preservation of natural resources. Instead, and as noted above, we believe

RESPONSES TO MAINE STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DELS

RESPONSES TO MAINE STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

1. Cont

that the State's effective and balanced land use protection measures should be recognized and deferred to by FERC.

In closing, I submit that the State of Maine has been a very effective steward of its natural resources. FERC staff apparently agrees, finding on page 5-24 that "the state's land use and forestry practices regulations adequately protect the natural resources of this area." Likewise, the land in public ownership is afforded quality resource management. This result is because the people of this State take very seriously their stewardship responsibilities. I hope that in preparing its final EIS, FERC will re-assess Maine's resource protection capabilities and commitment. Moreover, I hope that FERC will recognize in the State of Maine a potential partner rather than a barrier to forging a balanced public policy regarding these natural resources.

3

Thank you.

1. Cont



RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

au...,

and the second se

184 STATE STREET, STATE HOUSE STATION 30, AUGUSTA, MAINE 04333

BEFORE THE

UNITED STATES OF AMERICA

FEDERAL ENERGY REGULATORY COMMISSION

Penobscot River Basin Docket Nos 2572 and 2458

COMMENTS February 17, 1995

The Maine State Planning Office, in response to the <u>Draft Environmental Impact</u> <u>Statement on the Penobscot River Basin, Maine for Projects #2572 and 2458</u>, hereby moves to Comment in the proceeding in accordance with the requirements of the Rules of Practice and Procedure, 18 C F R §§385 210, 211, 214 The grounds for these comments are as follows

1 The Maine State Planning Office (SPO) is an agency of the State of Maine, with an office in Augusta, Maine The SPO consists of the Director of State Planning and his staff. Pursuant to the authority granted by Executive Order #13 FY86/87, the SPO is responsible for coordinating the State agency consultation for Federal Hydropower Licenses and Exemptions in the State of . Maine The SPO is also responsible for the development of a consistent State position in Federal Licensing and Relicensing Proceedings and for intervention on the State's behalf in all FERC licensing proceedings for hydropower projects in Maine

The Maine Department of Environmental Protection (DEP) is an agency of the State of Maine, with principal offices in Augusta, Maine. DEP consists of the Board of Environmental Protection (the Board), the Commissioner of Environmental Protection, and his staff. (38 MRSA § 341, et. seq.) DEP is responsible for administering and enforcing the Maine Waterway Development and Conservation Act (MWDCA), 38 MRSA §§ 630-636, in organized municipalities of the State

3. DEP is also responsible for certifying compliance with applicable water quality standards pursuant to Section 401 of the Clean Water Act, P L. 92-500 (as amended) 33 U. S. C §1341 (1988), for all activities located in whole or in part within organized municipalities subject to DEP's regulatory jurisdiction. In order to approve a certification under the Clean Water Act, DEP must find that there is a reasonable assurance that the proposed activity will not violate applicable Water Quality Standards A water classification system establishing such standards has been adopted by the State and has been duly approved by the U.S. Environmental Protection Agency pursuant to the provisions of the Clean Water Act. The decision made by the Board on the Penobscot Mills application (FERC #2458), and the terms and conditions contained therein, represents the official position of the State of Maine regarding this project.

4 The Maine Land Use Regulation Commission (LURC) is an agency of the State of Maine.

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

SPO-1 No response required.

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

COMMENTS FROM STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

with principal offices in Augusta. Maine LURC consists of a Commission of seven public members, an executive director and staff (12 MRSA §683, <u>st.seq.</u>) LURC and its staff are responsible for implementing zoning and sub-division control in the unorganized and deorganized townships of the state. The Commission was established to preserve public health, safety, general welfare, ecological and natural values in these areas and to prevent inappropriate development or intermixing of residential, recreational, commercial and industrial uses detrimental to the value of these areas, to provide and encourage sound land use planning; to prevent the development of substandard structures or structures located unduly proximate to waters or roads, to prevent despoliation, pollution and inappropriate uses of water in these areas.

5. In addition to the responsibilities described in paragraph 4., LURC is responsible for administering and enforcing the Maine Waterway Development and Conservation Act (MWDCA), 38 MRSA §§630-636, in unorganized and deorganized townships of the State

6. By authority granted in Executive Order #16 FY91/92,LURC is also responsible for certifying compliance with applicable water quality standards pursuant to Section 401 of the Clean Water Act, P.L. 92-500 (as amended) 33 U.S.C. §1341 (1988), for unorganized territories and townships. In order to approve a certification under the Clean Water Act, LURC must find that there is a reasonable assurance that the proposed activity will not violate applicableWater Quality Standards. A water classification system establishing such standards has been adopted by the State and has been duly approved by the U.S. Environmental Protection Agency pursuant to the provisions of the Clean Water Act. The decision by LURC on the Ripogenus application (FERC #2572), and she terms and conditions contained therein, represents the official position of the State of Maine regarding this project.

7 The Department of Marine Resources (DMR) is an agency of the State of Maine with principal offices in Hallowell, Maine. 12 MRSA §6001, et and DMR was established to regulate, conserve and develop marine and estuarine resources, to conduct and sponsor scientific research, to promote and develop marine coastal industries, to advise and cooperate with state, local and federal officials concerning activities in coastal waters and to implement, administer and enforce laws and regulations necessary for these purposes.

8. The Department of Inland Fisheries and Wildlife (IFW) is an agency of the State of Maine with principal offices in Augusta, Maine. 12 MRSA §7001, <u>et.seq</u>. IFW is responsible for the administration and enforcement of inland fisheries and wildlife laws and for the management of all inland fish and wildlife in the State.

9 The Atlantic Sea-Run Salmon Commission (ASRSC) is a commission that consists of the Commissioner of DMR or his designee, the Commissioner of IFW or his designee, and three members of the public. 12 MRSA §6251-A. et seq. The Commission was established to undertake projects in research, planning, management, restoration and propogation of the Atlantic sea-run salmon in the State of Maine.

10 The Department of Conservation (DOC) is an agency of the State of Maine with principal offices in Augusta, Maine 12 MRSA §5011-5012, et.seg. DOC is responsible for the review of

SPO-1 Cont

E-229

FERC licensed hydropower projects to ascertain the adequacy of recreational facilities adequacy of flow releases for boating and canoeing, effects of projects on publicly owned lands or private lands if available for public recreation.

SPO-1 Cont

SPO-2

E-230

11 The comments of the SPO in this proceeding are necessary to represent these agency interests and the public interest of the citizens of Maine Absent the State's participation and commentary in these proceedings, that public interest will not be served. The SPO is responsible for the intergovernmental coordination of state level planning activities with a goal of comprehensive planing. The SPO is also responsible for intervening on behalf of the State in all FERC licensing proceedings for projects in Maine and has intervened in the above captioned hydropower projects. In this instance, SPO represents the natural resource agencies described above in support of these comments which follow.

COMMENTS ON UPPER PENOBSCOT DEIS

Maine is committed to its current land use laws which are designed to balance the many and often conflicting uses of its public resources in the project areas. The draft EIS acknowledges Maine's effective stewardship in this regard but in the same breath requires Great Northern to forcibly acquire lands belonging to the State of Maine and private land owners, some of which have been in the same private ownership for over 100 years. That the DEIS attempts to encumber and limit use of these lands is inappropriate, inconsistent with the rights of existing users and in conflict with Maine's publicly developed, multiple use policy for shorefront lands

LURC Jurisdiction

References (page 4-45) to shoreland management on Gero Island (owned by the Maine Bureau of Public Lands), Nesowadnehunk Deadwater (under conservation easement of the State), and Caribou Lake's southern shore (already managed as eagle habitat by another timber company) are all inappropriate Management of these areas has been planned for and is administered by existing Department of Conservation (DOC)Programs and is subject to land use regulations administered by LURC LURC represents the public interest in a public process where balancing of multiple uses is paramount. We assert that the FERC staff has underestimated Maine's capability to develope creative and effective mechanisms to balance the competing needs of landowners and recreational users in the context of preserving environmental values. The State of Maine has been a very effective steward of its resources. Through a combination of regulation, public ownership and cooperation, Maine has earned a national reputation as a leader in resource protection. These State owned properties are managed within a publicly approved management plan. The DOC's Bureau of Public Lands policy is to issue no new residential or commercial leases on these lands. Moreover, timber harvesting on this property is subject to the Bureau's Integrated Resource Policy, developed after exhaustive study and public participation.

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

- SPO-2 Further staff review of LURC's land use regulations resulted in revised land use assessment and recommendations [see section 4.9]. The staff recommends under the recommended alternative two options: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion on GNP-owned lands. See section 4.9.3 of the FEIS for further discussion. For the Penobscot Mills project area, the staff recommends a 200-foot expansion of the project boundaries on GNP owned lands; existing structures would be grandfathered.
- SPO-3 See response SPO-2.

LURC is an agency of the State of Maine whose Comprehensive Land Use Plan is designed to balance the many and often conflicting uses of the public resources in the project areas. The State of Maine, based on sound management planning, has already made reasonable land use and shoreland zoning decisions which balance these resources and benefit the public interest. It is our understanding that while FERC intended the NEPA process to be an appropriate forum to resolve issues that might not be addressed in previous aspects of the hydro licensing process, the EIS procedure was not designed to be inconsistent with or deviate from existing State policy. LURC zoning, management policy and enforcement practices provide the preferred balance of our public resources and their uses. LURC is the steward and the local expert.

SPO-3 Cont In June of 1990, after having conducted an exhaustive review of the natural resources. recreational values and development pressures on the wildland lakes in its jurisdiction, LURC adopted an Amendment to its Comprehensive Land Use Plan, specifically to address and improve the managment of the lake resources in the unorganized territories. Specific zoning and regulatory standards changed as a result of this action, which provides additional protection to those lakes with exceptional values and guides lakeshore development toward those lakes best suited to accomodate additional development without adversely affecting the resource. We believe that the protection and mangement of these resources protection, recreation and development in the context of a comprehensive management study, rather than through imposition of arbitrary expanded project boundaries and requirements by FERC through the hydropower licensing process.

The DEIS characterization of the LURC regulatory program with respect to shoreline development (Section 4.9.1.1.) is misleading, as it exaggerates the actual shoreline build-out potential allowable under LURC standards based on a single factor (shoreline frontage), when in fact numerous site specific characteristics (eg. soil type, slope, lot size, access, compatability of the proposed development with surrounding natural resources and existing uses) are taken into account by LURC in determining the suitability of a shoreline parcel for development.

The timber harvesting section (Section 4.9.1.2.) of the DEIS indicates that the LURC regulations for vegetative buffers in the project areas range from 50 to 100 feet from the normal high water mark of lakes and streams and expresses concern that these buffers may not adequately protect aesthetics and water quality because the required buffers are unrelated to the slope of the land. The DEIS fails to acknowledge that, with respect to vegetative clearing for development, including tree removal, LURC regulations require a 100 foot vegetative buffer to be retained for any standing body of water 10 acres or greater and for all flowing waters draining 50 square miles or more. These requirements allow for limited cutting of trees (up to 30% provided a well distributed stand of trees is allowed to remain) and require that ground vegetation (less than 3 feet in height) remain undisturbed. Additional, less stringent clearing restrictions apply at distances from 100 feet to 250 feet on these waterbodies and, in fact, are also applied to the full 500 foot depth of the (P-AL) Accessible Lake Protection Subdistrict, which surrounds a significant portion of the project areas in LURC jurisdiction.

In addition, LURC timber harvesting regulations applying to the project areas zoned (P-GP) Great Pond Protection Subdistrict, (A-AL) Accessible Lake Protection Subdistrict and

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

SPO-4 The staff revised potential build-out estimates to reflect LURC's subdivision regulations. As stated in the DEIS, the staff acknowledges that these estimates do not account for development limitations such as steep slopes, poor soils, wetlands, or access (see section 4.9.1.1).

SPO-5 Opinion noted. See revised sections 3.11.1.1, 4.9.1.2, 4.9.2.2 and 4.9.3.2.

SPO-5

(P-SEE) Shoreland Protection Subdistrict, provide for managment of a 250 foot forested buffer around waterbodies so zoned. Specific limitations and performance based standards on timber harvesting and forest management activities are applied to maintain the aesthetic, recreational and water quality values of the area. Although limited harvesting is allowed within 50 feet of these waterbodies, it must be conducted in a manner that leaves a well distributed stand of trees and that maintains an unscarified vegetative filter strip of variable dpeth depending upon slope, between the operation and the waterbody.

Section 4 10.1.2 of the DEIS indicates that the LURC requires, " only a permit for permanent boat docks " This statement fails to clarify that such a permit is issued only by, "special exception," a category of LURC permit which, in addition to complying with all other applicable standards and requirments, requires the applicant to demonstrate that there is no reasonable alternative or suitable site available for the proposed use and that the use can be effectively buffered from all other uses and resources in the area with which it may be incompatible.

The State of Maine has no intention of selling its property within the project area. We strenuously oppose any effort to take it by emminent domain and resent being placed in a position of potential conflict with a landowner with whom we have enjoyed fruitful cooperation in the preservation of natural resources. The State of Maine requests that FERC honor existing lease arrangements in the project area and if further protection is required it should be in the form of voluntary partnership between the landowner and the State. Through a variety of programs the State of Maine has acquired land outright or negotiated easements for lands which the State has found warrant protection or preservation for public benefit. The proposed approach by FERC state to sustain cooperative, non-taking relationships with landowners.

The State of Maine requests prompt issuance of the new licenses for Ripogenus and Penobscot Mills and we request that these licenses be consistent with policies already established by the State We ask the FERC to defer to the publicly crafted and considered decisions of the states in these matters and to respect the primacy of State authority and not rely on the recommendations of out-of-state advocacy groups whose land use agenda has very little to do with hydropower licensing and balancing of immediate uses. We urge FERC to respond to local residents and State officials and revise the Final EIS to reflect the State of Maine's land use recommendations for these hydropower projects and to delete those recommendations which seek to alter existing landowner rights and relationships.

The Primacy of State Authority

The state is the stakeholder and the accountable party which lives most immediately with the consequences of these decisions. The state is the entity most familiar with its own resources and in cases where there is a conflict between the State's preference and an advocacy group or a Federal agency recommendation, we believe deference should be given to the State's preference

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

SPO-6 No response required.

- SPO-7 The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC's land use regulations. The staff's recommendations (see section 5.3.4) provide measures to protect shoreland resources within the project areas while considering existing land regulatory controls. See sections 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.
- SPO-8 Opinion noted.
- SPO-9 Opinion noted.

SPO-5

SPO-6

SPO-7

Cont

SPO-9

Fisheries Considerations

The official position of the State of Maine is concurrent with the State Water Quality. Certification and the recommendations for water levels, flows, and fisheries enhancements associated with the Ripogenus Project. Within the framework of seasonal flows that have been prescribed for the West Branch of the Penobscot River below McKay Station, maintaining stable river flows is a key component of these recommendations. We interpret stable flows to mean relatively consistent flows throughout each 24 hour period, with no short term special releases of water through Ripogenus Dam such as might occur purely for recreational boating.

Throughout the consultation process for the Ripogenus Project, IFW maintained a position that dramatic flow increases will adversely affect fishing opporutnities as well as fishing success in the West Branch Therefore, IFW opposes any last minute revisions in flow recommendations which do not thoroughly evaluate the adverse effects on fisheries. For example, any special recreational boating releases must be reviewed for their comprehensive effects upon fisheries habitat and angling use opportunity.

Wildlife Considerations/Engles

Several parties have raised concern for bald eagles in the project area. It is a State and Federally listed endangered species. The Maine Department of Inland Fisheries and Wildlife will clarify some inaccuracies found in the DEIS and recommend appropriate measures for project operation on eagles behalf

1 The West Branch basin is indeed an important area for bald eagles, especially the breeding population. Eleven of the 175 nesting pairs of eagles censused stateside in 1994 were resident in this watershed. This represents a significant inland aggregation but it is certainly an area of low nesting density relative to breeding eagle concentrations in coastal Maine.

2 The discussion of selective harvesting in Alternative 2 (page 4-45) to create new eagle perching areas, is confusing in intent and generally counter-productive to a more critical habitat management need: that of conserving nesting habitat. Harvesting in shoreland areas, an activity which is already subject to LURC standards, and the creation of snags are uncessary to enhance perching opportunities for bald eagles along lake shores which are as extensively forested as these impoundments are. A monitoring program to see if eagles perch along these manged shorelines is already incorporated into statewide nesting inventories.

3 Specific references (page 4-45) to shoreland management on Gero Island (owned by the Maine Bureau of Public Lands), Nesowadnehunk Deadwater (under conservation easement of the State), and Caribou Lake's southern shore (already managed as eagle habitat by another timber company) are all inappropriate. Great Northern Paper has a 25-year history of cooperative management of eagle nests, buffer areas and related perching/foraging sites on company-owned lands. Their resource managers have been working with DIFW to formalize long-term management of riparian zones and uplands in existing and potential eagle habitats in the project area. Forestry practices that perpetuate the availability of dominant white pines are featured in this

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

- SPO-10 We do not recommend any special boating releases from Ripogenus Dam, but do recommend notification of spill events (see section 4.8.3.1).
- SPO-11 Your statement that bald eagle is a state and federal listed endangered species is partially incorrect. The federal status of Baid eagle was officially downgraded under the Endangered Species Act from Endangered to Threatened in July 1995. It is still listed by the state of Maine, however, as endangered. We agree that the density of nesting eagles is low in the project area compared to concentrations in coastal Maine. Concerning the potential for creating snags in the vicinity of the project, we recommend that GNP confer with FWS and MDIFW to investigate the appropriateness of such actions. The availability of perching sites may be equally important as the availability of food resources for wintering bald eagles. Research has indicated that the most frequently used bald eagle perches are large mature live trees and snags that are no more than 15 meters from a shoreline. We have seen nothing in the record indicating that there is an over-abundance of such large trees and snags in this vicinity of the project shoreline. We view such enhancements as potentially appropriate, given the fact that the availability of perching sites are critical to wintering eagles, that such habitats would be relatively easy to create and are environmentally benign, and that such sites would only be created in consultation with FWS and MDIFW. Concerning shoreland management for eagles on Gero Island, Nesowadnehunk Deadwater, and the southern end of Caribou Lake, we have seen nothing in the record that indicates specifics for any existing cooperative habitat management program in these areas between GNP and MDIFW; therefore we cannot comment on it. Concerning vegetative buffers and conservation easements, building setbacks required under Alternative 1 could provide a minimum of habitat protection for nesting and non-nesting bald eagles. We have revised Table 5-2 to reflect that the proposed conservation easement and/or boundary expansion under Alternative 2 could also provide enhanced habitat protection for bald eagles. There is no reason that the protections under these alternatives would preclude GNP from further limiting development around areas they have designated as eagle nesting sites on their lands.

SPO-11

strategy

4 Vegetative buffers and conservation easements are discussed at length relative to water quality and aesthetics (Table 5-1) but the narrative does not reference these measures as enhancements for bald eagles. However, in the summary of environmental consequences. (Table 5-2), these measures are listed as enhancements for bald eagles and other endangered species. The benefits derived from use of vegetative buffers and conservation easements is defensible only in the context of possible water quality, and aesthetic improvements. The buffer/easement corridors cited in both alternatives are inadequate for eagles nesting habitat requirements which are the management priority. for this endangered species

Other eagle habitat uses (perching and foraging) are not well served by generic management prescriptions. Eagle activities away from nests have not required specific management because they have proven thus far to be in harmony with shoreland zoning. If shoreland buffers and easements were to be designed for eagles, they should focus on nesting areas as the priority. If enhancements of potential eagle habitat are an objective, shallow and slow flowing waters would accomodate their foraging needs.

Water Quality Certification Conditions

As noted in the DEIS, the Maine LURC and the DEP issued Water Quality Certifications for the Ripogenus Project and the Penobscot Mills Project on April 15, 1993 and April 22, 1993 respectively Each of these certifications is subject to a number of special conditions which are summarized on pages 2-17 through 2-21 of the DEIS

The FERC staff takes the position in the DEIS, in accordance with a recent Commission licensing decision (Tunbridge Mill Coroporation, 68 FERC ¶61,078, issued July 15, 1994), that FERC has the authority to review the validity of water quality certification conditions, that only "water quality related" conditions are lawful and are thus subject to inclusion as mandatory conditions of a license and that conditions unrelated to water quality are unlawful and are thus not subject to inclusion as mandatory conditions in a license.

After examining the State's certifications for the Ripogenus and Penobscot Mills Projects in accordance with the <u>Tunbridge</u> decision, the FERC staff concludes that several of the WQC conditions are not water quality related. These conditions include, for the Ripogenus Project, Condition #5 (requiring LURC approval for any changes in project as proposed) and Condition #8 (limiting applicability of the certification to upstream projects); and for the Penobscot Mills Project, Condition #6 (Millinocket Lake Storage Dam) and Condition #8 (North Twin, Millinocket, Dolby, and East Millinocket Dams) each requiring consultation regarding a study of navigation hazards on the applicable project impoundments.

The State strongly disagrees with FERC's position on its authority to review the validity of water quality certification conditions. The <u>Tunbridge</u> decision effectively changes FERC's longstanding practice of leaving any review of the propriety of WQC conditions to the State courts.

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

SPO-12 Opinions noted. The Commission's final determination on the legal status of the State of Maine's section 401 WQC conditions will be documented in any license orders for the projects.

SPO-11

Cont

We believe that this practice was in keeping with the express language of the Clean Water Act and controlling case law and should be continued. Further, we assert that FERC has no authority to either accept or reject state-imposed WQC conditions and that these conditions are subject to review by Maine's courts and ultimately, the U.S. Supreme Court. See generally, <u>PUD No.1.y.</u> <u>Washington Dept. of Ecology</u>, <u>U.S.</u>, 114 S.Ct. 1900, 1907-08, <u>Roosevelt Campobello</u> <u>Internation Park v. U.S.E.P.A.</u>, 684F2d 1041, 1056.

The State contends that all of the conditions attached to the LURC and DEP certifications for the Ripogenus and Penobscot Mills Projects set forth limitations and monitoring requirements necessary to assure that the projects will comply with State water quality standards, and that all such limitations and monitoring requirements, "shall become a condition on any Federal license," for the projects, in accordance with Section 401(d) of the Clean Water Act

The State of Maine is authorized to impose conditions necessary to insure that the continued operation of the Ripogenus and Penobscot Mills Projects will comply with all aspects of Maine's water quality standards, including designated uses, numeric and narrative criteria, and the State's antidegradation policy. See PUD No. 1 v. Washington Dept. of Ecology, $_U.S._$, 114S ct. 1900 FERC staff has ignored this holding, as all of the conditions determined by the FERC staff to be, "outside the scope," of the WQC are in fact necessary to (a) to meet the State's designated uses of the affected waters for recreation in and on the water and other lawful uses, (b) to satisfy the provision of the State's antidegradation policy that existing in-stream water uses and the level of water quality necessary to protect those existing uses must be maintained and protected, and (c) to insure that changes in project facilities or operation that may affect water quality are approved by the State.

Furthermore, under applicable agency enabling statutes and regulations, the terms and conditions of the LURC and DEP certifications for the Ripogenus and Penobscot Mills Projects were subject to appeal by Great Northern Paper during the 30 days following each respective decision date. GNP did not appeal the conditions of the certifications within this jurisdictional time period and thus has effectively agreed to comply with all conditions atched to the certifications. Compliance with these conditions is now enforceable by LURC and DEP under State law. We expect, and LURC and DEP will enforce, GNP's compliance with all WQC conditions, whether or not they are included in any FERC license for the projects.

Conclusion

The State of Maine requests that the new licenses for Ripogenus and Penobscot Mills be consistent with Maine State policy and that the FERC defer to the carefully crafted and considered decisions of the State in the matters discussed above. By deferring to the primacy of State authority FERC will be serving the true public interest rather than the interest of national advocacy groups whose agendas may be less sensitive to State and local interests than to winning support from their memberships whose concerns may be more philosophical than realistic. The DEIS does not recognize the balance reached by LURC in balancing project uses with non-project

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

SPO-13 No response required.

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

SPO-13 Cont uses in a public process. If further protection is required it should be in voluntary cooperation between the landowner and the State. We urge FERC to revise the Final EIS to reflect the positions of the State and local residents and to delete or modify recommendations which would alter existing landowner rights and land use options.

Wherefore, the Maine State Planning Office respectfully requests that these comments be given due weight and consideration by the FERC

Maine State Planning Office

Evan D Richert February 17, 1995 Maine State Planning Office State House Station #38 184 State Street Augusta, Maine 04333 207-287-3261

.

RESPONSES TO STATE PLANNING OFFICE ON UPPER PENOBSCOT RIVER BASIN DEIS

CERTIFICATE OF SERVICE

I hereby certify that I have served a copy of the foregoing Comments upon the parties in this proceeding by mail, prepaid, addressed to the parties herein as follows

(Name and Address of Applicant's Representative)

Mr Brian Stetson

Manager Environmental Affairs

Great Northern Paper Company/Bowater Inc.

One Katahdin Avenue

Millinocket, Maine 04462-1398

207-723-2664

Done and Dated at Augusta Maine this seventeenth day of February, 1995





The comments include discussion of these important points:

 The Schedule for the Final EIS Should Be Expedited. The licensing and related environmental review process is currently well behind the schedule detailed in the second scoping document for these projects. A release of the DEIS was predicted for the spring of 1994, with a final EIS in the fall of 1994. Given that the DEIS issued as of the end of November 1994, we urge FERC to issue a final EIS and a license by early summer 1995.

 The Final EIS Should Give Greater Consideration to the Broad Socioeconomic Impacts of Any Further Enhancements.
We gratefully acknowledge FERC's appropriate consideration

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-1 Opinion noted.

T&AS-2 The staff finds no conclusive evidence, based on our economic analysis, that the Applicant's Proposal or either version of Alternative 2, would adversely affect the competitive position of GNP's mills. These alternatives would produce only small increases in annual power costs, and small declines in annual power output, ultimately, these small changes would not translate into a significant increase in the cost of production at the two mills. As a result, these alternatives would not result in production cutbacks and the accompanying indirect regional losses in employment and income. The negative annual net benefits under Alternative 1 would be large enough to adversely affect the competitiveness of the two mills.

> The potential multiplier effects of closing a plant that could occur under Alternative 1 have been decreased from the levels presented in the DEIS, although there would still be a significant adverse economic effect on the regional economy if a shutdown took place. The multiplier effects presented in the FEIS are for the State of Maine's economy.

T&AS-1

T&AS-2

Loss - ashell, Selectary Federal Energy Keolia my Commission February 21, 1997 Page -

in the DELS of second compacts, particularly to those of the three towns in the immediate Katahdin Region.

Between 1980 and 1992, the population of the three area towns dropped by 7.5%, as compared to a statewide increase of 9.9% during the same period, representing a gap of 17.4%. The Great Northern mills have seen an approximate 50% work force reduction in the last decade, resulting in the loss of 2,000 well paid jobs. Federal and State agencies have had to respond to the new economic challenges facing the Katahdin region by investing several million dollars to stabilize the region's economy.

However, beyond the impacts to the Katahdin region, we recommend that FERC give careful consideration to the socioeconomic impacts of these Projects throughout Central and Northern Maine, and the State of Maine generally. Some examples include:

- As to regional impacts, in 1993, Great Northern's employees represented about 20% of the manufacturing employment base of all of Penobscot County. Indirect impacts extend to Piscataquis, Aroostook, and Washington Counties, which have seen the loss of hardwood mills, a 1994 potato blight, the closure of Loring Air Force Base, and potentially, the loss or reduction of an east-west rail line.
- Statewide, papermaking is important to the entire State economy. Paper accounts for 35% of the State's manufactured product and, with lumber and wood products, accounts for almost half of the State's manufactured product. The overall health of Maine's forest products economy is tied to competitive mills, including the Great Northern mills.

We recommend that FERC interview the proper officials at the Maine State Planning Office ("SPO") and Eastern Maine Development Corporation as to the broad socioeconomic impacts of the Projects.

As the DEIS reports, Great Northern's competitive position in the energy intensive paper industry depends on the availability of a reliable source of inexpensive electric power. Imposing additional production costs for replacement RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-2

Cont'd

Logic [asher1, Secretary] Federal Energy Regulatory Commission February 21, 1995 power and mandated land conservation will adversely affect the viability of the Millinocket and East Millinocket mills. which FERC properly characterize as the economic base for the Towns; as noted, the importance of the mills extends to Northern and Central Maine, and the State more generally. 3. Great Northern's Water Use Plan Requires No Further Changes Beyond Those Recommended in the DEIS. For a century, these hydro projects have provided very substantial natural resource and societal benefits. Those benefits now include: A world class landlocked salmon fishery: White water rafting on the West Branch, some of the best technical rafting on the East Coast; Significant flood control; and

> A guaranteed flow of 2000 cfs at Millinocket which provides instream river flows that enhance aquatic life.

These benefits exist because of the dams and how they are managed; they result from the storage and steady metering of water in and through the hydro system.

The Water Use Plan, as fine tuned during the State 401 certification proceedings and FERC's proceedings (including the summer 1993 negotiations in Orono, Maine with interest groups), only enhances these environmental, recreational and resource benefits.

FERC has correctly concluded that, given the far greater fisheries habitat and recreational opportunities present on other sections of the river system and the economic costs of greater flows, the EIS should recommend:

3

1. In Millinocket Stream, a minimum flow of 60 cfs between May 1 and October 15, and 20 cfs (rather than 30 cfs) during the rest of the year;

2. In the Back Channel, leakage and spillage flows;

3. In the Upper Gorge, only leakage from fall through spring; otherwise, the same as those under the

T&AS-3 No response required.

T&AS-2

Cont'd

T&AS-3

Page 3

.

٠

٠

page Cashel: Secretary Federal Energy Regulatory Commission February 21, 1995 Fage 4

Applicant's Proposal (as outlined in the Application, and the DEIS); and

4. At all other locations, the same as those under the Applicant's Proposal.

4. FERC Should Not Require an Expansion of the Project Boundaries. FERC should carefully re-evaluate the wisdom and costs of expanding the project boundaries. Before FERC succumbs to the allure of the environmental Coalition's logic for mandated land conservation, the Town and ASSSET respectfully ask that FERC consider:

• Deference to LURC. Consistent with well established principles of federalism, FERC should give Strong consideration and, as appropriate, defer to the State's regulation of lands in the unorganized territories, including those around the Ripogenus Impoundment. The proof of the State's success in managing thes shorline is the good water quality and aesthetics that FERC now seeks to protect. LURC, which oversees nearly twothirds of the State's lands, has effectively regulated these lands for many years as the steward of Maine's vast natural resources. Such land use regulation is largely a State matter. As Steve Adams of the Maine SPO quoted from the President's January 24, 1995 State of the Union address:

If the States can do it, we should let them do it. If they can do it better, we should get out of the way.

Before imposing this federal mandate, FERC should first consult with the State agencies who currently administer the State's land management and water quality protection programs.

* The Private and Public Landowners on Ripogenus Impoundment are Good Stewards. There is a strong tradition of resource stewardship practiced by the public and private landowners on the Ripogenus Impoundment. As stated in the testimony of Larry Philbrick, the private landowner tradition on the Rip Impoundment extends through five generations. The State's stewardship of lands in the area is also notable and includes Baxter State Park and, more

4

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-4 Opinion noted. Further staff review of LURC's land use regulations resulted in revised land use assessment and recommendations (see section 4.9).

T&AS-4

in affinity Secturary Federal Energy Regulatory Commission February 21, 1995 Fau - 5 specifically, Gero Island in the middle of Chesuncook Lake * Socioeconomic Impacts. The broad socioeconomic impacts of such measures on the majority of people who currently own or occupy land on the Project Impoundments and will be negatively impacted by the proposed land conservation requirements. * Direct Economic Impact on Great Northern/ Acquisition Cost. The economic impact of the land acquisition mandate on Great Northern. In the DEIS, FERC estimates that the cost of acquisition only is \$2,000,000, using the price per acre value of a "comparable sale." The \$2 million cost estimate which FERC assigns to the proposed land conservation proposal equates to \$4.59 per waterfront foot of land to be acquired. As set forth in the consulting report of Lowell T. Sherwood, ASA (Certified General Appraiser- WE #9), FERC derives that cost estimate from the voluntary discussion. "Rangeley Lake" acquisition by the Rangeley Lakes Beritage Trust. Indeed, that acquisition (on Mooselookmeguntic and Cupsuptic Lakes) assumes a price per waterfront foot of \$51 for a conservation easement and \$71 for the fee title. Combining severance damages and other likely costs in the eminent domain process (because both the State of Maine and the larger landowners on the Rip Impoundment have publicly stated their unwillingness to sell voluntarily), the likely cost falls within a range between \$13,000,000 and \$43,000,000, with an economically realistic cost of \$22,000,000 (based on the Rangeley Lakes Heritage Trust acquisition). Economic Impact on Great Northern to Conserve Lands. The economic impact of mandatory land conservation on Great Northern is also far more significant than FERC suggests, if at all, in the DEIS. As stated in the consulting report of Lowell T.

Sherwood, the "lost opportunity" cost to Great Northern of subjecting its own land on Rip Impoundment alone to a conservation easement equates to the likely cost of having to acquire the fee title or conservation easement on lands it does not own. Whereas Great

5

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-5 The staff revised its land use assessment and recommendations, and under the recommended alternative, the proposed easements and/or project boundary expansion would be only on GNP-owned land (see section 4.9). The staff determined, therefore, that the potential socioeconomic effects would be insignificant.
 - T&AS-6 The staff revised potential cost estimates for the shoreline easements upon review of comments received during the DEIS comment period and updated land valuation information. The staff estimated that the potential cost of approximately \$24.6 million, based on waterfront footage for the easements proposed in Alternative 1, would be greater than previously determined in the DEIS. Our evaluation of benefits suggests that the additional protection of the 500-foot expansion does not merit the much higher cost of that alternative as compared to the recommended alternative (\$24.6 million versus no direct costs). See section 4.9 for further

m -243

TRAS-5

T&AS-6

. — — Andria Sertelany Рефия. Energy Regulatory — элизэнэ Рефикату 21, 1995 Рафи С

> Northern would have to acquire 82.5 miles of shorefront from other landowners, it would have to give up developments rights on its 82 miles of shore frontage. The details supporting the discussion in this paragraph and the foregoing paragraph are contained in the Comments and the Sherwood report, attached as Exhibit A to the Comments.

* Speculative Positive Impacts. Such adverse impacts should be balanced against the speculative positive impacts in the eyes of "back country canonists" and other recreationalists who yearn for the remote "wilderness" experience free of human habitations.

The undersigned have each engaged in such recreation, albeit on the Allagash Wilderness Waterway. The remote wilderness experience abounds in Maine, but it is a genuine stretch of logic to characterize the present day working forest lands near the Ripogenus impoundments as remote "wilderness."

Review LURC Study Results. With regard to the mandatory land acquisition proposal for the Ripogenus Project, we also recommend that FERC carefully review the study results of the Maine Land Use Regulation ("LURC") in a Fact Sheet dated March 6, 1990 and titled "Land Use Plan and Rule Amendments Regarding the Development and Conservation of Lakes in Maine's Unorganized Areas", a copy of which is attached to our Comments as Exhibit B.

We ask FERC to consider that the need, if any, does not (1) warrant an intrusion into the Maine Land Use Regulation Commission's traditional jurisdiction for regulating such shorelands, (2) warrant disrupting the longstanding private and public landowner tradition and stewardship in the ownership of such lands, (3) justify the significant socioeconomic impacts that the imposition of such a mandate would cause, and (4) justify the very substantial direct economic costs Great Northern would have to incur in order to acquire lands it does not own and to conserve its own lands.

In conclusion, we believe that FERC has done an excellent job in sorting through the often competing, and sometimes inconsistent, requests and demands of the participating agencies and groups. We thank you for this opportunity to comment and

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-7 See response T&AS-4.

™ ₽**18AS-**7

T&AS-6

Cont'd

COMMENTS FROM THE TOWN OF MILLINOCKET AND ASSSET

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

Lui: Cashell, Secretary Federal Energy Regulatory Commission February 21, 1995 Page 7

T&AS-7wish FERC Staff and the Commission well as they develop the Final
Environmental Impact Statement and prepare to issue a final
license.

Andrew Hamler

7.

P. Andrew Hamilton Coupsel for ASSSET DOCCON Dean A. Beaupain

Dean A. Beaupain Counsel for the Towns

cc: Service List

1 VOI 76717AHILTEVCASOROS

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

TRUTED STATES OF AMERICA BEFORE THE FEDERAL ENERGY REGULATORY COMMISSION

- 1

- 1

Great Northern Paper

Penobscot Mills Project No. 2458, Maine and Ripogenus Project No. 2572, Maine

COMMENTS OF ASSSET, TOWN OF MILLINOCKET, TOWN OF EAST MILLINOCKET, MILLINOCKET CHAMBER OF COMMERCE, EAST MILLINOCKET.MEDNAY CHAMBER OF COMMERCE, EASTERN MAINE DEVELOPMENT CORPORATION, KATAHDIN REGIONAL DEVELOPMENT CORPORATION, INTERNATIONAL ASSOCIATION OF MACHINISTS AND AEROSPACE WORKERS LOCAL #156, UNITED BROTHERHOOD OF CARPENTERS AND JOINERS LOCAL #658, UNITED PAPERWORKERS INTERNATIONAL UNION LOCAL #24, INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL #567, AND UNITED PAPERWORKERS AND PLUMBERS LOCAL #685, UNITED PAPERWORKERS INTERNATIONAL LOCAL #685, UNITED PAPERWORKERS INTERNATIONAL DOCAL #12 CONCERNING DAFT ENVIRONMENTAL IMPACT STATEMENT

> By: Dean A. Beaupain, Esq. 4 Hill Street Millinocket, Maine 04462

> > And

P. Andrew Hamilton, Esq. Eaton, Peabody, Bradford, and Veague, P.A. P. O. Box 1210 Bangor, Maine 04402-1210

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

WRITTEN COMMENTS ON DRAFT ENVIRONMENTAL IMPACT STATEMENT TABLE OF CONTENTS

INTRODUCTION

- 1.0 PURPOSE AND NEED FOR POWER
 - 1.2 NEED FOR POWER
 - 1.3 SCOPE OF THE EIS
 - 1.3.2 Scoping Meetings
- 2.0 PROPOSED ACTION AND ALTERNATIVES
 - 2.4 ECONOMIC COMPARISON OF ALTERNATIVES
 - 2.3 PROJECT ALTERNATIVES
 - 2.3.1 Alternative 1
 - 2.3.2 Alternative 2
 - 2.4.1.2 Economic Comparison Alternatives 1 & 2
 - 2.4.1.3 Economic Comparison Alternatives 1 £ 2
 - 2.4.2.1 Cost Assessment of Potential Alternative Power Sources
 - 2.4.2.2 Estimated Cost of Lost Power Generation
 - 2.4.2.3 Estimated Cost of Lost Power Generation -Applicant's Proposal
 - 2.4.2.4 Estimated Cost of Lost Power Generation -Alternative 1
 - 2.4.2.5 Estimated Cost of Lost Power Generation -Alternative 2
 - 2.4.2.6 No-Action Alternative

3.0 AFFECTED ENVIRONMENT

- 3.3 STREAMFLOW
 - 3.3.1 Regional
 - 3.4 WATER QUALITY
 - 3.4.1 Regional
 - 3.5 FISHERIES RESOURCES
 - 3.5.2.12 Back Channel
 - 3.6 WETLANDS
 - 3.6.1 Basinwide
 - 3.6.2 Ripogenus Project
 - 3.6.3 Penobscot Mills Project
 - 3.11 LAND USE ISSUES
 - 3.11.1.1 Maine Land Use Regulation Commission
 - 3.11.1.4 Maine Forest Practices Act (MFPA)
 - 3.11.3 Proposed Land Uses
 - 3.12 AESTHETIC RESOURCES
 - 3.13 CULTURAL RESOURCES
 - 3.13.1.3 Pulp & Paper Industry
 - 3.14 SOCIOECONOMIC RESOURCES
 - 3.14.1.1 Population
 - 3.14.1.2 Employment
 - 3.14.1.3 Income
- i

COMMENTS FROM THE TOWN OF MILLINOCKET AND ASSSET

4.0 ENVIRONMENTAL IMPACTS 4.1 GEOLOGY AND SOILS

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

	4.1.1 Applicant's Proposal		
4.2	STREAMFLO	LOW	
	4.2.2	Alternative 1	
	4.2.3	Alternative 2	
	4.2.5	Fisheries - Summary	
4.3	WATER QU	ALITY	
	4.3.1.3	Back Channel	
4.4.	FISHERIES	S RESOURCES	
	4.4.2.1	Alternative 1 - Upper Gorge Flows	
	4.4.2.2	Alternative 1 - West Branch Flows	
	4.4.2.4	Alternative 1 - Back Channel Flows	
	4.4.2.5	Alternative 1 - Impoundment Drawdowns	
	4.4.3.1	Alternative 2 - Upper Gorge Flows	
	4.4.3.2	Alternative 2 - West Branch Flows	
	4.4.3.3	Alternative 2 - Millinocket Stream Flows	
	4.4.3.4	Alternative 2 - Back Channel Flows	
	4.4.3.5	Alternative 2 - Impoundment Drawdowne	
	4 4 4	No.action Alternative	
45	WETLANDS	NO ACTION AICEINACIVE	
	4 5 1	Applicant's Proposal	
	4 5 1 2	Wetlands Ethencement	
	4.5.1.4	Network Sumancement	
	1.3.2	Alteinative 1	
	9.3.3	Alternative 2	
	TEDBECTON	NO ACTION AILEFNALIVE	
4.0	IERRESIRI	AL RESOURCES	
	4.0.1	Applicant's Proposal	
	4.6.1.1	Site Specific Impacts	
	4.6.2	Alternative 1	
	4.6.3	Alternative 2	
4./	THREATENE	D AND ENDANGERED SPECIES	
	4.7.1	Applicant's Proposal	
	4.7.2	Alternative 1	
4.8	RECREATIC	resources	
	4.8.1	Applicant's Proposal	
	4.8.1.1	Recreation Flows	
	4.8.1.5	Access rees	
	4.8.2	Alternative 1	
	4.8.2.2	water level fluctuations	
	4.8.2.3	Recreational fisheries	
	4.8.2.4	Recreation access and facilities	
	4.8.3	Alternative 2	
	4.8.3.1	Recreational Flows	
	4.8.3.3	Recreational Fisheries	
	4.8.3.4	Recreation Access and Facilities	
	4.8.3.6	Cumulative Impacts	
4.9	LAND USE		
	4.9.1	Applicant's Proposal	
	4.9.1.1	Shoreline Development	
	4.9.1.2	Timber Harvesting	
	4.9.1.3	Expansion of Project Boundaries	
	4.9.2	Alternative 1	

ii

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

•

- 4.9.2.3 Expansion of Project Boundaries
- 4.9.2.4 Cumulative Impacts
- 4.9.3 Alternative 2
- 4.9.3.1 Shoreline Development
- 4.9.3.2 Timber Harvesting Practices 4.9.3.3 Expansion of Project Boundaries 4.9.3.4 Cumulative Impacts
- 4.10 AESTHETIC RESOURCES

 - 4.10.1 Applicant's Proposal 4.10.1.1 Water Level Fluctuations
 - 4.10.1.2 Development Along Shorelines
 - 4.10.1.3 Influence on Forestry Practices
 - 4.10.3 Alternative 2
 - 4.10.3.2 Development Along Shorelines
 - 4.10.3.3 Influence on Forestry Practices
- 4.11 CULTURAL RESOURCES
 - 4.11.1.2 PIN Claims to Islands in the West Branch
 - 4.11.1.3 Cumulative Impacts
 - 4.11.2 Alternative 1
 - Alternative 2 4.11.3
 - 4.11.4 No-Action Alternative
- 4 12 SOCIOECONOMIC RESOURCES
 - 4.12.1.1 Applicant's Proposal Employment and Fiscal Impact
 - 4.12.2.1 Alternative 1
 - 4.12.2.2 Effect on GNP Operations
 - 4.12.2.3 Cumulative Impacts
 - 4.12.3.1 Alternative 2 Employment and Fiscal Impact
 - 4.12.3.2 Effect on GNP Operations
 - 4.12.3.3 Cumulative Impacts
 - 4.12.4 No-Action Alternative
- 4.15 RELATIONSHIP BETWEEN SHORT TERM USES AND LONG TERM PRODUCTIVITY
- 5.3 ENVIRONMENTAL COMPARISON OF ALTERNATIVES
 - 5.3.1 Streamflow
 - Fisheries Enhancement 5.3.2
 - 5.3.3 Wetlands
 - 5.3.4 Land Use
- 5.3.5 Economics and Socioeconomic Effects
- 5.4 SUMMARY OF FINDINGS AND RECOMMENDATIONS
- 5.5 STATUTORY REQUIREMENTS
- 5.5.1 Water Quality Certificate Conditions

iii

5.6 SECTION 10(J) RECOMMENDATIONS

CONCLUSION

EXHIBITS

EXHIBIT A -

Consulting Land Appraisal Report of Lowell T. Sherwood, ASA Concerning Conservation Easements on 82.5 Miles of Shore Frontage on Ripogenus Impoundment.

EXHIBIT B -

Fact sheet of Maine Land Use Regulation Commission entitled "Land Use Plan and Rule Amendments Regarding the Development and Conservation of Lakes in Maine's Unorganized Areas."

EXHIBIT C -

Memorandum from James Haskell, Millinocket Town Planner, to James Kotredes, Millinocket Town Manager, dated February 16, 1995, attaching Table entitled "Great Northern Paper Company Employee Levels: 1985-1994."

EXHIBIT D -

Testimony of David Cole, Vice President of Eastern Maine Development Corporation, at FERC's January 25. 1995 Public Hearing on DEIS.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN
Comments In re: FERC's Draft Environmental Impact Statement Great Northern Paper Co. Penobscot Mills Hydroelectric Project (FERC No. 2458) Ripogenus Hydroelectric Project (FERC No. 2572)

INTRODUCTION

We commend you for a thorough effort and generally careful analysis of a significant range of complex economic and environmental issues. Moreover, we are impressed that FERC is now looking to complete the Environmental Impact Analysis promptly so that licenses can issue by early summer 1995.

However, there are several issues where we believe your analysis needs to be corrected by either minor adjustments or wholesale change, including the following:

 <u>Streamflows</u>. Your analysis of the Applicant's data concerning availability of water for Back Channel flows is inaccurate and flawed. The Applicant's data shows sufficient water is <u>not</u> available to sustain flows of either 350 cfs or 165 cfs, without violating water quality certifications issued by the Maine Land Use Regulation Commission and the Maine Department of Environmental Protection. These State certifications are FERC's legally binding, baseline constraints on the Applicant's operation of the project (as FERC properly notes at Section 5.5.1).

Notably, fisheries resources would also be jeopardized by such flows. Accordingly, while we agree with your conclusion that flows of either 165 cfs or 350 cfs should not be required, we assert that such a conclusion results not only from economic, but also from environmental requirements. We believe that your over-reliance on "average year" model simulations is misplaced and leads you to incorrect conclusions as to the availability of water for any significant Back Channel flows.

You have correctly concluded that higher streamflows are in many cases not warranted, given the far greater fisheries habitat and recreational Opportunities present in other sections of the river system. For the reasons detailed in these comments, we recommend that you adopt the following streamflows:

 a. in Millinocket Stream, a minimum flow of 60 cfs between May 1 and October 15, and 20 cfs during the rest of the year;

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-8 We have clarified the EIS to indicate that not all flow-related enhancements could be satisfied simultaneously if flows are provided to the Back Channel in addition to the 2000 cfs minimum flow required at Millinocket. We provide more detailed comments on various aspects of the water use model and streamflow requirements below.

T&AS-8	b. in the Back Channel, leakage and spillage flows;
Cont'd	c. in the Upper Gorge, only leakage from fall through July 1; otherwise, the same as those outlined under the LURC water guality certificate (as outlined in the DEIS); and
	d. at all other locations, the same as those under the Applicant's Proposal.
	 <u>Fisheries</u>. Your discussion of impacts on fisheries omits significant environmental impacts to fisheries by

TEAS-9 Initing the discussion to "average" years. The data demonstrates that you should (1) reconsider your conclusion that sufficient water is available for substantial Back Channel flows; and (2) discuss the fisheries impacts on West Branch salmon stocks below McKay Station and on establishment of a self-sustaining togue population. Your conclusion that flow enhancements would cause only minor changes in impoundment drawdowns may be accurate in average years, but it is certainly not accurate for drawdowns in worst case years. As an immediate example, requiring a flow of 165 cfs down the Back Channel during dry and worst case years would result in outflows from Ripogenus being reduced "below desired levels for 4 to 5 weeks."

> 3. <u>Wetlands</u>. In its Relicensing Regulations, FERC states its policy that "evaluation and consideration of the appropriateness of requiring enhancement measures is done in the context of today's environment and in relation to today's needs and problems, not in the context of the world as it existed 50 years ago." Similarly, in discussing water flows in the Back Channel, FERC specifically acknowledges that <u>existing</u> conditions provide the baseline for its environmental analysis. See section 4.3.1.3.

However, in analyzing wetlands impacts, FERC goes against its stated policy and its analysis of other environmental issues and looks to rectify "continuing adverse impacts." The language and analysis in the DEIS presupposes that <u>present</u> project operations are resulting in water level fluctuations that are now causing adverse impacts to wetlands, which would require mitigation even under the "no action" alternative.

You should reexamine your analysis of wetlands impacts and substantially reduce the mitigation required of the Applicant.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-9 Detailed responses are provided below.
- T&AS-10 Detailed responses are provided below.

E-25:

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-11 See response T&AS-4.

- 4 Expansion of Project Boundaries. You should carefully T&AS-11 reevaluate the wisdom, need and cost of expanding the project boundaries. Before you conclude that aesthetic impacts alone justify such radical surgery, we ask that FERC consider the following:
 - Deference to LURC. Consistent with well а. established principles of federalism, FERC should give strong consideration and, as appropriate. defer to the State's regulation of lands in the unorganized territories, including those around the Ripogenus impoundment. LURC, which oversees nearly two-thirds of the State's lands, has effectively regulated these lands for many years as the steward of Maine's vast natural resources. Such land use regulation is largely a state matter. Before imposing this federal mandate to expand project boundaries, FERC should first consult with the state agencies who currently administer the state's land management and water quality protection programs and reevaluate the need for further federal protection.
 - Land stewardship; unwilling sellers. The private b. and public landowners on Ripogenus impoundment are good stewards, have fought hard in getting and/or defending private and public landowner rights, and will not willingly give up those rights. There is a strong tradition of resource stewardship practiced by the public and private landowners on the Ripogenus impoundment. As stated in the testimony of Larry Philbrick, the private landowner tradition on the Ripogenus impoundment spans five generations. State stewardship of lands in the area is also notable and includes Baxter State Park and, more specifically, Gero Island in the middle of Chesuncook Lake.

As Steve Adams (State Planning Office). Senator Leo Keiffer (Maine Senate Majority Leader), and Representative Richard Gould (Cochair of the Maine Legislature's Natural Resources Committee) stated at the public hearing on January 25, 1995, the State is not interested in parting with public lands. Mr. Philbrick, on behalf of the many private landowners on Ripogenus impoundment, similarly observed that the private landowners on Ripogenus impoundment would not be willing sellers. Condemnation of such lands will therefore be required to implement an expansion of project boundaries on non-Applicant lands.

3

E-253

c. Undetermined socioeconomic impacts. You have yet to assess the broad, and as yet undetermined, socioeconomic impacts of this measure (expanding project boundaries) on the majority of people who currently own or occupy land on the project impoundments and who would be negatively impacted by the proposed mandated land conservation requirements.

d. Direct economic impacts/Great Morthern's acquisition cost. The economic impact of the land acquisition mandate on Great Northern has been severely undervalued. In the DEIS, FERC estimates that the cost of acquisition alone is \$2 million, using a "price per acre" value of a "comparable sale." The \$2 million cost estimate which FERC assigns to the proposed land conservation proposal equates to \$4.59 per waterfront foot of land to be acquired. As U.S. District Court Judge Hornby stated in a recent federal court decision here in Maine, the proper valuation basis for waterfront land is price per waterfront foot, not price per acre. See Exhibit A.

As set forth in the consulting report of Lowell T. Sherwood, A.S.A. (Certified General Appraiser - ME #9), a copy of which is attached as Exhibit A to these comments, FERC derives that cost estimate from a voluntary "Rangely Lake" acquisition by the Rangely Lakes Heritage Trust. Indeed, this acquisition (which actually occurred on Mooselookmeguntic and Cupsuptic Lakes) assumes a price per waterfront foot of \$51.00 for a conservation easement and \$71.00 for fee title. Combining severance damages and other likely costs as part of an eminent domain process (because neither the State of Maine nor the larger landowners on the Ripogenus impoundment will voluntarily sell), the likely cost falls within a range between \$13,068,000 and \$43,560,000. This cost is far in excess of \$9.5 million, which the DEIS found unacceptable.

e. Economic Impact/Great Northern's Mandatory Conservation of Its Own Lands. The economic impact of mandatory conservation on Great Northern is also far more significant than FERC suggests in the DEIS. Indeed, the DEIS does not discuss the "lost opportunity" cost to Great Northern of subjecting its own land on Ripogenus impoundment to a conservation easement. As noted in the consulting report of Lowell T. Sherwood, that lost

4

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-12 See response T&AS-5.

T&AS-13 See response T&AS-6.

T&AS-12

opportunity cost likely equates to the cost of having to acquire a fee title or a conservation easement on lands it does not own. Whereas Great Northern would have to acquire 82.5 miles of shorefront from other landowners, it would also have to give up development rights on its 82 miles of shore frontage.

- f. Speculative Positive Impacts. As stated in our comments as to recreational issues, the extreme minority of the boating population on the impoundments engage in back country canceing, which the environmental intervenors in part assert as the basis for the mandatory land conservation. Whereas many of the signatories to these comments have engaged in back country canceing, they have done it in Maine on the Allagash Wilderness Waterway. The remote wilderness experience abounds in Maine, but it is a genuine stretch of logic to characterize the present day working forest lands near the Ripogenus impoundments as remote "wilderness."
- 9. Review LURC Study Results. With regard to the mandatory land acquisition proposal for the Ripogenus Project, we also recommend that you carefully review the study results of the Maine Land Use Regulation Commission (*LURC*) in a fact sheet dated March 6, 1990 entitled *Land Use Plan and Rule Amendments Regarding the Development and Conservation of Lakes in Maine's Unorganized Areas,* a copy of which is attached to our comments as Exhibit B.

We ask you to consider that the need, if any, does not (1) warrant an intrusion into the Maine Land Use Regulation Commission's traditional jurisdiction for regulating such shorelands, (2) warrant disrupting the longstanding private and state landowner tradition in the stewardship and ownership of such lands on Ripogenus impoundment, (3) justify the significant socioeconomic impacts (which you did not evaluate) that imposition of such a mandate would cause, and (4) justify the very substantial direct economic costs (which you substantially undervalued) that Great Northern would have to incur in order to acquire lands it does not own and to conserve its own lands.

5. Broader Consideration to Socioeconomic Impacts in the Final EIS. You have appropriately considered in the DEIS a range of socioeconomic impacts, particularly to those of the three towns in the immediate Katahdin

5

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-14 See response T&AS-7.

T&AS-15 See response T&AS-2.

T&AS-13

Cont'd

region. Between 1980 and 1992, the population of the three area towns dropped by 7.5%, as compared to a statewide increase of 9.5% in the same period, representing a gap of 17.4%. The Great Northern mills have seen a 53% work force reduction in the last decade (between 1985 and 1994), resulting in the loss of 1,977 well paid jobs. See Exhibit C to these comments. Great Northern is currently pursuing further cost reductions. Federal and state agencies have had to respond to the new economic challenges facing the Katahdin region by investing several million dollars to stabilize that region's economy.

However, beyond the impacts to the Katahdin region, we recommend that you give careful consideration to the socioeconomic impacts of these projects throughout central and northern Maine, and the State of Maine generally.

As your DEIS notes, the pulp and paper industry is an extremely competitive and energy intensive industry. Great Northern's presence in certain markets and its competitive position in general depends upon the availability of a reliable source of inexpensive electric power. Great Northern mills are a critical component of the Maine forest products economy. Statewide, paper accounts for 35% of the State's manufactured product and, with lumber and wood products, accounts for almost half of the State's manufactured product.

Accordingly, we urge you to maintain your position as to the streamflows recommended above and, where your recommendations depart from those listed above, to reconsider your position and adopt all streamflows as recommended above.

For the reasons stated in these comments, we finally note that the proposed operations, in accord with Great Northern's Water Use Plan (as amended by the State Water Quality Certifications and the further reasonable conditions which FERC may impose), will guarantee very substantial natural resource and societal benefits, including:

a world class landlocked salmon fishery;

- whitewater rafting on the West Branch, some of the best technical rafting on the East Coast;
- significant flood control;

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

6

Cont'd

N

ũ

ŏ

- a guaranteed flow of 2,000 cfs at Millinocket to provide instream river flows for water quality; and
- 5. direct and indirect economic benefits (including highly paid jobs, production of high quality paper, purchase of forest products and other goods, both direct and indirect, in the Maine economy, and significant tax revenues to the Katahdin area towns).

These benefits exist because of the mills and the dams, and how they are managed. The resource benefits result from the storage and steady metering of water in and through the hydro system. The economic benefits result from the long term operation of the mills, which rely heavily on a reliable and inexpensive source of electric power from the hydro system.

Again, we congratulate you on the fine work that you have done thus far in the Environmental Impact Statement. We urge you to consider the comments set forth above and promptly complete the Environmental Impact Analysis. We wish you well as you complete this analysis, and urge FERC to promptly issue new licenses.

The future of this region turns on the opportunity and ability of Bowater to modernize these mills so that they can compete in today's global forest products industry.

7

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-16 No response required.

T&AS-15

T&AS-16

Cont'd

1.0 Purpose and Need for Action

1.2 Need for Power

You note that "GNP has limited options for replacing lost hydroelectric power." DEIS pg. 1-3. However, the information you cite to support that conclusion actually suggests that GNP has no means of replacing lost hydroelectric power. You cite three potential sources of replacement power: (1) cogeneration; (2) tie-in with Bangor Hydro-Electric Company (BHE); and (3) use of existing condensing turbines. You conclude, however, that based upon available data, those potential alternate sources of replacement power would be insufficient for GNP to sustain its operation and remain competitive in the industry.

Cogeneration does not represent a viable alternative because GNP's recent mill modernization efforts, undertaken to reduce energy usage by conserving steam, has reduced the amount of steam available, with a corresponding loss in cogeneration capacity. Reliance on BHE as a primary source of energy is not realistic given that, as you note, the utility "may not have sufficient capacity to meet the growing needs of all its customers." Finally, use of existing condensing turbines could only be expected to provide an additional 10MM of sustainable condensing capacity. None of these alternate sources, alone or in combination, could replace GNP's need for clean, renewable hydroelectric power.

1.3 Scope of the EIS

1.3.2 Scoping Meetings

You note that the scope of the EIS was expanded to include, inter alia, socioeconomics as a major issue. Socioeconomics are, indeed, a major issue and we applaud your recognition of this fact. We were disheartened to find, however, that socioeconomics received only minor attention in the draft EIS. We would suggest that you expand its discussion in Sections 3.13 and 3.14 of the draft EIS.

2.0 Proposed Action and Alternatives

2.3 Project Alternatives

2.3.1 Alternative 1

In Section 5.5.1 you have approved the State water quality
certificates which require minimum flows below McKay Station and
managing the North Twin impoundment for reestablishment of a self
sustaining lake trout population. In our comments at Sections
4.2 and 4.4, we demonstrate that Alternative 1 and Alternative 2
will repeatedly violate these State water quality certificate

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-17 Your comment does not contradict the statement made in DEIS.
- T&AS-18 Opinion noted.
- T&AS-19 Commented noted. Detailed responses are provided below in T&AS-52 through T&AS-56.

T&AS-17

T&AS-19conditions over the term of the new licenses. Therefore, we
question how either Alternative 1 or Alternative 2 (with Back
Channel flows of 165 cfs) can be considered reasonable
alternatives.

The fourth item of the second paragraph suggests that the creation of building setbacks and vegetative buffers within the watershed are necessary to prevent development and potential subsequent degradation of water quality. That hypothesis assumes that development cannot be done in an environmentally-friendly manner and that current LURC rules inevitably result in the degradation of water quality. Neither assumption is correct. See our discussion at 3.11, 3.12, 4.9, and 4.10.

2.3.2 Alternative 2

In Section 5.5.1 you have approved the State water quality certificates which require minimum flows below McKay Station and managing the North Twin impoundment for reestablishment of a self sustaining lake trout population. In our comments at Sections 4.2 and 4.4, we demonstrate that Alternative 1 and Alternative 2 will repeatedly violate these State water quality certificate conditions of the term of new licenses. Therefore, we question how either Alternative 1 or Alternative 2 (with Back Channel flows of 165 cfs) can be considered reasonable alternatives.

2.4.1.2 and 2.4.1.3 - Sconomic Comparison Alternative 1 & Alternative 2

The economic analysis of project costs for Alternatives 1 and 2 is seriously flawed primarily because you have grossly underestimated the cost of conservation easements as listed in Table 2-4 on page 2-27 and in Table 2-5 on page 2-28. See Exhibit A.

The January 25, 1995 public hearing, held at the Stearns High School Auditorium (which seats 450 people), was attended by over 500 people from all over Maine. Their comments clearly demonstrated that the concept of project boundary expansion must be reexamined not only from the perspective of need, but also cost. The affected landowners, including the State of Maine, made clear that no voluntary sales will occur and that the exercise of FERC's power of eminent domain will be necessary.

The need and justification for project expansion by fee or easement acquisition as well as the cost of acquisition is addressed in these comments in detail at sections 4.9, 4.10 and 4.12.

9

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-20 The staff review of LURC land use regulatory requirements resulted in revised land use assessment and recommendations (see section 4.9). Under the recommended alternative, the proposed conservation easements and/or boundary expansion would provide long term protection of valuable shoreland resources in accordance with LURC regulatory requirements (see section 4.9.3).
- T&AS-21 Comment noted. Detailed responses are provided below in T&AS-52 through T&AS-56.

T&AS-22 See response T&AS-6.

T&AS-22

T&AS-20

It is our position that project expansion is not justified by the record in this case and that you have not properly assessed the cost of any required acquisition.

With respect to economic comparison, we believe that the cost of project expansion will vary from a low of \$30.00 per waterfront foot to a high of \$100.00 per waterfront foot plus attorneys' fees, appraisal and expert fees, and severance damages as a result of condemnation proceedings. See sections 4.9, 4.10 and 4.12 and Exhibit A to these comments for detailed substantiation and documentation of this range of value.

This range of value would give a total cost of conservation easements ranging from \$13,068,000.00 (82.5 miles on Ripogenus not owned by Applicant x 5,280 feet/mile x \$30/foot = \$13,068,000) to \$43,560,000.00 (82.5 x 5,280 x \$100/foot = \$43,560,000) plus attorney's fees, appraisal and expert fees and severance damages, which could be significant. Given the unwillingness of private and public landowners on Ripogenus to sell and, therefore, the certainty of the necessity of condemnation proceedings, most costs would not be incurred until the condemnation proceedings were completed in the future. The total cost would be discounted to a present value but the discount would undoubtedly not be significant relative to the cost of acquisition of conservation easements. Also, these costs do not reflect lost opportunity costs for development which Great Northern would have to forego on its property on each impoundment.

Under Alternative 1, the total present value project costs would be:

	low	high
Ripogenus Wetlands		
Ennancemencs	\$ 1,235,000	\$ 1,235,000
Penobscot Wetlands Enhancements	\$ 46,000	\$ 46,000
Conservation Easements	\$13,068,000	\$43,560,000
Holbrook Enhancement	<u>\$ 20,000</u>	<u>\$ 20.000</u>
Total Present		
Value Cost:	\$14,369,000	\$44,861,000

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

Using a discount rate of 10% and a 30-year term results in a yearly cost of:

high

\$1,504,587 \$4,715,656

T&AS-22

Cont'd

With annual generation of 620,400 mwh the cost is:

104

low high

\$2.43 (mils/kwh) \$7.60 (mils/kwh)

The discussion at sections 2.4.1.2 and 2.4.1.3, as well as Tables 2-4 and 2-5, must be revised to reflect the vastly higher cost of expansion of project boundaries should you decide such expansion is necessary, notwithstanding comments made at the January 25th public hearing concerning the more than adequate protections provided by LURC zoning.

These costs also do not reflect the future lost opportunity benefits of landowners forced to sell their property against their will nor do these costs reflect the foregone economic benefits to the area population and regional economy of development in the future in accordance with LURC's rules.

2.4.2.1 - Cost Assessment of Potential Alternative Power Sources

T&AS-23 We agree with your assessment of potential alternative power sources and your conclusion that all alternatives are not only prohibitively expensive but are in addition to the cost of running the hydro system even if production from the hydro system is reduced through the use of alternative generation.

2.4.2.2 - Estimated Cost of Lost Power Generation

On page 2-29, you state the levelized cost of power from BHE is \$76 per mwh. On page 2-30, you state the cost of lost power is based on obtaining power from BHE. Table 2-6 lists the cost of power as \$83.22 per mwh. We assume the difference between \$76 per mwh and \$83.22 per mwh is the cost of upgrading the Applicant's existing tie to BHE.

2.4.2.3 - Estimated Cost of Lost Power Generation - Applicant's Proposal

Your discussion states that managing the elevation at North
Twin for recreation and fisheries will result in annual
generation losses of 5500 mwh and that increased flows to
Millinocket Stream will result in an additional loss of 350 mwh

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-23 No response required.

- T&AS-24 Levelized cost of power from BHE on page 2-29 is the applicant's estimate. Levelized cost of power in Table 2-6 is the staff's estimate. The difference is due to the different data sources and methodologies used to calculate alternative power values. The staff's estimate is calculated from BHE's sales of electricity to industrial customers that were reported in the 1994 FERC Form 1.
- T&AS-25 The 5,500 MWh estimate in the DEIS is a typographical error which should be 5,100 MWh. Total energy losses in Table 2-7 are mistakenly valued in dollars instead of MWh. Both errors are corrected in the FEIS. Annual costs associated with estimated energy losses in the FEIS are valued using an alternative power value of \$73.92 mills/kWh.

from North Twin and Millinocket for a total loss of 5850 mwh but Table 2-7 lists 5450 mwh.

Table 2-7 lists total energy losses as dollars and itappears that you meant megawatt hours. The annual cost for theapplicant's proposal should be:

19,450 mwh x \$.08322/kwh x 1,000 kwh/mwh = \$1,618,629.00.

Table 2-7 and Figure 2-4 should be corrected.

2.4.2.4 - Estimated Cost of Lost Power Generation - Alternative 1

Your discussion is not consistent with Table 2-7. Your discussion states that Ripogenus losses are 5355 mwh in addition to the applicant's loss of 5500 mwh. Table 2-7 shows a total of 8655 mwh for the Ripogenus project and should show 10,855 mwh.

Your discussion states that additional losses at Penobscot Mills would be 771 mwh for Millinocket Stream flows and 20,800 mwh for Back Channel flows so the total shown in Table 2-7 should be 27,071 not 26,671.

Therefore, the total for Alternative 1 should be 46,026 mwh not 43,426 mwh.

The annual cost of Alternative 1 should be:

46,026 mwh x \$.08322/kwh x 1,000 kwh/mwh = \$3,830,283.70.

Table 2-7 and Figure 2-4 need to be revised to show the correct mwh cost of Alternative 1 as well as the annual cost of energy losses caused by Alternative 1.

2.4.2.5 - Estimated Cost of Lost Power Generation - Alternative 2

Your discussion states that additional flows in the Upper Gorge will result in an annual loss of 3,787 mwh. Therefore, Table 2-7 for the Ripogenus project should list 9,287 mwh for leakage in the Back Channel and 9,287 mwh for 165 cfs in the Back Channel, not the 7,087 mwh listed.

12

Your discussion of the Applicant's proposal on page 2-31 states that managing North Twin for recreation and fisheries results in a loss of 5,500 mwh at North Twin. Your discussion on page 2-34 states that Millinocket Stream flows will result in an additional 388 mwh of loss. Therefore, Table 2-7 for the Penobscot Mills project - leakage in the Back Channel should list a loss of 5,888 mwh not 5,488 mwh. In addition, the Penobscot Mills project for 165 cfs in the Back Channel should list a loss

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-26 Additional flows in the Upper Gorge from Alternative 1 relative to the No-action alternative would result in the loss of 5,355 MWh of power generation from McKay Station. By contrast, GNP's proposal would result in the loss of 2,200 MWh. Both GNP and Alternative 1 would reduce annual generation at McKay Station by 3,300 MWh by maintaining flows from the Ripogenus impoundment to the West Branch of the Penobscot River below McKay Station to enhance recreation and fisheries. The additional energy loss to the Ripogenus Project from Alternative 1 relative to GNP's proposal is therefore 3,155 MWh for a total loss of 8,655 MWh, as stated in Table 2-7. The text in Section 2.4.2.4 has been corrected to clarify this point.

> As stated on page 2-34 of the DEIS, the proposal to maintain year-round flows of 60 cfs in Millinocket Stream would reduce power generation from the North Twin and Millinocket stations by 771 MWh versus 350 MWh in GNP's proposal. Maintaining flows of 350 cfs in the Back Channel would reduce power generation from Millinocket station by an additional 20,800 MWh. Power losses to the Penobscot Mills Project associated with Alternative 1 would therefore be 5,100 + 771 + 20,800 = 26,671 MWh, as stated in Table 2-7.

T&AS-27 Additional flows in the Upper Gorge from Alternative 2 relative to the No-action alternative would result in the loss of 3,787 MWh of power generation from McKay Station. By contrast, GNP's proposal would result in the loss of 2,200 MWh. Both GNP and Alternative 2 would reduce annual generation at McKay Station by 3,300 MWh by maintaining flows from the Ripogenus impoundment to the West Branch of the Penobscot River below McKay Station to enhance recreation and fisheries. The additional energy loss to the Ripogenus project from Alternative 2 relative to GNP's proposal is therefore 1,537 MWh for a total loss of 7,087 MWh, as stated in Table 2-7 of the DEIS. This information is now included in Table 2-4 of the FEIS.

T&AS-26

. ..

of 15,788 mwh and not 15,388 mwh.

The total energy losses on Table 2-7 for leakage in the Back Channel should be 23,275 mwh not 20,675 mwh and the total energy losses for 165 cfs in the Back Channel should be 33,175 mwh not 30,575 mwh.

Therefore, the annual energy loss cost of leakage in the Back Channel would be calculated as:

23,275 mwh x \$.08322/kwh x 1,000 kwh/mwh = \$1,936,945.00.

The annual energy loss cost for 165 cfs in the Back Channel is calculated as:

33,175 mwh x \$.08322/kwh x 1,000 kwh/mwh = \$2,760,823.00.

Corresponding changes should be made to Table 2-7 and Figure 2-4.

2.4.2.6 - No-Action Alternative

Your discussion should state that under this alternative the applicant would avoid the lost power generation cost of \$1,618,629.00 which it accepts as part of its proposal. Your discussion should also note that this alternative will avoid the following annual costs as compared to Alternative 1:

	low		high
(Table 2-4) (Table 2-7) Total:	project costs energy losses	\$1,504,587 <u>\$3.830.284</u> \$5,334,871	\$4,715,656 <u>\$3.830.284</u> \$8,545,940

Your discussion should also note that this alternative willT&AS-28avoid the following annual costs as compared to Alternative 2(leakage):

	low		high
(Table 2-5) (Table 2-7) Total:	project costs energy losses	\$1,504,587 <u>\$1,936,945</u> \$3,441,532	\$4,715,656 <u>\$1,936,945</u> \$6,652,601

and as compared to Alternative 2 (165 cfs Back Channel):

	low		high
(Table 2-5) (Table 2-7) Total:	project costs energy losses	\$1,504,507 <u>\$2,760,823</u> \$4,265,410	\$4,715,656 <u>\$2,760,823</u> \$7,476,479

13

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-27 As noted earlier, the generation loss at the North Twin Cont Station associated with the management of the elevation at the North Twin impoundment for recreation and fisheries during certain times of the year reported on page 2-31 should be 5,100 MWh not 5,500 MWh. Generation loss estimates reported in Table 2-7 are therefore correct.

T&AS-28 The No-action alternative provides a consistent baseline of comparison for other alternatives considered. Your suggestions simply introduce multiple baselines and needlessly complicate the analysis.

T&AS-27 Cont'd

3.0 Affected Environment

3.3.1 - Streamflow - Regional

We note that Dover-Foxcroft is located on the Piscataquis River and not on the West Branch or main stem of the Penobscot.

We also note that Great Northern's dams and storage capacity are primarily responsible for flood control in the region and that flooding in the main stem of the Penobscot is caused by uncontrolled run-off from the Mattawamkeag and Piscataquis Rivers as well as the East Branch of the Penobscot.

Flood control is a major public benefit of past and proposed operation of the system and should be so noted.

3.4.1 - Water Quality - Regional

Past and proposed operation provide predictable significant year round flows as compared to an unregulated river system. These flows provide irreplaceable dilution flows which allow attainment of water quality in the reach despite discharge from numerous industrial and municipal wastewater treatment flows. Dilution flows represent a major public benefit of past and proposed operation and should be so noted.

With respect to Table 3-5, we disagree that the "Back Channel", so-called, from Stone Dam to Shad Pond is classified by the State of Maine as Class C.

Maine law at Title 38 M.R.S.A. § 467 classifies river basins in the State of Maine. Subsection 7.C.1.f classifies the river in the area of the Millinocket mill. The statute reads as follows:

(f) From the outlet of Ferguson and Quakish Lakes to its confluence with the Bast Branch of the Penobscot River, including all impoundments-Class C.

You will note that the statutory definition uses the singular "outlet" to classify this section of the waterway. When Stone Dam was built in 1899 it diverted the West Branch to the east and through a canal which runs under the mill and is known as the West Branch Canal. A minimum of 2,000 cfs goes through the West Branch Canal and leakage flow goes through Stone Dam and down the Back Channel. Occasionally, spillage flow goes through the Back Channel.

Since the singular "outlet" is used, the Maine Legislature classified the West Branch Canal as Class C and did not classify the Back Channel at all.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-29 We have deleted the reference to Dover-Foxcroft to avoid implying that it is located within the Penobscot River drainage. We describe flood-control benefits of the projects in Appendix D.
- T&AS-30 We have added a statement regarding dilution flows from the projects which have lessened the impact of pollutants.
- T&AS-31 Table 3-5 does not specifically list the Back Channel and therefore does not explicitly state how it is classified. This table is listed as provided in the cited documents from the state of Maine, and we therefore are not changing the table. We note that Maine DEP classifies the Back Channel as Class C waters (letter from D. Marriott to FERC, April 22, 1993). The state further waived its authority to certify that the Back Channel will meet applicable water quality standards, as we note in section 2.2.1.2.

T&AS-30

-264

T&AS-29

Section 7.C.2.d is consistent with this construction by clarifying that <u>Millinocket Stream</u> from the terminus of the West Branch Canal to the terminus of the Back Channel is Class C (and not Class B), as it is upstream of the West Branch Canal. <u>See</u> 7.C.2.C.

T&AS-31 Cont'd

Table 3-5 should be revised to reflect that the Back Channel is not classified under Maine law.

3.5.2.1 - Back Channel

We note that Grand Falls is a barrier to movement of warmwater species. Flows in the Back Channel will govern the degree to which Grand Falls is a barrier to such movement. You should specifically state that Grand Falls and Stone Dam are important barriers to the spread of warm water species into the West Branch above Millinocket.

3.6 Wetlands

3.6,1 Basinwide

In the Draft EIS, you observe that "about 30% of the total land area of Maine is estimated to have been wetlands before the 19th century." Notably, you cite no authority for this proposition.

We are unsure of the relevance of this factual observation, particularly given the Commission's stated policy that applicants seeking to renew licenses do not need to characterize the resource that existed prior to construction, or even prior to initial licensing of the project. As the Commission has stated in its preamble to the relicensing regulations, "the valuation and consideration of the appropriateness of requiring enhancement measures is done in the context of today's environment and in relationship to today's needs and problems, not in the context of the world as it existed 50 years ago." 54 FR 23756 (1989); 55 FR 10768 (1990).

Moreover, since Great Northern is not proposing any substantial modifications to project operation, it is appropriate that Great Northern's application be judged by reference to conditions more than adequately described in its application. (See Penobscot Mills Project application, Vol. II, Section E3.2 and Ripogenus Project application, Vol. II, Section E3.2).

If you determine that it is relevant to include historic information, we agree with your observation that:

"[S]ince the beginning of the 19th century, wetlands throughout the Penobscot River basin have been altered or

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

- T&AS-32 We have revised the text of the EIS to better describe the extent to which Grand Falls acts as a barrier to fish movement.
- T&AS-33 The citation for this estimate of wetlands before the 19th century is Dahl (1990), as indicated on page 3-21 of the DEIS. We estimated the quantity of affected wetlands based solely on current conditions at both projects. We do not agree that these wetlands have, as you suggest, reached a state of "equilibrium" and have "adjusted to drawdowns;" they remain affected by the present operation of the project, and would also be affected by the proposed operation of the project.

E-265

inundated to some degree by large-scale projects such as dams for the logging industry, water storage, and hydro electric generation."

DEIS at p. 3-21. As set forth in Great Northern's application, historical records document as many as 137 dams on the West Branch. (See Penobscot Mills Project application, Vol. I, Exhibit C, pp. C-1 through C-10, and Figure C-1).

Because of the historic impoundment of waters in and along the West Branch, and the consistent drawdowns associated with releases from those impoundments, the overwhelming evidence presented during state water quality certification proceedings was that impoundment wetlands had reached a state of "equilibrium" and had adjusted to drawdowns. Again, your examination should compare proposed operations to current operations and not to pre-project conditions.

3.6.2 Ripogenus Project

Your discussion of the wetlands which border the Ripogenus impoundment accurately presents the findings of the applicant's wetland studies as reported in its application. We specifically note that the impoundment wetlands discussed at Page 3-22 of the DEIS accurately observes that "predominant plants within the shoreline zone are typical of those adapted to fluctuating water regime and disturbance." This observation supports the conclusion reached by State of Maine agencies during the water quality proceedings that the impoundment wetlands were in "equilibrium" and would not be impacted by proposed project operations any more than current operations.

3.6.3 Penobscot Nills Project

Your discussion of wetlands within the Penobscot Mills project boundaries accurately reports wetland study results from Great Northern's application. We specifically observe that the DEIS correctly notes that the substrate of the Main Channel and the Back Channel is predominantly large cobbles and boulders with some silt and sand accumulation in a few sections allowing development of small areas of wetlands, but only within narrow areas.

3.11 Land use Issues

3.11.1 Land Use Issues

3.11.1.1 Maine Land Use Regulation Commission

T&AS-34 The description of LURC's zoning classifications for the Penobacot Mills and Ripogenus Project areas, together with attendant setback and vegetative buffer standards, overlooks

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

í.

T&AS-34 See revised sections 3.11.1.1 and 4.9.

ő

T&AS-33

Cont'd

several key elements of LURC's overall protection scheme. First, while timber harvesting is permitted in all of the districts, though a permit is required, there are specific regulations which govern how timber harvesting activities must be conducted. For example, timber harvesting in the P-SL1 and P-GP Protection Subdistricts must be conducted in accordance with the following:

- (1) Within 50 feet of the normal high water mark, no clearcutting shall be allowed and harvesting operations shall be conducted in such a manner that a welldistributed stand of trees is retained so as to maintain the aesthetic and recreational value and water quality of the area and to reasonably avoid sedimentation of surface waters.
- (2) At distances greater than 50 feet from the normal high water mark, harvesting activities may not created single openings greater than 14,000 square feet in the forest canopy. In such areas single canopy openings of over 10,000 square feet shall be no closer that 100 feet apart.
- (3) Harvesting shall not remove, in any ten year period, more than 40 percent of the volume of trees 6 inches in diameter and larger measured at 4 1/2 feet above ground level. Removal of trees less than 6 inches in diameter, measured as above is permitted if otherwise in conformance with these regulations. For the purpose of these standards, volume may be determined as being equivalent to basal area.

Maine LURC Regs., Chapter 10.17.A.5.b.1-4.

Second, where the operation of machinery used in timber harvesting results in the exposure of mineral soil, LURC mandates that the minimum width of the unscarified filter strip meet the following standards:

Average Slope of Land Between Exposed Mineral Soil and Normal High Water Mark (Percent)	Width of Strip Between Exposed Mineral Soil and Normal High Water Mark (Feet Along Surface of the Ground)
0	25
10	45
20	65
30 -	85
40	105
50	125
60	145
70	165

m T&AS-34 267 Cont'd

T&AS-34 Maine LURC Reqs., Chapter 10.17.A.5.c. These standards are in Cont'd place to ensure that the quality of the impoundment waters. Finally, the discussion ignores the fact that the cottages constructed on both impoundments predate LURC. There has been very little development on either impoundment, comparatively speaking, since GNP adopted a moratorium on new leases over 20 years ago While LURC's regulations recognize the right of "grandfathered" uses to continue, LURC's regulations are somewhat unique when it comes to reconstructing nonconforming structures which are damaged or destroyed. In order to reconstruct, in its original location, a structure which is damaged or destroyed the **T&AS-35** landowner must show that there is no alternative site or location available which would cause the structure to be more conforming. Said differently, the Commission is only allowed to waive its requirements "to the minimum extent necessary" and, in no case, may the Commission waive such standards so as to increase the extent of the nonconformity. Maine LURC Regs., Chapter 10.11.C. GNP's current moratorium, together with LURC's regulations governing reconstruction of nonconforming structures, virtually ensures that the project waters will not be degraded.

3.11.1.4 Maine Forest Practices Act (MFPA)

The Forest Management Plans referenced in this section must include measures to address erosion control, windthrow, and wildlife.

Separation zones are required for any clearcut in excess of 36 acres. For clearcuts between 36 and 125 acres, those separation zones must be 1.5 times the number of acres in the clearcut. For clearcuts between 126 and 250 acres, the separation zone must be 2.0 times the number of acres in the clearcut.

3.11.3 Proposed Land Uses

It should be noted that LURC is currently undergoing a review of its Land Use Plan (its "comprehensive plan"), together with its underlying regulations. It is probably safe to assume that the resulting regulations will be more stringent than existing regulations.

It should also be noted that development of land along the impoundments may, in many instances, require a zone change from LURC. Such zone changes are often denied.

It is also important to note that neither GNP nor any other landowner or lessee within the project boundary have indicated that they have any future development plans.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-35 Opinion noted.

T&AS-36 Opinion noted.

T&AS-37 Opinion noted. The staff appropriately based the land use assessment on existing LURC regulations and not on assumptions that future regulations may be more stringent than existing regulations.

00 T&AS-36

3.12 Aesthetic Resources

You acknowledge that the Maine LURC conducted a Scenic Lakes Evaluation as part of its Wildland Lakes Assessment in 1990. Notably, LURC concluded as part of the Scenic Lakes Evaluation, that the Pemadumcook chain was identified as a chain which should be developed.

3.13.1.3 - Cultural Resources - Pulp and Paper Industry

You have failed to note that establishment of paper mills in
Millinocket and East Millinocket created these towns from the
wilderness and also created economic circumstances that supported
a significant population for generations.

Due to the geographic isolation of the area, a unique industrial and community culture developed based upon a unionized workforce, substantial immigration from northern Maine, Canada and foreign countries such as Italy. This unique culture is entirely dependent on the existence and operation of Applicant's mills and the mills are entirely dependent upon the hydro, fiber and skilled unionized workforce resources in this area. This unique symbiotic relationship between the area population, its culture and the Applicant should be noted as well as the dependence of any significant human population in the area on the continued economic viability of the Applicant.

3.14 - Socioeconomic Resources

We note that Penobscot and Piscataquis counties are not in "western Maine" but are located in north central Maine.

3.14.1.1 - Population

You have not noted continued population losses in the local area since 1990 caused by job losses at GNP and that continued job losses will result in additional area population losses because of the area's geographic isolation and lack of employment alternatives.

3.14.1.2 - Employment

The State of Maine Department of Labor has revised the employment multiplier for the paper and allied products industry from 1.61 to 3.2539. See Exhibit C to these comments. Thus, your calculation of area jobs which Great Northern supports should be revised from 3,200 jobs (2,000 x 1.61) to 6,508 jobs (2,000 x 3.2539). We note that Bowater/GNP employment in the Millinocket mill is 1,037 employees, and approximately 2,000 in the region.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-38 No response required.

- T&AS-39 Information regarding the history of paper mills in the region relevant to these proceedings is provided in FEIS (see section 3.13).
- T&AS-40 Opinion noted.
- T&AS-41 Text revised.
- T&AS-42 See section 3.14 of FEIS.
- T&AS-43 The multiplier of 3.2539 is a state-wide multiplier and can only be used to estimate state-wide employment losses. It was incorrectly applied to the regional economy and therefore over-estimates the total number of regional jobs supported by GNP. The correct multiplier to use in estimating total regional employment attributable to GNP is 1.61.

E-269

T&AS-41

T&AS-40

T&AS-42

You should also note that, with employment of 3,716 employees in 1985 and employment of 1,739 employees in 1994, Great Northern's workforce has been reduced by 1,977 employees between 1985 and 1994. See Exhibit C (Table attached to 2/16/95 Memorandum of James Haskell).

The impact of Great Northern on northern Maine and state wide employment is not noted.

3.14.1.3 - Income

We note that the 1990 labor contracts expire in 1995. We also note that declining employment opportunities, due to the geographic isolation of the area, lead directly to outmigration of population, lower regional income, and a lower ability of the Towns to provide community services. <u>See also</u> Testimony of David Cole attached as Exhibit D.

- 4.0 Environmental Impacts
 - 4.1 Geology and Soils
 - 4.1.1 Applicant's Proposal

Any construction activities associated with GNP's proposed enhancements at the Ripogenus Project, including improvement of boat ramps, construction of changing facilities and additional parking areas, and development of the Holbrook Stream nursery area must be in accordance with all local, state, and federal regulations, laws, and ordinances. Your conclusion that "a sediment control plan in accordance with local, state, and federal regulations is necessary to ensure that adequate precautions would be taken during construction of the proposed enhancements" is somewhat misleading to the extent that local, state and federal regulations may not, in all instances, require a "sediment control plan." We suggest that you consider concluding "<u>compliance</u> with all local, state, and federal laws" is necessary to ensure that adequate precautions would be taken during construction.

4.2.2 - Streamflow - Alternative 1

Your analysis of the Applicant's data concerning availability of water for Back Channel flows is inaccurate and flawed.

On page 4-5 you state only 2,350 cfs must be released at North Twin in order to pass 350 cfs down the Back Channel and 2,000 cfs through the mill. The Applicant calculated 2,610 cfs would be needed for this purpose because of physical equipment limitations for flow regulation, gate setting and control equipment sensitivities. See Applicant's April 5, 1994

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-44 No response required.

- T&AS-45 We have modified the text as you suggest. The orders for these projects will address FERC's requirements for erosion and sediment control plans for the projects.
- T&AS-46 We do not state in section 4.2.2 that only 2350 cfs would need to be released at North Twin to provide 2000 cfs at the mill and 350 cfs to the Back Channel. We agree that more flow than this amount would be required due to physical equipment limitations for flow regulation, etc. We also agree that it would not be possible to consistently provide 350 cfs to the BC in addition to the 2000 cfs at the mill along with the 10% buffer, without violating other elements of GNP's water use plan and the state's WQC.

Although GNP has not provided the information needed to determine exactly how much buffer is needed, the exact amount is irrelevant. The 2000 cfs minimum flow required at Millinocket by the WQC and the state charter could include flows provided to the BC, as we explain more fully in Appendix D. Therefore, the economic consequence relative to the environmental benefit of providing flow to the BC is the primary factor in determining its feasibility.

m T&AS-45 270

T&AS-44

additional response number 2 to your October 28, 1993 and January 14, 1994 Additional Information Request.

Your statement that only minor adjustments to 2,610 cfs must be made year round to save sufficient water for Back Channel flows assumes that the equipment is capable of such fine adjustment and it is not.

You should explain how the North Twin turbines can be adjusted to flow exactly 2,350 cfs. The Applicant's assumption merely reflects 10% error in the amount of water released because of inherent equipment and measuring device limitations. If the equipment is not capable of 100% accuracy in adjustment, it becomes the Applicant's burden to pass no less than 350 cfs in the Back Channel and 2,000 cfs through the mill. It is a real world constraint that 2,610 cfs must pass at North Twin to ensure these flows.

In response to your January 14, 1994 request for additional information, the Applicant on February 14, 1994 submitted simulations for Back Channel flows of 350 cfs and 165 cfs. The data shows sufficient water is <u>not</u> available to sustain such flows without violation of water quality certificates issued by LURC and DEP which are legal constraints on the Applicant's operation of the project as you note at Section 5.5.1 Fisheries resources are also jeopardized by such flows.

THE AVERAGE YEAR

Simulation for the average year (82SAI350) shows that 18 of 52 weeks or 35% of the time have flows of 2,610 cfs. Sufficient water may not be available without depletion of overwinter storage if more than a third of the time only the minimum which is required to meet the flows can be predicted.

The State water quality certificate for Penobscot Mills (L-17166-33-A-N) requires the water level at North Twin be maintained at or above the lake trout spawning/incubation level from October 15 through May 1. You note in Table 4-4 on page 4-23 that lake trout spawning occurs in October and November. The level on October 3 is 482.05 and drops thereafter for four weeks to levels as low as 478.61, violating the water quality certificate and dewatering lake trout eqgs.

THE AVERAGE DRY YEAR

Similar problems are apparent from the flow simulation for the average dry year (84SAI350).

until June 7 of the following year. See Water Quality

Salmon in the West Branch spawn between October 15 and November 15 of each year and incubation flows must be maintained

T&AS-48

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-47 We disagree that model runs of 350 cfs to the BC for the average year indicate possible depletion of overwinter storage. We also disagree that these model runs indicate a possible violation of lake trout spawning and incubation levels. As shown in Figure D-11, panel C, North Twin lake levels reach a minimum at the end of October rather than in the middle of the month, as required. However, panel D of this figure shows that North Twin outflows were simulated at close to 4000 cfs from the end of August through mid-October. A reduction of this outflow during the first half of October would have resulted in the correct lake elevation for lake trout. However, these are moot points, since as we point out above and in Appendix D, the flow through the mill could be reduced to make up any short fall if flows were required to the BC.
- T&AS-48 We did not evaluate the average dry year in the DEIS; however, our response to your comments on the dry year are applicable to the average dry year. As stated above, we agree that there is insufficient water in a dry year or a worstcase year (which you refer to as the typical year) to provide for all enhancements proposed by GNP in addition to 350 cfs to the BC and 2000 cfs through the mill plus an additional 10% buffer for equipment limitations, etc. However, the 2000 cfs minimum flow required at Millinocket by the WQC and the state charter could include flows provided to the BC, as we explain more fully in Appendix D. Therefore, the economic consequence relative to the environmental benefit of providing flow to the BC is the primary factor in determining its feasibility.

We disagree with your statements regarding simulations for the wet year. Figure D-10, panel B, clearly shows no difference between GNP's water use plan and the simulation of 350 cfs to the BC.

T&AS-46 Cont'd

Certification WQC-001 issued by Maine Land Use Regulation Commission Page 7, Section 16.A.2.(a), 3rd and 4th paragraphs.

Salmon spawning flows range from a high of 2,300 cfs to a low of 2,000 cfs. However, flows are 1,280 cfs for four weeks in March violating the water quality certificate, dewatering virtually the entire year's egg production and restricting habitat for all life stages.

Togue spawning elevations in North Twin range from 480.31 to a low of 478.73 during mid-November, but elevations drop below those levels in March violating the water quality certificate and dewatering the eggs.

During 15 weeks or 29% of the time, flows from North Twin are 2,610 cfs, the minimum necessary to pass 350 cfs down the Back Channel. Lack of available water is a distinct possibility.

Note that available storage in Ripogenus at the end of March is less than 4 bcf precluding the possibility of supplementing flows from storage.

THE DRY YEAR

The problems are magnified and readily apparent for the dry year (85SAI350) which one can assume will occur over the new license terms since it is an actual year from the period examined (3-31-85 to 3-30-86).

Salmon spawning flows range from 1,800 cfs to 1,400 cfs but flows fall to 690 cfs during March. Such flows grossly violate the water quality certificate, will dewater the entire egg production for the year and will severely compromise the habitat available for all life stages of salmon and all other fish species.

Togue elevations during October range from 483.56 to 480.34 but fall below the maximum for weeks which will violate the water quality certificate and dewater the eggs.

Flows from North Twin are 2,272 cfs for four weeks in March resulting in either partial or total dewatering of the back channel or passing less than 2,000 cfs through the mill which would violate state law and the water quality certificate.

Only during one week (11/10) will more than 2,610 cfs pass at North Twin; 51 of 52 weeks (or 98% of the time), only the minimum water required to pass 350 cfs and 2,000 cfs will flow through North Twin. Obviously, should an actual year drier than 1985 occur over the new license terms, sufficient water would not be available and an environmental and economic disaster would occur.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

∿ **T&AS-48** 77 **Cont'd**

THE WET YEAR

Even the wet year will violate the water quality certificates and dewater fish eggs.

The simulation (76RAI350) shows that salmon spawning flows are as high as 3,900 cfs but drop to 2,000 cfs violating the water quality certificate and dewatering the eggs laid earlier.

Togue spawning levels are as high as 481.55 but levels later drop as low as 478.95 violating the state water quality certificate and dewatering togue eggs.

THE TYPICAL YEAR

The typical year (MN3AI350) is a disaster. Salmon spawning flows range from 1,400 cfs to 1,800 cfs but drop to 400 cfs for five weeks in March. 400 cfs will violate the water quality certificate, dewater all eggs, and restrict habitat to such an extent that the continued existence of all fish stocks will be in question.

Togue spawning levels are more than 476 but the elevation drops for four weeks in March violating the water quality certificate and dewatering togue eggs.

Flows from North Twin are 2,610 cfs for 48 of 52 weeks and only half that level, 1,300 cfs, for the month of March; 1,300 cfs will result in dewatering the Back Channel, and violating both the water quality certificate and state law because 2,000 cfs cannot be passed through the mill. Once again, the data for this year simply cannot be used to justify the availability of water for 350 cfs down the Back Channel and, in fact, the data demonstrates that sufficient water for such flows is not available.

Note that available storage in Ripogenus at the end of March rises to only 3.5 bcf from 2 bcf and there is no possibility of supplementing flows from storage during this year.

SUMMARY

The data clearly shows violations of state water quality certificates and dewatering of salmon and/or lake trout eggs for all simulations with the possible exception of wet years. In addition, the simulations document that sufficient water is not available due to the high percentage of flows at or below the minimum of 2,610 cfs at North Twin which in dry years seriously deplete storage at Ripogenus.

T&AS-48 T-2 Cont'd Cont'd

You should note that supplementing flows from storage to boost flow in a particular year to avoid violations of water quality certificates or fisheries impacts is simply not possible during the average dry and dry year and during other years would be an unproven method of project operation with unknown impacts on future availability of water.

We understand that this alternative must be examined because the self proclaimed guardians of the environment have chosen it as their mantra. However, your discussion must point out these obvious failings if for no other reason than to educate project opponents to the dangers of their proposal.

4.2.3 - Streamflow - Alternative 2

Your analysis of 165 cfs flows down the Back Channel also overlooks significant water quality certificate violations and dewatering of salmon and togue eggs.

THE AVERAGE YEAR

The average year (82SAI165) simulation shows that North Twin flows will be at the minimum, 2,420, cfs for 11 weeks or 21% of the time. Given the equipment and measuring limitations, it is very likely that either 165 cfs will not be available or 2,000 cfs will not pass through the mill during these weeks. Minimum flows 20% of the time leads us to conclude that availability of water is a limiting factor.

Salmon spawning flows below McKay are as high as 2,400 cfs but drop below that level violating the water quality certificate and dewatering salmon eggs.

Togue spawning levels are as high as 481.94 but fall from that level resulting in dewatering togue eggs and violating the state water quality certificate.

THE AVERAGE DRY YEAR

As one would expect, the simulation for the average dry year (84SAI165) shows even greater impacts. Salmon spawning flows below McKay range from a high of 2,300 cfs to a low of 2,000 cfs, but flows are 1,680 cfs for the month of March violating the state water ouality certificate, dewatering salmon eggs, and reducing hatitat for all life stages.

Togue spawning levels are as high as 483.56 in North Twin but are lower than that level for the remaining weeks violating the state water quality certificate and dewatering togue eggs.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-49 We disagree with your statements regarding simulations of 165 cfs to the BC for the average year and the wet year. As figure D-14 illustrates, there is no difference in wet year simulations with and without 165 cfs provided to the BC. In the average year, simulation flows from North Twin for the 165 cfs to BC scenario are the same as or higher than GNP's proposal, except for the last three weeks of the simulation when they are slightly lower (panel D). North Twin lake elevations under the 165 cfs scenario are lower after mid-October when compared with GNP's proposal (panel C), but except for two weeks in late October, are not in violation of the lake trout spawning requirements. This exception could have been prevented by slightly lowering the outflow from North Twin in early October.

See our response to comment T&AS-48 (above) with respect to 165 cfs BC flows for the average dry year and worst-case dry year.

T&AS-49

T&AS-48

Cont'd

Flows from North Twin are at or below the minimum required to pass 165 cfs down the Back Channel and 2.000 cfs through the mill for 14 of 52 weeks or 27% of the time. Insufficient water is more than probable under these circumstances.

Note that available storage in Ripogenus at the end of March is less than 3.8 bcf precluding the possibility of supplementing flows from storage during this year. An additional 100 cfs for 100 days (27% of the time) would consume almost 1 bcf of water.

THE DRY YEAR

The dry year (85SAI165), which is an actual year during the period of study and will undoubtedly reoccur during the new license terms, shows severe state water quality certificate violations and impacts on fisheries.

Salmon spawning flows are 1,800 cfs or 1,900 cfs for all but one week but fall to 1,420 cfs during the month of March violating the state water quality certificate and dewatering salmon eggs.

Togue spawning levels in North Twin are as high as 481.51 and later as low as 479.51, violating the state water quality certificate and dewatering togue eggs.

Flows at North Twin are at the minimum required for 37 of 52 weeks or 71% of the time. It is a certainty that if a year drier than 1985 occurs, sufficient water will not be available and it is highly likely that at some point during those 37 weeks insufficient water will be available and either (1) flows in the Back Channel will be reduced from 165 cfs or (2) 2,000 cfs will not be passed through the mill. In either event, the flows will violate state law and/or the state water quality certificate.

Ripogenus storage at the end of March is 6.7 bcf and is insufficient to supplement flows without dangerously depleting available water since during this year minimum flows at North Twin occur 71% of the time. Depleting storage to supplement flows would be especially disastrous during back to back dry years.

THE WET YEAR

The wet year (76RAI165) also violates the state water quality certificate and impacts fisheries. Salmon spawning flows below McKay are 3,900 cfs and then drop to 2,000 cfs. However, flows are no higher than 3,300 cfs thereafter violating the state water quality certificate and dewatering salmon eqgs.

-275

m

Togue spawning levels in North Twin are as high as 481.55 but drop as low as 478.92 thereafter violating the state water quality certificate and dewatering togue eggs.

THE TYPICAL YEAR

The typical year (MN3AI165) also violates state water quality certificates and impacts fisheries. Salmon spawning levels below McKay range from a low of 1,800 cfs to a high of 2,000 cfs but flows during March drop to 400 cfs for 4 weeks violating the state water quality certificate, dewatering virtually all salmon eggs and reducing habitat for all life stages of salmon and all other fishery resources in the West Branch to such an extent that elimination of species must be considered a natural consequence of flows of 400 cfs for a month.

Togue spawning levels are as high as 477.865 in North Twin but fall as low as 475.37 thereafter violating the state water quality certificate and dewatering togue eggs.

The minimum amount of water passes at North Twin in 43 of 52 weeks or 83% of the time. It is highly likely that at some point during those 43 weeks insufficient water will be available and either the Back Channel will be wholly or partially dewatered or 2,000 cfs will not pass through the mill violating the state water quality certificate and state law.

Ripogenus storage at the end of March is 3.5 bcf and is insufficient to supplement flows during the 43% of the time that minimum flows pass at North Twin.

SUMMARY

Every simulation for Alternative 2 shows violations of the state water quality certificate, dewatering of salmon eggs, dewatering of togue eggs, and under some scenarios, conditions which place the entire West Branch salmon fishery in jeopardy, dewater the Back Channel, jeopardize passing 2,000 cfs at Millinocket (violating state law and the water quality certificate), and seriously depleting Ripogenus storage.

We question how Alternative 2 can be considered a reasonable alternative in light of flows and elevation levels which violate the state water quality certificates and the impact on fishery resources, flows at Millinocket, and Ripogenus storage. However, if you insist on studying the alternative, these adverse environmental impacts should be noted throughout the EIS in relevant sections.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

4.2.5 - Fisheries - Summary

Your discussion omits significant environmental impacts on fisheries by limiting the discussion to "average" years.

T&AS-50 The data demonstrates that you should:

- (a) reconsider your conclusion that sufficient water is available for substantial back channel flows;
- (b) discuss the fisheries impacts on West Branch salmon stocks below McKay Station, and establishment of a self sustaining togue population.

4.3.1.3 - Back Channel

We disagree with your premise that the Back Channel is Class C under Maine law. Our comments on this issue are detailed at section 3.4.1.

However, we agree that any comparison should be to existing
conditions since, under Maine law, designated uses to be
considered are only those that are actually present. Bangor
Hydro Electric Co. v. Bd. of Environmental Protection, 595 A.2d
438, 442 (Me. 1991).

Table 4-5, Page 4-24

Table 4-5 summarizes the impact on the impoundments for the Applicant's proposal and the alternatives. Because it is based on "average" flows, it masks the impact on the impoundments of Back Channel flows.

Simulation for the average year (82SAI350) shows a maximum difference of at least seven feet in North Twin elevations between historical (no action) operations and this alternative, and significant differences for many weeks. Simulation for the average year (84SAI350) shows even larger differences in North Twin elevation between historic (no action) and this alternative than are shown for the average year. The dry year (85SAI350) shows significant elevation differences at Ripogenus and North Twin for this alternative and historic (no action) operation. The wet year (76RAI350) shows significant elevation differences at Ripogenus and North Twin for this alternative and historic (no action) operation. The typical year (MN3AI350) shows significant elevation differences at North Twin for this alternative and historic (no action) operation.

Despite the simulation results, you have used "average" conditions in Table 4-5 and therefore erroneously shown Alternative 1 as having a smaller maximum change in elevation at North Twin than the "no action" case and a difference between the

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-50 We provided detailed responses above (see responses T&AS-46 to T&AS-49).

T&AS-51 See response T&AS-31.

T&AS-52 This table is not intended to present the full range of elevation changes for various alternatives. We discuss this in section 4.2 and Appendix D. Our discussion in section 4.4.3.5 covers alternatives 1 and 2. Changes in Ripogenus and North Twin elevations for average, dry, wet, and worst case typical year for GNP's WUP, versus 350 cfs in Back Channel, is also shown in panels A and C of Figures D-11, D-12, D-10, and D-13 in Appendix D. We have corrected the typographical error in the table.

T&AS-52

27

two at Ripogenus of .7 feet.

Your use of "averages" mischaracterizes the impact on elevation and surface area in the impoundment from the various proposals. In order to compare alternatives, your discussion needs to determine and discuss what will actually happen, not what "averages" show; the averages do not represent real world conditions. As presented, the reader would not suspect dewatering of eggs or overwinter storage depletion will occur repeatedly for the alternatives.

The last line of Table 4-5 refers to "minimum elevation" but apparently lists "maximum elevation".

4.4.2.1 - Fisheries - Alternative 1 - Upper Gorge Flows

The discussion does not mention that, under certain circumstances (i.e. dry years and worst case years), flows from Ripogenus would be compromised at various times. Such a discussion should be added so that the reader is informed of additional environmental impacts over the license term. The impact would seem to affect overwintering habitat.

4.4.2.2 - Fisheries - Alternative 1 - West Branch Flows

As noted in our comments to Section 4.2.2, this alternative has substantial impacts on fisheries below McKay Station that are not given even the briefest mention in your discussion.

Simulations for five scenarios for this alternative were submitted at your request. Three of those five simulations show substantial adverse impacts on spawning and incubation levels.

The average dry year (84SAI350) shows flows in March of slightly more than one half of spawning flows. These reduced flows will dewater a substantial portion of the year's eggs, substantially reduce available habitat for all life stages and violate the state water quality certificate. These impacts must be noted in your discussion.

The dry year (85SAI350) shows March flows of 690 cfs once again leading to a greater loss of salmon eggs, habitat for all life stages and violation of the state water quality certificate. Maintenance of the world class salmon fishery under these conditions will be difficult since this is an actual year from the period of study, the impacts must be noted because it is highly probable that a year as dry will be experienced over the new license terms.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-53 Leakage flows occur during the time of the year when Ripogenus flows would be impacted, so this concern is not relevant.
- T&AS-54 The model runs you describe were discussed fully in section 4.2 and in responses T&AS-46-49 above. Section 4.4.2.2 is only intended to indicate that no additional measures are proposed for that location.

T&AS-53

T&AS-52

Cont'd

The typical year (MN3AI350) is the worst of all with March flows of 400 cfs or 20% of spawning flows which will eliminate salmon eggs, eliminate almost all habitat and violate the state water quality certificate. This simulation for a "typical" year predicts elimination of the world class salmon fishery and this impact should be discussed.

Limiting the discussion to "average" years serves only to hide from the public the true environmental impacts which would result from implementation of this alternative. The public is entitled to a discussion of all reasonably expected impacts.

4.4.2.4 - Fisheries - Alternative 1 - Back Channel Flows

Of the five simulations studied for this alternative, all but the "wet year" revealed long time periods during which the flow at North Twin would be 2,610 cfs or less from which we conclude that dewatering of the back channel is likely a recurrent event under this alternative. The impact of this event needs to be discussed.

On page 4-5 (Section 4.2.2) you correctly state that during dry and worst case years "Back Channel flows would have to be curtailed", but in this section you fail to discuss the impact on environmental resources in the Back Channel which would be impacted by curtailed flows over the new license terms.

You note on page 4-29 that "portions of a salmon population ...probably would be displaced by ... high volume spillage ..." and note difficulties in recolonization. Impacts of dewatering the Back Channel during dry and worst case years would appear to be greater since those fish who do not detect diminishing flows and move downstream would be stranded and undoubtedly perish.

We also note that the extent that Grand Falls poses a barrier to fish migration would seem to be a function of the species involved as well as the flow. Flows that would allow warm water species to migrate above Grand Falls should be identified as contrary to DIFW's goal of preventing those species from moving into the West Branch above Millinocket because such follows enhance the risk of such movement.

We agree with your conclusion that 350 cfs flows would not significantly increase regional numbers of adult landlocked salmon, especially in light of recurrent dewatering of the Back Channel under this alternative.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-55 Back Channel flows would have to be curtailed in dry years only if flow through the mill were not reduced. Since no decision was ever made on how the Back Channel flows might be implemented, we have not addressed the issue of reduced Back Channel flows in dry years. Your other comments are noted.

T&AS-55

TRAS-54

Cont'd

29

4.4.2.5 - Pisheries - Alternative 1 - Impoundment Drawdowns

Your conclusion that flow enhancements would cause only minor changes in impoundment draw-downs may be accurate for average years but is certainly not accurate for dry and worst case years as noted on page 4-5 (Section 4.2.2).

The impact of impoundment levels several feet lower should be discussed, especially with respect to smelt and burbot spawning, Table 4-4, page 4-23. The simulation for the average dry year (84SAI350) shows elevation as low as 917.03 and storage as low as 3.3 bcf. Dry years will occur over the license term and this impact needs to be discussed in order to fully appreciate the alternative.

The impact on the North Twin impoundment also should be discussed, especially the effect of greater draw-downs on togue incubation levels and the impact on the togue restoration program.

On page 4-5 you note, with respect to Millinocket Lake, that increased flows in Millinocket Stream "would adversely affect fisheries..." but you omit any discussion of the adverse affect in this section as well as section 4.4.2.3. on page 4-26. This omission should be rectified and the "significant" effects on resources noted on page 4-5 detailed for average years as well as dry and worst case years. It seems apparent that spawning and incubation for all affected species in the lake would be adversely impacted as well as fisheries resources in the stream should flows be curtailed due to lack of water.

4.4.3.1 - Fisheries - Alternative 2 - Upper Gorge Flows

Page 4-6 notes that 165 cfs down the Back Channel during dry and worst case years would result in outflows from Ripogenus being reduced "below desired levels for 4 to 5 weeks ...".

This impact which will occur over the new license term is not discussed. At a minimum, adult habitat will be impacted when flows are reduced.

4.4.3.2 - Fisheries - Alternative 2 - West Branch Flows

Five simulations were examined for this alternative, three of the five show dewatering of salmon eggs, reduced habitat and perhaps loss of the entire West Branch salmon fishery. These impacts are omitted from your discussion.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-56 Your comments assume that flows would be curtailed or that impoundment levels would be reduced in dry years, rather than reducing the flow through the mill. Since no decision has been made on this point, we chose not to discuss potential inpacts to fisheries or other resources. We fully discuss flow impacts in section 4.2 and Appendix D, and in response T&AS-46 to T&AS-49 above.

The average dry year (84SAI165) shows March flows of 1,680 cfs as compared to spawning flows as high as 2,300 cfs. Salmon eggs would be dewatered and habitat greatly reduced stressing all life stages. Obviously, the state water quality certificate would be violated.

The dry year (85SAI165), which is an actual year that can be expected to occur over the term of new licenses, shows spawning flows as high as 1,900 cfs and March flows as low as 1,420 cfs. Eggs would be dewatered, habitat reduced and the state water quality certificate violated.

The typical year (MN3AI165) shows March flows of 400 cfs or less than 25% of spawning flows which will eliminate all incubating eggs and put the entire fishery resource at risk.

All simulations except the wet year show substantial time periods with minimum flows at North Twin. This alternative is subject to great risk of insufficient water especially during the average dry year, dry year and typical year when Ripogenus storage is seriously depleted and even lower flows from Ripogenus and greater stress on fishery resources are likely.

The foregoing impacts need to be discussed in order to fairly evaluate the alternative. Should you assume low flows can be supplemented from using storage from Ripogenus, you should discuss the impact of lower elevations on burbot spawning, smelt spawning, and the problems associated with lowering overwinter storage, especially in the context of several succeeding dry years.

4.4.3.3 - Fisheries - Alternative 2 - Millinocket Stream Flows

Your discussion concludes that enhanced flows "would provide some enhancement for salmon ..." The enhancement should be compared to the slight effect on fisheries and recreation noted on page 4-6, especially during dry and worst case scenarios.

4.4.3.4 - Fisheries - Alternative 2 - Back Channel Flows

On page 4-6 you note that during dry and worst case years flows from Ripogenus would be reduced and presumably flows in the Back Channel reduced or eliminated. The impact of reduced flows on fisheries resources in the Back Channel should be quantified since dry years will occur over the term of new licenses.

The combination of low Ripogenus storage and numerous weeks passing the minimum flow at North Twin for the average dry year, dry year, and typical year ensure that dewatering of the Back Channel will be a recurrent problem over the term of new licenses.

We agree that Back Channel flows would not significantly increase regional landlocked salmon stocks, especially in light of recurrent dewatering of the Back Channel.

4.4.3.5 - Fisheries - Alternative 2 - Impoundment Draw-downs

On page 4-6 you note that Ripogenus impoundment levels would be reduced under dry and worst case years. The impact on impoundment fisheries, especially smelt and burbot spawning (Table 4-4, page 4-23), should be discussed since dry years will occur over the new license term.

Simulations for the average dry year (84SAI165), dry year (85SAI165) and typical year (MN3AI165) will all have steadily lowering elevations due to storage depletion.

In addition, the impact on North Twin togue incubation should be discussed since page 4-6 notes dewatering of this resource during dry years. Also, the impact on establishing a self-perpetuating stock of togue in North Twin should be discussed. Simulations for the average year (85SAI165), the average dry year (84SAI165) and the typical year (MN3AI165) all show elevations in North Twin falling, to varying degrees, after the maximum elevation during togue spawning season.

4.4.4 - Fisheries - No-Action Alternative

You have omitted the fact that this alternative avoids the substantial adverse fisheries impacts and state water quality certificate violations that will occur on a regular basis under Alternative 1 and Alternative 2. In order to fairly assess the alternatives, this benefit of the no action alterative should be noted.

4.5.1 - Wetlands - Applicant's Proposal

In discussing Great Northern's proposed operation of the Ripogenus Project, and its asserted impact to wetlands within the project boundaries, you assert that the proposed operation "would continue to <u>adverselv</u> affect" those wetlands that are hydrologically dependent on impoundment levels. DEIS at Section 4.5.1.1 at p. 4-33. This observation appears wholly inconsistent with the Commission's statement in the preamble to its Relicensing Regulations that "evaluation and consideration of the appropriateness of requiring enhancement measures is done in the

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-57 The crux of your comments on wetlands relate to the notion that we have considered historic rather than existing conditions with regard to the project shoreline wetlands. You further suggest that there are no effects on shoreline wetlands owing to current or proposed project operation, and that it is therefore not appropriate to require enhancements for such effects. We disagree. It is clear that these shoreline wetlands are currently affected by the present operation of the projects, and would continue to be affected by the proposed operation of the projects. We have changed the language throughout the FEIS, however, to indicate that what we are recommending are enhancements, not mitigation. The requirements for the wetlands enhancements, however, will remain the same as in the DEIS for all of the project alternatives.

T&AS-56

Cont'd

context of today's environment and in relation to today's needs and problems, not in the context of the world as it existed 50 years ago." 54 Fed.Register 23756, 23758, 23775-76 (1989); 55 Fed.Register 4, 8-9 (1990).

This statement in the preamble to the Relicensing Regulations should be contrasted with your observation in the Executive Summary relating to your recommended action that:

"[T]he wetlands enhancements (affecting about 280 acres) included in Alternative 2 are sufficient to mitigate for the estimated acreage of wetlands that are adversely affected by impoundment draw-downs and would eliminate a <u>long-standing</u> <u>adverse</u> effect of the projects on the regional ecosystem."

DEIS at Executive Summary p. xvii.

Finally, this statement should be further contrasted with your observation on page 4-13 of the draft EIS:

"Using <u>existing</u> conditions <u>as the baseline</u>, we conclude that GNP's proposed projects would not affect water quality in the Back Channel because GNP proposes no changes in the existing flow regime."

Section 4.3.1.3, at p. 4.13 (emphasis supplied).

How do you justify using existing conditions as the baseline in one section of the DEIS consistent with FERC's Relicensing Regulations and, in the sections on wetlands, using pre-project conditions in a manner that is inconsistent with FERC's regulations?

As Great Northern noted in its application, in response to comments from the U.S. Fish & Wildlife Service, the assessment of project impacts is unique for relicensing of a project as it is directed at describing the existing environmental resources that have resulted from past years of project operation and assessing whether the effects of proposed project operation will be beneficial or adverse. Ripogenus Project Application, Vol. II at p. E3.2-30. Because Great Northern has proposed no significant changes to project operation, no additional impact to existing wetlands is likely. To the extent existing operations are being altered, in order to provide enhancements discussed in the Water Use Plan, those enhancements far outweigh the somewhat speculative impacts which longstanding impoundment drawdowns will have on impoundment wetlands. In this context, how can you justify wetlands mitigation estimated conservatively at \$750,000.007

33

As you note, wetlands that are hydrologically associated with the Ripogenus impoundment contain plant species that are somewhat tolerant of fluctuating water regimes and are not of high value to wildlife. This narrow band of affected wetlands occurs around the periphery of the impoundment and does not possess deep, organic substrates or sources of inflow independent of the impoundment. Thus, the vegetation, and the wetland systems which support it, are properly viewed as being "in equilibrium."

We agree with your observation that, because the Upper Gorge is extremely rocky and steep-sided, these conditions have precluded substantial wetlands development, and that the wetlands in the Upper Gorge would be adequately protected with leakage flows of about 12 CFS throughout most of the year.

Finally, we note a discrepancy between your recommended Alternative 2 and the acreage of impoundment wetlands within the Ripogenus project boundaries. Even if we assume that the Ripogenus project impoundment wetlands are "adversely impacted," the acreage of impacted wetlands totals only 250 acres. Without explaining the discrepancy in acreage, you propose under Alternative 2 that 280 acres of wetlands be created by Great Northern as mitigation. Again, even assuming (1) an adverse impact, and (2) an acre-for-acre compensation ratio for wetlands mitigation, FERC's Alternative 2 (as summarized at p. xvii of the DEIS) overcompensates. Given your subsequent observation that 350 acres of wetlands mitigation would provide "greater mitigation than is reasonable to compensate for the roughly 250 acres of presently affected wetlands, " why would 280 acres of wetlands mitigation not also provide greater mitigation than is reasonable? See DEIS at 4.5.2, p. 4-37.

PENOBSCOT MILLS PROJECT

Again, in light of FERC's stated policy in the preamble to its relicensing regulations, we do not understand the statement in the DEIS that "current operation of the Penobscot Mills project negatively affects wetland areas that are hydrologically dependent on the impoundments." Again, we understand FERC's policy to assess "adverse impact" by evaluating whether proposed operation of the project will have an additional negative impact beyond current operation. It appears that, in your assessment of impacts to wetlands, you are rewriting FERC's Relicensing Regulations to require a determination as to whether <u>current</u> operations result in an "adverse impact" and then requiring mitigation in conjunction with proposed operations. This is an inappropriate application of FERC's regulatory approach.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

M N T&AS-57 4 Cont'd

Your strained approach to assessing impacts to wetlands is further emphasized by your discussion of "status quo" wetlands conditions in the sections of the West Branch from the North Twin impoundment to Quakish Lake and from Millinocket Tailrace to Dolby Pond. Your statement suggests that relatively stable water levels can and should be viewed as a mode of project operation that will have no impact on wetlands within the project boundaries. From this approach, can we infer that (1) where project operations result in stable water levels, there is no or neutral impact on wetlands, but (2) where project operations result in fluctuating water levels, there will always be an "adverse impact" on wetlands within the project boundaries?

The problems with the methodology or approach you use are:

- It is contrary to FERC's own Relicensing Regulations, which require a comparison of proposed project operations against current project operations to determine whether or not there is an "adverse impact". If existing project operations cause fluctuating water levels, how can proposed project operations which will result in relatively similar water level fluctuations cause an additional "adverse impact" on wetlands?
- 2. This methodology or approach would appear to assume pre-project historical conditions, if not hypothetical conditions, as the baseline for determining whether proposed project operations as part of a relicensing will result in "adverse impact." As Great Northern noted in its application, in response to an assertion by U. S. Fish & Wildlife that project operations will result in a "net loss of wetlands":

"[T]he wetlands that now exist are the product of the construction of Ripogenus Dam. The question of what wetlands might exist if the Ripogenus impoundment were managed differently is entirely hypothetical. If a change in operation did create more wetlands, that would more properly be viewed as an enhancement of existing conditions."

Ripogenus Project Application, Vol. II at p. E3.2-20.

BACK CHANNEL

We agree with your observation that "existing wetlands along the Back Channel probably would remain stable under the proposed leakage flows" and that "Great Northern's proposed leakage flows would adequately maintain the existing small quantity of wetlands

다 **T&AS-57** N **Cont'd**

in the Back Channel." Thus, from the perspective of protecting wetlands resources in the Back Channel, there is no real justification for expanding or increasing flows in the Back Channel.

Further, as you note, increased flows could disrupt wildlife in the Back Channel, including the existing beaver population. To the extent that beavers have created dams in certain areas of the Back Channel that have created wetlands, disruption of the beaver population could adversely affect wetlands directly associated with the beaver dams. Thus, comparing to current conditions, an alternative to increase flows (that would impact the beaver population in the Back Channel) would result in an "adverse impact" to wetlands directly associated with those beaver dams.

4.5.1.2 Wetlands Enhancements

<u>Ripogenus Project</u>. Given the "equilibrium" condition of wetlands along the project impoundments, and the fact that proposed project operations will be relatively unchanged from current conditions, we again fail to see the "adverse impact" to wetlands (comparing to the baseline of existing conditions). If there is no "adverse impact" to wetlands, then proposals to enhance or mitigate in response to such impacts are irrelevant and inappropriate.

If you pursue a different analysis to wetland impacts, in order to appease the environmental Coalition intervenors, we would ask you to consider the socioeconomic impacts and determine whether the additional costs associated with requiring the applicant to undertake enhancements, which it has not offered as to the Ripogenus Project, justify such enhancements.

If, as stated above, you determine that, on an acre-to-acre basis, the Applicant should compensate for alleged adverse impacts to the 250 acres of impoundment wetlands in the Ripogenus project, a total of only 250 acres of wetlands for both projects should be enhanced, not 280 acres.

<u>Penobscot Mills Project</u>. Again, the justification for requiring wetlands enhancements is unclear under a proper analysis, particularly given FERC's Relicensing Regulations. However, if acre-to-acre compensation is to be required of Great Northern, Great Northern's proposal to enhance 45 acres of wetlands within the Penobscot Mills project boundaries (30 acres at Deep Cove and 15 acres at Wadleigh Brook) should be credited against the 250 acres of wetlands which you assert will be adversely impacted by proposed project operation.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-58 Current operations at the Ripogenus projects negatively affect approximately 250 acres of shoreline wetlands. This estimate is based on existing, not past, conditions. We have therefore determined that enhancements are appropriate.

T&AS-59 Comment noted.

M N 00 T&AS-58

T&AS-57

Cont'd
4.5.2 Alternative 1

T&AS-60 We agree, for all of the reasons stated above, that you could not reasonably require Great Northern to enhance 350 acres (100 acres at Quaker Brook and 250 acres at Black Pond) of existing, lower-quality, shoreline wetlands.

4.5.3 Alternative 2

Again, for all the reasons stated above, we do not agree that Great Northern should be required to enhance 250 acres of shoreline wetlands. In conjunction with your Relicensing Regulations, in comparing proposed project operations with existing project operations (as opposed to pre-project historical conditions), you should conclude that proposed operations will not result in "adverse impacts" to wetlands.

We agree with your assessment that wetlands in the Back Channel provide relatively high wildlife values at year-round leakage flows. Increasing flows above leakage flows in the Back Channel, however, would not increase the total area or quality of wetlands for wildlife and could decrease the area or quality of such wetlands to the extent the beaver population in the Back Channel is adversely impacted by higher flows. For these reasons, we agree that wetland resources would be best served by maintaining current leakage flows (the appropriate baseline under FERC's relicensing regulations) in conjunction with Great Northern's Back Channel Wildlife Habitat Management Plan.

4.5.4 No Action Alternative

We agree with FERC's assessment that the no action alternative would have no adverse effects on wetlands. If that is so, why do proposed operations result in "continued" adverse impacts to wetlands? <u>Compare second sentence of Section 4.5.1.1</u> at p. 4-33.

4.6 Terrestrial Resources

4.6.1 Applicant's Proposal

4.6.1.1 Site Specific Impacts

RIPOGENUS PROJECT

We agree with your observation that the use of herbicides to control vegetation along the transmission line corridor between McKay Station and Millinocket will not adversely affect water quality or wildlife habitat. As Great Northern observed in its application, Great Northern obtains the services of statelicensed contractors, who apply herbicides in accordance with

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-60 No response required.

T&AS-61 Opinion noted.

T&AS-62 We have changed the FEIS throughout to indicate that the no action alternative would have no <u>additional</u> adverse affects on wetlands (emphasis added).

T&AS-63 No response required.

E-287

T&AS-63

T&AS-62

T&AS-63state and federal licensing requirements, and with the
manufacturers' instructions. Ripogenus Application at Vol. II at
p.E3.2-31.

PENOBSCOT MILLS PROJECT

We agree with your observation that, under the requirements of Great Northern's Back Channel Wildlife Habitat Management Plan, continued operation of the Penobscot Mills project as proposed by Great Northern will not adversely affect or result in loss of additional terrestrial resources and will likely be beneficial to such resources within the project area.

4.6.2 Alternative 1

As to the Penobscot Mills project, we agree that with a flow of 350 cfs under Alternative 1, project operations would then have adverse affects on the existing beaver population in the Back Channel. This disruption could adversely affect wetland resources directly related to beaver dams created in the Back Channel.

4.6.3 Alternative 2

We agree that Alternative 2 would protect less terrestrial habitat in the immediate vicinity of the impoundment than the larger zone proposed in Alternative 1; however, as under Alternative 1, we see no justification for imposing any new mandatory easement or expanding project boundaries around the impoundment beyond what the Maine Land Use Regulation Commission already requires.

4.7 Threatened and Endangered Species

4.7.1 Applicant's Proposal

We agree with your observation that "project operations are not related to elevated mercury levels in fish within the project impoundment."

We further observe and agree with your concurrence that:

- **T&AS-67**1. nothing in the current literature directly or
indirectly links impoundment drawdowns with elevated
levels of mercury (a point which was clearly
established after hours of expert witness testimony
before the Maine Land Use Regulation Commission as part
of Section 401 Water Quality Certification hearings);
and
 - no evidence suggests that the fecundity of the eagle population at the Ripogenus project is affected by

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-64 No response required.

T&AS-65 No response required.

T&AS-66 No response required.

T&AS-67 No response required.

N T&AS-66

current drawdowns in the Ripogenus project operation. This accords with the expert testimony of Dr. Ian Nisbet at the LURC Water Quality Certification hearings. LURC Transcript I, pp. 145-50 and 235-38.

Indeed, the evidence before you suggests that the effective existing operations on threatened and endangered species (i.e., bald eagles) in the project study area has been beneficial. Open water below McKay Station has provided foraging opportunities for eagles overwintering along the West Branch. Ripogenus Project Application at p. E33-69.

Concerning wildlife associated with open water, fluctuating water levels on the storage impoundments have not had a significant impact on loon nesting success. Because nesting occurs after ice-out on the impoundments during the period mid-May to late June, and water levels typically recede less than one foot during the incubation period, no significant effects on nesting success is likely. <u>See</u> Ripogenus Project Application at p. E3.3-63.

You properly observe that, to the extent that U.S. Fish 4 Wildlife representatives believe that the long-tailed shrew and the North American lynx have ranges which overlap the project area, proposed operation would not have adverse impacts on such species because of their relative mobility.

4.7.2 Alternative 1

Does CLF's assertion that artificial nesting platforms would help to alleviate some of the effects of impoundment fluctuation on nesting common loons and other aquatic birds assume a definite adverse impact to the nesting loon population? Such proof or evidence of adverse impacts due to impoundment fluctuations is neither provided nor referenced in the DEIS.

4.8 Recreation Resources

4.8.1 Applicant's proposal

4.8.1.1 Recreation Flows

Multiday Canoe Trips

We fundamentally disagree with two of your observations: (1) that the West Branch offers one of the few remaining opportunities in the eastern United States for multiday cance trips in a wilderness setting and (2) that there is any concreted evidence of such <u>multiday</u> trips on this section of the West Branch (particularly the Ripogenus Impoundment).

39

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-68 As we indicated in the DEIS (4.7.1, p. 4-44), and subsequently in the FEIS, receding water levels of more than 2 feet can occur during the later part of the loon nesting cycle, potentially exposing the loon chicks, and increasing the likelihood of predation. Therefore, it is our view that recommending the artificial nesting platforms is appropriate.
- T&AS-69 We agree that the West Branch above Ripogenus dam is not a "pristine wilderness". We use the term "wilderness" in the context of a relatively large natural area with little development and few permanent residents.

T&AS-68

T&AS-69

m

T&AS-67

Cont'd

As to the first point, this section of the West Branch is not true "wilderness" as compared to the Allagash <u>Wilderness</u> Waterway located to the north of the Ripogenus Impoundments. If the Rip area is to be considered "wilderness" (because it is not significantly developed), then it must be recognized that it is not pristine "wilderness" in a spiritual or emotional sense, because it <u>has</u> been touched by man's hand. You should not buy into the emotional argument urged by the CLF and the environmental Coalition, but rather recognize that this section of the West Branch has for a century been shaped by man's hand.

This section of the West Branch is perhaps the best case for the concept of a "multi-use forest." Harvesting activities, a quality road network, a world class salmon fishery, and remarkably open public recreational access all coexist. You should review the record in these proceedings and reexamine whether there is any evidence that shoreline development (particularly on the Rip impoundments), forestry practices, and water level fluctuations are, or will be, radically different than current operations. Also, please consider whether current operations are disturbing anything other than the sensibilities of an elite few who don't want to see a cabin in the woods when they cance, and on most lakes, for every mile of shore frontage.

As for the second point, again, there is no evidence that there is significant use of the Rip impoundments for multiday cance trips. In support of this observation, please note that page 56 of the DeLorme Maine Atlas and Gazetteer denotes that the start of the Allagash River Cance Trip is at the Telos Lake/Chamberlain Lake thoroughfare, located approximately 5 to 7 miles northeast of the north end of Chesuncook Lake.

In further support, we ask you to reexamine the use survey results in Great Northern's application. Unlike Penobscot Mills, access to Ripogenus Impoundment is gate-controlled; GNP has therefore been able to do studies based on gate surveys. This ability to more precisely survey facilitated the summer use assessment found at Ripogenus, Vol. III, p. E5.32. This assessment indicates that about 8,000 persons used the impoundment for business purposes (rather than used it to recreate). 4,000 visitors recreated. As to the 4,000 recreators, please note that there are eleven canoeing/kayaking sites (the study lumps them together) on the West Branch (Ripogenus, Vol. 3, p. E5-43).

Also, the Applicant conducted a study of residents, leaseholders and visitors on the Ripogenus Impoundment.

This study reveals that only a fifth of the area residents use the impoundment waters for canoeing; of those who do, 56.9 percent only canoe one to two times a year. Another third canoe

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-70

T&AS-69

Cont'd

between three and nine times a year. Of the leaseholders surveyed, we find only 11 percent of the leaseholders cance at all; of that 11 percent, 59.2 percent cance once or twice a year; and another third cance between three and nine times a year (Ripogenus, Vol. 3, E5-52) just as the resident, nonleaseholders do. If the nonresident, nonleaseholder use is similar, most of the canceing/kayaking occurs in the West Branch, not the impoundment lakes; the West Branch is not burdened with development nor the threat of the same.

Further, kayaking, which presumably is almost exclusively a West Branch function, is included with canceing, inflating the still-small number of participants. Notable again, however, is the much more prevalent use of presumably more intrusive recreational activities; fishing, motor boating, ice fishing, snowmobiling and game hunting are far more common activities than canceing. (Ripogenus, Vol. III, p. E5-51 & 52).

As three quarters of the users are Maine residents (Ripogenus, p. E5-3) and many of that three quarters are Millinocket area residents, several other statistics in the study bear note:

Of area residents, about 61 percent said Ripogenus was being properly used or under-used; 49 percent said Chesuncook was being properly used of under-used; and 53 percent said Caribou Lake was either properly used or under used. About half the leaseholders also shared these assessments. Ripogenus, Vol. III, p. 85-50.

4.8.1.5 Access Pees

The settlement between GNP and the Fin and Feather Club to provide free day use to Maine residents and the settlement agreement between GNP and MEPRO should be considered enhancements. Both agreements provide tremendous benefits to the public, especially low-income users and frequent local users. These benefits more than offset any adverse impacts.

4.8.2.2 Water Level Fluctuations

We note that you have failed to discuss the impact of dry years, worst case years, and back to back dry years. You suggest that extra drawdowns would improve recreational access and use. We would note, however, that the opposite is true. Extra drawdowns would simply make it harder to use recreational docks and increase navigational hazards. This adverse impact, especially on Millinocket Lake, is not noted.

41

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

- T&AS-71 We agree that the proposed elimination of access fees for Maine residents would be beneficial by making access more affordable to low-income and local residents and would potentially increase recreational opportunities and use. The staff finds, however, the elimination of fees for Maine residents and the maintenance of fees for out-of-state users would be discriminatory. In the preferred alternative, we recommend that GNP either eliminate fees or maintain reasonable fees for both Maine residents and out-of-state residents.
- T&AS-72 We evaluated all of the alternatives using the same scenarios. We do not suggest that extra drawdown would improve recreational access and use. We have clarified Section 4.8.2.2 to indicate that increased drawdown at Millinocket Lake would adversely affect recreational uses.

T&AS-70

Cont'd

E-291

T&AS-71

4.8.2.3 Recreational Figheries

We note that the potential severe impact on fisheries is not **T&AS-73** noted from dewatering salmon eggs, dewatering lake trout eggs, dewatering smelt eggs, dewatering burbct eggs, and dewatering the Back Channel over the term of new licenses.

4.8.2.4 Recreation Access and Facilities

Access would be more difficult in dry years and worst case **T&AS-74** years because of the attendant decreases in water levels.

4.8.3 Alternative 2

4.8.3.1 Recreational Flows

Spillage flows are usually greater than 1000-2000 cfs. If, at any time subsequent to the relicensing of the project, it can **T&AS-75** be shown that a minimum number of paddlers are taking advantage of the whitewater boating opportunities, then the flows should be terminated.

4.8.3.3 Recreational Fisheries

We note that you have failed to consider, or at least **T&AS-76** articulate, the impact of dry years and worst case years on E-292 fisheries

4.8.3.4 Recreation Access and Facilities

We note that you have failed to consider, or at least **T&AS-77** articulate, the impact of dry years and worst case years on accéss.

4.8.3.6 Cumulative Impacts

You have failed to mention the impact of successive dry **T&AS-78** years.

4.9 Land Use

4.9.1 Applicant's Proposal

4.9.1.1 Shoreline Development

Your discussion suggests that LURC's existing regulations governing shoreline development are inadequate and that they do not quarantee long-term protection of the project waters for the duration of the project license. Setting aside momentarily the adequacy of LURC's regulations, it is our basic contention that, for a number of reasons, LURC's regulations should be used as the

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET **ON UPPER PENOBSCOT RIVER BASIN**

- T&AS-73 See response T&AS-56.
- T&AS-74 The FEIS has been modified to reflect this comment.
- T&AS-75 Opinion noted.
- T&AS-76 We only discuss those areas potentially affected by the alternative. See response T&AS-56.

T&AS-77 See response T&AS-56.

- T&AS-78 See response T&AS-56.
- T&AS-79 The staff revised its land use assessment and recommendations in the FEIS (see section 4.9).

basis for ensuring continued protection. First, the preference of the people affected by the project for local control should be given deference. Admittedly, LURC is a state agency and, as a result, a significant amount of local control has already been lost. However, it is considerably easier for local preferences to be considered at the state level than at the federal level. Moreover, LURC, while it does so infrequently, has the capacity to amend its rules and regulations to reflect local conditions. Regulations or restrictions imposed by you are not so easily modified, even where appropriate.

Second, the State won't sell and unwilling private landowning sellers will force expensive and needlessly difficult condemnation proceedings. For the reasons stated below (and particularly as detailed under Section 4.9.2.3), condemnation proceedings would be an unfortunate and senseless result as to private landowners who have owned, in some cases, for five generations on the Ripogenus impoundments and have been good stewards of the resource. It is simply unreasonable to consider forcing acquisition of public lands owned by the State of Maine.

Third, LURC zoning will reduce the amount of land otherwise involved in condemnation proceedings, enhance the local tax base, and encourage local private development for recreation. Zoning is considerably less costly than acquisition of a fee interest or a conservation easement. As discussed under Section 4.9.3.3, you have severely underestimated the costs that will be incurred to acquire such interests. See also Exhibit A to these comments.

Fourth, and in response to the concern that LURC may relax it regulations, we would note that LURC must meet certain statutory obligations in carrying out its zoning obligations. These statutory obligations are just as effective, if not more so, as a means of shoreline control.

By providing a technique for assessing the adequacy of local zoning laws and their enforcement, concerns that zoning would be unreliable are met. Moreover, this situation is considerably different from the <u>Alabama Power</u> case where it was held that local zoning ordinances were unreliable. In this situation we have a state governmental entity, fully-staffed by paid professionals, fully-funded, and closely monitored, which oversees zoning. UURC is not some small municipal governmental entity made up of citizens which may have little or no experience regarding administration and enforcement of zoning ordinances, where variances, zone changes, and lax enforcement are often the rule rather than the exception.

You have grossly overestimated the potential extent of residential development in the project area. A single-family dwelling is allowed in the P-GP subdistrict only if the lot has a minimum of 200 feet of frontage (and the dwelling unit meets

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-80 The staff revised potential build-out estimates to reflect LURC subdivision regulations. As stated in the DEIS, the staff acknowledges these estimates do not take into account development limitations such as steep slopes, poor soils, wetlands, or access (see section 4.9.1.1). Also, see response T&AS-79.

÷

T&AS-80

43

other extensive requirements). LURC Regs., Section 10.17.B.1. However, only two such lots may be created in this fashion over a five-year period. If a parcel of land is divided into more than two lots over a five-year period, a subdivision is created. Subdivisions are <u>prohibited</u> in P-GP subdistricts. LURC Regs. Chapter 10.16.B.3.d. Consequently, even if each individual lake and impoundment were considered a "parcel", the 2-lot/5-year limitation would result in a maximum number of only 201 dwellings for all impoundments.

You should also analyze the extent to which the shores of the Ripogenus Project impoundments are protected by LURC's Protection-Accessible Lakes (P-AL) Zone. Almost all of the shoreline of Chesuncook Lake, for example, is zoned P-AL which only allows one camp or residential structure for every mile of shore frontage.

LURC's existing regulations not only severely restrict the number of residential dwellings that can be constructed, they also control how that development must take place. These requirements set forth detailed provisions with respect to subsurface disposal systems, minimum road frontage and setbacks. as well as building heights and dimensions. Moreover, if the development would have an "undue adverse impact on existing uses, scenic character, or natural and historic resources in the area, LURC may impose additional or more protective standards with respect to clearing, frontage, and setback requirements . . . " LURC Reqs. Chapter 10.17.B.1(g)(6).

As the above demonstrates, the applicable regulatory requirements absolutely prohibit the type of development you suggest may occur. Even if it wanted to, the only way GNP could exceed the 2-lot/5-year limitation would be to petition LURC to rezone the lands as a development district, thereby permitting the creation of a subdivision. By statute, land cannot be rezoned unless there is <u>substantial evidence</u>, among other things, that

- a. The change would be consistent with . . . the Comprehensive Land Use Plan; and the purpose, intent and provisions . . . [of the LURC law]; and
- b. The change . . . will satisfy a demonstrated need in the community or area and will have no undue adverse impact on existing uses or resources . . . "

12 M.R.S.A. \$ 685-A(6).

You reference the fact that LURC approved 17 of 23 rezoning petitions between 1985 and 1992 to support the claim that LURC regulations and existing zoning patterns provide very little long-term protection for the duration of the project license. As

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

44

Charles J. Gadzik testified, the screening process that LURC staff pursues has prevented many rezoning requests from getting to a decision before the Commission. See FERC Transcript, Gadzik Testimony at p. 64. Without reviewing the merits of each of these rezoning petitions, it is impossible to know whether they somehow resulted in the degradation of the resource or that somehow the area was left "unprotected."

If LURC were to rezone the land in the P-GP subdistrict to a development district, the regulatory standards that would have to be met for approval of a subdivision to allow more than the two lots are extensive and fully protective of environmental values. LURC's regulations provide that the following standards "shall be considered in evaluating whether subdivisions . . . proposed on land adjacent to lakes":

The proposal will not adversely affect natural and cultural resource values identified as significant or outstanding:

The proposal will not, alone or in conjunction with other development, have an undue adverse impact on water quality:

The proposal will not have an undue adverse impact on traditional uses;

The proposal will not substantially alter the diversity of lake-related uses afforded within the region in which the activity is proposed;

Adequate provision has been made to maintain the natural character and shoreland;

The proposal is consistent with the management intent of the effected classification.

LURC Regs., Section 10.13-B.2.

In any event, determining the maximum number of lots which could be created is only half the equation. You candidly admit that the estimates you use "are based on current LURC regulations and do not account for development limitations such as steep slopes, poor soils, woodlands, or access." (DEIS pg. 4-57). These limitations would reduce the potential number of camps. But even more important than the number of camps that could be constructed, are the standards which must be met under today's standards to secure building permits.¹ It is difficult to argue that, if permit standards are met, that the resources would not be adequately protected. Consequently, the issue must be LURC's

¹/ <u>See LURC's land use standards for construction within 250</u> feet of a water body for both P-GP and P-AL zones.

capacity and willingness to administer and enforce its
 regulations. Again, we are talking about a state agency with a complete enforcement staff, with legal support provided by the Maine Attorney General's office. This is not some local zoning board operating on a shoestring budget.

You also state that future increased development could adversely affect project aesthetics and the wilderness recreational experience. (DEIS pg. 5-58). We question whether Ripogenus or North Twin even provides a wilderness experience. Certainly, North Twin does not. From the most remote portion of the North Twin impoundment, a radio tower on Black Cat Mountain (which is on the south shore of Millinocket Lake near Spencer Cove on Ambajesus Lake) is visible as well as the vapor column from the Millinocket Mill, and occasionally from the East Millinocket mill. Such views are not consistent with a wilderness experience.

TGAS-81 In terms of aesthetics, it is a subjective judgment that a person paddling on the impoundment is offended or his experiences impacted by cottages on the shore, especially cottages built in accordance with LURC's regulations. LURC requires that buildings be painted a neutral color; colors which clash with the natural setting are prohibited. These tests have been developed under LURC's mandate to ensure, as a criterion for approval of any permit, that "adequate provision has been made to "fit the proposal harmoniously into the existing natural environment in order to ensure that there will be no undue adverse affect on . . . scenic character . . . " LURC Regs., Section 10.13-B.1.C.

4.9.1.2 Timber Harvesting Practices

You have misconstrued LURC's regulations on timber harvesting, especially section 10.17.A.5.J (which makes clear that the prior paragraphs in that Chapter are <u>sinimum</u> requirements) and that the timber harvesting operations must be conducted in order to reasonably avoid sedimentation of surface waters. Once again, section 3.4.1 of the DEIS, which deals with a broad range of likely causes of pollution, does not mention timber harvesting activities and does note that water quality is much improved. In addition the record in these proceedings contains little, if any, evidence that timber harvesting activities are adversely impacting water quality at this time. One can certainly see that LURC's regulations and GNP's practices over the past 20 years are responsible for the present water quality in the watershed.

You state that vegetative buffers range from 50 to 100 feet. However, the P-GP subdistrict and all other LURC shoreland protection subdistricts are 250 feet from the high water mark (LURC Regs. Chapter 10.16.b.2). Timber harvesting is regulated within that 250-foot strip; Section 10.17.A.5.b.3 provides that

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-81 Opinion noted.

T&AS-82 Opinion noted. See revised sections 3.11.1.1 and 4.9.

only 40% of the volume of larger trees can be removed in any ten year period. Therefore, one can argue that the vegetative buffer is actually 250 feet since removal of 40% by volume obviously leaves 60%; those remaining trees will be a vegetative buffer. In addition, from a timber harvesting perspective, it does not seem likely that a landowner would remove 40% in year 1 and 40% of the remainder in year 11, and so on. One would expect the harvesting cycle would be greater than 10 years.

You state that timber harvesting practices can degrade water quality by altering temperature, lowering DO concentrations, and increasing concentrations of nitrates and suspended sediments. We certainly do not quarrel with that observation. However, it is important to note that LURC's timber harvesting regulations were specifically designed to prevent these very problems in the 250-foot buffer from the normal high water line of the impoundments. In a word, the primary focus of LURC's shoreland protection measures is "water" and its quality.

Finally, the first paragraph on page 4-49 of the DEIS needs to be corrected. Pursuant to LURC's regulations (Section 10.17.A.5.B), no clearcutting is allowed in the first 50 feet and a well-distributed stand of trees must be retained "so as to maintain the aesthetic and recreational value and water quality of the area and to reasonably avoid sedimentation of surface waters." In the next 200 feet, or the remainder of the P-GP subdistrict, harvesting activities may not create single openings greater than 14,000 square feet or approximately 120 feet by 120 feet; in such areas, single canopy openings of over 10,000 square feet shall be no closer than 100 feet apart. In the entire 250 foot section, only 40% of the trees by volume can be removed in any 10-year period. Timber harvesting operations, beyond the P-GP area and in the P-GP zone, have occurred continuously around the impoundments and throughout the West Branch watershed with no adverse water quality impacts evident in the record of this proceeding.

The record in this proceeding is devoid of any significant adverse impact on water quality or aesthetics from timber harvesting. Indeed, the record is the best evidence that current LURC rules regulating timber harvesting have protected water quality throughout the West Branch watershed. There is no factual basis to conclude that expansion of project boundaries is necessary to protect water quality or aesthetics from timber harvesting. LURC is a land use agency with almost twenty-five years experience in these matters. Defer to the judgement of the State of Maine on this issue.

T&AS-82 Cont'd

E-297

47

4.9.1.3 Expansion of Project Boundaries

You suggest in this section that you are concerned about "potential shoreline development and timber harvesting practices." It is imperative to note, however, that current development and timber harvesting practices are not causing any significant water quality or aesthetic problems. Potential impacts from timber harvesting activities are more than adequately addressed by LURC's regulations. (See comments under Section 4.9.1.2 above). The present quality of the impoundment water is the best indicator of that fact.

Potential development concerns should be allayed by consideration of the three likely development options. The first scenario is the continuation of present landowner policies, that is, no new leases. Obviously, the development threat posed by this scenario is nil.

The second development scenario is gradual development under the 2-lot/5-year approach. Recall, under this scenario, each landowner can create only two new leases or two buildings in any five-year period in that landowner's contiguous ownership. Moreover, any development that does occur must be done in strict conformance with LURC's development standards. The development threat posed by this scenario is also very small.

The third potential development scenario is the potential that landowners will apply to secure a zone change and intensify the developments. Zone changes, as noted above, can only be approved if (1) the change would be consistent with (a) the Comprehensive Land Use Plan, and (b) the purpose, intent and provisions of the LURC law, and (2) the change will satisfy (a) demonstrated need in the community or area and (b) will have no undue adverse impact on existing uses or resources. 12 M.R.S.A. § 685-A(6).

The reference to LURC's Comprehensive Plan brings into play the goals and policies of the Commission concerning development, including policies to:

Discourage growth which results in scattered and sprawling development patterns.

Require that provision be made for fitting development harmoniously into the existing natural environment.

Encourage orderly growth within and proximate to existing, compatible developed areas, particularly near towns and communities.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-83 The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff's recommendations (see section 5.3.4) provide measures to protect shoreland resources within the project areas while considering existing land use regulatory controls.

> Within the Ripogenus Project area, the recommended alternative proposes two options: (1) the adoption of the proposed 250-foot conservation easements for GNP owned land as defined by the MOU; or (2) a 200-foot boundary expansion on GNP-owned lands (see section 4.9.1). The proposed 250-foot easements would lie outside the project boundary and would remain under LURC land use regulatory control. For the Penobscot Mills Project, the recommended alternative proposes expanding project boundaries to generally extend 200 feet from the high water mark of the impoundments within the project area, only on land currently owned by GNP.

> Within the proposed boundary expansion areas, existing structures would be grandfathered and GNP would have the authority to review and approve proposed actions as established by the Commission under the Standard Land Use Article or SMP. See Section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.

T&AS-83

E-298

Permit subdivision developments only in areas zoned for development.

Limit residential densities on the basis of soil suitability and other site limitations.

Prevent the degradation of natural and cultural values resulting from cumulative impacts of incremental development.

Require the use of buffers, building setbacks, and landscaping to minimize the impact of land use activities upon one another and to maintain the scenic quality of shorelines and roadways.

Regulate the disposal of sewage, solid waste, manure, and septic sludge and prohibit their disposal in flood prone areas, on unsuitable soils, or in other inappropriate areas.

In addition, LURC's Regulations at Chapter 10.08, state that when lakes are involved, the review standards listed at Section 10.13-B.2 shall also be considered. That section protects natural and cultural resource values, water quality, traditional uses, natural character, and lake management goals, and requires consideration of all relevant information, including the Maine Wildlands Lake Assessment findings.

We believe the present water quality in the impoundments and the present aesthetic benefits seen from the impoundments are a result of LURC's rules and wise landowner management of their property. You should reconsider expansion of project boundaries because you have misinterpreted LURC's rules and their ability to protect these resources in the future. You also have not considered the impacts on affected lease holders, private landowners and the State of Maine.

The record in this proceeding does not justify expansion of project boundaries and issues concerning future water quality and aesthetics are best left to the people of the State of Maine who have amply demonstrated not only their concern about these resources, but their ability to protect them through appropriate regulatory agencies such as LURC.

In addition, expansion of project boundaries raises a number of issues that were not addressed in the draft EIS such as:

- (1) How are existing cottages and leases affected?
- (2) How are property owners other than the Applicant affected?
- (3) How is the Applicant affected?

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

m Cont'd

I. Existing Cottages and Leases.

On page 4-57 you stated that there are 430 dwelling units on Penobscot Mills and 70 dwelling units on Ripogenus. On page 3-47 you state GNP has granted 60 leases on Ripogenus and on page 3-48 you state GNP has granted 823 leases on Penobscot Mills.

Cottage owners have made significant investments in their property; the vast majority were built before creation of the Land Use Regulation Commission. Changes in their property are, however, subject to LURC's rules.

A number of questions were raised at the public hearing concerning the effect of expansion of project boundaries on these cottages and leases such as:

- Will leaseholders who have not built on their lot be allowed to build in the future in accordance with LURC's rules? You note GNP has issued 883 leases, but you document only 500 dwelling units on those leases. We assume other landowners on Ripogenus have also issued leases but buildings have not been erected on all of them;
- 2. Will leaseholders be able to have permanent or temporary docks in the future if they presently do not have one?
- Will cottage owners be able to maintain existing building and docks within the proposed setback zone?
- 4. Will cottage owners be able to replace buildings in the future that are destroyed or expand existing buildings?
- Will cottage owners be able to build new buildings within the setback zone in the future?
- 6. Will cottage owners be allowed to remove any vegetation within the proposed vegetation buffer?
- If changes in existing cottage are allowed or new buildings are allowed, how do affected individuals get permission from you?
- 8. Will any restrictions be placed on GNP's ability to sell existing leased lots to the leaseholder?
- 9. How will existing leases be affected if the lease is not renewed for any reason?

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

The high level of concern from GNP leaseholders was apparent at the public hearing. Answers to these questions as part of your explanation of any "grandfather" concept omitted from the draft EIS is necessary to alleviate public concerns.

II. Affected Landowners on Ripogenus.

Some 82.5 miles of shoreline on the Ripogenus impoundment is not owned by the Applicant. More than 55 miles of frontage are owned by private landowners; the rest is owned by the State of Maine.

T&AS-83 Cont'd

As noted at the public hearing, we do not believe it is appropriate to require private landowners who happen to own land on the impoundment to sacrifice their property to protect such a subjective resource as aesthetics. The present aesthetics exist in large part because of the wise stewardship of the private landowners. It is a poor reward for good land management to deprive these people of their property. Nothing in the record suggests that these landowners have adversely affected water' quality or aesthetics or that their use of their property will do so in the future.

Expansion of project boundaries on land not owned by the Applicant is in poor taste, is not mandated by the facts and may well be untenable legally. We suggest that you delete the requirement.

III. The State of Maine.

The State of Maine owns Gero Island at the north end of Chesuncook along with several miles of shorefront. Maine holds its land for its people and can be trusted to manage its property appropriately without guidelines from you. We suggest you delete any requirement of expansion of project boundaries on land owned by the State of Maine.

IV. Future Development Impacts.

We believe development on the impoundments in accordance with LURC's rules is compatible with the environment or it would not be approved.

We view some degree of development in the future on the impoundments in accordance with LURC's rules as an important potential source of diversification for the local economy. Not only will some limited development generate construction dollars, but it also will increase at least seasonal visitors to the area. As the paper industry employment has contracted, it is increasingly important to explore and develop all alternatives. It is not fair for you to preclude this option for the affected landowners and local population since you have not considered the

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-84 As noted earlier, staff revised recommendations for conservation easements and project boundary expansion (see section 4.9). The staff never intended to require the State of Maine to relinquish any land holdings for the project boundary expansion recommendations put forth in the DEIS.

T&AS-85 Opinion noted.

T&AS-85 Cont'd	economic impacts on them from the perspective of lost future opportunity.	T&AS-86	Opinion noted.
	4.9.1.4 Cumulative Impacts	T&AS-87	Opinion noted.
T&AS-86	For the reasons developed more fully in Section 4.9.1.1 hereof, we take issue with the conclusion that potential development "would contribute to the loss of wilderness character in the Upper Penobscot drainage and could degrade water quality	T&AS-88	See response T&AS-82.
	because of inadequate buffers."	T&AS-89	The staff revised its land use assessment and
	4.9.2 Alternative 1		recommendations in the FEIS (see section 4.9). We
	4.9.2.1 Shoreline Development		considered comments received during the DEIS comment period, GNP's proposed conservation easements for the
T&AS-87	The imposition of a 250-foot vegetative buffer to protect against degradation of project waters does not seem to fit within the permissible bases for requiring an expansion of project boundaries.		Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff's recommendations (see section 5.3.4) provide
	4.9.2.2 Timber Marvesting Practices		measures to protect shoreland resources within the project areas while considering existing land use regulatory
T&AS-88	We would simply reiterate that there is absolutely no evidence that anything in addition to LURC's existing regulations is needed to protect water quality, especially since the present buffer is already 250 feet.		controis.
-30	4.9.2.3 Expansion of Project Boundaries		
2	We question whether your eminent domain authority, as set forth in Section 21 of the Federal Power Act (16 U.S.C. § 814) is available in connection with the aesthetic considerations raised in the DEIS. Your regulations provide that:		
T&AS-89	<pre>[t]he [project] boundary must be located no more than 200 feet (horizontal measurement) from the exterior margin of the reservoir, defined by the normal maximum surface elevation, except where deviations may be necessary in describing the boundary according to the above methods, or where additional lands are necessary for project purposes, such as public recreation. shoreline control. or protection of environmental resources.</pre>		

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

ON UPPER PENOBSCOT RIVER BASIN

18 C.F.R. § 4.51 (h) (2) (B) (emphasis supplied). This provision is by no means a model of clarity, but seems to limit the purposes for which project boundaries can be expanded beyond 200 feet from the exterior margin of the impoundment. It does not, however, detail the bases for requiring an expansion of project boundaries up to the 200-foot mark.

52

FERC's authority to require the expansion of project boundaries up to the 200-foot mark may be extracted from two other regulatory provisions. First, you are authorized to require GNP to include in its relicensing application an analysis of costs and constraints in developing a protective zone around the impoundments to ensure public access and to protect aesthetic values. 18 C.F.R. § 4.51(f) (6) (iv).

The regulations also provide that an applicant will assume the

. . . responsibility to acquire in fee and include within the project boundary enough land to assure optimum development of the <u>recreational resources</u> afforded by the project. To the extent consistent with the other objectives of the license, such lands to be acquired in fee for <u>recreational purposes</u> shall include the lands adjacent to the exterior margin of any project reservoir plus all other project lands specified in any approved recreational use plan for the project.

18 C.F.R. § 2.7(a) (emphasis supplied).

Boiled down, beyond performing an analysis of the costs to secure a buffer to protect aesthetic values, these provisions, would seem to limit the purposes for which a project expansion could be required up to the 200-foot mark to protection of recreational resources of the impoundment.

You have concluded that the proposed project boundaries are inadequate and recommend:

creating a 200-foot building setback with a 100-foot vegetative buffer around the Ripogenus impoundment and GNP-owned property along Penobscot Mills Project impoundments would preserve existing <u>aesthetic values</u> and protect the project areas against potential future degradation of impoundment <u>water guality</u>. The building setbacks and vegetative buffers also would protect against <u>habitat modifications</u> that could adversely affect bald eagles, terestrial wildlife, and vegetation in valuable habitat around the impoundments.

DEIS, pg. 5-16 (emphasis supplied).

Of your stated purposes for requiring the 200-foot expansion of project boundaries, only preserving aesthetic values is arguably consistent with the regulations, but only to the extent that FERC wants Great Northern to study costs to acquire buffer areas. If you actually want FERC to properly exercise its authority under its own rules, its eminent domain authority would RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

Τ&AS-89 Cont'd Δ Δ Δ Δ Δ

be limited to requiring fee acquisition for recreation which, as quoted above, is not included in your motivating purposes for land acquisition. Water quality and habitat protection may serve as legitimate bases for extending the project boundaries beyond the 200-foot mark; however, that particular proposal has been appropriately rejected by you as providing only marginal additional benefits.

The existence of limitations on the eminent domain right is established by case law that deals principally with recreational or flooding needs of a project. <u>See</u>, <u>Louisiana through the</u> <u>Sabine River Authority v. Kindsey</u>, 524 F.2d 1934 (5th Cir. 1975) and <u>Chapman v. Public Utility District No. 1</u>, 367 F.2d 163 (9th Cir. 1966).

Before forcing use of condemnation proceedings in this case, you should recognize that this is not the ordinary "run of the mill" eminent domain case. The limits of your eminent domain authority in this particular case must be reviewed at three separate levels, each posing their own unique issues. First, whether you can compel GNP to impose the aforementioned buffer zones and building setbacks on land that it owns. Second, whether you can compel existing private landowners to sell their land solely for aesthetic purposes. Third, and perhaps most difficult, whether you can require that the State of Maine give up that land that it owns and manages.

As noted above, you cannot seriously ask FERC to exercise its eminent domain authority to address purely aesthetic concerns. Such exercise is completely outside of its statutory and regulatory reach. Moreover, it is clear that LURC's existing rules already protect the aesthetic values sought to be protected by you. The Federal Power Commission, in <u>Appalachian Power</u> <u>Company</u>, Project No. 2317, Opinion No. 698, June 14, 1974, held:

if adequate zoning laws are enacted by the Counties - and, we would add, if such zoning laws are adequately enforced - they should be permitted to substitute for Appalachian's acquisition, in fee or as scenic easements, of both the 200-foot [buffer] strip [].

In the same case, FPC also noted that

[if] adequate shoreline control beyond the three foot vertical line can be achieved through local zoning, such arrangements should be considered by the Commission as an acceptable alternative to additional costly fee acquisition or scenic easements.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

In this case, we are fortunate in many respects. First, existing zoning laws sufficiently stringent to address even the most outrageous concerns regarding aesthetics (as well as, water quality) are already in place. Second, those zoning laws are administered by LURC, a fully-staffed, well-administered, state agency. Third, LURC has an impressive track record for enforcing its rules and regulations. The situation present in this case appears to be exactly the situation the FPC was describing in ruling on the <u>Appalachian Power Company</u> matter; this is not the <u>Alabama Power</u> case.

With respect to LURC, and the commitment the State of Maine has towards resource protection, Steve Adams, Director of Policy Coordination for the Maine State Planning Office, testifying on behalf of the Governor King Administration at the January 25, 1995 public hearing stated that:

[the] draft environmental impact statement includes much useful information. However, the analysis and recommendations are colored by the archaic and increasingly discredited notion that somehow Washington-based bureaucrats are better able to manage Maine's resources than are the people of Maine. We believe that you have underestimated Maine's capacity and commitment to preserving this State's natural resources. Maine is very proud of it track record in developing creative and effective measures that balance the competing needs of landowners, recreational users and the imperative to protect our natural environment. Through a combination of regulation, public ownership and cooperation, Maine has earned a national reputation as a leader in resource protection.

Adams Testimony at Pg. 23.

Your notion that you should (or even can) compel private landowners to sell their respective property interests to GNP to preserve aesthetic values is even more ludicrous. The cost of acquiring those rights is grossly disproportionate to any benefit accruing (over and above those already existing by virtue of LURC's regulations).

Let us also consider the legal impediments to such a "taking." While property can, in some instances, be taken for "public purposes," it is debatable whether the purpose you articulated to justify the expansion of project boundaries is, in fact, a "public purpose." It is a very narrow class of "backcountry canceists" on multiday trips who would perceive an injury to aesthetic values, to the extent such injury were even allowed by LURC's regulations.

55

۳۱ ۵۰ **T&AS-89** ۵۰ **Cont'd**

The cost of taking the property by eminent domain, assuming it is even possible under the law, would be staggering. (See these comments Section 4.9.3.3). Given the testimony at the January 25, 1995 public hearing in Millinocket, it is abundantly clear that there are very few individuals prepared to sell their land to GNP.

A number of landowners and their representatives that would be impacted by your decision to require the taking of private property testified at the public hearing on January 25, 1995. For example, Larry Philbrick, Vice President, Prentiss & Carlisle Management Company, Inc. (which manages the timberland interests of private clients who would probably be most impacted by a decision to require acquisition of boundary lands), testified as follows:

Our clients in this area consist of three large timberland owning families who collectively own 274,853 feet - - that's 52 miles - - of shore frontage in this project area. At a depth of 200 feet, this amounts to 1,262 acres of land within this area.

Our clients have owned this property for a long period of time. The shortest length of family ownership is 50 years and one family has owed here for 110 years, now in the sixth generation of family ownership and they have a strong sense of stewardship.

These owners have not abused this land over the years. In fact, the aesthetics and water quality of this property are judged high today because of our clients' care and management of these resources. Our clients have no interest in selling the fee or conservation easements on this property.

Philbrick Testimony, FERC Transcript at pg. 72.

. . .

These landowners ought not be penalized for their stewardship of the resource by being forced to sell their lands to GNP. Rather, their history and commitment should be recognized and encouraged.

The third tier of the analysis is whether it is appropriate to take the public lands owned and managed by the State of Maine for the benefit of the public. Again, Steve Adams spoke eloquently on this point at the January 25, 1995 public hearing. Mr. Adams testified:

[the King Administration] take [s] exception to the recommendation that would require the licensee to purchase

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

or otherwise take property in the project area to implement protection measures. The State of Maine owns Gero Island and shoreland in the vicinity of Chesuncook Village which is managed by the Bureau of Public Lands. This property is managed consistent with an approved management plan. The Bureau's policy is to issue no new residential or commercial leases on these lands. Moreover, timber harvesting on this property is subject to the Bureau's integrated resource policy developed with public participation and subjected to interdisciplinary review.

The State of Maine has no intention of selling its property or its property rights within the project area. We will strenuously oppose any effort to take it by eminent domain and resent being placed in a position of potential conflict with the corporate citizen with whom we have enjoyed fruitful cooperation in the preservation of natural resources.

Instead, as noted above, we believe that the State's effective and balanced land use protection measures should be recognized and deferred to by you.

In closing, I submit that the State of Maine has been a very effective steward of its natural resources. Your staff apparently agrees, finding on page 5-24 [of the DEIS] 'the State's land use and forestry practices regulations adequately protect the natural resources of this area.' Likewise, the land and public ownership is afforded quality resource management. This result is because the people of this State take very seriously their stewardship responsibilities.

Adams Testimony, FERC Transcript at pgs. 23-24.

Based on the above discussion, and consistent with your earlier rulings, please consider that FERC may not have the authority to require the imposition of a 200-foot buffer around impoundment waters to preserve aesthetic values. That being the case, please also consider whether FERC <u>should</u> exercise that authority to compel private landowners, much less the State of Maine, to sell their lands. In our view, FERC should not.

4.9.2.4 Cumulative Impacts

While we agree in the abstract that the conservation easement could provide some additional marginal benefits through additional regulation of development and timber harvesting, the cost of obtaining those marginal benefits far outweighs the speculative benefit accrued. As discussed above, LURC should and does regulate development and timber harvesting in a manner fully protective of aesthetics and water quality.

57

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-90 Opinion noted.

T&AS-89 Cont'd

4.9.3 Alternative 2

4.9.3.1 Shoreline Development

Please see the discussion under Sections 4.9.1.1. and 4.9.2.3.

Notwithstanding the legal implications of imposing building setbacks discussed in Section 4.9.2.1 above, we would note that such setbacks also raise a number of interesting issues with respect to <u>existing</u> camps and leases. For example:

- Will leaseholders who have not built on their lot be allowed to build in the future in accordance with LURC's rules? You note GNP has issued 883 leases, but you document only 500 dwelling units on those leases. We assume other landowmers on Ripogenus have also issued leases but buildings have not been erected on all of them;
- Will leaseholders be able to have permanent or temporary docks in the future if they presently do not have one?
- Will cottage owners be able to maintain existing building and docks within the proposed setback zone?
- 4. Will cottage owners be able to replace buildings in the future that are destroyed or expand existing buildings?
- 5. Will cottage owners be able to build new buildings within the setback zone in the future?
- 6. Will cottage owners be allowed to remove any vegetation within the proposed vegetation buffer?
- If changes in existing cottage are allowed or new buildings are allowed, how do affected individuals get permission from you?
- 8. Will any restrictions be placed on GNP's ability to sell existing leased lots to the leaseholder?
- How will existing leases be affected if the lease is not renewed for any reason?

You state that, for the Penobscot Mills Project, GNP would have the option of developing a Shoreline Management Plan (SMP) providing the means to establish different building setback and lease granting requirements. You also state that the SMP should include, but not be limited to, provisions to "plan timber harvesting activities within and adjacent to the protection zones

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-91 See response T&AS-83.

T&AS-91

E-308

to follow guidelines established by LURC for timber harvesting and requiring permits from LURC before timber harvesting." We guestion whether requiring a permit for timber harvesting would solve any perceived pollution problems from timber harvesting.

T&AS-91

Cont'd

You also state that the SMP should contain provisions for "maintain[ing] project transmission line rights-of-way in a manner that minimizes adverse aesthetic effects caused by clearing vegetation." We question whether the transmission line is even visible from any of the project impoundments. Only a very short section of the transmission line is even visible from Millinocket Lake and Ambajejus Lake.

4.9.3.2 Timber Harvesting Practices

T&AS-92 See discussion under Section 4.9.2.2 above.

4.9.3.3 Expansion of Project Boundaries

Please see discussion under Section 4.9.2.3.

As a preliminary matter, GNP owns 82 miles of the 162.5 shoreline miles (or 50%) of the Ripogenus shoreline, and 82% of the North Twin Impoundment shoreline, 40% of the shoreline on Polby and other lower impoundments, or 72% of the shoreline on the entire Penobscot Mills project.

You have grossly underestimated the cost of securing the proposed conservation easement over the property. Based upon an independent appraiser's review of the proposed conservation easement on the 82.5 miles of Ripogenus Project shoreline which Great Northern does not own, we believe that the actual cost of project expansion will vary from a low of \$30.00 per waterfront foot to a high of \$100 per waterfront foot. That report of Lowell T. Sherwood, ASA, is attached as Exhibit A. These figures do not include attorneys' fees, appraisal and expert witness fees, and severance damages as a result of condemnation proceedings to assert your eminent domain authority. These additional costs could increase the actual cost of project boundary expansion by a factor of 2 or 3.

Your approach to estimating the cost of the project expansion is seriously flawed. First, the threshold matter is whether the loss of development rights equates with fee simple interest under the highest and best use of the property. The appraiser concluded that "if the conservation easement includes the development rights and prohibits use of property for private or public development, including seasonal building sites, then the remainder encumbered fee rights have no measurable market value." Exhibit A at pg. 2. As a result, he concluded that "there should be no doubt that the development rights equate to the fee simple interest in terms of market value." Id.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-92 See response T&AS-82.

T&AS-93 The staff revised potential cost estimates for the shoreline easements upon review of comments received during the DEIS comment period and updated land valuation information. The staff determined the potential cost, based on waterfront footage for the easements proposed in Alternative 1, approximately \$24.6 million, would be greater than previously determined in the DEIS. Our evaluation of benefits suggests that the additional protection of the 500-foot expansion does not merit the much higher cost of that alternative as compared to the recommended alternative (\$24.6 million versus no direct costs). See section 4.9 for further discussion.

T&AS-93

59

Second, your use of the \$/acre unit comparison is inappropriate. A unit of comparison such as \$/acre is a way of comparing data in the appropriate language of the marketplace. There is no doubt that 82.5 miles of lake frontage to a depth of 200 feet is not an acreage purchase, but rather a waterfront purchase. Buyer and seller will not speak in terms of 1.624 acres but of 82.5 miles of water frontage. The appropriate unit of comparison, therefore, should be \$/waterfront foot, not S/acre. In United States of America v. 1.499 Acres of Land. et al., the Department of Interior, in pursuing a 250-foot strip around Crawford Pond for the Appalachian Trail, attempted to use \$/acre in establishing "just compensation" in an eminent domain taking of 18,660 feet of lake frontage to a depth of 200 feet. The Court held, however, that "in placing a value on parcels involving shore frontage, the market focuses primarily on the amount of water frontage, rather than the total acreage." Court's decision attached as part of Exhibit A at page 10. The Court went on to determine a land value based on S/waterfront foot.

Third, your estimated cost assumes an artificial market of willing sellers, always at market value, and ignores the reality of severance damages, unwilling sellers, and the costs and risks associated with eminent domain. Market value assumes a willing seller, a component missing in this equation. An estimated "cost" of acquiring conservation easement derived from a single open-market sale ("Rangely Lake shoreline") between two willing parties assumes an artificial marketplace comprised of individuals lined up to convey their interests because it is the right thing to do.

No economically feasible amount of money will acquire 82.5 miles of waterfront from a multitude of owners and lessees without the benefit of eminent domain authority. Without question, there will be landowners who will be unwilling to sell at any price, and litigation will be necessary. With litigation comes legal fees, the risk of an unfavorable jury's opinion of "just commensation" and the potential for severance damages.

While our appraiser did not appraise the subject property, he did personally view the Chesuncook, Caribou, and Ripogenus shorelines on several occasions and made the following observations regarding the estimated acquisition cost of \$4.59/wff:

 The "comparable land values" used in the DEIS to estimate the potential cost of acquiring conservation easements over the subject property was probably the sale of a fee interest from Oxford Paper Company to Rangeley Lakes Heritage Trust on December 29, 1993. This property was not located along the Rangely Lake shoreline but rather along the shore of Cupsuptic Lake

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

-310

COMMENTS FROM THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN DEIS and Mooselookmeguntic Lake in Adamstown Township. This sale, at \$1,193,000, involved 16,480' of lake frontage at a price of \$72.39/wwf. Shortly after, a conservation easement including development rights was conveyed to the United States of America for \$843.000 or \$51.15/wff. The sale was not to the Maine Forestry Legacy Program and the property was not on Rangely Lake. However, regardless of the errors and omissions in the DEIS reference, this is a sale of the fee simple interest at \$72.39/wff and a conservation easement of \$51.15/wff. 2 From a list of sales of large tracts of lakefront acreage, the two most comparable parcels sold for \$26.14/wff and \$30.51/wff. The \$26.14/wff involved an unwilling seller and required condemnation. The 3/06/90 Fact Sheet published by LURC outlines the 3. "Land Use Plan and Rule Amendments Regarding the Development and Conservation of Lakes in Maine's Unorganized Areas." In that Fact Sheet, relying heavily on the Maine Bureau of Taxation, LURC estimates the current market value of Class 1, 2, 3 and 6 shorelands. "As to Shoreland, the Bureau develops estimates of front foot value for each lake based on recent sales on that lake or sale on comparable lakes." This includes Chesuncook Lake @ \$100/wff. Brandy Pond @

The second paragraph of Section 4.9.3.3 of the DEIS states that your regulations require the applicant to provide access to recreational resources and protect aesthetic resources within the project boundaries. Certainly, the building setback and vegetative buffer do nothing to provide access to recreational resources; however, no one has said that access is not sufficient. Once again, you have the authority to protect aesthetic and recreational resources, but we question whether there is a factual basis in this particular case in light of actual LURC regulations and actual conditions on the impoundments.

\$75/wff. and Caribou Lake @ \$100/wff.

Once again, there is no problem with water quality and it is not clear how the enhanced conservation measures, by repeating existing LURC setbacks, would enhance the present water quality. There is no problem with the present levels of recreation and it is not clear how the proposed conservation easement would either maintain the present levels or expand them and, if so, what impact will expanded recreation have on those resources. The present level of development does not seem to have adversely impacted either water quality or recreation. It seems

61

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-94 Opinion noted.

T&AS-93

Cont'd

inappropriate for you to speculate that any future development **T&AS-94** within 200 feet of the shoreline will always have adverse impacts Cont'd on water quality and recreation.

4.9.3.4 Cumulative Impacts

Please see discussion under Section 4.9.2.4. The expanded project boundaries may have cumulative benefits, however, those additional benefits are marginal at best and do not outweigh the **T&AS-95** cost of obtaining the additional incremental benefits.

4.10 Aesthetic Resources

Applicant's Proposal 4.10.1

4.10.1.1 Water Level Fluctuations

You state that "exposed rocky shorelines . . . can detract from overall visual experience, particularly when associated with recreational use of the area." This observation seems to be highly subjective in that it does not seem to have affected recreation to any measurable degree. Given your proposal for a 200 foot buffer area around Ripogenus Project impoundments based on aesthetic impacts, it is curious that you find a minimal negative effect on visual elements due to the Ripogenus drawdown, but are concerned with the aesthetic impact of timber harvesting and/or future development.

4.10.1.2 Development Along Shorelines

In the first paragraph, we would note that the vegetative buffer is actually 250 feet and the building setback 100 feet or more.

Your projections in the third paragraph are grossly out of line with what LURC would likely approve. There is simply no basis for the argument that temporary boat docks are a problem, especially along the Ripogenus impoundments. Moreover; LURC does not routinely approve the construction of permanent boat docks.

Given LURC's regulations, your conclusions in the fourth paragraph are plain wrong, especially when new development in accordance with existing regulations is compared to existing pre-LURC development. Indeed, your conclusion supports our claim that additional restrictions on North Twin are not necessary.

4.10.1.3 Influence on Forestry Practices

The vegetative buffer under LURC's rules is actually 250 T&AS-98 feet

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-95 Opinion noted.

T&AS-96 Opinion noted.

T&AS-97 The staff revised its land use assessment and recommendations in the FEIS (see section 4.9).

T&AS-98 No response required.

T&AS-96

T&AS-97

-312

P0.244T	4.10.2.2 Development Along Shorelines	T&AS-99 1	
1873-33	The impact of gradual development over a number of years has not been, but should be, considered.	۲ a	
	4.10.3 Alternative 2	T&AS.100	
T&AS-100	LURC has an actual 250 foot vegetative buffer. When you observe "restrict excessive shoreline development", please note: first, development on the Ripogenus Impoundment is not excessive. In fact, virtually all of Chesuncook Lake is zoned P-AL (Protection-Accessible Lake), which only allows one camp for every mile of shore frontage. That is not excessive. Second, banning all development is an odd way to restrict excessive		
	development.	T&AS-101	
4.10.3.2 Development Along Shorelines		T&AS-102	
T&AS-101	We question whether, if all that is to be gained is a "minor" visual benefit, why is it necessary to radically expand project boundaries?	T&AS-103	
	4.10.3.3 Influence of Forestry Practices		
	A No mention is made of the P-GP zone which is hy far the		
က် T&AS-102 ယ	greater in terms of land coverage, as compared to P-SL and MUN- SL.	T&AS-104	
ω	4.11.3 - Cultural Resources- Alternative 2		
	On pages 4-6 and 4-7, you document that impoundment levels, flow levels, and environmental resources are all at risk, to varying degrees during dry and worst case years because of Back Channel flows of 165 cfs. In addition, we have detailed environmental impacts, which you have failed to discuss, in our comments under Sections 4.2.3 and 4.4.3. In this section, you have omitted any discussion of impacts on cultural resources from these events.		
T&AS-103	varying degrees during dry and worst case years because of Back Channel flows of 165 cfs. In addition, we have detailed environmental impacts, which you have failed to discuss, in our comments under Sections 4.2.3 and 4.4.3. In this section, you have omitted any discussion of impacts on cultural resources from these events.		
T&AS-103 T&AS-104	<pre>varying degrees during dry and worst case years because of Back Channel flows of 165 cfs. In addition, we have detailed environmental impacts, which you have failed to discuss, in our comments under Sections 4.2.3 and 4.4.3. In this section, you have omitted any discussion of impacts on cultural resources from these events. With respect to the Penobscot Indian Nation (*PIN*), reduced impoundment levels and reduced or eliminated flows will have obvious adverse impacts on fishery resources throughout the West Branch and the main stem of the Penobscot. Also, loss of dilution flows could have direct impacts on PIN islands and reservation lands in the main stem of the Penobscot. These impacts should be noted and discussed.</pre>		
T&AS-103 T&AS-104 T&AS-105	<pre>varying degrees during dry and worst case years because of Back Channel flows of 165 cfs. In addition, we have detailed environmental impacts, which you have failed to discuss, in our comments under Sections 4.2.3 and 4.4.3. In this section, you have omitted any discussion of impacts on cultural resources from these events. With respect to the Penobscot Indian Nation (*PIN*), reduced impoundment levels and reduced or eliminated flows will have obvious adverse impacts on fishery resources throughout the West Branch and the main stem of the Penobscot. Also, loss of dilution flows could have direct impacts on PIN islands and reservation lands in the main stem of the Penobscot. These impacts should be noted and discussed. With respect to the local population, the significant loss of power production and the significant cost of project costs for this Alternative 2 (see our discussion of Sections 2.4, 4.9, 4.10 and 4.12 for details) will greatly weaken the Applicant's competitive position over the new license terms and directly</pre>		
T&AS-103 T&AS-104 T&AS-105	 Now revers, and environmental resources are art at risk, to varying degrees during dry and worst case years because of Back Channel flows of 165 cfs. In addition, we have detailed environmental impacts, which you have failed to discuss, in our comments under Sections 4.2.3 and 4.4.3. In this section, you have omitted any discussion of impacts on cultural resources from these events. With respect to the Penobscot Indian Nation ("PIN"), reduced impoundment levels and reduced or eliminated flows will have obvious adverse impacts on fishery resources throughout the West Branch and the main stem of the Penobscot. Also, loss of dilution flows could have direct impacts on PIN islands and reservation lands in the main stem of the Penobscot. These impacts should be noted and discussed. With respect to the local population, the significant loss of power production and the significant cost of project costs for this Alternative 2 (see our discussion of Sections 2.4, 4.9, 4.10 and 4.12 for details) will greatly weaken the Applicant's competitive position over the new license terms and directly 		

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-99 The staff considered the effects, gradual or otherwise, of potential long term development in our land use assessment and recommendations (see section 4.9).
- T&AS-100 The recommendations in the FEIS would not ban development. Development could occur behind the proposed conservation easement or project boundary expansion areas and along other portions of the shoreline areas (see sections 4.9 and 4.12).
- T&AS-101 See revised section 4.10.
- &AS-102 See revised section 4.10.
- **&AS-103** As stated in the FEIS, the effects of this alternative would be the same as those of the Applicant's proposal (see section 4.11).
 - The staff reviewed available information regarding the Penobscot Indian Nation's claims to lands and resources within the West Branch region. The Commission concurs with the Maine State Department of the General Attorney findings that the Penobscot Indian Nation retains no ownership or title to lands within the branches of the Penobscot River (see section 4.11.1.2). The staff finds that consideration of the Penobscot Indian Nation's traditional practices within the project area are outside the scope of Section 106. The staff does not recommend the Penobscot Indian Nation be included as a concurring party to the Programmatic Agreement associated with the Ripogenus and Penobscot Mills projects. The staff acknowledges, however, the interest of Penobscot Indian Nation in the management of historic properties potentially eligible for listing in the National Register of Historic Places within the project areas. Accordingly, the staff recommends the Penobscot Indian Nation be consulted during the development of the revised Cultural Resource Management Plans for the Penobscot Mills and Ripogenus Projects.

T&AS-105 Opinion noted.

result in significant additional job losses in the mills and throughout the region. Each of these impacts will adversely affect local cultural resources such as the population base, employment levels, increased social services costs, reduced local tax payments and will also lead to increased crime and domestic violence. These cultural impacts must be discussed in order for you to assess the impact of this alternative on the Applicant's "workers, and the related community" as required by Title 16 U.S.C. Section 808.a.2.D.

Area residents view access to and enjoyment of recreational benefits such as fishing to be part of their culture. Adverse impacts of this alternative on fishery resources also need to be discussed since recreational use of the impoundments and river by the area population is heavy.

Expansion of project boundaries will adversely impact the culture of the local population by artificially limiting the number of cottages on the impoundments. This impact should be noted.

4.11.4 - Cultural Resources - No-Action Alternative

Your discussion fails to note that this alternative will avoid the project costs and energy losses detailed in section 2.4 of these comments. Those costs are significant and will adversely affect the Applicant's competitive position, thereby adversely affecting all cultural resources in this area as the local economy weakens and can no longer support the current population.

This benefit of the no-action alternative on local cultural resources must be discussed in order to quantify losses to local cultural resources from the other proposals. Even the voluntary environmental enhancements of the Applicant's proposals will have an adverse cultural impact from effects on the local economy and that impact needs to be identified and justified.

4.12 - Socioeconomic Resources

You are required by Title 16, Section 808.a.2.D to consider the impact on the Applicant's workers and the related community. The section reads, in part, as follows:

T&AS-107"In the case of an applicant using power for the applicant's own industrial facility related operations, the effect on the operations and efficiency of such facility or related operations, <u>its workers</u>, and the related community." (emphasis added)

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

- T&AS-106 Opinion noted.
- T&AS-107 The FEIS presents the total state-wide declines in employment and wages that would be produced by a plant shut-down. Using the correct regional multiplier of 1.61 indicates that closing down the coater would decrease regional employment by a total of approximately 383 jobs. As noted above, the staff's finding is that neither the preferred Alternative Two (both versions), nor the Applicant's Proposal, would result in a plant shut-down.

E-314

T&AS-106

T&AS-105

Cont'd

Your socioeconomic discussion is woefully inadequate in documenting the dependence of the local economy on the direct and indirect purchases of Great Northern and the lack of viable alternatives due to the geographic isolation of the area.

You have also omitted any discussion of the socioeconomic impacts on the surrounding area, northern Maine economy and State wide economy.

These omissions are a disservice to the thousands of Maine residents who depend for their livelihood on the Applicant's economic contributions. In addition, you have failed to document the losses which will be suffered by Maine people if the Applicant cannot successfully compete over the new license terms. Thousands expect you to reaffirm that commitment of resources to their economy.

4.12.1.1 - Socioeconomic Resources - Applicant's proposal - Reployment and Fiscal Impact

Your discussion should note the Applicant's need for low cost power to compete and that its proposal will best maintain or enhance the existing economy of the local region as well as the thousands dependent on the Applicant's purchases of goods and services.

The geographic isolation of the area and lack of economic alternatives needs to be discussed as well as recent loss of population and economic impacts from the Applicant's downsizing in response to competitive pressures.

Recent history well demonstrates area dependence on the Applicant and the economic and population losses which have occurred. See also Testimony of David Cole, Vice President of Eastern Maine Development Corporation, attached hereto as Exhibit D. This discussion needs to be included in order to fully assess the impact on the alternatives.

You should also note that the beneficial economic impacts from increased whitewater boating and recreational fishing are minor as compared to the potential negative economic impacts on the Applicant.

4.12.2.1 - Socioeconomic Resources - Employment and Fiscal Impact - Alternative 1

Flows of 350 cfs down the Back Channel would reduce power generation at the Millinocket station by 20,800 MNH as you state on page 2-34 in section 2.4.2.4. The Applicant would need to replace that power at a cost of \$83.22 / mwh x 20,800 mwh * \$1,730,976 per year. The Applicant has stated that this cost, alone, will lead to the loss of its coated paper operation, 238

65

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-108 We agree that GNP will continue to need low-cost hydroelectric power to remain competitive. The recent downsizing was due primarity to declines in demand for GNP's output from the two mills, significant world-wide over-capacity, and strong price competition. The staff agrees that GNP is a major component of the regional economy. The preferred Alternative Two would not adversely affect the mills' competitiveness.
- T&AS-109 See response T&AS-2. We agree that Alternative One could require the coated paper operation to shut down, and could result in a decline in direct employment. The precise magnitude of this direct decline is difficult to determine. It is not necessarily true that an increase in power costs would produce a dollar-for-dollar decline in wages, and ultimately in employment. The extent of any reduction in employment would depend on the percent increase in the total cost of production that occurs as a result of an increase in power, and on the percent of total production costs that go to labor. The multiplier of 3.2539 cannot be used to estimate the total decline in regional employment, the appropriate multiplier is 1.61.

μ UTLAS-108

T&AS-109

T&AS-107

Cont'd

related mill jobs and lower purchases throughout Maine's economy.

Our review of the total cost of this alterative reveals an annual range from a low of \$5,334,871 to a high of \$8,545,940.

The Applicant's mills have not made a profit in a number of years and the paper industry appears to be in the most severe recession in its history; this recession has resulted in numerous mill and machine closing throughout North America. The Applicant must become a low cost producer if it is to survive.

This alternative imposes substantial additional annual costs that will either result in job layoffs or less funds available for necessary modernization.

We estimate each mill job costs the Applicant \$50,000 per year for wages and benefits. This alternative will cost from \$5,334,871 / \$50,000 = 107 jobs to \$8,545,940 / \$50,000 = 171 jobs even if the Applicant is not forced to shut down its coated paper operation.

Using the revised Maine Department of Labor multiplier effect (see Exhibit C), the losses will range from 107 x 3.2539 = 348 area jobs to 171 x 3.2539 - 556 area jobs.

Given the annual costs in addition to water down the Back Channel, the conclusion is inescapable that the local population will bear the full brunt of all changes by the Applicant to absorb such substantial increases in its operating costs.

In addition, even if the Applicant did not offset its increased operating costs by reducing employment, \$5.3 million to \$8.5 million per year would not be available to pay wages and wage increases, pay municipal taxes, purchase goods and services or modernize the facilities.

Workers in the mills have made major monetary concessions to the Applicant in light of increasing competitive pressures in the marketplace which have resulted in widespread layoffs and machine shutdowns locally and machine shutdowns and plant closing elsewhere. Arbitrarily mandating increased costs from \$5.3 million to \$8.5 million per year will reduce the ability of mill workers to regain past concessions to the Applicant or gain wage increases to improve their standard of living. This impact must be noted in your discussion and decision making process as part of evaluating the impact of this alternative on the workers and related community. 16 U.S.C. Section 808.a.2.D.

Many of the Applicant's suppliers of goods and services (including many of the more than 200 members of ASSSET) have also made substantial economic concessions due to the same competitive pressures. Arbitrarily mandating increased costs from \$5.3

66

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

္က r&AS မ် Cont'd ဂ m T&AS-109

million to \$8.5 million per year will divert substantial resources away from the businesses that work with the Applicant and who are part of the affected community impacted by this alternative.

Increased costs of the magnitude imposed by this alternative could result in a radically smaller company or perhaps lead to a shutdown of the Applicant's mills.

T&AS-109 Cont'd In addition, simulations show that the average dry year (84SAI350), the dry year (85SAI350) and the typical year (MN3AI350) result in numerous weeks passing the minimum flow at North Twin and seriously deplete Ripogenus storage. It is inevitable that average power production will be reduced under these circumstances, forcing the Applicant to replace the lost power at \$83.22/mwh thereby increasing its costs, and during extreme periods (such as March during the typical year), power production may be curtailed altogether. These events may lead to mill shutdowns throwing most, if not all, mill workers out of work.

It is hard for us to believe that any rational person could advocate this alternative in light of these repercussions.

Your analysis of the 500-foot conservation easement is unrealistic. Conservation easements will preclude all future development and deprive the local economy of an important possibility for diversifying its economy.

We totally disagree with your statement that "lot values would adjust to the revised setbacks." A piece of property 500 feet back from the water is not shorefront property and no one would pay as much for such a lot as for shorefront property. Truly, one would have to "take a taxicab" to get to the shore from a cottage set that far back from the shore.

The economic impact of the 500-foot expansion of project boundaries needs to be discussed, not only from the Applicant's perspective of out-of-pocket costs and lost development opportunity on its own property, but also from the perspective of (a) existing landowners who will be forced to sell their property, (b) existing leaseholders affected by expansion, and (c) the public who will be precluded from building on presently undeveloped areas in accordance with LURC's rules.

The economic impact of expansion of project boundaries on the Applicant under this alternative has been discussed in these comments at Sections 2.4, 4.9 and 4.10. Our discussion established a range of value to acquire the remaining non-Applicant lands around Ripogenus in fee or by conservation easement from \$13,068,000 to \$43,560,000. It seems obvious that acquisition of a 500 foot strip will cost the Applicant more than

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-110 The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). Under the revised recommendations. GNP would have to acquire conservation easements on non-GNP owned land only under Alternative 1. The staff revised potential cost estimates for the shoreline easements upon review of comments received during the DEIS comment period and updated land valuation information. The staff estimated that the potential cost of approximately \$24.6 million, based on waterfront footage for the easements proposed in Alternative 1, would be greater than previously determined in the DEIS. Under the recommended alternative (Alternative 2) the proposed conservation easements and project boundary expansion would only be on GNP owned lands. Our evaluation of benefits suggests that the additional protection of the 500-foot expansion does not merit the much higher cost of that alternative as compared to the recommended alternative (\$24.6 million versus no direct costs). See section 4.9 for further discussion.

E-317

acquisition of a 200 foot strip. The difference in cost to acquire a fee interest will not vary appreciably from the cost to acquire a conservation easement because the easement will prohibit all practical use of the strip other than limited timber harvesting in a portion. Your analysis of the cost of acquisition is erroneous and needs to be revised. The impact of this cost on the Applicant, its workers and the related community needs to be discussed.

As noted earlier, the impact on those landowners forced to sell their property should be discussed. At the very least, that impact should be discussed from the perspective of lost opportunity in the future.

In addition, the economic cost of any impacts on existing leases must be evaluated and discussed. If existing leaseholders can no longer build if they have not yet done so, the impact on that individual should be evaluated. If expansion or repair of existing cottages will be limited, the economic impact needs to be discussed. If limitations on removing vegetation will be imposed, the impact on value needs to be evaluated as compared to cottages on other lakes without such restrictions.

Also, the impact on the local economy of no future development on the impoundments needs to be evaluated and the economic cost of the foregone opportunities quantified. The impact on the public of limiting the availability of cottages on the impoundments should also be discussed.

You note on page 4-5 that this alternative will result in reduced impoundment levels and flows during dry or worst case years which will adversely impact recreation and fisheries. In your discussion of section 4.12.1.1 on page 4-71 you note the dollar value of whitewater boating and fishing to the local economy. However, in this section you have omitted the impact on those economic activities of adverse environmental impacts from Alternative 1. In order to fully evaluate this alternative, you need to quantify those impacts and show the economic impact of whitewater boating and fishing from reduced flows and reduced impoundment levels.

4.12.2.2 - Effect On GMP Operations

Your discussion concerning the impact on Great Northern of
lost generation needs to be revised. Your discussion is not
consistent with section 2.4.2.2 or Table 2-7. See our comments
at section 2.4.2.2. The annual cost of alternative 1 to Great
Northern in terms of lost power alone is \$3,830,283.00. We
believe that this impact would not only jeopardize the coated
operation in the Applicant's Millinocket mill but could
jeopardize the viability of the entire operation.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-111 Comment noted. See response T&AS-56.

T&AS-112 See response T&AS-25, T&AS-26, T&AS-27, and T&AS-110.

T&AS-111

T&AS-110 Cont'd

We totally disagree with your estimate of the cost of acquiring easements for the Ripogenus and Penobscot Mill projects at approximately \$9.5 million. As detailed in these comments at section 4.9, we estimate a range of value from approximately \$13 million to approximately \$43 million for this cost. That cost also does not include the lost opportunity cost on the Applicant's own property which would be restricted from further use.

T&AS-112You have also omitted any discussion of the impact of these
costs on Great Northern's ability to buy goods and services,
compete economically, maintain competitive wage rates, or
modernize its operations.

4.12.2.3 - Cumulative Impacts

You have omitted the lost opportunity cost to the region of
precluding any development on the impoundments over the new
license terms. This lost opportunity to diversify the local
economy is a significant negative impact of Alternative 1.

4.12.3.1 - Alternative 2 - Employment and Fiscal Impact

Your discussion needs to be revised in light of project costs and energy losses for this alternative as discussed under Sections 2.4.1.2, 2.4.1.3, 2.4.2.5 and 2.4.2.6 of these comments.

This alternative will impose increased annual operating costs on the Applicant ranging from \$3,441,532 to \$7,476,479. Costs of this magnitude are difficult to distinguish from those imposed by Alternative 1.

The costs imposed by this alternative greatly exceed those of 165 cfs down the Back Channel which, costs 9,900 mwh x \$83.22 / mwh = \$823,878.00 per year.

Given the highly competitive nature of Applicant's business, increased costs can only be dealt with by reducing expenses elsewhere. Therefore, this alternative will cost between \$3,441,532 / \$50,000 = 69 jobs and \$7,476,479 / \$50,000 = 150jobs with 165 cfs down the Back Channel and (\$3,441,532 - \$823,878 = \$2,617,654 / \$50,000 = 52 jobs to (\$7,476,479 - \$823,878 = \$6,652,601 / \$50,000 = 133 jobs without Back Channel flows.

Using the revised multiplier from the Me. Dept. of Labor (<u>see</u> Exhibit C), these 52 jobs translate to a loss of a total of 52 jobs x 3.2539 = 169 area jobs; if 133 mill jobs are lost, a total of 433 area jobs are lost.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-113 Opinion noted.
- T&AS-114 The Commission has changed its approach to economic analysis to better reflect the changing conditions in the electric power market brought about by deregulation. We have adopted a current cost approach under which a project's cost of producing power and the value of its output (i.e., the cost of acquiring a project's output from the least-cost alternative) are based on current market costs and prices. We identified the most feasible least-cost source of alternative power for GNP; it was found to be firm purchases from Bangor Hydro Electric (BHE).

The staff assumes that there will be no inflation and escalation of capital costs fuel prices, operating and maintenance costs, and power prices over the 30 year analysis period. Current year capital costs are amortized over the 30 year financing period at the cost of capital appropriate for the applicant. No future inflation or escalation is assumed beyond the license issuance date, and no long-term levelized values for energy and capacity are used (i.e., their use implicitly incorporates assumptions about long-term inflation and escalation rates into an analysis). The staff's position is that our revised approach to economic analysis explicitly includes the appropriate economic factors, and accurately reflects the current conditions in the electric power industry. All the economic analyses in Sections 2.4 and 5.1 of the EIS have been redone using our revised methodology. The staff's position is that our revised economic methodology is consistent with the FPA.

See also responses T&AS-2, T&AS-43, and T&AS-107.

The brunt of increased costs will be borne by the local and regional economy and even if layoffs should not occur, the funds diverted to the costs of this alterative will not be available to maintain wage rates, increase wage rates, purchase goods and services, pay taxes, or modernize operations.

Workers in the mills have made major monetary concessions to the Applicant in light of increasing competitive pressures in the marketplace which have resulted in widespread layoffs and machine shutdowns locally and machine shutdowns and plant closing elsewhere. Arbitrarily mandating increased costs from \$3.4 million to \$7.5 million per year which reduce the ability of mill workers to regain past concessions to the Applicant or gain wage increases to improve their standard of living. This impact must be noted in your discussion and decision making process as part of evaluating the impact of this alternative on the workers and related community. 16 U.S.C. Section 808.a.2.D.

Many of the Applicant's suppliers of goods and services (including many of the more than 200 members of ASSSET) have also made substantial economic concessions due to the same competitive pressures. Arbitrarily mandating increased costs from \$3.4 million to \$7.5 million per year will divert substantial resources away from the business community which works with the Applicant and who are part of the affected community impacted by this alternative.

In light of these repercussions, it is difficult to view this alternative as reasonable.

4.12.3.2 - Effect on GMP Operations

The economic impact of 200 foot expansion of project boundaries needs to be discussed, not only from the Applicant's perspective of out of pocket costs and lost development opportunity on its own property, but also from the perspective of (a) existing landowners who will be forced to sell their property, (b) existing leaseholders affected by expansion, and (c) the public who will be precluded from otherwise building on presently undeveloped areas in accordance with LURC's rules.

The economic impact of expansion of project boundaries on the Applicant under this alternative has been discussed in these comments at Sections 2.4, 4.9 (are especially Section 4.9.3.3), and 4.10. Our discussion established a range of value to acquire the remaining non-Applicant lands around Ripogenus in fee or by conservation easement from \$13,068,000 to \$43,560,000. The difference in cost to acquire a fee interest will not vary appreciably from the cost to acquire a conservation easement because the easement will prohibit all practical use of the strip

T&AS-115

T&AS-114

Cont'd

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-115 See response T&AS-110.

of the cost of acquisition is erroneous and needs to be revised. We believe that \$22,000,000 is a reasonable estimate of the cost to the Applicant of a voluntary acquisition of the fee or conservation easements required by this alternative. However, because the State and private landowners are not willing sellers, condemnation costs (including severance damages) will push fee or easement costs for the 82.5 miles of Ripogenus Project shorelands higher than this figure. The impact of this cost on the Applicant, its workers and the related community needs to be discussed.

As noted earlier, the impact on those landowners forced to sell their property should be discussed, at least from the perspective of lost opportunity in the future and, for the State of Maine, loss of its sovereign right of self-determination over use of its property.

In addition, the economic cost of any impacts on existing leases must be evaluated and discussed. If existing leaseholders can no longer build if they have not yet done so, the impact on those individuals should be evaluated. If expansion or repair of existing cottages will be limited, the economic impact needs to be discussed. If limitations on removing vegetation will be imposed, the impact on value needs to be evaluated as compared to cottages on other lakes without such restrictions.

Also, the impact on the local economy of no future development on the impoundments needs to be evaluated and the economic cost of the foregone opportunities quantified. The impact on the public of limiting the availability of cottages on the impoundments should also be discussed.

As noted in these comments, we estimate that this alternative imposes annual costs on the Applicant ranging from a low of \$3,441,532 per year to as much as \$7,476,479 per year.

You should discuss the impact of costs of these magnitudes on the ability of the Applicant to compete, maintain employment, purchase goods and services, pay taxes, pay increased costs and modernize its facilities.

As compared to your original annual cost estimates of \$213,000 (Table 2-5) plus \$1,136 million or \$1.679 million (Table 2-7), the revised cost of this alternative are substantial and prohibitive in terms of impacts on the Applicant.

71

4.12.3.3 - Cumulative Impacts

Increased costs ranging from \$3,441,532 to \$7,476,479 peryear for this alternative will have devastating cumulativeimpacts which you have not discussed. Such impacts includefurther job losses ranging from 52 to 150 mill jobs (and 169 to

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-116 This alternative is not being recommended. The staff's finding is that its percent increase power cost, and its percent decline in power production, may be at the threshold where the competitiveness of the two mills would be adversely affected. If this occurs, there would be additional regional declines in employment and wages. The employment decline should be calculated using the multiplier of 1.61.

T&AS-115 Cont'd

T&AS-11 Cont'd	488 area jobs, using the revised MDOL multiplier) further population losses, further area economic losses and further regional impacts. All of these impacts should be discussed. In addition, expansion of project boundaries will restrict future possibilities to diversify the local economy.
	4.12.4 - No Action Alternative
	The socioeconomic benefits of this alternative are apparent by noting the annual expenses imposed by the Applicant's proposal and the alternatives.
T&AS-11	The Applicant's proposal incurs increased costs of \$1,618,629 per year (<u>see</u> 2.4.2.3 and 2.4.2.6 of these comments). Increased costs of this magnitude will adversely affect the Applicant's ability to compete and will adversely affect the local economy.
	Alternative 1 imposes costs over the no action alternative ranging from a low of \$5,334,871 to a high of \$8,545,540 per year.
	Alternative 2 imposes costs over the no action alternative ranging from a low of $$3,441,532$ to a high of $$7,476,479$.
មា ស្	4.15 - Relationship Between Short Term Uses and Long Term Productivity
N N TEAS.11	You state that "The recommended alternative is designed to provide significant long-term enhancement of biological and recreational resources of the system, while meeting energy and economic needs of GNP and the local population."
1443-11	In light of the costs of the recommended alternative (\$3,441,532 to \$6,652,601 per year), we recommend that you revise the recommended alternative by deleting the conservation easement and wetland enhancements beyond the Applicant's Proposal.
	5.3 Environmental Comparison of Alternatives
T8.8 C 44	Table 5-2 summarizes the effects or costs of each alternative for each resource. Our comments on sections 2.4, 4.9, 4.10, 4.11 and 4.12 detail the costs and effects of the alternatives from our perspective.
10(43-11	With respect to levelized annual net costs listed in Table 5-2, we note:

 (a) the cost of the Applicant's proposal is \$1,618,629 per year (See comments at 2.4.2.3);

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-117 No response required.
- T&AS-118 Opinion noted.
- T&AS-119 See responses T&AS-25, T&AS-27, and T&AS-28. The staff's economic analysis has been revised so that the annual net benefits are present in current 1996 dollars. Levelized power values were not used in the revised analysis. Table 2-8 presents the revised annual net benefits. Table 5-2 presents only the direct net economic benefits at the two projects based on changes in the value of the power they produce, and their cost of producing power.
- (b) the cost of Alternative 1 ranges from a low of \$5,334,871 to a high of \$8,545,940 per year (<u>See</u> comments at 2.4.1.2, 2.4.2.4 and 2.4.2.6); and
- (c) the cost of Alternative 2 with leakage flows in the Back Channel ranges from a low of \$3,441,532 to a high of \$6,652,601 per year (<u>See</u> comments 2.4.1.3, 2.4.2.5 and 2.4.2.6).

Levelized annual net costs for each proposal need to be revised.

With respect to streamflows, our comments at 4.2 and 4.4 detail the circumstances under which flows equal to the Applicant's proposed flows cannot be maintained and state water quality certificates will be violated for 350 cfs and 165 cfs down the Back Channel. Appropriate changes to note these impacts should be made in Table 5-2.

T&AS-119 Cont'd

With respect to fisheries, our comments at 4.2 and 4.4 detail adverse fisheries impacts from 350 cfs and 165 cfs down the Back Channel. Appropriate changes to note these impacts should be made in Table 5-2.

With respect to socioeconomics, our comments at 2.4, 4.9, 4.10, 4.11 and 4.12 establish that the costs of expansion of project boundaries are substantial and range from \$3,441,532 to \$6,652.601 per year (jeopardizing at least 238 mill jobs and well more than 1,200 jobs total, including multiplier), as well as additional impacts on affected landowners forced to sell their property and foregone opportunity benefits for the entire area. These impacts should be noted in Table 5-2.

Table 5-2 should be revised to include all of the foregoing impacts.

5.3.1 Streamflow

As detailed in our comments on Sections 4.2 and 4.4, we question your conclusion that water is available for substantial Back Channel flows because of the state water quality violations and adverse fisheries impacts which occur under almost all simulations of 350 cfs or 165 cfs down the Back Channel.

Flows of 350 cfs.

T&AS-120

With respect to flows of 350 cfs, the average year (82SAI350) simulation shows dewatering of lake trout eggs in North Twin which violates the state water quality certificate. The average dry year (84SAI350) shows extensive dewatering of salmon eggs below McKay and dewatering of lake trout eggs, with each of these simulations violating the applicable state water quality certificate and insufficient storage available to further

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-120 We have already responded to these comments in responses T&AS-46 and T&AS-49. We have clarified the language in section 5.3.1 with respect to water availability.

supplement flows. The dry year (85SA1350) shows flows below McKay which not only dewater salmon eggs, but severely restrict habitat for all species during March (violating the state water quality certificate), lake trout eggs are dewatered (violating the state water quality certificate), and the Back Channel will be dewatered in March. In addition, storage is insufficient to augment flows.

The wet year (76RAI350) shows dewatered salmon and lake trout eggs violating both state water quality certificates. The typical year (MN3AI350) shows 400 cfs flows below McKay during March dewatering all salmon eggs, jeopardizing all species by severely restricting habitat and grossly violating the state water quality certificate, with insufficient storage to supplement flows, lake trout eggs are dewatered (violating the state water quality certificate), flows from North Twin fall to 1,300 cfs during March (dewatering the Back Channel and violating the state law requirement to pass 2,000 cfs at Millinocket).

In light of these impacts, we ask that you reconsider whether sufficient water is available for 350 cfs flows down the Back Channel and, at a minimum, list the adverse environmental impacts and state water quality violations which will repeatedly occur over the new license terms for this alternative.

Flows of 165 cfs.

With respect to flows of 165 cfs, the average year (82SAI165) simulation shows dewatering of salmon eggs and dewatering of lake trout eggs, which violates the applicable state water quality certificates. The average dry year (84SAI165) shows dewatering salmon eggs, dewatering of toque eggs, violations of each state water quality certificate, and insufficient storage for supplementing flows. The dry year (85SAI165) shows dewatering of salmon eggs, dewatering of togue eggs, violations of both state water guality certificates and. since minimum flows pass at North Twin 71% of the time, insufficient storage to supplement flows. The wet year (76RAI165) shows dewatering of salmon eggs and dewatering of lake trout eggs, violating both state water quality certificates. The typical year (MN3AI165) shows flows below McKay as low as 400 cfs during March, grossly violating the state water quality certificate, dewatering virtually all salmon eggs, reducing habitat for all life stages of salmon and all other fishery resources (to such an extent that elimination of species will occur), lake trout eggs will be dewatered violating the state water quality certificate and, since minimum flows pass at North Twin 03% of the time, it is likely that the Back Channel will be dewatered and insufficient water will be available to pass 2,000 cfs at Millinocket. Storage is insufficient to supplement the flows.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-120 Cont'd

E-324

In light of these impacts, we ask that you reconsider whether sufficient water is available for 165 cfs flows down the Back Channel and, at a minimum, list the adverse environmental impacts and state water quality violations which will repeatedly occur over the new license terms for this alternative.

5.3.2 Fisheries Enhancement

We generally agree with your discussion but ask that you add the adverse impacts on fisheries below McKay station which will result from substantial flows in the Back Channel and reconsider the impact on lake trout since we believe the data shows the impact will be greater than that included in your discussion. You also have not noted that dewatering of the Back Channel would eliminate whatever marginal salmon fishery might be established in the Back Channel. You have also failed to note that the primary food source for the salmon stock below McKay station is smelt drift from Ripogenus; even though substantial flows down the Back Channel would create habitat for salmon, nothing in the record indicates that smelt drift would accompany those flows. In the absence of smelt drift, sufficient food would not be available to support several hundred adult salmon in the Back Channel.

5.3.3 Wetlands

There is a clear inconsistency between the language of section 4.5.4 and the first sentence of section 5.3.3. When the no action alternative is referenced under section 4.5.4, you properly observe that the no action alternative would have no adverse effects on wetlands. How then, under section 5.3.3, do you determine that fluctuating water levels under the no action alternative would continue to adversely affect dispersed areas of wetlands along the shorelines of project impoundments that experience significant water level fluctuations? Again, under your Relicensing Regulations, adverse impact must be determined by reference to a baseline. The proper baseline is current project operations, not pre-project historic conditions.

When compared to the proper baseline of current projectT&AS-122Conditions, the Applicant's mitigation proposal to provide 45acres of wetlands enhancement is perfectly adequate.

We also note that, in several locations of the DEIS, the acreage impacts to wetlands is variously estimated at 250 acres, 280 acres, and 300 acres. It would appear that, under your analysis, there is a floating target for wetlands mitigation.

We agree that given that the potential adverse impacts of higher flows to the beaver population which creates wetlands in the Back Channel, no greater flows in the Back Channel are justified from the perspective of avoiding wetlands impacts or

75

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

- T&AS-121 We have already responded to these comments in responses T&AS-53 through T&AS-56.
- T&AS-122 We do not find the inconsistency to which ASSSET refers. Current operations at the projects negatively affect approximately 250 acres of shoreline wetlands. Under the no action alterative, no additional wetlands would be affected. Any inconsistencies contained in the DEIS concerning the quantity of affected wetlands have been corrected in the FEIS. Your other comments are noted.

T&AS-120

T&AS-121

Cont'd

enhancing wetlands

Further, given the costs and scientific uncertainty of successful creation of wetlands, we believe that you should be more careful and consistent in addressing adverse impacts to the resource under FERC's relicensing regulations, which clearly require that current conditions be used as the baseline for assessing whether proposed operations will have any greater impact. Under the proper analytical framework, it is doubtful that greater flows in Millinocket Stream and the Upper Gorge would provide anything more than dubious wetland habitat and cannot be justified. We concur that some wetlands mitigation may be appropriate, but more in line with what the Applicant has agreed to under the water quality certificates as outlined in the summary of findings.

5.3.4 Land Use

We strongly disagree with your conclusions in this section.

The record in this proceeding demonstrates no adverse impacts to water quality or aesthetics from current operation. LURC rules are more than sufficient to protect water quality and aesthetics over the term of new licenses.

Of equal importance is the right of the people of the State of Maine to determine the appropriate standards for water quality and aesthetics. You should defer to Maine's right to make this judgement, as well as its judgement as expressed by LURC rules and regulations.

We expect that you will delete the requirement for expansion of project boundaries as factually unjustified on this record, an unwarranted intrusion on the right of Maine to govern water quality and aesthetics, and a drastic response to a problem that does not exist at this time and could either be dealt with when a problem develops or could be dealt with through less intrusive means.

Your estimate of the cost of project boundary expansion is woefully inadequate. The cost of expansion (\$13,068,000 to \$43,560,000) coupled with the impact on existing leaseholders, private landowners on Ripogenus, who would lose their lands and the State of Maine, which would lose its land, are sufficient independent reasons to delete the requirement for expansion of project boundaries.

The impoundments are not part of the wilderness. The impoundments are in the heart of GNP's working forest. You do a disservice to the people of Maine by trying to justify expansion of project boundaries as a means to preserve the wilderness character of the area for a very few tourists who come for a few

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-123 Comment noted. The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff's recommendations (see section 5.3.4) provide measures to protect shoreland resources within the project areas while considering existing land use regulatory controls.

T&AS-123

T#AS-122

Cont'd

T&AS-123weeks during a few summers out of their lives. The costs of
expansion of project boundaries will be borne by the Applicant,
local residents, and area businesses forever.

5.3.5 Economics and Socioeconomic Effects

Table 5-3 summarizes levelized annual net costs for the proposal. The table is not consistent with your discussion in other sections of the draft and needs to be revised.

With respect to the Applicant's proposal in Table 5-3, we note the following:

- (a) Stable flows below McKay. On page 2-31, you state stable flows cost 3,300 mwh a year or 3,300 mwh x \$83.22/mwh = \$274,626, rather than \$181,269 listed in Table 5-3;
- (b) Upper Gorge. On page 2-31 you state Upper Gorge flows cost 2,200 mwh a year or 2,200 mwh x \$83.22/mwh = \$183,084, rather than \$120,846 listed in Table 5-3;
- (c) Stable North Twin. On page 2-31 you state managing North Twin elevations will cost 5,500 mwh x \$83.22/mwh = \$457,710, rather than \$280,143 listed in Table 5-3;
- (d) Millinocket Stream. On page 2-31 you state Millinocket Stream flows cost 350 mwh x \$83.22/mwh = \$29,127, rather than \$19,226 listed in Table 5-3;
- (e) Hydro efficiency losses. On page 2-31 you state hydro efficiency losses cost 4,100 mwh x \$83.22 = \$341,202, rather than \$225,213 listed in Table 5-3;
- (f) Steam losses. On page 2-31 you state steam efficiency losses cost 4,000 mwh x \$83.22 = \$332,880, rather than \$219,720 listed \$219,720.

Therefore, losses for the Applicant's proposal total:

(a)	McKay Flows	\$ 274,626
(Ь)	Upper Gorge Flows	\$ 183,084
(c)	Holbrook	\$ 1,929
(đ)	North Twin	\$ 457,710
(e)	Wetlands - North Twin	\$ 4,476
(f)	Millinocket Stream	\$ 29,127
(g)	Hydro losses	\$ 341,202
(g)	Steam losses	<u>\$ 332,860</u>
5		\$1.625.034

The total losses based upon your figures on page 2-31 are over 50% higher than those listed on Table 5-3. We ask that you revise Table 5-3 to reflect the correct totals.

With respect to Alternative 1, and based on the figures you list on page 2-34 and our comments on Section 2.4.1.2 through 2.4.2.6 and 4.9, 4.10 and 4.12, the losses should be:

77

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-124 The revised annual net benefits are presented in Tables 2-8 and Table 5-2 of the FEIS. The annual losses in power that would occur under each alternative are presented in 2-4, and the annual costs of purchasing replacement power are presented in Table 2-5. The staff used a value of \$73.92/MWh as the cost of purchasing replacement power from the least-cost alternative. As a result, our estimates of the increases in GNP's power costs are slightly lower than those you calculated.

E-327

T&AS-124

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

		low	high
(a) (b) (c) (d)	McKay and Upper Gorge 10,855 mwh x \$83.22 Holbrook Wetlands Enhancements Conservation Easement	\$ 903,353 \$ 1,929 \$ 119,098 \$1,367,580	\$ 903,353 \$ 1,929 \$ 119,098 \$4,578,650
(e) (f)	North Twin Millinocket Stream 771 x \$83.22	\$ 457,710 \$ 64,163	\$ 457,710 \$ 64,163
(g) (h) (i)	Back Channel 20,800 x \$83.22 Hydro losses Sream losses	\$1,730, 9 76 \$ 341,202 <u>\$ 332,680</u>	\$1,730,976 \$ 341,202 <u>\$ 332,880</u>
Tot /		\$5,318,891	\$8,529,961

Table 5-3 should be revised to reflect the correct totals for this alternative.

With respect to Alternative 2, Final Recommendation (leakage in Back Channel), and based on your figures on page 2-34 and our comments on Sections 2.4.1.2 through 2.4.2.6 and 4.9, 4.10 and 4.12, the losses should be:

		low	high
(a)	McKay and Upper Gorge 9,287 mwh x \$83.22/mwh	\$ 772,864	\$ 772,864 \$ 65.094
(Ъ)	Wetlands	3 03,074	
(c)	Holbrook	S 1,929	\$ 1,949
(d)	Conservation Easement	\$1,367,580	\$4,578,650
(e)	North Twin	\$ 457,710	\$ 457,710
(£)	Millinocket Stream 388 mwh x \$63.22	\$ 32,289	\$ 32,289
(a)	Wetlands	\$ 4,436	\$ 4,436
120	Nydro losses	5 341.202	\$ 341,202
(i)	Steam losses	\$ 332.880	\$ 332,880
тоти	ALS:	\$3,375,984	\$5,814,190

Table 5-3 should be revised to reflect the correct totals for this alternative.

The only difference between Alternative 2 without Back Channel flows and with 165 cfs is 9,900 mwh x \$83.22 mwh = \$823,878, so totals for this alternative would be the same as previously stated plus the Back Channel loss:

T&AS-124 Cont'd E-328

E.		low	high
(a) McKay and Upper Gorge		
	9,287 mwh x \$83.22/mwh	\$ 772,864	\$ 772,864
I (b) Wetlands	\$ 65,094	\$ 65,094
(c) Holbrook	Š 1,929	\$ 1,929
1 (4) Conservation Easement	\$1,367,580	\$4,578,650
(e) North Twin	\$ 457,710	\$ 457,710
l (f) Millinocket Stream		
	388 mwh x \$83.22	\$ 32,289	\$ 32,289
) (a) Wetlands	\$ 4.436	\$ 4.436
1 (ň) Hydro losses	\$ 341.202	\$ 341,202
1 G) Steam losses	\$ 332,880	\$ 332,880
6) Back Channel	<u>\$ 823,878</u>	<u>\$ 823,878</u>
то	TALS:	\$4,199.862	\$6,638,068

T&AS-124 Cont'd

Table 5-3 should be revised to reflect the correct totals for this alternative.

Since your entire discussion is based upon the totals found in Table 5-3, the discussion needs to be redone to reflect correct costs for each proposal.

We agree with your reasoning but note that the actual cost to the Applicant and the local and regional economies are far greater than stated in the draft and result in the inescapable conclusion that the final recommendation must be revised by <u>deleting</u> the conservation easement and wetlands enhancement beyond the Applicant's Proposal. It is not needed in light of LURC rules and regulations and the cost is prohibitive.

Finally, we note that the predicted area job loss for Alternative 1 should be revised to reflect the revised MDOL multiplier of 3.2539, rather than 1.61. <u>See</u> Exhibit C to these comments.

5.4 Summary of Findings and Recommendations

With respect to your "reasons" listed on pages 5-16 and 5-17:

 (a) With the exception of your observation that some enhancement of flows in Millinocket Stream beyond those proposed by Great Northern should be required, we agree with Reason #1;

(b) Reason \$2, you should add that significant Back Channel flows would result in state water quality certificate violations, adversely impact salmon and lake trout, and threaten 2,000 cfs flows at Millinockt as well as downstream dilution flows (because of insufficient availability of water to sustain such flows); T&AS-125 Where appropriate, we have revised our reasons for recommending Alternative 2 and the economics on the projects based on comments from various parties on the DEIS and our revised economic analysis procedure.

T&AS-125

79

- (c) We agree with Reason #3;
- (d) We agree with Reason #4;
 (e) We disagree that wetlands mitigation (Reason #5) is
- necessary. See our comments under Sections 4.5 and 5.3.4;
- (f) We disagree that expansion of project boundaries (Reason #6) is necessary or justified given past landowner practice, LURC rules, and cost. See our comments at Sections 2.4, 4.9, 4.10, 4.11, 4.12, 5.3.4 and 5.3.5;
- (g) We agree with Reason #7;
- (ħ) We disagree with your conclusions (Reason #8) concerning Alternative 2. Wetland mitigation is not required and expansion of project boundaries is not necessary or cost effective. The difference in cost between Alternative 1 and your recommendation is negligible.
- (i) We disagree that the difference in cost (Reason #9) between Alternative 2 and Alternative 1 is significant; the cost of either is far higher than \$9.5 million.
- For the reasons stated in these comments concerning Section 2.4, Table 2-5, Table 2-7, Sections 4.9, 4.10, 4.12, 5.3.4, 5.3.5, Table 5-2 and Table 5-3, the figures listed in Table 5-5 are incorrect.

In our comments concerning Sections 2.4.1.2 and 2.4.1.3, we calculated a range for annualized project costs for your recommendation (Alternative 2, leakage in Back Channel) from \$1,504,587 to \$4,715,656 or from \$2.43 (mils/kwh) to \$7.60 (mils/kwh). Energy losses for this Alternative are discussed at Section 2.4.2.5 and are:

23,275 mwh x \$.08322/kwh x 1,000 kwh = \$1,936,955

or (\$1,936,955/620,400 mwh) x 1,000 mwh/kwh = \$3.12 mils/kwh.

Therefore, the total annualized cost of the alternative ranges from \$3,441,542 to \$6,652,611 or \$5.55 mils/kwh to \$10.72 mils/kwh.

5.5.1 Water Quality Certificate Conditions

As suggested in Section 2.3, the objectives of the state water quality certificates for both the Ripogenus and Penobscot Mills projects include (1) providing flows below McKay Station which support the West Branch fishery; and (2) maintenance of a North Twin drawdown to benefit lake trout spawning. Given these objectives and the fact that the state water quality certificates represent a minimum baseline, which cannot be reduced (provided the baseline conditions relate to water quality), we fail to see how either Alternative 1 or Alternative 2 (to the extent it requires 165 cfs of flow) can be considered reasonable alternatives. Both Alternative 1 and Alternative 2 (to the extent it suggests a flow in the Back Channel of 165 cfs) would

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-126 We disagree that WQC conditions would necessarily have to be violated to provide flows to the Back Channel. See Appendix D.

T&AS-125

T&AS-126

Cont'd

T&AS-126 Cont'd	violate the baseline objectives of the state water quality certifications as recited in section 5.5.1 and Tables 5-6 through and including 5-8 at pp. 5-18 through 5-20 of the DEIS.
	5.6 Section 10(J) Recommendations
	We agree with your positions on Interior's 10(J) recommendations 1, 2, 3, and 6. We specifically note that, in your statement of its position on recommendation number 6 of Interior's 10(J) recommendations, watershed protection is a matter best addressed on a comprehensive basis by state agencies under the statutory provisions cited by you at pp. 5-23 and 5-26 of the DEIS.
TEAS.127	CONGLUEICH
	We commend you for a thorough effort and generally careful analysis of a significant range of complex economic and environmental issues. Moreover, we are impressed that FERC will now look to complete the Environmental Impact Analysis promptly so that licenses can issue by early summer 1995.
m	However, there are several issues where we believe your analysis needs to be corrected by either minor adjustments or wholesale change, including the following:
T&AS-128	1. <u>Streamflows</u> . Your analysis of the Applicant's data concerning availability of water for Back Channel flows is inaccurate and flawed. The Applicant's data shows sufficient water is not available to sustain flows of either 350 cfs or 165 cfs, without violating water quality certifications issued by the Maine Land Use Regulation Commission and the Maine Department of Environmental Protection. These State certifications are your legally binding, baseline constraints on the Applicant's operation of the project (as you properly notes at Section 5.5.1).
	Notably, fisheries resources would also be jeopardized by such flows. Accordingly, while we agree with your conclusion that flows of either 165 cfs or 350 cfs should not be required, we assert that such a conclusion results not only from economic, but also from environmental requirements. We believe that your over-reliance on "average year" model simulations is misplaced and leads you to incorrect conclusions as to the availability of water for any significant Back Channel flows.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-127 Section 5.6 of the EIS has been revised subsequent to revised 10(j) recommendations from Interior following the DEIS and further discussions with Interior on those recommendations.

T&AS-128 We disagree. See responses T&AS-46 through T&AS-49.

.

81

You have correctly concluded that higher streamflows are in many cases not warranted, given the far greater fisheries habitat and recreational opportunities present in other sections of the river system. For the reasons detailed in these comments, we recommend that you adopt the following streamflows:

in Millinocket Stream, a minimum flow of 60 cfs

between May 1 and October 15, and 20 cfs during

T&AS-128 Cont'd

T&AS-129

T&AS-130

332

b. in the Back Channel, leakage and spillage flows;

the rest of the year;

- c. in the Upper Gorge, only leakage from fall through spring; otherwise, the same as those outlined under the LURC water quality certificate (as outlined in the DEIS); and
- d. at all other locations, the same as those under the Applicant's Proposal.
- 2. <u>Fisheries</u>. Your discussion of impacts on fisheries omits significant environmental impacts to fisheries by limiting the discussion to "average" years. The data demonstrates that you should (1) reconsider its conclusion that sufficient water is available for substantial Back Channel flows; and (2) discuss the fisheries impacts on West Branch salmon stocks below McKay Station and on establishment of a self-sustaining togue population. Your conclusion that flow enhancements would cause only minor changes in impoundment drawdowns may be accurate in average years, but it is certainly not accurate for drawdowns in worst case years.

3. <u>Hetlands</u>. In its Relicensing Regulations, FERC states its policy that "evaluation and consideration of the appropriateness of requiring enhancement measures is done in the context of today's environment and in relation to today's needs and problems, not in the context of the world as it existed 50 years ago." Similarly, in discussing water flows in the Back Channel, you specifically acknowledges that <u>existing</u> conditions provide the baseline for its environmental analysis. See section 4.3.1.3.

> However, in analyzing wetlands impacts, you go against FERC's stated policy and its analysis of other environmental issues and look to rectify "continuing adverse impacts." The language and analysis in the DEIS presupposes that <u>present</u> project operations are

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-129 We disagree. See response T&AS-56.

T&AS-130 Current operations at the projects negatively affect approximately 250 acres of shoreline wetlands. This analysis is based on existing conditions. We see no conflict in our policy with regard to this criterion.

COMMENTS FROM THE TOWN OF MILLINOCKET AND ASSSET

resulting in water level fluctuations that are now causing adverse impacts to wetlands, which would require mitigation even under the "no action" alternative.

You should reexamine your analysis of wetlands impacts and substantially reduce the mitigation required of the Applicant.

 Expansion of Project Boundaries. You should carefully reevaluate the wisdom, need and cost of expanding the project boundaries. Before you conclude that aesthetic impacts alone justify such radical surgery, we ask that you consider the following:

a. Deference to LUEC. Consistent with well established principles of federalism, you should give strong consideration and, as appropriate, defer to the State's regulation of lands in the unorganized territories, including those around the Ripogenus impoundment. Before imposing this federal mandate to expand project boundaries, you should first consult with the state agencies who currently administer the state's land management and water quality protection programs and reevaluate the need for further federal protection.

b. Land stewardship; unwilling sellers. The private and public landowners on Ripogenus impoundment are good stewards, have fought hard in getting and/or defending private and public landowner rights, and will not willingly give up those rights. There is a strong tradition of resource stewardship practice by the public and private landowners on the Ripogenus impoundment. As stated in the testimony of Larry Philbrick, the private landowner tradition on the Ripogenus impoundment spans five generations. State stewardship of lands in the area is also notable and includes Baxter State Park and, more specifically, Gero Island in the middle of Chesuncook Lake.

As Steve Adams (State Planning Office), Senator Leo Keiffer (Maine Senate Majority Leader), and Representative Richard Gould (Cochair of the Maine Legislature's Natural Resources Committee) stated at the public hearing on January 25, 1995, the State is not interested in parting with public lands. Mr. Philbrick, on behalf of the many private landowners on Ripogenus impoundment, similarly observed that the private

83

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

TS&A-131 See responses T&AS-11 through T&AS-16.

က ယ ယ **T&AS-13**'

T&AS-130

Cont'd

landowners on Ripogenus impoundment would not be willing sellers. Condemnation of such lands will therefore be required to implement an expansion of project boundaries on non-Applicant lands.

- C. Undetermined socioeconomic impacts. You have yet to assess the broad, and as yet undetermined, socioeconomic impacts of this measure (expanding project boundaries) on the majority of people who currently own or occupy land on the project impoundments and who would be negatively impacted by the proposed mandated land conservation requirements.
- d. Direct economic impacts/Great Northern's acquisition cost. The economic impact of the land acquisition mandate on Great Northern has been severely undervalued. In the DEIS, you estimates that the cost of acquisition alone is \$2 million, using a "price per acre" value of a "comparable sale." The \$2 million cost estimate which you assigns to the proposed land conservation proposal equates to \$4.59 per waterfront foot of land to be acquired. As U.S. District Court Judge Hornby stated in a recent federal court decision here in Maine, the proper valuation basis for waterfront land is price per waterfront foot, not price per acre. See Exhibit A.

As set forth in the consulting report of Lowell T. Sherwood, A.S.A. (Certified General Appraiser - ME #9), a copy of which is attached as Exhibit A to these comments, you derives that cost estimate from a voluntary "Rangely Lake" acquisition by the Rangely Lakes Heritage Trust. Indeed, this acquisition (which actually occurred on Mooselookmeguntic and Cupsuptic Lakes) assumes a price per waterfront foot of \$51.00 for a conservation easement and \$71.00 for fee title. Combining severance damages and other likely costs as part of an eminent domain process (because neither the State of Maine nor the larger landowners on the Ripogenus impoundment will voluntarily sell), the likely cost falls within a range between \$13,068,000 and \$43,560,000. This cost is far in excess of \$9.5 million, which the DEIS found unacceptable.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

T&AS-131

T&AS-131 m **Cont'd** ω ω

- e. Economic Impact/Great Northern's Mandatory Conservation of Its Own Lands. The economic impact of mandatory conservation on Great Northern is also far more significant than you suggest, in the DEIS. Indeed, the DEIS does not discuss the "lost opportunity" cost to Great Northern of subjecting its own land on Ripogenus impoundment to a conservation easement. As noted in the consulting report of Lowell T. Sherwood, that lost opportunity cost likely equates to the cost of having to acquire fee title or a conservation easement on lands it does not own. Whereas Great Northern would have to acquire 82.5 miles of shorefront from other landowners, it would also have to give up development rights on its 82 miles of shore frontage.
- f. Speculative Positive Impacts. As stated in our comments as to recreational issues, the extreme minority of the boating population on the impoundments engage in back country canceing, which the environmental intervenors in part assert as the basis for the mandatory land conservation.

We ask you to consider that the need, if any, does not (1) warrant an intrusion into the Maine Land Use Regulation Commission's traditional jurisdiction for regulating such shorelands, (2) warrant disrupting the longstanding private and state landowner tradition in the stewardship and ownership of such lands on Ripogenus impoundment, (3) justify the significant socioeconomic impacts (which you did not evaluate) that imposition of such a mandate would cause, and (4) justify the very substantial direct economic costs (which you substantially undervalued) that Great Northern would have to incur in order to acquire lands it does not own and to conserve its own lands.

5. Broader Consideration to Socioeconomic Impacts in the <u>Final EIS</u>. You have appropriately considered in the DEIS a range of socioeconomic impacts, particularly to those of the three towns in the immediate Katahdin region. Between 1980 and 1992, the population of the three area towns dropped by 7.5%, as compared to a statewide increase of 9.5% in the same period, representing a gap of 17.4%. The Great Northern mills have seen 53% work force reduction in the last decade (between 1985 and 1994), resulting in the loss of 1,977 well paid jobs. See Exhibit C to these comments. Great Northern is currently pursuing further cost reductions. Federal and state agencies have had to

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-131

T&AS-131 m**Cont'd** ່ ຜ ເມ

respond to the new economic challenges facing the Katahdin region by investing several million dollars to stabilize that region's economy.

However, beyond the impacts to the Katahdin region, we recommend that you give careful consideration to the socioeconomic impacts of these projects throughout central and northern Maine, and the State of Maine generally.

As your DEIS notes, the pulp and paper industry is an extremely competitive and energy intensive industry. Great Northern's presence in certain markets and its competitive position in general depends upon the availability of a reliable source of inexpensive electric power. Great Northern mills are a critical component of the Maine forest products economy. Statewide, paper accounts for 35% of the State's manufactured product and, with lumber and wood products, accounts for almost half of the State's manufactured product.

Accordingly, we urge you to maintain your position as to streamflows except where your recommendations depart from those listed above; as to the areas of departure, we ask you to reconsider your position and adopt <u>all</u> streamflows as recommended above.

For the reasons stated in these comments, we finally note that the proposed operations, in accord with Great Northern's Water Use Plan, will guarantee very substantial natural resource and societal benefits, including:

- 1. a world class landlocked salmon fishery;
- whitewater rafting on the West Branch, some of the best technical rafting on the East Coast;
- significant flood control;
- 4. a guaranteed flow of 2,000 cfs at Millinocket to provide instream river flows for wastewater treatment facilities up and down the Penobscot River; and
- 5. direct and indirect economic benefits (including highly paid jobs, production of high quality paper, purchase of forest products and other goods, both direct and indirect, in the Maine economy, and significant tax revenues to the Katahdin area towns).

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET

T&AS-131

COMMENTS FROM THE TOWN OF MILLINOCKET AND ASSSET

Again, we congratulate you on the fine work that you have done thus far in the Environmental Impact Statement. We urge you to consider the comments set forth above and promptly complete the Environmental Impact Analysis. We wish you well as you complete this analysis, and urge FERC to promptly issue new licenses.

T&AS-131 Cont'd

The future of this region turns on the opportunity and ability of Bowater to modernize these mills so that they can compete in today's global forest products industry.

RESPONSES TO THE TOWN OF MILLINOCKET AND ASSSET ON UPPER PENOBSCOT RIVER BASIN

UNITED STATES OF AMERICA BEFORE THE FEDERAL ENERGY REGULATORY COMMISSION

Project Nos. 2458 and 2572

<u>,</u> ```

Great Northern Paper Co., <u>et al.</u>) Penobscot Mills and Ripogenus)

PH 4:59

COMMENTS OF TROUT UNLIMITED REGARDING FERC/DEIS-0075

Pursuant to Federal Energy Regulatory Commission ("FERC" or "Commission") notices dated December 9, 1994 and February 2, 1995 in the above-referenced proceedings, Trout Unlimited ("TU") hereby files its comments concerning FERC's Draft Environmental Impact Statement ("DEIS") No. 75.

General Comments

DEIS No. 75 fails to give equal consideration to mitigation and restoration of fish and fish habitat or to alternatives other than that proposed by Great Northern Paper Co. ("GNP"). The DEIS does not even mention, let alone respond to, scientifically-based studies regarding fishery and recreational resources submitted by TU in these proceedings.¹ The DEIS fails to provide a sufficient record upon which the Commission can reach a reasonable decision regarding the best adaptation of these projects to a comprehensive plan for the West Branch. Additional review and analysis of alternatives and additional information regarding instream flows should be included in a final EIS.

The DEIS continues to propagate erroneous interpretations of the terms "no action" alternative and "baseline." "No action" properly means no issuance of a new license and, now that the Commission has determined in RM93-23 that it has

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

- TU-1 Staff gave equal consideration to fish and wildlife resources in the DEIS by analyzing a range of alternatives from no-action (relicensing the projects as they currently exist) to Alternative 1 (the Conservation-Intervenor oriented alternative, providing many additional environmental enhancements beyond those proposed by GNP). We cited TU's letter to which the Vadas study was attached; our discussion of it was included only in Appendix D. We did not consider the Vadas study and TU's recommendations which are based on it, as reasonable alternatives to be evaluated for the following reasons: 1) the study does not take into account the state's recommendations, as codified in the 401 WQC, and would render the other goals of the state agencies and the Water Use Plan unobtainable; 2) the study does not take into account the economic penalty to GNP: 3) the study does not take into account loss of flood control benefits under the flow regime suggested by the study.
- TU-2 FERC has not changed its definition of baseline and no-action. Baseline conditions continue to be existing conditions, not preproject conditions. No-action refers to continued project operation under the existing license.

TU-1

¹See e.g., "Evaluation of Instream Flow Needs for Fisheries and Recreational Resources in the lower West Branch of the Penobscot River," submitted with TU's comments of May 24, 1993

- 2 -

the authority not to issue a new license, the Commission should correct this mistake. "Baseline" property means pre-project, unregulated, natural riverine conditions. The failure of the Commission to consider basinwide impacts, decommissioning, and full restoration alternatives necessarily skews its analysis of all other alternatives.

As to proposed "buffer zones," while TU agrees that activities on adjacent lands can adversely affect riverine conditions, TU is not willing to "trade" land use restrictions for biologically-sound flows, as is clearly the intent of the DEIS. FERC has a duty to balance energy and non-energy uses of *rivers*. The achievement and benefit of land use restriction, above and beyond current LURC set-back requirements, *in this case* is not certain, while the need for flows other than those recommended in the DEIS is clearly defined and documented. There is little benefit to land use restrictions if they merely provide access to or visual enhancement of uplands adjacent to a dewatered riverbed or lake bottom.

Specific Comments

On page i, it is noted that, "This DEIS [was] prepared by Commission Staff in connection with *an application* filed by GNP for FERC Projects Nos. 2572 and 2458 ..." (emphasis added). *This is incorrect*. GNP filed two, separate license applications. This issue goes to balancing and determining appropriate environmental protection, miligation, and enhancements within each license.² In this case, while GNP is offering some enhancements at Ripogenus, there is very little consideration of environmental concerns at Penobscot Mills. The failure to provide reasonable mitigation at Penobscot Mills will have a

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

- TU-3 Opinion noted. FERC staff did not recommend land use restrictions in exchange for flow restrictions. These decisions were made independently but within the context of FERC's balancing requirements.
- TU-4 We have corrected the EIS to indicate that it covers 2 applications from GNP. Your other comments are noted.

TU-3

E-340

TU-2

Cont

TU-4

²FPA, 16 USC §803(a).

- 3 -

deleterious effect on the West Branch as a whole.

On page ii, it is noted that, pursuant to the FPA, hydropower projects that are licensed by FERC must "be best adapted to a comprehensive plan for improving or developing a waterway ..." for purposes of commerce, power, "adequate protection, mitigation, and enhancement of fish and wildlife " and other beneficial uses (including irrigation, flood control, water supply, recreation and other 4(e) purposes). In its unopposed motions to intervene in opposition to licensing, dated August 22, 1993, and again in its comments of May 24, 1993. TU requested that the Commission prepare an Environmental Impact Statement for the entire West Branch of the Penobscot River to assess the basin-wide impacts of operations at this project and related projects. GNP's upriver storage projects regulate flows throughout the West Branch.³ The Commission must include in its assessment at least GNP's upriver storage projects in order to comply with the FPA's directive regarding "a single comprehensive plan,"4 to be able to determine the cumulative impacts of hydropower generation on brook trout and landlocked Atlantic salmon, and to have a sufficient record upon which a reasonable decision can be based regarding appropriate protection, restoration, and enhancement of fish and fish habitat. At Section 4.1, the Scoping Document stated that the "river reach to be considered for cumulative impact evaluation includes the entire West Branch of the Penobscot River;" the DEIS fails to meet this goal.

· On page xiii, it is stated that "[s]everal intervenors seek enhancements beyond

³E.g., GNP Storage Project No. 2634, which includes several dams and an impoundment, and a number of other unicensed storage projects owned and operated by GNP ⁴Section 10(a).

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

TU-5 The upper storage projects are included in GNP's water use model and the alternatives evaluated with that model, including additional model runs requested of GNP by FERC staff. These projects are not up for relicensing now but these projects are included in the water use model as a combined input. This input was kept constant in developing the water use plan although the model could be used to vary the input. We saw no need to do this however, since no one has suggested alternative management schemes for the upstream projects that would provide additional downstream benefits. In addition, GNP has agreed to inclusion of a reopener to allow modification of the water use plan, should analysis of the upper projects during relicensing result in unanticipated findings that desired changes could be made. We will recommend that the orders for both the Ripogenus and Penobscot Mills Projects include an article containing a reopener clause for consideration of the upper storage projects.

TU-6 Comment noted. FERC considers baseline to be existing conditions; hence environmental benefits provided beyond existing conditions are considered enhancements and not mitigation.

TU-5

- 4 -

TU-6 those proposed by GNP to improve existing fisheries. recreational opportunities. Cont and other environmental conditions." (emphasis added). This is incorrect. TU is seeking mitigation and restoration of fisheries and fish habitat in Penobscot Mills' "Back Channel" and Millinocket Stream and throughout the West Branch.

> Currently, federal and state agencies are actively involved in Atlantic salmon restoration efforts for the Lower Penobscot. In addition, the U.S. Fish and Wildlife Service ("USFWS") is contemplating a petition to have Atlantic salmon listed as an endangered species. The final EIS should allow for future restoration efforts and examine current agency efforts in light of the potential need to expand available habitat to approximate the historic range of Penobscot River salmon.

> The DEIS focuses on the needs of species for which the Maine Department of Inland Fisheries and Wildlife ("MDIFW") is managing the West Branch to the detriment of other species. The DEIS fails to balance the needs of successful, non-native species (*e.g.*, landlocked salmon) with the needs of declining native species (*e.g.*, brook trout). All of the Instream Flow Incremental Methodology studies ("IFIM") that are referred to in the DEIS (*e.g.*, Millinocket Stream and the Back Channel) refer to flow needs of landlocked salmon, not brook trout. By contrast, TU's instream flow analysis, submitted with its recommended terms and conditions, comprehensively analyzed habitat needs of native and nonnative beneficial species.⁵ The final EIS should address the need for biodiversity, protection and restoration of native species, and the benefits of restoring natural conditions to a river basin.

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

- TU-7 Comments noted. These are not reasonably foreseeable actions that could be considered in this EIS. We note, however, that Interior has reserved the right to prescribe fishways under section 18 of the FPA and this reservation will be included in the orders for the projects.
- TU-8 State and federal resource agencies had the opportunity during initial stage consultation to recommend species to be considered. The state and interior chose not to consider brook trout in the Back Channel. Restoration of native species was not a goal of the state so the FEIS does not consider it.

TU-7

E-342

TU-8

⁵See Vadas study, as referenced above

- 5 -

Ripogenus -- Impoundment, Upper Gorge, and below McKay Station:
 As pointed out above, the DEIS fails to consider the impacts of hydropower operations on native species such as brook trout. Drawdowns of the Ripogenus impoundment should be limited in order to foster the restoration of: (1) brook trout in the impoundment; (2) associated wetland regions; and, (3) general biodiversity throughout the project area.⁶

- Given the position of USFWS and the Penobscot Indian Nation ("PIN") regarding flows in the "Upper Gorge" of the Ripogenus Project, *i.e.*, that they will forego recommendations to provide higher-than proposed flows in exchange for other mitigation associated with Holbrook Channel, TU will also support the arrangement between MDIFW and GNP with regard to flows in this reach. (see page 4-15).
 - The DEIS fails to mention the above-cited study regarding instream flow needs for fish and recreation.⁷ As TU has asserted a number of times in this proceeding, without additional instream flow and habitat studies below McKay Station, the Commission does not have sufficient information upon which to base a decision that allows for flow levels other than those recommended by USFWS. Without the information provided by an IFIM and a HEP study, FERC must deter to USFWS recommendations in the interests of fish and fish habitat. Either GNP must conduct these studies below McKay Station or FERC must provide minimum flows as recommended by USFWS and supported by the data collected by GNP, *i.e.*, 1,422 cfs from October 15 to June 7 and 711 cfs at all other times or inflow, whichever is less. As pointed out in earlier TU comments.

See Vadas sludy, as referenced above.
 See Vadas sludy, as referenced above.

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

- TU-9 As we stated in Appendix D of the DEIS, limiting drawdowns of Ripogenus is not a reasonable alternative, for at least 2 reasons: 1) downstream flow needs could not be met; and 2) some flood control benefits would be lost, depending on how much of a limit was imposed.
- TU-10 No response required.
- TU-11 See TU-1 above. In addition, approximately the minimum flows recommended by FWS will be provided, except during unplanned outages when 400 cfs would be provided for no more than 3 days. At other times, the minimum would be 1422 or 1000 cfs or inflow, at times when FWS recommends 1422 or 711 cfs or inflow. We discuss flows below McKay Station more fully in section 4.2.

TU-11

TU-9

TU-10

E-343

- 6 -

 TU-11
 because GNP studied only the needs of one species (landlocked salmon) and

 TU-11
 one form of recreation (whitewater boating), its calculations for summer flows

 Cont
 are too high and for spring flows are too low. Flow criteria could be corrected by

 including the needs of trout, wetlands, recreational fishing, other types of

 boating, and general biodiversity.

 The DEIS fails to provide for safe ramping rates below McKay Station. GNP has failed to conduct ramping studies. TU has requested a number of times in these proceedings that GNP conduct ramping studies and that safe ramping rates be required. Safety of anglers and other river users should be a major concern of the Commission in determining project operations.

E-344

TU-12

TU-13

Penobscot Mills -- Millinocket Stream and Back Channel:
 The DEIS frequently repeats that there are few/no fish in these areas, that small populations do not warrant expensive measures, and that there is no recreational or navigational demand for such fisheries. This is patently absurd. These conditions exist because GNP has destroyed the fisheries within the Penobscot Mills project area through substantial manipulations of flow for the purpose of hydropower generation. To now say that GNP has been so successful in wreaking havoc in the project area that FERC will not consider mitigation or restoration ignores the Commission's mandate that relicensing examine preproject conditions.

TU-14

As noted, Maine waived jurisdiction for the Penobscot Mills' Back Channel in its 401 certification and there are questions as to the effect of such a waiver.⁶

⁸ DEIS at page 4-13.

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

- TU-12 We do not believe that ramping rates below McKay Station are necessary. GNP has instituted a public warning system consisting of posted notices and a siren that is sounded prior to the release of flows from Ripogenus dam. During the whitewater boating season, a recreation attendant registers and controls access through the McKay Station gate and warns anglers of impending releases.
- TU-13 The DEIS stated that there would be few fish produced in these areas even with additional flows, relative to the costs due to lost power and within the context of fisheries within the region. FERC is not mandated to examine preproject conditions but does consider enhancements of existing resources within the context of balancing power and nonpower uses of the project area.
- TU-14 The legal status of the Back Channel relative to Maine's 401 WQC will be addressed in the order for the project.

- 7 -

TU-14 Cont

Since the issuance of its certification for Penobscot Mills. Maine's Department of Environmental Protection has found that neither the Back Channel nor Millinocket Stream meet state water quality standards.⁹ FERC must report current water quality conditions and, since the state waived jurisdiction over the Back Channel and finds that current flows in Millinocket Stream are inadequate, in compliance with *Jefferson County v. Washington*,¹⁰ should require hydropower operations that provide flows which attain state water quality standards.

 The DEIS also claims that there is "no apparent demand for navigation in" the Back Channel.¹¹ This ignores PIN comments, as supported by TU, USFWS, and other recreational and environmental groups. The DEIS should reevaluate flows needed for navigation in the Back Channel.

In its assessment of regional recreation, the DEIS purports that one of the reasons that it does not recommend flows other than those proposed by the applicant is that there is no recreational demand for fishing in Millinocket Stream or the Back Channel.¹² First, this is incorrect. Both TU and PIN have expressed the need for additional, safe fishing areas. Second, this is not logical. Fishermen do not show up at a dried out reach, waders and rod in hand, demanding that fish and water be provided; fishermen go where there are

¹⁰ U.S. Supreme Court Case No. 92-1911, decision issued May 31, 1994. In other DEIS's, e.g., the DEIS for the Lower Penobscot River, FERC has "deleted" some conditions of state water quality certificates as being "unlawful" in that they are "beyond the scope intended by *Jetterson Co.*" As a corollary, in the case at hand, FERC is responsible for setting minimum flows that will ensure that state water quality standards are attained where state water quality certifications have failed to do so. The intent of *Jetterson Co.* is that states may be more stringent than a national standard in implementing state standards and anti-degradation policies, but not less so. Syllabus at p.17.

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

- TU-15 The Back Channel would only provide an approximately 4.5-mile-long boating trip in a less aesthetically pleasing setting than along the West Branch. We state in Section 4.8.1.1 that significant opportunities exist within the project area to meet recreational boating demand, and that the significant flows that would be required to enable recreational boating activities would jeopardize other environmental enhancements.
- TU-16 We conclude that high-quality recreational fishing waters are abundant in the project region. We also conclude that increasing flows in Millinocket Stream and Back Channel would not result in any significant benefits to recreational fishery resources. Therefore, we do not support the provision of increased flows in these areas.

TU-15

⁹¹⁹⁹⁴ Statewide Water Quality Assessment.

¹² See e.g. page xvii.

- 8 -

fish and where they have access to safe fishing.

Because MDIFW has reversed its earlier position regarding Millinocket Stream (i.e., that it could be managed for salmon), the DEIS has dismissed the need for adequate flows (see page 5-7). As noted, MDIFW does not manage the stream for brook trout. Brook trout can flourish with lower flows than salmon. The IFIM for Millinocket Stream did not address brook trout needs. Therefore, TU maintains its request for USFWS recommended flow levels in the Stream, *i.e.*, 120 cfs from October 15 to June 7 and 60 cfs at all other times or inflow, whichever is less. FERC has no information in the record upon which it could base a decision to allow flow levels other than those recommended by USFWS.

TU-17

E-346

- As stated in earlier comments, landlocked salmon in Millinocket Lake seeking to drop into Millinocket Stream have no reasonable avenue of passage. TU reiterates its request for up- and down-stream fish passage at Millinocket Lake. Such facilities address the DEIS's concerns about limited habitat for adults. Furthermore, neither MDIFW nor USFWS agree with the DEIS's theory that increased flows in the stream and fish passage to and from the lake will lead to competition from downstream warmwater species.¹³
- The DEIS fails to require that GNP install downstream passage facilities for trout and salmon at Penobscot Mills' North Twin, Dolby, and East Millinocket Dam.
 Currently, coldwater lish must pass through turbines at those facilities, where fish mortality is high.

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

- TU-17 Based on the IFIM study for Millinocket Stream, which included brook trout, a year-round flow of 60 to 80 cfs would provide optimal habitat for this species. Following a 10(j) meeting with Interior, we recommend a year-round flow of 60 cfs or inflow, whichever is less, for Millinocket Stream.
- TU-18 FWS and DIFW did not express disagreement in their respective comment letters on the DEIS. We continue to believe that fish passage at Millinocket Lake Dam is unnecessary to sustain the Millinocket Lake or Stream fish populations and that the absence of fish passage facilities could prevent the undesirable spread of non-indigenous warmwater species into Millinocket Lake from downstream project waters.
- TU-19 The only fish entrainment within project waters for which there is data available, results in what has been termed "smelt drift". This phenomenon appears to be advantageous to piscivorous species for which smelt form a forage base while at the same time, not adversely affecting the smelt populations. Entrainment-related mortality of trout or salmon has not been raised as an issue in this proceeding by FWS or DIFW. We further believe that the life history patterns exhibited by project salmonid populations do not necessitate fish passage facilities at these dams. Therefore, we will not recommend that fish passage facilities be constructed at this time.

TU-19

¹³ DEIS at page 4-28 CI, personal communications/telephone conversations with MDIWF and USFWS. MDIFW has not expenenced this situation in other circumstances, e.g., Grand Lake.

- 9 -

The DEIS fails to require that drawdowns of the associated impoundments be limited. Excessive drawdowns have contributed to stress on lake trout, severely limiting populations throughout the Penobscot Mills project, and may be contributing to mercury contamination of fish and wildlife. High levels of mercury have been verified in bald eagles that nest within the Penobscot Mills project boundaries. Since bald eagles feed on fish, it is implicit that the fish within the Penobscot Mills boundaries are also contaminated with mercury.¹⁴ Before a new license is issued for Penobscot Mills, the Commission should explore the relationship between the operation of Penobscot Mills, particularly drawdowns of the impoundments, and mercury levels in resident fish. The Commission should adopt the recommendation of USFWS that Penobscot Mills be operated in a run-of-river mode in order to minimize impoundment fluctuations.

Respectfully submitted,

TROUT UNLIMITED

B١

Mona M. Janopaul Conservation Counsel Trout Unlimited 1500 Wilson Blvd., Suite 310 Artington, VA 22209 (703)522-0200

February 22, 1995

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

TU-20 GNP, the state of Maine (during the 401 WQC proceedings) and FERC staff considered drawdown limits at the major impoundments. As explained above (TU-9), Ripogenus drawdown limits are not reasonable for a number of reasons. North Twin water levels are proposed for management of lake trout and the state's 401 WQC requires such management.

> There is no evidence that annual drawdowns are contributing to mercury mobilization within the project area. Nevertheless, the state's 401 WQC contains conditions for both projects for further study of this issue and the orders for these projects will also require these studies or cooperation by GNP in studies conducted by other agencies.

¹⁴In the water quality certification that it issued for Penobscot Mills, the Mane Department of Environmental Protection required that GNP will conduct studies of mercury contamination in bald eagles, fish, and other wildlife.

RESPONSES TO TROUT UNLIMITED ON UPPER PENOBSCOT RIVER BASIN DEIS

CERTIFICATE OF SERVICE

I hereby certily that I have served on this 22nd day of February, 1995, by first TU-20 Cont

class mail, a copy of the foregoing Petition upon each person designated in the

Secretary's service list for this proceeding.

Mona M. Janopau

COMMENTS FROM SUSAN DOLAN ON UPPER PENOBSCOT RIVER BASIN DEIS

Susan Mackin Dolan 139 Bates St. Millinocket. Me 04462 January 31, 1995 Lois B. Cashell, Secretary Federal Energy Regulatory 825 N. Capital Street, N.E. Washington, D.C. 20426 Dear Ms. Cashell:

It has come to my attention that FERC has proposed new building setback guidelines as part of its environmental impact statements, in connection with the relicensing application submitted by the Great Northern Paper Company, (Ripogenus Hydroelectric Project, Ferc # 2572 – 00 and Penobscot Mills Hydroelectric Project, Ferc # 2458).– 009

I have leased land from Great Northern Paper for many years on North Twin Lake and have a small cottage on this land that my family returns to every summer. My parents and four other aunts and uncles also own cottages within two miles of mine, that were all built in accordance with the state of Maine's Land Use Regulatory Commission, LURC, in the 1970's and 1980's. We are concerned that our cottages may be within the proposed 200' setback and would like to know exactly how these new proposed restrictions would impact the existing buildings. As we have many years and a lot of time and money invested in our properties, we would like assurances that they would be protected from any new restrictions. We would also like to know if Great Northern Paper will be allowed to, (if they so decide), sell this land to the current leaseholders under the terms of the new licenses. This seems to be happening in other leased land recreation areas in the state.

This area is very special to all of us who live here and we work hard to mantain its unique beauty while enjoying the lakes and forests. One thing that greatly disturbs us when we return every year is the amount of clear cutting the paper company is doing so close to the shores of these lakes. This has a much greater impact on the environment than the few cottages scattered around the area. What restrictions will Ferc be putting on those activities? Your help is answering these questions would be greatly appreciated.

Thank you for your time and consideration.

Sincerely. Swan Maden Dran Susan Mackin Dolan

RESPONSES TO SUSAN DOLAN ON UPPER PENOBSCOT RIVER BASIN DEIS

The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period. GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff proposes, for the Ripogenus project area, two options: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion on GNP-owned lands. For the Penobscot Mills project area. the staff recommends a 200 foot expansion of the project boundaries on GNP owned lands. The staff recommends that existing structures would be grandfathered under any of the project boundary expansion alternatives. See section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.

2. See sections 3.11.1.1 and 4.9.

1.

1.



I am Dr. John Frachella, West Branch Regional Representative of the American Whitewater Affiliation (AWA). I am also a noncommercial, recreational boater and a camp owner within the 200 foot setback zone in the Ripogenus impoundment area.

AWA strongly supports the concept of recreational whitewater releases in the Upper Gorge. Alternative 2 calls for two scheduled recreational releases of 1,000 to 2,000 cubic feet per second, restricted to May, to avoid conflicts with fishery goals. However, <u>after</u> the license has been issued, we would like to be able to examine more closely the volume necessary for whitewater recreation in this stretch of the river. Those of us who have paddled this stretch in the past have reason to believe that lower flows <u>may</u> provide suitable runs. Therefore, we would like to design and implement a <u>post-license</u> recreational whitewater study. At this time we want to be reassured that less water per run <u>may</u> translate into more releases.

We are also requesting that <u>evening flows</u> out of McKay Station from May 1 to September 15 be reduced from commercial levels for private use thus preserving the natural river fluctuations found in nature and providing flows more comfortable for less experienced paddlers and for those looking for a diversity of experience.

We support the 200 foot setback buffer but <u>only</u> if the present existing lots on the shore frontage of the impoundments are grandfathered. We support FERC pursuing <u>willing</u> sellers only. We do not support the condemnation of lands through eminent domain.

> John (Frachella, DMQ John C. Frachella, DMD

RESPONSES TO JOHN FRACHELLA ON UPPER PENOBSCOT RIVER BASIN DEIS

- Opinion noted. Staff recommends that GNP consult with whitewater interest groups regarding appropriate flows for the scheduled releases in the West Branch during two weekends in May (see section 4.8.3.1).
- As stated in section 4.8.1, GNP would provide minimum flows from McKay Station during daytime hours. GNP usually maintains flows at or above 1800 cfs throughout the year, however, evening flows may be reduced or varied. We agree that the reduced or varied evening flows would provide additional opportunities for less experienced paddlers or those desiring a diversified experience.
- The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff proposes, for the Ripogenus project area, two options: (1) accepting the conservation easement proposed by GNP and the State of Maine: or (2) a 200-foot boundary expansion on GNP-owned lands. For the Penobscot Mills project area, the staff recommends a 200 foot expansion of the project boundaries on GNP owned lands. The staff recommends that existing structures would be grandfathered under any of the project boundary expansion alternatives. See section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.

1.

COMMENTS FROM CHARLES GADZIK ON UPPER PENOBSCOT RIVER BASIN DEIS

RESPONSES TO CHARLES GADZIK ON UPPER PENOBSCOT RIVER BASIN DEIS

OFFICE OFFILED 95 FEB -9 AN 10: 03 TO: Ed Meyer Federal Energy Regulatory Commission Office of Hydropower Licensing 825 North Capitol Street, N.E. Washington, DC 20426

٢. •...

FROM: Charles Gadzik RR#1, Box 947 ECOLO3101 51 Carroll Plantation, Maine 0448

RE: Comments on Draft Environmental impact Statement, Penobscot River Basin FERC No. 2527 & No. 2458 - 004

Members and Staff of the Federal Energy Regulatory Commission:

My name is Charles Gadzik, I am a resident of Carroll Plantation (about 40 miles east of Millinocket) Maine. My interest in this relicensing process is as a resident of the region for nearly 20 years, and in my participation in land use issues of northern Maine. During the years of 1988 through early 1994 I served as a member of Maine's Land Use Regulation Commission. For 26 months of my service I was Commission Chairman, as one of my responsibilities, I presided over 401 Water Quality Certification for the Ripogenus Project. I am here tonight representing myself.

In preparation of this hearing tonight, I've tried to recall all the details of LURC's 401 process two years ago. 1 remember as we began the review we were inundated with intervener applications which resulted in endless briefs on the subjects of mercury, access, wetlands, flows recreation and fisheries

As our review progressed, the question I asked myself as a veteran regulator was - Where are the substantive concerns? As each issued was raised for both relevance and then substance, the frequent conclusion for myself and colleagues was that there was no real conflict with public needs. Over time the picture emerged of a project that, due to its management for base load power, was being managed very closely to ideal for the principal public concerns of fisheries and water-use recreation. The truth, of course, is that both No response required.

1.

COMMENTS FROM CHARLES GADZIK ON UPPER PENOBSCOT RIVER BASIN DEIS

the world class land-locked salmon fishery and the white water rafting use exist because of the dams and how they are managed (There was the potential for fine tuning the optimization of all uses, and under FERC's guidance and GNP's discussion with interest groups these tunings are part of a new license). Yet despite this reality, some of the voices we heard were insistent on being sure that a sufficient pound of flesh was extracted from the applicant. These voices moved on to your proceedings with the same demands. It is with this background I ask you to reconsider two provisions in the Alternative 2 of the draft EIS.

1. On page 4-57, section 4.9.1.1, Shoreline development, there is discussion of the rationale for requiring GNP to acquire a shoreland easement on Chesuncook lake. As a veteran of LURC and its process I find this discussion surprising and disappointing. As is discussed on page 3-45. LURC has gone through a very extensive review of the lake resources in northern Maine. Lakes were classified by their resources and zoned in a manner that protected the highest value shoreline from any or excessive development. Much of Chesuncook's GNP frontage has the P-AL zoning which limits structures to one per mile of frontage. With this knowledge, it is very difficult to understand FERC's statement "...they still allow for a potential increase in development of more than 1,000 percent in the Ripogenus Project...". It is inconceivable to me that LURC will permit 700 dwelling units in the Ripogenus Project area. To accept this conclusion two things completely inconsistent with practices of the past 24 years would have to happen; 1) the landowners in the project area would have to seek every possible shorefront lot theoretically possible, and 2) LURC would have to blindly approve a massive rezoning for this shoreline. Neither of these actions are real world possibilities. (I noticed on page 4-57, that Mr. Sosland has suggested in his letter of September 3, 1993 that LURC is prone to approve the majority of rezoning applications it receives. The valid statistics should include the large number of applications that are pulled when LURC staff advise an applicant it does not have a chance for approval by the Commission.)

I believe that LURC's 24 year history is strong evidence that extensive development in the Ripogenus Project are is not a real threat. This history is not one of capitulation before landowners wishing to maximize development. I believe the representations of CLF, American Rivers Association, Appalachian Mountain Club, and Maine Audubon, are either out of ignorance or purposeful misrepresentation for ends that are not justified nor appropriate.

RESPONSES TO CHARLES GADZIK ON UPPER PENOBSCOT RIVER BASIN DEIS

- Comment noted. The staff revised potential build-out estimates (see section 4.9).
- 3. Opinion noted.

2.

COMMENTS FROM CHARLES GADZIK ON UPPER PENOBSCOT RIVER BASIN DEIS

Concerning development in the Penobscot Mills Project area; this area is regarded as already hosting significant development, is adjacent to the community of Millinocket, and in the broad perspective of all the water bodies in the region, a suitable place for some additional development. This is a rational position for land use planning in this region. However, to suggest that the existing 430 units can expand to 5,000 is again totally inconceivable. Nothing in LURC's history or in the history of this region suggest this course is a possible one.

t urge you to allow that land use planning and permitting stay in the capable hands of LURC, an agency with a long track record of success.

II. Page 4-58, 4.9.1.2 Timber harvesting Practices.

The impacts of timber harvesting on water quality is an issue important to Maine, and one that has been the attention of much research and technical review. An accurate representation of the current LURC status is that the zone for buffering is 250 feet from the high water mark. It has long been recognized that in order to maintain a fully functioning buffer zone, it must be maintained in a healthy condition. To this end, timber harvesting can fulfill both economic goals of the landowner as well as a healthy vegetative filter strip. The current standards allowing some occasional removal of trees from this zone is fully consistent with water quality objectives. The assertion by CLF, American Rivers Association, Appalachian Mountain Club, and Maine Audubon Society that water quality and wildlife objectives are best accomplished without any timber harvesting is completely inconsistent with the conclusions of many technical forums discussing such issues.

Finally, I want to thank FERC for negotiating through many complex issues and recognizing the quality of GNP's application for recertification. I believe there are few examples in this country of such a fine balance between the production of power and the maintaining of many outstanding public values. I hope my views can figure into your deliberations.

RESPONSES TO CHARLES GADZIK ON UPPER PENOBSCOT RIVER BASIN DEIS

- 4. See response #2.
- 5. Opinion noted. See revised sections 3.11.1.1 and 4.9).
- 6. No response required.

5.

6.

RESPONSES TO JIM HASKELL ON UPPER PENOBSCOT RIVER BASIN DEIS

SC FI

1. Opinion noted.

PUBLIC HEARING TESTIMONY: EIS FERC PHOJECTS 1058 2572 AND 2458 January 25, 1995 Stearns High School Millingerent. Maine

-2 ANII: 67 Ecolor -

My name is Jim Haskell.

Although I am the Town Planner, Code Enforcement Officer & Community Development Director of the Town of Millinocket, I am testifying tonight as a private citizen.

I have been involved in Community, Regional, Environmental and Land Development Planning in Maine for more than 26 years.

I hold a Masters Degree in Landscape Architecture from Harvard University. My Maine Professional Registration Number is 27, which I am told accounts for the gray in my beard. I am also a State Certified Code Enforcement Officer, certified in the areas of Zoning, Land Use & Shoreland Zoning.

I did my graduate thesis on Wildland Planning Issues in Maine and was the first Executive Director of the Maine Land Use Regulation Commission (LURC), I helped write the LURC law.

I would like to make the following points:

The First Point: I do not envy the job before you.

The Congress gave you no easy task when it required your agency to comply with the National Environmental Policy Act (NEPA) and the Regulations of the Council on Environmental Quality. (CEQ)

The Second Point: I realize that you can not possibly satisfy everyone.

The record is clear...in this situation...that the environmental intervenors involved will never be satisfied until all or most of the North Maine Woods are taken from private hands and made into public domain.

In recent years we have been hammered with one confiscatory proposal after another:

 the 10.5 Million acre of <u>North Woods Conservation Area</u> proposed to LURC by the Natural Resources Council of Maine and the Maine Audubon Society.

1

PUBLIC HEARING TESTIMONY: EIS FERC PROJECTS NOS 2572 AND 2458. January 25, 1995 Stearns High School Millinocket, Maine

 the 3.2 Million acre <u>Maine Woods National Park</u> proposed by the Massachusetts based environmental group...Restore.

1. Cont

- the 2.6 Million acre <u>Maine Woods Preserve</u> proposed in the May-June 1994 edition of the National Audubon Society Magazine.
- the 2.8 Million acres of <u>Conservation Priority Areas</u> proposed by the Maine Audubon Society.
- and now the intervenor's "Land for Dams" scam.

<u>The Third Point:</u>! realize the political realities of the power environmental groups have over federal agencies, through NEPA.

In spite of the fact that the current intervenors, collectively represent less than 1 percent of America's (94 million) households... They are allowed to push their own self-serving agendas as if they were speaking for the majority of the public.

Conservation Law Foundation Boston, Messachusetts	5,000 members
Appelechien Mountain Club Boston, Messachusetts	65,000 members
American Filvers Washington, DC	15,000 members
American Whitewater Affiliation Phoenicia, NY	(unknown)
Trout Unlimited Arlington, Virginia	75,000 members
Maine Audubon Society Felmouth, Maine	7,000 members

2

RESPONSES TO JIM HASKELL ON UPPER PENOBSCOT RIVER BASIN DEIS

1. Cont

2.

Opinion noted.

PUBLIC HEARING TESTIMONY: EIS FERC PROJECTS NOS 2572 AND 2458 January 25, 1995 Stearns High School Millinocket, Maine

2. Cont

3.

You know and I know that these self-appointed, un-elected, Environmental Lawyers and Lobbyists do not represent the public interest and really Could care less about the public Cont

> They are simply a well organized elitist minority...Funded by public tax credits...With their own well-documented agendas. (See attachment A and B)

> Their selfishness knows no bounds. Consider the testimony of the representative of the American Whitewater Affiliation, in which he clearly asked FERC not to adversely affect his existing camp on the Ripogenus Impoundment, but to apply the proposed restrictions to everyone else.

> The Fourth Point:....Regarding the building setbacks and vegetative buffers and forced acquisition of private property being proposed in the draft EIS:

- Α. It is my opinion that the rationale put forth in the Draft EIS for these proposed drastic actions:
 - is not based on. "accurate scientific analysis" or "high quality Ó information As required by NEPA Regulations in 40 CFR 1500.1(b) and CEQ Regulations 40 CFR 1502.24.

The US Forest Service's 1990 "Northern Forest Land Study", used as the basis for projected land use development (pages 3-48 & 3-49), is a four State study (ME,NH,VT & NY) of generalized comments regarding the 26 million acre Northern Forest and is without any specificity to the project area. This study is clearly not a sufficient fact basis to support the proposed setbacks, buffers and property rights acquisitions.

The most accurate data in terms of historic and projected rates of development can easily be had from the annual records of the landowners or the Maine State Bureau of Taxation, Both maintain files, dating back beyond 1970 (when LURC was formed) up to this year on the number of leased lots and the number of structures on these lots. A straight line projection of this 20+ year historic data base would result in a more "accurate scientific analysis" and higher "quality information* than the documentation used by FERC staff.

RESPONSES TO JIM HASKELL ON UPPER PENOBSCOT RIVER BASIN DEIS

The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period. GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff proposes, for the Ripogenus project area, two options: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion on GNP-owned lands. For the Penobscot Mills project area, the staff recommends a 200 foot expansion of the project boundaries on GNP owned lands. The staff recommends that existing structures would be grandfathered under any of the project boundary expansion alternatives. See section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.

ON UPPER PENO8SCOT RIVER BASIN DEIS

RESPONSES TO JIM HASKELL

PUBLIC HEARING TESTIMONY: EIS FERC PROJECTS NOS 2572 AND 2458 January 25, 1995 Stearns High School Millinocket, Marie

3. Cont

In addition, FERC knows that the US Forest Service is clearly not an unbiased player in these matters.

- B. It is my opinion that the rationale put forth in the Draft EIS for these proposed drastic actions:
 - is inconsistent with "approved state plans and laws" and these inconsistencies have not been adequately "reconciled" in the Draft E[S...As required by CEQ Regulations in 40 CFR 1506.2(d).

Clearly the minimal environmental problems sought to be mitigated by these measures are adequately controlled by existing State Land Use and Environmental Protection Laws including water quality laws mandated by the Federal Government and administered by the State.

C. In my opinion the proposed building setbacks, vegetative buffers and the forced acquisition of private property <u>constitute the equivalent of an Unfunded Federal</u> <u>Mandate.</u>

Every dollar FERC forces B-GNP to spend to meet frivolous, minimally important, undocumented and unnecessary environmental impacts is a dollar not invested in plant and production capacity and therefore has a significant adverse impact on the economy of the towns of East Millinocket, Millinocket and the Region.

D. In my opinion the proposed building setbacks, vegetative buffers and the forced acquisition of private property is <u>un-warranted in terms of the number of people</u> who would benefit.

4

My personal experience and knowledge of the gates and road system leads me to believe that less than 1 percent of the 130,000 visitors logged at B-GNP gates ever use the Ripogenus impounded lakes. Most end up recreating between the Big Eddy Bridge and the Abol Bridge or are simply traveling through to destinations outside the project area such as, the Allagash, traveling between Greenville and Millinocket, or to other lakes and ponds elsewhere.

Cont

PUBLIC HEARING TESTIMONY: EIS FERC PROJECTS NOS 2572 AND 2458 January 25, 1995. Stearns High School. Millinocket, Maine

Can you really justify using such extreme measures for the limited and unnecessary benefit of less than 1,000 persons?

Did you poll any of these 1,000 persons,...To ask them if they thought such drastic measures were necessary? How many of the intervenors' members are included in this 1,000? Did the intervenors poll their membership?

My family and I go through these B-GNP gates at least once per weekend and sometimes twice, from spring to fall. I do most of my fishing and photography in the watershed above Ripogenus dam. No one from FERC ever asked us, for our opinion.

From what factual basis did FERC and the intervenors come to the conclusion that we needed protection from ourselves...And that they know what is best for us.

<u>The Fifth Point</u>....I strongly believe that you should be rewarding B-GNP for the significant recreation opportunities they provide the general public through their generous *multiple-use management policies*...Instead of punishing them on behalf of the intervenors.

These are not urban dams. This is not the Androscoggin or the Kennebec or the Lower Penobscot. These dams are part of an integrated industrial forest product manufacturing system.

The lands in the project area are not "Wilderness" or "Semi-Wilderness", they are managed private fiber producing lands, producing jobs and Local, Regional, State and National wealth.

These private lands are the last of their kind. No where else in the world can you find such a relatively contiguous private timberland holding...of 2.1 million acres...With such an extensive private road system direct from the stump to the mill...With its own integrated private hydro-electric system. Not in North or South America. Not in Europe or Africa.

RESPONSES TO JIM HASKELL ON UPPER PENOBSCOT RIVER BASIN DEIS

3. Cont

4. Opinion noted.

3.

4.

Cont

PUBLIC HEARING TESTIMONY: EIS FERC PROJECTS NOS 2572 AND 2458 January 25, 1995 Stearns High School Millinocket, Maine

4. They can not be replicated anywhere else in the world, except perhaps in China or Russia fifty years from now. This set of circumstances is rare, unique and exemplary. It deserves to be left alone in private hands, and allowed to continue as a National Historic Area of private capital and job producing property. It is a fast disappearing bit of working (Pre-Information Age) American History.

<u>The Sixth Point:</u>....I really do not believe that you have carefully thought out your proposed regulation of existing and future development and vegetation removal in the proposed extended project area boundary. As a Code Enforcement Officer and as Consultant who spends a lot of time drafting and/or revising Ordinances and Codes, I can assure you that FERC or GNP does not want the job of administering the "grandfathered" uses.

Who is going to be responsible for the administration, inspections, and enforcement required? B-GNP? FERC? LURC? or some new bureaucratic entity?

Can FERC legally (and morally) supersede State of Maine environmental and land use law?

Are you prepared to construct clear language that will give clear answers to the following consumer questions?

- Is my building grandfathered?
- Is my unbuilt upon lot grandfathered?
- Is my commercial use grandfathered?
- Is my illegal building grandfathered?
- Is my illegal use grandfathered?
- If my non-conforming camp/house burns down can I rebuild?
- If my non-conforming camp/house burns down can I rebuild a slightly bigger camp/house?
- If my non-conforming camp/house burns down can I rebuild in a different location?
- If my non-conforming camp/house burns down how long do I have before I have to rebuild?
- If my non-conforming camp/house burns down is my shoreland setback grandfathered?

6

Can I convert my existing non-conforming camp to a year-round house?

RESPONSES TO JIM HASKELL ON UPPER PENOBSCOT RIVER BASIN DEIS

5. The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). Within the Ripogenus Project area, the recommended alternative proposes two options: (1) the adoption of the proposed 250-foot conservation easements for GNP owned land as defined by the MOU; or (2) a 200-foot boundary expansion on GNP-owned lands (see section 4.9.1). The proposed 250-foot easements would lie outside the project boundary and would remain under LURC land use regulatory control. For the Penobscot Mills Project, the recommended alternative proposes expanding project boundaries to generally extend 200 feet from the high water mark of the impoundments within the project area, only on land currently owned by GNP.

Within the proposed boundary expansion areas, existing structures would be grandfathered and GNP would have the authority to review and approve proposed actions as established by the Commission under the Standard Land Use Article and/or SMP. See Section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.
COMMENTS FROM JIM HASKELL

PUBLIC HEARING TESTIMONY: EIS FERC PROJECTS NOS 2572 AND 2458 January 25, 1995 Stearns High School Millinocket Maine

 Can I continue to operate my existing home occupation from my nonconforming camp/house?

Can I start a new home occupation from my non-conforming camp/house?

• Can I build an accessory building/structure on my non-conforming lot?

- o How much can I expand the footprint of my non-conforming camp/house?
- How much can I expand the volume of my non-conforming camp/house?
- How much can I remodel/repair my non-conforming camp/house?
- o Etc. Etc. Etc. Etc.

Clearly LURC has much more experience with these questions than B-GNP or FERC. Under State Law won't LURC still be involved regardless of what FERC does? Will people have to get more than one permit for everything?

<u>Finally</u>....I urge you to consider making a finding of no significant impact with regard to shoreland development in light of existing State Laws and Regulations and leave the Project Area boundaries where they are.

Being unnecessarily forced into becoming a tool of the intervening special interests, in their "Land for Dams" scam, is of no gain whatsoever to the credibility or the future of FERC. Clearly FERC should be acting in the best interests of the majority of the households in America and the majority of the households in Maine...And they have not requested that you replace LURC on their behalf.

If we really must put things back to the way they were in pre-European settlement...As the intervenors advocate....Lets start on Cape Cod or the Potomac River or the Hudson River or Northeast Harbor...Leave us alone to continue to enjoy Maine....The way it is....Or as our State Bureau of Tourism says..."The Way Life Should Be." Let it be the way it was before the intervenors came up with their "Land for Dams" scam.

James S. Haskell, Jr.

206 Highland Avenue Millinocket, Maine 04462

2

(207) 723-8543

attach:

RESPONSES TO JIM HASKELL ON UPPER PENOBSCOT RIVER BASIN DEIS

5.

Cont

6. Opinion noted.

6.

5.

Cont

NANCY L JOHNSON

COMMITTEE ON WAYS AND MEANS

SUICOMMITTEES HEALTH TRADE CONNENTTEE ON

COMMITTEE ON STANDARDS OF OFFICIAL CONDUCT CREMM EXPORT VASK FORCE Congress of the United States House of Representatives Mashington, DC 20513-0706

1.

The optimized

.

NUM BRITHE DIFICE ABC MINETLE STORET-SUITE 200 REW BRITHE CT 08052 TELEPHONE (203) 223-8410

95FEB 21 PILI2: 08 A LATORY CONST.

15-50054

February 16, 1995

Ms. Becky Schaffer Federal Energy Regulatory Commission 825 North Capital Street, NE Room 9200E Washington, D.C. 20426

Dear Ms. Schaffer:

E-36

1

I am writing on behalf of my constituent, Mrs. Betty Trott-Tessman, who has contacted me regarding her concerns with the Hydroelectric licenses applied for by Great Northern, Inc. located in Millinocket, Main (FERC No. 2572 and FERC No. 2458).

Mrs. Trott-Tessman has indicated to me that Great Northern, Inc. owns the land her camp is on. She claims that the Land Use Regulatory Commission (LURC) of Maine has been adequately serving the needs of the community. She feels that the involvement of FERC will cause over-regulation and will impose unrealistic standards for use of summer homes, new construction and general land use. Mrs. Trott-Tessman believes that it would be in the best interests of the town of Millinocket if FERC supervises the Hydroelectric operations and leaves the land and waterway regulation issues to LURC.

I respectfully request your investigation into this matter and that you provide an explanation of the details of your findings to my caseworker, Betty Butler, 480 Myrtle Street, New Britain, Connecticut 06053 or at (203) 223-8412.

Verv truly yours Nancy . Johnson

Member of Congress

NLJ/bjb Enclosure

RESPONSES TO NANCY JOHNSON ON UPPER PENOBSCOT RIVER BASIN DEIS

The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff proposes two options for the Ripogenus project area: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion on GNP-owned lands. For the Penobscot Mills project area, staff recommends a 200 foot expansion of the project boundaries on GNP owned lands. Staff recommends that existing structures would be grandfathered under any of the project boundary expansion options.

心心

RESPONSES TO NANCY JOHNSON ON UPPER PENOBSCOT RIVER BASIN DEIS

27 Hutchinson Street New Britain, Connecticut 06053

January 26th, 1995

Representative Nancy Johnson U.S. House of Representatives Washington, D.C. 20510

Dear Representative Johnson:

Enclosed please find a copy of a letter which my sister and I have sent to Lois Cashell, Secretary of the Federal Energy Regulatory Commission regarding the reneval of two Hydroelectric licenses applied for by Great Northern, Inc. located in Hillinocket, Maine (FERC No. 2572 and FERC No. 2458).

Whatever actions FERC may take on these applications are ones potentially having very long term effects not only on the town of Millinocket, Maine, which has been economically fragile for many years now, but on my entire present and future family members, most of whom reside legally in Connecticut.

Much of the town of Millinocket is built upon land owned by Great Northern, Inc. It is often leased to individuals such as me to build upon. As a camp owner and summer resident one Cont of my greatest concerns is that FERC, in renewing these aforementioned licenses, apparently plans to inflict unrealistic standards upon the use of land surrounding these two projects. In fact, I question FERC's overlap of authority with the Land Use Regulatory Commission (LURC) of the State of Maine.

> Having invested more than \$40,000. in property improvements to our camp in just the last two years, all of which was done with the prior approval of LURC, I am even more concerned about any transfer of land use regulation to FERC, who failed to mention existing camp owners who fall into my position--land renters, camp owners -- in their Alternative I to FERC Nos. 2472 and 2458. Will FERC allow Great Northern to renew my land lease? If so would I then be subjected to their proposed and unrealistic standards for new construction in the area? Nould I even be allowed the use of the camp which I own? If my rented land which is on an island in the middle of a lake does not conform to FERC standards would my camp have to be moved and if so at whose expense?

RESPONSES TO NANCY JOHNSON ON UPPER PENOBSCOT RIVER BASIN DEIS

FERC Nos. 2572 and 2458

Page 2

The conditions which are proposed by FERC in approving the applications of Great Northern. Inc. are ones which would severely interfere with the autonomy of an entire region which is already, in my opision, adequately regulated by LURC. Not only do I consider this as extreme interference with a town's autonomy, but complete duplicity of the services already adequately rendered by LURC. Why must we, as taxpayers, pay for land use regulation services twice?

I hope that you will look into this problem for me and if you see fit encourage FERC to approve the applications as proposed by Great Northern, Inc. In other words, let's leave all land and waterway regulation of this area to LURC which is already doing an excellent job free of Federal interference. And, let us leave the actual supervision for the HydrGelectric operations to FERC. Let's pay for the job only once.

Thank you for whatever help you may be able to give my family in investigating this matter which FERC plans to act upon by February 8th, 1995.

Yours truly, Netter liast lessman

Betty Trott Tessman Legal resident of Connecticut Camp owner, Town of Millinocket, Maine

ET

Enclosure

1.

Cont

RESPONSES TO NANCY JOHNSON ON UPPER PENOBSCOT RIVER BASIN DEIS

4. A. A. L. L. -

27 Hutchinson Street New Britain, Connecticut 06053

January 26th, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

Dear Ms. Cashell:

As the owner of a camp on North Twin Lake in Millinocket, Maine I am writing in support of the applications of Great Northern, Inc. for renewal of their thirty year license for the Ripogenus Hydroelectric Project (FERC No. 2572) and the Penobscot Mills Hydroelectric Project (FERC No. 2458).

Our family goes back many generations in the area encompassed by the Penobscot Project. Additionally, we have owned a camp on land leased from Great Northern on North Twin Lake for the last sixty years. During this time we have seen a good situation environmentally decline and then improve greatly after the establishment of Maine's Land Use Regulatory Commission in 1951. There have been far reaching effects from this point forward. Logging practices have been greatly improved, our beautiful lakes are in pristine condition, and most recently we have observed great improvement in the quantity and diversification of fish and water fowl populations. We have never in our lengthy lifetimes seen the area in better condition than it is nov. This pleases us greatly.

That this is the case is in no small way due to the fact that LURC - Land Use Regulation Commission - serving approximately two thirds of the state of Maine, has already set and enforced on its own with no Federal suggestion solid environmental standards for the area in which both FERC No. 2572 and FERC No. 2458 fall. These LURC standards are tough, environmentally sound, working well, fair to area residents and visitors alike, and already paid for by the state of Maine and its residents.

If the Draft Environmental Impact Statement were so accurate, and tighter environmental controls needed, why are residents seeing continuous, steady environmental improvement such as that which we outlined? Perhaps the excellent controls already

1. Cont

RESPONSES TO NANCY JOHNSON ON UPPER PENOBSCOT RIVER BASIN DEIS

FERC Nos. 2572 and 2458

Page 2

in place in the state of Maine are not always in evidence in other areas serviced by FERC. But, for us in Maine your attempt to over regulate leads me to think of this as unneccesary Federal intervention. This seriously interferes with the individual rights of many people and the autonomy of an entire area of already environmentally conscious people.

Once again, we reiterate our total support of the renewal applications of Great Northern, Inc. as submitted and hope you will in wisdom leave matters affecting surrounding property in these areas encompassed by both projects up to the state of Haine and their Land Use Regulatory Commission which appears to already be doing a more than adequate job of regulating their own environment.

Respectfully,

Betty Trott Tessman 27 Hutchinson St. New Britzin, CT 06053 Camp Owner Millinocket, ME

Sandra Trott Davenport 12 Eton Place Nev Britain, CT 06051

Camp Owner Millinocket, ME

1.

Cont



Dear Ms. Cashell:

For many generations my family has been involved in the stewardship of Maine land, especially in the Ripogenus area. When I was first introduced to the magnificence of the Maine North Woods, the Chesuncook - Ripogenus area was then being used for log drives, so much a part of Maine's past. Since that time I have introduced to the area two generations of Rogers, with pride and deep appreciation for the woods, the fauna, and the waters. So too do we harbor environmentalist concerns - they flow deep in each family member. I have canoed the West Branch with my sons, then six and seven, reading from Thoreau's journals to them at night; the next year we canoed the Allagash. Since that time my family has renovated an old camp that was abandoned on Ripogenus.

Our love of Maine is deep and broad, my mother and wife being Mainiacs. Loving the tranquillity of slipping in a cance up on moose wading from the shore, being lulled by the loon, watching the sun rise over Kathadin - these are wondrous experiences alforded by north woods. Surely they merit our protection and respect. But as I have canced these shores, been encompassed by the beauties of the shoreline I, perhaps, unlike some ardent environmentalists, do not find the presence of man an anathema in nature. Man too in his simple pursuits is very much a part of nature and her rich bounty. The aroma of coffee wafting above a campfire, the sound of children's chatter along the shore skipping rocks - in moderation, with wise regulations, the presence of man is not unpleasant. After a two week cance trip, slipping by pristine shoreline, soaked by unexpected rain, the smell of a fire; the sound , yes, of even a chain saw of a leasee getting wood, the presence of a camp along the shore, this manifestation to the tired traveler is welcome indeed. From the heart of nature civilization can be a wonder.

Though I endorse protecting the environment from the abuses of development, I feel the FERC proposal to force by eminent domain Great Northern to buy easements on the Ripogenus impoundment area is a travesty, for several reasons:

1 am a quiet voice, but I have canoed, camped, and lived in this area, and I do not feel that the quality of the natural environment is threatened; but, if an area is deemed necessary for the federal government to take it from private

RESPONSES TO LANCE ROGERS ON UPPER PENOBSCOT RIVER BASIN DEIS

1. Opinion noted.

2.

The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff proposes two options for the Ripogenus project area: (1) accepting the conservation easement proposed by GNP and the State of Maine; or (2) a 200-foot boundary expansion on GNP-owned lands. For the Penobscot Mills project area, the staff recommends a 200 foot expansion of the project boundaries on GNP owned lands. The staff recommends that existing structures would be grandfathered.

COMMENTS FROM LANCE ROGERS ON UPPER PENOBSCOT RIVER BASIN DEIS

ownership because of environmental abuse, then the government must prove its case and then under its own auspices legally take over ownership of the land. Surely for such a scenario the cogency of the government claim to the land must be proven. That FERC is forcing Great Northern to obtain fee interests or conservation easements on 2,000 acres of Ripogenus / Chesuncook land is unconscionable. Does it not smack of blackmail, to obtain re licensing the corporation which is very much opposed to the forced acquisition must purchase land from disjunctioned land owners who have a long record of good stewardship.

Cont

2.

3.

unconscionable. Does it not smack of blackmail, to obtain relicensing the corporation which is very much opposed to the forced acquisition must purchase land from distinctined land owners who have a long record of good stewardship. If this land is threatened to the degree that some environmentalists assert, then a federal agency, perhaps the Dept. of the Interior, should assess the threat to the area and act under federal auspices and congressional authority. To force a company to purchases this land in order to obtain re licensing is irrational and simply wrong. I was under the distinct impression that the 1994 congressional elections suggested a message that federal regulations and authority had over-reaches themselves - that the later 90's were a time for state authority to exercise leadership.

Since 1974 I have had several dealings with the state of Maine's Land Use Regulations Commission. By the state of Maine and her citizens this state commission has been entrusted with the authority of regulating development in the unorganized towns. By the criteria and regulations to which I was asked to comply I have been impressed with their stewardship of the lands with which they have been entrusted. By nature the layman does not enjoy (and sometimes mistrusts) the state authority with which he must work. But always both by phone conversation and personal inspection I found the L.U.R.C. authorities to be responsive and fair. That another agency, a federal agency, would be superimposed over a state agency which is already in place and functioning with prudence and discretion seems arrogant and counter productive. Maine has a tradition of deciding what is best for her citizens, her lands, and her economies. That in 1995 the federal government would supersede the authority of the Maine citizenry seems ludicrous. Maine and her citizens are eminently qualified to govern themselves.

I believe in ecology as stated above, but so too do I recognize that there has to be an equitable balance between the aesthetics of the natural world and man who has an economic presence in that world. That Great Northern/Bowater has been forced to spend \$9,000,000 in over eight years to be granted re licensing permission for its hydro electric plant seems a travesty. This senseless expenditure and inordinate time period well represent the pejorative views we all have of Big Government. Surely there have been controversial views, ecological issues, corporate issues, not always easily reconciled. But not \$9,000,000 and eight years. The individuals, families, town members of

Millinocket and adjacent areas, not to mention two major corporations, Great Northern and now Bowater, working in the State of Maine, have paid a grievous price for Federal regulation.

This prodigal expenditure placed into plant modernization and corporate ecological cognizance surely would have benefited the state of Maine and her

RESPONSES TO LANCE ROGERS ON UPPER PENOBSCOT RIVER BASIN DEIS

- 3. Opinion noted. See response 2.
- 4. Opinion noted.

COMMENTS FROM LANCE ROGERS ON UPPER PENOBSCOT RIVER BASIN DEIS

RESPONSES TO LANCE ROGERS ON UPPER PENOBSCOT RIVER BASIN DEIS

citizenry rather than unreasonable regulations , proposed forced land purchases, and countless legal haggling in the halls of Washington. Please, consider in your FERC deliberations that the best place for

4. Cont

Please, consider in your FERC deliberations that the best place for regulation is in the state of Maine which has prudent and proven authorities in place and which has an enlightened citizenry capable of and eager to govern themselves.

Sincerely,

Janue D. Rogues

COMMENTS FROM JAMES W. SEWALL COMPANY ON UPPER PENOBSCOT RIVER BASIN DEIS JAMES W. SEWALL COMPANY Express. Surveyors and Pascaros Company 95 FEB - 6 PM 2: 57 ULATORY January 31, 1995 MMISSION Ms. Lois Cashell, Secretary FERC Commission 825 Capitol Street, N.E. Washington, DC 20426 005

RE: Project #2572 (Ripogenus) & #2458 (Penobscot Mills)

Dear Commission Member:

At the recent hearing in Millinocket, I offered testimony which included reference to 24 sales of shorefront property in Maine. Herein I've provided additional information identifying each sale according to the type of sale (fee/easement) as well as certain other brief comments. Differences between fee and easement sales can only be assessed on the basis of the details of each transaction and, as comparables, related to a particular subject property(s); however, it has been my observation that easement transactions usually command a high percentage of full fee value. I hope this elaboration will be helpful to you.

Sincerely,

Robert B. Fiske, L

Chief Forester/Appraiser

Enclosure

147 Center Street - P.O. Box 423 - Old Town, Maine 04/69 USA - 207 / 827-4466 - FAV - 207 / 827-3641

RESPONSES TO JAMES W. SEWALL COMPANY ON UPPER PENOBSCOT RIVER BASIN DEIS

Comment noted.

COMMENTS FROM JAMES W. SEWALL COMPANY ON UPPER PENOBSCOT RIVER BASIN DEIS

4

RESPONSES TO JAMES W. SEWALL COMPANY ON UPPER PENOBSCOT RIVER BASIN DEIS

FERC/TENORSCOT RIVER BASEN HYDROGLECTERC RE-LACEMENC HEARING

•

(1)11 (1)160 Took Lake, T10 S.D. Fee. (2)1309 Finguard: Excit Fee. below market. (2)1309 Finguard: Excit Fee. below market. (2)1309 Finguard: Excit Fee. below market. (3)1319 Finguard: Excit Fee. with convenents. (4)1319 Finguard: Excit Fee. with convenents. (4)1319 Finguard: Excit Fee. with convenents. (4)132 20190 Finguard: Excit Fee. with convenents. (4)132 20190 Finguard: Adomations Fee. with convenents. (4)133 14. Outstoped. Errol Fee. on socres. (4)133 14. Outstoped. Errol Fee. on socres. (4)133 14. Outstoped. Errol Fee. on socres. (5)133 13. Outstoped. Errol Fee. on socres. (6)1339 14. Outstoped. Errol Fee. on socres. (6)1339 14. Outstoped. Fee. (7)2 102309 Monoschead LJ. Toolnegan Fee. (7)3 102309 Monoschead LJ. Toolnegan <	1122 1128 1128 1128 1128 1128 1128 1128	1,200 760 760 3,600 3,000 3,100 4,607 4,100 23,390 23,390	00000000000000000000000000000000000000	000'0615 000'0615	160'68 51,5'68	8 3 3
18.1 [21)309 Flaguarf. Evois Fee, below market. 18.0 [210]30 Flaguarf. Evois Fee, with conversion 18.1 21409 Londword LA, Lakeville Fee, with conversion 18.3 21409 Londword Fee, with conversion 18.3 21409 Londword Fee, post with conversion 17.3 2160 Conversion Fee, post with conversion 17.3 2160 Li Conversion Fee, with conversion 18.4 11/07 Lk Londworgk, Errol Fee, with conversion 18.6 10/1309 Lk Conversion Fee 18.6 10/1309 Lk Conversion Fee 18.6 12/209 Kordak Lk, Fills Niller Fee 18.7 12/201 Foe M, L, 108.1 Fee 18.8 12/203 Kordak Lk, T14.1 Fee 18.9 212/09 Londwork Ellisterille Fee 18.9 12/203 Kordak Lk, L14.1 Fee 18.9 212/09 Londwork Ellisterille Fee 18.9 212/09 Kordak Lk, T14.1 Fee 18.9	160 1716 1716 1716 1716 1716 1716 1716 1	750 760 760 760 75,000 75,000 75,000 75,000 75,000 75,000 75,000 75,000	5 5 5 5 5 5 5 5 6 6 6 6 6 6 6 6 6 6 6 6	000'04'15 000'945 000'945		33
170 2409 Lombard LL, Lakeville Fee with convensus 100 181 2409 Lombard LL, Lakeville Fee with convensus 2409 Copyelle 183 2409 Copyelle Lakeville 172 2409 Copyelle Kap 172 2409 Copyelle Kap 172 2409 Copyelle Kap 173 2409 Copyelle Fee, post with convension 173 101399 LL Owew, Elikestwilke Fee, ensement Fee, post with conversion 1725 107399 Fee, ML, TIGR.D. Fee, mith convension 1725 107399 Ext, ML, TIGR.D. Fee, fee, mith convension 1729 127993 Copyelie LL, TIR.D. NBPP Fee, fee, mith convension 1720 127993 Copyelie LL, TIR.D. Fee, fee, mith convension 1720 127993 Copyelie LL, TIR.D. Fee, fee, mith convension 1720 127993 Copyelie LL, TIR.D. Fee, fee, fee, fee, fee, fee, fee, fee,	500 316 316 316 316 312 312 312 312 3270 3270 3270 3270 3270 3270 3270 327	760 1,000 1,000 1,000 1,000 4,000 4,000 23,000	6 6 8 8 8 6 8 8 8 8	000'WS		5
 100 1173/09 Janice LJ, Lakreiki, Fee with concentral (18) 21(49) Capangie, Adamstone Fee, post with concentral (17) 22(49) Capangie L, Rangeley Fee, post with concentral (17) 20(49) Concer A, Leadan Fee, and concentral (17) 20(49) Concer A, Leadan Fee, and concentral (17) 20(49) Concer A, Dedam Fee, and concentral (17) 20(49) Lobaren, Eliatorial E, Fee with concentral (17) 20(49) Lobaren, Eliatorial E, Fee M, Bernera (17) 20(2)/09 Laboren, Eliatorial E, Fee M, Bernera (17) 20(2)/09 Laboren, Eliatorial E, Fee M, Bernera (17) 20(2)/09 Laboren, Eliatorial E, Toninega (17) 20(2)/09 Laboren, Eliatorial (17) 20(2)/09 Laboren, Eliatorial (17) 20(2)/09 Laboren, Eliatorial (17) 20(19) Manusantag (13, T(41) 74)/1 Fee M 2007 1112 20(9) Laboren, Eliatorial (13, 14) 14) Fee M 2007 1113 20(9) Laboren, Eliatorial (13, 14) 14) Fee M 2007 1113 20(9) Laboren, Eliatorial (13, 14) 14) Fee M 2007 1113 20(9) Laboren, Eliatorial (13, 14) 14) Fee M 2007 1113 20(9) Laboren, Eliatorial (13, 14) 14) Fee M 2007 1113 20(9) Laborentarial (13, 17) 15 1114 20(9) Laborentarial (13, 14) 14) Fee M 2007 1115 20(9) Laborentarial (13, 17) 14) Fee M 2007 1115 20(9) Laborentarial (13, 17) 14) Fee M 2007 1116 10(19) Manusantaria (13, 17) 14) Fee M 2007 1113 20(9) Laborentarial (13, 17) 14) Fee M 2007 1114 20(10, 14) 14) 14) 14) 14 	716 1111 1111 1111 1111 1111 1111 1111	00 3,000 3,000 3,000 4,667 4,000 4,100 4,100 4,100 5,300 4,100	6 0 0 0 7 8 8 8	006'65		
1148 31/474 Capaceptic, Adamstone Fee, poor wfl. 1723 47/975 Likagetry Fee, no accoss. 1724 47/976 Gones PA, Langetry Fee, no accoss. 1724 47/976 Gones PA, Landenger, Errol Fee, no accoss. 1729 47/976 Gones PA, Londanger, Errol Fee, semenary. 1737 47/976 Lik Undinger, Errol Fee, semenary. 1747 Statisticalis Fee Fee 1745 Statisticalis Fee Fee 1745 Statisticalis Fee Fee 1745 Statisticalis Fee Fee 1753 Fea J. Fankegan Fee 1753 Fea J. Fankegan Fee 1753 Fea J. Admissione Fee 1753 Fee J. Admissione Fee 1753 Fee J. Admissione 1753 J. Connew, Ellictorilis<	11.0 11.0 11.0 11.0 11.0 11.0 11.0 11.0	000,0 2001,0 201,0 201,0 201,4 201,4 201,4 201,4 2004,000,000,000,000,000,000,000,000,000	888	200'0015	21.11 11.12	615
(91) S(147) M. Blue Poul, Avon Fee, no access. (72) 42601 Cupromptie: Li, Rangelery Fee, no access. (72) 21072 Li, Unadorgi, Errol Fee, on access. (73) 21072 Li, Unadorgi, Errol Fee, on access. (73) 21070 Li, Channe, Elihearolite Fee, on 500 anip. (63) (101799) Li, Channe, Elihearolite Fee (63) (101799) Lourane, Elihearolite Fee (64) (101799) Lourane, Elihearolite Fee (64) (101790) Monetopie et al., Admission Fee (112) (111, R.) NBPP Fee (112) (111, R.) (111, R.)	174.0 112.4 112.4 112.4 112.0 112.0 112.0 112.0 112.0 12.0 1	000,1 201,1 201,2 201,4 201,4 201,4 201,4 201,4 200,4	900 100		51,205	2
1721 47691 Copensitie Li, Rangeley Fee, generality point 1721 41999 Gones Mi, Diedkam Fee with consensations 1899 11972 Li, Undinging, Errol Fee with consensations 1800 11972 Specializhand Mi, Forris City Fee excenses 1800 11972 Specializhand Mi, Forris City Fee 1800 11072 Specializhand Mi, Forris City Fee 1800 11072 Specializhand Mi, Forris City Fee 1800 11072 Specializh Ministra Fee 1801 107309 Li, Gonew, Ellinstville Fee 1801 107309 Specializi Li, Tombegan Fee 1803 127993 Cupenptic et al. Adamstorm Fee 1809 127993 Cupenptic et al. Adamstorm Fee 1809 127993 Li Li Onarea, Elliostville Fee 1809 127993 Cupenptic et al. Adamstorm Fee 1809 127993 Li Li Onarea, Elliostville Fee 1811 279943 Li Manuvakage Li Li Ti Ri Li Ti Ri Li Fee 1813 21193 2119 1816 2119 Fee 1816 Fee 2007 1813 21991 Li Onarea Filostville Fee 1813 21993 Li Onarea Li Li Ti Ri	Mf. 1010 1010 1011 1012 1	2,105 198,5 198,5 1001,4 10000	0.06		295	8
(17) 4/1999 Goose PA, Dedhan Free with concentration (89) 7/1972 LL. Uushappa, Ernel Free with concentration (89) 7/1973 LL. Uushappa, Ernel Free with concentration (89) 7/1973 LL. Uushappa, Ernel Free with concentration (80) 7/1973 LL. Uushappa, Ernel Free (80) 7/1973 LL. Uushappa, Ernel Free (80) 7/1973 LL. Oushappa, Ernel Free (80) 107399 LL. Guewa, Elikotsville Free (81) 1172291 Frankegan Free (83) 1172291 Frankegan Free (183) 1172291 Constructed LL. TORNEgan Free (84) 1127993 Cupenprix et al. Adamstown Free (184) 1127991 LL. Ouenwa Elitotsville Free (192) 711291 LL. Ouenwa Elitotsville Free (193) 211991 Mansurvaturage LL, TTRI LTALLI Free (116) 711930 Mansurvaturage LL, TTRI L Free (116) 1116694 Sanke & Copenet LLI, TTRI L Free (116) 1116694 Sanke & Copenet LLI, TTRI L Free	201.0 147.0 151.1 151.0 152.0	4,667 35,150 4,100		1280,000	\$1,335	8
 7/1072 LL. Undiaged, Errol 1699 7/1072 Lg. Undiaged, Errol 1630 1011349 Ld. Conserv. Elitionshile 1600 1011349 Ld. Conserv. Elitionshile 1725 102349 Monochend Lk. Tonikegan 1603 1172391 Ex. T. 105.D. 1603 1172391 Ex. T. 105.D. 1603 1172391 Ex. T. 105.D. 1603 1172991 Ex. Conserve. Elitionshile 1604 127391 Conserve. Elitoteville 172991 Lk. Conserve. Elitoteville 172991 Lk. Conserve. Elitoteville 172991 Lk. Conserve. Elitoteville 172991 Lk. Conserve. Elitoteville 172930 Mataneouse (Lk. T1R11, T2R11) 172030 Mataneouse (Lk. T1R11, T2R11) 172030 Mataneouse (Lk. T1R11, T2R11) 113 2/1991 Londinghyje Systadobis (Lk. L1R11, T2R11) 113 2/1991 Londinghyje Systadobis (Lk. L1R11, T7R11) 114 	17.0 11.0 12.12 12.12 12.12 12.10 12.12 10 12.10 12.10 10 10 10 10 10 10 10 10 10 10 10 10 1	35,350 201,4 2004,62	1 0:0	\$500,000	22. 4m	1015
 (69) (1) 309 LL Conver, Eliteraritit. (100) 71/132 Spechaiz/Maid ML, Forersi city Fee (101) 209 LL Conver, Eliterari city Fee (11) 201 LL Conver, Eliteraritita (12) 101/209 LL Conver, Eliteraritita (12) 101/209 LL Conver, Eliteraritita (12) 101/209 LL Conver, Eliteraritita (12) 71/209 LL, Conver, Eliteraritita (13) 71/209 LL, Conver, Eliteraritita (14) 2019 Neteraritita (15) 71/209 LL, Conver, Eliteraritita (15) 71/209 LL, Conver, Eliteraritita (16) 71/209 Manametrity Systacheti LLA, T181 L (11) 21/991 Londardytita Systacheti LLA, T181 L (11) 21/991 Londardyty Systacheti LLA, T181 L (12) 71/2093 Londardyty Systacheti LLA, T181 L (12) 71/2091 Londardyty Systacheti LLA, T181 L (13) 71/2091 Londardyty Systacheti LLA, T181 L (14) 71/201 L (15) 71/201 Londardyty Systacheti LLA, T181 L (15) 71/201 Londardyty Systacheti LLA, T181 L (16) 71/201 L (17) 71/201 L (17) 71/201 L (17) 71/201 L (18) 71/201 L (19) 71/201 L (19) 71/201 L (19) 71/201 L (10) 71/201 L (11) 71/201 L <	511.0 523.0 527.0 528.0 528.0 600.0 600.0	A 10 00 1 10	0.0	21,300,000	111,22	2
 1/192 Spechric/Meal PL, Fortest city 1/192 Spechric/Meal PL, Fortest city 1/25 10/2399 Lk. Convex. Elitistiville 1/251 10/2399 For ML, T. 108.D. 1/2529 For ML, T. 108.D. 1/2591 Lk. Convex. Elitotiville 1/2501 Lk. Convex.	5210 527.0 528.0 528.0 822.0 600.0	23,400	0.12	\$455,000		1115
 [60] [01] JPB Li. Gueve, Elitersville [723] For Manuel Li. Tombegan [744] T. T. 1923 [744] T. T. 1923 [744] T. T. 1923 [745] T. 1949 [745] T. 1949 [745] T. L. Oanwe, Elitosville [744] Y. 1959 [746] T. 1950 [746] T. 1970 [747] T. 1971 [746] T. 1970 [747] T. 1971 [746] T. 1970 [748] Lomburdhage Lit., T1811 [748] T. 1970 [748] Lomburdhage Lit., T1811 [749] T. 20199 [749] Lomburdhage Lit., T1811 [740] T. 1970 [740] Sould Systachobis Lit., T1811 [740] T. 1970 [740] T. 1970<	527.0 528.0 228.0 221.0 8221.0 660.0		80	2772,000	925 5	12
 1725 [1073/09] Monochead LJ. Tombegan Fac. 1633 [1723/09] Fox Put, T. 105.D. 1649 [1729/09] Cuproptic et al. Adamstown Fac. 1649 [1729/09] Cuproptic et al. Adamstown Fac. 1729 [17.049] Cuproptic et al. Adamstown Fac. 1720 [17.049] Cuproptic et al. Adamstown Fac. 1721 [1729/19] LL. Onawu, Elitoteville Fac. 1723 [1719] U. Onawu, Elitoteville Fac. 1723 [1719] Matterwarkage [1.1, T181] Fac. 1732 [1719] CumberdNyn Systahobist Lb. 1. Jateville Fac. 1738 [1716/94] Sanke & Corporate Lb. 1. T181] Fac. 1738 [1716/94] Sanke & Corporate Lb. 1. T181] 	528.0 822.0 clow appraisal. 860.0	2,500	0.21	200,2002	\$655	
 [63] 11/2291 Fox M, T.105.D. [49] 12/793 Sequence La, T118.3 NBPP Examination 2307.1 [49] 12/793 Sequence La, T118.3 NBPP Examination 7507.1 [40] 12/793 Sequence La, Admissionn Fox [41] 27/291 Pierce Pound [42] 12/41 Sequence La, T118.1 [43] 27/991 Pierce Pound [43] 27/991 Pierce Pound [44] 27/41 Fox [45] 27/991 Contractivity [46] 27/41 Fox [46] 27/493 Sequence Lib., T78.11 [46] 11/1699 Sealacheis Lib., T78.11 [46] 11/1699 Sealacheis Lib., T78.11 [46] 11/1699 Sealacheis Lib., T78.11 	\$22.0 clow appraisal. B60.0	000.6	90.0	100,000	1963	8
(45) 12/(5) Speduic (L, T11R.) NB/P Easement on 250:1 (60) 12/593 Cupanylic et al., Adamssown Fee. 1723 7/1291 Lk. Onawa, Eliostwille Fee. 1841 7/1291 Lk. Onawa, Eliostwille Fee. 1843 7/1291 Lk. Onawa, Eliostwille Fee. 1841 10/1500 Manuschaus Lit, T1R11, T2R11 Fee. on 250'. 1842 2/1991 Lomburd/ups Systabobis Lit, T1R11, T2R11 Fee. 1843 2/1991 Lomburd/ups Systabobis Lit, T1R11, T2R11 Fee. 1845 2/1991 Lomburd/ups Systabobis Lit, T1R11, T2R11 Fee. 1846 11/1604 Samk & Corporate Lit, T7R11 Fee.	clow appraisal. B60.0	11,200	0.0	\$500,000	19 ,	3
 1689 12/2993 Copendric et al. Adamssown Fee. 1723 71/291 Lk. Onawa, Elitostvilke Fee. 1344 9/391 Pierce Pouel 1497 10/190 Mathematical Lk. T1811, T2811 1201 72659 Mathematical Lk. T1811, T2811 1315 2/19991 Lomburdhige Systacholosis Lbk. Labevilke Fee. 1315 2/19991 Lomburdhige Systacholosis Lbk. Labevilke Fee. 1316 11/1694 Sanke & Component Lbk. T7811 Fee. 		65,032	0.0	000,8173	HOG4	115
 1723 7/12/91 Lk. Onawa, Elicotsville Fee. 1344 9/19/1 Pierce Prand 1497 1/19/10 Nudmickang Li, T1R11, T2R11 Fee on 2507. 111 7/26/93 Mananumakang Li, T1R11, T2R11 Fee on 2507. 111 7/26/93 Mananumakang Li, T1R11, T2R11 Fee. 112 2/19/91 Lombard/Upr Systachobist Lib., LTR11 Fee. 113 2/19/93 Lombard/Upr Systachobist Lib., LTR11 Fee. 		16,400	0.0	000'000" IS	Xe	E
 1344 9/391 Pierce Pond 147 10/190 Nabimutauta Lii, TTR11, TZR11) 148 10/190 Nabimutauta Lii, TTR11, TZR11) 148 2011 7/2693 Muntavenutang Lii, T4R3 150 201991 Londburdhije Systakobois Lii, TZR11 150 11/16091 Sante & Cimponer Lii, TTR11 156. 	0.552.1	9,055 201,0	0.17	1200,000	203	3
 IV/190 National auto Lit, TUR11, T2R11) Fee on 250°. Z111 Z/2693 Manawamizing Lit, T4R3 Z113 Z/19991 LombardVijn Systahobois Lits, T7R11 Fee. F116 L1116694 Sanite & Carponec Lits, T7R11 Fee. 	Below market. 1,587.0	42.240	60	\$345,000	9075	2
 2111 7/26/93 Manawamicag LJ, T4R3 Fee 211991 LondbardVijer Systakobois LLA, Lakteville Fee 1116 11116694 Sandle & Camponer LLA, TTR.11 Fee 	0.044,1	40,900	0.0	000/9/2/15	1213	
1315 2/19791 Lowburd/Lips Systachobis Like, Lakewille Fee. 1176 11/1694 Samile & Camponer Like, T7R11 Fee.	1,23.0	CSN,M	0.0	\$1,026,316	813	5
1176 11/16/94 Smalle & Carpenner Lits, T78.11 Fee.	3,367.4	24,840	0.14	2926,145	5635	125
	7,144.0	61,135	0.12	\$2,150,000	IOC\$	8
1204 BASM Tunk & Spring et als, T. 105.D. Fee + Lan gift.	0,1301,5	962,01	0.0	32,600.000	2963	8
1721 6/1/93 Mooudhead L.L. Tomkegan For	0940	092.60	0.12	\$5,100,000	95	55
Mean All Acres:	1.757.0	24,216.0	0.0	696'9LIS	51,653	Z
Mean 500+ Acres:	2,736.2	35,335	60.0	167,291,18	055\$	S
Mean 1000+ Acres	5.141,5 2	46,044	0.09	ETT, EA3, 12	83	53
Mean 5000+ Acres:	0.275/8/2	71,810	0.11	51,002,02	1374	Ī
	LL ALONA	3515				
	12 m 2 SI					
Defer Detr Curte sources I Luck W Sewill Creature Or Ba						

COMMENTS FROM SPRAGUE ENERGY ON UPPER PENOBSCOT RIVER BASIN DEIS

RESPONSES TO SPRAGUE ENERGY ON UPPER PENOBSCOT RIVER BASIN DEIS

SPRAGUE ENERGY

1. Opinion noted.

SEATSPORT TEIMINAL PO Box 435 Mach I Pour Searaport ME 04974 16L (207) 548-2531 FAX (207) 548-8582

January 25,1995

Good evening! My name is Bruce Probert. I once lived in Millinocket, but now live and work in Searsport, Maine for Sprague Energy. Where I am also chairman of the local planning board.

Speaking on behalf of myself and also Sprague Energy, we continue to urge the Pederal Energy Regulatory Commission to restrict the scope of its environmental impact review to the relevant issues which pertain to water flows and the area in immediate proximity to the impoundments and dams. Land conservation is not, in our understanding, a FERC priority nor mandate.

Overall, environmental impact statements should be an aid to assist regulatory agencies to make an informed decision. In some cases, they have been used as a tool against the applicant to impede the the decision making process. In Searsport, we have seen this first hand after more than 15 years of legal maneuvering, court challenges, and one impact statement after another which have stymied the development of Sears Island and the construction of a modern cargoport for the State of Maine.

It appears that FERC has reached an informed decision on the significant environmental issues relating to the use of waters. This is a water issue, and not a land issue. We in Maine have the Land Use Regulation Commission to rule on development in the unorganized townships, therefore any development set back should be decided by them not by any federal agency. To establish a development set back regulation for the duration of this license deprives us of one of our cherished freedoms, namely

AN ANEL JOHNSON INC COMPANY

COMMENTS FROM SPRAGUE ENERGY ON UPPER PENOBSCOT RIVER BASIN DEIS

RESPONSES TO SPRAGUE ENERGY ON UPPER PENOBSCOT RIVER BASIN DEIS

(2)

SPRAGUE ENERGY SEARGPORT TERMINA. PO Bin 436 Mach & Port Soursport ME 04974 *EL 12071548-2531 FAX (2071548-8562 1/25/95

1. Cont

that of " home rule" . We in Maine are most capable of making Cont these decisions for the general good of us all.

> Please move forward with the relicensing of Penobscot Hills and the Ripogenus Projects as expeditiously as possible. Thank you. We wish you well as you move forward to a final resolution.

> > Sincerely,

J. Bruce Probert J. Bruce Machart

Maine Division Manager

COMMENTS FROM STEVE TUCKERMAN ON UPPER PENOBSCOT RIVER BASIN DEIS



Secretary FERC 825 North Capitol St NE Washington DC 20426

Re: Ripogenus (P-257Z) and Penobscot Hills (P-2458)

Comments are as follows:

1. Whitewater flows in Back Channel (Penobscot Hills) Notify through river phone what the flows are. Clearly mark portage channel around dam. Improve portage trail.

2. Whitewater flows from Ripogenus Gorge (McKay Station down) Allow lower flows evenings (after 5) and Wednesdays. Provide flow information through river phone (toll free) March 1 through December 1. Provide gages at McKay Station, Cribworks, Abol Bridge. Allow same access to private boaters at McKay as enjoyed by commercial rafters.

Thank you Steve .uckerman

20M Broadway Bangor ME 04401

RESPONSES TO STEVE TUCKERMAN ON UPPER PENOBSCOT RIVER BASIN DEIS

- Back Channel would only provide an approximately 4.5-mile-long boating trip in a less aesthetically pleasing setting than along the West Branch. We state in Section 4.8.1.1 that significant opportunities exist within the project area to meet recreational boating demand, and that the significant flows that would be required to enable recreational boating activities would jeopardize other environmental enhancements. As part of the settlement agreement with MEPRO, GNP would maintain establish a telephone message system to provide information about flow conditions along the West Branch and scheduled releases from Ripogenus dam. GNP would also notify a designated representative of MEPRO of any unscheduled releases from Ripogenus Dam.
- 2. See response #1. As stated in section 4.8.1, GNP would provide minimum flows from McKay Station during daytime hours. GNP usually maintains flows at or above 1800 cfs throughout the year, therefore, evening flows may be reduced or varied. The reduced or varied evening flows would provide additional opportunities for less experienced paddlers or those desiring a diversified experience. As part of the settlement agreement with MEPRO, GNP would provide readily visible staff gauges calibrated to river flow levels at McKay Station, the cribworks, and Abol Bridge. We do not recommend gauges below Ripogenus dam or in Back Channel. We believe that the existing parking and boat unloading areas are adequate.

<u>COMMENTS TO CONCERNED CITIZENS</u> <u>CONCERNING LAND USE</u> <u>ON PENOBSCOT/RIPOGENUS MILLS DEIS</u>

<u>Group - A</u>

We received comments from many concerned citizens regarding our recommended land use recommendations in the Ripogenus/Penobscot Mills DEIS. Our response to these comments are provided here and the comments follow.

Comment noted. The staff revised its land use assessment and recommendations in the FEIS (see section 4.9). We considered comments received during the DEIA comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations. The staff's recommendations (see section 5.3.4) provide measures to protect shoreland resources within the project areas while considering existing land use regulatory controls.

Within the Ripogenuse Project area, the preferred alternative proposes two options: (1) the adoption of the proposed 250-foot conservation easements for GNP owned land as defined by the MOU; or (2) a 200-foot boundary expansion on GNP-owned lands (see section 4.9.1). The proposed 250-foot easements would lie outside the project boundary and would remain under LURC land use regulatory control. For the Penobscot Mills Project, the preferred alternative proposes expanding project boundaries to generally extend 100 feet from the high water mark of the impoundments within the project area, only on land currently owned by GNP.

Within the proposed boundary expansion areas, existing structures would be grandfathered and GNP would have the authority to review and approve proposed actions as established by the Commission under the Standard Land Use Article. See section 4.9 and 5.3.4 for further discussion regarding proposed protection zones for the Ripogenus and Penobscot Mills Project areas.



As a Senator in the Maine Legislature, I consider the ability of our State's paper industry to compete and thrive in the global market, of great importance. Accordingly, I have serious concerns regarding the Draft Environmental Impact Statement issued for FERC #2572 and FERC #2458.

It's my understanding that the Draft will increase the applicant's annual cost by about 30 percent. I oppose this increase. Continued operation of the projects as proposed by the applicant will provide much need economic and environmental benefits to Mainers.

Also, I oppose the idea of Great Northern taking over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment. I believe these agencies are capable of continuing that effort.

Sincerely,

Joel Abromson State Senator

IJA/skf



Senator John W. Benoit District 17 State House Station 3 Augusta, Maine 04333

THE MAINE SENATE 117th Legislature

P.O. Box 890 Rangeley, Maine 04970

4

<u>,</u>

February 1, 1995

Lois B Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) - 005 = Penobscot Mills Hydroelectric Project (FERC No. 2458) - 005

Gentlemen:

I write at this time to submit written comments concerning the draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this state.

I oppose the conditions continued in the Draft which will increase the applicants annual cost by an stimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Sincerely,

John W. Benoit Senator, District #17

JWB/law



Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: RIPOGENUS HYDROELECTRIC PROJECT (FERC NO. 2572) -009PENOBSCOT MILLS HYDROELECTRIC PROJECT (FERC NO. 2458) -009

Dear Mr. Cashell and members of FERC:

I am writing to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the 117th Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market. The applicant, Great Northern Paper, is an important part of the paper industry in this state.

I believe that continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of Maine.

I oppose the conditions contained in the Draft Environmental Impact Statement which will increase the applicant's annual cost by an estimated 30 percent. I believe such an icrease will impair Great Northern's ability to modernize its facilities, pay decent wages and compete in today's global markets. I especially oppose the staff proposal that Great Northern Paper take over control of regulating shoreline development from the state. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of state jurisdiction is unwarranted and unneeded.

I strongly suggest that you do not include conditions for this project that would increase the cost of doing business in the State of Maine.

Sincerely,

Gerald N. Bouffard State Representative



Richard H. Campbell Rt. 2, Box 1575 Eastern Avenue Brewer, Maine 04412 Residence Tel: 207-989-6055 Business Tel: 207-945-3585

HOUSE OF REPRESENTATIVES AUGUSTA 04333-0000 FEB 9 AM 10: 07

February 1, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street NE Washington, DC 20426

> RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458) ~

Gentlemen:

As a member of the Penobscot County Legislative delegation, I would like to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

Very truly yours,

Richard H. Campbell State Representative

V

Dietriat 116 Bradley nart of Brewesso Eddington, Holden and Vezzie



80 Cottage Street Sanford, Maine 04073

January 27, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC #2572) Penobscot Mills Hydroelectric Project (FERC #2458) - CO

Gentlemen:

As a Senate Chair of the Maine Legislature's Joint Standing Committee on Utilities and Energy, I consider Maine's paper industry's ability to compete in the global market, of great importance. Therefore, I have serious concerns regarding the Draft Environmental Impact Statement issued for FERC #2572 and FERC #2458.

I oppose the applicant's annual cost increase of approximately 30 percent. Continued operation of the projects as proposed by the applicant will provide economic and environmental benefits to Mainers.

I also disagree with the idea of Great Northern taking over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment. These agencies are capable of continuing that effort in the future.

Sincerely,

David L. Carpenter State Senator

DLC/skf

 \mathcal{U}^{\prime}



Dean F. Clukey

19 Alfred Street Houlton, Maine 04730 Tel: 207-532-6363

HOUSE OF REPRESENTATIVES

OFFICE SEALE SECRETARY 287-1400

95 JAN 23 AM 10: 01

FEDERAL ENERGY REGULATORY COMMISSION 2 AUGUSTA 04333-0002

VERSAR, INC FCOLOGICAL SCIENCES A

January 18, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital St. Washington, D.C. 20426

> RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hvdroelectric Project (FERC No. 2458)

Gentlemen:

I write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the above captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this state.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this state.

l oppose the conditions contained in the draft which will increase the applicant's annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the state, control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of state jurisdiction is unwarranted and unneeded.

Sincerely.

E-382



HOUSE OF REPRESENTATIVES

AUGUSTA 04333-0002 STATE HOUSE 287-1400

OFFICE OF THE SECRETARY 95 FEB - 3 PM 1:54

FEDERAL ENERGY

REGULATORY

COMMISSION

ames O. Donnelly P.O. Box 1883 Presque Isle, Maine 04769

VERSAR, INC. FF 2 4 1995 ECOLOGICAL SCIENCES AND ANALYSIS

January 25, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

Dear Colleagues,

I write to you with regards to the Ripogenus Hydroelectric Project (FERC No. 2572) and the Penobscot Mills Hydroelectric Project (FERC No. 2458). The following are my written comments concerning the Draft Environmental Impact Statement for the aforementioned projects.

It is crucial to the workers of our State that the paper industry is able to compete in the international markets. The applicant, Great Northern Paper, will provide economic and environmental benefits to the people of the State with the approval of these projects.

Bowater's financial commitment to a cleaner environment is proven by the millions of dollars they have committed to a de-inking machine to facilitate recycling of waste paper. Further burdening this environmentally friendly company will be counterproductive to the goal of a cleaner, safer environment with good paying jobs for our people.

It concerns me that conditions in the Draft will increase the applicant's annual costs by 30 percent. Another disturbing element is the proposal that Great Northern would be required to assume control of regulating shoreline development. The land use agencies in Maine have traditionally overseen the preservation and protection of the environment and they are fully capable of continuing to do so. I object to any FERC intrusion into this area of what has always been State jurisdiction.

Sincerely.

ames O. Donnelly

State Representative

 $_{\rm V}$ cc: Great Northern Paper Co. Herbert E. Clark

District 145 Pressure Isle



Gentlemen:

In spite of reports of a "robust" U.S. economy, Maine is still climbing slowly out of the last recession. Our paper industries make up a vital part of our State's economy. Accordingly, I have serious concerns regarding the Draft Environmental Impact Statement issued for FERC #2572 and FERC #2458.

Our State Legislature has taken steps to help our paper industry compete in the world market. The industry's ability to do so is critical. Therefore, I join other members of the Maine Legislature in opposing the applicant's annual cost increase of approximately 30 percent. Continued operation of the projects as proposed by the applicant will provide economic and environmental benefits to Mainers.

Also, I disagree with Great Northern taking over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment. These agencies are capable of continuing that effort in the future.

Sincerely,

Norman K. Ferguson, Jr. State Senator

NKF/skf

<u>\</u>{

E-384



Dear Commissioners:

As a member of the Maine Legislature, I am taking the stand that continued operation on these projects will have a positive impact on this region. Maine is still suffering from the most recent economic slowdown. The cornerstone of our economy is the paper industry and that industry has been especially hard hit. At this time, new regulations and greater responsibilities are not what this industry needs.

While campaigning for my office last summer and during the fall, one message came through loud and clear. That being that the regulatory process is killing the state of Maine. Nationally, the same message was sent. It is my hope that you will take the voters sentiments to heart as you consider FERC 2572 and FERC 2458.

With respect

Charles D. Fisher State Representative

1 =



FILED UFFICE OF THE SECRETARY

95 JAN 27 AN 10: 02

FEDERAL ENERGY

REGULATORY

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) .009 Penobscot Mills Hydroelectric Project (FERC No. 2458)

Gentlemen:

I write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Very truly yours,

 \checkmark

Albert P. Lamache

OFFICE OF THE SECRETARY

5 IAN 23	PH	12:	09
----------	----	-----	----



1	VERSAR. INC.
	· N 0 1775
ECOL	DGICAL SCIENCES AND ANALYSIS

FEDERAL ENERGIA STATE OF MAINE REGULATORY HOUSE OF REPRESENTATIVES AUGUSTA, MAINE 04333-0002

January 20, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital St., N.E. Washington, D.C. 20426

Re: <u>Great Northern Paper Company</u> <u>Repogenus Hydroelectric Project (FERC #2572)</u> <u>Penobscot Mills Hydroelectric Project (FERC #2458)</u>

Dear Ms. Cashell:

I'm writing you in support of Great Northern Paper Company's activity to relicense its various hydroelectric facilities for its operations in the Millinocket area. In my view as a State of Maine House of Representatives member, their ability to produce power for their operations is critical to Maine and our economy. This project must be approved.

I oppose any conditions contained in your Draft that would increase costs of manufacturing to Great Northern. We must do all we can to reduce costs to insure that this mill and other mills are competitive in the industry elsewhere. I understand that the Draft will increase costs by an estimated 30%. This is contrary to what we in Maine are trying to do for business.

Finally, I wish to voice my concern over your proposal to having a change in control over shoreline development. I believe our agencies here in Maine have done an outstanding job over the years in the unorganized territories and they should be allowed to continue to do so. I do not believe it is in our best interests to have the federal government intrude into this area of state jurisdiction where it is not needed.

Because I cannot attend the public hearing you have scheduled in the near future, I hope my written letter will be included in the record.

Sincerely,

Ernest C. Greenlaw State Representative

ECG/hjp

E-387



DAN A. GWADOSKY

Ú

STATE OF MAINE HOUSE OF REPRESENTATIVES SPEAKER'S OFFICE AUGUSTA, MAINE 04333-0002



DEFICE OF FILED 95 FEB -9 AM 10: 09

February 1, 1995

Lois B. Cashell, Secretary F.E.R.C. 825 North Capital St., N.E. Washington, D.C. 20426

Dear Ms. Cashell:

I am writing to raise some concerns which I have pertaining to the Draft Environmental Impact Statement (DEIS) of Great Northern Paper's relicensing request of the Ripogenus Hydroelectric Project (FERC No. 2572) and the Penobscot Mills Hydroelectric Project (FERC No. 2458).

Great Northern Paper (GNP) has been an important part of northern Maine's economy for almost 100 years. In an increasingly more competitive global economy, paper producers are forced to be as cost conscious as possible in order to remain viable. GNP is no exception.

It is my understanding that the DEIS is recommending a 200-foot setback requirement in the federal dam license. This recommendation conflicts with Maine law which requires a setback of at least 75 to 100 feet from rivers, depending on whether the river flows to a great pond. Please consider that these setback requirements were developed with open participation by the people of Maine, reflecting the values and intentions of our citizens.

Another concern which I have relates to the 200-foot extension of the project boundary. This recommendation would, in effect, require GNP to purchase state-owned land around the Ripogenus impoundment. Our state has a great deal of time and money invested in the acquisition of this property. And we would like to protect this land for the use of our citizens.

I am also concerned about the proposed requirement to force private landowners around the impoundment to sell their property to GNP. These landowners have cooperated with state and local efforts to protect the rivers and lakes. They have lived and played by the rules. They should not be punished by being required to forfeit their rights to the land. Letter to F.E.R.C. Page 2

In summary, I urge the Commissioners of the FERC not to adopt the 200-foot setback and not to expand the project boundary, thus requiring GNP to purchase state and privately-owned property from people who do not wish to sell it.

Thank you for the opportunity to comment on these recommendations. Please contact me if you have any questions about my comments.

Sincerely, Gwadosky Dan A. Speaker of the House



House of Representatives

STATE HOUSE AUGUSTA 04333-0002 287-1400

Charles H. Heino P.O. Box 113 Boothbay, Maine 04537

January 24, 1995

JAN 30 AN 10: 38 JAN 30 AN 10: 38 EDERAL EINERGY REGULATORY

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458)

Gentlemen:

i write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Charles H. Heino State Representative

District 58 Boothbay, Boothbay Harbor, Edgecomb, Newcastle and Southport

INFINE OF THE SECHETARY 95 JAN 30 AM 11: 37 EDERAL ENERGY

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) - 005 Penobscot Mills Hydroelectric Project (FERC No. 2458) - 009

Gentlemen:

I write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Very ruly yours,

 \checkmark



Priscilla Lane HCR 1159 Enfield, Maine 04433 Tel: 207-732-4783

STATE	House	August	` A
		287-1400	1
1	VERSAR	130.	
	FEB 2 4	1 1995	
ECOL	OGICAL SCIENC	ES AND ANALYSI	5

House of Representatives



January 26, 1995

04333-0002

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) and Penobscot Mills Hydroelectric Project (FERC No. 2458) ---

To whom it may concern:

The following are my written comments regarding the Draft Environmental Impact Statement issued for the above projects.

As a Maine Representative, I am aware of the severe recession we've gone through and how it is still effecting this state's economy. Maine's paper industry is a crucial part of our economy and it is critical that it continues to compete worldwide. The applicant, Great Northern Paper, is a major player in Maine's paper industry.

The continued operation of the projects as proposed by the applicant, Great Northern Paper, will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants' annual cost by an estimated 30 percent. I am also concerned about the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have traditionally overseen the preservation and protection Maine's environment and are fully capable of continuing that effort. To divide responsibility between the government and private industry would be opposite to what we are currently trying to accomplish. I oppose any FERC intrusion into this traditional area of State jurisdiction

Sincerely.

Priscilla Lane State Representative

District 138 Enfiald HarE-392 d and nort of I incoln

OFFICE CE THE COPETARY





P.O. Box 14 North Waterboro, Maine 04061 (207) 247-5331



Dear Ms. Cashell:

I write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the above captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this state.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarrented and unneeded.

Sincerely,

reen a bord

Willis A. Lord State Senator

WAL/jr



Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

n02

RE: Ripogenus Hydroelectric Project (FERC No. 2572) and Penobscot Mills Hydroelectric Project (FERC No. 2458)

To whom it may concern:

The following are my written comments regarding the Draft Environmental Impact Statement issued for the above projects.

As a Maine Representative, I am aware of the severe recession we've gone through and how it is still effecting this state's economy. Maine's paper industry is a crucial part of our economy and it is critical that it continues to compete worldwide. The applicant, Great Northern Paper, is a major player in Maine's paper industry.

The continued operation of the projects as proposed by the applicant, Great Northern Paper, will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants' annual cost by an estimated 30 percent. I am also concerned about the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have traditionally overseen the preservation and protection Maine's environment and are fully capable of continuing that effort. To divide responsibility between the government and private industry would be opposite to what we are currently trying to accomplish. I oppose any FERC intrusion into this traditional area of State jurisdiction

Sincerely,

Rooney State Representative

District 110 Burnham, Freedom, Knox, Liberty, Montville, Palermo Thorndike, Troy and Unity



FILED UFFICE OF THE SECRETARY UFFICE OF THE SECRETARY Senator Judy Paradig5 JAN 27 N IC OI District 1 State House Station 3 FEDERAL ENERGY Augusta, Maine 04333 FEDERAL ATORY COMMISSION

THE MAINE SENATE 117th Legislature 40 US Route 1 Frenchville, Maine 04745 Fax (207) 728-6374 Home (207) 728-4854

January 20, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, DC 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458)

Gentlemen:

Please allow me to comment, regarding the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Senate, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Sincerely,

Judy Paradis

JP/cam



Penobscot Mills Hydroelectric Project (FERC No. 2458)

Dear Mr. Cashell:

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State. Please consider this statement as part of the official record.

I oppose the conditions contained in the Draft Environmental Impact Statement which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

The impact statement issued for the captioned projects will provide significant economic and environmental benefits to the people of this State. Continued operation of these two projects as proposed by the applicant is desirable for all concerned.

Sincerely,

Thomas E. Poulin State Representative

District 103 Oakland and Sidney E-396



Chester A. Rice P.O. Box 99 Damariscotta, Maine 04543 Tel: 207-563-3661

HOUSE OF REPRESENTATIVES

STAFT FDUSTARY AUGUSTA 04333-0002 287-1400 95 JAN 23 PH 12: 09

FEDERAL ENERGY REGULATORY

	VEED			
	. EN	C	-	
FCOL	OGICAL SCI	ENCES	AND 4	

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

005

Penobscot Mills Hydroelectric Project (FERC No. 2572) RE:

Gentlemen:

I write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that FERC's intrusion into this traditional area of State effort. jurisdiction is unwarranted and unneeded.

Very truly yours,

he AZ

District 56 Bremen, Bristol, Cushing, Damariscotta, Friendship
Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458) - O(

ECOLOGICAL SCIENCES AND ANAL

Gentlemen:

I write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Very truly yours,

Rep. Robert Spen



HOUSE OF REPRESENTATIVES

STATE HOUSE AUGUSTA 04333-0002 287-1400



FEDERAL ENERGY

REGULATORY

Rep. Richard I. Stone 18 Milton Place Bandor, Maine 04401 Tel: 207 942-0692



Lois B. Cashell Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

.002 RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458) -009

Gentlemen:

I am submitting written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a Representative in the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market. Great Northern Paper is an important part of the paper industry in this State.

The continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

I strongly oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30 percent. The staff proposal that Great Northern take over from the State control of regulating shoreline development is also disturbing to me. Maine's land use agencies have done a good job of preserving and protecting Maine's environment in the past and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Sincerely,

2-1~

Richard I. Stone State Representative

Distance 117 E399 of Bangor



RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458)

Gentlemen:

I write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Very truly yours,

Representative District 42

ERDI L. TRIPP

27 Ward Road Topsham, ME 04086 ~ 729-0336

VERSAR, INC

ECOLOGICAL SCIENCES

n 1

District 52 (Topsham)

OFFICE OF THE SECRETA STATE REPRESENTATIVE

95 JAN 24 PM 1:15

FEDERAL ENERGY REGULATORY MICCINN

January 17, 1995

Lois B. Cashell, Secretary Fedferal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

.005 Ripogenus Hydroelectric Project (FERC No. 2572) RE: Penobscot Mills Hydroelectric Project (FERC No. 2458) -

Please include the following remarks regarding the above projects into your record.

As a new Maine Legislator, I intend to do all I can to keep the business and industry we have strong, maintain and increase the job force and lobby for projects which will do just these things.

I believe the Draft Environmental Impact Statement on the two above projects will act just the opposite. Maine has a difficult time competing for new business as it is without driving up the costs of existing business (30% in this particular case) through tougher regulations.

The American people have asked for change and believe me that is coming across the board particularly in the area of Federal Regulations. We must begin to give projects like this some latitude to provide a more significant economic and environmental benefit to the people of this state.

l oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agensies have done a good job of preserving and protecting Maine's environment and are fully apable of continuing that effort. FERC's intrusion into this traditional area of State jurisdicion is unwarranted and unwanted.

'ery truly yours.

LT:tfm "Rep. Herbert E. Clark, District 140

erdi L. Tripp

E-401

Lois B. Cashell, Secretary Federal Energy Regulatory Commission Gical Sciences and AMALYSIS EDERA 825 North Capital Street, N.E. Washington, D.C. 20426	FILEO THE SECRETARY 3 PM 1:47 L ENERGY LATORY IISSION
RE: Ripogenus Hydroelectric Project (FERC No. 2572) - () Penobscot Mills Hydroelectric Project (FERC No. 245)	s)-009

Gentlemen:

I write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maime's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Very truly yours,

1/ hall hangt

Robert L. Tufts State Representative District 107 P. O. Box 124 Stockton Springs, Maine 04981



Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: RIPOGENUS HYDROELECTRIC PROJECT (FERC NO. 2572) PENOBSCOT MILLS HYDROELECTRIC PROJECT (FERC NO. 2458) *

Dear Mr. Cashell and members of FERC:

I am writing to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the 117th Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market. The applicant, Great Northern Paper, is an important part of the paper industry in this state.

I believe that continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of Maine.

I oppose the conditions contained in the Draft Environmental Impact Statement which will increase the applicant's annual cost by an estimated 30 percent. I believe such an icrease will impair Great Northern's ability to modernize its facilities, pay decent wages and compete in today's global markets. I especially oppose the staff proposal that Great Northern Paper take over control of regulating shoreline development from the state. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of state jurisdiction is unwarranted and unneeded.

I strongly suggest that you do not include conditions for this project that would increase the cost of doing business in the State of Maine.

Sincereh Marc J. Vique State Representative

001

District 107-40917inslow

1

667



Rep. Julie Winn RR #4, Box 570 Glenburn, Maine 04401 Tel: 207-884-7836 HOUSE OF REPRESENTATIVES





January 27, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458) - 009

Gentlemen:

I write at this time to submit written comments concerning the Draft Environmental Impact Statement issued for the captioned projects.

As a member of the Maine Legislature, I am concerned about the ability of our paper industry to compete in an increasingly global market and the applicant, Great Northern Paper, is an important part of the paper industry in this State.

In my opinion, continued operation of the projects as proposed by the applicant will provide significant economic and environmental benefits to the people of this State.

I oppose the conditions contained in the Draft which will increase the applicants annual cost by an estimated 30%. I especially oppose the staff proposal that Great Northern take over from the State control of regulating shoreline development. Maine's land use agencies have done a good job of preserving and protecting Maine's environment and are fully capable of continuing that effort. FERC's intrusion into this traditional area of State jurisdiction is unwarranted and unneeded.

Very truly yours,

Lele.

Julie Winn State Representative

District 122 Glenburn, Kenduskeag, part of Levant

COMMENTS TO CONCERNED CITIZENS CONCERNING LAND USE ON PENOBSCOT/RIPOGENUS MILLS DEIS

Group - B

We received comments from many concerned citizens regarding our recommended land use recommendations in the Ripogenus/Penobscot Mills DEIS. Our response to these comments are provided here and the comments follow.

Comment noted. The staff's recommendations (see section 5.3.4) provide measures to protect shoreland resources within the project areas while considering existing land use regulatory controls. The staff's final assessment and recommendations consider comments received during the DEIS comment period, GNP's proposed conservation easements for the Ripogenus Project area, updated land valuation information, and further assessment of LURC land use regulations and resource protection measures (see section 4.9).

1/25/95 Lois Cashell 005 (DPY Dear Us Cashell -**.** Thank you for on the Ripoguns Prog. EDERADENER Project Wills. FEB 2 4 1955 COMMISSION This is a wonderful opportunity to protect some of the most wonderful lake vistas in the U.S. at almost not cost . to taxpagers. A 200' buffer seems really not which easier to achieve than 500' and 500' feet would de thejde right. Hease, shease support the first alternative and trow that it nears so much to those of no who have on wild Maine lates Best to you, Wet lignen Alex Agnew Box 4733 DTS Portland, ME 04112-4733



95 FEB 27 PM 12: 16

FILED OFFICE OF THE SECRETARY

> FEDERAL ENERGY REGULATORY COMMISSION

January 23, 1995

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

> Re: Draft Environmental Impact Statement -- Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458)

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, I am concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500' is the minimum that has been used on other buffer zones in the immediate area. A 500' wide buffer should not be much more costly to Bowater/Great Northern than a 200' buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch.

Second, I an concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours,

Ann Remark Bechen





 $\mathbf{\hat{y}}$

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, I am concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500' is the minimum that has been used on other buffer zones in the immediate area. A 500' wide buffer should not be much more costly to Bowater/Great Northern than a 200' buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours,

H. M. Bliss

VFPRNI FFR 2 1 15.5

ECOLOGICAL SCIENCES AND ANALYSIS

Dear Secretary Cashel, = - I am swriting in reference to the Draft Enouron! mental Impact Statement - Repogenus Project (FERC# 257 + Pensbacat Mills Project (FERC # 245F) - 009 I am sorry that I was not able to attend the hearing on Jan 25, & am not aware of ite onterme-I am pleased to see FERC's active , fositive (I fail) response to this request, and would like to support The hufler zone of soo! My faling (atthough not scientific) in that this is a beautiful and important area. I that it needs all the protection it can yet My husband & I have been in the area at various times, I while we have in sinne, and in a relatively uncposti east. The northern part calle you to it in a long

م م م م and pressing side Come server and arrive -faities concerned co - spected, compromised & c Ap with a workable plan. Orwater - sthe insuch was it should look to incidences like this for guidance and enlighterment, and perhaps with you help they will . Thank you for your time . ` Amerely, Margery & Blander

== 2205 Peter J. Beehn **10 Lobster Cove Road** Monhegan Island, ME 04852 OFFICE OF THE SECRETAR January 23, 1995 95 FEB -3 PH 1: 39

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

FEDERAL ENERGY REGULATORY COMMISSION

Re: Draft Environmental Impact Statement -- Ripogenus project (FERC No. 2572) and Penobscot Mills project (FERC No. 2458)

Dear Secretary Cashell:

As a citizen of Maine, I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great northern.

Although I am pleased by FERC's recognition of the need to protect shoreland. I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First. I am concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500, is the minimum that has been used on other buffer zones in the immediate area. A 500, wide buffer should not be much more costly to Bowater/Great Northern than a 200' buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive

Boehmer to Cashell Page 2 of 2 Jan 23, 1995

but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500 buffers in the West Branch.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the ahoreland protection plan set forth in Alternative $1 - 500^{\circ}$ buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment, especially as the State affected is my state.

Sincerer Peter J/Boehmer

OFFICE OF THE SECRETARY		Copy
SULATORY COMMISSION	January 23,	1995
Lois D. Cashell, Secretary Federal Energy Regulatory Com 825 North Capitol Street, NE Washington, DC 20426	aission	FEB 2 4 19-5
Re: Draft Environmental	Impact Statement	Ripogenus

Re: Draft Environmental Impact Statement -- Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458) ~ 00 %

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, I am concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500' is the minimum that has been used on other buffer zones in the ismediate area. A 500' wide buffer should not be much more costly to Bowater/Great Northern than a 200" buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours, Cuitin Bank

ustice

ME 04421

FILED OFFICE OF THE SECONETAR R.F.D.#3 Box 39 Libby Avenue 95 FEB 27 MIL: 39 Gorham, Maine 04038 February 14, 1995

DUR

Lois Cashell FLOERAL ENERGY REGULATORY Secretary, Federal Energy Regulatory Continuission 825 North Capitol Street, NE Washington, D.C. 20426

and the second second

Dear Secretary Cashell:

I am writing in regard to the draft environmental impact statement (DEIS) issued by FERC on the Ripogenus (FERC #2572) and Penobscot Mills (FERC #2458) – 009Hydroelectric Projects in Maine. I did not attend the January 25, 1995 public hearing in Millinocket, but I would like to express my opinions now. Thank you for hearing my comments and for preparing an environmental impact statement on these projects.

My husband and I have monitored a lake in Maine on which my family owns property for about twenty years. We have tested for several water quality parameters and have worked under the auspices of the Maine D.E.P. We hold a strong belief that the lakes of Maine should be protected as much as possible considering all the pressures from human intrusion so that the best water quality possible will be preserved for as long as possible. It makes good sense to require shoreland protection zones on the lakes, ponds, and streams in the vicinity of The Ripogenus and Penobscot Mills project areas. Considering the fact that the West Branch of the Penobscot is associated with one of the most significant watersheds in Maine, protection of its lakes, ponds, rivers, and streams is crucial.

I appreciate hearing that FERC believes that these shorelines should be protected and also that easements should be installed on all land. Great Northern could procure land which the company does not now own by means of a conservation easement trust. I would like to see the proposed buffer strips increased from 200' to 500'. The cost should not be excessive since it is not shorefront, and the 500' buffer

E-418

zone would be the same as those in other areas of the West Branch of the Penobscot as well as in the Allagash.

My belief is that along with the applicant's right to generate power should also come the responsibility to procure shoreland easement with a 500' buffer zone so that the natural beauty will be preserved for future generations. Thank you for your concerned involvement in this process.

.

Sincerely,

Eilen D. Burnell



Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

> Re: Draft Environmental Impact Statement -- Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458) - 009

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, I am concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500' is the minimum that has been used on other buffer zones in the immediate area. A 500' wide buffer should not be much more costly to Bowater/Great Northern than a 200' buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours, Of Calidual HCSO Box 33 Penalescat, Mo. 04476

P.S. Two additional comments

() I saw historical to converg friends rature isontancies for trips on niness they expected to please, only to find these hifter genes so painfully, minibly closete the water that the character of the area was destrayed. (c) Our sheeting company reasond rates partly because its anotamers used in a manger abatricity. I'm sure Barrate l'Sheet historican can doel with it if the score buffer zero in fact is a handship for them. It is infinitely more important to protect our emission. Tail integrate than to protect cours emission. Tail

		1 AT			The North Woods				COP Salit
	R	E	S	T	0	R	E :		ૼૼૼ૱
	P.O	. Box 44	ю, Con	cord, N	1A 0174:	2 • 508-2	87-0320		ALIA CA
				F	ebruary 2	2, 1995	م با معنی ا	د مدید مدانو مارسانی	507 PC
Lois D. Cashell							FEE 2	4.85	-
Federal Energy I 825 North Capit	Regula al S tre	tory Co et. NE	mmissio	n		101	OLCGICAL SCIE	HCES AND ANA	LYSIS
Washington, DO	2042	26					-		

Dear Secretary Cashell:

These comments are submitted by RESTORE: The North Woods on the draft environmental impact statement issued by Federal Energy Regulatory Commission (FERC) on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

We are pleased that FERC has prepared an environmental impact statement (EIS) for these projects. Although we have not participated in the formal process of relicensing the dams, we do care deeply about the West Branch watershed and its future. We were unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7:00 pm public hearing and therefore want to make our comments to you in writing.

We strongly support the need to require shoreland protection zones on the lakes, ponds, and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers, and streams for aesthetic, ecological, and recreational values is critically important. We are pleased that FERC recognizes the need to protect these shorelines.

Although we are encouraged by FERC's recognition of the need to protect shoreland, we are concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, we are concerned that buffers of only 200 feet are not sufficient to achieve the goals of aesthetic, recreational, and ecological protection. We believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500 feet is the minimum that has been used on other buffer zones in the immediate area. A 500 foot wide buffer should not be much more costly than a 200 foot buffer; the most expensive land is that immediately fronting on water. The land behind the shorefront is much less

1

"I wish to speak a word for Nature. ..." -Henry David Thoreau

expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500 foor buffers in the West Brinch

Presently, the U.S. Eish and Wildlife Service is reviewing a petition to list the North American wood turtle (*Clemmy insculpts*) as a threatened species under the Endangered Species Act. Wood turtles, which should be found throughout Maine, are know to travel up to 600 meters from a water source. A buffer zone of 200 feet or less will not protect the required habitat of this imperiled species.

We are also concerned that the option to allow Great Northern to install a shoreland management plan at Penobscot Mills with buffers narrower than 200 feet could be used to defeat the purpose of the buffer zones. We believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500 feet, subject to change only on a very strong demonstration that aesthetic, recreational, and ecological values will be adequately safeguarded.

We urge FERC to adopt the recommended buffer zone proposals as modified above. We believe that FERC should adopt the shoreland protection plan set for in Alternative 1—500 foot buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in the naturally significant Maine woods.

Thank you for the opportunity to comment.

Sincerely,

Anthe Col

David N. Carle Associate Executive Director

RE Dreft Environmental Impact Statement - Rigo genus Project (FER GUENOUS 2572) and Pendrocot Millo Project (FER C NO. 2458 UFFICE 21 PM 3:53 OG H.C.R.32 BOX 253 VERSAR INIC. OC Smallpoint Rd. Sebasco Estates, Me February 14, 1995

Lear persons Lois Cashe

.....

L

ECOLOGICAL SCIENCES AND ---

Thank you for your wise requirement attached to Bowater's dam relicensing process that Bowater establish a conservation easement along 110 miles of lake and pond shoreline in the West Branch of the Penobscot drainage. I have canoed on Chesuncook Lake and have worked for four years to help plan with responsible stewardship for future resource extraction and conservation and human development in the Northern Forest. I believe your action was both timely and appropriate as the West Branch is an area of statewide significance which is threatened by sale for recreational dwellings.

(node you will increase the size of the conservation buffer to 500 feet. to maintain a wide, viable wildlife corridor and to prevent erosion cause by logging. 150 acre clearcuts at the edge of a conservation corrigor can cause wind restruction of adjacent trees in the lake oufter, have soil resulting trom crearcute low feet from the rake can channer water during neavy storms into the outter cone, gouging out revines and destroying the soul texture and quality within the suffer. Here species of migratory songoings will not nest in a 200 foot wide morested area, but need 500 contiguous meet of morest to provide the quality of habit they need for reproduction. Many of these songoinds, including several species of warblers and torushes, eat spruce budworm, and serve to reduce forest lamage from this destructive insect.

I node that you will ansure that current leasengigens on the lakes will have grandtathered leases, that they will have the right to renew. I also hope you will require Great Northern (Bowater) to ourchase from willing sellers, lands deeded to install the easements.

These conservation easements, if enlarged to 500 feet. will protect water quality. Wighte mabitat, recreation and aesthetics, are a mandatory and necessary part of relicensing, returning to the public good stewardship for Bowater's theritable use of public water for hydropower.

Sincerely. Mancy B. Chandler



FILED OFFICE OF THE SECRETARY

95 FEB 27 AM 11: 16

REGULATORY COMMISSION

Colby Environmental Council Colby College Waterville, ME 04901

_005

February 15, 1995

Lois Cashell Secretary, Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

,009

RE: Draft Environmental Impact Statement -Kipogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458)

Dear Secretary Cashell:

These comments are submitted by the Colby Environmental Council on the draft environmental impact statement (DEIS) issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although we have not participated in the formal process of relicensing the dams, we do care deeply about he West Branch watershed and its future. We were unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 PM evening public hearing and therefore want to make our comments to you in writing.

We strongly support the need to require shoreland protection zones on the lakes, ponds, and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the state of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological, and recreational values is critically important. We are pleased that FERC recognizes the need to protect these shorelines. We are equally glad to see that FERC understands that the goal of protecting these resources can only be achieved on the Ripogenus impoundment if easement are installed on all land, including those not owned by Great Northern. We believe that these lands should be purchased on a willing-seller basis and that Great Northern should be obligated to establish a conservation easement trust fund to purchase presently unavailable lands as they become available.

Although we are pleased by FERC's recognition of the need to protect shoreland, we are concerned with several aspects of the approach recommended in the DEIS.

First, buffers of only 200 feet are not sufficient to achieve the goals of aesthetic, recreational, and ecological protection stated in the DEIS. The proposed buffer zones should be expanded to a width of 500 feet. A 500' wide buffer should not be much more costly than a 200' buffer; the most expensive land is that fronting immediately on the water. The land behind the

shorefront is much less expensive because it is less desirous for development, although it has great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch of the Penobscot River and the Allagash. Also, we believe that the buffer width should be 500' throughout the Ripogenus and Penobscot Mills Project area.

We also agree with FERC that any existing commercial and private leases as identified in the DEIS should continue to be honored and renewable, subject to LURC zoning regulations.

We urge FERC to adopt the recommended buffer zone proposals and modified above. Adequate shoreland protection zones are a routine and necessary aspect of ownership of a valuable hydropower license. Then need for shoreland protection is particularly keen in the Northern Forest region, as the DEIS acknowledges. The public is entitled to assurances that its use and enjoyment of the natural character of this region, like the applicant's right to generate power, will be secure for the term of this license. Shoreland easements are a critical element of that assurance.

Thank you for the opportunity to comment.

Sincerely,

Joshua B. Illoson Friter & Wilson Jennife Lemnaco 4 Hoynne Rogers Amme 21/ C. Connell Ethica thompson Fauren Vitiano Die Final Time Robinson

Loio R. Cashell, Socratary Federal Energy Regulating Communing 825 houth Capital Street NE Washington, D.C. 20426 57 J.C. 500 Ξ

VERSAR, INC. it with and loinga FFB 2 A 1995 1-2-1-94 ECOLOGICAL SCIENCES AND ANALYSIS EGRETARY Streibbar Secretar Regulation Com 825 month Council Street, NE Dashington; DC 20426 622 Dact Cronomental Impart Startend Stipsgerine prized CZehcz 2572-002 penderent mills project 2458-009 Dear Secretary Cashell _ I would like to comment on the droft EIS isome by Febcon the above prejects in maine I was unable to attend the milinochet public hearing on Jan. 25th So would like to make my commuter Writing. my wife & I are life - long resuccents of mains I use the west granch region extensively far recreationof Junpours. I are placed that FERC'S

red secret cere a second has about t especial internets of this protent th tesiche Dierra - liland to hi o only often th _ lasure - expratio - I dequite protection of the she lime should be for Barreter Valt use of the States - maturel sesaue - Thank for your cond Segman P Richard Z Colo - Xaa la لو_ AGT BOY 740 the, me 04434

20 February 1995

5

Lois Cashell

OFFICE OF THE SECRETARY

95 FEB 27 AMII: 16

FEDERAL ENERGY REGULATORY COMMISSION

Secretary, Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

RE: Draft Environmental Impact Statement-Ripogenus Project 009 (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458)

Dear Secretary Cashell;

These comments are submitted by myself on the draft environmental impact statement (DEIS) issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 PM public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds, and streams in the Ripogenus and Penobscot Mills projects areas. The West Branch is one of the most significant watersheds in the state of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers, and streams for aesthetic, ecological, and recreation values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved on the Ripogenus impoundment if easements are installed on all land, including those not owned by Great Northern. I believe that these lands should be purchased on a willing seller basis and that Great Northern should be obligated to establish a conservation easement trust fund to purchase presently unavailable lands as they become available.

Although I am pleased by FERC's recognition of the need to protect shoreline, I am concerned with several aspects of the approach recommended in the DEIS.

First, buffers of only 200 feet are not sufficient to achieve the goals of aesthetic, recreational, and ecological protection stated in the DEIS. The proposed buffer zones should be expanded to a width of 500 feet. A 500 foot wide buffer should not be much more costly that a 200 foot buffer; the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive because it is less desirous for development, although it has great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500 foot buffers in the West Branch of the Penobscot River and the Allagash. Also, I believe that the buffer width should be 500 feet throughout the Ripogenus and Penobscot Mills Project area.

I also agree with FERC that any existing commercial and private leases as identified in the DEIS should continue to be honored and renewable, subject to LURC zoning regulations.

I urge FERC to adopt the recommended buffer zone proposals as modified above. Adequate shoreland protection zones are a routine and necessary aspect of ownership of a valuable hydropower license. The need for shoreland protection is particularly keen in the Northern Forest region, as the DEIS acknowledges. The public is entitled to assurances that its use and enjoyment of the natural character of this region, like applicants right to generate power, will be secure for the term of the license. Shoreland casements are a critical element of the assurance.

· • • •

Thank you for the opportunity to comment.

Sincerely yours,

Ste lon

Steven Corman 104 Main Street Apt. 2B Orono, ME 04473



Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and racreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, I am concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500' is the minimum that has been used on other buffer zones in the immediate area. A 500' wide buffer should not be much more costly to Bowater/Great Northern than a 200' buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours,

fundy of Duris

LIVINIT B. Javis PO BON 278 JAMANSLOTTA, ME 04543






January 23, 1995

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

> Re: Draft Environmental Impact Statement -- Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458).

Dear Secretary Cashell:

/

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sharow Drake Key RV Park - 53 W Marathon, FI 33050

COG

FILED OFFICE OF THE SECRETARY

RFD#3 Boy227

VFRSAR

1001004024

autom, Me 04210

February 10, 1995

95 FEB 21 AH 10: 40

FEDERAL ENERGY Lois Cashell Secretary, Federal Energy Regulatory Commissi NE Street NE Street NE 825 North Capitol Street, NE Washington, DC 20426

RE: Draft Environmental Impact Statement - Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458)

005

Dear Secretary Cashell,

These comments are submitted by [organization/individual] on the draft environmental impact statement (DEIS) issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although we have not participated in the formal process of relicensing the dams, we do care deeply about the West Branch watershed and its future. We were unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 PM evening public hearing and therefore want to make our comments to you in writing.

We strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project ares. The West Branch is one of the most significant watershed in the state of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers, and streams for aesthetic, ecological, and recreation values is critically important. We are pleased that FERC recognizes the need to protect these shorelines. We are equally glad to see that FERC understands that the goal of protecting these resources can only be achieved on the Ripogenus impoundment if easements are installed on all land, including those not owned by Great Northern. I believe that these lands should be purchased on a willing seller basis and that Great Northern should be obligated to establish a conservation easement trust fund to purchase presently unavailable lands as they become available.

Although We are pleased by FERC's recognition of the need to protect shoreland, we are concerned with several aspects of the approach recommended in the DEIS.

First, buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational, and ecological protection stated in the DEIS. The proposed buffer zones should be expanded to a width of 500' feet. A 500' wide

buffer should not be much more costly than a 200' buffer; the most expensive land is that <u>Immediately fronting</u> on <u>water</u>. The land benind the shorefront is much less expensive because it is less desirous for development, although it has great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch of the Penobscot River and the Allagash. Also, we believe that the <u>buffer</u> width should be 500' throughout the Ripogenus and Penobscot Mill's Project area.

We also agree with FERC that any existing commercial and private leases as identified in the DEIS should continue to be honored and renewable, subject to LURC zoning regulations.

We urge FERC to adopt the recommended buffer zone proposals as modified above. Adequate shoreland protections zones are a routine and hecessary aspect of ownership of a valuable hydropower license. The need for shoreland protection is particularly keen in the Northern Forest region, as the DEIS acknowledges. The public is entitled to assurances that its use and enjoyment of the natural character of this region, like the applicants right to generate power, will be secure for the term of the license. Shoreland easements are a critical element of that assurance.

Thank you for the opportunity to comment.

R Stephen Drane, Ph. P

FILED OFFICE OF THE SECRETARY 95 FEB -9 AN 10: 06	COPY
NERGY	
Commission	January 23, 1995
	FED Z L T S
Lois D. Cashell, Secretary Federal Energy Regulatory Commiss: 825 North Capitol Street, NE Washington, DC 20426	Lon ECOLOGICAL SOIENUSS
Re: Draft Environmental Impa Project (FERC No. 2572) (FERC No. 2458).	and Penobscot Mills Project

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Frances E-Dunn Box 22 Round Pond, ME. 04564

VERSAR, INC. T M. EWING 2 1 1995 Neal:St. FF .3 PH 1:57 DEFICE 177 PORTLAND, MAINE 04102 ECOLOGICAL SCIENCES AND ANALYS (207) 774-8009 tois Cashell, Secret F.E.R.C. 825 N. Capitol St. NE UL С. Washington, D.C. 20426

Re: Environment Impact Statement - Ripogenus Project and Penobscot Mills Project

Dear Secretary Cashell,

 \mathcal{V}

Although I do not know all the conditions and all the problems concerning the projects at Ripogenus and Penobscot Mills, I feel very strongly about the need for shoreline protection and I have confidence in the expertise of the C nservation Law Foundation. My own experience is limited to the problems of shoreline protection in the southern dounties of Maine.

I applaud FERC 's attempt to require adequate buffer zones, and I hope that you will withstand any pressure to defeat or reduce the proposed standards.

Sincerely.

836 River Bozel Brunewich ME 04011 : 🖅 🎽 January 26, 1995 Lois D. Cashell, Secretary Federal Energy Regulatory Commission 95 JAN 31 AM 11:51 825 North Capitol Street, NE Washington, D.C. 20426 Re: Oraft Environmental Impact Statement-Dipogenue Project (FERC No. 2572) and Penolscot Mille Project (FERC no. 2458)-009 Dear Secretary Cashell, as a member of AMC, Maine auditon and Conservation Law Trandation, I am submitting these commente on the draft environmental impact statement relative to the chose projects, since I was unable to be present at the January 25 hearing in Millinocket. My concern is for the shoreland protection zones on lakes, ponde and streams in the above project, areas which are so important to maine future because of high aesthetic and recreational value as well as for the enhancement of biodiversity in Maine. I am grateful that FERC has recognized the importance of such protection. I would urge that FERC require the fuffer zones of Alternative (1) so that really adequate and effective protection is maintained for the West Branch. Included should be the requirement that Bowater/GNP obtain easements for all of the 500 foot buffer zones, not solely on land it owns. For the Penotscot Mills project, if the proposed option of a Shoreland management Plan prepared by GNP is allowed, FERC needs to insist that minimum standards are mainteined including the full 500 feet of fuffer zone, subject to modification only under strict requirements for safeguarding recteational aesthetic and ecological resources. I have been impressed by the hydro licensing

process as it affected me personally, since our home adjoins the property along the andress correction of a fairly the Peretsest Dam I is located. Construction of a fairly alattrate canse-launching facility below the dam after extended consultation with local officiale and with reighbors, demonstrated FERC's commitment to and concern for recreational values and its fleribility in planning for enhancement. Thank you for soliciting and considering my commente on the West Branch projects. Sincerely Virginia Vi Hammond



Dear Secretary Casheli:

You may wonder why I am writing you, given the address on my letterhead. However, my family has been tied to the town of Castine for generations. I try to spend as much time as I can in the family homestead north of the town on the Penobscot River. Over the years I have become somewhat familiar with the upper reaches of the river.

Unfortunately, I was unable to make the trek to Millinocket last week to comment on the Draft Environmental Impact Statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine. Therefore, I am commenting by letter.

Your commission certainly deserves credit for preparing an environmental impact statement on these projects. It is a demanding process and, to its credit, FERC did it.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its valuable lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC also recognizes the need to protect the shorelines involved. It is heartening to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased that FERC recognizes the need for shore protection, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, it appears to me that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet, a width that is the minimum used for other buffer zones in the immediate area. A 500' wide buffer should not be much more costly to

Bowater/Great Northern than a 200' buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch.

Second, it appears that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', reduced only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely.

Francis W. Hatch

FWH/jp



BOOTH HEMINGWAY 94 PEPPERRELL RD. P.O. BOX 42 MAINE KITTERY PT.03905-0042

OFFICE OF THE SECRETARY 95 FEB - 3 PM 1: 38 FEDERAL ENERGY REGULATORY COMMISSION

January 23, 1995

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

> Re: Draft Environmental Impact Statement -- Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458)

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours,

BOOTH HEMINGWAY 94 PEPPERRELL RD. P.O. BOX 42 MAINE KITTERY PT. 1360%



Dear Secretary Cashell:

I have not participated in the formal process of relicensing these dams, but I do care deeply about the West Branch watershed and its future. I am unable to travel to Millinocket tonight to comment on the draft at the public hearing and therefore hope I may make my comments in writing.

The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast, and the protection of its shorelines is critically important. I'm glad to see that FERC understands that protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

However, I'm concerned with a couple of aspects of the approach recommended in Alternative 2 of the draft EIS.

First, buffers of only 200' are not sufficient to achieve any meaningful goals of aesthetic, recreational and ecological protection. A width of 500' would be much more reasonable for purposes of protection and is the minimum that has been used on other buffer zones in the immediate area. There is not much point in requiring Bowater/Great Northern to protect only the first 200' (the most valuable) if that doesn't do the job, and it wouldn't. Let's face it, as a practical matter in most stretches of shoreline, 200' of "protection" is so transparent aesthetically, or for recreational consideration, and so meaningless biologically that it serves little purpose. If we are going to bother, let's do it right. Ms. Lois D. Cashell

Similarly, with respect to the Shoreland Management Plan at Penobscot Mills, FERC should set minimum guidelines for any management plan, and that plan should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

2

It seems to me that adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Sincerely,

Horace A. Hildreth

HAH:bpl



January 23, 1995

AN II:

Auvi Jafi

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

> Draft Environmental Impact Statement -- Ripogenus Re: Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458) - 004

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Howard E . Jones

COMMITTER /4



Julia Khorana, Chair 3 Birch Hill Road Stow, MA 01775

February 7, 1995

Norman Refin, Treasurer 1 Ancher Way Newbury, MA 01931

Re: Draft Environmental Impact Statement - Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458)

Dear Secretary Cashell,

I am writing on behalf of the Interchapter Canoe Committee of the Appalachian Mountain Club.

ALACHIAN

INTERCHAPTER

I applaud FERC for all their work in preparing an environmental impact statement on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine. I have enjoyed many river trips in the area, and hopefully I will in the future, thus I would like to submit the following comments.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenous and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the state of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers, and streams for aesthetic, ecological, and recreational values is critically important. I am pleased that FERC recognized the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved on the Ripogenus impoundment if easements are installed on all land, including those not owned by Great Northern. I believe that these lands should be purchased on a willing seller basis and Great Northern should establish a conservation easement trust fund to be able to purchase the lands, should they become available.

Although we are pleased by FERC's recognition of the need to protect shoreland, we are concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, we are concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational, and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500' is the minimum that has been used on other buffer zones in the immediate area. A 500' wide buffer should not be much more costly than a 200' buffer; the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch of the Penobscot River and the Allagash.

Second, we are concerned that the option to allow Great Northern to install a Shoreland Managemen Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines subject to change only on a very strong demonstration that aesthetic, recreational, and ecological resources will be adequately safeguarded.

We also agree with FERC that any existing commercial and private leases as identified in the EIS should continue to be honored and renewable under the conditions of LURC zoning regulations

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set for in Alternative 1 - 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

aulic Khorama

Julia Khorana Chair, ICC

VERSAR. \$10329 Dearlow Hell Dear Secretary Case of the Street leter Button me 04093 law writing about the draft superimmented mant statement issued by Earch Gon Fit 1. ER ip ogener and Venaliset wells Hydrocleating Vameta marine We could not travel to willerecht on fan 25 1995 to comment and do want to make commente to The West Branch is a most significant water sleep. -----lede must protest the lales, fonde, rever and streams . " Caremente must be installed on all land, including these not among great Northern . . Buffus of 200' are not sufferent to acheve the gaale. The buffer stupic should be 500' The same as ather 500' buffers on The West Branch of the Senderent and the allegash. We agree will FERC that existing leave shareld continue. The public is entitled to use and enjoyment of the natural character of thes region, in return for The applecants using an water to generate forme. _ J hand you for The opportunity to comment Emert and Corsere Kenney-

E-454

and a state of the second ERNEST A. KINNEY ;-やてつ ž 4 Frederick Ľ 5 アメロ Ť 12 12 395 PM ç 20426 2 5 ۲,



January 23, 1995

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

Re: Draft Environmental Impact Statement -- Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458) - 00

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Charles E. Kitin

11 h. Shore Dr -Treeport me 04032

FILED OFFICE OF THE SECRETARY

95 FEB 27 AMII: 41

REGULATORY

205

February 16, 1995

Lois Cashell Secretary, Federal Energy Regulatory Commission 825 North Capitol St., NE Washington, D.C. 20426

Dear Secretary Cashell,

This letter is in reference to the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine (FERC No. 2572 and FERC No. 2458). Although I was not able to travel to Millinocket on January 25, 1995 to provide public comment, I am very concerned about this issue and offer my written comments.

The natural resources of Maine are very dear to me and shoreland protection in the Ripogenus and Penobscot Mills project areas is critical. I am very pleased that FERC also recognizes the importance of the areas and the need for protection. My concern is with the buffer areas being proposed. I feel that 500' buffer widths will provide the protection which is necessary rather than the 200' buffer. This would place these buffers on a par with other 500' buffers in the West Branch of the Penobscot River and the Allagash and would not add substantially to the cost of the buffer zones.

I applaud you for the fine work which you have accomplished to date on the draft environmental statement and urge you to adopt the modified buffer width.

Thank you for the opportunity to comment.

Locus in Same

Valarie C. Lamont 34 Chenery St. Portland, ME 04101

7.7.**225** - A. A. A. A. T 7 000

OFFICE OF THE SECRETARY 95 MAR -3 PN 3= 19 REDERAL ENERGY REGULATORY COMMISSION

January 23, 1995

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

> Draft Environmental Impact Statement -- Ripogenus Re: Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458) 009 005

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing. 1.5 ×02 .5 Å

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and the streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I an equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the appreach recommended in Alternative 2 of the draft EIS.

First, I am concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500' is the minimum that has been used on other buffer zones in the

E-459

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region. na se anti e se se se anti e a

Thank you for the opportunity to comment.

Sincerely yours,

Ivunda lang

· • · · · · ·

LUCINDA LANG P.O. **BOX 28**2 TENANTS HARBOR, ME 04880



Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

> Re: Draft Environmental Impact Statement -- Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458)-()04

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

month.B. Hayth.

GRENVILLE B. LLOYD JR. BOX 55 SCUTHWEST HARBOR, MAINE 04679

VEPSAR. INC. Kelly McClintock 95 FEB 16 AM 9: 51 65 Woodland Street COLO NOME SU Sherborn, MA 01770 FEDERAL ENERGY 2/10/95 Join Carhell FERC 825 N Capital St ME Marlington DC 20421 Re: FERC No. 2572 and No. 2450-009 I have just read about your proceedings on the allest Branch of the Renobicot. I have canced the river all my life, and care deeply that it be preservedparticularly shouline buggers. I an glod you are considering buggers - but 200' is, in my opinion, nouhere near encrique. 200 yards Mean - protest the fabricon area! is necessary. Thank you for considering these connect. Sincerel, MM Mellintone



JANUARY 25, 1995 -

Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

UFFICE OF THE SECRETARY 95 jan 30 an 10: 38 FEDERAL EN

Re: Ripogenus Hydroelectric Project (FERC No. 2572) COMMON Penobscot Mills Hydroelectric Project (FERC No. 2458 SION

Gentlemen:

I know how difficult it is for the Federal Government to get anything done right and I am one of many who expressed this sentiment in the last election.

Regarding the above Projects on which you recently issued a draft Environmental Impact Statement, come on! Do some research with the locals and make your Impact study clear and understandable so you won't get swamped with letters like this.

There are people like Daniel Sosland of the Conservation Law Foundation saying "owners of camps are frantic for no reason...the setbacks would only apply to future development... existing camps would not have to be pushed back 200 feet from the shore; camp owners could add a room; and permits would come only from LURC". Sounds reasonable to me. If this is so, why didn't you spell it out in your draft? I think one this is done, the local "panic" will ease up a little.

I am unable to attend the meeting tonight in Millinocket that addresses this issue so felt compelled to write and at least show support for my fellow camp owners on Ambajejus Lake.

Sincerely,

Paul J. McPheters 15 Prospect Street Brewer, ME 04412

GNP Lease No. 3165 Lot 234 T1 R9 WELS PISCATAQUIS



COPY

January 23, 1995

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

> Re: Draft Environmental Impact Statement -- Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458)

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

Second, I am concerned that the option to allow Great Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer sones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours, Konn Mo RRI BOK 150 White Field Me 04353 207549 7122 Fox 207589 7172

•	COBY
HUGH MONT	IGOMERY
P. O. BC	
PHILLIPS, MAINE	: 04966 - 1509
TELEPHONE 207 - 639 - 2881	FILED
Ja	inuary 23, 1995
Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE	
Re: Draft Environmental Tanat	
Project (FERC No. 2572) and (FERC No. 2458) - 00	Penobscot Mills Project

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.



Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours, Heigh Montgoming



January 24, 1995

RT 129

130

LES

مہ چاچ چین

5

Lois D. Cashell, Secretary FERC 825 North Capitol Street, NE Washington, D. C. 20426

GFLB-1 PN 12:30 Sec EDERAL ENERGY DMMISSION ODE

Re: Draft DEIS, Ripogenus Project (2572) and Penabscot Mills Project (2459) - 009

Ry dear Secretary Cashell:

Because we are unable to attend tomorrow's meating in Millinocket, Mrs. Myers and I wish to convey to you by this letter our concerns about the two projects indicated above.

Everyone seems agreed on the need for shoreland protection of these magnificent watersheds, and that is all to the good, but in the matter of degree of protection, we would urge the adoption of Alternative 1, requiring 500' buffers. That footage should be the minimum protection, except possibly when awkward topography mandatas shading it five or ten feet. In the event of heavy rains or accelerated snowmelt, Alternative 1 is better protection from erosion and siltation.

The discussion of dollars leaves us bemused. If we use 110 miles of shore, Alternative 2, and an average \$3,000,000 cost, the shoreside buffer occupies 2,650 acres, thus valued at \$1,100 per acre. The additional 300' for Alternative 1 is claimed at a cost of \$6,000,000 for 4,000 acres, thus valued at 31,500 per acre. \$1,100 per acre for shoreland and \$1,500 per acre for back land seems beyond the canons of good sense; neither figure bears a sensible relationship to assessed galuation.

However the dollar confusion is resolved, Alternative 1 saems much the nore urgently better for every consideration in this case.

Faithfily your:

Equard Dyers: g



UFFICE OF THE SECRETARY 95 JAN 30 AM 10= 40 FEDERAL ENERGY REGULATORY COMMISSION

Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

> Draft Environmental Impact Statement -- Ripogenus Project (FERC No. 2572) and Penobscot Mills Project Re: (FERC No. 2458) 009

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.
immediate area. A 500' wide buffer should not be much more costly to Bowater/Great Northern than a 200' buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours,

Peter Nesser

Chairman NCN prinancial Corporation



Re: Draft EIS -- Ripogenus Project (FERC No. 2572)

Dear Secretary Cashell:

I am writing this letter in reference to the West Branch of the Penobscot river. I am a kayaker and enjoy kayaking on the West Branch very much. I enjoy the current situation, water levels and access. I hope in your relicensing you will keep in mind the interests of whitewater boaters.

I am also told that a buffer zone is to be established. This sounds like a good idea in that it keeps the river corridor aesthetically pleasing. It would be ideal to set a 500 foot buffer zone. I hope the agreement honors the existing leases along the river.

Thank you very much, and if you have never been to the Penobscot River you should. It is one of the most beautiful rivers in the country. The wildlife and scenery are spectacular.

Thank you for your help.

Sincerely,

J

George B. O'Connell

INC 28 5

 $\langle \overline{\langle} \rangle$

Lois Cashell SCOLOGICAL St. Secretary, Federal Energy Regulatory Commission 825 North Capitol St. NE Washington, DC 20426

Dear Secretary Cashell,

SFEB 21 PM 3: 56 We are writing this concerning the draft environmental impact statement on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine. Although we have not been able to participate in the formal process of relicensing the dams by attending the hearings, we both care very deeply about the future of the West Branch watershed. Because we were unable to travel to Millinocket on January 25 to comment on the draft at the 7 pm meeting, we want to make our comments in writing.

According to our understanding, the FERC has recognized the need to protect shoreline, and the prepared environmental impact statement addresses many clear and specific issues. The West Branch is certainly one of the most important (and most used!) watersheds in Maine. Its lakes, ponds, rivers and streams need protection for a host of reasons - aesthetic, ecological and recreational primarily. FERC certainly understands that this protection can only happen on the Ripogenus impoundment if easements are installed on all land, including what is not owned by Great Northern. We both think that these lands should be purchased on a willing seller basis and that Great Northern should be required to establish a conservation easement trust fund to purchase presently unavailable lands as they become available.

There are several aspects of FERC's approach, however, that we are concerned about. First, we think the proposed buffer zones should be expanded to 500' instead of the proposed 200'. We would like this buffer width to be required throughout the Ripogenus and Penobscot Mills project area. Acquiring the additional land should not prove very expensive since once one moves away from actual shore frontage, land is considerably cheaper.

In addition we would like to comment that conservation easements should be mandatory and necessary in any relicensing. We need the assurance that the public's use and enjoyment of the land is secure for the full term of the license, as is the applicant's right to generate power. The conservation easements of 500' should respect the rights of current leaseholders on the lakes by making the leases renewable and grandfathered.

Great Northern, which owns 80-85% of the shorelands under consideration, must be required, as suggested by the FERC environmental impact draft, to purchase from willing sellers lands it does not own in order to install the easements.

We realize the complexity of environmental regulation. We ourselves have enjoyed the West Branch area in particular for many years and purposes. Effective compromise is certainly the way to go, as FERC is demonstrating.

Thank you very much for the opportunity to present our views.

Sincerely yours,

۶.

 $\boldsymbol{\mathcal{M}}$

Lucia B. Owen

James B. Owen

P.O Box 613 Bethel, Me. 04217 207-928-2062

My 15, 1995



UFFICE OF THE SECRETARY 95 JAN 30 AN II: 57 FEDERAL ENERGY

REGULATORY Lois D. Calify ATORY Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, D.C. 20426

Dear Secretary Cashell:

January 26, 1995



I am writing to express my comments on the draft environmental impact statement issued by FERC on the Ripogenus Hydroelectric Project (FERC No. 2572) and the Penobscot Mills Hydroeclectric Project (FERC No. 2458) in Maine.

005

I strongly urge you to adopt Alternative 1, which would expand the shoreland buffer zone to 500', consistent with other existing easements on the West Branch of the Penobscot River and Chesuncook Lake. It is essential that FERC require Bowater/Great Northern Paper to obtain easements on land it doesn't own, and that any shoreland management plan prepared by Bowater sets minimum standards including the 500' buffer zone. I do not believe that a 500' buffer would be three times as expensive as a 200' buffer, since land behind the immediate shorefront is much less expensive.

The West Branch of the Penobscot River is one of the most significant watersheds in the State of Maine and indeed in the entire Northeastern United States. Protecting the ecology of this extraordinary watershed is extremely important. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in the northern forest region of Maine.

Sincerely,

Wayne Berson-

Wayne Persons RR 1 Box 348 Bradford, Maine 04410



Dear Secretary Cashell:

I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the public hearing and therefore want to make my comments to you in writing. I am pleased with Ferc's recognition to protect shoreline, but concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

I strongly support the need to require shoreline protection zones the lakes, ponds, and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the state of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers, and streams for ecological, aesthetic, and recreational values is critically important.

My Concerns:

First, I am concerned that a 200' buffer zone is not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the zones should be expanded to 500'-a width that is the minimum used on other buffer zones in the immediate area. This should not be more costly because the most expensive land is that immediately fronting on water. Buffers included in the hydropower licenses should be consistent with the other 500' buffers in the West Branch of the Penobscot and in the Allagash.

Second, I am concerned that allowing Great Northern the option to install a Shoreland Management Plan at Penobscot Mills with a buffer narrower than 200' could defeat the purpose of buffer zones. I believe that FERC must set minimum guidelines subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I also agree that any existing commercial and private leases as identified in the DEIS should continue to be honored and renewable under the conditions of LURC zoning regulations.

T where the adapt the short from the short for the standard the in

Alternative 1 -- 500' feet buffers. Adequate shoreline protection zones should be considered a routine cost of holding valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours,

Ì

Connie Peterson, Chair Connie Peterson, Chair Chapter Canoe Committee Chair, Berkshire Chapter Appalachian Mountain Club

	VERSAR. INC.]
	2.7 5	•
FCOL	OGICAL SCIENCES ACHER	

152 Bennoch Rd Orono, Marine D4473

Dear Secretary Castell, detrongly support the read for should protection on the nativery official by the Regiones on Perdecot Wills project areas been maine. I have conversed on these waters and give the wildeness character worth protecting waters and find the wildeness character worth protecting with at least a 200' buffers. a feel that if breat worther is the least they is allowed to continue to use our waters it is the least they and in return. Sincerety, Slam Pompe

U-2572-005 2458-009



Copy

Barbara & Ray Rappaport

33 Cleftstone Road Bar Harbor, ME. 04609

January 24, 1995

UFFICE OF THE SECRETARY

95 JAN 30 AH 10= 41

FEDERAL ENERGY REGULATORY COMMISSION

Ms Lois D. Cashell, Secretary Fed. Energy Regulatory Commission 825 N. Capitol St., NE Washington, DC 20426

These two projects have considerable impact for our whole state. I have no selfinterest or financial interest in either project, but I believe that FERC's environmental impact statements reflect much careful work. I am unable to attend the public hearing at Millinocket, but wish to have my comments on the record.

I support the need for shoreland protection zones on the lakes, ponds and streams of this northern area. I believe it is highly important to protect the watershed of the West Branch. Maine has little in the way of resources except for its natural environment, and so it is of utmost importance to protect the lakes, ponds, rivers and streams for ecological and recreational values.

The buffer zones in these areas are particularly important, and I would favor a width of 500 feet, the minimum that has been used on other buffer zones in the immediate area. This would maintain consistency also.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I favor the adoption of the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Shoreland protection zones should be a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for recording my comments.

÷

Sincerely yours,

. Eulan M. Kappigou

FILED OFFICE OF THE SECRETARY SFICE UT NORMAN SIMS 95 FEB 27 143 FLAT HILLS ROAD • AMHERST, MA 01002 • (413) 253-7922 THAL ENERY FAX: (413) 253-2636 . E-MAIL: SIMS@JOURN.UMASS.EDU REJULATISSION COMMISSION

Feb. 20, 1995

Lois Cashell Secretary Federal Energy Regulatory Commission 825 North Capitol Street N.E. Washington, D.C, 20426 わりり

Re: Draft Environmental Impact Statement: Ripogenus Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458) - 009

Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in the State of Maine.

I feel the EIS should be more strongly worded to insure adequate shoreland protection zones in these projects. I believe the 500' buffer zone set out in Alternative 1 is preferable to the 200' zone set in the draft EIS.

My objection to the narrower buffer zone is essentially aesthetic. These projects are located in the midst of some of the finest scenery on the East Coast. I believe the FERC should protect the lakes, ponds, rivers and streams in these projects from unsightly industrial and commercial activity. The 200' buffer is simply not big enough for that job. Nor does the narrower buffer protect these water resources from ecological damage.

As a longtime tourist, boater, and conservationist, dating from my first climb up Katahdin in 1971, I encourage you to provide more protection for these valuable natural resources.

In summary, I recommend that you expand the buffer zone to 500' so that these water resources receive protection similar to the Allagash and other easements on the West Branch; that you require Bowater and Great Northern to obtain easements on land it does not own, from willing sellers, to ensure protection; and that you permit existing commercial and private lease holders to continue their leases.

Norman Lins

CC Gov. King



Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, I am concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500' is the minimum that has been used on other buffer zones in the

immediate area. A 500' wide buffer should not be much more costly to Bowater/Great Northern than a 200' buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch.

alah sa tang s

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours,

the the

Rotanni German Course 1 (0.00-2 1789-72



Dear Secretary Cashell:

I am submitting these comments on the draft environmental impact statement issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although I have not participated in the formal process of relicensing the dams, I do care deeply about the West Branch watershed and its future. I was unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 p.m. evening public hearing and therefore want to make my comments to you in writing.

I strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas. The West Branch is one of the most significant watersheds in the State of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers and streams for aesthetic, ecological and recreational values is critically important. I am pleased that FERC recognizes the need to protect these shorelines. I am equally glad to see that FERC understands that the goal of protecting these resources can only be achieved if easements are obtained on land not owned by Great Northern.

Although I am pleased by FERC's recognition of the need to protect shoreland, I am concerned with several aspects of the approach recommended in Alternative 2 of the draft EIS.

First, I am concerned that buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational and ecological protection. I believe that the proposed buffer zones should be expanded to a width of 500 feet. A width of 500' is the minimum that has been used on other buffer zones in the

immediate area. A 500' while buffer should not be much more costly to Bowater/Great Merthern than a 200' buffer since the most expensive land is that immediately fronting on water. The land behind the shorefront is much lass expensive but of great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch.

Second, I am concerned that the option to allow Great Northern to install a Shoreland Management Plan at Penobscot Mills with buffers narrower than 200' could be used to defeat the purpose of the buffer zones. I believe that FERC must set minimum guidelines for any management plan, which should mandate a buffer of 500', subject to change only on a very strong demonstration that aesthetic, recreational and ecological resources will be adequately safeguarded.

I urge FERC to adopt the recommended buffer zone proposals as modified above. I believe that FERC should adopt the shoreland protection plan set forth in Alternative 1 -- 500' buffers. Adequate shoreland protection zones should be considered a routine cost of holding a valuable hydropower license, particularly in Maine's northern forest region.

Thank you for the opportunity to comment.

Sincerely yours,

Eleanor M Stenson RR2 Box 719 Wells Me, 04090

PH A 3 Zone if established now will remain for ever. 5 "Atradue It. will never be missed or grast med Unv State. Alternative #1 - 500H Buffer 210 - 009 84 West Street, Portland, Maine 01102 JAN 24 lear hois and Membersof F.E.R.L. Eliter of Maine will be ever thun kful Great Northern are Borhater are transients States Alternative -009 The established . Since Ailin Theriosi 114.

VERSAR, INC. FEB 2 4 195 ECOLOGICAL SCIENCES AND ANALYSIS



86

COP

VFRSAR

FCOLOGICAL

February 10, 1995

Lois Cashell Secretary, Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

Draft Environmental/Impact Statement - Ripogenus RE: Project (FERC No. 2572) and Penobscot Mills Project (FERC No. 2458)

Dear Secretary Cashell,

Kathy Winslow These comments are submitted by [organisation/individual] on the draft environmental impact statement (DEIS) issued by FERC on the Ripogenus and Penobscot Mills Hydroelectric Projects in Maine.

FERC deserves considerable credit for preparing an environmental impact statement for these projects. Although we have not participated in the formal process of relicensing the dams, we do care deeply about the West Branch watershed and its future. We were unable to travel to Millinocket on January 25, 1995 to comment on the draft at the 7 PM evening public hearing and therefore want to make our comments to you in writing.

We strongly support the need to require shoreland protection zones on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project ares. The West Branch is one of the most significant watershed in the state of Maine and in the Northeast. Protecting its high value lakes, ponds, rivers, and streams for aesthetic, ecological, and recreation values is critically important. We are pleased that FERC recognizes the need to protect these shorelines. We are equally glad to see that FERC understands that the goal of protecting these resources can only be achieved on the Ripogenus impoundment if easements are installed on all land, including those not owned by Great Northern. believe that these lands should be purchased on a willing seller basis and that Great Northern should be obligated to establish a conservation easement trust fund to purchase presently unavailable lands as they become available.

Although We are pleased by FERC's recognition of the need to protect shoreland, we are concerned with several aspects of the approach recommended in the DEIS.

First, buffers of only 200' are not sufficient to achieve the goals of aesthetic, recreational, and ecological protection stated in the DEIS. The proposed buffer zones should be expanded to a width of 500' feet. A 500' wide

buffer should not be much more costly than a 200' buffer; the most expensive land is that immediately fronting on water. The land behind the shorefront is much less expensive because it is less desirous for development, although it has great value ecologically. Buffers included in the hydropower licenses should be consistent with other 500' buffers in the West Branch of the Penobscot-River and the Allagash. Also, we believe that the buffer width should be 500' throughout the Ripogenus and Penobscot Mills Project area.

We also agree with FERC that any existing commercial and private leases as identified in the DEIS should continue to be honored and renewable, subject to LURC zoning regulations.

We urge FERC to adopt the recommended buffer zone proposals as modified above. Adequate shoreland protections zones are a routine and necessary aspect of ownership of a valuable hydropower license. The need for shoreland protection is particularly keen in the Northern Forest region, as the DEIS acknowledges. The public is entitled to assurances that its use and enjoyment of the natural character of this region, like the applicants right to generate power, will be secure for the term of the license. Shoreland easements are a critical element of that assurance.

Thank you for the opportunity to comment.

Sincerely yours,

Kathy K. Unilow

K. Winslow 211 Rocky Dundee Ow West Bankn Me West Bankn Me S I O I 825 North Papitol 14, NE Washington DC, 20426 s Cashell Sacretary. Federal Energy Negulatory 1955 19 Commission ÷ E-489

RESPONSES TO CONCERNED CITIZENS CONCERNING LAND USE ON PENOBSCOT/RIPOGENUS MILLS DEIS

Group - C

We received comments from many concerned citizens regarding our recommended land use recommendations in the Ripogenus/Penobscot Mills DEIS. Our response to these comments are provided here and the comments follow.

Comment noted. The staff developed revised land use recommendations (see section 4.9). The staff's recommendations provide measures to protect shoreland resources within the project areas while considering existing land use regulatory controls (see section 4.9, 4.10 and 5.3.4).

DEFICE OF THE SECRETARY

95 FEB 27 MII: 41

February 16, 1995

ECERAL ENERGY REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 -00 Ripogenus Project, FERC No. 2572 - 015

Dear Ms. Cashell:

旧花

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours ena

FILED OFFICE OF THE SECRETARY 95 FEB-3 PM 1:42 **160** VERSAR, INC. 71 Foreside Road Falmouth, Maine 04105 January 25, 1995 FEB 2 4 1995 Lois D. Cashell cretary FERC 825 North Capitol Street NE ECOLOGICAL SCIENCES AND ANA Washington, DC 20426

RE: Ripogenus Project (FERC#2572) & Penobscot Mills Project (FERC#2458) = 009

Dear Secretary Cashell:

As citizens of Maine, who have canoed the Chesuncook and rafted on the West Branch of the Penobscot, we are concerned about the West Branch watershed and its future. We are unable to travel to Millinocket for the hearing tonight, but send this letter regarding the Draft Environmental Impact Statement.

We are pleased that the Federal Energy Commission recognizes the need to protect shoreland on the lakes, ponds and streams in the Ripogenus and Penobscot Mills project areas, including land not owned by Great Northern.

However, we urge you to expand the proposed buffer zones to a width of 500 feet to ensure adequate protection for water quality and wildlife resources in the watershed. Five hundred foot buffers would be consistent with other buffers in the West Branch. Since the land closest to the water is the most valuable, additional width should not add greatly to the cost of easements. We also ask you to mandate 500' buffers for the Shoreland Management Plan at Penobscot Mills unless a clear and convincing demonstration is made that resources will be protected adequately.

Great Northern is asking for license to use a public resource. Sufficient shoreland buffers are a reasonable cost of obtaining these valuable rights. Please adopt the shoreland protection plan described in Alternative 1.

Thank you for your consideration.

Sincerely. Abre C Arde C. Anderso Richard G. Rockefeller



I - 26 - 95 2 Norse Place Athol, Na. 01331

Lois E Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital St. N.E. Washington, D. C. 20426

Dear Sir,

RE: Great Morthern Paper Company Ripogenus Hydroelectric Project (FERC # 2572) Penobscot Mills Hydroelectric Project FERC # 24581 - 009

The proposal for a five hundred foot buffer zone would take all my cottage and most all the land on my lease Lease number 25A on plan 8-573.

25A on plan 5-5/3. HUNDRED The proposed two/foot zone would take part of my camp. Is FERC prepared to buy the camp?

I have a cottage on South Twin Lake (Penobscot Hill Impoundment) which I purchased inI970. It may be fifty yards from my porch steps, down a steep hill, across a raised R. R. track, and down to the highigh water mark. The R. R. follows the eastern shoreline of the lake very closely. There are a lot of cottages with the R. R. between the lake and the camps. It seems to me the R. R. is about the buffer youcould find.

I can make very little sense of this proposal and am very much against it. This is an unwarranted restriction on my right to use and enjoy my property.

Sincerely, Phad Walustin

cc to Senators William C Cohen and Olympia Snowe Representatives James Longley and John B Aldacci





Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - 000

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely, Bruce Bailev

Woodruffe Bantiey Pritham Ave FO^P Hox B Greenville ME 04444EB 2 4 1995 January 29, 1995

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - 000

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely. 2 Bar

DEFICE OF THE SECRETARY

95 FEB 27 AN 11: 44

February 16, 1995

REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

- - - Lun

Daivd Beaulieu Millinocket ME OFFICE OF THE SECRETARY

95 FEB 28 PN 1:41

February 16, 1995

ELLEFAL ENERGY REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory.Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

V

Linda Belmont Tario PO Box 536 Millinocket, Me 04462

E-499

FILED OFFICE OF THE SECRETARY

95 FEB 27 Mil: 22

February 16, 1995

REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 095

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Mary Bernier 12 South St Millinocket, Me 04462

E-500



Bernard Bienkowski 316 Elmyood Drive Cliftand Beach NJ 07735 January 29, 1995

UFFICE OF THE SECRETARY 95 FEB -8 PM 12:00 SEDERAL ENERGY REGULATORY OMMISSION

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426.

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely, Samarel Dienkowski

Bernard Bienkowski

Ē N N с с RECULATION COMMISSION, LURIC, THE IN MICT OF MORE RESTRICTIONS WILL BE VERY COSTLY TO DEAR SECRETARY DOG & 2458,007 PO Box 7C ALREADY OVERLY INPACTED BY THE MAIN ł I AM SERIO WAY CONCERNED A BOUT HYDROELBUTHIC PROJECTS, I HAVE A SI IN VESTMONT IN SHORE PROUT PROPE FERCH 2572, AND PENOPSCOT MILLS LOIS B CASHEIL, SEC. OWMEN SUCH AS 1, FERC.

E-502-



ORIGINAL gen 31, 1785 LEC - ETARY and the second s Meurip hay miller Hydroneter Free SI, 1785 Martin Program the true Party of FERCAG 2572 Martin Provide Martine Program FERCAG 2572 A: 2454 A: 2457. Eran Frin Caulatt de ECOLOGICAI STER Strike it many du to bring think it has taken "I have the gran and 10 million dallans for FERC to maticina the califert stand - - norder call thes a day mai and a good in file dim Josty the Fallenel growing of sorthe thing happy grandfatten in the totangly in law that yeller instanded to part in working and your first - popen abuch you have reducted for march & a house them around the It call some to devise for be your and I have that any or white nent little degradation of the com this finite. How that a compan Finelizated fater for 42 your and the water quality has remained The Regulation Commission although intercome attimes fine dow a great for and I strongly ful that we do not mad any half from FERL in antaling mildlande of Ripagement is videralon. Bowater bes a quet for with it loving program and I feel that any me not fintificial Print dragging your feel and go on Epor tonly Co den ho Colon Jun G. Sum P.J. Toughay appropring E-504

95 FEB 28 PM 1: 46

- AL ENE-L

OFFICE OF

February 16, 1995

REGULATORY

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 ___005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Aller C. Aler

Very truly yours,



January 17th, 1995

1

Lois B. Cashell, Secretary F. E. R. C. 825 North Capital Street, N. E. Washington, D. C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572)

Dear Ms. Cashell:

As a Leaseholder from Great Northern Paper on the Ripogenus and Penobscot Mills Impoundments I wish to give my opinion in this letter and express my views on your Draft environmental Impact Statement covering development, setback requirements, etc.

FEDERAL ENERGY REGULATORY COMMISSION

.005

Development is not rampant in this area, and due to Forestry and Economic conditions it is doubtful that development will ever be an issue. This area is already under the Maine Land Use Regulation Commission regulations and as such, has effectively prohibited any real development. New, and old Businesses as well as campowners are affected. I find it very discouraging to think that I would have to deal with yet MORE restrictions if I should want to do anything to my existing property. I do not want to move my existing buildings, or my existing "vegetation". This should rightfully be my choice.

In closing I urge you to PLEASE help eliminate the proposed building setback and proposed vegetative buffer.

Thank you very much.

Sincerely,

Marjørie Briggs, Campowner Golden Road Millinocket, Maine Ø4462

Mrs. Frank Browning Morgan Bay Rd. FILED OFFICE OF THE SECRETAR RFD 1, Box 158+ 1925 95 MAR -2 AM 9: 55 Surry, Maine 04684 FELLINAL ENERGY REGULATORY Feb 18, 1955 COMMISSION Dear Sceretary Cashell, I wish to voice my support for the Draft Environmental Sugar Statement on the Ripogenus Project (Fere # 2572) and the Penebscot Mills Project (Fere # 2458) - \$09 strongly support protection for the lakes and rivers in These areas, Jeeling such protection is essential in Maine 5 Tualian of life & to Maine's Juture I would suggest through that a 200 ft. buffer zone is not Wide ensagh. My husband + I went down the allaghash Rive last

Mrs. Frank Browning Morgan Bay Rd. RFD 1, Bax 156A 1985 Surry, Maine 04684

Ot Laura Burch OFFICE OF THE SECRETARY 96 1/2 Quarry Road Ithdca NY 14850-8801. January 29, 1995 VCOCAR 95 FEB -9 AM 10: 18 HE JULATORY 프로 2 소 . Secretary ñ Federal Energy Regulatory Commission MMISSION 825 North Capitol Street NE Washington DC 20426 Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 -005

Dear Secretary, FERC;

Ń

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely. Laura Burch

annon (Marian Burch

E-508
COPY

VEEC

ECOLOGICAL SCIENCES AND

FILED OFFICE OF THE SECRETARY 17 January 1995

95 JAN 20 PH 12: 00;

Lois B. Cashell, Secretary Federal Energy Regulatory Commission ENERGY 825 North Capital Street Washington, D.C. 20426

Reference: Ripogenus Hydroelectric Project (FERC No 2572) Penobscot Mills Hydroelectric Project (FERC No 2458)

Dear Ms Cashell:

I lease a lot from Great Northern Paper which will be effected by the decisions pending on reference projects. I am writing to express my concern regarding the proposals to increase the building setback and vegetative buffer. I do not believe that expansion of project boundaries is necessary nor is it justified by the record of use in the case of these projects. In this I disagree with the Draft Environment Impact Statement and find the proposal for a 500 foot building setback ludicrous.

I am also concerned that the Draft Environment Impact Statement fails to address grand fathering of the current construction, deal with the issue of new construction as part of existing facilities within the proposed setback area or the issue of Great Northern selling the leased property during the license period.

I request these comments be taken into account as the Commission reviews the Draft Impact Statement.

Douglas B. Campbell 222 North Dickinson School Road Carlisle, PA 17013

CC: Congressman Goodling



95 FEB 10 AM 9:40

Management Company Inc. Timberland Service 107 COURT STREET • P O BOX 637 • BANGOR MAINE 04402-0637 TELEPHONE 207 942-8295 • FAX 202-942-1668

January 31, 1995

Lois B. Cashell Federal Energy Regulatory Commission 825 North Capital Street, NE Washington, DC 20426

Re: <u>Ripogenus Hydroelectric Project (FERC No. 2572)</u> -005

Dear Ms. Cashell:

Our company represents three large timberland owning families who collectively own fifty two miles of shore frontage on the waters behind Ripogenus Dam. Of the three family groups, the shortest length of ownership has been over fifty years and the longest is in excess of one hundred and ten years. In the case of the latter family, they are now into the sixth generation of ownership.

These owners strongly object to the provision in your November 1994 FERC/DEIS-0075 which calls for FERC to transfer its eminent domain power to Bowater and force Bowater to exercise this power to buy the fee or a conservation easement on their fifty two miles of frontage as a condition of granting a hydropower license to Bowater.

Maine's Land Use Regulation Commission (LURC) has adequate rules and regulations already in place to protect the economic and recreational value of the land adjacent to the impoundments.

While our clients have no interest in selling their land or conveying conservation easements, I can tell you if they were forced by eminent domain powers to sell they would vigorously oppose the values you have proposed. We have well-documented evidence that the use of \$5-\$5 per frontage foot is not even close to the value of what FERC proposes to take. Your proposal would cost Bowater well over \$10,000,000. This would be an undo and unfair burden to add to the already high cost of relicensing.

As you know, on January 25th a public hearing was held in Millinocket, Maine on your DEIS. About four hundred people attended. The hearing started at 7:00 p.m. and ended at 2:15 a.m. the next morning. Seventy one people testified with the main issue being the need for the 200 foot FERC buffer strip. Sixty seven of the seventy one people who testified were opposed to the 200 foot taking. Only four supported it. I have enclosed a <u>Bangor Daily News (BDN)</u> account of the hearing, our company's testimony, my letter to the editor of the <u>BDN</u> that appeared the day of the hearing, and a copy of a letter to the <u>BDN</u> from the town manager of Millinocket expressing the town's feeling. The articles represent the overwhelming opinion of Maine people and the specific tone of the January 25th hearing. You can confirm this latter point with the FERC people who were at the hearing.

v

Lois B. Cashell Federal Energy Regulatory Commission Page Two January 31, 1995

We in Maine value the economic and recreational benefits of the West Branch of the Penobscot River. We feel we have and will continue to protect these benefits without a FERC taking. I encourage you to get on with the business at hand and grant Bowater a thirty year license without the provision that they be forced to buy our clients' land as a condition of relicensing.

Sincerely,

PRENTISS & CARLISLE MANAGEMENT COMPANY, INC.

President

DMC/jm Enclosures

cc: The Honorable Elizabeth A. Moler, Chair (w/enclosures) Federal Energy Regulatory Commission

Prentiss & Carlisle Management Company, Inc. Timberland Service

Mainers take aim at FERC proposals

Hundreds at hearing on Bowater dams

By Andrew Kei Of the NEWS Staff

MILLINOCKET - The Federal Energy Regulatory Commission threatens to usurp state powers by propering limits to development on miles of shore front along the West Branch of the Penobecot River.

That was the overwhelming mee sage Wednesday night from state efficials and average citizens who spoke at a public hearing on the relicensing of hydroelectric dams ewned by Bowater Inc.

FERC is considering a draft environmental impact statement that impress several conditions on the relicensing. Hundreds of peois filled the auditorium at Stearns High School to demounce one re-

mirement that Beweter buy conpervation ensemble of miles of waterfront land in the region and strict future development within 200 feet of the shore.

Sen. Leo Kieffer, R-Caribou, majority leader in the state Senate. expressed grave concern about the requirements. Some of the shore front belong: to the state of Maine. "I urge FERC to disregard the reoommendation," said Kieffer. "... Federal imposition of land-use standards will raise an immediate conflict with the state of Maine."

Rep. Richard Gould, D-Greenville, House chairman of the Legislature's Energy and Natural Resources Committee, said the Maine Land Use Regulation Commission was well-qualified to take care of development in the unorganized townships.

"I know that you are interfering with the sovereign rights of the state of Maine when you attempt to regulate development," said Gould. "Maine has no intention of giving up any rights on any land it owns.

Stephen Adams of the State Planning Office said the King ad-٠. ministration would fight any attempt by FERC to impose land-use restrictions.

"The state of Maine has no intention of selling its property or property rights in the area, and it will strongly resist any attempt to take them by eminent domain," said Adams.

Residents of northern Maine also blasted the proposal. "Money meets to be speat in the

mills, not in buying share-front property to protect the view." anid Maureen Bacon, a local real estate agent.

Bowater's hydroelectric system on the West Branch includes five generating stations that produce \$2.8 megawatts of electricity. It is said to be the nation's largest hydropower system owned by a private manufacturer.

The dams provide about half the electricity needed by the compa-ny's Greet Northern Paper mills in Millinochet and East Millinocket.

Great Northern spent most of a decade and millions of dollars trying to relicense the dams. The pro-. cess is finally drawing to a close, as FERC commissioners consider a draft environmental impact statement released by the staff in December.

In the EIS, the FERC staff proposed only minor modifications to Bowater's proposal to regulate water flows on the West Branch. Environmentalists had sought substantial changes to the company's proposed use of water in the region.

Even so, environmentalists were delighted by the staff's proposals that Bowater develop a shoreland management plan for the watershed and restrict further development.

"These waters have extraordinary value," said Daniel Sosland of the Conservation Law Foundation. "The setback zones will help to protect the scenic values that are so important to this part of Maine."

If FERC accepts the draft EIS as it is written, Bowater would be obligated to buy conservation easements on about 2,000 acres of shore-front land on Ripogenus, Chesuncook and Caribou lakes at E-512

an estimated cost of 22 millio

"FERC would require GNP to obtain the title or a essements for the land area it do not currently own." read the draft EIS. "This protection zone would require a 200-foot building setback and a 100-foot vegetative buffer. prohibit individual and temperary docks and piers, and require ap-proval for public docks and piers." Closer to Millinockst, the com-

pany would be required to prohibit. waterfront development only on land it already owns. That includes much of the shoreline on Dolby Pond and Persodum.cook, Ambajejus, Twin, Millinockst and Quakish laines.

Environmentalists said the setback provisions were a bold step by FERC stations to protect resources in the watershed. They had see a 500-foot seipask, but said the more modest recommendation would help safeguard the aesthetic, recreations) and ecological values of the West Branch.

The proposal, however, worried some of the 1,000 camp owners who lease land from Great Northern. They questioned whether the setback would make it impossible to improve their camps or rebuild in the event of fire.

Some feared the government ultimately would force them to give up their leases on the pristine backwoods lakes. A FERC staff member assured the audience that the restrictions would not apply to existing camps, only to future development.

Doug Schmidt of the Maine Leaseholders Association, which represents hundreds of people who lease land from Great Northern. threatened to sue FERC if it attempted to regulate existing camps.

FERC	Нe	arir	ıg	(N	ο.	25	7 <u>2)</u>
Bang	or	(ME)	Ī)ai	1y	Ne	ws
;	ลถน	ary	4	5,	199	75	
-	(P.	age	l	of	1)	•	

CLF asks for what already exists on rivers

By James P. Keineles

would like to correct miscanceptions and inaccuracies fostered by the Conservation Law Foundation's recent Oped column in the Bangor Dasly News concerning the draft environmental impact statement usuad by the Federal Energy Regulatory Commission on Great Northern Paper Co.'s Ripaganus and Penobacot shills relicensing projects.

The Conservation Law Foundation waxed elequently and at great length about the benefits of building setbacks and vegetative builters around the impoundments but failed to mention that there is no factual need for either in these cases.

Past operation of the projects in question by Great Northern Paper Co. has given the area and the state of Maine all of the environmental benefits that the Conservation Law Poundation feels ments to be protocted, including semic views, a world-class salmon fishery, whitewater rafting flows, steedy year-round flows that benefit the entire watershed and fleed courted, which has prevented millions of dollars of devestation from flooding over the years. Great Nerthers's proposed operation under the provide and protect these impertant resources without the necessity of expanded project boundaries and more government regulation.

The expension project to additional more government regulation. Not only has the Conservation Law Foundation ignored Great Northern Paper Co.'s contribution to providing these benefits to the public as a whole, it has also accused the critians of the state of Maine and our Land Use Regulation Commission of having inadequate regulations in place to reviser these removements.

EURC's rules and requisitions. EURC's rules and requisitions were adopted after extensive input from Maine classes is accordance with Maine law, and represent a judgment by the classes of this state of the measures necessary to project our resources in the unifganized territories, including the project areas. CLF obvisually domnot feel Maine's citizens have done an dequate job and wants federal regulations to correct our mistakes as they perceive them. CLF also has failed to explain why these projects should be subject to rules dufferent than them which apply to Maine's thousands of other lakes, streams and rivers.

streams and rivers. Those of us who chose to live in and near Mains's unerganised territories believe that LURC's reprlations are extremely restrictive and are more than sufficient to protect the aestbetic, recreational and ecological values of the West Branch River and the surrounding territory. The sroay is CLF's arguments is that the resources CLF claims to cherich are present because of LURC's rules and not in spite of them.

Maine people are capable of protecting environmental resources

'CLF and its out-of-state allies want to teck up these lakes and prevent all future land development. Their instent is, first and foremost, part of a larger scheme to create a national park in northern Maine which would replace the privately owned working forests which have steadfastly proven their worth to Maine's scoonny and, secondly, is protect the local residents of methern Maine from themselves. I believe that the poople of this state, and of northern Maine is particular, are fully capable of protecting their environmental researches through agencies such as LURC and that private ownmuck.

Water quality in the impoundments and throughout the West Branch is excellent by any standards, and diverse recreational opportunity is widespread. Additional restrictions are not justified by present conditions or any reasonably foreseeable events which may occur over the new li-

cense terms. It's especially unfortunate that CLF has created great uncertainty over the continued use and property rights of more than 900 lesses on the impoundments. These uncertainties include:

James F. Kotredes Letter to the Editor Bangor Daily News January 26, 1995 (Page 1 of 1) 1. Will camp owners be allowed to maintain their existing buildings and docks within the proposed 200fast setback zone?

2. Will camp owners be allowed to construct any new buildings, expand existing buildings or new decks within the 200-foot setback zone?

 Will Great Northern be allowed to sell the leased land to camp owners during the term of the new licenses should it decide to sell and camp owners decide to huv?

4. Will camp owners be allowed to remove any vegetation within the proposed 100-feet vegetative buffer?

5. How will the value of the camp owners' property be affected by the proposed 100-foot vegetative buffer?

5. Will relicensing opponents and federal agencies convince FERC to increase the proposed building sethack to 500-feet and the vegetative builfer to 250 feet?

7. Will camp owners have to deal with LURC and FERC concerning future changes to camp owners' property?

It is equally unfortunate that CLF wishes to force Great Northern Paper Co. to buy through the power of emment domain if necessary, some 67 miles of shoreline around Chesencook Lake that Great Northern does not presently own in order to expand project boundaries. The affected landowners have not been parties to this proceeding and should not be forced to give up their land in order to foster CLF's land grab. The affected landowners have been good stewards of the land for generations and should be allowed to continue their ownership.

The draft issued by FERC's staff is not final and its purpose is to allow the public an opportunity to comment. We feel confident that once FERC understands LURC's rules, it will withdraw its proposal for expansion of project boundaries.

The projects and their impoundments are not just the wilderness. They are an integral part of a modern working forest that supports a good part of Maine's economy and certainly is of utmost importance to the Millinocket area.

The draft environmental impact statement rejected most all of CLF's other proposals, and FERC should reject expansion of project boundaries for the same reason; expansion is simply not justified by the facts in these cases, and any costs unnecessarily passed on to Maine's businesses results in direct job reductions and further erosion of the state's economy.

James F. Kotredes is the town manager of Millinocket

TESTIMONY OF LAWRENCE E. PHILBRICK / FEDERAL ENERGY REGULATORY COMMISSION DRAFT ENVIRONMENTAL IMPACT STATEMENT / GREAT NORTHERN PAPER HYDRO RELICENSING APPLICATION / RIPOGENUS IMPOUNDMENT JANUARY 25, 1995

My name is Lawrence Philbrick, I am the vice president of Prentiss & Carlisle Management Company Inc. which is a timbertand management company located in Bangor Maine. We manage the timbertand interests of clients who have substantial ownership within the FERC project area behind the Ripogenus impoundment. It is our understanding that FERC in it's Draft Environmental Impact Statement on the GNP relicensing application is requesting that Great Northern Paper should acquire for or conservation essences on all of the land within a 200 foot setback of the shoreline of the waters impounded by Ripogenus Dam in order to maintain the aesthetics and water quality of the area.

Our clients in this area consist of three large timberland owning families who collectively own 274, 853 feet (52 miles) of shore frontage in this project area. At a depth of 200 feet this amounts to 1,262 acres of land within this area. Our clients have owned this property for a long period of time (the shortest length of family ownership) is 50 years and one family has owned here for 110 years - now in the sixth generation of family ownership) and they have a strong sense of stewardship. Most of these owners are Maine residents and many of them actively use the area for recreation and family activities. These owners have not abused this land over the years. In fact the aesthetics and water quality of this property are judged high today because of our client's care and management of these resources. Our clients have no interests or conservation easemnents on this property.

We and our clients also feel that it is not fair for the Federal government to force GNP to acquire fee or conservation easements on property of others as a condition of having this impoundment relicensed. This seems to us to be excessive use of the powers of the Federal government.

The Maine Land Use Regulatory Commission (LURC) has the regulatory authority and the zoning responsibility over the subject lands. We sincerely feel that LURC's present rules and regulations provide more than adequate insurance that the water quality and aesthetics of this area will be carefully protected

In summary, we and our clients are very much opposed to FERC's suggestion that GNP should acquire the land or rights of others within this impoundment area. We feel that this land is currently well managed by it's owners and it is adequately protected by the existing rules and regulations of the State of Maine. We furthermore feel that the federal government should not attempt to override the authority and powers of the State of Maine and the property rights of its citizens.

Thank you.

Lawrence E. Philbrick Testimony FERC Hearing (No. 2572) January 25, 1995 (Page 1 of 1)

FERC's proposals don't ease dispute

Devid M. Carlisle

am writing in response to the guest column by Daniel Sosland of the Conservation Law Foundation (BDN, Dec. 27), regarding the Federal Energy Regulatory Commission and Great Northern Paper Inc.'s hydro relicensing efforts.

Sosiand is correct in stating that the West Branch of the Penobscot River is an extraordinary resource - both from the point of view of low-cost power and varied recreational use. We. too, support efforts to protect these economic and recreational benefits. However, we oppose the means by which Sosland, the CLF and FERC propose to protect these special features of the West Branch. We view the CLF/FERC approach as an egregious abuse of the power of the federal government. It is yet another case of the federal government attempting to override the authority and powers of the state of Maine and the property rights of its citizens.

In December, FERC issued its Draft Environment Impact Statement on GNP's hydro relicensing application. As a condition of its relicensing, FERC proposes to force GNP (using FERC's eminent domain power) to "grab" the land that GNP does not presently own on the shores of the impounded waters behind Ripogenus Dam. The protection zone that GNP would have to acquire would be a 200-foot setback with no building allowed within the 200 feet. There would also be a 100-foot vegetation buffer. It appears that FERC would not allow permanent and temporary docks on the shores of the impoundment area.

Our company represents three large timberland owning families who collectively own 52 miles of shore frontage on the waters behind Ripogenus Dam. Of the three family groups, the shortest length of family ownership h been over 50 years and the lon has been in excess of 110 years. In the case of the latter family, they are now into the sixth generation of family ownership. Most of the owners are Maine residents and many of them actively use the area for recreation and family pursuits. All three of these families have a strong sense of stewardship. They have no interest in sellthis land that the CLF and FERC are trying to grab. To me, it is an outrage that the federal government could force GNP through eminent domain to buy others' land as a condition of GNP being granted a dam license. Great Northern Paper Co., for its part, does not want to have any part in being forced to take land under the threat of the eminent domain power of the federal government. Maine's Land Use Regulation Commission has the regulatory authority and responsibility over the land involved. LURC's present rules and regulations provide ample regulatory authority and power to protect the future aesthetics and water quality. Sosland, the Conservation Law Foundation and FERC seem to be saying that LURC and the people of Maine cannot be trusted to protect the aesthetics and water quality of the West Branch. I do not agree.

Editorial Page Bangor Daily News January 25, 1995 E-515 Sociand refers to FERC's DEDS as gains in the interest of all; in other words, "no winners or losers." He obviously forgot to take into account the owners of land on the shore of the impoundment area, the lossess who have camps on the shore of the impoundment area and, for that matter, all of the residents of the state of Maine who own Gero Island in the middle of Chesuncook Lake — which also would be subject to the facturally mandated taking. FERC's landtaking proposal is not in the interest of the Maine people.

All Maine citizens are losers when we allow the might of the federal government to overpower us and usurp our rights.

David M. Cartisle is president of Prentiss & Cartisle Management Co. Inc.

David M. Carlisle
Letter to the Editor
Bangor Daily News
January 25, 1995
Re: FERC No. 2572
(Page 1 of 1)

FILED AM 11: 38 95 JAN 30 FEDERAL ENERGY REGULATORY

January 25, 1995

Peter J. & Norma F. Cesare 9844 Burke Pond Lane Burke, VA. 22015

or

116 Penobscot Ave, P.O. Box 849 Millinocket, Maine 04462

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

Re; Ripogenus Hydroelectric Project (FERC No.2572) -007 Penobscot Mills Hydroelectric Project (FERC No. 2458) -009

We feel compelled to add our voices to the many who will be attending the public hearing concerning the Draft Environmental Impact Statement for these projects on this date at Stearns High School in Millinocket, Maine.

We have been lease holders and owners of a cabin on Caribou Lake (adjoining Chesuncook Lake) for close to twenty years. This cabin has been a very special place for our family and friends during this period and we were hoping for many more years to come. However the proposed regulations for building and development setbacks along with vegetation buffers are completely inappropriate for the type of leased lot that we possess on the lake. The proposed 500 foot setback would place us on or near the hauling road which runs behind the cabin. The 200 foot setback would place us very near to the loud sounds of the passing trucks and the huge clouds of dust and dirt caused by their passing. In effect you will destroy our ability to enjoy the wonderful lakeside and wilderness experiences that we have been priviledged to share for so many years. It represents another glaring example of big goverment intruding upon the lives of ordinary people because it has some honorable goal which has been distorted by over-zealous extremists. We also feel that the State of Maine has been very responsive to the local environmental issues and does not require any assistance from the Federal level of government. Therefore we are hopeful that you will become more respectful of local government on these issues.

We feel we should be allowed to continue to enjoy our cabin at its present location and to be allowed to make improvements to it as we deem necessary for our family's use. We also have great respect for the land and water in and around Caribou Lake. We insist that all of our visitors to the cabin do nothing to harm the environment. Also we have observed that our neighbors on the lake feel and behave in a like manner. Therefore we find your proposed regulations as unnecessary and not appropriate for this region of Maine. We feel strongly that these proposed regulations will severly impact the ambience of our vacation home and the homes of our Maine neighbors and friends, which is a very important part of all of our lives. Therefore we urge FERC to eliminate the proposed building setback and vegetation buffer regulations.

Thank you for your understanding in this matter of great concern and worry.

Very truly yours,

Re J. Cenn + Ninsteene

Peter J. & Norma F. Cesare

cc Senator William S. Cohen Senator Olympia Snowe Congressman John E. Baldacci Congressman James Longley Govenor Angus King Mr. Dean A. Beaupain, Attorney for the Town of Millinocket



Kendall Chevalier 287 Atlantic Ave North Hampton NH 03862 January 29, 1995



Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - $00^{>}$

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

- Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

- Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely,

Kendall Chevalier

dall

i⁄

I	
. Van Briten an Bart C	95 FED
SET. 2.2.17.5	FEDERAL ENFRON
100113-031 1 4 FT	COMMISSION
Describer Discourse Had	
	J Recording: Ripogenus Hyd

Federal Energy Regulatory Commission 825 North Capitol Street, N.E. Washington, DC 20426

Penobscot Mills Hydroelectric Project, FERC No 2458

Dear Ms. Cashell:

IJ

The Maine Conservation Rights Institute MECRI is a 501-3c institution whose purpose is to promote through education and action good conservation practices within the Constitutional framework of the Fifth Amendment and property rights. MECRI filed as an intervenor for Great Northern in 1993. I also attended the January 25, 1995 public hearing held by FERC in Millinocket, Maine.

MECRI commends the action taken by FERC to not require flows into the backwater channel as proposed in Alternative #1. We agree that the requirement would provide only marginal environmental gains at unacceptable costs to Great Northern.

However, we are very concerned that you elected Alternative #2 which requires the project boundaries to be expanded to include a 200 foot no development buffer, and a 100 foot no timber harvesting buffer zone around all waterways within the project. Of particular concern is the condition that Great Northern be given powers of eminent domain to purchase, through fee or easement, lands not presently owned by Great Northern but which are within a 200 foot buffer around waterways bounded by the project area. Such requirements represent:

- 1. An expansion of the powers of eminent domain for purposes that are totally unjustified and which establish a serious precedent for fascist-like collusion between government and private business.
- 2. An unacceptable intrusion of Federal jurisdiction into state and private rights as guaranteed by the United States Constitution. The Maine Land Use Regulatory Commission already has stringent setback requirements that more than adequately protect water quality. There is simply no justification for FERC to require their own set-back requirements. Not only does it create jurisdictional nightmares, but is an insult to the State of Maine and our ability to control our own destiny while protecting our resources.
- There is no specific language to protect existing development along the shoreline other than a vague 3. grandfather clause. If FERC is to assume jurisdiction in this matter, it must develop full guidelines governing existing development and establish an office to administer and enforce the rules. FERC must spell out whether a camp owner can repaint his house another color, or whether the owner can improve his home, build a garage, or replace his home if it burns down. Such minutia are not hypothetical but are real-world issues which consume tremendous time of the local code enforcement officer whom FERC would have to employ or contract. Do you want to assume this headache?

MECRI Page: 2 January 28, 1995

Although the above represent major concerns of MECRI, we also have serious questions about the process FERC has used in Great Northern's application for relicensing. This process has now taken nearly 10 years and \$10 million. Great Northern has demonstrated that it is a good steward and a good corporate citizen. From the very start, it was obvious that environmental groups were using nuisance issues to stall and otherwise harass Great Northern in their efforts to renew a simple license for operating its hydropower facilities. It is obvious that these environmental groups are using these nuisance issues to not only drive up the cost of the application, but are attempting to create uncertainty about the applicants financial future and stability. In the case of the paper industry, competition is fierce and requires continuous major capital improvement to remain competitive. It is well known that Great Northern was to receive an essential \$150 plus million capital improvement project. But as long as there remains doubt about the relicensing of its hydro-power facilities there is no way Bowater could provide that level of capitalization. If the licensing process is delayed long enough, the mill might become so uncompetitive that capitalization is no longer feasible. Great Northern and the State of Maine need a decision soon for the health of both.

The environmentalists have made much of the so-called public good in their arguments. But the public good that Great Northern and the forest products industry in general have provided in the State of Maine is even better documented and has been vital to Maine's economy. By including conservation buffers demanded by environmental groups, FERC extended its powers beyond its specific mandate and effectively secured the values of a tiny, but powerful, handful of environmentalists that otherwise had little if any legal or political support. Furthermore, FERC catered to the special interest groups who have no real stake in the issue while simultaneously playing Russian roulette with local residents whose lives depend on your decision. It was insulting to the citizens of northern Maine that your consultant during the public hearing on January 25, 1995 stated that the 200 foot conservation restriction was a *balance* between the Applicant's proposal and Alternative #1, which required a 500 foot buffer. To use that logic is like saying a thief is balanced because he only stole half your belongings.

This is not a new issue. The Northern Forest Lands Study suggested similar mechanisms to give land use jurisdiction to federally sanctioned agencies. The U.S. Forest Service then spent \$4 million on the Northern Forest Lands Council, which took four years and heard testimony from several thousands of people to investigate and recommend where government should go from there. An unjustified 25 percent of the seats on the Council were reserved for profession environmental representatives, and the Council's consensus process gave each of them virtual veto authority, yet the Council recanted support for each and every suggestion which would give jurisdiction over land use issues to federal agencies. The Council's Final Recommendations adopted the following principles:

- · The rights of private property owners must be respected.
- States must be allowed to determine their own futures.

The Maine Conservation Rights Institute recommends that FERC learn from this exhaustive study and adopt the Applicant's Proposal instead of Alternative #2 which requires a 200 foot setback and a 100 foot no harvest buffer.

Sincerely yours,

Robert Voight President

Michael & Coffman, Ph.D. Board Member

MANY CONTERVATION RESEATON INSTITUTE. Lubec. Maine 04852



825 North Capital Street, N. E. Washington, D. C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572)

Dear Ms. Cashell:

As a Leaseholder from Great Northern Paper on the Ripogenus and Penobscot Mills Impoundments I feel compelled to write this letter to express my views on your Draft Environmental Impact Statement covering development, setback requirements, etc.

Contrary to some beliefs, development is not rampant in this area, and due to Forestry and Economic conditions it is doubtful that development will ever be an issue. This area is already under the Maine Land Use Regulation Commission regulations and as such, has effectively prohibited any real development in this area.

As President and owner of Katahdin Air Service Inc., a business which has been in existence here since 1946, I find it very discouraging to do business under existing regulations. To add further restrictions would be economically, and in some cases, physically impossible to comply with. This would inhibit the small businesses in this area from continuing and the resultant loss of jobs amd economic benefits to an already depressed area.

I would hope that the current political climates in Washington would not be in favor of unneccessarily restricting small business and higher unemployment because of these unneccessary bureaucratic decisions.

Singerel

Rand**ál /** E. Comber, Pres. Katahdin Air Service Inc.

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 - U Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

OF FICE OF

95NAR - 2 11 3 3

FEDERAL ENERG REGULATORY. COMMISSION

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Ganet & Correau Cicalice Paper Salis Clerk

COPT VERSAR 17.88 Knople millinocked m. OFFICE OF THE SECRETARY Jan 27, 1995-FERC Washington DC 95 FEB - 3 PM 904.7 FEDERAL ENERGY REGULATORY P-2572-005 P-2458-009 Simi I attended the fan 25 the meeting in milling his which drew a full house plus standers. I had no previous connection with or knowledge of the matter in hand - I lasted three hours and about 80 speakers and left with 30 od speakers left to Several thing impressed me -I The quality and caliber, the breath and depth of knowledge of these who spoke. I the unanimity of opinion and good sense enders. 3. the wasteful and burdensome cost of the whole federal regulatory process - a seriour indictions of FERC's ability to do its fol properly. 4. the gratuitour intrusiveness of federal regulator expanding from legitimete energy and fickeries lonsions into land use regulation where it having authority or business to be - political blackmail, really; and in an area already will taken care of 523 the state and it people.

5 the valation of state rights, property rights personal rights by our own alleged public servante - the insufferable federal attitude - that we are relly not much nor than retarded wards of the state. The whole affair inspressed me as wasternel. - inefficient, meddlisome, expensive, incompetent and abusine - 8 years and ten million dollare badly nieded for plant modernization down the tribes and still no beense renewal - what an atuse of perver! The wonder the voters last fall said get off my back and out of my buince I say stop statting, gelout of the land we rachel, get in with your proper fot, and rine the licenses emmediately. Too many "deallines" have come and you I really think your group shall be Erripelle. to reembridge the Great northern for its washed time and money. They and all the Nestig us have totos surely been neidlessly damaged.

Marian Consistork

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

ECOLOGICAL SCIENCES AND AMALYSIS Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572

Robert Cressey HC 74 Box 232C

East Baldwin ME 040 January 29, 1995

Dear Secretary, FERC:

 $\sqrt{}$

J

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

· Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

· Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

· How will the value of property be affected by the proposed setbacks and vegetative buffers?

· Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

 Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

· Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Robert Cressey

CEX 2410

OFFICE OF THE SECRETARY

95 FEB -9 AM 10: 19

FILED OFFICE OF THE SECRETARY

95 MAR -2 AM 9:56

February 16, 1995

REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Eteptica Collect Proprietorie There Unhai Home - Whoreket , ME

95 FEB 28 PH 1145

EELERAL ENERGY STREGULATORY COMMISSION

NGTT

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

OFFICE OF

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

-Very truly yours,

semily Cille

FILED OFFICE OF THE SECRETARY

95 FEB 27 Mil: 17

REGULATORY

COMMISSION

009

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 00.5

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

V

153 marice et Nulimenter MAR 04462 723-8-96

CO9¥ SAR. INC VER EFR 24 10:5 Alyce Cusson 163 Hebron Road Bolton CT 06043-7809 . ECOLOGICAL SCIENCES AND ANALYSIS January 29, 1995

OFFICE OF THE SECRETARY 95 FEB -9 AM 10: 02

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 -000

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Curso Alyce Cusson

William Van Deventer 154 East Nassau Street Islip Terrace, NY 11752



OFFICE OF THE SECRETARY 95 JAN 27 AN 9: 55 FEDERAL ENERGY REGULATORY COMMISSION

January 23, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC No.2572) Penobscot Mills Hydroelectric Project (FERC No. 2458)

Dear Lois Cashell:

I am a lease holder with the Great Northern Paper Company (Lease #3948). This lease resides on property within the Ripogenus Impoundment. Although I hold permanent residence in New York State, it is my hope to purchase this land when the opportunity is made possible. I have visited this area since 1969 and have utmost respect for the land and the environment.

I recently applied for and received approval from Maine Land Use Regulatory Commission (LURC) to construct a seasonal camp. I can attest first hand that the guidelines provided by LURC meet all environmental concerns and more. There is absolutely no reason for further restrictions placed on this land use by the Federal Energy Regulatory Commission. Maine's LURC guidelines are more than adequate for building setbacks and vegetative buffers within watershed areas.

I fully support Great Northern's proposal to add no additional restriction on existing leases within the impoundment.

Sincerely,

William Van Deventer

WVD/wvd cc: Dean A. Beaupain Attorney for Town of Millinocket



John & Charlotte Dilworth Box 10 Jonesboro ME 04648 January 29, 1995



Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - 000

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

- How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

- Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

John & Charlotte Dilworth

E-531





1/27/95

To: Lois B. Cashell Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

From: Robert L. Dishon 33 Independence Lane East Millinocket, ME 04430



1

OFFICE OF THE SECRETARY

95 FEB -2 AN 9: 28

FEDERAL ENERGY

COMMISSION

To Whom It May Concern;

As a lease holder and campowner on South Twin Lake, Maine (an area controlled by the Penobscot Mills and Ripogenus Hydroelectric projects, FERC No.'s 2458 & 2572) - 005 I wish to make my views known to you.

We strongly oppose FERC's proposal to impose building setbacks or vegetative buffers on property surrounding these projects. (Whether it be 500' & 200' or 250' & 100' respectively.) We feel this would be extremely restrictive to property owner's in these areas, especially since FERC has not elaborated on whether any current lease holder's or campowner's would be "grandfathered" under this ruling.

We understand that relicensing of these dams is imperative to the survival of Great Northern Paper Co., but feel that FERC should allow the Maine LURC to continue to make decisions on this matter.

Thank you.

Robert L. Dishon & family



Richard Doane 291 Tuttle Road Cumberland ME 04021 January 29, 1995

UFFICE OF THE SECRETARY 95 FEB -8 AN 11: 59 FEDERAL ENERGY REGULATORY

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 7000

Dear Secretary, FERC;

Ĵ

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.



John Downing Box 655 Rt #76, Chesuncook Villiage Greenville ME 04441 January 29, 1995 VERSAR, INC.

-10 👘

ECOLOGICAL SCIENCES AND ANALYSIS

៍ភ

95 FEB 27 MII: 29

FELERAL ENLIGY Secretary GULATORY Federal Bhergy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely. John Downing

V



000

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

ROBERT ERICKSON, D.M.D. 50 Summer St., P.O. Box 829 Millinocket, ME 04462-0829 (207) 723-4543

AS PLEASE ISSUE THE CREESE JOON SO WE CAN VEFT ON MEH ONE LIVES



COP Linocket, ME 04462g January 26, 1995

rection in dk

Ms. Lois D. Cashell Federal Energy Regulatory Commission 825 North Capitol Street, N.E. Vashington, D.C. 20426

Dear Ms. Cashell:

I attended the FERC public hearing held in Hillinocket, ME on January 25th; concerning Project Nos. 2572 and 2458; however I did not speak at that meeting. I have two points I wish to make pertaining to the DEIS-0075, issued in November 1994.

First, I oppose the recommendation that would require GNP to purchase lands that border the project area. This is a needless expense to GNP, comes upon the heels of an economic recession, and comes at a time of tough economic conditions for GNP and for Bowater, Inc. The region's economy, and especially the people of northern Maine, has suffered greatly due to the economic conditions that have prevailed for the past several years. The added burden placed upon GNP to purchase lands from uninterested sellers, takes capital away that is needed to improve mill facilities. The improvement of mill facilities will ultimately lead toward improving our region's economy.

Second, I oppose the recommendation that requires a 200' set back from the high water mark for buildings, and a 100' vegetative buffer zone, within the project area. Current regulations enforced by Maine's Land Use Regulation Commission, are adequately protecting the environment and need not be duplicated or superseded by the Federal Government.

I urge FERC to expeditiously issue a long term license to GNP for Project Nos. 2572 and 2458 and that the license not be encumbered with the above mentioned conditions of approval.

With this letter I am urging Maine's congressional delegation to press for the approval of Project Nos. 2572 and 2458, as outlined in this letter.

I thank you for the opportunity to comment in support of GNP's re-licensing effort.

Sincerely.

Anthony Filauro

cc: Congressman John E. Baldacci Senator William S. Cohen Congressman James Longley Senator Olympia J. Snove





February 13, 1995

VERSAR.

Honorable Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, N. E. Washington D. C. 20426

RE: Penobscot Mills Project, FERC No. 2458 & Ripogenus Project, Ferc No. 2572 _ OOh

Dear Secretary Cashell:

Overall, I was pleased with the content and recommended alternatives in the Draft Environmental Impact Statement for the above referenced projects, but there were several areas which I feel should be changed.

Your conclusion and recommendation that no water be released in the Back Channel is correct. However, even though your modeling indicates that with flows up to 165 cfs there would be no significant impact on the WUP during most years, the negative impact would be much more significant during dry years because of the unpredictability and timing of rainfall and snow runoff. The language used on this subject should leave no doubt that no water is to be released into the Back Channel except for spillage in the case of a full impoundment due to excessive rainfall or spring runoff.

Any expansion of project boundaries is completely unnecessary. If regulation is necessary with anything above the high water mark of the impoundments, it should be specifically stated that any existing development would not be affected - it would be "grandfathered" - and would not be subject to any conditions or regulations other than those currently in place by landowners and state agencies.

The most unreasonable aspects of the DEIS are the expansion of the boundaries to include 200 feet of shoreline above the high water mark, and the requirement that in the Ripogenus Project, Great Northern Paper Company purchase lands they do not currently own, or acquire conservation easements in order to achieve this. Any transaction which takes place for the conservation of any land should take place on a willing seller - willing buyer basis. This would not be the case in this instance; Great Northern would obviously not be interested in purchasing land from which it would gain nothing. I also doubt that current landowners would be willing to sell this same property, which may have been in family ownership for generations. If it is necessary to further restrict activities on Chesuncook Lake, conservation easements such as are in effect for the West Branch of the Penobscot itself should be negotiated by the state with all landowners involved, and should be satisfactory to all. Governmentally mandated purchase of land by one owner from another flies in the face of the principles of freedom on which this country was founded.

It is difficult to believe that Federal officials located hundreds of miles from this state, prompted by environmental groups similarly located, feel more qualified to determine what is best for us, the people of Maine, or that our state government and regulatory agencies such as the Land Use Regulation Commission cannot adequately protect the state's environment. The Federal government and envirnomental groups appear to have lost sight of a unique and critical fact: within the state of Maine is a vast area which remains forested and undeveloped - even though timber has been harvested and the water resources used for log driving and power generation for well over 100 years - because of the management policies of Great Northern and the regulatory policies of Maine's state government. In its comprehensive use plan, LURC obviously recognizes the value of Chesuncook Lake by classing it 1A, the highest classification, which indicates it is of statewide significance. Its management class indicates an accessible. undeveloped lake with exceptional values. LURC, through its own regulations, would endeavor to protect the lake for the values it deems important in order to preserve its character. LURC's zoning for most of the Ripogenus Project is among its most restrictive. I believe that current policies and regulations are certainly adequate to preserve the aesthetics desired for the area.

For twenty-two years I have owned and operated Frost Pond Camps, a sporting camp which is located 3 miles north of Ripogenus Dam. Many of our guests fish both the lower West Branch of the Penobscot and Chesuncook Lake, and use the lake for boating as well. I have also owned and operated a cance trip outfitting business serving the West Branch and Chesuncook for 28 years. Logically, I would be supportive of any measure ensuring the preservation of the wilderness character of the area. However, I recognize that private management of the area for production of forest products as well as the recreational needs of the public has worked extremely well in the past, and I see no need to undermine this unique system of stewardship. During the past twenty-eight years of living and working in the West Branch-Chesuncook region, I have noticed few changes to its wilderness character other than those resulting from the public's increasing recreational use of the area. We serve numerous clients each year, and I cannot recall any comments or concerns about how the lake is managed or regulated. In fact, the opposite is true; most are amazed that this area in private ownership has remained as it is, and appreciate that Great Northern has chosen to continue their policies and that current state regulations ensure that it will remain as it is.

I find a certain irony in the concern by what I feel is a vocal minority to prohibit any future development, while these same people would like to preserve such past developments as the Ambejejus Boom House and Chesuncook Village for their cultural and historical value. If the restrictions currently proposed for these projects had been in effect in the past, these sites of such great cultural and historical value would never have been constructed.

Requiring Great Northern to purchase additional land around Chesuncook Lake is entirely unnecessary. I am personally offended that others who know very little about my home, lifestyle, and livelihood feel that they have the right to tell me what is best for me. The public benefit so often referred to must include not only the pristine aesthetics of the area, but also recreational activities, the need for the forest products produced here, and the many jobs and other widespread economic benefits resulting from a healthy and profitable Great Northern Paper Company.

A situation which had some similarities to this one was the creation of the Allagash Wilderness Waterway in 1965 - 67. The Allagash River had been Fedrally designated as a Wild and Scenic River and a decision needed to be made as to whether the area should be managed by Federal or State agencies. Maine state government chose to create a State Park in order to keep management and control at the state level rather than in the hands of bureaucrats from a Federal agency hundreds of miles away. The State of Maine, through its management and regulation, has done an admirable job of maintaining the wild character of the Allagash. I see no reason why the State of Maine cannot continue to regulate the lands around the projects waters through LURC as it has in the past, and as it has done with the Allagash. Obviously, the people of Maine want to keep their lands in their control, as was evidenced by the decision with the Allagash almost 30 years ago, and by the comments made at the hearing on January 25 in Millinocket. As indicated by the results of the November elections, the public wants less governmental interference and control in their lives. Where possible, local and statewide control should be handled at the local and state level, not by the Federal government. In this instance, the regulations in place which are administered by LURC are more than adequate and should continue. Keep regulation of Maine's wildlands in the hands of those who have successfully preserved them for many years and who know best what the people of Maine need - that is, our own local Company and State government.

> Sincerely yours, fue Junens Eric Givens Maine Sporting Camp Association

P. O. Box 552 Kingfield, ME 04947 207-265-2049

January 27,1995

Office of Hydro Power Licensing Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC

Dear Commissioners:

Re: FERC Case #2458 FERC Case #2572



Please accept this written testimony regarding the above referenced cases.

My major concern is with the proposed addition of land use restrictions within the project area. I heard much excellent testimony at your your public hearing on January 25 in Millinocket, but missed any discussion of the following two points.

BALANCE

Your consultant opined that the 250 foot conservation restriction was a <u>balance</u> between the Applicant's proposal and Alternative #1. I do not believe that the concept of <u>balance</u> has any relevance to the issue. Meaningful <u>balance</u> must hang from legitimate <u>principle</u>. We do not excuse a thief who takes only half of his victim's possessions.

Your alternatives would put FERC in the position of securing the values of a tiny, but powerful, handful of environmental organizations, though the legal and political rationale for such action is questionable. Effective arrogation of such land use authority requires not only a legal basis but the fundamental consent of the people whose lives are most affected.

This is not a new issue. The Northern Forest Lands Study suggested similar mechanisms to give land use jurisdiction to federally sanctioned agencies. The U. S. Forest Service then spent \$4,000,000 on the Northern Forest Lands Council, which took four years and heard testimony from several thousands of people to investigate and recommend where government should go from there. An unjustified 25% of the seats on the Council were reserved for professional environmental representatives, and the Council's consensus process gave each of them virtual veto authority, yet the Council recanted support for each and every suggestion which would endorse federal land use jurisdiction. The Council's <u>Final Recommendations</u> adopted the following principles:

* The rights of private property owners must be respected.

* States must be allowed to determine their own futures.

I believe FERC should think hard before dismissing the recommendations which have flowed from such an exhaustive and public process.

Office of Hydro Power Licensing January 27, 1995 Page 2 Case 2458 Case 2572

ADMINISTRATION

Your consultant emphasized that existing uses within the buffer area would be grandfathered. You are deluded if you think the distinction between old and new development is a bright line. Can a camp owner repaint his house another color? Can he use a better material on the roof? Can he plant different shrubbery or extend his lawn? Such minutia are not hypothetical but are the very real issues which consume a good deal of the time of local code enforcement officers every day.

If you take jurisdiction of land use, they are issues which <u>you</u> must decide every day, and it is a lead pipe cinch that environmental organizations will be nipping at your heels with every step. Does FERC really want the headaches that come with this responsibility? Would you set up a full time permitting office in the area? I don't think this is what the citizens of Millinocket had in mind when they emphasized the need for economic development.

I leave it to others to argue whether or not FERC has the legal authority to assume land use control authority. I question whether FERC has the <u>competence</u>. Land use regulation is not a hobby. If you accept the responsibility, you will be thrust into an arena where your jurisdiction overlaps with that of other government agencies. The simplest request can blossom into a time and cost consuming nightmare. This is the sort of situation in which environmentalists can work their manipulative machinations like a virtuoso. Unless there is a truly compelling reason for getting into land use (and none has yet been advanced), I believe you'd be well advised to stick to your strict mandate.

Once again I appreciate your giving the public opportunity to have their values considered in your process.

Very truly yours, Cur hung

David W. Guernsey



.OFFICE OF THE SECRETARY

William S. Haggerty 36 Bokum Rd. Deep River CT 06417

95 JAN 31 PH 12: 07

FEDERAL ENERGY REGULATORY MMISSION

۰.

Lois E. Cashell Secretary Federal Energy Regulation Commission 825 North Capitol St. N.E. Washington D.C. 20426

Dear Ms. Cashell. I am writing to you to inquire about the Ripogenus Hyperb Electric Project. #2572 -005

It seems that this project will impact very neavily upon the residents of Chesuncook Village. I have had a nome there for the last twenty five years. Like most people who live in Chesthousek I am a second beneration resident. In fact, many people are thind and founts benevation regidents. This evening I received a prope call from a friend in Millingsket Maine asking me if I planned to attend a meeting at Eterne migh Eindel to discuse project 2571. It is the middle of winter and Willingskat is over 500 miles awey. Many litesyncopy residents find Themselves in similar dircumstances. I am very conderned addit this lat: of notice. It would seem that those of us who are most greatly impacted sesenve the might to be neero and be informed accut this plan.

Flease each we a copy of this proposal. I will distribute a TIP - TO REPORT CHOCKNOY GHORMS, WORD TOLE IS COMPLETE HE HELD DESTRICT TO TO THE nalo e neering where we der de heerd.

I am sume that this proposal is stable for many areas or the country and it bight be viable for most parts of the floogenus area, but not for Cresuncook Village. I am centain that no one has told you shat Inequalize village is an nistorical site. Many of the nomes have been restored and maintained as they were originally durit. Mu nome is the tr the new meralting examples of a period logging deep nome. Engeundoor wet events the third less examples of the early logging and economy tame ree l'évenu archificant toopiet luige for New England.

Lasti, I would like you to evolain to me how it was decembined from the meeting in the dead of winter with less ofer 14 ocurs portice is fair on monally connect. WE WERE NOT INFORMED [0 4 7 1 (EL -FACH(CA) who is resconsible for the scheduling of meetings, and how did the, detendine who to contect?

I look forwerd to neering more jok as soon ve cosaidle concerning this matter.

William & Haggerty

⊌11.1288 **8.** ∃ejjen



Fin and Feather Club

P. O. Box 123 MILLINOCKET, MAINE 04462



SPEECH TO FERC 1/25/95

THE MILLINOCKET FIN & FEATHER CLUB SUPPORTS THE RELICENSING OF G.N.P.'S HYDROELECTRIC DAMS IN THE PENOBSCOT MILLS AND RIPOGENOUS PROJECTS.

WE FEEL THAT G.N.P. HAS BEEN A GOOD STEWARD OF THE BASINS FOR THE PAST 97 YEARS, AND WE FEEL THAT THEY WILL BE A BETTER ONE FOR THE NEXT 30 YEARS.

WE STRONGLY SUPPORT THE ALIENABLE RIGHTS OF BOTH THE CAMP OWNERS AND GREAT NORTHERN PAPER, AND WE FEEL THAT ALL EXISTING CAMP LEASES SHOULD BE GRANDFATHERED, BOTH IN THEIR ALIENABLE RIGHTS, AND CONTINUE TO BE UNDER L.U.R.C. JURISDICTION. WE WOULD WANT THE FINAL DRAFT TO CONTAIN A CLAUSE STATING SUCH, INCLUDING GREAT NORTHERN PAPER'S RIGHT TO SELL EXISTING LOTS TO PRESENT LEASEHOLDERS SHOULD THEY SO DESIRE.

WE DO NOT FEEL THAT ANY CONDITION SHOULD BE PLACED UPON THE RELICENSING OF THESE DAMS THAT WOULD HAVE AN ADVERSE IMPACT UPON THE ECONOMY, RIGHTS, AND WAY OF LIFE OF THIS AREA'S CITIZENS, AND COMMERCIAL VENTURES.

Vern A sines

VERNON HAINES, DIRECTOR

FILED

95 FEB 27 PH 12: 01

February 16, 1995

REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

, ⁰⁰⁹ Penobscot Mills Project, FERC No. 2458 RE: Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Chester L. Hamm Jr. P.O. Box 808 Ambajejus Lake Millinocket, ME _04462

Matel Ken


0

FILED OFFICE OF THE SECRETARY

95 FEB 27 NH 11: 22 FLUE HAL ENERGY REGULATORY COMMISSION

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 025

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Arthur Idansen 213 Congress St Millimordset ME 04462

FILED OFFICE OF THE SECRETARY

95 FEB 27 MII: 22

FLEERAL ENERGY REGULATORY COMMISSION

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

v'

mand ane ME 04462-

FILED OFFICE OF THE SECRETARY

95 FEB 27 AMII: 22

February 16, 1995

FEDERAL ENERGY REGULATORY COMMISSION

2.5

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Finnick h Hansen

29 Grand Que Millinocket ME 04462

ORIGINAL



95 FEB 27 MII: 21

+ LUERAL ENERG . REGULATORY

COMMISSION

February 16, 1995

: 1

Lois B. Cashell, Secretary Federal Energy Regulatory Commission ... 825 North Capital Street, N.E. Washington D.C. 20426

Penobscot Mills Project, FERC No. 2458 - 009 Ripogenus Project, FERC No. 2458 - 009 RE Ripogenus Project, FERC No. 2572 - 00.5

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

-Mulanie Hanser 1. 29 Grand Avenue Millinocket, M2 04462



Michael Harney 15 Walnut Ave North Hampton NH 03862 January 29, 1995

UFFICE OF THE SECRETARY 95 FEB -8 AM 11: 57 FEDERAL ENERGY REGULATORY OMMISSION

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - 005

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

michael Phones

Michael Harney

FEB 2 4 1995 95 JAN 27 M 10:00 FERC 825 NORTH CARTAL STREET, NE FEDERAL ENERGY COMMISSION WASHINGTON, D.C. 20426 RE: GREAT NORTHERN PAPER, INC. RIPOGENUS HYDROELECTRIC PROJECT (FERC NO PENOBSCOT MILLS NYPROFLECTRIC PROJECT (FERC DÉAR MS. CASHELL :

MY WIGE AND I PRESENTLY LEASE 3 LOTS ON AMBAJENUS LAKE WHICH IS LOCATED IN ONE OF THE PENDESCOT MILLS IMPOUNDMENTS. WE ORIGINALLY LEASED THESE LOTS IN 1968. SINCE THAT TIME WE HAVE BRONGANT UP & DAUGHTERS SPENDING OUR COMPLETE SUMMERS (DURING THE TIME SOMOCH WAS IN RECESS), MANY WEEKENDS AND WINTER VACATIONS AT OUR COTTAGE. SINCE MY RETIREMENT APAROXIMATELY 5 YEARS AGD MY WIFE AND I MAVE UTILIZED THE CAMP A MINIMUM OF 5 MONTHS /YK, AS YOU CAN SEE FROM THE ARSUTE A GOOD PART OF OUR LIVES OVER THE PAST 27 YEARS HAS ISEEN AT CAMP.

DURING THESE MANY YEARS AS A FAMILY HONE WITH FRIENDS THERE HAS BEEN A TERRIFIC AMOUNT OF PLEASURE ENJOYING NATURE - THE FEEDING OF DUCKS AND HUMMING BIRDS, WHITCHING THE MANY LOONS, CATCHING FISH, SWMMING, BOATING ON THE LAKES AND UP THE PENOBSCOT RIVER, WATCHING THE DEER, MODSE AND AT ONE TIME 4 BALD EAGLES.

<u>E-550</u>

- -1

PRESENTLY OUR CAMP IS LOCATED ARAVIT SO FLET FROM THE HIGH WATER MAKE AND IF YOU THINK IT IS AN "EYE SORE" OR IT BOTHERS THE PREABLE NATURE LOVER YOU ARE WRONG - IT IS A LOG CONSTRUCTED BUILDING THAT LOOKS VERY NECE FROM THE WATER. IF WE WERE REQUIRED TO SET IT BACK 200 OR 500 FRET AS SOME SMALL PERSENTAGE OF THE PEOPLE INVOLVED WOULD LIKE WE COULD NOT EVEN SEE THE LAKE SEEING THE WATER, WATCHING THE DUCKS, LOONS, PEOPLE CANDEND OR BOATING, LISTENING TO THE WATER LAP ONTO THE SHORE IS PART OF WHAT LIFE IS ALL ABOUT).

<u>(2</u>)

AS FARAS DOCK CONSTRUCTION BEING ALLOWED-IT WOULD BE REDICULOUS TO STOP THIS. OVER THESE 27 YEAKS I HAVE BUILT/MODIFIED MY DOCKS AT LEAST 15 TIMES AND HAVE NOT HURT THE ENVIRONGMENT ONE BIT. IN FACT WITH THE AMERICAN FLAG FLYING AT THE END IT LOOKS PRETTY DARN NICE,

ALSO OVER THIS TIME PERIOD I HAVE OUT UP APPROXIMATELY SO TREES WHICH HAVE DIED OR BLOWN OVER DUE TO HIGH WINDS, SPRUCE BUDWORM OR SATIN MOTH THIS AGAIN DID NOT HURT ANYTHING.

I HAVE LIVED IN THIS AREA FOR MOST OF MY 61 YEARS AND HAVE FOUND GREAT NORTHERN PAPER TO BE A VERY RELIABLE COMPANY AS AN OPERATOR OF THE DAMS IT CONTROLS AND WORKING WITH THE PEOPLE IT LEASES LOTS TO FOR THE ENJOYMENT OF NATURE.

A PERSON LEASING A LOT TODAY SHOULD ALSO HAVE THE SAME OPPORTUNITY TO BUILD WITHIN THE PRESENT GUIDE LINES IN ORDER TO ENJOY THE SAME

3 THINGS THAT OTHERS HAVE HAD THE PRIVILEGE TO GASOY. THERE IS NO NEED (OR SOUND LOGIC) TO BE 200 OR 500 FEET BACK FROM THE KIGH WATER MARK (THAT IS LIKE WATCHING A FOOTBALL GAME FROM OUTSIDE THE STADIUM) IN SUMMARY, ALLOW BREAT NORTERN TO CONDUCT BUSINESS LIKE IT HAS IN THE PAST AND ALLOW THE PRESENT AND FUTURE LEASE HOLDERS TO ENJOY THEIR LOTS AND CAMPS WITHOUT INCREASED REGULATIONS OR RESTRICTIONS. IT HAS BEEN FINE FOR YEARS AND WILL CONTINUE TO BE FINE IF A FEW PEOPLE LEAVE IT ALONE. SINCERELY,____ Walter R Hal E-552

Dan & Deborah Hoffses RR #2 Box 1310 New Gloucester ME 9426RSAR, INC January 29, 1995 FEB 2 4 1995 ECOLOGICAL SCIENCES AND ANALY

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 ~005

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

· How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

 Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

 Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely, June R& Datanh L 13/1/m Dan & Deborah Hoffses

	FILED OFFICE OF THE SECRETAR	Y	e Presidente de la companya de la comp	- 4
e da la constante de la constante	95 FEB 27 MII: 43)		
February 16, 1995	EDERAL ENERGY		ि ित्रासट्टेस्टन्ट्रस्ट्रान्स् २	
	COMMISSION		1999 - 1999 - 1997 - 2020 - 2020 1999 - 1979 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	

009

Lois B. Cashell, Secretary Federal Energy Regulatory Commission -825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 005

This is the second

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Ĵ

Deucithy Thimieson

FILED

•

- - -

95 EEB 27. AM 11: 42

February 16, 1995

REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Jon Jamieson

 $\dot{\nu}$

February 16, 1995

-- -

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

.009

OFFICE

95 MAR -6

FEDERAL ENE

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Carolyn J. Johnson



February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

./

Stonald M. Johnson

,023

95 TAR

009

FEDERAL ENE REGULATOR







COMMISSION

January 26, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, M.E. Washington, D.C. 20426

09

RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458)

Dear Ms. Cashell:

I am aware that Great Northern Paper has applied for new 30 year licenses for projects which are known as the Ripogenus Hydroelectric Project (FERC No. 2572) and the Penobscot Hills Hydroelectric Project (FERC No. 2458) and that the Staff of FERC has recently issued a Draft Environmental Impact Statement against such.

I lease a lot from Great Northern Paper on the Penobscot Hills Impoundments. As a leaseholder, I wish to express my opposition to the restrictions and all future restrictions imposed by the Federal Energy Regulatory Commission.

Cordially,

Donna Kelly Leaseholder

	VERSAS, INC	<u>.</u>	
	FEB 2 4 15:5	;	
:00L	OBICAL SCIENCE		• .

OFFICE n 95 FEB -3 PH 2: 05 FEDERAL ENERGY REGULATORY COMMISSION

January 27, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobacot Hills Hydroelectric Project (FERC No. 2458) - 009

Dear Ms. Cashell:

I an aware that Great Northern Paper has applied for new 30 year licenses for projects which are known as the Ripogenus Hydroelectric Project (FERC No. 2572) and the Penohscot Hills Hydroelectric Project (FERC No. 2458) and that the Staff of FERC has recently issued a Draft Environmental Impact Statement against such.

I lease a lot from Great Northern Paper on the Penobscot Hills Impoundments. As a leaseholder, I wish to express my opposition to the restrictions and all future restrictions imposed by the Federal Energy Regulatory Commission.

Cordially,

Dora J. Kelly

Dora Kelly Leaseholder



95 MAR -2

FEDERAL ENERG

REGULATOR COMMISS

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Kathy Kenneson Creative Capier

(Retail Office & Lift Store

2

Contraction of the second s	
OFFICE OF FILED	
95 FFR 20	
11 CO 28 PM 1:43	
REGULATORY	
SION	

004

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscat Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 -005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Lenda & Labby



Tom Lambert 93 Mill Road North Hampton NH 03862 January 29, 1995

UFFICE OF THE SECRETARY 95 FEB -8 PH 12: 11 FEDERAL ENERGY REGULATORY COMMISSION

• ••• · ·

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 – DUD

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

incerely, Tom Lambert



95 FEB - 3 PM 1: 4 J FEDERAL ENERGY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N. E. Washingto D.C.

Subject: Ripogenus Hydroelectric Project (FERC # 2572 – 002Penobscot Mills Hydroelectric Project (FERC # 2458) – 009

Dear secretary:

ā

- · · **C**

I am writing this letter in support of Great Northern Paper Co. and in protest of the proposed building setbacks and vegetative buffers recommended by opponents of the relicensing of these projects.

I am familiar with the licensing process having been the project manager of the Big A Project which was proposed by Great Northern Paper. I am also familiar with the self serving interests of the environmental groups opposing the relicensing.

Great Northern Paper and the State of Maine have done an excellent job of managing the project areas over the past 95 years and proven that no added restrictions or control based on the demands of special interest groups is necessary or desirable. The fact that these projects were built and managed so well is the very reason that the excellent fisheries and other wilderness experiences are available to the public.

I own a camp on one of the lakes and have concern as to the implications of the proposed protection zone. We residents of this part of the state of Maine have chosen to live and work here enduring the hard winters and short summers to be able to enjoy the lakes, forests and wildlife God has given us. We do not need special interest groups placing controls on the area where we have chosen to live.

The controls governmental bodies have placed upon us as landowners and camp owners on leased land are already too restrictive and taking away further rights to safe guard aesthetics for the sake of visitors to the area, who live in cities and communities where they have the luxuries of life that they want, is unwarranted and selfserving on their part.

Finally, no American company should be forced to spend the years and Millions of dollars to continue their operation as a company that this relicensing process imposes. While americans are losing their jobs and the foreign trade deficit is growing, companies like Great Nortthern Paper are forced to spend wasted dollars driving up the price of their products making them less competitive. We as a country cannot afford to be so wasteful of our time and money and let foreign competition out us as a country out of business. We will become a country of monuments to place on the federal register for visitors to come and see and learn a lersson from.

Please be considerate of the individual resident of this area of our great country in your decision making.

Yours truly Galen Løgger

୍ର 🛛 ୮୦୦ରି Federal Energy Regulatory Commissio 825 North Capital Street, N.E. ECOLOGICAL SCIENCES AND ANALYS Washington, D.C. 20426 009 Re: Ripodenus Hydroelectric Project (FERC No. 2572)

Penciscat Mills Hydroelectric Project (FERC No. 2438)

Cear Sirs:

1

I am writing this letter in hopes I can receive some assurance that the recently issued traft environmental impact statement will not affect me and my seasonal dwelling on South Twin Lake which is part of the Pencosict Mills Hydroelectric Project. As I have reviewed the populary it is apparent to as that the environmental impact study if FERC established a protection zone around these waterways at a depth of five rundred feet. From the high water mark. i aiso inderstand that FERC staff has studied a two hundred foot building set back and a possible one hundred foot vegetative buffer as part of a second alternative. I would like to have my input on the implementation of these proposals.

I prw have a season dwelling in a lot that at no point reaches two Now dred feet. At the most, from the high water mark my lot is one -unared fifty feat.

Yow do you shopped by season owelling will be affected which a two numbred foot tet lety tone? Does that mean I have to up root my case and out it on acheone else's property just simply to satisfy vour regulations? Also, I have not come across any notation of a grandfather clause, consequently, if these buffers go into affect it would totally negate my lot and render it valueless. This concerns we since over the last two years I have put in over forty thousand dollars in renovations.

I would like some indications that by property will come under a grandfather clause as it relates to these buffer zones.

I have been renovating my camp in a three year project clan. At Federal Energy Regulatory Commission Re: FERC 2572 & FERC 2458 January 20, 1995

and a second
this point my camp is done and I am happy to say I have done all renovations according to LURC licensings and permits. I am totally legal. My next step is to place, through proper permitting, a garage on this property as well. But, according to your impact studies and statements, I will not be able to build any new buildings or docks within the two hundred foot set back zone. Here egain I am stymied, and the simple reason I am stymied is I simply to not have two bundred feet. I need to have some adsurance, that stoke I do not have that much land, that I will still be able to itild a muchly needed garage. I feel it is foolishness that this and product oureaucracy is occurring. I have been on this sight for over ten years and have enjoyed it immensely, and now I have some pancil busher from Washington telling me that I am going to lose my dream, this is absurd.

• •

My int is heavily wooded in the back end and since I do not have but whiches feet, that of my int falls within the one number foot the the cyling tree that is narmful to my grandchildren and thildren you are telling my I can not remove it! This again is absure, if I move a particular tree that proposes a hazard to me and my family, I should have the right to dispatch that tree without breaking the law. If this law does go into affect, and such a situation occurs, whether it's my property or someone else's and the bazard is not properly disposed of there could be law suits bending for the particular agency that is in control. Please control that twist to your vegetative tuffer gone.

Used: . Eites I do dit dave anything that goes beyond one bundred wit fift, feat from the high water mark, your proposed two hundred from building set back and one hundred foot vegative buffer renders to ist and ty samp totally worthless. I do not want nor do I wish to lose my dream of my seasonal dwelling or see it altered by sureaucratic tampering.

Flease use this latter as my view point concerning this and please file it where it needs to be filed Phowing full well they need to be in your hand by February 3, 1995. Flease be considerate to the

Federal Energy Regulatory Commission Re: FERC 2572 & FERC 2458

January 20, 1994 Page: 3 · . .

land owners and lease holders of this great area in the state of Maine. .

10

Thank you for your time and consideration in this matter.

. .

Sincerely, formallant X

Lawrence Lankhorst, Jr. Chiropractor

L1/sh

.

DEFICE OF THE SECRETARY

95 FEB 28 PM 1:45

ELEFAL ENERGY REGULATORY

COMMISSION

009

February 16, 1995

Lois B. Cashell, Secretary Pederal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 -005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

-Very truly yours,

Mathalie Leet

FILED OFFICE OF THE SECPETARY

95 FEB 27 PH 12: 01

February 16, 1995

REGULATORY

,00⁹

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Carol Mackin

Carol J. Mackin P.O. Box 808 Ambajejus Lake Millinocket, ME 04462



COMMER

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

~ 0⁰⁴ Penobscot Mills Project, FERC No. 2458 RE: Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

υ

syme M. macking

FILED OFFICE OF THE SECRETARY INC

ĥ

ECOLOGICAL SCIENCES AND ANALYSIS

95 FEB 28 PM 1: 43

February 16, 1995

REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission -825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Paul C Mpi 15 Men yulist Muinaket Me

VER	Louisa P. Malizia P.O. Box 215
	22K5
ECOLOGICA	FILED FILED GENCES AND ANALYSIS GEFICE OF THE SECRETARY March 30, 1995 GENCES AND ANALYSIS GEFICE OF THE SECRETARY MArch 30, 1995
Fe 82	ieral Energy Regulatory (95mill) -5 Mili'll
wa RE	Ripogenus Hydroelectric Project (FER? No. 2572)
De	Penobscot Mills Hydroelectric Project (FERC No. 2458)

As leaseholders from Great Northern Paper Co. on the shores of Caribou Lake in the Ripogenous Impoundment, we are very distressed by the continual encroachment of government. solving problems which they and the elite Conservation Societies deem important.

Over the thirty years that my father and I have held this lease the costs and restrictions have increased monumentally, while the so called "wilderness experience" allows every snowmobiler, vandal (we've been vandalized twice) and bozo to cut through our lots, cut our trees, use our firewood, use our spring water. break down our doors and steal equipment -- all on our nickle.

We are totally and irrevocably opposed to any government rules, regulations, laws, etc., and as soon as possible will sell our lease and cabin. It's just not worth it any more. ₩e are retired and we certainly do not need government or conservation people telling us what to do or how to do it.

We have been responsible stewards of this land for 30 years, and now must pay for the arrogance and stupidity of the Federal and State governments and the elite conservationist movement.

We will take our chances and let a new owner demolish our architect designed cabin, and let the land revert to the woods primeval, so that today's Thoreaus from out of state can come and contemplate their navels.

We have always been happy with Great Northern and we certainly wish them well in their dealings with the government.

Very truly yours, Lauiza P. Malign

Louisa P. Malizia Joseph P. Malizia

Dean A. Beaupain, Atty. cc: Great Northern Paper Company

This letter is arriving a but late, Din april that perhaps you could find comeone there who might 57 be interested. Thank

VENOAN, INC.

FCNI DGICAL SCIENCES AND IN

FEB

Grace Maloney Box 804 Madison CT 06443 January 29, 1995



Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - 005

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

- Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely, Grace E. Maloner Grace Maloney



Michael Maloney Madison CT 06443 January 29, 1995

UFFICE OF THE SECRETA 95 FEB -8 PN 2: 05 FEDERA

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - 00

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

· Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

- How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

- Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely Michael Maloney



Richard & Lenore Maloney FILED OFFICE OF THE SECRETARY Madison CT 06443 January 29, 1995

95 FEB 10 AM 9: 43

AL ENERGY LATORY

HISSION

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 -005

Dear Secretary, FERC;

Ń

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

· Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

- Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

· Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

· How will the value of property be affected by the proposed setbacks and vegetative buffers?

- Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

 Will relicensing opponents and federal agencies convince FERC. to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

 Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely,

Richard & Lenore Maloney

Senor Wal

FILED OFFICE OF THE SECRETARY

1.2.2

95 FEB 28 PH 1:43

February 16, 1995

FEREFAL ENERGY REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours Georgen Mary

V



Lois B. Cashell, Secretary Federal EnergyRegulator Commission 825 North Capital Street, N.E. Washington D.C. 20426 001

Dear Secretary:

I understand that the FERC is considering additional restrictions on land adjoining the Ripogenus Lake Hydroelectric Project, (FERC 2572/ relative to the relicensing application from Great Northern Paper, Inc.

I have a leased lot in this area and am interested in knowing what impact the new restrictions will have on the use of my cabin and lot.

Will I be able to continue using my lot, can I continue to clean up down trees on my lot, and a lot of other questions.

I am opposed to the increasing of restrictions until I know how they will affect me.

Thank you for your help.

Sincerely vours



Chesuncook Lake House

Bert & Maggie McBurnie Rt. 76, Box 656 Chesuncook Village Greenville, Maine 04441



Daily • (207) 745-5330 (8 p.m. - 10 p.m.) • Preferably

Federal Energy Regulatory Commission P-2572-005 P-2458-009 825 North Capital St. N.E. Washington DC 20426

Dear Ms. Cashell;

After attending the Jan. 25th public hearing of the FERC board on the Ripogenus Hydro project at Millinocket Maine, I would like to comment on the proposed regulation of the 200ft building set back and looft vegetative zone.

Since 1935 I have been a resident of Chesuncook Village, which is located on the north west shore of Chesuncook Lake, the major part of the Ripogenus inpoundment in question. Here, for nearly 40 years now, my wife and I have owned and operated a lodge known as the Chesuncook Lake House. This old inn, having been in existence since 1863, and along with the village, is on the Federal Historical Register. This small hamlet comprises of fifty odd lots and buildings, owned not leased, by individuals from far and wide. Chesuncook Village is primarily accessible by air or water only. A logging road ends some four miles south of the village, .From there only tractors, log skidders or special four wheel drive vehicles can negotiate.

I am very concerned with the 200 ft building setback as over 50 percent of the village lies within its boundaries.

We were told, by a FERC member, that all present structures are Grandfathered. I cannot find any reference to this in the DEIS I have studied.

If this clause exists, could you please explain how it will effect the private owners here in this village. Will we also be allowed, as in the past, to have temporary floating docks to load unload and moor our boats to?

I would like to be placed on the mailing list of any future proposals, also could I have a copy of the present DEIS, as the one I have is borrowed.

I leave you with the hope that FREC will speedily issue a license for the Ripogenus Hydro project, allowing us to deal with facts and regulation rather than the threatening proposals we hear.

Sincerely

Chesuncook Lake House

FILED OFFICE OF THE SECRETARY

95 FEB 28 PH 1:44

February 16, 1995

REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project. FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penohscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

1

Madin Milleton

and the second s

OFFICE OF THE SECRETARY

95 JAN 23 PH 12: 14

FEDERAL ENERGY REGULATORY

OMMISSION

Maurice C. McLean 194 Highland Avenue Millinocket, ME 04462

January 20, 1995

Federal Energy Regulatory Commission 825 North Capital Street NE Washington, DC 20426

Attn: Lois B. Cashell Secretary

RE: Repogenus Hydro-electric Project (FERC No. 2572) Penobscot Mills Hydro-electric Project (FERC No. 2458)

Dear Mr. Lois Cashell:

I have great concerns on the proposal by FERC to establish a protection zone on the impounded shoreline of waters known as the Ripogenus and Penobscot Mills Hydro-electric Projects.

I have been a camp owner on land leased from Great Northern Paper Company for more than thirty years and I hope that my children will own the camp for another thirty years.

We presently have the State Land Use and Regulations that are more than adequate to preserve the impounded water shorelines. We do not need or want the Federal Government taking over what should be and is a State responsibility.

Additional restrictions on Great Northern Paper Company will prevent the mills from being competitive and restrict any further economic development in this area which is greatly needed.

These restrictions should be removed from the Draft Environmental Impact Statement.

Sincerely.

Taurie

Maurice C. McLean

CC: Senator Olympia Snowe Senator William Cohen Rep. John Baldacci Rep. James Longley, Jr. Rep. Herbert Clark


RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458)

At this time we would like to voice our concerns regarding the Draft Environmental Impact Statement which will affect all of the lease lots on North and South Twin Lakes.

We have owned and paid the cost of a leased lot since the summer of 1981 waiting until the appropriate time prior to our retirement when we could begin building a cottage. We have held the land regardless of the fact that there is no electricity knowing that there are to be no future available lots on any one of these lakes. We only have several years to go before retirement, and are planning to start building at the first opportunity.

If this ruling is accepted, we will not be able to build on the lot we presently hold. We all realize the concern with degradation of water quality. You also need to understand that those presently on these lakes are concerned daily with these problems and have been for years - this is not a new concern to any of them. They take special care in observing all of the existing regulations and you must admit that they have been extremely successful over a very long period of time. There are not that many lots still existing that have not been built on - why at this time are you considering additional building setbacks and vegetative buffers?

We are planning to submit a potential plan for a retirement cottage to Lurc sometime this summer and would sincerely hope that consideration will be given to the many years we have held this lot, paid for it, and been very concerned with the environment and everything that keeps it healthy/clean/ and something to be enjoyed by families in the area.

Most if not all of the opposition to the current existing plans come from people/s who are not involved with the lake and its many existing and planned homes. They cannot know the great care that is currently being taken to keep it the way it has been for many years. We are not dealing with condominiums or retirement villages Ms. Lois B. Cashell, Secretary Page 2

or large extravagant additions that would substantially affect our water - these are the same families who have lived in the Millinocket area for years and plan to remain there for many more. We find it extremely difficult to understand the logic behind any of these new rules and regulations being proposed.

It would be a devastating blow to us at this time to learn that we would not be able to build the summer cottage we've dreamed of. Please reconsider any changes you may be considering for the existing lots on these lakes.

Sincerely. laven O. meny

Robert A & Charlene O. Meininger

cc: Senator William S. Cohen Senator Olympia J. Snow Congressman James Longley, Jr. Governor Angus King

١

OFFICE OF THE SECRETARY

95 FEB 27 AM 11: 44

February 16, 1995

ELLEFAL ENERGY REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572_005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

_ Very truly yours,

MU xue

John H. Michaud Millinocket ME



.0⁰⁹

FILED OFFICE OF THE SECRETARY

95 MAR 24 PH 1:44

Box 326 East Baldwin, ME 04024 March 11. 1995

RECULATORY COMMISSION Lois G. Cashell. Secretary Federal Energy Regulatory Commission 825 North Capital Street. H.E. Washington, D. C. 20426

Dear Ms. Casnell.

,0⁰⁵ RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penopscot Mills Hvaroelectric Provect (FERC No. 2458)

In 1981 my huspang ang I executed a 99 year lease of land from Great Northern Paper. Inc. on which we pullt a seasonal camp. This leased land is on Turkey Tail Lake. That Lake and the surrounding land are apparently considered part of the Rippgenus or Penopscot Mills Impoundments for which new, retroactive restrictions are being considered in the context of the referenced religensings.

I am writing to oppose any such new restrictions as pelno unwarranted and inappropriate and as an infrindement on my right to use and endoy my property.

Dur camp means a great deal to our family and friends. Although its remote location means we can only visit it a few times a rear for limited periods, those have all been duality leisure times in an otherwise difficult existence. You see, my nuspand and I, and thus many of our plosest triends are deat. The damp has been a haven for us. For our two sons, both nearing persons, the damp sympolizes the ovs of outdoor living and family togetherness that we were able to provide them because of our lease.

The damp has become even more important to us recently. In husband died of a heart attack at age 53 last Octoper. The came meant so much to him that we had a drawing of the URE econed on nus dravestone as a perpetual remembrance. Having lost my nuspang and the poys, father, the threat of the opponents of relidensing to take our gamp is especially cruel. It is also an unwarranted taking of our property alanca. There is simply no way we could afford now to repulse our camp to meet some new set of requiations.

To aroue that the handful of camps on Turkey Tall Lake and neighboring lakes impose a threat to the water quality of those lakes of the envricomment in general of the areas SHERS DITERT & IDSULG. We view the proposals of the separente of revidensing as an effort to steal from us. In

our case a modest Maine family. Our property. We use the FERC to reject the restrictions being proposed by organized lobbying groups for their own nefarious objectives.

Thank you for your consideration of our plea.

Sincerely.

Des Ot Mori-

. -

cc: Dean A. Beaupain. 4 Hill Street. Millinocket ME 04462 Senator Olympia Snow Representative James Longley

95 FEB 27 - PH 12: 13

REGULATORY

FICE OF

IL FIT

SECRETARY?

00

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.



Jonuary 27, 1995-Low B. Cashell, Secretary Federal Energy Fegulatory Comm 825 Morek Capital St. M.E. 95 FEB -2 AM 9 28 Washington, D.C. 20426 EDERAL ENE Re: Ripogenus Hydroelectric tro (FERC No. 25/2)-004 Penoloscol Mills Hydroelectric Project ECOLOGICAL SCIENCES AND ANALYSIS <u>(FERC</u> No. 2458)-009 He are land leaseholders from Great Northern Paper Co. (Bowater). It strongly oppose the unwarranted restructions and vigetative buffers as proposed in the above captioned matter He also urge the relicencing of the dams for the benefit and welfare of the people who depend on Great Northern Paper to for their livelihood. CC; Sex, William Cohen Mrimrs Robert Mosca Bep. John Balilacci P.O. Brk 969 Millinocket, Maine

04462

COPY

VERSAR, INC.

FEB 2 4 1925



Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

- Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely,

Charles Paul

Charle a Brid

FILED OFFICE OF THE SECRETARY

95 FEB 28 PH 1: 48

February 16, 1995

REGULATORY

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 ____005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

V

Beverly Kelletier)

2458-009 Cupe Common R. 329 2527 Second Ariell OFPICE OF LERETARY ~ 5.2 Please Except This NOTE yearsy my goosino of your Days: (FEAR REGUL bly sarance and VELETATIVE BUFFEL. I make pageery locare on the societies Ausser Mills pajeer (Myons) and it seems carsmi Enour to me, that the numerous pine trees. for 6" to 36" , a Diameric ther some ou My Lots, WHICH is an a point, at South Turi LAKE. This point is appreximately 100' white. THESE TREES of 24" to 36" DIAMOSTRE TOOR MAN. many years to grow to this size. FUTTING IT USED BLOWTLY, FERC DIDN'T Exist 1/20 Th of the time those there's Howe, so I see Apsolutely, positively NO NEED for FERC'S so eall'so proposal. Ture You Alter WINTER) THOMAS FELLETIER 8700 Rissanco Ars A-402 RESIDENCE) Cape Commense, R. 32920 Summer HC 74 BOX 557 Millinger, ME. 04462 (OVER)

FER 24

FILED OFFICE OF THE SECRETARY

95 FEB 27 AM 11: 32

February 16, 1995

EDERAL ENERGY REGULATORY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

edian Mi, an he 15 Chio St Millen Leket, ME 04462 S.T- 723-6202

February 16, 1995

. . . .

n, 👬

. .

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: -Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

DFFICE 8

001

95 MIR - 2 - 18 9

FEDERAL ENERGY REGULATORY

COMMISSION

E SECRETA

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

Line Ming Cheatine Repair



UPPOR OF THE STOTETARY

95 FEB -9 AM 10: 01

CHERGY CHERCENT COMMISSION Philip and Carol Ramu P.B. Box390 Mt. Sinai, N.Y. 11766

ECOLOGICAL

VFRS

FEB 2 to the s

January 31, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Carital Street, N.E. Washington, D.C., 20426

Dear Ms. Cashell:

V

I am a leaseholder with Bowater/Great Northern Paper Co. in Northern Maine. We have been informed of the current proposals that FERC has put forth in response to licensing applications introduced by Great Northern (Ripogenus hydroelectric project, FERC # 2572 & Penobscot Mills Hydroelectric Project FERC #2458).

Please be advised that we are against any legislation that would cause setbacks for our home in Maine. The current Maine State LURC regulations are more than sufficient in protecting the environmental concerns. Additional setbacks are not indicated and would cause great problems for myself and other fellow homeowners in the Penobscot Region.

Flease allow Great Northern the licenses they require without adding additional burden to the citizens who enjoy the current pleasures of life in this area of Maine.

Sincerely,

Philip & Carof M, Rami

Philip M. Ramu Carol M. Ramu

COPI Mo Xois Cashell FERCENCE FILED 2/6/95 Nachington, DC 20426 PHR: 17 FFR ZLERS Den Mo. Cashell ECOLOGICAL SOLL ST fam writing in appoint to the Gue Northern Japan Co. Dans Re licensing Represent 5 toto of Mainle Sur who who was the 005 impoundment apen Ripogenes Dam 1.9 Chemmerk Lake, thet in Millimoplet at a public hearing regarding FERC- 2572. about a hundred people got to speak. Only 5 on 6 were in farms of the provision of the project the rest were against it. For the life of me & cannot understand why FERC has any fusines forcing a private obvined people conforcing to use fiderall powers of eminent domain to enforce a 200 ft fullor. State of Mainle ... eminent domain to enforce a 200 ft fuffer strip along the slores of Cherman Loke. I believe that Maine's Land Use Regulatory Commission has done an excellent job of regulating water quality and land use in the region . I am a lence on the lake and have been since 1970. We have summer place there. Liked in the area. I am extrem by concerned of our will lands generally. I have never met a land owner on leave on the lake who does not feel as I do. In fast & have never even heard of an infrastion under LURC rules and regulations. regulations theteen that all of us in the impoundment area deserve a lot of credit for our (192) E-594

2.00--Contractor - 1.1 - 0 31 ----J-Jis-a-vis-5mmil . _____ ••••• madrie 6 Court ST **.** Earnington Me 0493

OFFICE OF BELECTMEN 53 Main Street East Millinocket. Me 04430-1199



, TEL. 746-3376 The FILLED X 746-3550 95 JAN 37 ANII: 41

January 25, 1995

Lois D. Cashell, Secretary TEAT VE Federal Energy Regulatory commission 025 North Capitol Street, NE Washington, DC 20426

RE: FERC PROJECTS #2572 (RIPOGENUS) AND #2458 (PENOBSCOT MILLS)

Dear Secretary,

Below is my testimony as prepared for presentation at the FERC public hearing on the Draft Environmental Impact Statement, Wednesday, January 25, 1995, in Millinocket, Maine:

My name is John Rouleau and I am Chairman of the Board of Selectmen in the Town of East Millinocket, an Employee of Great Northern and a campowner. I would like to welcome the Commission to our Region and thank you for allowing the public to speak at this public hearing.

There are two issues in this DETS which I hope the Commission will reconsider before issuing the final license to Great Northern:

- a) 200' setback along the water areas; and
- b) the requirement of FERC for GNP to purchase land that is not already in their possession.

As others have said before me, I reiterate that:

- a) The State of Maine, Land Use Regulation Commission very adequately addresses all of the environmental concerns concerning shoreland properties and timber harvesting.
- b) The Commission has grossly understated the market value of the 54+ miles of land which will need to be purchased. Therefore, the additional several million dollars which the company will be required to spend should be directed toward the mill expansion to secure the economic stability of The Katahdin Region, Penobscot County and The entire State of Maine.

I would like to thank the Commission for the decision <u>not to</u> <u>require</u> Great Northern to put flows down the back channel.

Thank you for your time tonight and for expediting a decision on these licenses.

Sincerely.

Jefen S. Rouliauga John E. Rouleau Chairman, Board of Selectmen





V

	VERSAR. INC.	3
21 Jan Lois B Federa 825 No	Lengy Regulatory Commission	OFFICE OF THE SECRETARY 95 FEB - 3 PM 1
Weshi RE:	Ripogenus Hydroelectric Project (FERC No. 2572) - 002 Penobscot Mills Hydroelectric Project (FERC No. 2458) - 009	FEDERAL ENERGY REGULATORY COMMISSION

Dear Ms. Cashell:

We recently received a letter from Mr. Dean A. Beaupain, Esquire informing us that "opponents of relicensing and certain federal agencies recommended to FERC that it create building setbacks and vegetative buffers" within the watershed of the dams listed above, and that from these comments, "FERC, as part of its environmental impact statement, studied establishing a protection zone around the impoundments to a depth of 500 feet from the high water mark" in Alternative #1 and 200 fact in Alternative #2.

We have been leaseholders with the Great Northern Paper Company since 1981, leasing a two acre island lot located in Indian Purchase T4, North Twin Lake. FERC's proposed increase to restrict a building setback from the current restriction of 100 feet, for "shorelines of a flowing water draining 50 square miles or more and a body of standing water greater than 10 acres in size," as written in the Maine Land Use Regulation Commission's, (LURC), Building Pennit Application, 9/94, would make it impossible to build any structure on our island lot and without a structure we will never be able to transfer or sell the lease. We have a considerable amount invested in this property and ironically were planning to start construction on a small camp this summer. If FERC should proceed with establishing a setback beyond that currently required by LURC, our property will be deemed valueless, and we would lose not only the dream of a modest cabin on the lake, but the thousands of dollars we have invested in this dream. We believe that this would infinge upon our rights as leaseholders and law abiding citizens of the United States. If after 14 years we are denied the opportunity to construct a camp, due to new government regulations, we will be forced to seek whatever legal recourse possible to either receive an exception or sue for reimbursement.

We have family and friends in Millinocket who are dependent upon the Great Northern Paper Company. Without the relicensing of the dams, it is our understanding that the company would be forced to either burn greater quantities of fossil fuel, having a greater impact on the environment and the already cancer laden population, or the mills could ultimately close. By adding more government regulations to the ridiculous skyrocketing land use fees, it is the local leaseholders, not the companies, who are penalized.

To the best of our knowledge, the Great Northern Paper Company stopped giving leases for shorefront property back in the early 1990's, therefore, only those leaseholders who haven't yet built will be adding to the visual impact to the lake shorefronts and reduced water quality. For anyone who has not been in the Maine woods recently, 100 feet of dense Maine woods is ample buffer to obscure any man-made obstruction from the water. In addition, if gray water is properly drained and filtered into the sandy soils of the terrain, there should be no resulting loss of water quality. To control pollution and water quality, we suggest that the federal regulationists think more about limiting gasoline motor boats, sulfur smokestack emissions, and paper company clear cutting, where the chemical and visual pollution is most prevalent.

Please leave the existing Maine Land Use Regulation Commission setbacks as written.

Sincerety Wilf. Contreme Celer tu / Sand

Patrick J Santerre Celeste M. Bard 604 South Race Street, #2N Urbana, IL 61801



SAWYER ENVIRONMENTAL SERVICES

TEL: (207) 947-5100 + FAX: (207) 947-5822 115 FRANKLIN STREET + BANGOR, MAINE 04401-4938

FILE PLOT NO. 20

VE-SAR INC

TOOLD GIGHT UDIENCES AND ANALYSIS

Fig. 19075 February 3, 1995



Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capitol Street, N.E. Washington, D.C. 20426

RE: COMMENTS ON DRAFT ENVIRONMENTAL IMPACT STATEMENT/ RIPOGENUS AND PENOBSCOT MILLS HYDRO ELECTRIC PROJECTS. (FERC NUMBERS 2572 AND 2458) -009

n05

Dear Ms. Cashell:

By this letter, I am writing to provide my comments on the Draft Environmental Impact Statement. As the President of a Maine business with a strong track record in protecting Maine's environment through proper management of solid waste, and as an employer of 92 employees, I know the importance of this relicensing to both Maine people and to Maine's environment.

Great Northern's hydro system provides a guaranteed flow of water at Millinocket of 2,000 CFS on which the municipal and industrial wastewater treatment facilities on the Penobscot River rely. Moreover, through management of the cold waters from the bottom of the Ripogenus Impoundment, Great Northern's hydro system provides clean, cool waters for a world-class landlocked salmon fishery below McKay Station. Some of the best whitewater rafting on the East Coast can be found in the Gorge on the West Branch.

With regard to the specific determinations reached in the Draft Environmental Impact Statement, I praise FERC for a careful review of the Water Use Plan. FERC has reasonably concluded that it is unnecessary to spill water down the Upper Gorge or the Back Channel (beyond what the Water Use Plan, as revised, already requires). This conclusion is particularly correct in light of the substantial fisheries habitat and boating opportunity available elsewhere on the West Branch.

As a business owner who has been through the environmental permit process a number of times, I am concerned, however, that FERC not give in to arguments made by intervenors in environmental permit proceedings. The requirement that Great Northern acquire a 200' strip of land all the way around every one of the Ripogenus Impoundment lakes is disturbing to me. Great Northern, and all the other landowners in the Ripogenus Impoundment, have a proud tradition of limiting development. LURC's current zoning, as I understand it, limits most development on these impounded waters to no more than one cottage for every mile of shore frontage. This suggests to me that the Maine Land Use

Printed on recycled paper

Lois D. Cashell, Secretary February 3, 1995 page 2.

Regulation Commission is doing its job.

In these days, when both the State and Federal Governments are trying to reduce the number of mandate they impose, it is unclear to me whey this specific mandate is necessary or should be imposed given the adequacy of the State's protection. If there is any evidence that the water quality of the impounded waters has been in any way impaired, there would likely be greater support.

I ask that your take these comments into consideration as you prepare the Final Environmental Impact Statement. I would be happy to address any questions that you may have.

Sincerely. The for

W. Tom Sawyer, Jr. President

DWL

FILED OFFICE OF THE SECRETARY

VERSAR, INC FEB 24 F 20010070

95 FEB 16 AM 11:56

FEDERAL EMERGY Lois D. Cashell, Secretary FEGULATORY Federal Engery Regulatory Commission HHISSION 825 North Capitol Street, N.E. Washington, DC 20426

Re: Ripogenus Hydro-Electric Project FERC #2572 09 Penobscot Mills Hydro-Electric Project FERC #2458 Please accort to

Please accept these comments on the captioned draft environmental impact statement.

Millinocket, East Millinocket and Medway are dependent on Great Northern for the economy of our area. We are isolated geographically and have few opportunities to diversity.

Past operation of the dams have proven their environmental and economic worth to the entire state.

I want to commend you for approving Great Northern's water use plan with minor modification. Dedication of the west branch to multiple use is appropriate and will allow Great Northern to effectively compete in changing global markets.

We hope to see Great Northern embark on a massive modernization plan in the next few years. Your decision will certainly lay the groundwork for Great Northern to move forward.

As Great Northern moves forward, we expect our economy will improve and that we will regain some of what has been lost during the recent past.

I want to commend you for the job you have done but I would like to point out that you seem to have misconstrued not only the ability of the Land Use Regulation Commission's Rules to protect water quality and aesthetics but also the right of the State of Maine to act in these areas.

Lurc rules govern over one half of the state and have proven their worth in protecting the quality of our water as well as aesthetics for recreation and all other purposes. Those rules have been and are sufficient to protect these impoundments and there is no reason to have different rules here than apply elsewhere in Lurc's jurisdiction.

On a more fundamental level, it is Maine's job to protect these impoundments. Maine has done a good job and there is no basis for you to usurp Maine's authority and judgment in these matters.

COPY

Expansion of project boundaries is not needed.

I think the present level of use of the impoundments shows at present level of development does not hurt recreation. Lurc rules provide that new development must be done in a manner that is compatible with the environment. Therefore, any new development would not adversely impact recreation. Our economy has suffered over the last few years and we need to diversify. Some level of development in accordance with Lurc's rules should be allowed.

Your draft was very unclear concerning existing camps and leases. Please answer the following questions:

- (1) will campowners be allowed to maintain their existing buildings and docks within the proposed 200 foot setback zone;
- (2) will campowners be allowed to construct any new buildings or docks within the 200 foot setback zone;
- (3) will Great Northern be allowed to sell the leased land to campowners during the term of the new licenses should it decide to sell and campowners decide to buy;
- (4) will campowners be allowed to remove any vegetation within the proposed 100 foot vegetative buffer;
- (5) how will the value of the campowners property be affected by the proposed 200 foot building setback and 100 foot vegetative buffer;
- (6) do you intend to increase the proposed building setback to 500 feet and the vegetative buffer to 250 feet;
- (7) will campowners have to deal with Lurc and FERC concerning future changes to campowners property.
- (8) If existing camps, docks and leases are grandfathered, what does "grandfathered" mean:
 - (a) if a camp burns, can it be rebuilt.
 - (B) can an existing camp be expanded.
 - (C) can new accessory buildings be built.
 - (D) can vegetation be cut in front of existing camps.

Sincerely,

WAYNE SCARANO BGG00303

43 Katahdin Avenue Millinocket, ME 04462

VERSAR, INC. CO 77 ian 5 0 1995 ECOLOGICAL SCIENCES 1-21-95 To: Low B. Casher Re: (FERC) NO. 2458 - NO. 2572" Will CONTRACT ON THE CONTRACT CONTRACT ON THE OFFICE AND STREET To: Low B. Cashell FILED THE DECRETARY the new proposalo re: the afford • • • • I own a campon South Twin Lake within 200 ft of the high. vater make mark. I unge Fere to elimite the pro posed building set back and proposed vegetative buffer tor instance one example of a campon an island without 200 ft Space on either side of Camp to the high water mark. What happens here? If allowed an island camp to remain as is . Why not the others? Sincerely, Charles a. Shorten P.S. I lease Lot #281 Plan B-573 #2058 my address 166 Blake Rd Stanchisk M.e. 04084



Raymond Sikoski 6 Hillside Road North Hampton NH 03862 January 29, 1995

OFFICE OF THE SECPETARY 95 FEB 10 AM 9: 43 AL ENERGY

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - UUC

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

- Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

- Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

FILED OFFICE OF THE SECRETARY



95 FEB - 9 AM 10: 18

LICAL ENERGY DESULATIONY MMISSION Jan. 31, 1995

Ms. Lois D. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington , D.C. 20426

Dear Ms. Cashell, 505

The purpose of my Letter is to address and comment on the relicensing of Projects Nos. 2572 (Ripogenus) and 2458 (Penobscot Mills) located in Maine, owned and operated by Bowater, Great Northern Paper, Inc.

As an employee of Great Northern Paper, I am supportive to the relicensing of these two projects and hope that the license will be issued very soon.

The DEIS addresses many concerns, areas, public and private issues. Having read the draft, I have several concerns also. The following statements pertain to some of which were mentioned in the DEIS.

My residence lies in an unorganized township, Indian #3. One of the many areas LURC has regulated very professionally and strictly. LURC has done an excellent job, why change the responsibility over to another agency? Think of the costs in terms of federal dollars lost to all taxpayers. These monies could support other programs needing more attention. Consider the hardships imposed on the many citizens because of new government regulations.

Changing the regulations on future developments creates major problems for landowners, regardless of who they are.

I do not agree with the requirement imposed on Bowater to aquire lands, 67 miles, belonging to public and private owners along the impoundments. The capital expenses here, to aquire these areas, should instead be spent on repairs to the mills which are badly needed.

After attending the public hearing, I was amazed to see such a small turn out on the opposition's side. Special interest groups have hired representatives to address their concerns. A handful attended the public meeting held in Millinocket on Jan. 25, 1995. Their comments presented at the public meeting seem to have made a great impact regarding the commissions decisions on land use regulations in the future; way out of proportion to what they have at stake or their numbers.

This area is facing a major change. Our community and surrounding areas are counting on this license and having Bowater continue capital investments. Outside factions are imposing unwanted restrictions in how the area is to be used. We plain folk want to continue living in this area without unnecessary changes. I can understand the need to preserve and protect. We, living in this area take pride to what is here and will protect the area in which we live!

PLEASE, for the PEOPLE, (first and foremost) living and working in this area reconsider these sensitive issues raised at the public hearing and ones addressed in this letter.

Thank you.

Sincerely,

Cardyn J. Simme

Carolyn J. Simone



Doc Simmons 8 Exeter Road - Box 550 North Hampton NH 03862 January 29, 1995

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572-005

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

- Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely,

Doc Simmons -ridd Devennon DV M.

SECRETARY



Thomas Simmons 8 Exeter Road North Hampton NH 03862 January 29, 1995

· ** ----UFFICE OF FILED THE SECRETARY 95 FEB -8 AN 11: 57 FEDERAL ENERGY COMMISSION

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572-005

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerel 21/1 Thomas Simmons





Fred & Marianne Smith 106 Euclid Avenue Massapequa, Long Island NY 11758 January 29, 1995

⁵FEB-7 AMII: 36

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 🦳

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

- Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely,

Fred & Marianne Smith WHILE

ř 🖌 🗁 🍸

IVIC

FEB 2 4 955

ECOLOGICAL SCIENCES AND ANAI

Richard and Jean Smith 6 Greystone M.H. Park Veamie ME 04401-7059VERSAR. January 29, 1995

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - 00°

Dear Secretary, FERC;

11

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

- Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

- How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Inuth Sincerely, C

Richard and Jean Smith



Sandra Smith 263 State Road Eliot ME 03903 January 29, 1995

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 - 005

Dear Secretary, FERC;

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerel andra Smith

后 SECRETARY

2

ŝ

95 FEB 27 PH 12: 00

FELERAL ENERGY RLGULATORY 10 Maine Avenue COMMISSION Millinocket, Maine, 04462 Feb. 20, 1995

Federal Energy Pegulation Commission

Gentlemen: re: Penobscot Piver and Rirosenus Dam Water Use

I attended both of the hearings in Millinocket, Maine. As a life- onresident of Millinocket, I am much concerned both for the present and for the future. Whath r it affects me or my grandchildren.

My family has had a cottage (we conversely call camp) on Ambajejus wake on a small hill overlooking the lake and three islets. I now an area 72, an lar blind, and have rheumatoid arthritis; I am thankful that I am close enough to the shore to be able to see as such as I am able to see, that I am able to set down to the shore in my condition. If I had to put a comp 500 feet away from the shore with specified version between me and that shore, then 4 would not be close shough to see the lake and I would be forever trying to headble or or from my camp to the shore through the version. So, wen if 4 an areadfathered by the FEEC proposed regulation, I am concerned low

others in future without erondfathering; for literwand infire surgers, you would be provided in fire surgers, you would be provided in the fire of a stars and vides: if it are to be the stars and vides: if it are to be the stars and vides: if it are to be the stars and vides: if it are to be the stars and vides: if it are to be the stars and vides: if it are to be to sell my the stars and vides: if it are to be to sell my the stars and vides: if it are to be to be the stars and vides: if it are to be to sell my the stars and vides: if it are to be to sell my the stars are sell to just the stars are sell on a context that the set of the from the stars are to be a stars and to be a friend of the stars are to provide that to be a friend of the set of the set of the stars are to be a stars and to be a friend of the set of the se

Turthermore, tourism and terrestion is important to us in frine Your crossel has unforming to tourism and to represtion, and that is unfrien by an fairs morele and to show who desire to visit us from other as ss. These lat us do ove own revening in such metters we do not not federal-very lation is such metters. Lat us be driends, Woine and you.

You this State Stand

Cercego,



Dear Madam:

Please be advised that as a tax payer and a leased land holder on Ambajejus Lake, Millinocket, Maine, I see no reason to expand the project boundaries and are not justified by the record of the case.

I am urging FERC to eliminate the proposed building setback and proposed vegetative buffer.

> Thomas and Patricia E. St. John 9 Orchard Street Millino ket, Maine 04462

Thomas W. It John F Patricia &, St. John





ECOLOGICAL SCIENCES AND ANALYSIS

Mir and Mrs. Robert Van Deventer 45 Grenadier Lane East Islip, N.Y. 11730 January 23, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N. E. Washington, D.C. 20426

RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458)

Dear Lois Cashell,

I am a lease holder with the Great Northern Paper Co. on property within the Ripogenus Impoundment. My family has held this lease for over 5 years. We hold permanent residence in New York State and utilize the property in Maine on a seasonal basis. We hope some day to be able to purchase this land. Our camp was built, on this lease property, within the strict guidelines of the Maine Land Use Regulatory Commission (LURC). The guidelines provided by LURC, were developed to meet all environmental needs and concerns. I urge no further restrictions placed on this land use by the Federal Energy Regulatory Commission (FERC). Any further restrictions are unwarranted and unnecessary. I am in full support of Great Northern's proposal to add no additional restrictions on existing leases within the impoundment. Please consider the vast number of people you will affect with the proposal's that are being considered by FERC. Please take into consideration less government regulation and recognize Maine's LURC guidelines as adequate for building setbacks and vegetative buffers within watershed areas. Thank you.

Sincerely.... 1-1

Robert Van Deventer and Family

cc Dean A. Beaupain Attorney for Town of Millinocket



RE: Ripogenus Hydroelectric Project (FERC No. 2572) Penobscot Mills Hydroelectric Project (FERC No. 2458)

Dear Lois Cashell,

I am a lease holder with the Great Northern Paper Co. on property within the Ripogenus Impoundment. My family has held this lease for over 10 years. We hold permanent residence in New York State and utilize the property in Maine on a seasonal basis. We hope some day to be able to purchase this land. Our camp was built, on this lease property, within the strict guidelines of the Maine Land Use Regulatory Commission (LURC). The guidelines provided by LURC, were developed to meet all environmental needs and concerns. I urge no further restrictions placed on this land use by the Federal Energy Regulatory Commission (FERC). Any further restrictions are unwarranted and unnecessary. I am in full support of Great Northern's proposal to add no additional restrictions on existing leases within the impoundment. Please consider the vast number of people you will affect with the proposal's that are being considered by FERC. Please take into consideration less government regulation and recognize Maine's LURC guidelines as adequate for building setbacks and vegetative buffers within watershed areas. Thank you.

Sincerely,

; /

Parker 1/2 S

Glen H. Van Deventer and Family

GHV/ghv cc: Dean A. Beaupain Attorney for Town of Millinocket



Robert Wells 70 Walnut Ave North Hampton NH 03862 January 29, 1995

UFFICE OF THE SECRETA 95 FEB -8 PH 12: 0 FEDERAL ENERG REGULATORY COMMISSION

· · ·

Secretary Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572 -005

Dear Secretary, FERC;

. .

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

• Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

• Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

• Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

• How will the value of property be affected by the proposed setbacks and vegetative buffers?

• Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

• Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely,

Robert Wells

Robert al all:
FILED OFFICE OF THE SECRETARY

95 FEB 27 AM 11: 18

February 16, 1995

FEDERAL ENERGY REGULATORY COMMISSION

009

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

RE: Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 _ 004

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

uley littleton

1

١

Perley Wheaton, President

Millinocket Ford-Mercury, Inc. 1009 Central Street Millinocket, ME 04462

E-617

Richard J. Wheaton Jr. 398 So. Main St. Old Town, ME 04468

January 31, 1995

Lois B. Chashell, Secretary Federal Energy Regulator Commission 825 North Capitol Street, N. E. Washington D. C. 20426

Re Great Northern Paper. Inc. (FERC No. 2572) 005 **Ripogenus Hydroelectric Project** (FERC No. 2458) - 00 9 Penobscot Mills Hydroelectric Project

Dear Ms. Cachell,

As a typical member of the silent majority this is only the second time I have been compelled to speak. The first was a year ago with regard to the James River Land Fill.

COP

ND AMALYSIS

I work at the James River mill in Old Town, Maine, where we have spent \$2,000,000, six years and produced five volumes of paper, in an effort to procure a license for a land fill.

Now, Great Northern Paper, Inc. is in the process of relicensing their dams, Ripogenus Hydroelectric Project (FERC No. 2572) and Penobscot Mills Hydroelectric Project (FERC No. 2458).

While growing up in Millinocket, we lived six months of the year on Ambajesus Lake and I have owned a camp on Millinocket Lake for the last 15 years. It is my hope to spend my summer retirement there.

The Draft Environmental Impact Statement suggest the need for a 500 foot building setback and 250 foot vegetative buffer.

I do not understand how the relicensing of the existing dams will cause further degradation of the waterways. The dams and waterways are there and will not change with a new license. In fact, if the dams weren't there, much of the waterway we seek to protect would not exist. Development around these waterways is a completely separate issue and should not be a condition of relicensing.

This would seem to me to be a ploy of special interest groups to impose unreasonable and possibly unnecessary restrictions, using the relicensing process as a lever. This misuse of well intended processes needs to stop. Industry in Maine, and the country cannot compete in this rapidly shrinking world if we continue to impose costly, drawn out processes with

unreasonable expectations and often minimal or unmeasurable impact. Of course there are those who would have Maine become a pure Vacation Land. This provides them a place to escape their fast paced, economical environment. Unfortunately, for those of us who live in Maine and make their living here, it promotes fewer and fewer opportunities, and Millinocket can certainly attest to that.

Much of the industry that existed in Maine is gone and what is left has downsized significantly. A salesman that comes to the mill told me he doesn't make as many calls as he used to, many of the businesses he called on, just aren't there any more". One does not have to travel very to find falling down buildings that represent an industry that has become extinct. With this extinction went the jobs that the industry provided. If there was an endangered specious list for industries, many Maine industries are showing the signs that would place them on the list.

I do not wish to destroy the environment in which I hope to retire, but I have seen the water quality of these waterways improve in the thirty some years that I have hunted and fished them. I have also seen many jobs disappear. We must do something to improve the ability of industry survive in this state.

LURC has many restrictions that are intended to protect the quality of Maine's waterways. The limit on the number of leases that may be issued makes it almost impossible to get a new lease for the purposes of development. Are the present restrictions on development in these areas not working? Will the proposed restrictions do anything more than current restrictions? If there are any gains, will they be of sufficient significance to justify the restrictions? Will the gains be measurable or even be noticeable?

Somewhere, someone has to apply some basic logic and good common sense! Are the purposed restriction of value? Will there be sufficient gains to justify the cost or impact? What will be the impact on industry and the general public? I'm sure that there are many more basic questions that should be considered before allowing the vocal minorities and special interest groups dictate the destiny of the silent majority.

It is time for the silent majority to be heard, before the special interest groups force us to extinction!

Cordially,

Rulland June herry Richard J. Wheaton Jr.

E-619

February 16, 1995

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

Penobscot Mills Project, FERC No. 2458 RE: Ripogenus Project, FERC No. 2572 - 005

Dear Ms. Cashell:

This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

CHENTARY

95 BR -2 11 3-55

FEDERAL ENERGY

OFFICE OF

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

John Olerk Guatice Payer

Jopy David Whorf 4 Tanya Lane VERSAR, INC E SECRETARY Falmouth ME 04105 : As January 29, 1995 ġ. FEB 2 4 1995 . Secretary ECOLOGICAL SCIENCES AND ADA Federal Energy Regulatory Commission 825 North Capitol Street NE Washington DC 20426

Re: Ripogenus Hydroelectric Project, Maine, Project No. 2572-005

Dear Secretary, FERC;

.

.....

I (We) support the relicensing of Bowater's hydroelectric project under the terms of Bowater's original application, and do not support either alternative proposed by FERC.

I (We) further demand that FERC answer the following questions about its proposed alternatives to Bowater's application before it takes further action.

· Will land owners be allowed to maintain existing buildings and docks within the proposed setback zones?

· Will land owners be allowed to construct any new buildings or docks within the proposed setback zones?

- Will land owners be allowed to remove any vegetation within the proposed vegetative buffers?

- How will the value of property be affected by the proposed setbacks and vegetative buffers?

- Will any grandfathering in regard to new regulations be transferable to next of kin, or on upon sale of the property?

- Will relicensing opponents and federal agencies convince FERC to increase the proposed building setback and vegetative buffer in proposal 2 to 500 feet, and 250 feet respectively?

• Will land owners have to deal with both LURC and FERC concerning future changes to their property?

A written response to these questions may be sent to the above address.

Sincerely, David Whorf

Hutchinson Street 27 New Britain, Connecticut 06053

January 26th, 1995



Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, D.C. 20426

Dear Ms. Cashell:

002 As the owner of a camp on North Twin Lake in Millinocket, Maine I am writing in support of the applications of Great Northern, Inc. for renewal of their thirty year license for the Ripogenus Hydroelectric Project (FERC No. 2572) and the Penobscot Mills Hydroelectric Project (FERC No. 2458). -00°

Our family goes back many generations in the area encompassed by the Penobscot Project. Additionally, we have owned a camp on land leased from Great Northern on North Twin Lake for the last sixty years. During this time we have seen a good situation environmentally decline and then improve greatly after the establishment of Maine's Land Use Regulatory Commission in 1981. There have been far reaching effects from this point forward. Logging practices have been greatly improved, our beautiful lakes are in pristine condition, and most recently we have observed great improvement in the quantity and diversification of fish and water fowl populations. We have never in our lengthy lifetimes seen the area in better condition than it is This pleases us greatly. now.

That this is the case is in no small way due to the fact that LURC - Land Use Regulation Commission - serving approximately two thirds of the state of Maine, has already set and enforced on its own with no Federal suggestion solid environmental standards for the area in which both FERC No. 2572 and FERC No. 2458 fall. These LURC standards are tough, environmentally sound, working well, fair to area residents and visitors alike, and already paid for by the state of Maine and its residents.

If the Draft Environmental Impact Statement were so accurate, and tighter environmental controls needed, why are residents seeing continuous, steady environmental improvement such as that which we outlined? Perhaps the excellent controls already FERC Nos. 2572 and 2458

in place in the state of Maine are not always in evidence in " other areas serviced by PERC. But, for us in Maine your attempt to over regulate leads me to think of this as unneccesary Federal intervention. This seriously interferes with the individual rights of many people and the autonomy of an entire area of already environmentally conscious people.

Once again, we reiterate our total support of the renewal applications of Great Northern, Inc. as submitted and hope you will in wisdom leave matters affecting surrounding property in these areas encompassed by both projects up to the state of Maine and their Land Use Regulatory Commission which appears to already be doing a more than adequate job of regulating their own environment.

Respectfully,

Betty Trott Tessman 27 Hutchinson St. New Britain, CT 06053

Camp Owner Millinocket, ME

That lessman

Sandra Trott Davenport 12 Eton Place New Britain, CT 06051

Camp Owner Millinocket, ME

Sandra hatt Lever at

95 FEB 28 PH 1: LL

February 16, 1995

FELERAL ENERGY COMMISSION

Lois B. Cashell, Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington D.C. 20426

LagRE --- Penobscot Mills Project, FERC No. 2458 Ripogenus Project, FERC No. 2572 __ 006

Dear Ms. Cashell:

.

n an an Arrange an Arr Arrange an A This letter is written in support of Great Northern's applications for new licenses for its Ripogenus and Penobscot Mills Projects.

 \mathbf{T}_{i}

I have reviewed the Draft Environmental Impact Statement and generally agree with its conclusions but strongly disagree that project boundaries must be expanded by means of a conservation easement or by prohibiting all future development.

Land use of property around the impoundments is presently regulated by Maine's Land Use Regulation Commission which is well able to discharge its duties and is largely responsible, along with the landowners, for conditions as they presently exist.

I object to a Washington bureaucracy dictating to the State of Maine and local residents how land use decisions should be made. This State has done a good job to date and the residents of this State are perfectly capable of making our own decisions.

Under no circumstances should anyone be forced to sell their land.

Very truly yours,

TECHAICIAN

FREEDENT

CARLE A Stedical OFFICE MARNEER

Srint Rollins TECHNICIAN

Cit members & Xayle Filler & Mating Suc.